LYAPUNOV APPROACH ON A HOMOGENEOUS FAMILY OF Controllers for Robotic Manipulator

Jang Wilson

Thesis submitted to the University of Nottingham for the degree of Doctor of Philosophy

February 2019

ABSTRACT

Second order sliding mode has been successfully implemented for solution of real problems for its inherent features such as finite-time convergence and robustness to disturbances. For the first order sliding modes, it is common to deal with the issues of stability, robustness, and convergence rate of the equilibrium by means of a Lyapunov approach. For higher order sliding modes, however, a similar treatment has not been developed until recently. The focus of this thesis is the construction of strong Lyapunov function, i.e. for which its time derivative can be upper bounded by negative-definite functions, for the design of control strategies for robotic manipulator, which is a nonlinear system, subject to combinations of parametric uncertainty, bounded disturbances, actuator saturation, and output feedback.

The first contribution of this work is the development of a strong Lyapunov function for a parameterized family of homogeneous sliding mode based controller comprising twisting algorithm, continuous finite time control, linear PD control law, and uniformly stable control law, all of which belongs to a general homogeneous family of control algorithms. The strict locally Lipschitz homogeneous Lyapunov function proposed permits the estimation of convergence time for the trajectories of the system to the origin, in finite-time, exponentially, or uniformly asymptotically, even in the case when it is affected by bounded non-vanishing or growth bounded vanishing external perturbations. Moreover, the relationship between the control gains and its convergence performance can be analyzed.

Leveraging on these results, a strong Lyapunov function is developed for a closely related second order sliding mode algorithm, the super-twisting algorithm based controller. In particular, the construction of these strong homogeneous Lyapunov function is able to show

the relationship between the twisting and super-twisting algorithms and allows linear combination of two homogeneous control of different degree.

Extending the results for MIMO robot manipulator, a type of Euler-Lagrange dynamic systems, a family of integral sliding mode-based controller is introduced for trajectory tracking. In particular, the homogeneous dynamics is employed as the desired error dynamics for the controller. Additionally, the conventional PID control is shown to be a special case and the present formulation presents the relationship between the gains of the controller and the desired performance, which provides a systematic method for gain selection for a robust PID control. In addition, for the special problem of regulation, employing the results of homogeneous control, finite-time regulation of the robot manipulator is achieved.

Since actuator saturation is a phenomenon that affects the performance of dynamic systems under closed-loop control, a saturated version of the controller is also developed that achieved global stability while maintaining the features of the unbounded version of the controller in terms of trajectory tracking and finite time regulation. Extending the results for system with position measurements only, a saturated output feedback version of the controller is introduced that can achieved global stability as well. Each of the proposed controllers provides advantages over the previous literature in their ability to design desired error dynamics and the time derivative of the disturbance is not required in the stability analyses.

Throughout the work, Lyapunov-based stability, in particular the nonsmooth Lyapunov analysis techniques, and numerical experiments are provided to highlight the performance of each controller design.

ACKNOWLEDGEMENTS

I would like to express sincere gratitude to my supervisor, Dr. Haider Abbas, whose motivation and experience have been immense in my life as a researcher. As an advisor, he has provided guidance in my work and encouragement in developing some new ideas. During our regular meetings, it is always exciting in our exchange of ideas and I am always left with new inspiration on the direction of my research after the meeting. I would like to extend my appreciation to all the faculty members for their kindness and understanding. Particularly, I would like to extend my gratitude to the university for providing the scholarships for my PhD studies. Lastly, I would like to thank my family for their continuous support and care.

TABLE OF CONTENTS

ABSTRACT	
ACKNOWLEDGEMENTS	4
LIST OF FIGURES	
CHAPTER 1: INTRODUCTION	
1.1 MOTIVATION	
1.2 LITERATURE REVIEW	
1.3 PROBLEM STATEMENT	
1.4 Contributions	
CHAPTER 2: LYAPUNOV APPROACH ON TWISTIN	G AND SUPER-TWISTING
BASED SECOND ORDER SLIDING MODE	
2.1 Preliminaries	
2.2 TWISTING ALGORITHM	54
2.2.1 System description	
2.2.2 Stability analysis	
2.3 GENERIC TWISTING	
2.3 GENERIC TWISTING2.3.1 System description	
2.3 GENERIC TWISTING2.3.1 System description2.3.2 Stability analysis	

2.4.1 System description	
2.4.2 Stability analysis	
2.5 GENERIC SUPER-TWISTING	
2.5.1 System description	
2.5.2 Stability analysis	
2.6 GENERIC SECOND ORDER ALGORITHM	
2.6.1 System description	
2.6.2 Stability analysis	
2.7 NUMERICAL SIMULATIONS	
2.7.1 Simulation Setup	
2.7.2 Results and Discussions	
2.8 SUMMARY	
CHAPTER 3: ROBOT MANIPULATOR CONTROL	: FULL STATE FEEDBACK
APPROACH	
3.1 System Description	
3.1.1 Manipulator Dynamics	
3.1.2 Control Development	
3.2 STABILITY ANALYSIS	
3.2.1 Construction of Lyapunov Function	

3.2.2 Stability Criterion Determination	
3.3 NUMERICAL SIMULATIONS	
3.3.1 Simulation Setup	196
3.3.2 Results and Discussions	197
3.4 SUMMARY	
CHAPTER 4: ROBOT MANIPULATOR CONTROL: OUTPUT	FEEDBACK
APPROACH	207
4.1 Observer Dynamics	
4.1.1 System Description	
4.1.2 Stability Analysis	
4.2 MANIPULATOR DYNAMICS	
4.2.1 Control Development	
4.2.2 Stability Analysis	
4.3 NUMERICAL SIMULATIONS	
4.3.1 Simulation Setup	
4.3.2 Results and Discussions	
4.4 Summary	
CHAPTER 5: ROBOT MANIPULATOR CONTROL: BOUNDED	CONTROL
APPROACH WITH FULL-STATE FEEDBACK	

5.1 Nonlinear Robot Dynamics	
5.1.1 Control Development	
5.1.2 Stability Analysis	
5.2 NUMERICAL SIMULATIONS	
5.2.1 Simulation Setup	
5.2.2 Results and Discussions	
5.3 SUMMARY	
CHAPTER 6: ROBOT MANIPULATOR CONTROL: OUTPUT	FEEDBACK
BOUNDED CONTROL APPROACH	
6.1 Observer Dynamics	
6.1.1 System Description	
6.1.2 Stability Analysis	
6.2 MANIPULATOR DYNAMICS	
6.2.1 Control Development	
6.2.2 Stability Analysis	
6.3 NUMERICAL SIMULATIONS	
6.3.1 Simulation Setups	
6.3.2 Results and Discussions	
6.4 Summary	

CHAP	TER 7: CONCLUSION AND FUTURE WORK	
7.1 I	DISSERTATION SUMMARY	
7.2 I	JIMITATIONS AND FUTURE WORK	
APPE	NDIX A	
A.1	PROOF OF DESIRED DYNAMICS (CH3 & 4)	
A.2	PROPOSITION 1	
A.3	PROPOSITION 2	
APPE	NDIX B	
B.1	PROOF OF DESIRED ERROR DYNAMICS (CH 5 &6)	
B.2	PROPOSITION 1	
REFE	RENCES	

LIST OF FIGURES

Figure 2.1. Twisting based algorithm. States of (2 - 6) with $k_1 = 1$, $k_2 = 2$, and three different values of $b = \{0.6, 0.7, 1.2\}$.

Figure 2.2 Twisting based algorithm. Disturbance $d = |z_2|^{\frac{2b}{1+b}} \operatorname{sign}(z_1) \cdot \operatorname{sin}(t)$ for $b = \{0.6, 0.7, 1.2\}.$

Figure 2.3 Twisting based algorithm. States of (2 - 6) for b = 0 with $k_1 = 4$, and $k_2 = 2$ under persistent disturbance d = 1.

Figure 2.4 Super-twisting based algorithm. States of (2 - 17) with $k_3 = 2$, $k_1 = 0.3$, and three different values of $p = \{0.5, 0.7, 1.2\}$.

Figure 2.5 Super-twisting based algorithm. Disturbance $d_1 = |z_1|^p \operatorname{sign}(z_2) \cdot \operatorname{sin}(t)$ for $p = \{0.5, 0.7, 1.2\}.$

Figure 2.6 Super-twisting based algorithm. States of (2 - 17) for p = 0.5 with $k_3 = 1$ and $k_1 = 1$ without perturbation.

Figure 2.7 Generic super-twisting based algorithm. States of (2 - 23) with

$$k_1 = k_{1n} = 4$$
, and $k_3 = k_{3n} = 2$ under three values of initial conditions
 $(z_1(0), z_2(0)) = \{(-0.1, -0.1), (-1, -1), (-10, -10)\}$.

Figure 2.8 Generic super-twisting based algorithm. Disturbance

$$d_{1} = \operatorname{sign}(z_{2}) \cdot \sin(t) ||z_{1}|^{p} + |z_{1}|^{q} \text{ and } d_{2} = \operatorname{sign}(z_{2}) \cdot \sin(t) ||z_{1}|^{2p-1} + |z_{1}|^{2q-1} \text{ for three}$$

$$157$$
different values of initial conditions $(z_{1}(0), z_{2}(0)) = \{(-0.1, -0.1), (-1, -1), (-10, -10)\}.$

Figure 2.9 Generic twisting based algorithm. States of (2 - 11) with $p = \frac{1}{1.4}, q = \frac{1}{0.4},$ $k_1 = k_{1n} = 4$, and $k_2 = k_{2n} = 2$ in (a)-(b). (c)Disturbance $d = \operatorname{sign}(z_2) \cdot \sin(t) \cdot \left(|z_1|^{2p-1} + |z_1|^{2q-1}\right) + \operatorname{sign}(z_1) \cdot \sin(t) \cdot \left(|z_2|^{2-\frac{1}{p}} + |z_2|^{2-\frac{1}{q}}\right).$ 158

Figure 2.10 Generic second order based algorithm. States of (2 - 29) with

$$p = \frac{1}{1.4}, q = \frac{1}{0.4}, \quad k_1 = k_{1n} = k_2 = k_{2n} = 4, \text{ and } k_3 = k_{3n} = 2 \text{ in (a)-(b). (c)Disturbances}$$

$$d_1 = \operatorname{sign}(z_2) \cdot \operatorname{sin}(t) (|z_1|^p + |z_1|^q) \text{ and}$$

$$d_2 = \operatorname{sign}(z_2) \cdot \operatorname{sin}(t) (|z_1|^{2p-1} + |z_1|^{2q-1}) + \operatorname{sign}(z_1) \cdot \operatorname{sin}(t) (|z_2|^{2-\frac{1}{p}} + |z_2|^{2-\frac{1}{q}}).$$
(159)

Figure 3.1 Simulation 1. Tracking errors using control (3 - 2) for three different values of $a = \{0.7, 0.8, 0.9\}$.

Figure 3.2 Simulation 1. Control input of joint 1 and joint 2.	200
Figure 3.3 Simulation 2. Tracking errors using control $(3 - 2)$ with $a = 0.9$ with three different values of control gains <i>K</i> .	201
Figure 3.4 Simulation 2.Control input of joint 1 and joint 2.	202
Figure 4.1 Simulation 1. Tracking errors and control input of joint 1 and joint 2.	271
Figure 4.2 Simulation 1. Position observation errors using observer (4 - 6) of joint 1 and joint 2.	272
Figure 4.3 Simulation 1. Velocity observation errors using observer (4 - 6) of joint 1 and joint 2.	273
Figure 4.4 Simulation 2. Observation errors without L_3 , small L_1 and L_2 .	274

Figure 4.5 Simulation 2. Observation errors without L_3 , large L_1 and L_2 .	275
Figure 4.6 Simulation 3. Observation errors with $p = 0.5$.	276
Figure 4.7 Simulation 3. Tracking errors using control $(3 - 2)$, control $(4 - 4)$ with $p = 0.5$, and control input of joint 1 and 2.	277
Figure 5.1 Simulation 1. Tracking errors, sigma, and control input of joint 1 and joint 2.	303
Figure 5.2 Simulation 2. Tracking errors using control (3 - 2), control (5 - 2).	304
Figure 6.1 Simulation 1. Tracking errors, sigma and control input of joint 1 and joint 2.	352
Figure 6.2 Simulation 1. Observation errors for joint 1 and joint 2.	353
Figure 6.3 Simulation 2. Tracking errors using control (4 - 4), control (6 - 3).	354

Chapter 1: INTRODUCTION

1.1 Motivation

Sliding mode approach to control design has received great amount of attentions of late due to its inherent attractive features such as finite-time convergence and robustness to disturbances. It introduces a nonlinear discontinuous term where its gain must be designed so that the trajectories of the system are forced to remain on some user-defined sliding hyperplane in the error state space. The resulting motion is called sliding mode. It is this discontinuous term that provides the abilities to reject perturbations and some classes of uncertainties between the actual system and the nominal model used in the control design stages. In [1], a definition of the sliding mode order is given and consists of trajectories in the sense of Filippov [2]. The standard sliding mode is of the first order and is known as robust and very accurate with respect to various classes of internal and external perturbations, but it is restricted to the case in which the output relative degree is one. Besides, the high frequency switching that produces the sliding mode may cause chattering effect.

Higher order sliding modes (HOSM) appears sometimes in systems with traditional sliding mode control or they are deliberately introduced because it has been found that finite time convergent HOSMs preserve the features of the first order sliding modes and can improve them, if properly designed, by eliminating the chattering, for instance see [3] and [4]. While finite-time convergent arbitrary order sliding mode controllers are mostly still theoretically studied, 2-sliding controller or second order sliding mode(SOSM) with finite-time convergence have already been successfully implemented for solution of real problems.

However, there are only few SOSM that are widely used namely, the sub-optimal controller [3], [5], [6], the terminal sliding mode controllers [7], [8], [9], twisting controller, and the super-twisting controller; the last two being the most popular.

Particularly, the super-twisting control found its application on wind energy conversion system [10], velocity observer of mechanical systems [11], and uncertainty observers [12] [13]. On the other hand, application of twisting algorithm can be found in the adaptive tracking control of intelligent vehicle system [14] and trajectory tracking of crane [15] and [16]. Also of interest, are the development of family of controllers that are based upon the twisting and super-twisting algorithms [17] and [18]. For the first order sliding modes, it is common to deal with the issues of stability, robustness, and convergence rate of the equilibrium by means of a Lyapunov approach. For higher order sliding modes, however, a similar treatment has not been developed until recently. Instead it is usual to use majorant curves [11], homogeneity based methods [19], or a weak Lyapunov function together with geometric approach [20]. The focus of Chapter 2 is the development of strict Lyapunov functions for the super-twisting, twisting algorithms, and the corresponding family of controllers, which can be used as design and analysis tool whose time derivative can be bounded by negative definite functions.

The research on the control theory of serial mechanical systems has been a topic that is actively studied. The asymptotic stability of robot manipulators can be achieved by computed torque method or inverse-dynamics control [21]. While asymptotic stability implies that the system have convergence to the origin as time goes to infinity, finite-time stabilization can ensure convergence to the origin in finite time, as discussed above. On the other hand, robustness in control systems is an equally important property as stability and convergence. It is the property by which a system preserves a tolerable behaviour under the influence of uncertainty, perturbations, external disturbances, etc. In particular, it is known that finite-time stabilization of dynamical systems can provide high-precision performance and improved rejection of low-level persistent disturbances [22] [23]. This can be achieved by continuous non-Lipschitz feedback controllers such as twisting-based algorithm in [24]. However, the robustness issue is not specified clearly as the algorithms require exact knowledge of the dynamics of the manipulators. Leveraging the outcomes developed in Chapter 2, Chapter 3 presents full state feedback approach by integrating SOSM into the controller.

The previous discussion on robot manipulator control assumes that the joint velocity is available from measurement. If only position information is available, one has to employ output feedback control, which has received considerable interest in robotics literature due to its possibility to avoid the need of a tachometer hence simplifying the robot design [12]. The main problem in output feedback control is the need for the control law to not only compensate uncertainties of the system but also the lack of link joint velocity measurements. Employing the outcomes of Chapter 2 and 3, Chapter 4 presents an output feedback controller for robot manipulator.

While the robust control methods mentioned above for robot manipulators have been shown to be effective for the compensation of uncertainties and disturbance in their respective context, generally, the fact that the required input torques may command more actuation than is physically possible by the system for instances such as large perturbations, initial conditions that are far from the equilibriums, or fast desired trajectory. These may lead to unexpected or undesirable closed-loop behaviours, for instance, in the nonlinear PID control for the global regulation problem of robot manipulator of [25], it requires unbounded state dependent control gains, which may easily causes saturation of actuator if the initial conditions of the system is not restricted. Owing to these risks, control law that is bounded while ensuring performance when operating within actuator limits are motivated. By means of the results of Chapter 2 and 3, Chapter 5 presents a bounded full state feedback controller for robot manipulator which limits the control authority at or below an adjustable a priori limit. Bounded control designs are available in literature; however, the integration of SOSM into the bounded structure that has a strict Lyapunov function has remained an open problem.

Motivated by the same concerns presented in Chapter 4 on the lack of joint velocity measurements and that of actuator constraints in Chapter 5, Chapter 6 develops a control strategy for robot manipulators with a bounded control approach with output feedback. Previous techniques and outcomes obtained in Chapter 4 and 5 are utilized which allows for the bound on the control to be adjusted a priori provided through strict Lyapunov functions.

1.2 Literature Review

A literature review of Chapters 2-6 is presented below.

Chapter 2: Lyapunov approach on twisting and super-twisting based second order sliding mode: By means of strong Lyapunov functions, the study of stability on super-twisting algorithm and its finite time convergent characteristics was carried out for the first time by [26] and later in [27]. This approach allows a wider class of perturbations and uncertainties originally admitted by SOSM. Another advantage of the use of Lyapunov functions is that it is possible to obtain explicit relations for the design parameters. On the other hand, [17] developed a Lyapunov method for the analysis of a generic second order algorithm, which is a family of controllers of which the super-twisting algorithm is a special case. It extends the results of [26] by allowing the positive power of the control terms to range from less than one to more than one. By doing so, a range of stability results were obtained. Essentially, three types of stability can be achieved for the generic super-twisting system namely, finite-time, exponential, and uniform convergence. A remarkable fact from the approach is that the stability of the equilibrium of the system is completely determined by the stability of its associated linear counterpart. Additionally, non-homogeneous super-twisting algorithms have been studied using strict Lyapunov functions as well [28].

A closely related algorithm, which is equivalently important, is the twisting algorithm. The twisting algorithm [29] is given by

$$\dot{z}_1 = z_2, \ \dot{z}_2 = -k_1 sign(z_1) - k_2 sign(z_2)$$

where z_1 and $z_2 \in \mathbb{R}$ are scalar state variables, $k_1 > 0$ and $k_2 > 0$ are control parameters.

Similar to that of super-twisting, it is common to find its stability analyses using the homogeneity approach [1], [20], [30], [31] or a weak Lyapunov function together with geometric approach [32]. The algorithm is globally uniformly finite time stable if the inequality $k_1 > k_2 > 0$ is satisfied [20]. Of recent, there is a growing interest in identifying strict Lyapunov function for this algorithm. A strict Lyapunov design and estimation for reaching time, based on Zubov method, is presented in [33]. It employs the idea of using the solution of a partial differential equation as the Lyapunov function. For design control purposes, this methodology becomes difficult. Indeed it requires some handicraft techniques, like fixing the discontinuities. In [34], a strict non smooth Lyapunov function is proposed for the twisting algorithm. The strictness of this function allows estimation of the convergence time of the closed loop system to the origin. While the in work of [35], an alternative proof is established with the same Lyapunov function as in [34]. It provides additional properties that are not shown in [34] as well as a simple rule of thumb pertaining to the relationship between the control gains and the finite settling time of the system.

Similar to the super-twisting algorithm, which has a generic second order algorithm developed based on it [17], for twisting algorithm, in [18], a parameterized family of homogeneous continuous controllers, inspired from the twisting algorithm, is proposed

$$\dot{z}_1 = z_2, \, \dot{z}_2 = -k_1 |z_1|^{\frac{\alpha}{2-\alpha}} \operatorname{sign}(z_1) - k_2 |z_2|^{\alpha} \operatorname{sign}(z_2) + \delta, \quad (1-1)$$

where $\alpha \in [0,1)$, δ is an external disturbance, and $k_1, k_2 > 0$. Note that the twisting algorithm is a special case of this generalized controller. This continuous controller with unity gains was first proposed by [36] to develop a class of continuous second order finite time systems. However, the approach employed there for finite time stability does not allow an upper bound of the settling time to be obtained.

Pertaining to the approaches mentioned above for the twisting algorithm, they cannot be applied directly for the parameterized family of homogeneous continuous controllers (1 - 1) that includes the twisting algorithm as its special case. In [18], for $\alpha \in (0, 1)$, the use of a weak Lyapunov function with invariance principle, only global asymptotic stability can be guaranteed. It requires application of Theorem 4.2 of [20], which depends on the weighted homogeneity properties to infer finite time convergence for the case of $\alpha = 0$.

In contrast, from the results of [37] and [38], a strict non-smooth Lyapunov function is proposed for the family of controllers (1 - 1), where the upper bound of the settling time can be obtained. On a related development in [24], a non-smooth proportional-derivative (PD) controller is proposed that is of the same form as the controllers (1 - 1) shown above. There, an explicit construction of Lyapunov function, based on the method of [39], is given. In spite of that, the Lyapunov function given cannot accept zero fractional power (i.e. $\alpha = 0$), the Lyapunov function does not work for the twisting algorithm. Within the work of [40] the conditions on the gains of the controllers (1-1) are developed for $\alpha \in (0, 1)$ that ensures finite time stability.

While in [41], an important link between asymptotic stability with finite-time stability is provided. It is shown that a homogeneous system is finite-time stable if and only if it is asymptotically stable and has a negative degree of homogeneity. On the other hand, a double integrator nonlinear system [42],

$$\dot{z}_1 = z_2, \, \dot{z}_2 = -f(z_1) - g(z_2)$$

where f(0) = 0 and g(0) = 0 are continuous functions verifying the sign conditions, $z_1 f(z_1) > 0$, for $z_1 \neq 0$, and $z_2 g(z_2) > 0$, for $z_2 \neq 0$, is shown to be globally asymptotically stable. Since, the parameterized controllers (1-1) for $\alpha \in (0, 1)$ with gains k_1 and $k_2 > 0$ satisfy the above sign conditions, the system is asymptotically stable.

In addition, it is noted that the parameterized family of system (1-1) is homogeneous of degree ($\alpha - 1$), which is negative for $\alpha \in (0,1)$, with respect to dilation (($2-\alpha$), 1) (see definitions of [13]). This implies that the family of controllers are finite time stable when k_1 and $k_2 > 0$ and $\alpha \in (0, 1)$. In [37] and [38] the gains are required to be greater than some positive values. Similarly the strict Lyapunov function of [24] requires k_2 to be sufficiently large, while k_1 has to belong to a certain positive interval in the results of [40]. Thus, it suggests that the gain restrictions given by these results are conservative.

Pertaining to the robustness of this family of controllers, two types of perturbation are analyzed in [18], namely $|\delta| \leq M$, and $|\delta| \leq \mu |z_2|^{\alpha}$ where δ is an external disturbance, M and μ are positive constants, and α has the same value as that of the controller (1-1). The first one is a uniform upper bounded type and the second one is a vanishing perturbation. Note that both conditions are identical when $\alpha = 0$. However, finite-time stability can only be proved for the uniform upper bounded type for $\alpha = 0$ (i.e. the discontinuous control). For the vanishing perturbation with $\alpha \neq 0$, the results of [18] can guarantee asymptotic stability only. Furthermore, the vanishing perturbation constraints are with respect to one state only. In [43], a recent survey on robust finite time stability is presented. It suggested that robustness of continuous finite time controllers to disturbances that are bounded by continuous bound with fractional exponent has not been proven. Furthermore, they noted that a recent result in this direction without proof appears in [37]. In [44], a similar form of perturbation is considered as well, however the range of α is limited and it requires additional condition on the gain of i.e. $k_1 > k_2 + \mu$. This system is also studied in [45] which provides asymptotic stability through explicit Lyapunov method, and concludes the finite-time stability through implicit Lyapunov method for the system (1-1) and that of the super-twisting observer. However, due to the structure of the Lyapunov function, the results are only reported for $\alpha \in [0, 1)$. While in [46], the same upper bound on the perturbation is considered but it allows discontinuity in the perturbation, which is not considered in prior contributions. However, the analyses are tedious since no strong Lyapunov function is available.

The vanishing perturbation that is with respect to both state variables,

$$\left|\delta\right| \le M_1 \left|z_1\right|^{\frac{\alpha}{2-\alpha}} + M_2 \left|z_2\right|^{\alpha} \tag{1-2}$$

where M_1 and M_2 are positive constants, are studied in [37] and [38]. Note that the vanishing perturbation considered in [18] is a special case of this when $M_1 = 0$. It should be noted that this perturbation has a non-Lipschitz continuous bounds that vanish at the origin. While [18] only manage to guarantee asymptotic stability under the vanishing perturbation, [38] manage to prove finite-time stability through its strict Lyapunov function. However, similar to its unperturbed analysis, the gains have to satisfy a more conservative constraints for the case of $\alpha = 0$ (i.e. twisting algorithm). In particular, when $\alpha = 0$, with $M_1 + M_2 = M$, the well known inequality [20] for the twisting algorithm under bounded perturbation, $k_1 - M > k_2 > M$ cannot be obtained from the results of [38], which suggest the stability properties of the system not being fully characterize by the Lyapunov function presented there.

Of recent, [47] put forward a stability analysis of second-order sliding mode for a discontinuous control by means of a Lyapunov function. The main benefits are to fully characterize the stability properties of the system so that finite-time convergence can be concluded without resorting to geometric methods, to provide a relationship between the gains and its estimation of convergence time. Pertaining to Lyapunov function candidate, [48] provides an important theorem that relates the finite-time stability of a system with the existence of a Lyapunov function. It states that there exists a strict Lyapunov function for a system if the origin of the system is a finite time equilibrium. In particular, the Lyapunov function satisfy the differential inequality $\dot{V} + kV^{\beta} \leq 0$ where k > 0 and $\alpha \in (0, 1)$. That contribution provides not only an existence of strict Lyapunov for the stability analysis, but also an estimate on the upper bound of the convergence time of the system based on the differential inequality.

Regarding the construction of Lyapunov function, [49] provides a method to construct strict Lyapunov function for a class of HOSM algorithms. In particular, the twisting algorithm is considered and a Lyapunov function is shown in the work. However, the construction is highly dependent on the knowledge of an expression for the solutions of the system, which is difficult to obtain for nonlinear systems in general. In [50] and [51], Lyapunov functions, obtained through solving partial differential equation, are applied to study the finite-time stability of some finite-time 2-sliding mode algorithms, including the twisting, and super-twisting algorithm, and terminal controllers [52]. It is found in a recent study of HOSM

control schemes [53] that these Lyapunov functions are indeed homogeneous in nature. However, the form of the function makes it difficult to operate with for applications or developments [17].

Furthermore, recently a Lyapunov-based homogeneous controllers is presented by [54] for perturbed integrator chains, of which the results of universal SISO sliding mode of [55] are a special case of. In particular, greater simplicity in analyses can be achieved by taking advantage of the homogeneity properties of the system by using a homogeneous Lyapunov function. In [56], the existence of a homogeneous Lyapunov function for a homogeneous asymptotically stable system is shown. In particular, a homogeneous Lyapunov functions for a terminal sliding mode-like second order system is constructed in [57]. Due to the homogeneous nature of the Lyapunov function, several interesting properties can be obtained that simplifies the construction of a differential inequality that ultimately concludes finite-time stability.

Other than finite-time stability, there is another relevant development on homogeneous systems that provides a convergence time independent of the initial conditions. In [58], a lemma that relates the convergence rate of a homogeneous system with its degree of homogeneity is provided. In particular, the origin of a system is rationally stable if it is homogeneous with degrees greater than zero, exponentially stable if it is equal to zero, and finite-time stable if it is negative. For the rational stability, the states of the system will asymptotically converge to zero. It differs from the exponential convergence in the sense that it converge to a bounded region in finite time independent of the initial conditions of the system. The upper-bound of the settling time is dependent only on the said bounded region.

This condition is similar to the uniform convergence shown in [59].

This attractive feature of uniform convergence has attracted numerous attentions lately. For instance, by employing a control term having exponent greater than unity, fast and uniform convergence can be found in the work of double power reaching law for sliding mode control [60], uniform sliding dynamic of [59], generic second order algorithm of [17], and fixed-time stabilization of [61] and in [62] a fixed-time convergent super-twisting-like control inspired from [63] is presented, to name a few.

Since the parameterized family of controllers (1 - 1) is homogeneous, it is of interest to study its uniform convergence characteristic when its range of $\alpha \in [0, 1)$ is allowed to extend beyond unity, which to the best of the authors' knowledge, has not been reported in the published literature.

The properties of (1 - 1) and the kind of perturbations that each of the member of the algorithms can tolerate are different depending on the parameter α . The member with $\alpha \ge 1$ has correction terms that are stronger further away from the origin and not as strong near the origin as those members with $0 \le \alpha < 1$. These differences are related to the kind of perturbations that each member of the algorithm is able to tolerate as suggested by the form of (1 - 2), although no conclusive result has been shown in the literature

Hence a natural question arises whether the linear combination of both members of the family can inherits the best properties of both. A result that is close to answering the question is found in [64], where a finite-time dynamics which has a fast transient process is introduced. Instead of having a linear sum of two homogeneous controls with different exponent, the fast transient process is achieved by modifying the finite-time homogeneous control law to have

greater correctional values when the states are far away from the origin. On the other hand, when the states are smaller than a threshold, the control law is essentially a double integrator with control (1 - 1) of negative degree of homogeneity hence providing finite-time stability. Nevertheless, only weak Lyapunov function is given for analysis, as such no conclusion is obtained on its robustness. Another question is, if a strong homogeneous Lyapunov function can be developed for the family of algorithms, how can it be employed for this new combinational algorithm which is not homogeneous, that comprise of sum of two homogeneous algorithm.

The above questions are answered positively however, not for the twisting based family of controllers (1 - 1), but for a closely-related SOSM based controllers, namely, a generalized super-twisting algorithm that comprises the super-twisting algorithm with additional linear correctional term found in [65]. In it a linear framework is put forward for the algorithm that allows the construction of a strong Lyapunov functions. While in [17] a generic second order sliding mode (super-twisting based), which extends the result of [65], where two Lyapunov functions, which are structurally different, are developed to show different aspects of convergence properties of the system. Both in [65] and [17], the control comprise of linear combination of two homogeneous control term of different homogeneity plus a additional term that has an exponent that depends on both the individual component. Hence, several questions arise here for the super-twisting based family of controllers, as whether the additional term in the linear combination is necessary in order to attain the best properties of both components, and whether the structurally different Lyapunov functions are required to obtain the same results. To that end, it is imperative to study the relationship between the

super-twisting and the twisting algorithms and their respectively based family of controllers. In [32] and [66] the same mechanical energy are employed as the basis for constructing Lyapunov function to analyse the stability of both twisting and super-twisting algorithms. However, the resulting Lyapunov function is a weak one; i.e. its time derivative is only negative semi-definite.

While in [67] a super-twisting algorithm plus an adaptive term to compensate structured uncertainty is proposed. The additional adaptive term is based on the certainty equivalence principle, in which the controller is designed initially under the assumption of known parameters by means of a nominal Lyapunov function, in which case an adaptation law is derived from. There, it shows the importance of having a strong nominal Lyapunov function for the super-twisting algorithm, as a weak nominal Lyapunov function when applied to obtain adaptation law, stability of the system states cannot be concluded. At the same time, the strong Lyapunov function required here have to be at least Lipschitz continuous, since the strong Lyapunov function for the super-twisting algorithm developed in [26], which is non-Lipschitz, resulting singularity to appear in the adaptation law. Thus, it is of importance to not only develop a strong Lyapunov function but a locally Lipschitz one as well for the super-twisting algorithm in order for it to have a wider applications. In [67], a Lyapunov function that satisfies these criteria, developed by [68], is used. However, it is mentioned that the said Lyapunov function requires a more conservative condition on the controller gains than would be obtained employing the previously weak Lyapunov function or the non-Lipschitz one.

The preceding results and arguments point to the direction of finding strict and locally Lipschitz Lyapunov functions that are able to characterize the various stability properties of the family of controllers (1 - 1), which is a twisting-based algorithm, and correspondingly a super-twisting based algorithm, together with a derivative algorithm that are based on the sum of two homogeneous ones.

However, constructing Lyapunov function for a system is difficult, let alone for a family of system. In essence, some of the works above employ strict Lyapunov function for stability analyses without considering the homogeneity properties of the system. Interestingly, in [69] a constructive method is proposed for generating Lyapunov function for a class of homogeneous systems using Polya's theorem. However, there are some drawbacks mentioned in the method including the selection of monomials and the exponent in the Polya's theorem which are unknown in the initial selection process. On the other hand, others rely on a weak Lyapunov function for asymptotic stability, together with a homogeneity approach to prove finite-time stability. These methods inevitably provide restrictions on the range of usability of the Lyapunov function even when the controllers are from the same parameterized family.

At the same time, the system described above may have differential equations with discontinuous right-hand side (i.e. the twisting or the super-twisting algorithm or having perturbations that is discontinuous). According to Filippov's theory, a solution to a differential equation with discontinuous right-hand side is an absolutely continuous function that satisfies a suitable differential inclusion associated to the differential equation [2]. In particular, for some nonsmooth dynamic system, it is natural for the system to assume a nonsmooth Lyapunov function (Example 1, [70]). Due to the lack of differentiability of nonsmooth Lyapunov function, the usual Lyapunov's theorem [71] cannot be applied. Instead, we need some tools of generalized Lyapunov analysis for which the stability properties of nonsmooth

dynamic systems can be determined such as in [70] [72] [73] [74].

Chapter 3: Robot manipulator control - full state feedback approach: An approach with sliding mode control has also been followed extensively on robot manipulator systems. However, such an approach will lead to discontinuous control [75]. In [76] the authors employ a smooth robust controller that comprises a proportional term and an integral term of a linear sliding mode variable for the trajectories control of robot manipulator. The stability analyses ensure the states to have asymptotic convergence. However, to ensure robustness, the robust gains of the control law have to dominate the first and second derivative of the uncertainties, which are difficult to obtain. While, in [77] the authors provides a chattering free sliding mode based control for trajectory tracking of robot manipulator. It can ensure global invariance by having the system to be on the sliding surface from the initial conditions. This is achieved through an integral sliding surface, hence producing a high order sliding manifold. However, due to the formulation, extended state variables, the acceleration, have to be available for computation o the control law. Higher order sliding mode algorithm for chattering reduction and finite-time stability for the control of robot manipulator is reported in [78]. However, the robust gains of the system have a singularity problem. In particular, when the sliding variable is zero while its derivative is non-zero, singularity will occur. In order to produce a smooth variable structure control on robot manipulator, the use of a low pass filter is presented [79]. Essentially, a virtual controller is designed based on sliding mode approach for a virtual plant that comprises the actual plant in cascade with a low pass filter. Hence, the switching action will be filtered before being applied to the actual plant.

Beside the discontinuous control, application of sliding mode control entails a certain reaching phase. The robustness property that sliding mode is well-known for can be achieved only after the occurrence of sliding mode. During the reaching phase, however, no guarantee on robustness is available [75]. In [80], a time-varying sliding manifold that comprise a conventional linear sliding manifold together with an exponentially decaying term is proposed to overcome this issue. By designing the initial condition of the decay term to be equal to the initial value of the linear sliding manifold, the time-varying manifold will be zero initially, hence avoiding reaching phase altogether. The concept of integral sliding mode to the trajectory tracking of robot manipulators is put forward in [75]. In particular, a low pass filter is added to reduce chattering effect of the discontinuous control term. By the means of adjusting the time constant of the low pass filter, the algorithm has a characteristic of a perturbation estimator to that of a pure integral sliding mode. Also, due to the use of an integral sliding surface, initial conditions of the algorithm can be chosen to match those of the robot manipulators so that the states of the systems are on the sliding manifold initially, thus having robustness throughout an entire response of the system starting from initial time instance [81]. Essentially, the integral sliding mode control leads to a sliding manifold that spans the whole state space [82]. Thus the tuning of the time constant of the filter, the tradeoff between chattering reduction and robustness can be adjusted.

To apply second order sliding mode, in [83] a time varying nonsingular terminal sliding mode (NTSM) control for robot manipulator is presented. It is able to eliminate reaching phase by formulating a time-varying nonsingular terminal sliding surface. Essentially, the surface is augmented by an additional time varying function that provides the sliding function

to be zero at the initial time instance and decay to zero in finite time. Together with a switching type robust control term, the system will be in sliding mode from the initial time instance. Also, in [84] the NTSM is use as the sliding manifold and a fast terminal sliding mode type reaching law for ensuring finite time convergence. The control law also comprise of nominal parts of the robot dynamics for compensation of the nonlinearities. Another example can be found in [85], where a type of fast nonsingular terminal sliding mode for the control of robot manipulators is presented. The sliding manifold consists of a nonsingular terminal sliding mode together with a proportional term.

Effectively, the main feature of sliding mode utilized in the previously mentioned controller is its inherent robustness properties to uncertainty and disturbances, in which case is also its main drawback because to have the said robustness properties, its gain for the discontinuous control term has to be sufficiently larger the upper bound of the disturbances. As such, a lot of effort has been done to research on the method of reducing the gain of the discontinuous term while at the same time having the same level of robustness in the system. Towards this direction, a sliding mode algorithm for the control of robot manipulator with an efficient online compensation for tracking of trajectories is found in [86]. The compensation is computed from the acceleration information and the torque applied to the robot manipulator. This method of uncertainties compensation is the essence of time-delay estimation, and the conventional sliding mode is applied not on the uncertainty of the system itself but on the error between the uncertainty and its compensated (estimated) form, which is assumed to be smaller.

A slightly different approach is found in [87], where a gradient estimator, instead of sliding

mode approach, is applied on time-delay control (TDC) for improving the robustness of robot manipulator control under the presence of nonlinear friction. The control law consists of a time-delay estimation term to estimate the nonlinearities of the system, a desired error dynamics term, and a gradient estimator term as a compensator for the time-delay estimation error. The algorithm is shown to provide similar performance to the case of time-delay control with switching action which is sliding mode based. Similarly, an approach called time-delay control with ideal velocity feedback (TDCIVF) for controlling tracking problem of robot manipulators is given [88]. The control structure is simple; it has three distinct elements, namely the soft nonlinearity compensation term, hard nonlinearity cancelling term, and a desired error dynamics injection term. The soft nonlinearity compensation term is of the timedelay estimation form, while the hard nonlinearity is taken care of by the ideal velocity feedback. The so-called ideal velocity feedback term can be viewed as a proportional control term of an integral sliding surface variable, which its derivative is the desired error dynamics.

It is worth mentioning in [89], a simple decentralized linear time-invariant control for robot manipulator as a alternative to computation intensive computed torque method. The algorithm also uses a time-delayed control together with a specially designed constant diagonal gain matrix for the decoupling and linearization of the robot joint dynamics. The desired error dynamics are of a linear PD structure. Sufficient condition for ensuring stability for the design of the gain matrix is given, however, it requires knowledge of the inertial matrix of the robot manipulator. In [90], a similar robust control for the trajectory tracking of robot manipulators is developed that is based on a disturbance and uncertainty estimation (UDE), instead of time-delayed control. The control formulation consists of two parts, one to inject a desired linear

error dynamics, and another term is based on the UDE [91] to compensate for the uncertainties and disturbances of the system. Essentially, the compensation includes the design of a constant diagonal matrix and a time constant for a first order low pass filter. The stability of the system is dependent on the effectiveness of this compensation. In particular, if exact compensation is achieved, exponential convergence can be achieved based on the injected desired linear error dynamics. On the other hand, if the derivatives of the uncertainties are non-zero but finite, uniform ultimate boundedness can be attained. However, the existence of the estimation term is not properly shown and it is mentioned that the compensation does not exist for systems that have discontinuous disturbances and uncertainties.

It is of interest to note that the time-delay estimation method of TDC has similar structure as the UDE algorithm above. Similar to the UDE, the time-delay-estimation method for the estimation of uncertainties includes the design of a constant diagonal matrix for decoupling of the nonlinear dynamics of the robot manipulator. In particular, it is shown by [92] that the time-delay estimation does indeed behave like a first-order digital low-pass filter, in the sense that the diagonal elements of the constant matrix is related to the cutoff frequency of the digital low-pass filter. Besides that, the time-delay estimation also has similar drawbacks as of the UDE. For instance in [93] the authors show that the time-delay estimation algorithm have difficulty in estimating hard nonlinearity or discontinuous uncertainty. Also, the time delay estimation has an inherent property of an integrator as shown by [94], which is similar to the integration action of the UDE. Hence, time-delay estimation can be seen as a discrete form of the UDE algorithm. In [95] UDE-based control was proposed as a replacement of the timedelay control [96]. Along the same line of estimating disturbance in robotic manipulators, a relationship between the discrete TDC and the discrete PID controller is established in [97]. In particular, the gains of the discrete PID controller can be selected such that it has same properties as the TDC. While in [98] and [99], based on the concept of modelling error compensation, the PID control is formulated as a composition of modelling error estimator and a certainty equivalent feedback function for regulation and tracking problem of robot manipulators. As most of the literature on the robot joint position control problem deal solely with the stability problem rather than the system performance in a transient situation, it has been early recognized that transient performance guarantees deserve further research as noted by [98], where a PID control scheme with acceptable transient performance guarantee is proposed. Several works on prescribed performance guarantees have been presented that utilised error transformation, see [100], [101]. A sliding mode controller with guaranteed transient performance is proposed for application on robot manipulators, [102], where by choosing proper initial value of the controller, reaching transient is eliminated

Note that TDE, UDE, and the modelling error observer discussed above, all have a similar structure, in which they can be reformulated to be PID control that inherently comprise of a linear desired error dynamics and a form of uncertainty compensation. The closed-loop error dynamics in all cases above involve the time derivative of the lumped disturbances and uncertainties, which its upper bound may not be easily evaluated in practical applications. Also, for both the TDE [97] and modelling error compensation [99] approaches it is necessary to choose an appropriate inertia matrix estimate term.

Alternatively, robustness of PID-controlled manipulators is studied differently in [103] as

opposed to the formulation in [99]. It is shown that uniform semiglobal practical asymptotic stability can be achieved and a tuning procedure of the PID gains is given in order to obtain any given precision from any given bounded set of initial conditions. It is worth noting that the results are obtained through analysis of a strict Lyapunov function. Also, the perturbations that are considered there include the discontinuous functions of the state such as Coulomb friction and it does not require an appropriate inertia matrix estimate term. Furthermore, in [104], [105] a tuning procedure for the PID gains that ensures semiglobal asymptotic stability for the regulation problem of rigid robots. Although the tuning procedure there allows the selection of PID gains that ensures stability in a specified arbitrary domain, the transient performance of the closed-loop system is unclear from the gain selection procedures. As such, the performance in terms of desired error dynamics is not clear in these approaches.

Hence, a natural question arises here as to whether a particular formulation of stability analysis is available, such that it provides a strict Lyapunov function that ensures semiglobal practical stability of PID-controlled manipulators and at the same time, provides a linear desired error dynamics and an uncertainty compensation components that relates directly to the PID gains. This relationship between conventional PID control and the nonlinear formulation of TDE or modelling error estimator is useful, as it provides a systematic way of PID gain selection as opposed to heuristic gain tuning approach which has its own problems due to too many gains to tune simultaneously [97].

Next, it should be noted that in the previously mentioned integral sliding mode, TDE-based, UDE-based, and PID control of robot manipulators, even in the case of exact model compensation, only exponential convergence is attained due to the inherent linear desired

error dynamics as opposed to that of NTSM, which can ensure finite time convergence [84]. The linear error dynamics is actually the response of linear second order system, and by designing the desired dynamics gains, the well-known responses such as stable node or stable focus in the phase portraits can be achieved (see Chapter 2 of [71]). However, due to the structure of NTSM, which comprise of two first order sliding mode, the type of desired error dynamics is limited. Essentially, under the conditions of no perturbations, it has two distinct phases, namely a reaching like phase, which will bring the trajectories towards the NTSM sliding surface if it is no already there, and once the NTSM surface, it will have a finite time sliding phase towards the origin, thus it is not possible for it to exhibit stable focus in its phase portraits. Hence, another question emerges here as to whether one can design a desired error dynamics that can retain finite time convergence property of NTSM while allowing the flexibility of selection of desired responses either stable node-like or stable focus-like in its dynamics. To answer it, one can consider the dynamics (1 - 1), which is a second order sliding mode based algorithm. In [106], for motion control of permanent-magnet linear motors, an integral sliding mode control with (1 - 1) as desired error dynamics is presented. It comprise of the usual discontinuous reaching law to enforce sliding mode, in which case, the system behaved like (1 - 1) in sliding mode. However, if the sliding surface is not reached exactly but reached within a bounded region (i.e. when a boundary layer method is employed to reduce chattering), the convergence of the states are not clearly shown due to the lack of strong Lyapunov function for the desired error dynamics of (1 - 1).

Chapter 4: Robot manipulator control - output feedback approach: Several outcomes

in this direction have been developed. For instance, in [12], super-twisting algorithm is employed as observers for both velocity and uncertainties estimation. The same structure is employed in [24] as well, but the robot dynamics is required in its implementation. While in [90] a Luenberger-like plus UDE-based robust observer was proposed to solve the problem of requiring joint velocities for control. Additionally, a filter based on so-called "dirtyderivative" is used for finding the velocity from position measurements in [107], [108], [109], [110].

However, the Luenberger-like observer and the "dirty-derivative" can only provide asymptotical convergence. On using super-twisting algorithm as a velocity observer, the admissible upper bound of unknown disturbances and its finite-time convergence properties is studied in [111]. In addition, it is mentioned that finding the tradeoff between the gains of the algorithm to minimize chattering amplitude at the presence of unmodelled dynamics is an open problem. In [112] a modified super-twisting algorithm with double closed-loop feedback regulation is proposed.

$$\dot{\sigma} = -k_1 |\sigma| \operatorname{sign}(\sigma) - k_2 \sigma + \omega,$$

$$\dot{\omega} = -k_3 \operatorname{sign}(\sigma) - k_4 \omega + \phi(t),$$

Essentially, it added a linear correction term $-k_4\omega$ to the $\dot{\omega}$ dynamics as opposed to the linear correction term of the sliding variable proposed by [26]. It is reported in their results that the modified super-twisting algorithm can improve the convergence of the sliding variable by accelerating the approaching speed and at the same time limiting the overshoot. Note that in the observation error dynamics, both the Luenberger-like and the super-twisting algorithm do not have a velocity observation error term in the dynamics, which in part due to
the lack of information on the velocity information, and hence the need of this observer in the first place. However, from the "dirty-derivative" formulation it is indeed possible to get a linear correction term into the observer error dynamics. Hence, another question arises as to whether the combination of super-twisting algorithm with "dirty-derivative" can complement each other, and if yes, can a strict Lyapunov function be developed that characterize its features.

Chapter 5: Robot manipulator control - bounded control approach with full-state feedback: Motivated by issues with actuator constraints for robot manipulator control, some efforts have been proposed in the literature. For instance, assuming exact value of robot manipulator parameters, a bounded static feedback for trajectory tracking for robot manipulator is proposed in [113]. In achieving semiglobal finite-time tracking, a saturated control law plus desired trajectories based dynamics term is found in [114]. It has the ability to ensure that actuator constraints are not violated by selecting control gains *a priori*. In [115] an asymptotic tracking control for robot manipulators with actuator saturation is presented. The control law comprises saturated hyperbolic tangent function and computed feed-forward of robot dynamics terms. In [116] a static nonlinear controller is added to an existing PD control plus exact gravity compensation to guarantee global asymptotic stability for the Euler-Lagrange system with input saturation. However, these methods require full dynamics of the system. This is not desirable because it requires *a priori* knowledge of parameter values of the system, including that of payload, which is particularly restrictive, because in typical tasks many different payloads are encountered and it is unrealistic to assume that the properties of all payloads are accurately known [117]. Also, the transient performance of these methods is not clearly linked to the control gains. Particularly, if the desired error dynamics is to be injected into the control, it should be expected for it to be modified due to the bounded nature of the actuator.

While in [118] global asymptotic stability for the tracking control of robot manipulators can be achieved in the presence of sufficiently large viscous friction by additionally including a feed-forward compensation term of the viscous friction. Additionally, in a frictionless setting, the control scheme of [119] is proven to only semi-globally stabilize the closed-loop system. Semiglobal stability for robot manipulator regulation problem is presented through a saturated linear PID control in [120]. In particular, their stability analysis showed that the semiglobal stability is due to the Coriolis term. Hence, it is concluded in [120] that the need for nonlinear integral function is justified to dominate the effects of Coriolis term at high velocities, for instance see [121], [122], [123], [124]. In the work of [125], theoretical justification is provided on the exponential stability for regulation problem of classical PID used in industrial robots in the presence of saturation effects, essentially the controller comprise nonlinear integral term as well.

Of interest is the new approach for integral action within a continuous sliding mode control design framework in [126]. The integrator presented is modified to provide integral action only inside the boundary layer. In particular, the anti-reset windup structure of conditional integrator is explored in [127]. In essence, the conditional integrator has an inherent anti-reset windup built-in. The conditional integrator bears resemblance to the above mentioned nonlinear integral in which it contains an integration of a saturation function.

Thus, by accounting the presence of viscous friction together with a nonlinear integral term, yet another question emerges as to whether a bounded control law for global tracking can be developed that answers all the previously mentioned questions and the modifications that is required, if necessary. Additionally, when no saturation occurs, it is expected for the system to behave similarly to its unbounded-control counterpart.

Chapter 6: Robot manipulator control - output feedback bounded control approach: Bounded controller without velocity measurements for robot manipulator have been studied in the literature. For example, for global regulation of robots using position measurements only are achieved in [128] [129]. However, both requires the desired gravity compensation term. While for global stabilization, in [130] a bounded output-feedback PID-type controller of robot manipulators is proposed. However, for the velocity observation, they share a structure similar to that of the "dirty-derivative".

For Luenberger-like observer, a saturated output feedback based PID control is proposed in [131]. The resulting controller is simple to implement. It is robust to parameter uncertainties, decentralized, and saturated. However, it only achieves semiglobal stability but for sufficiently high gains, the controller, locally, can achieve exponential stability for regulation problem. On the other hand, global regulation is achieved in [132] through adaptive control that is output feedback-based in a bounded control approach. As per the literature review of Chapter 4, question arises as to whether the combination of super-twisting algorithm of Chapter 2 with "dirty-derivative" can complement each other, and if yes, can a strict Lyapunov function be developed that characterize its features. Moreover, if the answer to the

question positive, it is of interest to know as to whether it can be extended to a bounded control approach while maintaining its original features.

1.3 Problem Statement

With the above motivations, the problems investigated in this dissertation are now presented.

- The twisting based dynamics (1 1) is to be explored not only for α ∈[0, 1) as per the literature, but for all α≥ 0. In particular, strict Lyapunov is to be constructed that can allows the full set of stabilizing constant gains.
- Due to the different convergence properties of the above mentioned system when, 0
 ≤ α < 1 and that of α > 0, a non homogeneous algorithms that comprise of linear
 sum of the two different system is to be studied, as to ascertain whether this new
 combination exhibits the properties of their individual components by means of
 strict Lyapunov function.
- 3. The super-twisting based homogeneous algorithm is revisited to study the possibility of having a single structure of locally-Lipschitz Lyapunov function that can fully characterize the various convergence properties. The locally-Lipschitz is important as to avoid singularity in its time derivative (as mentioned in Section 1.1).
- 4. Similar to point 2, the problem of a linear combination of two different degree of homogeneity of the super-twisting based algorithms is investigated. In particular, the necessity of an additional term found in the work of [26] is to be explored and the possibility of construction of a single strict Lyapunov function that fully characterize the system while at the same time avoiding singularity.
- 5. The similarities of the mechanical energy of the super-twisting and twisting based

algorithms are to be examined. In particular, the idea of combining both algorithms is to be explored through strict Lyapunov functions.

- 6. The types of disturbances that the systems can tolerate for each system (point 1-5) are to be considered. The possibility of robustness towards non-Lipschitz and discontinuous disturbances are looked into through the generalized Lyapunov framework and the construction of strict Lyapunov functions.
- 7. The control problem of the highly coupled nonlinear dynamics of robot manipulator, which is a second-order system, is to be dealt with by investigating the idea of injecting the second order sliding mode algorithms, mentioned previously, into the systems and how it affects the type of convergence attainable.
- 8. The application of the super-twisting and twisting based algorithms on the issue of lack of velocity measurements of the robot manipulator is to be investigated as well.
- 9. Another practical problem of the control of robot manipulator, the actuation limits of the control input to the system is looked into for the possibility of application of the said second order sliding mode based algorithms.
- 10. Finally, the idea of combining the constraints point 8 and 9 into the trajectory tracking control of robot manipulator is explored.

1.4 Contributions

The contributions of the main chapters of the dissertation are discussed as follows:

Chapter 2: Lyapunov Approach on Twisting and Super-twisting based Second Order Sliding Mode: Strict Lyapunov functions are developed for twisting and super-twisting based family of algorithms by using the generalized Lyapunov theorem for non-smooth systems and using the Filippov solutions. Due to the strictness of the proposed Lyapunov functions, whose time derivative can be bounded by negative definite functions, settling time for the finite time convergence member of the algorithms can be obtained. In addition, the strict Lyapunov functions are employed throughout the chapter to study the type of disturbances that the algorithms can tolerate, which include the non-Lipschitz type as well (Preliminary conference version of the some results here can be found in [133] and [134]).

Chapter 3: Robot Manipulator Control: Full State Feedback Approach: The main contribution of Chapter 3 is the development of control law that generalised the well-known PID control, which comprises a desired error dynamics injection and uncertainty and disturbance compensation. Nonsmooth analysis methods introduced in Chapter 2 are used throughout the stability analysis. In particular the twisting based algorithms developed in Chapter 2 are to be employed as the desired error dynamics in the control law. The technical challenge presented by this aim, is the need to avoid differentiation of the non-Lipschitz twisting based algorithms, which will lead to singularity. To achieve this objective, an auxiliary desired error dynamics variable is introduced, which in effect avoid the singularity problem by having the non-Lipschitz desired error dynamics in to an integral term. Through this formulation, semiglobal practical trajectory tracking is achieved, where the region of

attractions is directly dependent on a single gain parameter.

Chapter 4: Robot Manipulator Control: Output Feedback Approach: An observer based on the super-twisting based algorithms of Chapter 2 is utilised to tackle the problem of lack of velocity measurements. In particular, due to the non-Lipschitz nature, properties like finite time convergence of observation errors can be obtained rendering the controller development to be akin to that of full-state feedback in Chapter 3. However, the non-Lipschitz gains of the observer can be high even for large initial conditions of the closed-loop system. To overcome this issue, the proposed observer contains an additional linear damping term that can aids in reducing the said non-Lipschitz. Not only that, to increase the size of the region of attraction only the linear damping term of the observer gain has to be increased without requiring that of the non-Lipschitz gains.

Chapter 5: Robot Manipulator Control: Bounded Control Approach with Full-State Feedback: The controller in Chapter 3 is redesigned to accommodate the issue of saturation of actuators. The main problem that hinders global stability of robot manipulator is the quadratic nature of the Coriolis and centrifugal terms in the dynamics. To tackle this issue, the stability analysis takes into account of the inherent viscous friction of the robot manipulator, which has an additional damping effect on the system. In addition, the desired trajectory and the desired error dynamics that is to be injected to the system through the control law has to be modified accordingly to account for the bounded nature of the control. At the same time, it is desired for the control to have the same behaviours as their unbounded counterpart in Chapter 3 when the control is unsaturated. All of these are achieved through the special form of the integral term that injects saturated version of the twisting based desired error dynamics, and its inherent anti-windup nature that conditionally injects the desired error dynamics depending on the saturation condition of the control.

Chapter 6: Robot Manipulator Control: Output Feedback Bounded Control Approach: Based on the previous saturated controller utilized in Chapter 5 and the observer of Chapter 4, an output feedback version of saturated controller in Chapter 5 is developed. Due to the inherent boundedness of states of the closed-loop system (by taking into accounts of the issues mentioned in the previous paragraph), the observer applied here can achieve global stability results independent of initial conditions. Once the observation error is stabilized, it essentially acts as additional disturbances to the closed-loop system that affects the size of the ultimate bound of the states. Due to this nature, the proposed observercontroller structure is able to assure global practical stability for trajectory tracking, and if the actuator limits is sufficiently high, the desired error dynamics can be unsaturated, and similar behaviours observed in the unbounded output feedback control in Chapter 4 can acquired.

Chapter 2: LYAPUNOV APPROACH ON TWISTING AND SUPER-TWISTING BASED SECOND ORDER SLIDING MODE

In this chapter, several families of controllers based on twisting and super-twisting algorithms are presented. Strict Lyapunov functions, whose time derivative are negative definite, are developed for each family of controllers, with or without perturbations. The Lyapunov functions presented will fully characterize the type of stability and robustness properties of each algorithm. For the case of finite time convergence, the estimation of the finite settling time is provided by means of the Lyapunov function.

2.1 Preliminaries

In this section, several technical lemmas, some important definitions, and theorems for nonsmooth analysis [135], which will be employed throughout the thesis, are presented.

Lemma 2.1: (Young's inequality [17], and [136]) For every real numbers a > 0, b > 0, c > 0, p > 1, q > 1, with $\frac{1}{p} + \frac{1}{q} = 1$ the following inequality is satisfied

$$ab = \left(ca\right)\left(\frac{b}{c}\right) \le c^p \frac{a^p}{p} + c^{-q} \frac{b^q}{q}$$

Lemma 2.2: If $a_i \ge 0$, for i = 1, ..., n, while $0 < c \le 1$, and $d \ge 1$, the following inequalities hold:

$$\sum_{i=1}^{n} a_{i}^{d} \leq \left(\sum_{i=1}^{n} a_{i}\right)^{d} \leq n^{d-1} \sum_{i=1}^{n} a_{i}^{d}, \qquad (2-1)$$

and

$$\frac{1}{n^{1-c}} \sum_{i=1}^{n} a_i^{\ c} \le \left(\sum_{i=1}^{n} a_i\right)^c \le \sum_{i=1}^{n} a_i^{\ c}$$
(2 - 2)

Proof of lemma 2.2: For (2 - 1) note that when d = 1 or $\sum_{i=1}^{n} a_i = 0$, the results follow directly.

Now, consider the case when d > 1 and $\sum_{i=1}^{n} a_i \neq 0$. Let $A = \sum_{i=1}^{n} a_i$ and note that

$$\frac{a_i}{A} \le 1, \quad \forall i = 1, \dots, n, \text{ hence, } \sum_{i=1}^n \left(\frac{a_i}{A}\right)^d \le \sum_{i=1}^n \left(\frac{a_i}{A}\right) = 1, \text{ since } d > 1,$$

Consider the following,

$$\sum_{i=1}^{n} a_i^{d} = \sum_{i=1}^{n} \left(A \times \frac{a_i}{A} \right)^d = A^d \sum_{i=1}^{n} \left(\frac{a_i}{A} \right)^d \le A^d . 1$$
$$\therefore \sum_{i=1}^{n} a_i^{d} \le \left(\sum_{i=1}^{n} a_i \right)^d$$

Next, let $B = \left(\sum_{i=1}^{n} a_i^{d}\right)^{\frac{1}{d}}$, and $C = n^{1-\frac{1}{d}}$, then,

$$\sum_{i=1}^{n} a_i = BC \cdot \sum_{i=1}^{n} \left(\frac{a_i}{B}\right) \left(\frac{1}{C}\right) \leq BC \cdot \sum_{i=1}^{n} \left(\frac{1}{d} \left(\frac{a_i}{B}\right)^d + \frac{d-1}{d} \left(\frac{1}{C}\right)^{\frac{d}{d-1}}\right), \text{ from lemma 2.1, since } d > 1,$$

$$= BC \cdot \left(\frac{1}{d} \frac{\sum_{i=1}^{n} \left(a_i^d\right)}{B^d} + \frac{d-1}{d} \frac{\sum_{i=1}^{n} \left(1\right)}{C^{\frac{d}{d-1}}}\right)$$

$$= BC \cdot 1$$

$$\therefore \sum_{i=1}^{n} a_i \leq n^{1-\frac{1}{d}} \cdot \left(\sum_{i=1}^{n} a_i^d\right)^{\frac{1}{d}} \Rightarrow \left(\sum_{i=1}^{n} a_i\right)^d \leq n^{d-1} \cdot \sum_{i=1}^{n} a_i^d$$

Thus, we obtained (2 - 1).

Now, for (2 - 2), note that when c = 1, the results follow directly as well. So, consider the case when, 0 < c < 1.

Now, consider,

$$\left(\sum_{i=1}^{n} a_{i}\right)^{c} = \left(\sum_{i=1}^{n} \left(a_{i}^{c}\right)^{\frac{1}{c}}\right)^{c} \le \left(\left(\sum_{i=1}^{n} a_{i}^{c}\right)^{\frac{1}{c}}\right)^{c} \text{ from the left hand side of } (2 - 1), \text{ since } \frac{1}{c} > 1,$$
$$= \sum_{i=1}^{n} a_{i}^{c}$$
$$\therefore \left(\sum_{i=1}^{n} a_{i}\right)^{c} \le \sum_{i=1}^{n} a_{i}^{c}$$

Similarly,

$$\left(\sum_{i=1}^{n} a_{i}\right)^{c} = \left(\sum_{i=1}^{n} \left(a_{i}^{c}\right)^{\frac{1}{c}}\right)^{c} \ge \left(\frac{1}{n^{\frac{1}{c}-1}} \left(\sum_{i=1}^{n} a_{i}^{c}\right)^{\frac{1}{c}}\right)^{c} \quad \text{from the right hand side of } (2-1), \text{ since } \frac{1}{c} > 1,$$
$$= \frac{1}{n^{1-c}} \sum_{i=1}^{n} a_{i}^{c},$$
$$\therefore \left(\sum_{i=1}^{n} a_{i}\right)^{c} \ge \frac{1}{n^{1-c}} \sum_{i=1}^{n} a_{i}^{c}$$

Thus, we obtained (2 - 2).

Lemma 2.3: Let $0 with <math>a \ge 0$, then the inequality hold, $|a|^p + |a|^q \ge |a|^r$.

Proof of lemma 2.3: For $0 \le |a| \le 1$, note that, $|a|^p \ge |a|^r$, since $p \le r$, and $\max\{a|^p, |a|^q\} = |a|^p$, since $p \le q$, hence we have $\max\{a|^p, |a|^q\} \ge |a|^r$. While for $|a| \ge 1$, note that, $|a|^q \ge |a|^r$, since $q \ge r$, and $\max\{a|^p, |a|^q\} = |a|^q$, since $q \ge p$, hence, $\max\{a|^p, |a|^q\} \ge |a|^r$. Thus, for all $|a| \ge 0$, we have $\max\{a|^p, |a|^q\} \ge |a|^r$. Since $|a|^p + |a|^q \ge \max\{|a|^p, |a|^q\}$, hence for all $|a| \ge 0$, one obtains $|a|^p + |a|^q \ge |a|^r$.

Consider the vector differential equation,

$$\dot{x} = f(x,t), \ x(t_0) = x_0$$
 (2-3)

where $x \in \mathbb{R}^n$ is a state vector, and $f: \mathbb{R}^n \times \mathbb{R} \to \mathbb{R}^n$ is defined by almost all points on an open subset $Q \subset \mathbb{R}^{n+1}$ and is measurable. For an arbitrary compact set $D \subset Q$, a Lebesgue integrable function A(t) exists and satisfies the following:

$$||f(x,t)|| \le A(t)$$
, a.e. in D .

Under the above conditions, the existence of a Filippov solution is guaranteed.

Remark 2.1. Throughout the subsequent discussion, for brevity of notation, let a.e. refer to almost everywhere, i.e., for almost all $t \in [0, \infty)$.

Definition 2.1: [70]When a vector function x(t) meets the following conditions, the solution to (1) in the interval [t_0 , t_1] in Filippov's sense is called a Filippov solution.

- a) x(t) is a absolutely continuous on $[t_0, t_1]$.
- b) For almost every $t \in [t_0, t_1]$, the following differential inclusion is satisfied:

$$\dot{x} \in K[f](x,t) \tag{2-4}$$

Here,

$$K[f](x,t) = \bigcap_{\delta > 0} \bigcap_{\mu N = 0} \overline{co} f(B(x,\delta) - N,t)$$
(2-5)

In addition, $\bigcap_{\mu N = 0}$ is the intersection over all sets *N* of Lebesgue measure zero , \overline{co} is the convex hull, and $B(x, \delta) = \{y \in \mathbb{R}^n \mid ||y - x|| < \delta\}$ is an open sphere.

Definition 2.2: [72]A function $f : \mathbb{R}^n \to \mathbb{R}$, which is locally Lipschitz near $x \in \mathbb{R}^n$, is said to be regular at *x* if the following holds. For all directions $v \in \mathbb{R}^n$, there exists the usual one-sided directional derivative

$$f'(x,v) = \lim_{\rho \downarrow 0} \frac{f(x+\rho v) - f(x)}{\rho} \quad \text{and we have } f'(x, v) = f^{0}(x, v), \text{ where}$$
$$f^{0}(x,v) = \limsup_{\substack{y \to x \\ \rho \downarrow 0}} \frac{f(y+\rho v) - f(y)}{\rho}$$

is the generalized directional derivative of f at x in the direction v. The function is said to be regular in \mathbb{R}^n , if it is regular for any $x \in \mathbb{R}^n$.

Remark 2.2. A useful property is that a locally Lipschitz and convex function in \mathbb{R}^n is also regular in \mathbb{R}^n (see [137], Proposition 2.3.6). A feature of Filippov's solution is that it is defined by the condition (2 - 4) for \dot{x} . This approach leads to generalization of the Lyapunov stability theory so that the solution x(t) of the differential equation is not needed explicitly.

Lemma 2.4: (Chain rule [70]) Let the vector function x(t) be Filippov's solution to (2 - 3) on an interval containing t and $V: \mathbb{R}^n \times \mathbb{R} \to \mathbb{R}$ be a Lipschitz-continuous regular function. In this case, V(x(t), t) is absolutely continuous, (d/dt)V(x(t), t) exists for almost everywhere (a.e.), i.e., for almost every t, and satisfies

$$\frac{d}{dt}V(x(t),t) \stackrel{a.e.}{\in} \dot{\widetilde{V}}(x(t),t) = \bigcap_{\xi \in \partial V(x(t),t)} \xi^{T} \begin{bmatrix} K[f](x(t),t) \\ 1 \end{bmatrix},$$

Here $\dot{V}(x(t), t)$ is the generalized time derivative of V(x(t), t), while $\partial V(x(t), t)$ is the Clarke's generalized gradient [138], defined as follows:

$$\partial V(x,t) = \overline{co} \left\{ \lim_{i \to \infty} \nabla V(x_i,t_i) | (x_i,t_i) \to (x,t), (x_i,t_i) \notin \Omega_V \right\}$$

where Ω_V is the set of measure zero where the gradient of V is not defined.

Lemma 2.5: (Lyapunov's Theorem Generalized [73], [70]). Suppose that $V: \mathbb{R}^n \times \mathbb{R} \to \mathbb{R}$ is a Lipschitz-continuous regular function satisfying V(0, t) = 0, and

$$0 < V_1(||x||) \le V(x,t) \le V_2(||x||), \quad \text{for} \quad x \neq 0$$

for some $V_1, V_2 \in \text{class } \mathcal{K}$. Then,

1) $\dot{\tilde{V}}(x, t) \le 0$ in implies $x(t) \equiv 0$ is a uniformly stable solution.

2) If in addition, there exists a class \mathcal{K} function $\omega(.)$ that satisfies $\dot{\tilde{V}} \leq -\omega(x) < 0$,

then the solution $x(t) \equiv 0$ is uniformly asymptotically stable.

- 3) Furthermore, if the function $\omega(x) = cV^a$, where c > 0 and a > 0, then we have
 - a) Finite time convergence [72], for 0 < a < 1, with the settling time estimate,

$$T(x_0) \leq \frac{\left[V(x_0, t_0)\right]^{1-a}}{c(1-a)}$$
, where x_0 is the initial states at $t = t_0$.

- b) Exponential convergence [72], for a = 1.
- c) Asymptotical convergence, for a > 1, with convergence time to bounded level set V = μ, for any μ > 0, uniformly upper bounded with respect to the initial condition [17],

$$T_{\max}(\mu) = \left(\frac{1}{c(a-1)}\right) \left(\frac{1}{\mu^{a-1}}\right)$$

Lemma 2.6: (Uniformly ultimate boundedness [139], Theorem 3.3 of [140]). Assume that there exists r > 0, $\sigma > 0$, and $V: \mathcal{D} \times R \to R$ is a globally Lipschitz continuous function such that for any initial condition, $\forall x(t_0) = x_0 \in \mathcal{D}$ be a domain that contains the origin and $||x_0|| \le \sigma$, any Filippov solution of (2 - 3) $x(t) \in S(x_0)$ satisfies:

1) There exist two functions α_1 and α_2 of the class \mathcal{K}_{∞} such that

 $0 < \alpha_1(\|x\|) \le V(x,t) \le \alpha_2(\|x\|),$

2) $\exists 0 < \mu < \alpha_2^{-1}(\alpha_1(r))$ while $||x|| \ge \mu$, there exists a function α_3 of the class \mathcal{K} such

that
$$\dot{\widetilde{V}} \leq -\alpha_3(\|x\|)$$

Then, the origin of the discontinuous system (2 - 3) is globally strongly uniformly ultimately bounded. In particular, there exist a finite $T(r, \sigma)$ such that $\forall t \ge t_0 + T(r, \sigma)$, all the Filippov

solutions $x(t) \in S(x_0)$ of the system (2 - 3) with the initial condition x_0 satisfy $V(x(t), t) \le \alpha_2(||\mu||)$, and by point (1) there holds $||x(t)|| \le \alpha_1^{-1}(\alpha_2(\mu)) < r, \forall t \ge t_0 + T(r, \sigma)$.

For the system described in (2 - 3) with a continuous right-hand side, existing Lyapunov theory can be used to examine the stability of the closed-loop system using continuous techniques such as those described in [71]. However, these theorems must be altered for the set-valued map $\tilde{V}(x(t),t)$ for systems with right-hand sides which are not Lipschitz continuous [73], [70]. Lyapunov analysis for nonsmooth systems is analogous to the analysis used for continuous systems. The differences are that differential equations are replaced with inclusions, gradients are replaced with generalized gradients, and points are replaced with sets in several places.

In the following subsections, locally Lipschitz strict Lyapunov functions will be developed for second order sliding mode that are based on twisting and super-twisting algorithms.

2.2 Twisting Algorithm

In this section, a twisting based family of algorithms is developed. Particularly, the family of algorithms generalised the twisting and linear proportional-derivative (PD) algorithms. Different types of convergence (i.e. finite-time, exponential, and uniform) and the disturbances (i.e. which include non-Lipschitz type) that the algorithms can tolerate are presented by employing strict Lyapunov function throughout the stability analyses, in which its time derivative can be bounded by negative definite function. The inherent homogeneity

properties of the algorithms and that of the proposed strict Lyapunov function are utilised, together with lemmas shown in the preliminaries (section 2.1), to obtained the results that characterize the algorithm.

2.2.1 System description

Consider the twisting based family of algorithms:

$$\dot{z}_1 = z_2,$$

$$\dot{z}_2 = -k_1 |z_1|^b \operatorname{sign}(z_1) - k_2 |z_2|^{\frac{2b}{1+b}} \operatorname{sign}(z_2) + d,$$
 (2 - 6)

where $z_1, z_2 \in \mathbb{R}$ are the scalar state variables, k_1, k_2 are positive constants, $b \ge 0$ real number, and *d* is time-varying and/or nonlinear term of uncertainty bounded by

$$|d| \le M_1 |z_1|^b + M_2 |z_2|^{\frac{2b}{1+b}} + M_3 \text{ with } |d| := \sup \{ |\delta| : \delta \in \mathbf{K}[d] \},\$$

where $M_1 \ge 0$, $M_2 \ge 0$, and $M_3 \ge 0$, with the same $b \ge 0$ as that in (2 - 6). Note that since no continuity assumption is made on d, it may contain discontinuities and hence we define its upper bound through Filippov set-valued map. Note that for the case of b = 0, the algorithm contains discontinuity and the uncertainty is upper bounded by nonvanishing constant.

2.2.2 Stability analysis

For system (2 - 6), the following function

$$V(z_1, z_2) = \left(\frac{1}{2}z_2^2 + \frac{k_1}{1+b}|z_1|^{1+b}\right)^2 + r|z_1|^{\frac{3+3b}{2}}|z_2|\operatorname{sign}(z_1z_2)$$

where r is a positive constant scalar, will be shown as a strict Lyapunov function.

Remark 2.3. Note that the above function is locally Lipschitz and regular. It is differentiable everywhere for b > 0, and not differentiable on $z_1 = 0$ for b = 0. The proposed Lyapunov function comprises the energy function, which is known as a weak Lyapunov function, and an additional cross-term that consists of two states. It is worth mentioning that in studying the stability of system (2 - 6), only a single structure of Lyapunov function is employed in the following analysis for both unperturbed and perturbed cases. However, in [37] and [38], a separate Lyapunov function is required to study the perturbed system. In particular, their Lyapunov function comprise the upper bound of the disturbance, in which case making it unclear on extending the analysis to the cases of perturbations that might includes additive non-Lipschitz vanishing perturbations and non-vanishing perturbations that are considered here.

Note that the Lyapunov function V can be lower bounded by,

$$V(z_1, z_2) \ge \frac{1}{4} |z_2|^4 + \left(\frac{k_1}{1+b}\right)^2 |z_1|^{2+2b} + \left(\frac{k_1}{1+b}\right) |z_1|^{1+b} |z_2|^2 - r|z_1|^{\frac{3+3b}{2}} |z_2|$$

Using lemma 2.1

$$\frac{1}{8}|z_2|^4 + \frac{1}{2}\left(\frac{k_1}{1+b}\right)^2 |z_1|^{2+2b} \ge \left(\frac{4}{6}\left(\frac{k_1}{1+b}\right)^2\right)^{\frac{3}{4}} |z_1|^{\frac{3+3b}{2}} \left(\frac{4}{8}\right)^{\frac{1}{4}} |z_2|$$

Hence, for

$$\left(\frac{k_{1}}{1+b}\right)^{\frac{3}{2}} \left(\frac{2}{3}\right)^{\frac{3}{4}} \left(\frac{1}{2}\right)^{\frac{1}{4}} > r$$

$$(2 - 7)$$

$$\therefore V(z_{1}, z_{2}) \ge \frac{1}{8} |z_{2}|^{4} + \frac{1}{2} \left(\frac{k_{1}}{1+b}\right)^{2} |z_{1}|^{2+2b} \ge \underline{\pi}_{1} \left(|z_{2}|^{4} + |z_{1}|^{2+2b}\right)$$
where $\underline{\pi}_{1} \coloneqq \min\left\{\frac{1}{8}, \frac{1}{2} \left(\frac{k_{1}}{1+b}\right)^{2}\right\}.$

Similarly, it can be upper-bounded by,

$$V(z_1, z_2) \le \frac{1}{4} |z_2|^4 + \left(\frac{k_1}{1+b}\right)^2 |z_1|^{2+2b} + \left(\frac{k_1}{1+b}\right) |z_1|^{1+b} |z_2|^2 + r|z_1|^{\frac{3+3b}{2}} |z_2|$$

Using lemma 2.1:

$$\left(\frac{k_1}{1+b}\right) |z_1|^{1+b} |z_2|^2 \le \left(\frac{k_1}{2+2b}\right) |z_1|^{2+2b} + \left(\frac{k_1}{2+2b}\right) |z_2|^4$$

$$r|z_1|^{\frac{3+3b}{2}} |z_2| \le \frac{3r}{4} |z_1|^{2+2b} + \frac{r}{4} |z_2|^4$$

Thus,

$$V(z_1, z_2) \le \overline{\pi}_1 (|z_1|^{2+2b} + |z_2|^4)$$

where
$$\overline{\pi}_1 := \max\left\{\frac{3r}{4} + \left(\frac{k_1}{1+b}\right)^2 + \left(\frac{k_1}{2+2b}\right), \quad \left(\frac{k_1}{2+2b}\right) + \frac{r}{4} + \frac{1}{4}\right\}$$

Thus, V is positive definite and radially unbounded. Since (2 - 6) is a differential equation that has discontinuous right-hand side, i.e. when b = 0 and since no continuity assumption is

made on d, its solutions are understood in the sense of Filippov (see definition 2.1). According to lemma 2.4, the time derivative of non-smooth Lyapunov function exists almost everywhere

$$\frac{d}{dt}V(\mathbf{z}(t),t) \stackrel{a.e.}{\in} \dot{\widetilde{V}}(\mathbf{z}(t),t) = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix} (\mathbf{z},t),$$

where $\mathbf{z} = (z_1, z_2)^T$ and, $\boldsymbol{\xi} = (\boldsymbol{\xi}_1, \boldsymbol{\xi}_2)^T \in \partial V(\mathbf{z}, t)$.

Since the controller (2 - 6) is discontinuous when b = 0, for ease of presentation, the analysis is separated for two different cases of $b \ge 0$, i.e. b > 0 and b = 0.

a) Case 1: For b > 0

Note that for b > 0, *V* is continuously differentiable, hence

$$\begin{split} \dot{\tilde{V}}(z_{1}, z_{2}) &= \bigcap_{\xi \in \partial V(\mathbf{z}(t), t)} \xi^{T} K \begin{bmatrix} \dot{z}_{1} \\ \dot{z}_{2} \end{bmatrix} = \nabla V^{T} K \begin{bmatrix} \dot{z}_{1} \\ \dot{z}_{2} \end{bmatrix} \\ &\subset -\frac{2k_{1}k_{2}}{(1+b)} |z_{1}|^{1+b} |z_{2}|^{\frac{1+3b}{1+b}} - k_{1}r|z_{1}|^{\frac{3+5b}{2}} - k_{2}r|z_{1}|^{\frac{3+3b}{2}} |z_{2}|^{\frac{2b}{1+b}} \operatorname{sign}(z_{1}z_{2}) - k_{2}|z_{2}|^{\frac{3+5b}{1+b}} \\ &+ \frac{3r(1+b)}{2} |z_{1}|^{\frac{1+3b}{2}} |z_{2}|^{2} \\ &+ K [d] \Big(|z_{2}|^{3} \operatorname{sign}(z_{2}) + r|z_{1}|^{\frac{3+3b}{2}} \operatorname{sign}(z_{1}) + \frac{2k_{1}}{(1+b)} |z_{1}|^{1+b} |z_{2}| \operatorname{sign}(z_{2}) \Big) \end{split}$$

After rearrangement:

$$\begin{split} \dot{\tilde{V}}(z_{1},z_{2}) &\leq -\frac{2k_{1}(k_{2}-M_{2})}{(1+b)}|z_{1}|^{1+b}|z_{2}|^{\frac{1+3b}{1+b}} - \frac{r(k_{1}-M_{1})}{2}|z_{1}|^{\frac{3+5b}{2}} - \frac{(k_{2}-M_{2})}{2}|z_{2}|^{\frac{3+5b}{1+b}} \\ &+ \dot{V}_{1} + M_{3} \bigg(|z_{2}|^{3} + r|z_{1}|^{\frac{3+3b}{2}} + \frac{2k_{1}}{(1+b)}|z_{1}|^{1+b}|z_{2}|\bigg) \end{split}$$

where

$$\dot{V}_{1} = -\frac{r(k_{1} - M_{1})}{2} |z_{1}|^{\frac{3+5b}{2}} - \frac{(k_{2} - M_{2})}{2} |z_{2}|^{\frac{3+5b}{1+b}} + r(k_{2} + M_{2})|z_{1}|^{\frac{3+3b}{2}} |z_{2}|^{\frac{2b}{1+b}} + r\frac{3+3b}{2} |z_{1}|^{\frac{1+3b}{2}} |z_{2}|^{2} + M_{1}|z_{1}|^{b} |z_{2}|^{3} + \frac{2k_{1}}{(1+b)} M_{1}|z_{1}|^{1+2b} |z_{2}|$$

Applying lemma 2.1,

$$\begin{split} -\left|z_{1}\right|^{\frac{3+5b}{2}} - \left|z_{2}\right|^{\frac{3+5b}{1+b}} &\leq -\left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{3+5b}} \left|z_{1}\right|^{\frac{3+3b}{2}} \left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} \left|z_{2}\right|^{\frac{2b}{1+b}}, \\ -\left|z_{1}\right|^{\frac{3+5b}{2}} - \left|z_{2}\right|^{\frac{3+5b}{1+b}} &\leq -\left(\frac{3+5b}{1+3b}\right)^{\frac{1+3b}{3+5b}} \left|z_{1}\right|^{\frac{1+3b}{2}} \left(\frac{3+5b}{2+2b}\right)^{\frac{2+2b}{3+5b}} \left|z_{2}\right|^{2}, \\ -\left|z_{1}\right|^{\frac{3+5b}{2}} - \left|z_{2}\right|^{\frac{3+5b}{1+b}} &\leq -\left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} \left|z_{1}\right|^{b} \left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{3+5b}} \left|z_{2}\right|^{3}, \\ -\left|z_{1}\right|^{\frac{3+5b}{2}} - \left|z_{2}\right|^{\frac{3+5b}{1+b}} &\leq -\left(\frac{3+5b}{2b}\right)^{\frac{2+4b}{3+5b}} \left|z_{1}\right|^{1+2b} \left(\frac{3+5b}{1+b}\right)^{\frac{1+b}{3+5b}} \left|z_{2}\right|, \end{split}$$

Thus, if the following inequalities

$$\min \begin{cases} \lambda_{1} \frac{\left(k_{1} - M_{1}\right)^{\frac{3+3b}{2b}} \left(k_{2} - M_{2}\right)}{\left(k_{2} + M_{2}\right)^{\frac{3+5b}{2b}}}, \\ \lambda_{2} \left(k_{1} - M_{1}\right)^{\frac{1+3b}{2+2b}} \left(k_{2} - M_{2}\right) \end{cases} > r > \max \begin{cases} \lambda_{3} \frac{M_{1}^{\frac{3+5b}{2b}}}{\left(k_{1} - M_{1}\right) \left(k_{2} - M_{2}\right)^{\frac{3+3b}{2b}}}, \\ \lambda_{4} \frac{k_{1}^{\frac{3+5b}{2+4b}} M_{1}^{\frac{3+5b}{2+4b}}}{\left(k_{1} - M_{1}\right) \left(k_{2} - M_{2}\right)^{\frac{1+b}{2+4b}}} \end{cases}$$
(2-8)

where

$$\begin{split} \lambda_1 &= \left(\frac{3+5b}{24+24b}\right)^{\frac{3+3b}{2b}} \left(\frac{3+5b}{16b}\right), \quad \lambda_2 = \left(\frac{3+5b}{8+24b}\right)^{\frac{1+3b}{2+2b}} \left(\frac{3+5b}{16+16b}\right) \left(\frac{2}{3+3b}\right)^{\frac{3+5b}{2+2b}}, \\ \lambda_3 &= \left(\frac{16b}{3+5b}\right) \left(\frac{24+24b}{3+5b}\right)^{\frac{3+3b}{2b}}, \quad \lambda_4 = \left(\frac{16+32b}{3+5b}\right) \left(\frac{8+8b}{3+5b}\right)^{\frac{1+b}{2+4b}} \left(\frac{2}{1+b}\right)^{\frac{3+5b}{2+4b}} \end{split}$$

hold then the function \dot{V}_1 is negative definite. Then,

$$\begin{split} \dot{\widetilde{V}}(z_1, z_2) &\leq -\frac{r(k_1 - M_1)}{2} |z_1|^{\frac{3+5b}{2}} - \frac{(k_2 - M_2)}{2} |z_2|^{\frac{3+5b}{1+b}} \\ &+ M_3 \bigg(|z_2|^3 + r|z_1|^{\frac{3+3b}{2}} + \frac{2k_1}{(1+b)} |z_1|^{1+b} |z_2| \bigg) \end{split}$$

Applying lemma 2.1,

$$|z_1|^{1+b}|z_2| \le \frac{2}{3}|z_1|^{\frac{3+3b}{2}} + \frac{1}{3}|z_2|^3$$

together with lemma 2.2, and the bounds of the Lyapunov function, we have:

$$\begin{split} \dot{\tilde{V}}(z_{1},z_{2}) &\leq -\omega_{1} \left(\left| z_{1} \right|^{\frac{3+5b}{2}} + \left| z_{2} \right|^{\frac{3+5b}{1+b}} \right) + M_{3} \omega_{2} \left(\left| z_{1} \right|^{\frac{3+3b}{2}} + \left| z_{2} \right|^{3} \right) \\ &= -\omega_{1} \left(\left\| z_{1} \right|^{2+2b} \right)^{\frac{3+5b}{4+4b}} + \left\| z_{2} \right|^{4} \right)^{\frac{3+5b}{4+4b}} \right) + M_{3} \omega_{2} \left(\left\| z_{1} \right|^{2+2b} \right)^{\frac{3}{4}} + \left\| z_{2} \right|^{4} \right)^{\frac{3}{4}} \\ &\leq - \left(\frac{\omega_{3}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{3+5b}{4+4b}} + M_{3} \left(\frac{2^{\frac{1}{4}} \omega_{2}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} \end{split}$$

$$(2 - 9)$$

where

$$\omega_{1} = \min\left\{\frac{r(k_{1} - M_{1})}{2}, \frac{(k_{2} - M_{2})}{2}\right\}, \quad \omega_{2} = \max\left\{\left(r + \frac{4k_{1}}{3 + 3b}\right), \left(\frac{2k_{1}}{3 + 3b} + 1\right)\right\},\\ \omega_{3} = \begin{cases}\omega_{1} & \text{for } 0 < b \le 1,\\ \frac{\omega_{1}}{2^{\frac{b-1}{4+4b}}} & \text{for } b > 1\end{cases}$$

Remark 2.4. Note that the nonlinear inequalities (2 - 7) and (2 - 8) are feasible for sufficiently large $k_1 > 0$, and $k_2 > 0$. Thus, an r > 0 can always exists. Consider some particular cases:

- Note that (2 8) is feasible with respect to k₁, k₂ for any M₁ ≥ 0 and M₂ ≥ 0. This implies that for any given upper bound on the disturbances M₁ and M₂, sufficiently large k₁ and k₂ always exist to render V positive definite and V

 i negative definite.
- 2. If $M_1 = 0$, then (2 8) is satisfied for any $k_2 > M_2$. This coincides with the conditions obtained through the weak Lyapunov function [18].

Remark 2.5. Hence, when $M_3 = 0$, i.e. without persistent perturbations, from lemma 2.5-3, the system (2 - 6) will have finite time convergence for 0 < b < 1, with the settling time estimate,

$$T(z_{10}, z_{20}) \leq \left(\frac{\overline{\pi}_1^{\frac{3+5b}{4+4b}}}{\omega_3}\right) \left(\frac{4+4b}{1-b}\right) [V(z_{10}, z_{20})]^{\frac{1-b}{4+4b}},$$

where (z_{10}, z_{20}) are the initial states of the system. Similarly, exponential and asymptotical convergence for a = 1 and a > 1 respectively, can be concluded from lemma 2.5 as well. The above results are possible due to the negative definiteness of the time derivative of the

Lyapunov function, i.e. strict Lyapunov function.

Remark 2.6. When persistent perturbations occur on the system, $M_3 \neq 0$, from (2 - 9)

$$\begin{split} \dot{\widetilde{V}}(z_{1}, z_{2}) &\leq -\frac{1}{2} \left(\frac{\omega_{3}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{3+5b}{4+4b}} - V^{\frac{3}{4}} \left(\frac{1}{2} \left(\frac{\omega_{3}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{b}{2+2b}} - M_{3} \left(\frac{2^{\frac{1}{4}} \omega_{2}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right) \\ &\leq -\frac{1}{2} \left(\frac{\omega_{3}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{3+5b}{4+4b}}, \text{ for } V \geq \left(2M_{3} \left(\frac{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}}{\omega_{3}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{2}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right)^{\frac{2+2b}{b}} \end{split}$$

hence, from lemma 2.6, the system (2 - 6) is uniformly ultimately bounded.

b) Case 2: For b = 0

For b = 0, V is not differentiable on $z_1 = 0$

$$\dot{\widetilde{V}}(z_1, z_2) = \bigcap_{\xi \in \partial V(\mathbf{z}(t), t)} \xi^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix} \subset \dot{\widetilde{V}}_1 + \dot{\widetilde{V}}_2$$

where

$$\dot{\widetilde{V}}_1 = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^T K[f](z_1, z_2), \quad \dot{\widetilde{V}}_2 = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^T K\begin{bmatrix}0\\d\end{bmatrix},$$

$$K[f](z_{1}, z_{2}) = \begin{cases} \begin{cases} z_{2}, \\ -k_{1} \operatorname{sign}(z_{1}) - k_{2} \operatorname{sign}(z_{2}) \end{cases} \forall z_{1} \neq 0, z_{2} \neq 0 \\ \begin{cases} z_{2}, \\ -k_{1}[-1, 1] - k_{2} \operatorname{sign}(z_{2}) \end{cases} \forall z_{1} = 0, z_{2} \neq 0 \\ \begin{cases} 0, \\ -k_{1} \operatorname{sign}(z_{1}) - k_{2}[-1, 1] \end{cases} \forall z_{1} \neq 0, z_{2} = 0 \\ \begin{cases} 0, \\ -k_{1} \operatorname{sign}(z_{1}) - k_{2}[-1, 1] \end{cases} \forall z_{1} = 0, z_{2} = 0 \end{cases}$$

$$\partial V = \mathbf{K} [\nabla V] = \mathbf{K} \begin{bmatrix} \frac{\partial V}{\partial z_1} \\ \frac{\partial V}{\partial z_2} \end{bmatrix} \\ \subset \begin{bmatrix} \mathbf{K} \begin{bmatrix} \frac{\partial V}{\partial z_1} \\ \frac{\partial V}{\partial z_1} \end{bmatrix} \\ \mathbf{K} \begin{bmatrix} \frac{\partial V}{\partial z_2} \end{bmatrix} \end{bmatrix} = \begin{cases} \left\{ \begin{pmatrix} 2k_1^2 z_1 + \frac{3}{2}r|z_1|^{\frac{1}{2}} z_2 + k_1 \operatorname{sign}(z_1)|z_2|^2 \\ |z_2|^3 \operatorname{sign}(z_2) + r|z_1|^{\frac{3}{2}} \operatorname{sign}(z_1) + 2k_1|z_1|z_2 \end{pmatrix} \right\} \forall z_1 \neq 0, z_2 \neq 0, \\ \left\{ \begin{pmatrix} [-1, 1]k_1 z_2^2 \\ z_2^{-3} \end{pmatrix} \right\} \forall z_1 = 0, z_2 \neq 0, \\ \left\{ \begin{pmatrix} 2k_1^2 z_1 \\ r|z_1|^{\frac{3}{2}} \operatorname{sign}(z_1) \end{pmatrix} \right\} \forall z_1 \neq 0, z_2 = 0, \\ \left\{ \begin{pmatrix} 0 \\ 0 \end{pmatrix} \right\} \forall z_1 = 0, z_2 = 0, \end{cases}$$

Let us define:

$$\left| \frac{\partial V}{\partial z_1} \right| := \sup \left\{ \left| \xi_1 \right| : \xi_1 \in \mathbf{K} \left[\frac{\partial V}{\partial z_1} \right] \right\}, \text{ and } \left| \frac{\partial V}{\partial z_2} \right| := \sup \left\{ \left| \xi_2 \right| : \xi_2 \in \mathbf{K} \left[\frac{\partial V}{\partial z_2} \right] \right\}, \text{ with} \\ \left| \frac{\partial V}{\partial z_2} \right| \le \left| z_2 \right|^3 + r \left| z_1 \right|^{\frac{3}{2}} + 2k_1 \left| z_1 \right| \left| z_2 \right|$$

Thus, the term

$$\dot{\widetilde{V}}_{2} = \bigcap_{\xi_{2} \in \mathbf{K}\left[\frac{\partial V}{\partial z_{2}}\right]} \xi_{2} \mathbf{K}\left[d\right] \le \left|\frac{\partial V}{\partial z_{2}}\right| d \le M\left(\left|z_{2}\right|^{3} + r\left|z_{1}\right|^{\frac{3}{2}} + 2k_{1}\left|z_{1}\right| \left|z_{2}\right|\right)$$

where M be defined as, $M := M_1 + M_2 + M_3$, since for b = 0, $|d| \le M_1 + M_2 + M_3$.

Computing $\dot{\tilde{V}}_1$ for each case, we have

For $z_1 \neq 0$ and $z_2 \neq 0$:

$$\dot{\widetilde{V}}_{1} = -2k_{1}k_{2}|z_{1}||z_{2}| - k_{1}r|z_{1}|^{\frac{3}{2}} - k_{2}r|z_{1}|^{\frac{3}{2}}\operatorname{sign}(z_{1}z_{2}) - k_{2}|z_{2}|^{3} + \frac{3}{2}r|z_{1}|^{\frac{1}{2}}|z_{2}|^{2}$$

For $z_1 = 0$ and $z_2 \neq 0$: Let $(\xi_2 k_1 z_2^2, z_2^3)^T$ with $\xi_2 \in [-1, 1]$ be an arbitrary element of $\partial V(\mathbf{z}, t)$, then

$$\xi^{T}K[f](z_{1},z_{2}) = [\xi_{2}-1, \xi_{2}+1]k_{1}z_{2}^{3}-k_{2}|z_{2}|^{3}$$

hence

$$\dot{\vec{V}}_{1} = \bigcap_{\xi_{2} \in [-1, 1]} \left[\xi_{2} - 1, \xi_{2} + 1 \right] k_{1} z_{2}^{3} - k_{2} |z_{2}|^{3} = -k_{2} |z_{2}|^{3}$$

For $z_1 \neq 0$ and $z_2 = 0$:

$$\dot{\tilde{V}}_1 = -k_1 r |z_1|^{\frac{3}{2}} - [-1, 1] k_2 r |z_1|^{\frac{3}{2}} \operatorname{sign}(z_1)$$

For $z_1 = 0$ and $z_2 = 0$: $\dot{\tilde{V}}_1 = 0$.

Thus, for all $(z_1, z_2) \in \mathbb{R}^2$

$$\dot{\tilde{V}}_{1} = -2k_{1}k_{2}|z_{1}||z_{2}| - k_{1}r|z_{1}|^{\frac{3}{2}} - k_{2}r|z_{1}|^{\frac{3}{2}}\operatorname{sign}(z_{1})\operatorname{SGN}(z_{2}) - k_{2}|z_{2}|^{3} + \frac{3r}{2}|z_{1}|^{\frac{1}{2}}|z_{2}|^{2}$$

Hence, the generalized time derivative of the Lyapunov function, after rearrangement:

$$\dot{\widetilde{V}} = \dot{\widetilde{V}}_1 + \dot{\widetilde{V}}_2 \le -2k_1(k_2 - M)|z_1||z_2| - r(k_1 - k_2 - M)|z_1|^{\frac{3}{2}} - (k_2 - M)|z_2|^3 + \frac{3r}{2}|z_1|^{\frac{1}{2}}|z_2|^2$$

Applying lemma 2.1,

$$-|z_1|^{\frac{3}{2}}-|z_2|^{3} \le -3^{\frac{1}{3}}|z_1|^{\frac{1}{2}}\left(\frac{3}{2}\right)^{\frac{2}{3}}|z_2|^{2}$$

Thus, if the following inequality holds

$$\frac{\left(k_1 - k_2 - M\right)^{\frac{1}{2}} \left(k_2 - M\right)}{2} > r \tag{2-10}$$

employing lemma 2.2 and the bounds on the Lyapunov function, then

$$\dot{\tilde{V}} \leq -\omega_4 \left(\left| z_1 \right|^{\frac{3}{2}} + \left| z_2 \right|^{\frac{3}{2}} \right) = -\omega_4 \left(\left\| z_1 \right\|^2 \right)^{\frac{3}{4}} + \left\| z_2 \right\|^4 \right)^{\frac{3}{4}} \right) \leq -\omega_4 \left\| z_1 \right\|^2 + \left| z_2 \right\|^4 \right)^{\frac{3}{4}} \leq -\left(\frac{\omega_4}{\overline{\pi_1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}}$$
where $\omega_4 = \min\left\{ \frac{r\left(k_1 - k_2 - M\right)}{2}, \frac{\left(k_2 - M\right)}{2}\right\}.$

Remark 2.7. Note that the nonlinear inequalities (2 - 7) and (2 - 10) can be satisfied for any $k_1 > k_2 + M$, and $k_2 > M$, which guarantees the existence of an r > 0. The conditions on the gains obtained here for the special case of b = 0 coincides with [20]. Utilising lemma 2.5-3, the system (2 - 6) will have finite time convergence for b = 0, with the settling time estimate,

$$T(z_{10}, z_{20}) \leq \left(\frac{4\overline{\pi}_1^{\frac{3}{4}}}{\omega_4}\right) [V(z_{10}, z_{20})]^{\frac{1}{4}}, \text{ where } (z_{10}, z_{20}) \text{ are the initial states of the system.}$$

2.3 Generic Twisting

Leveraging the results of section 2.2.1, a generic twisting based family of controllers is presented. Essentially, the algorithm comprise linear sum of two system that have different homogeneity (hence different convergence properties), thus yielding a family of non-homogeneous algorithms. By means of strict Lyapunov functions, the family of algorithms are shown to exhibit the properties of their individual components, i.e. finite-time and uniform convergence, thus, yielding uniform finite-time convergence that is independent of initial conditions of the system.

2.3.1 System description

Consider the following generic twisting based dynamics:

$$z_{1} = z_{2},$$

$$\dot{z}_{2} = -k_{1}|z_{1}|^{2p-1}\operatorname{sign}(z_{1}) - k_{1n}|z_{1}|^{2q-1}\operatorname{sign}(z_{1}) - k_{2}|z_{2}|^{2-\frac{1}{p}}\operatorname{sign}(z_{2}) - k_{2n}|z_{2}|^{2-\frac{1}{q}}\operatorname{sign}(z_{2}) + d \qquad (2-11)$$

where $z_1, z_2 \in \mathbb{R}$, are the scalar state variables, k_1, k_{1n}, k_2, k_{2n} are positive constants, $0.5 \le p \le 1$, and $1 \le q$. Also, *d* is time-varying and/or nonlinear term of uncertainty bounded by:

$$|d| \le M_1 |z_1|^{2p-1} + M_2 |z_1|^{2q-1} + M_3 |z_2|^{2-\frac{1}{p}} + M_4 |z_2|^{2-\frac{1}{q}} + M_5 \text{ with } |d| := \sup\{|\delta| : \delta \in \mathbf{K}[d]\},$$

where $M_1 \ge 0$, $M_2 \ge 0$, $M_3 \ge 0$, $M_4 \ge 0$, and $M_5 \ge 0$ with the same *p* and *q* as that in (2 - 11). Note that for the case of p = 0.5, the algorithm contains discontinuity and the uncertainty is upper bounded by non-vanishing constant.

2.3.2 Stability analysis

The system above (2 - 11) is essentially a summation of two different degree-ofhomogeneity of the twisting-based algorithm considered in section 2.2.1. Hence, a Lyapunov function, which consists of linear summation of two different degree-of-homogeneity Lyapunov functions based on section 2.2.1, is proposed for system (2 - 11).

$$V(z_1, z_2) = \left(\frac{1}{2}z_2^2 + \frac{k_1}{2p}|z_1|^{2p} + \frac{k_{1n}}{2q}|z_1|^{2q}\right)^2 + r|z_1|^{3p}|z_2|\operatorname{sign}(z_1z_2) + r|z_1|^{3q}|z_2|\operatorname{sign}(z_1z_2)$$

where *r* is a positive constant scalar. Note that the above function is locally Lipschitz and regular. It is differentiable everywhere for p > 0.5, and not differentiable on $z_1 = 0$ for p = 0.5.

Note that the Lyapunov function V can be lower bounded by,

$$V(z_{1}, z_{2}) \geq \frac{1}{4} |z_{2}|^{4} + \frac{k_{1}^{2}}{4p^{2}} |z_{1}|^{4p} + \frac{k_{1n}^{2}}{4q^{2}} |z_{1}|^{4q} + \frac{k_{1}}{2p} |z_{1}|^{2p} |z_{2}|^{2} + \frac{k_{1n}}{2q} |z_{1}|^{2q} |z_{2}|^{2} + \frac{k_{1}k_{1n}}{2pq} |z_{1}|^{2p+2q} - r|z_{1}|^{3p} |z_{2}| - r|z_{1}|^{3q} |z_{2}|$$

Using lemma 2.1,

$$|z_{1}|^{4p} + |z_{2}|^{4} \ge \left(\frac{4}{3}\right)^{\frac{3}{4}} |z_{1}|^{3p} (4)^{\frac{1}{4}} |z_{2}|$$
$$|z_{1}|^{4q} + |z_{2}|^{4} \ge \left(\frac{4}{3}\right)^{\frac{3}{4}} |z_{1}|^{3q} (4)^{\frac{1}{4}} |z_{2}|$$

Hence, for

$$\min\left\{ \left(\frac{k_1^2}{6p^2}\right)^{\frac{3}{4}} \left(\frac{1}{3}\right)^{\frac{1}{4}}, \left(\frac{k_{1n}^2}{6q^2}\right)^{\frac{3}{4}} \left(\frac{1}{3}\right)^{\frac{1}{4}} \right\} > r$$
 (2 - 12)

 $V(z_1, z_2) \ge \frac{1}{12} |z_2|^4 + \frac{k_1^2}{8p^2} |z_1|^{4p} + \frac{k_{1n}^2}{8q^2} |z_1|^{4q} \ge \underline{\pi}_1 \left(|z_1|^{4p} + |z_1|^{4q} + |z_2|^4 \right)$

where $\underline{\pi}_1 = \min\left\{\frac{1}{12}, \frac{k_1^2}{8p^2}, \frac{k_{1n}^2}{8q^2}\right\}.$

Also, it can be upper-bounded by,

$$V(z_{1}, z_{2}) \leq \frac{1}{4} |z_{2}|^{4} + \frac{k_{1}^{2}}{4p^{2}} |z_{1}|^{4p} + \frac{k_{1n}^{2}}{4q^{2}} |z_{1}|^{4q} + \frac{k_{1}}{2p} |z_{1}|^{2p} |z_{2}|^{2} + \frac{k_{1n}}{2q} |z_{1}|^{2q} |z_{2}|^{2} + \frac{k_{1}k_{1n}}{2pq} |z_{1}|^{2p+2q} + r|z_{1}|^{3p} |z_{2}| + r|z_{1}|^{3q} |z_{2}|$$

Using lemma 2.1,

$$\begin{aligned} \frac{k_1}{2p} |z_1|^{2p} |z_2|^2 &\leq \frac{k_1}{4p} |z_1|^{4p} + \frac{k_1}{4p} |z_2|^4, \\ \frac{k_{1n}}{2q} |z_1|^{2q} |z_2|^2 &\leq \frac{k_{1n}}{4q} |z_1|^{4q} + \frac{k_{1n}}{4q} |z_2|^4, \\ \frac{k_1 k_{1n}}{2pq} |z_1|^{2p+2q} &\leq \frac{k_1 k_{1n}}{4pq} |z_1|^{4p} + \frac{k_1 k_{1n}}{4pq} |z_1|^{4q}, \\ r |z_1|^{3p} |z_2| &\leq \frac{3r}{4} |z_1|^{4p} + \frac{r}{4} |z_2|^4, \\ r |z_1|^{3q} |z_2| &\leq \frac{3r}{4} |z_1|^{4q} + \frac{r}{4} |z_2|^4. \end{aligned}$$

Then,

$$V(z_1, z_2) \le \overline{\pi}_1 \left(|z_1|^{4p} + |z_1|^{4q} + |z_2|^4 \right)$$

where

$$\overline{\pi}_{1} = \max\left\{ \left(\frac{k_{1}^{2}}{4p^{2}} + \frac{k_{1}}{4p} + \frac{k_{1}k_{1n}}{4pq} + \frac{3r}{4} \right), \left(\frac{k_{1n}^{2}}{4q^{2}} + \frac{k_{1n}}{4q} + \frac{k_{1}k_{1n}}{4pq} + \frac{3r}{4} \right), \left(\frac{1}{4} + \frac{k_{1}}{4p} + \frac{k_{1n}}{4q} + \frac{r}{4} + \frac{r}{4} \right) \right\}$$

Thus, *V* is positive definite and radially unbounded. Since (2 - 11) is a differential equation that has discontinuous right-hand side, i.e. when p = 0.5 and since no continuity assumption is made on *d*, its solution are understood in the sense of Filippov (see definition 2.1). According to lemma 2.4, the time derivative of non-smooth Lyapunov function exists almost everywhere

$$\frac{d}{dt}V(\mathbf{z}(t),t) \stackrel{a.e.}{\in} \overset{\cdot}{\widetilde{V}}(\mathbf{z}(t),t) = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix} (\mathbf{z},t).$$

Since the controller (2 - 11) is discontinuous when p = 0.5, for ease of presentation, the analysis is separated for the case of p > 0.5 and p = 0.5.

a) Case 1: For
$$0.5 , and $1 \le q$$$

Note that for p > 0.5, V is continuously differentiable, hence

$$\begin{split} \hat{V}(z_{1},z_{2}) &= \bigcap_{\xi \in \partial V(z(t),t)} \xi^{T} K \begin{bmatrix} \dot{z}_{1} \\ \dot{z}_{2} \end{bmatrix} = \nabla V^{T} K \begin{bmatrix} \dot{z}_{1} \\ \dot{z}_{2} \end{bmatrix} \\ &\subset -\frac{k_{1}k_{2}}{p} |z_{1}|^{2p} |z_{2}|^{3-\frac{1}{p}} - \frac{k_{1n}k_{2}}{q} |z_{1}|^{2q} |z_{2}|^{3-\frac{1}{p}} - \frac{k_{1}k_{2n}}{p} |z_{1}|^{2p} |z_{2}|^{3-\frac{1}{q}} \\ &- \frac{k_{1n}k_{2n}}{q} |z_{1}|^{2q} |z_{2}|^{3-\frac{1}{q}} - rk_{1n} |z_{1}|^{2q+3p-1} - rk_{1} |z_{1}|^{2p+3q-1} - k_{2} |z_{2}|^{5-\frac{1}{p}} \\ &- k_{2n} |z_{2}|^{5-\frac{1}{q}} - rk_{1} |z_{1}|^{5p-1} - rk_{1n} |z_{1}|^{5q-1} - rk_{2} |z_{1}|^{3p} |z_{2}|^{2-\frac{1}{p}} \operatorname{sign}(z_{1}z_{2}) \\ &- rk_{2n} |z_{1}|^{3p} |z_{2}|^{2-\frac{1}{q}} \operatorname{sign}(z_{1}z_{2}) + r3p |z_{1}|^{3p-1} |z_{2}|^{2} \\ &- rk_{2} |z_{1}|^{3q} |z_{2}|^{2-\frac{1}{p}} \operatorname{sign}(z_{1}z_{2}) - rk_{2n} |z_{1}|^{3q} |z_{2}|^{2-\frac{1}{q}} \operatorname{sign}(z_{1}z_{2}) + r3q |z_{1}|^{3q-1} |z_{2}|^{2} \\ &+ K [d \begin{bmatrix} |z_{2}|^{3} \operatorname{sign}(z_{2}) + \frac{k_{1}}{p} |z_{1}|^{2p} z_{2} + \frac{k_{1n}}{q} |z_{1}|^{2q} z_{2} \\ &+ r |z_{1}|^{3p} \operatorname{sign}(z_{1}) + r |z_{1}|^{3q} \operatorname{sign}(z_{1}) \end{bmatrix} \end{split}$$

After rearrangement,

$$\begin{split} \dot{\tilde{V}}(z_{1},z_{2}) &\leq -\frac{k_{1}}{p}(k_{2}-M_{3})|z_{1}|^{2p}|z_{2}|^{3-\frac{1}{p}} - \frac{k_{1n}}{q}(k_{2}-M_{3})|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{p}} \\ &\quad -\frac{k_{1}}{p}(k_{2n}-M_{4})|z_{1}|^{2p}|z_{2}|^{3-\frac{1}{q}} - \frac{k_{1n}}{q}(k_{2n}-M_{4})|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{q}} \\ &\quad -r(k_{1n}-M_{2})|z_{1}|^{3p+2q-1} - r(k_{1}-M_{1})|z_{1}|^{2p+3q-1} - \frac{(k_{2}-M_{3})}{2}|z_{2}|^{5-\frac{1}{p}} \\ &\quad -\frac{(k_{2n}-M_{4})}{2}|z_{2}|^{5-\frac{1}{q}} - \frac{r(k_{1}-M_{1})}{2}|z_{1}|^{5p-1} - \frac{r(k_{1n}-M_{2})}{2}|z_{1}|^{5q-1} \\ &\quad +\dot{V}_{1} + M_{5}\left(|z_{2}|^{3} + \frac{k_{1}}{p}|z_{1}|^{2p}|z_{2}| + \frac{k_{1n}}{q}|z_{1}|^{2q}|z_{2}| + r|z_{1}|^{3p} + r|z_{1}|^{3q}\right) \end{split}$$

where

$$\begin{split} \dot{V_{1}} &= -\frac{\left(k_{2}-M_{3}\right)}{2} |z_{2}|^{5-\frac{1}{p}} - \frac{\left(k_{2n}-M_{4}\right)}{2} |z_{2}|^{5-\frac{1}{q}} - \frac{r\left(k_{1}-M_{1}\right)}{2} |z_{1}|^{5p-1} - \frac{r\left(k_{1n}-M_{2}\right)}{2} |z_{1}|^{5q-1} \\ &+ r\left(k_{2}+M_{3}\right) |z_{1}|^{3p} |z_{2}|^{2-\frac{1}{p}} + r\left(k_{2n}+M_{4}\right) |z_{1}|^{3p} |z_{2}|^{2-\frac{1}{q}} + r3p |z_{1}|^{3p-1} |z_{2}|^{2} \\ &+ r\left(k_{2}+M_{3}\right) |z_{1}|^{3q} |z_{2}|^{2-\frac{1}{p}} + r\left(k_{2n}+M_{4}\right) |z_{1}|^{3q} |z_{2}|^{2-\frac{1}{q}} + r3q |z_{1}|^{3q-1} |z_{2}|^{2} \\ &+ M_{1} |z_{1}|^{2p-1} |z_{2}|^{3} + \frac{k_{1}}{p} M_{1} |z_{1}|^{4p-1} |z_{2}| + \left(M_{1}\frac{k_{1n}}{q} + M_{2}\frac{k_{1}}{p}\right) |z_{1}|^{2p+2q-1} |z_{2}| \\ &+ M_{2} |z_{1}|^{2q-1} |z_{2}|^{3} + M_{2}\frac{k_{1n}}{q} |z_{1}|^{4q-1} |z_{2}| \end{split}$$

Applying lemma 2.1,

$$\begin{split} &-|z_{1}|^{5p-1}-|z_{2}|^{5-\frac{1}{p}}\leq-\left(\frac{5p-1}{3p}\right)^{\frac{3p}{5p-1}}|z_{1}|^{3p}\left(\frac{5p-1}{2p-1}\right)^{\frac{2p-1}{5p-1}}|z_{2}|^{2-\frac{1}{p}},\\ &-|z_{1}|^{5p-1}-|z_{2}|^{5-\frac{1}{p}}\leq-\left(\frac{5p-1}{3p-1}\right)^{\frac{3p-1}{5p-1}}|z_{1}|^{3p-1}\left(\frac{5p-1}{2p}\right)^{\frac{2p}{5p-1}}|z_{2}|^{2},\\ &-|z_{1}|^{5p-1}-|z_{2}|^{5-\frac{1}{p}}\leq-\left(\frac{5p-1}{2p-1}\right)^{\frac{2p-1}{5p-1}}|z_{1}|^{2p-1}\left(\frac{5p-1}{3p}\right)^{\frac{3p}{5p-1}}|z_{2}|^{3},\\ &-|z_{1}|^{5p-1}-|z_{2}|^{5-\frac{1}{p}}\leq-\left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}}|z_{1}|^{4p-1}\left(\frac{5p-1}{p}\right)^{\frac{p}{5p-1}}|z_{2}|,\\ &-|z_{1}|^{5q-1}-|z_{2}|^{5-\frac{1}{q}}\leq-\left(\frac{5q-1}{3q}\right)^{\frac{3q}{5q-1}}|z_{1}|^{3q}\left(\frac{5q-1}{2q-1}\right)^{\frac{2q-1}{5q-1}}|z_{2}|^{2-\frac{1}{q}},\\ &-|z_{1}|^{5q-1}-|z_{2}|^{5-\frac{1}{q}}\leq-\left(\frac{5q-1}{3q}\right)^{\frac{3q-1}{5q-1}}|z_{1}|^{3q-1}\left(\frac{5q-1}{2q}\right)^{\frac{2q-1}{5q-1}}|z_{2}|^{2-\frac{1}{q}}, \end{split}$$

$$\begin{split} &-|z_1|^{5q-1} - |z_2|^{5-\frac{1}{q}} \leq -\left(\frac{5q-1}{2q-1}\right)^{\frac{2q-1}{3q-1}} |z_1|^{2q-1} \left(\frac{5q-1}{3q}\right)^{\frac{3q}{5q-1}} |z_2|^3, \\ &-|z_1|^{5q-1} - |z_2|^{5-\frac{1}{q}} \leq -\left(\frac{5q-1}{4q-1}\right)^{\frac{4q-1}{5q-1}} |z_1|^{4q-1} \left(\frac{5q-1}{q}\right)^{\frac{2q-1}{5q-1}} |z_2|, \\ &-|z_1|^{5p-1} - |z_1|^{5q-1} - |z_2|^{5-\frac{1}{q}} \leq -|z_1|^{5p-\frac{p}{q}} - |z_2|^{5-\frac{1}{q}} \leq -\left(\frac{5q-1}{3q}\right)^{\frac{3q}{5q-1}} |z_1|^{3p} \left(\frac{5q-1}{2q-1}\right)^{\frac{2q-1}{5q-1}} |z_2|^{2-\frac{1}{q}}, \\ &-|z_1|^{5p-1} - |z_1|^{5q-1} - |z_2|^{5-\frac{1}{p}} \leq |z_1|^{5q-\frac{p}{q}} - |z_2|^{5-\frac{1}{q}} \leq -\left(\frac{5p-1}{3p}\right)^{\frac{3p}{5q-1}} |z_1|^{3q} \left(\frac{5p-1}{2p-1}\right)^{\frac{2p-1}{5p-1}} |z_2|^{2-\frac{1}{p}}, \\ &-|z_1|^{5p-1} - |z_1|^{5q-1} - |z_2|^{5-\frac{1}{p}} \leq |z_1|^{5q-\frac{q}{p}} - |z_2|^{5-\frac{1}{p}} \leq -\left(\frac{5p-1}{3p}\right)^{\frac{3p}{5p-1}} |z_1|^{3q} \left(\frac{5p-1}{2p-1}\right)^{\frac{2p-1}{5p-1}} |z_2|^{2-\frac{1}{p}}, \\ &-|z_1|^{5p-1} - |z_1|^{5q-1} - |z_2|^{5-\frac{1}{p}} \leq |z_1|^{(2p+2q-1)} \left(\frac{5p-1}{4p-1}\right) - |z_2|^{5-\frac{1}{p}} \\ &\leq -\left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}} |z_1|^{2p+2q-1} \left(\frac{5p-1}{p}\right)^{\frac{p}{5p-1}} |z_2|, \end{split}$$

where lemma 2.3 has been employed in the last three inequalities since

$$5p-1 \le (2p+2q-1)\left(\frac{5p-1}{4p-1}\right) \le 5q-1, \ 5p-1 \le 5q-\frac{q}{p} \le 5q-1, \text{ and}$$

$$5p-1 \le 5p - \frac{p}{q} \le 5q - 1,$$

Thus, if the following inequalities
$$\min \begin{cases} \lambda_{1} \frac{(k_{1} - M_{1})^{\frac{3p}{2p-1}}(k_{2} - M_{3})}{(k_{2} + M_{3})^{\frac{5p-1}{2p-1}}}, \\ \lambda_{2}(k_{1} - M_{1})^{\frac{3p-1}{2p}}(k_{2} - M_{3}), \\ \lambda_{3} \frac{(k_{1n} - M_{2})^{\frac{3q}{2q-1}}(k_{2n} - M_{4})}{(k_{2n} + M_{4})^{\frac{5q-1}{2q-1}}}, \\ \lambda_{4}(k_{1n} - M_{2})^{\frac{3q-1}{2q}}(k_{2n} - M_{4}), \\ \lambda_{5} \frac{\omega_{1}^{\frac{3q}{2q-1}}(k_{2n} - M_{4})}{(k_{2n} + M_{4})^{\frac{5q-1}{2q-1}}}, \\ \lambda_{6} \frac{\omega_{1}^{\frac{3p}{2q-1}}(k_{2n} - M_{4})}{(k_{2} + M_{3})^{\frac{5q-1}{2q-1}}}, \\ \lambda_{6} \frac{\omega_{1}^{\frac{3p}{2q-1}}(k_{2} - M_{3})}{(k_{2} + M_{3})^{\frac{5q-1}{2p-1}}} \end{cases} \right\} > r > \max \begin{cases} \lambda_{7} \frac{M_{1}^{\frac{5p-1}{2p-1}}}{(k_{1} - M_{1})(k_{2} - M_{3})^{\frac{3p}{2p-1}}}, \\ \lambda_{8} \frac{k_{1}^{\frac{5p-1}{4p-1}}M_{1}^{\frac{5p-1}{4p-1}}}{(k_{1} - M_{1})(k_{2} - M_{3})^{\frac{3q}{2p-1}}}, \\ \lambda_{9} \frac{M_{2}^{\frac{5q-1}{2q-1}}}{(k_{1n} - M_{2})(k_{2n} - M_{4})^{\frac{3q}{2q-1}}}, \\ \lambda_{10} \frac{k_{1n}^{\frac{5q-1}{4q-1}}M_{2}^{\frac{5q-1}{4q-1}}}{(k_{1n} - M_{2})(k_{2n} - M_{4})^{\frac{3q}{4q-1}}}, \\ \lambda_{10} \frac{(M_{1} \frac{k_{1n}}{q} + M_{2} \frac{k_{1}}{p})^{\frac{5p-1}{4p-1}}}{(k_{1n} - M_{2})(k_{2n} - M_{4})^{\frac{4q-1}{4p-1}}}, \\ \lambda_{11} \frac{(M_{1} \frac{k_{1n}}{q} + M_{2} \frac{k_{1}}{p})^{\frac{5p-1}{4p-1}}}}{(k_{1n} - M_{2})^{\frac{4p-1}{4p-1}}}, \end{cases}$$

$$(2 - 13)$$

$$\begin{split} \lambda_{1} &= \left(\frac{5p-1}{42p}\right)^{\frac{3p}{2p-1}} \left(\frac{5p-1}{24p-12}\right), \quad \lambda_{2} &= \left(\frac{5p-1}{42p-14}\right)^{\frac{3p-1}{2p}} \left(\frac{5p-1}{24p}\right) \left(\frac{1}{3p}\right)^{\frac{5p-1}{2p}}, \\ \lambda_{3} &= \left(\frac{5q-1}{42q}\right)^{\frac{3q}{2q-1}} \left(\frac{5q-1}{20q-10}\right), \quad \lambda_{4} &= \left(\frac{5q-1}{42q-14}\right)^{\frac{3q-1}{2q}} \left(\frac{5q-1}{20q}\right) \left(\frac{1}{3q}\right)^{\frac{5q-1}{2q}}, \\ \lambda_{5} &= \left(\frac{5q-1}{42q}\right)^{\frac{3q}{2q-1}} \left(\frac{5q-1}{20q-10}\right), \quad \lambda_{6} &= \left(\frac{5p-1}{42p}\right)^{\frac{3p}{2p-1}} \left(\frac{5p-1}{24p-12}\right), \\ \lambda_{7} &= \left(\frac{28p-14}{5p-1}\right) \left(\frac{36p}{5p-1}\right)^{\frac{3p}{2p-1}}, \quad \lambda_{8} &= \left(\frac{56p-14}{5p-1}\right) \left(\frac{12p}{5p-1}\right)^{\frac{p}{4p-1}} \left(\frac{1}{p}\right)^{\frac{5p-1}{4p-1}}, \\ \lambda_{9} &= \left(\frac{28q-14}{5q-1}\right) \left(\frac{30q}{5q-1}\right)^{\frac{3q}{2q-1}}, \quad \lambda_{10} &= \left(\frac{56q-14}{5q-1}\right) \left(\frac{10q}{5q-1}\right)^{\frac{q}{4q-1}} \left(\frac{1}{q}\right)^{\frac{5q-1}{4q-1}}, \\ \lambda_{11} &= \left(\frac{56p-14}{5p-1}\right) \left(\frac{12p}{5p-1}\right)^{\frac{p}{4p-1}}, \quad \omega_{1} &= \min\{(k_{1}-M_{1}), (k_{1n}-M_{2})\} \end{split}$$

hold then function \dot{V}_1 is negative definite. Then,

$$\begin{split} \dot{\tilde{V}}(z_{1},z_{2}) &\leq -\frac{\left(k_{2}-M_{3}\right)}{2} |z_{2}|^{5-\frac{1}{p}} - \frac{\left(k_{2n}-M_{4}\right)}{2} |z_{2}|^{5-\frac{1}{q}} - \frac{r\left(k_{1}-M_{1}\right)}{2} |z_{1}|^{5p-1} - \frac{r\left(k_{1n}-M_{2}\right)}{2} |z_{1}|^{5q-1} \\ &+ M_{5} \left(\left|z_{2}\right|^{3} + \frac{k_{1}}{p} |z_{1}|^{2p} |z_{2}| + \frac{k_{1n}}{q} |z_{1}|^{2q} |z_{2}| + r|z_{1}|^{3p} + r|z_{1}|^{3q} \right) \end{split}$$

Applying lemma 2.1,

$$\begin{split} & \left| z_1 \right|^{2p} \left| z_2 \right| \leq \frac{2}{3} \left| z_1 \right|^{3p} + \frac{1}{3} \left| z_2 \right|^3, \\ & \left| z_1 \right|^{2q} \left| z_2 \right| \leq \frac{2}{3} \left| z_1 \right|^{3q} + \frac{1}{3} \left| z_2 \right|^3, \end{split}$$

and lemma 2.3:

$$\begin{split} &-\left|z_{1}\right|^{5p-1}-\left|z_{1}\right|^{5q-1}\leq-\left|z_{1}\right|^{5p-\frac{p}{q}},\\ &-\left|z_{1}\right|^{5p-1}-\left|z_{1}\right|^{5q-1}\leq-\left|z_{1}\right|^{5q-\frac{q}{p}}, \end{split}$$

together with lemma 2.2, and the bounds of the Lyapunov function, we have:

$$\begin{split} \dot{\tilde{V}}(z_{1}, z_{2}) &\leq -\omega_{2} \left(\left| z_{1} \right|^{5p-1} + \left| z_{1} \right|^{5q-\frac{q}{p}} + \left| z_{2} \right|^{5-\frac{1}{p}} \right) - \omega_{3} \left(\left| z_{1} \right|^{5q-1} + \left| z_{1} \right|^{5p-\frac{p}{q}} + \left| z_{2} \right|^{5-\frac{1}{q}} \right) \\ &+ M_{5} \omega_{4} \left(\left| z_{2} \right|^{3} + \left| z_{1} \right|^{3p} + \left| z_{1} \right|^{3q} \right) \\ &= -\omega_{2} \left(\left(\left| z_{1} \right|^{4p} \right)^{\frac{5p-1}{4p}} + \left(\left| z_{1} \right|^{4q} \right)^{\frac{5p-1}{4p}} + \left(\left| z_{2} \right|^{4} \right)^{\frac{5p-1}{4p}} \right) \\ &- \omega_{3} \left(\left(\left| z_{1} \right|^{4q} \right)^{\frac{5q-1}{4q}} + \left(\left| z_{1} \right|^{4p} \right)^{\frac{5q-1}{4q}} + \left(\left| z_{2} \right|^{4} \right)^{\frac{5q-1}{4q}} \right) \\ &+ M_{5} \omega_{4} \left(\left(\left| z_{2} \right|^{4} \right)^{\frac{3}{4}} + \left(\left| z_{1} \right|^{4p} \right)^{\frac{3}{4}} + \left(\left| z_{1} \right|^{4q} \right)^{\frac{3}{4}} \end{split}$$

$$\leq -\left(\frac{\omega_{2}}{\bar{\pi}_{1}^{\frac{5p-1}{4p}}}\right)V^{\frac{5p-1}{4p}} - \left(\frac{\omega_{3}}{\frac{q-1}{3^{\frac{q-1}{4q}}\bar{\pi}_{1}^{\frac{5q-1}{4q}}}}\right)V^{\frac{5q-1}{4q}} + M_{5}\left(\frac{3^{\frac{1}{4}}\omega_{4}}{\underline{\pi}_{1}^{\frac{3}{4}}}\right)V^{\frac{3}{4}}$$
(2 - 14)

$$\omega_{2} = \min\left\{\frac{r(k_{1} - M_{1})}{4}, \frac{r}{8}\omega_{1}, \frac{(k_{2} - M_{3})}{2}\right\}\omega_{3} = \min\left\{\frac{r(k_{1n} - M_{2})}{4}, \frac{r}{8}\omega_{1}, \frac{(k_{2n} - M_{4})}{2}\right\},\\ \omega_{4} = \max\left\{\left(r + \frac{2k_{1}}{3p}\right), \left(r + \frac{2k_{1n}}{3q}\right), \left(1 + \frac{k_{1}}{3p} + \frac{k_{1n}}{3q}\right)\right\}$$

Remark 2.8. Note that the nonlinear inequalities (2 - 12) and (2 - 13) are feasible for sufficiently large $k_1 > 0$, $k_{1n} > 0$, $k_2 > 0$, $k_{2n} > 0$. Thus, an r > 0 always exists. Consider some particular cases.

- 1. Note that (2 13) is feasible with respect to k_1 , k_{1n} , k_2 , k_{2n} for any $M_1 \ge 0$, $M_2 \ge 0$, $M_3 \ge 0$, and $M_4 \ge 0$.
- 2. If $M_1 = M_2 = 0$, then (2 13) is satisfied for any $k_2 > M_3$ and $k_{2n} > M_4$. This coincides with the conditions obtained through the weak Lyapunov function, i.e. the energy function.

Remark 2.9. Hence, when $M_5 = 0$, i.e. without persistent perturbations, from lemma 2.5-3,

1. For 0 , and <math>q > 1 the system exhibit uniform asymptotical convergence, where

$$T_{\max}(\mu) = \left(\frac{3^{\frac{q-1}{4q}} \bar{\pi}_1 \frac{5q-1}{4q}}{\omega_3}\right) \left(\frac{4q}{q-1}\right) \left(\frac{1}{\mu^{\frac{q-1}{4q}}}\right),$$

is the time at which the trajectories reach the surface level $V = \mu$. At the same time, the system (2 - 11) will have finite time convergence, in particular after reaching the surface level $V = \mu$, from lemma 2.5-3, the settling time estimate,

$$T(\mu) \leq \left(\frac{\overline{\pi}_1^{\frac{5p-1}{4p}}}{\omega_2}\right) \left(\frac{4p}{1-p}\right) [\mu]^{\frac{1-p}{4p}},$$

where the initial starting states is changed to $V = \mu$. Hence, the total time to reach the origin can be estimated as

$$T_{total}(\mu) = \left(\frac{3^{\frac{q-1}{4q}} \overline{\pi}_1^{\frac{5q-1}{4q}}}{\omega_3}\right) \left(\frac{4q}{q-1}\right) \left(\frac{1}{\mu^{\frac{q-1}{4q}}}\right) + \left(\frac{\overline{\pi}_1^{\frac{5p-1}{4p}}}{\omega_2}\right) \left(\frac{4p}{1-p}\right) \left[\mu\right]^{\frac{1-p}{4p}}.$$

The minimum of this function can be found at
$$\mu = \left(\frac{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}{\omega_3}\right)^{\frac{4pq}{q-p}} \left(\frac{\omega_2}{\overline{\pi}_1 \frac{5p-1}{4p}}\right)^{\frac{4pq}{q-p}}.$$

Substituting into the function, a finite settling time independent of initial conditions is obtained,

$$\begin{split} T_{total} = & \left(\frac{3^{\frac{q-1}{4q}} \bar{\pi}_1 \frac{5q-1}{4q}}{\omega_3}\right) \left(\frac{4q}{q-1}\right) \left(\frac{\omega_3}{3^{\frac{q-1}{4q}} \bar{\pi}_1 \frac{5q-1}{4q}}\right)^{\frac{p(q-1)}{q-p}} \left(\frac{\bar{\pi}_1 \frac{5p-1}{4p}}{\omega_2}\right)^{\frac{p(q-1)}{q-p}} \\ &+ \left(\frac{\bar{\pi}_1 \frac{5p-1}{4p}}{\omega_2}\right) \left(\frac{4p}{1-p}\right) \left(\frac{3^{\frac{q-1}{4q}} \bar{\pi}_1 \frac{5q-1}{4q}}{\omega_3}\right)^{\frac{q(1-p)}{q-p}} \left(\frac{\omega_2}{\bar{\pi}_1 \frac{5p-1}{4p}}\right)^{\frac{q(1-p)}{q-p}} \end{split}$$

- 2. For 0 , and <math>q = 1, finite time convergence can be concluded.
- 3. For p = 1, and $q \ge 1$, exponential convergence can be concluded.

These results are possible due to the negative definiteness of the time derivative of the Lyapunov function, i.e. strict Lyapunov function.

Remark 2.10. While if persistent perturbations occur on the system, $M_5 \neq 0$, from (2 - 14),

$$\begin{split} \dot{\widetilde{V}}(z_{1},z_{2}) &\leq -\frac{1}{2} \left(\frac{\omega_{2}}{\overline{\pi}_{1}^{\frac{5p-1}{4p}}} \right) V^{\frac{5p-1}{4p}} - \left(\frac{\omega_{3}}{3^{\frac{q-1}{4q}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}} \right) V^{\frac{5q-1}{4q}} - V^{\frac{3}{4}} \left(\frac{1}{2} \left(\frac{\omega_{2}}{\overline{\pi}_{1}^{\frac{5p-1}{4p}}} \right) V^{\frac{2p-1}{4p}} - M_{5} \left(\frac{3^{\frac{1}{4}} \omega_{4}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right) \\ &\leq -\frac{1}{2} \left(\frac{\omega_{2}}{\overline{\pi}_{1}^{\frac{5p-1}{4p}}} \right) V^{\frac{5p-1}{4p}} - \left(\frac{\omega_{3}}{3^{\frac{q-1}{4q}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}} \right) V^{\frac{5q-1}{4q}}, \text{ for } V \geq \left(2M_{5} \left(\frac{\overline{\pi}_{1}^{\frac{5p-1}{4p}}}{\omega_{2}} \right) \left(\frac{3^{\frac{1}{4}} \omega_{4}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right)^{\frac{4p}{2p-1}} \end{split}$$

or

$$\begin{split} \dot{\tilde{V}}(z_{1}, z_{2}) &\leq -\left(\frac{\omega_{2}}{\overline{\pi}_{1}^{\frac{5p-1}{4p}}}\right) V^{\frac{5p-1}{4p}} - \frac{1}{2} \left(\frac{\omega_{3}}{\frac{q-1}{3}\frac{q-1}{4q}}\right) V^{\frac{5q-1}{4q}} \\ &- V^{\frac{3}{4}} \left(\frac{1}{2} \left(\frac{\omega_{3}}{\frac{q-1}{3}\frac{q-1}{4q}}\right) V^{\frac{2q-1}{4q}} - M_{5} \left(\frac{3^{\frac{1}{4}}\omega_{4}}{\underline{\pi}_{1}^{\frac{3}{4}}}\right)\right) \\ &\leq -\left(\frac{\omega_{2}}{\overline{\pi}_{1}\frac{5p-1}{4p}}\right) V^{\frac{5p-1}{4p}} - \frac{1}{2} \left(\frac{\omega_{3}}{\frac{q-1}{3}\frac{q-1}{4q}\frac{5q-1}{\pi_{1}^{\frac{5q-1}{4q}}}}\right) V^{\frac{5q-1}{4q}}, \end{split}$$
for $V \geq \left(2M_{5} \left(\frac{3^{\frac{q-1}{4q}}\overline{\pi}_{1}\frac{5q-1}{4q}}{\omega_{3}}\right) \left(\frac{3^{\frac{1}{4}}\omega_{4}}{\underline{\pi}_{1}^{\frac{3}{4}}}\right)\right)^{\frac{4q}{2q-1}}. \end{split}$

Thus,

$$\dot{\tilde{V}}(z_{1}, z_{2}) < 0, \text{ for } V \ge \min \left\{ \begin{pmatrix} 2M_{5} \left(\frac{\overline{\pi}_{1}^{\frac{5p-1}{4p}}}{\omega_{2}} \right) \left(\frac{3^{\frac{1}{4}} \omega_{4}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right)^{\frac{4p}{2p-1}}, \\ \left(2M_{5} \left(\frac{3^{\frac{q-1}{4q}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}{\omega_{3}} \right) \left(\frac{3^{\frac{1}{4}} \omega_{4}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right)^{\frac{4q}{2q-1}} \right\},$$

in which case, from lemma 2.6, the system (2 - 11) is uniformly ultimately bounded.

b) Case 2: For p = 0.5, and $1 \le q$

For p = 0.5, *V* is not differentiable on $z_1 = 0$:

$$\dot{\widetilde{V}}(z_1, z_2) = \bigcap_{\xi \in \partial V(\mathbf{z}(t), t)} \xi^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix} \subset \dot{\widetilde{V}}_1 + \dot{\widetilde{V}}_2$$

$$\dot{\widetilde{V}}_1 = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^T K[f](z_1, z_2), \quad \dot{\widetilde{V}}_2 = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^T K\begin{bmatrix}0\\d\end{bmatrix},$$

$$K[f](z_1, z_2) \in \left[\left(-k_1 \text{SGN}(z_1) - k_{1n} |z_1|^{2q-1} \operatorname{sign}(z_1) - k_2 \text{SGN}(z_2) - k_{2n} |z_2|^{\frac{2q-1}{q}} \operatorname{sign}(z_2) \right) \right]$$

$$\begin{aligned} \partial V &= \mathbf{K} [\nabla V] \\ &\subset \begin{bmatrix} \mathbf{K} \Big[\frac{\partial V}{\partial z_1} \Big] \\ \mathbf{K} \Big[\frac{\partial V}{\partial z_2} \Big] \end{bmatrix} \\ &= \begin{cases} \forall z_1 \neq 0, z_2 \in \mathbb{R} : \\ \left[\Big(|z_2|^2 + 2k_1|z_1| + \frac{k_{1n}}{q} |z_1|^{2q} \Big) (k_1 + k_{1n}|z_1|^{2q-1}) \operatorname{sign}(z_1) + r (1.5|z_1|^{0.5} + 3q|z_1|^{3q-1}) z_2 \\ \\ \left(|z_2|^2 + 2k_1|z_1| + \frac{k_{1n}}{q} |z_1|^{2q} \Big) (z_2) + r |z_1|^{1.5} \operatorname{sign}(z_1) + r |z_1|^{3q} \operatorname{sign}(z_1) \end{bmatrix} \\ &\forall z_1 = 0, z_2 \in \mathbb{R} : \\ \begin{bmatrix} |z_2|^2 k_1 [-1,1] \\ z_2^{-3} \end{bmatrix} \end{aligned}$$

Let us define:

$$\left|\frac{\partial V}{\partial z_1}\right| \coloneqq \sup\left\{\left|\xi_1\right| \colon \xi_1 \in \mathbf{K}\left[\frac{\partial V}{\partial z_1}\right]\right\}, \text{ and } \left|\frac{\partial V}{\partial z_2}\right| \coloneqq \sup\left\{\left|\xi_2\right| \colon \xi_2 \in \mathbf{K}\left[\frac{\partial V}{\partial z_2}\right]\right\},$$

with $\left|\frac{\partial V}{\partial z_2}\right| \le \left|z_2\right|^3 + 2k_1\left|z_1\right|\left|z_2\right| + \frac{k_{1n}}{q}\left|z_1\right|^{2q}\left|z_2\right| + r\left|z_1\right|^{1.5} + r\left|z_1\right|^{3q}$

Thus, the term

$$\begin{split} \dot{\vec{V}}_{2} &= \bigcap_{\xi_{2} \in \mathbf{K} \left[\frac{\partial V}{\partial z_{2}} \right]} \xi_{2} \mathbf{K}[d] \\ &\leq \left| \frac{\partial V}{\partial z_{2}} \right| d | \\ &\leq \left(M + M_{2} |z_{1}|^{2q-1} + M_{4} |z_{2}|^{2-\frac{1}{q}} + M_{5} \right) \left(|z_{2}|^{3} + 2k_{1} |z_{1}| |z_{2}| + \frac{k_{1n}}{q} |z_{1}|^{2q} |z_{2}| + r |z_{1}|^{1.5} + r |z_{1}|^{3q} \right) \end{split}$$

where M be defined as, $M := M_1 + M_3$, since for p = 0.5, $|d| \le M_1 + M_3 + M_2 |z_1|^{2q-1} + M_4 |z_2|^{2-\frac{1}{q}} + M_5$.

Computing $\dot{\vec{V}_1}$ for each case, we have

For $z_1 \neq 0$, $z_2 \neq 0$:

$$\dot{\tilde{V}}_{1} = \left(z_{2}^{2} + 2k_{1}|z_{1}| + \frac{k_{1n}}{q}|z_{1}|^{2q}\right) \times \left(-k_{2}|z_{2}| - k_{2n}|z_{2}|^{\frac{3q-1}{q}}\right) - rk_{1n}|z_{1}|^{2q+0.5} - rk_{1}|z_{1}|^{3q} - rk_{1}|z_{1}|^{1.5}$$
$$- rk_{1n}|z_{1}|^{5q-1} - rk_{2}|z_{1}|^{1.5}\operatorname{sign}(z_{1}z_{2}) - rk_{2n}|z_{1}|^{1.5}|z_{2}|^{2-\frac{1}{q}}\operatorname{sign}(z_{1}z_{2}) + 1.5r|z_{1}|^{0.5}|z_{2}|^{2}$$
$$- rk_{2}|z_{1}|^{3q}\operatorname{sign}(z_{1}z_{2}) - rk_{2n}|z_{1}|^{3q}|z_{2}|^{2-\frac{1}{q}}\operatorname{sign}(z_{1}z_{2}) + 3rq|z_{1}|^{3q-1}|z_{2}|^{2}$$

For $z_1 = 0, z_2 \neq 0$:

Let $(\xi_2 k_1 z_2^2, z_2^3)^T$ with $\xi_2 \in [-1, 1]$ be an arbitrary element of $\partial V(z_1, z_2)$, then

$$\xi^{\mathrm{T}}K[f](z_{1},z_{2}) = \begin{bmatrix} \xi_{2}k_{1}z_{2}^{2} \\ z_{2}^{3} \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} z_{2} \\ -[-1,1]k_{1} - k_{2}\mathrm{sign}(z_{2}) - k_{2n}|z_{2}|^{\frac{2q-1}{q}}\mathrm{sign}(z_{2}) \end{bmatrix}$$
$$= [\xi_{2} - 1, \xi_{2} + 1]k_{1}z_{2}^{3} - k_{2}|z_{2}|^{3} - k_{2n}|z_{2}|^{\frac{5q-1}{q}}$$

hence

$$\dot{\widetilde{V}}_{1} = \bigcap_{\xi_{2} \in [-1, 1]} \left[\xi_{2} - 1, \ \xi_{2} + 1 \right] k_{1} z_{2}^{3} - k_{2} |z_{2}|^{3} - k_{2n} |z_{2}|^{\frac{5q-1}{q}} = -k_{2} |z_{2}|^{3} - k_{2n} |z_{2}|^{\frac{5q-1}{q}}$$

For $z_1 \neq 0$, $z_2 = 0$:

$$\dot{\vec{V}}_{1} = -rk_{1}|z_{1}|^{1.5} - rk_{1}|z_{1}|^{3q} - rk_{1n}|z_{1}|^{2q+0.5} - rk_{1n}|z_{1}|^{5q-1} - rk_{2}[-1,1]z_{1}|^{1.5}\operatorname{sign}(z_{1}) - rk_{2}[-1,1]z_{1}|^{3q}\operatorname{sign}(z_{1})$$

For $z_1 = 0$, $z_2 = 0$:

$$\dot{\widetilde{V}}_1 = 0$$

Thus, for all $(z_1, z_2) \in \mathbb{R}^2$:

$$\dot{\tilde{V}}_{1} = \left(z_{2}^{2} + 2k_{1}|z_{1}| + \frac{k_{1n}}{q}|z_{1}|^{2q}\right) \times \left(-k_{2}|z_{2}| - k_{2n}|z_{2}|^{\frac{3q-1}{q}}\right) - rk_{1n}|z_{1}|^{2q+0.5} - rk_{1}|z_{1}|^{3q} - rk_{1}|z_{1}|^{1.5}$$
$$- rk_{1n}|z_{1}|^{5q-1} - rk_{2}|z_{1}|^{1.5}\operatorname{sign}(z_{1})\operatorname{SGN}(z_{2}) - rk_{2n}|z_{1}|^{1.5}|z_{2}|^{2-\frac{1}{q}}\operatorname{sign}(z_{1}z_{2}) + 1.5r|z_{1}|^{0.5}|z_{2}|^{2}$$
$$- rk_{2}|z_{1}|^{3q}\operatorname{sign}(z_{1})\operatorname{SGN}(z_{2}) - rk_{2n}|z_{1}|^{3q}|z_{2}|^{2-\frac{1}{q}}\operatorname{sign}(z_{1}z_{2}) + 3rq|z_{1}|^{3q-1}|z_{2}|^{2}$$

Hence, the generalized time derivative of the Lyapunov function can be obtained, after rearrangement:

$$\begin{split} \dot{\tilde{V}} &= \dot{\tilde{V}_{1}} + \dot{\tilde{V}_{2}} \\ &\leq -2k_{1}(k_{2} - M)|z_{1}||z_{2}| - \frac{k_{1n}}{q}(k_{2} - M)|z_{1}|^{2q}|z_{2}| - r(k_{1} - k_{2} - M)|z_{1}|^{3q} \\ &- 2k_{1}(k_{2n} - M_{4})|z_{1}||z_{2}|^{3 - \frac{1}{q}} - \frac{k_{1n}}{q}(k_{2n} - M_{4})|z_{1}|^{2q}|z_{2}|^{3 - \frac{1}{q}} - r(k_{1n} - M_{2})|z_{1}|^{2q + 0.5} \\ &- \frac{(k_{2} - M)}{2}|z_{2}|^{3} - \frac{(k_{2n} - M_{4})}{2}|z_{2}|^{5 - \frac{1}{q}} - r\frac{(k_{1} - k_{2} - M)}{2}|z_{1}|^{1.5} - \frac{r(k_{1n} - M_{2})}{2}|z_{1}|^{5q - 1} \\ &+ \dot{V}_{2} + M_{5}\left(|z_{2}|^{3} + 2k_{1}|z_{1}||z_{2}| + \frac{k_{1n}}{q}|z_{1}|^{2q}|z_{2}| + r|z_{1}|^{1.5} + r|z_{1}|^{3q}\right) \end{split}$$

$$\begin{split} \dot{V}_{2} &= -\frac{\left(k_{2}-M\right)}{2} |z_{2}|^{3} - \frac{\left(k_{2n}-M_{4}\right)}{2} |z_{2}|^{5-\frac{1}{q}} - r\frac{\left(k_{1}-k_{2}-M\right)}{2} |z_{1}|^{1.5} \\ &- \frac{r\left(k_{1n}-M_{2}\right)}{2} |z_{1}|^{5q-1} + r\left(k_{2n}+M_{4}\right) |z_{1}|^{1.5} |z_{2}|^{2-\frac{1}{q}} + 1.5r|z_{1}|^{0.5} |z_{2}|^{2} \\ &+ r\left(k_{2n}+M_{4}\right) |z_{1}|^{3q} |z_{2}|^{2-\frac{1}{q}} + 3rq|z_{1}|^{3q-1} |z_{2}|^{2} \\ &+ M_{2} |z_{1}|^{2q-1} |z_{2}|^{3} + M_{2} \frac{k_{1n}}{q} |z_{1}|^{4q-1} |z_{2}| + 2M_{2}k_{1}|z_{1}|^{2q} |z_{2}| \end{split}$$

Applying lemma 2.1,

$$\begin{split} &-\left|z_{1}\right|^{1.5}-\left|z_{2}\right|^{3}\leq-3^{\frac{1}{3}}\left|z_{1}\right|^{0.5}\left(\frac{3}{2}\right)^{\frac{2}{3}}\left|z_{2}\right|^{2},\\ &-\left|z_{1}\right|^{5q-1}-\left|z_{2}\right|^{5-\frac{1}{q}}\leq-\left(\frac{5q-1}{3q}\right)^{\frac{3q}{5q-1}}\left|z_{1}\right|^{3q}\left(\frac{5q-1}{2q-1}\right)^{\frac{2q-1}{5q-1}}\left|z_{2}\right|^{2-\frac{1}{q}},\\ &-\left|z_{1}\right|^{5q-1}-\left|z_{2}\right|^{5-\frac{1}{q}}\leq-\left(\frac{5q-1}{3q-1}\right)^{\frac{3q-1}{5q-1}}\left|z_{1}\right|^{3q-1}\left(\frac{5q-1}{2q}\right)^{\frac{2q}{5q-1}}\left|z_{2}\right|^{2}, \end{split}$$

$$\begin{split} -|z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q-1}\right)^{\frac{2q-1}{5q-1}} |z_{1}|^{2q-1} \left(\frac{5q-1}{3q}\right)^{\frac{3q}{5q-1}} |z_{2}|^{3}, \\ -|z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{4q-1}\right)^{\frac{4q-1}{5q-1}} |z_{1}|^{4q-1} \left(\frac{5q-1}{q}\right)^{\frac{q}{5q-1}} |z_{2}|, \\ -|z_{1}|^{1.5} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -|z_{1}|^{\frac{5q-1}{2q}} - |z_{2}|^{5-\frac{1}{q}} \\ &\leq -\left(\frac{5q-1}{3q}\right)^{\frac{3q}{5q-1}} |z_{1}|^{1.5} \left(\frac{5q-1}{2q-1}\right)^{\frac{2q-1}{5q-1}} |z_{2}|^{2-\frac{1}{q}}, \\ -|z_{1}|^{1.5} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -|z_{1}|^{2q} \left(\frac{5q-1}{4q-1}\right) - |z_{2}|^{5-\frac{1}{q}} \\ &\leq -\left(\frac{5q-1}{4q-1}\right)^{\frac{4q-1}{5q-1}} |z_{1}|^{2q} \left(\frac{5q-1}{q}\right)^{\frac{q}{5q-1}} |z_{2}|, \end{split}$$

where lemma 2.3 has been employed in the last three inequalities since

$$1.5 \le \frac{5q-1}{2q} \le 5q-1$$
, and $1.5 \le 2q \left(\frac{5q-1}{4q-1}\right) \le 5q-1$.

Thus, if the following inequalities

$$\min \begin{cases} \left(\frac{1}{2}\right) \left(\frac{1}{3^{\frac{1}{2}}}\right) (k_{1} - k_{2} - M)^{\frac{1}{2}} (k_{2} - M), \\ \lambda_{12} \frac{(k_{1n} - M_{2})^{\frac{3q}{2q-1}} (k_{2n} - M_{4})}{(k_{2n} + M_{4})^{\frac{5q-1}{2q-1}}}, \\ \lambda_{13} (k_{1n} - M_{2})^{\frac{3q-1}{2q}} (k_{2n} - M_{4}), \\ \lambda_{14} \frac{\omega_{5}^{\frac{3q}{2q-1}} (k_{2n} - M_{4})}{(k_{2n} + M_{4})^{\frac{5q-1}{2q-1}}} \end{cases} > r > \max \begin{cases} \lambda_{15} \frac{M_{2}^{\frac{5q-1}{2q-1}}}{(k_{1n} - M_{2})(k_{2n} - M_{4})^{\frac{3q}{2q-1}}}, \\ \lambda_{16} \frac{k_{1n}^{\frac{5q-1}{4q-1}} M_{2}^{\frac{5q-1}{4q-1}}}{(k_{1n} - M_{2})(k_{2n} - M_{4})^{\frac{q}{4q-1}}}, \\ \lambda_{17} \frac{k_{1}^{\frac{5q-1}{4q-1}} M_{2}^{\frac{5q-1}{4q-1}}}{\omega_{5} (k_{2n} - M_{4})^{\frac{q}{4q-1}}}, \end{cases}$$
(2 - 15)

83

$$\begin{split} \lambda_{12} &= \left(\frac{5q-1}{36q}\right)^{\frac{3q}{2q-1}} \left(\frac{5q-1}{24q-12}\right), \ \lambda_{13} = \left(\frac{5q-1}{36q-12}\right)^{\frac{3q-1}{2q}} \left(\frac{5q-1}{24q}\right) \left(\frac{1}{3q}\right)^{\frac{5q-1}{2q}}, \\ \lambda_{14} &= \left(\frac{5q-1}{3q}\right)^{\frac{3q}{2q-1}} \left(\frac{5q-1}{24q-12}\right), \ \lambda_{15} = \left(\frac{24q-12}{5q-1}\right) \left(\frac{36q}{5q-1}\right)^{\frac{3q}{2q-1}}, \\ \lambda_{16} &= \left(\frac{48q-12}{5q-1}\right) \left(\frac{12q}{5q-1}\right)^{\frac{q}{4q-1}} \left(\frac{1}{q}\right)^{\frac{5q-1}{4q-1}}, \ \lambda_{17} = 2^{\frac{5q-1}{4q-1}} \left(\frac{4q-1}{5q-1}\right) \left(\frac{12q}{5q-1}\right)^{\frac{q}{4q-1}}, \\ \omega_{5} &= \min\left\{\frac{\left(k_{1}-k_{2}-M\right)}{6}, \frac{\left(k_{1n}-M_{2}\right)}{12}\right\} \end{split}$$

hold then function \dot{V}_2 is negative definite. Then,

$$\begin{split} \dot{\vec{V}} &\leq -r(k_1 - k_2 - M)|z_1|^{3q} - \frac{(k_2 - M)}{2}|z_2|^3 - \frac{(k_{2n} - M_4)}{2}|z_2|^{5 - \frac{1}{q}} - r\frac{(k_1 - k_2 - M)}{2}|z_1|^{1.5} \\ &- \frac{r(k_{1n} - M_2)}{2}|z_1|^{5q - 1} + M_5 \left(|z_2|^3 + 2k_1|z_1||z_2| + \frac{k_{1n}}{q}|z_1|^{2q}|z_2| + r|z_1|^{1.5} + r|z_1|^{3q}\right) \end{split}$$

Applying lemma 2.1,

$$|z_1||z_2| \le \frac{2}{3} |z_1|^{1.5} + \frac{1}{3} |z_2|^3,$$
$$|z_1|^{2q} |z_2| \le \frac{2}{3} |z_1|^{3q} + \frac{1}{3} |z_2|^3,$$

and lemma 2.3:

$$-|z_1|^{5q-1}-|z_1|^{1.5} \le -|z_1|^{\frac{5q-1}{2q}},$$

together with lemma 2.2, and the bounds of the Lyapunov function, we have:

$$\begin{split} \dot{\tilde{V}} &\leq -\omega_{7} \left(\left| z_{1} \right|^{1.5} + \left| z_{1} \right|^{3q} + \left| z_{2} \right|^{3} \right) - \omega_{8} \left(\left| z_{1} \right|^{5q-1} + \left| z_{1} \right|^{\frac{5q-1}{2q}} + \left| z_{2} \right|^{5-\frac{1}{q}} \right) + M_{5} \omega_{6} \left(\left| z_{1} \right|^{1.5} + \left| z_{1} \right|^{3q} + \left| z_{2} \right|^{3} \right) \\ &= -\omega_{7} \left(\left(\left| z_{1} \right|^{2} \right)^{\frac{3}{4}} + \left(\left| z_{2} \right|^{4} \right)^{\frac{3}{4}} + \left| \left| z_{2} \right|^{4} \right)^{\frac{3}{4}} \right) - \omega_{8} \left(\left(\left| z_{1} \right|^{4q} \right)^{\frac{5q-1}{4q}} + \left| \left| z_{1} \right|^{2} \right)^{\frac{5q-1}{4q}} + \left| \left| z_{2} \right|^{4} \right)^{\frac{5q-1}{4q}} \right) \\ &+ M_{5} \omega_{6} \left(\left(\left| z_{1} \right|^{2} \right)^{\frac{3}{4}} + \left| \left| z_{1} \right|^{4q} \right)^{\frac{3}{4}} + \left| \left| z_{2} \right|^{4} \right)^{\frac{3}{4}} \right) \\ &\leq - \left(\frac{\omega_{7}}{\overline{\pi_{1}}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} - \left(\frac{\omega_{8}}{3^{\frac{q-1}{4q}} \overline{\pi_{1}}^{\frac{5q-1}{4q}}} \right) V^{\frac{5q-1}{4q}} + M_{5} \left(\frac{3^{\frac{1}{4}} \omega_{6}}{\underline{\pi_{1}}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} \tag{2 - 16} \end{split}$$

$$\omega_{6} = \max\left\{ \left(\frac{4k_{1}}{3} + r\right), \left(\frac{2k_{1n}}{3q} + r\right), \left(1 + \frac{2k_{1}}{3} + \frac{k_{1n}}{3q}\right) \right\},\$$
$$\omega_{7} = \min\left\{\frac{r(k_{1} - k_{2} - M)}{4}, r(k_{1} - k_{2} - M), \frac{(k_{2} - M)}{2}\right\},\$$
$$\omega_{8} = \min\left\{\frac{2r(k_{1n} - M_{2})}{5}, \frac{(k_{2n} - M_{4})}{2}, r\omega_{5}\right\}$$

Remark 2.11. Note that the nonlinear inequalities (2 - 12) and (2 - 15) is feasible with respect to k_1 , k_{1n} , k_2 , k_{2n} for any $M \ge 0$, $M_2 \ge 0$, and $M_4 \ge 0$. As such an r > 0 always exists for sufficiently large k_1 , k_{1n} , k_2 , k_{2n} . In particular, consider the case where $M_2 = 0$, inequalities (2 -15) can be easily satisfied for any $k_1 > k_2 + M > 0$, $k_2 > M > 0$, $k_{2n} > M_4$.

Remark 2.12. Hence, when $M_5 = 0$, it is not difficult to show that the system achieved finite time convergence for $q \ge 1$. In particular for the case of q > 1, following similar arguments in

section 2.2.2(a), finite convergence time independent of initial conditions for q > 1,

$$\begin{split} T_{total} = & \left(\frac{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}{\omega_8}\right) \left(\frac{4q}{q-1}\right) \left(\frac{\omega_8}{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}\right)^{\frac{0.5(q-1)}{q-0.5}} \left(\frac{\overline{\pi}_1 \frac{3}{4}}{\omega_7}\right)^{\frac{0.5(q-1)}{q-0.5}} \\ & + \left(\frac{4\overline{\pi}_1 \frac{3}{4}}{\omega_7}\right) \left(\frac{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}{\omega_8}\right)^{\frac{0.5q}{q-0.5}} \left(\frac{\omega_7}{\overline{\pi}_1 \frac{3}{4}}\right)^{\frac{0.5q}{q-0.5}} \end{split}$$

and when $M_5 \neq 0$, from (2 - 16),

$$\begin{split} \dot{\tilde{V}} &\leq -\left(\frac{\omega_{7}}{\overline{\pi}_{1}^{\frac{3}{4}}}\right) V^{\frac{3}{4}} - \frac{1}{2} \left(\frac{\omega_{8}}{\frac{q^{-1}}{3^{\frac{q-1}{4q}}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}\right) V^{\frac{5q-1}{4q}} - V^{\frac{3}{4}} \left(\frac{1}{2} \left(\frac{\omega_{8}}{\frac{q^{-1}}{3^{\frac{q-1}{4q}}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}\right) V^{\frac{2q-1}{4q}} - M_{5} \left(\frac{3^{\frac{1}{4}}\omega_{6}}{\underline{\pi}_{1}^{\frac{3}{4}}}\right)\right) \\ &\leq -\left(\frac{\omega_{7}}{\overline{\pi}_{1}^{\frac{3}{4}}}\right) V^{\frac{3}{4}} - \frac{1}{2} \left(\frac{\omega_{8}}{\frac{q^{-1}}{3^{\frac{q-1}{4q}}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}\right) V^{\frac{5q-1}{4q}}, \quad \text{for} \quad V \geq \left(2M_{5} \left(\frac{3^{\frac{q-1}{4q}}}{\omega_{8}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}\right) \left(\frac{3^{\frac{1}{4}}\omega_{6}}{\underline{\pi}_{1}^{\frac{3}{4}}}\right)\right)^{\frac{4q}{2q-1}} \end{split}$$

and uniform ultimate boundedness is implied from lemma 2.6.

Remark 2.13. Note that even in the case of $M_5 = 0$, the system is able to be exactly robust with respect to persistent perturbations with an upper bound of M. This interesting feature is possible due to the discontinuous nature of the algorithm when p = 0.5.

2.4 Super-twisting

In this section, the super-twisting based algorithm is revisited by presenting a locally-Lipschitz strict Lyapunov function. The various convergence properties of the algorithm can be fully described through a single Lyapunov function structure. The time derivative of the Lyapunov function is able to avoid singularity due to its locally Lipschitz property. Due to the strictness of the Lyapunov function, different types of disturbances that the algorithm can withstand are shown, including non-Lipschitz disturbances.

2.4.1 System description

Consider the following super-twisting based family of algorithms:

$$\dot{z}_{1} = -k_{3}|z_{1}|^{p}\operatorname{sign}(z_{1}) + z_{2} + d_{1},$$

$$\dot{z}_{2} = -k_{1}|z_{1}|^{2p-1}\operatorname{sign}(z_{1}) + d_{2}$$
(2 - 17)

where $z_1, z_2 \in \mathbb{R}$, are the scalar state variables, k_1, k_3 are positive constants, $p \ge 0.5$, and d_1 and d_2 are time-varying and/or nonlinear term of uncertainties bounded by:

$$|d_1| \le M_1 |z_1|^p + M_2$$
, and $|d_2| \le M_3 |z_1|^{2p-1} + M_4$,

with $|d_1| := \sup \{ \delta \mid : \delta \in \mathbf{K}[d_1] \}$, and $|d_2| := \sup \{ \delta \mid : \delta \in \mathbf{K}[d_2] \}$,

where $M_1 \ge 0$, $M_2 \ge 0$, $M_3 \ge 0$, and $M_4 \ge 0$ with the same $p \ge 0$ as that in (2 - 17). Note that for the case of p = 0.5, the algorithm contains discontinuity and the uncertainty d_2 is upper bounded by nonvanishing constant.

2.4.2 Stability analysis

For system (2 - 17), the following function

$$V(z_1, z_2) = \left(\frac{1}{2}z_2^2 + \frac{k_1}{2p}|z_1|^{2p}\right)^2 - r|z_1||z_2|^{4-\frac{1}{p}}\operatorname{sign}(z_1z_2)$$

where *r* is a positive constant scalar, will be shown as a strict Lyapunov function. Note that the above function is locally Lipschitz and regular. It is differentiable everywhere for p > 0.5, and not differentiable on $z_1 = 0$ for p = 0.5.

Remark 2.14. It should be noted that the above Lyapunov function is still locally Lipschitz even when the algorithm is discontinuous, i.e. p = 0.5 (the conventional super-twisting algorithm). On the other hand, the Lyapunov functions presented in the literature, (i.e. [26] and [27]), for the conventional super-twisting algorithm, are not locally-Lipschitz, which in turn will cause singularity in its time derivative. This singularity will cause issue as noted by [77], when the conventional super-twisting algorithm is use in tandem with adaptive controller. It is also mentioned in [77], that the problem may be resolved whenever a continuously differentiable Lyapunov function is available. Particularly, they proposed to use the continuously differentiable Lyapunov function of [68], albeit, it produces a more conservative condition on the controller gains than would be obtained from the non-Lipschitz one. Thus, it is important to develop a locally Lipschitz Lyapunov function. Note that the Lyapunov function V can be lower bounded by,

$$V(z_1, z_2) \ge \frac{1}{4} |z_2|^4 + \left(\frac{k_1}{2p}\right)^2 |z_1|^{4p} + \left(\frac{k_1}{2p}\right) |z_1|^{2p} |z_2|^2 - r|z_1||z_2|^{4-\frac{1}{p}}$$

Using lemma 2.1,

$$|z_1|^{4p} + |z_2|^4 \ge (4p)^{\frac{1}{4p}} |z_1| \left(\frac{4p}{4p-1}\right)^{\frac{4p-1}{4p}} |z_2|^{4-\frac{1}{p}}$$

Hence, for

$$\left(\frac{p}{8p-2}\right)^{\frac{4p-1}{4p}} \left(\frac{1}{2p}\right)^{\frac{1}{4p}} k_1^{\frac{1}{2p}} > r$$

$$V(z_1, z_2) \ge \frac{1}{8} |z_2|^4 + \frac{1}{2} \left(\frac{k_1}{2p}\right)^2 |z_1|^{4p} \ge \underline{\pi}_1 \left(|z_2|^4 + |z_1|^{4p}\right)$$
where $\underline{\pi}_1 = \min\left\{\frac{1}{8}, \frac{1}{2} \left(\frac{k_1}{2p}\right)^2\right\}.$

$$(2 - 18)$$

Similarly, it can be upper-bounded by,

$$V(z_1, z_2) \le \frac{1}{4} |z_2|^4 + \left(\frac{k_1}{2p}\right)^2 |z_1|^{4p} + \left(\frac{k_1}{2p}\right) |z_1|^{2p} |z_2|^2 + r|z_1||z_2|^{4-\frac{1}{p}}$$

Using lemma 2.1,

$$\begin{aligned} |z_1|^{2p} |z_2|^2 &\leq \frac{1}{2} |z_1|^{4p} + \frac{1}{2} |z_2|^4 \\ |z_1| |z_2|^{4-\frac{1}{p}} &\leq \frac{1}{4p} |z_1|^{4p} + \left(\frac{4p-1}{4p}\right) |z_2|^4 \end{aligned}$$

Thus,

$$V(z_{1}, z_{2}) \leq \overline{\pi}_{1} \left\| \left| z_{1} \right|^{4p} + \left| z_{2} \right|^{4} \right\}$$

where $\overline{\pi}_{1} = \max \left\{ \left(\frac{k_{1}}{2p} \right)^{2} + \left(\frac{k_{1}}{4p} \right) + \frac{r}{4p}, \frac{1}{4} + \left(\frac{k_{1}}{4p} \right) + r \left(\frac{4p-1}{4p} \right) \right\}.$

Thus, *V* is positive definite and radially unbounded. Since (2 - 17) is a differential equation that has discontinuous right-hand side, i.e. when p = 0.5 and since no continuity assumption is made on d_1 , and d_2 , its solution are understood in the sense of Filippov (see definition 2.1). According to lemma 2.4, the time derivative of non-smooth Lyapunov function exists almost everywhere

$$\frac{d}{dt}V(\mathbf{z}(t),t) \stackrel{a.e.}{\in} \overset{\dot{V}}{V}(\mathbf{z}(t),t) = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^{T} K \begin{bmatrix} \dot{z}_{1} \\ \dot{z}_{2} \end{bmatrix} (\mathbf{z},t),$$

Since the controller (2 - 17) is discontinuous when p = 0.5, for ease of presentation, the analysis is separated for the case of p > 0.5 and p = 0.5.

a) Case 1: For p > 0.5

Note that for p > 0, V is continuously differentiable, hence

$$\dot{\widetilde{V}}(z_1, z_2) = \bigcap_{\xi \in \partial V(\mathbf{z}(t), t)} \xi^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix} = \nabla V^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix}$$

$$\subset -\frac{k_1^2 k_3}{p} |z_1|^{5p-1} - k_1 k_3 |z_1|^{3p-1} |z_2|^2 - r|z_2|^{5-\frac{1}{p}} + rk_3 |z_1|^p |z_2|^{4-\frac{1}{p}} \operatorname{sign}(z_1 z_2)$$

$$+ r \left(4 - \frac{1}{p} \right) k_1 |z_1|^{2p} |z_2|^{3-\frac{1}{p}}$$

$$+ K \left[d_1 \left(k_1 |z_1|^{2p-1} |z_2|^2 \operatorname{sign}(z_1) + \frac{k_1^2}{p} |z_1|^{4p-1} \operatorname{sign}(z_1) - r|z_2|^{4-\frac{1}{p}} \operatorname{sign}(z_2) \right)$$

$$+ K \left[d_2 \left(z_2^3 + \frac{k_1}{p} |z_1|^{2p} z_2 - r \left(4 - \frac{1}{p} \right) |z_1| |z_2|^{3-\frac{1}{p}} \operatorname{sign}(z_1) \right) \right]$$

After rearrangement:

$$\begin{split} \dot{\tilde{V}}(z_{1},z_{2}) &\leq -k_{1}(k_{3}-M_{1})|z_{1}|^{3p-1}|z_{2}|^{2} - \frac{k_{1}^{2}(k_{3}-M_{1})}{2p}|z_{1}|^{5p-1} - \frac{r}{2}|z_{2}|^{5-\frac{1}{p}} \\ &+ \dot{V}_{1} + M_{2}\left(k_{1}|z_{1}|^{2p-1}|z_{2}|^{2} + \frac{k_{1}^{2}}{p}|z_{1}|^{4p-1} + r|z_{2}|^{4-\frac{1}{p}}\right) \\ &+ M_{4}\left(|z_{2}|^{3} + \frac{k_{1}}{p}|z_{1}|^{2p}|z_{2}| + r\left(4 - \frac{1}{p}\right)|z_{1}||z_{2}|^{3-\frac{1}{p}}\right) \end{split}$$

where

$$\dot{V}_{1} = -\frac{k_{1}^{2}(k_{3} - M_{1})}{2p} |z_{1}|^{5p-1} - \frac{r}{2} |z_{2}|^{5-\frac{1}{p}} + r(k_{3} + M_{1})|z_{1}|^{p} |z_{2}|^{4-\frac{1}{p}} + r\left(4 - \frac{1}{p}\right)(k_{1} + M_{3})|z_{1}|^{2p} |z_{2}|^{3-\frac{1}{p}} + M_{3}|z_{1}|^{2p-1} |z_{2}|^{3} + M_{3}\frac{k_{1}}{p} |z_{1}|^{4p-1} |z_{2}|$$

Applying lemma 2.1,

$$\begin{split} &- \left|z_{1}\right|^{5p-1} - \left|z_{2}\right|^{5-\frac{1}{p}} \leq - \left(\frac{5p-1}{p}\right)^{\frac{p}{5p-1}} \left|z_{1}\right|^{p} \left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}} \left|z_{2}\right|^{4-\frac{1}{p}}, \\ &- \left|z_{1}\right|^{5p-1} - \left|z_{2}\right|^{5-\frac{1}{p}} \leq - \left(\frac{5p-1}{2p}\right)^{\frac{2p}{5p-1}} \left|z_{1}\right|^{2p} \left(\frac{5p-1}{3p-1}\right)^{\frac{3p-1}{5p-1}} \left|z_{2}\right|^{3-\frac{1}{p}}, \end{split}$$

$$\begin{split} &-\left|z_{1}\right|^{5p-1}-\left|z_{2}\right|^{5-\frac{1}{p}}\leq-\left(\frac{5p-1}{2p-1}\right)^{\frac{2p-1}{5p-1}}\left|z_{1}\right|^{2p-1}\left(\frac{5p-1}{3p}\right)^{\frac{3p}{5p-1}}\left|z_{2}\right|^{3},\\ &-\left|z_{1}\right|^{5p-1}-\left|z_{2}\right|^{5-\frac{1}{p}}\leq-\left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}}\left|z_{1}\right|^{4p-1}\left(\frac{5p-1}{p}\right)^{\frac{p}{5p-1}}\left|z_{2}\right|, \end{split}$$

Thus, if the following inequalities

$$\min\left\{\lambda_{1}\frac{k_{1}^{2}(k_{3}-M_{1})}{(k_{3}+M_{1})^{\frac{5p-1}{p}}}, \lambda_{2}\frac{k_{1}^{2}(k_{3}-M_{1})}{(k_{1}+M_{3})^{\frac{5p-1}{2p}}}\right\} > r > \max\left\{\lambda_{3}\frac{M_{3}^{\frac{5p-1}{3p}}}{k_{1}^{\frac{4p-2}{3p}}(k_{3}-M_{1})^{\frac{2p-1}{3p}}}, \frac{1}{k_{4}}\frac{M_{3}^{\frac{5p-1}{p}}}{k_{4}^{\frac{3p-1}{p}}(k_{3}-M_{1})^{\frac{4p-1}{p}}}\right\}$$

$$(2 - 19)$$

where

$$\begin{split} \lambda_{1} &= \left(\frac{5p-1}{8p^{2}}\right) \left(\frac{5p-1}{8p-8}\right)^{\frac{4p-1}{p}}, \quad \lambda_{2} = \left(\frac{5p-1}{16p^{2}}\right) \left(\frac{5p-1}{24p-8}\right)^{\frac{3p-1}{2p}} \left(\frac{p}{4p-1}\right)^{\frac{5p-1}{2p}}, \\ \lambda_{3} &= \left(\frac{16p^{2}-8p}{5p-1}\right)^{\frac{2p-1}{3p}} \left(\frac{24p}{5p-1}\right), \quad \lambda_{4} = \left(\frac{32p^{2}-8p}{5p-1}\right)^{\frac{4p-1}{p}} \frac{8p}{(5p-1)p^{\frac{5p-1}{p}}} \end{split}$$

hold then the function \dot{V}_1 is negative definite. Then,

$$\begin{split} \dot{\widetilde{V}}(z_1, z_2) &\leq -\frac{k_1^{2}(k_3 - M_1)}{2p} |z_1|^{5p-1} - \frac{r}{2} |z_2|^{5-\frac{1}{p}} \\ &+ M_2 \bigg(k_1 |z_1|^{2p-1} |z_2|^2 + \frac{k_1^{2}}{p} |z_1|^{4p-1} + r |z_2|^{4-\frac{1}{p}} \bigg) \\ &+ M_4 \bigg(|z_2|^3 + \frac{k_1}{p} |z_1|^{2p} |z_2| + r \bigg(4 - \frac{1}{p} \bigg) |z_1| |z_2|^{3-\frac{1}{p}} \bigg) \end{split}$$

Applying lemma 2.1,

$$\begin{split} &|z_1|^{2p-1} |z_2|^2 \leq \left(\frac{2p-1}{4p-1}\right) |z_1|^{4p-1} + \left(\frac{2p}{4p-1}\right) |z_2|^{4-\frac{1}{p}}, \\ &|z_1|^{2p} |z_2| \leq \frac{2}{3} |z_1|^{3p} + \frac{1}{3} |z_2|^3, \\ &|z_1| |z_2|^{3-\frac{1}{p}} \leq \frac{1}{3p} |z_1|^{3p} + \left(\frac{3p-1}{3p}\right) |z_2|^3, \end{split}$$

together with lemma 2.2, and the bounds of the Lyapunov function, we have:

$$\begin{split} \dot{\tilde{V}}(z_{1},z_{2}) &\leq -\omega_{1} \left(\left| z_{1} \right|^{s_{p-1}} + \left| z_{2} \right|^{s-\frac{1}{p}} \right) + M_{2} \omega_{2} \left(\left| z_{1} \right|^{4_{p-1}} + \left| z_{2} \right|^{4-\frac{1}{p}} \right) + M_{4} \omega_{3} \left(\left| z_{1} \right|^{3_{p}} + \left| z_{2} \right|^{3} \right) \\ &= -\omega_{1} \left(\left\| \left| z_{1} \right|^{4_{p}} \right)^{\frac{s_{p-1}}{4_{p}}} + \left\| \left| z_{2} \right|^{4} \right)^{\frac{s_{p-1}}{4_{p}}} \right) + M_{2} \omega_{2} \left(\left\| \left| z_{1} \right|^{4_{p}} \right)^{\frac{4_{p-1}}{4_{p}}} + \left\| \left| z_{2} \right|^{4} \right)^{\frac{4_{p-1}}{4_{p}}} \right) \\ &+ M_{4} \omega_{3} \left(\left\| \left| z_{1} \right|^{4_{p}} \right)^{\frac{3}{4}} + \left\| z_{2} \right|^{4} \right)^{\frac{3}{4}} \right) \\ &\leq - \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{s_{p-1}}{4_{p}}}} \right) V^{\frac{s_{p-1}}{4_{p}}} + M_{2} \left(\frac{2^{\frac{1}{4_{p}}} \omega_{2}}{\underline{\pi}_{1}^{\frac{4_{p-1}}{4_{p}}}} \right) V^{\frac{4_{p-1}}{4_{p}}} + M_{4} \left(\frac{2^{\frac{1}{4}} \omega_{3}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} \end{split}$$

$$(2 - 20)$$

where

$$\omega_{1} = \min\left\{\frac{k_{1}^{2}(k_{3} - M_{1})}{2p}, \frac{r}{2}\right\}, \quad \omega_{2} = \max\left\{k_{1}\left(\frac{2p - 1}{4p - 1}\right) + \frac{k_{1}^{2}}{p}, k_{1}\left(\frac{2p}{4p - 1}\right) + r\right\}$$
$$\omega_{3} = \max\left\{\frac{2k_{1}}{3p} + \left(4 - \frac{1}{p}\right)\left(\frac{r}{3p}\right), 1 + \frac{k_{1}}{3p} + r\left(4 - \frac{1}{p}\right)\left(\frac{3p - 1}{3p}\right)\right\},$$
$$\omega_{4} = \begin{cases}\omega_{1} \quad \text{for } 0.5 1\end{cases}$$

Remark 2.15. Note that the nonlinear inequalities (2 - 18) and (2 - 19) are feasible for sufficiently large $k_1 > 0$, and $k_3 > 0$. Thus, an r > 0 can always exist. Consider some particular cases:

- 1. Note that (2 19) is feasible with respect to k_1 , k_3 for any $M_1 \ge 0$ and $M_3 \ge 0$.
- 2. If $M_3 = 0$, then (2 19) is satisfied for any $k_3 > M_1$. This coincides with the conditions obtained through the weak Lyapunov function, i.e. the energy function

$$E(z_1, z_2) = \frac{1}{2} z_2^2 + \frac{k_1}{2p} |z_1|^{2p}.$$

Remark 2.16. Hence, when $M_2 = M_4 = 0$, i.e. without persistent perturbations, from lemma 2.5-3, the system (2 - 17) will have finite time convergence for 0.5 , with the settling time estimate,

$$T(z_{10}, z_{20}) \leq \left(\frac{\overline{\pi_1}^{\frac{5p-1}{4p}}}{\omega_4}\right) \left(\frac{4p}{1-p}\right) [V(z_{10}, z_{20})]^{\frac{1-p}{4p}}, \text{ where } (z_{10}, z_{20}) \text{ are the initial states of the system.}$$

Similarly, exponential and asymptotical convergence for p = 1 and p > 1 respectively, can be concluded from lemma 2.5 as well. The above results are possible due to the negative definiteness of the time derivative of the Lyapunov function, i.e. strict Lyapunov function.

Remark 2.17. When persistent perturbations occur on the system, $M_2 \neq 0$ and/or $M_4 \neq 0$, from (2 - 20)

$$\begin{split} \dot{\widetilde{V}}(z_{1},z_{2}) &\leq -\frac{1}{3} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{5p-1}{4p}}} \right) V^{\frac{5p-1}{4p}} - V^{\frac{4p-1}{4p}} \left(\frac{1}{3} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{5p-1}{4p}}} \right) V^{\frac{1}{4}} - M_{2} \left(\frac{2^{\frac{1}{4p}} \omega_{2}}{\underline{\pi}_{1}^{\frac{4p-1}{4p}}} \right) \right) \\ &- V^{\frac{3}{4}} \left(\frac{1}{3} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{5p-1}{4p}}} \right) V^{\frac{2p-1}{4p}} - M_{4} \left(\frac{2^{\frac{1}{4}} \omega_{3}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right) \right) \\ &\leq -\frac{1}{3} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{5p-1}{4p}}} \right) V^{\frac{5p-1}{4p}}, \quad \text{for} \quad V \geq \max \begin{cases} \left(3M_{2} \left(\frac{\overline{\pi}_{1}^{\frac{5p-1}{4p}}}{\omega_{4}} \right) \left(\frac{2^{\frac{1}{4p}} \omega_{2}}{\underline{\pi}_{1}^{\frac{4p-1}{4p}}} \right) \right)^{4}, \\ \left(3M_{4} \left(\frac{\overline{\pi}_{1}^{\frac{5p-1}{4p}}}{\omega_{4}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{3}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right)^{\frac{4p}{2p-1}}, \end{split}$$

hence, from lemma 2.6, the system (2 - 17) is uniformly ultimately bounded.

b) Case 2: For p = 0.5

For p = 0.5, *V* is not differentiable on $z_1 = 0$:

$$\dot{\widetilde{V}}(z_1, z_2) = \bigcap_{\xi \in \partial V(\mathbf{z}(t), t)} \xi^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix} \subset \dot{\widetilde{V}}_1 + \dot{\widetilde{V}}_2$$

where

$$\dot{\widetilde{V}}_{1} = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^{T} K[f](z_{1}, z_{2}), \quad \dot{\widetilde{V}}_{2} = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^{T} K\begin{bmatrix}d_{1}\\d_{2}\end{bmatrix},$$
$$K[f](z_{1}, z_{2}) \in \begin{bmatrix}-k_{3}|z_{1}|^{0.5} \operatorname{sign}(z_{1}) + z_{2}\\-k_{1} \operatorname{SGN}(z_{1})\end{bmatrix}$$

$$\partial V = \mathbf{K} [\nabla V] = \mathbf{K} \begin{bmatrix} \frac{\partial V}{\partial z_1} \\ \frac{\partial V}{\partial z_2} \end{bmatrix}$$

$$\subset \begin{bmatrix} \mathbf{K} \begin{bmatrix} \frac{\partial V}{\partial z_1} \\ \frac{\partial V}{\partial z_2} \end{bmatrix} \end{bmatrix}$$

$$= \begin{cases} \begin{bmatrix} (z_2^2 + 2k_1 |z_1|)(k_1 \operatorname{sign}(z_1)) - r |z_2|^2 \operatorname{sign}(z_2) \\ (z_2^2 + 2k_1 |z_1|)(z_2) - 2r z_1 |z_2| \end{bmatrix}, \quad \forall z_1 \neq 0, \ z_2 \in \mathbb{R}$$

$$= \begin{cases} \begin{bmatrix} z_2^2 k_1 [-1,1] - r |z_2|^2 \operatorname{sign}(z_2) \\ z_2^3 \end{bmatrix}, \quad \forall z_1 = 0, \ z_2 \in \mathbb{R} \end{cases}$$

Let us define:

$$\left|\frac{\partial V}{\partial z_1}\right| \coloneqq \sup\left\{\left|\xi_1\right| \colon \xi_1 \in \mathbf{K}\left[\frac{\partial V}{\partial z_1}\right]\right\}, \text{ and } \left|\frac{\partial V}{\partial z_2}\right| \coloneqq \sup\left\{\left|\xi_2\right| \colon \xi_2 \in \mathbf{K}\left[\frac{\partial V}{\partial z_2}\right]\right\},$$

with $\left|\frac{\partial V}{\partial z_1}\right| \le k_1 |z_2|^2 + 2k_1^2 |z_1| + r|z_2|^2, \text{ and } \left|\frac{\partial V}{\partial z_2}\right| \le |z_2|^3 + 2k_1 |z_1||z_2| + 2r|z_1||z_2|.$

Thus, the term

$$\begin{split} \widetilde{V}_{2} &= \bigcap_{(\xi_{1},\xi_{2})^{T} \in \partial V(\mathbf{z}(t),t)} \xi_{1}K[d_{1}] + \xi_{2}K[d_{2}] \\ &\leq \left| \frac{\partial V}{\partial z_{1}} \right| d_{1} | + \left| \frac{\partial V}{\partial z_{2}} \right| d_{2} | \\ &\leq \left(M_{1}|z_{1}|^{0.5} + M_{2} \right) (2k_{1}^{2}|z_{1}| + (k_{1}+r)|z_{2}|^{2}) + M \left(|z_{2}|^{3} + 2k_{1}|z_{1}||z_{2}| + 2r|z_{1}||z_{2}| \right) \end{split}$$

where *M* be defined as, $M := M_3 + M_4$, since for p = 0.5, $|d_1| \le M_1 |z_1|^{0.5} + M_2$, and $|d_2| \le M_3 + M_4$.

Computing $\dot{\widetilde{V}}_1$ for each case, we have

For $z_1 \neq 0, z_2 \in \mathbb{R}$:

$$\dot{\tilde{V}}_{1} = -2k_{1}^{2}k_{3}|z_{1}|^{1.5} - k_{1}k_{3}|z_{1}|^{0.5}|z_{2}|^{2} - r|z_{2}|^{3} + rk_{3}|z_{1}|^{0.5}|z_{2}|^{2}\operatorname{sign}(z_{1}z_{2}) + 2rk_{1}|z_{1}||z_{2}|$$

For $z_1 = 0, z_2 \in \mathbb{R}$:

Let $(\xi_2 k_1 z_2^2 - r|z_2|^2 \operatorname{sign}(z_2), z_2^3)^T$ with $\xi_2 \in [-1, 1]$ be an arbitrary element of $\partial V(z_1, z_2)$, then

$$\xi^{\mathrm{T}}K[f](z_{1},z_{2}) = \begin{bmatrix} \xi_{2}k_{1}z_{2}^{2} - r|z_{2}|^{2}\operatorname{sign}(z_{2}) \\ z_{2}^{3} \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} z_{2} \\ -[-1,1]k_{1} \end{bmatrix}$$
$$= [\xi_{2}-1,\xi_{2}+1]k_{1}z_{2}^{3} - r|z_{2}|^{3}$$

hence $\dot{\tilde{V}}_1 = \bigcap_{\xi_2 \in [-1, 1]} [\xi_2 - 1, \xi_2 + 1] k_1 z_2^3 - r |z_2|^3 = -r |z_2|^3$

Thus, for all $(z_1, z_2) \in \mathbb{R}^2$:

$$\dot{\tilde{V}}_{1} = -2k_{1}^{2}k_{3}|z_{1}|^{1.5} - k_{1}k_{3}|z_{1}|^{0.5}|z_{2}|^{2} - r|z_{2}|^{3} + rk_{3}|z_{1}|^{0.5}|z_{2}|^{2}\operatorname{sign}(z_{1}z_{2}) + 2rk_{1}|z_{1}||z_{2}|$$

Hence, the generalized time derivative of the Lyapunov function, after rearrangement:

$$\dot{\widetilde{V}} = \dot{\widetilde{V}}_{1} + \dot{\widetilde{V}}_{2} \le -k_{1}^{2} (k_{3} - M_{1}) |z_{1}|^{1.5} - \frac{r}{2} |z_{2}|^{3} + \dot{V}_{2} + M_{2} (2k_{1}^{2} |z_{1}| + (k_{1} + r) |z_{2}|^{2})$$

where

$$\dot{V}_{2} = -k_{1}^{2}(k_{3} - M_{1})|z_{1}|^{1.5} - \frac{r}{4}|z_{2}|^{3} - \left(\frac{r}{4} - M\right)|z_{2}|^{3} - (k_{1}(k_{3} - M_{1}) - r(k_{3} + M_{1}))|z_{1}|^{0.5}|z_{2}|^{2} + 2r(k_{1} + M)|z_{1}||z_{2}| + 2Mk_{1}|z_{1}||z_{2}|$$

Applying lemma 2.1

$$-|z_1|^{1.5} - |z_2|^3 \le -(1.5)^{\frac{1}{1.5}}|z_1|(3)^{\frac{1}{3}}|z_2|$$

Thus, if the following inequalities

$$\min\left\{\frac{\frac{k_{1}(k_{3}-M_{1})}{(k_{3}+M_{1})},}{\left(\frac{3}{4}\right)^{\frac{1}{2}}\left(\frac{1}{2^{\frac{3}{2}}}\right)\frac{k_{1}^{2}(k_{3}-M_{1})}{(k_{1}+M)^{\frac{3}{2}}}\right\} > r > \max\left\{\frac{4M}{\left(\frac{4}{3}\right)^{2}\left(\frac{8}{3}\right)^{2}\frac{M^{3}}{k_{1}(k_{3}-M_{1})^{2}}\right\}$$
(2 - 21)

hold then the function \dot{V}_2 is negative definite. Then,

$$\dot{\tilde{V}} \leq -k_1^2 (k_3 - M_1) |z_1|^{1.5} - \frac{r}{2} |z_2|^3 + M_2 (2k_1^2 |z_1| + (k_1 + r) |z_2|^2)$$

Employing lemma 2.2, and the bounds of the Lyapunov function, we have:

$$\begin{split} \dot{\vec{V}} &\leq -\omega_{5} \left(\left| z_{1} \right|^{1.5} + \left| z_{2} \right|^{3} \right) + M_{2} \omega_{6} \left(\left| z_{1} \right|^{2} \right)^{\frac{1}{2}} \\ &= -\omega_{5} \left(\left| \left| z_{1} \right|^{2} \right)^{\frac{3}{4}} + \left| \left| z_{2} \right|^{4} \right)^{\frac{3}{4}} \right) + M_{2} \omega_{6} \left(\left| \left| z_{1} \right|^{2} \right)^{\frac{1}{2}} + \left| \left| z_{2} \right|^{4} \right)^{\frac{1}{2}} \right) \\ &\leq -\omega_{5} \left(\left| z_{1} \right|^{2} + \left| z_{2} \right|^{4} \right)^{\frac{3}{4}} + M_{2} 2^{\frac{1}{2}} \omega_{6} \left(\left| z_{1} \right|^{2} + \left| z_{2} \right|^{4} \right)^{\frac{1}{2}} \\ &\leq - \left(\frac{\omega_{5}}{\overline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} + M_{2} \left(\frac{2^{\frac{1}{2}} \omega_{6}}{\underline{\pi}_{1}^{\frac{1}{2}}} \right) V^{\frac{1}{2}} \end{split}$$

$$(2 - 22)$$

where

$$\omega_5 = \min\left\{k_1^2(k_3 - M_1), \frac{r}{2}\right\}, \quad \omega_6 = \max\left\{2k_1^2, (k_1 + r)\right\}$$

Remark 2.18. Note that the nonlinear inequalities (2 - 18) and (2 - 21) is feasible with respect to k_1 , k_3 for any $M \ge 0$ and $M_1 \ge 0$, which guarantees the existence of an r > 0. Also, when M = 0, (2 - 21) is easily satisfied for any $k_1 > 0$ and $k_3 > M_1$.

Remark 2.19. Thus, when $M_2 = 0$, utilising lemma 2.5-3, the system (2 - 17) will have finite time convergence for p = 0.5, with the settling time estimate,

$$T(z_{10}, z_{20}) \leq \left(\frac{4\overline{\pi}_{1}^{3}}{\omega_{5}}\right) [V(z_{10}, z_{20})]^{\frac{1}{4}},$$

where (z_{10}, z_{20}) are the initial states of the system, in which case the system is able to withstand persistent perturbations, bounded by *M*.

Remark 2.20. While for the case of $M_2 \neq 0$, from (2 - 22)

$$\begin{split} \dot{\widetilde{V}} &\leq -\frac{1}{2} \left(\frac{\omega_{5}}{\overline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} - V^{\frac{1}{2}} \left(\frac{1}{2} \left(\frac{\omega_{5}}{\overline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{1}{4}} - M_{2} \left(\frac{2^{\frac{1}{2}} \omega_{6}}{\underline{\pi}_{1}^{\frac{1}{2}}} \right) \right) \\ &\leq -\frac{1}{2} \left(\frac{\omega_{5}}{\overline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}}, \quad \text{for } V \geq \left(2M_{2} \left(\frac{\overline{\pi}_{1}^{\frac{3}{4}}}{\omega_{5}} \right) \left(\frac{2^{\frac{1}{2}} \omega_{6}}{\underline{\pi}_{1}^{\frac{1}{2}}} \right) \right)^{4} \end{split}$$

where uniformly ultimately bounded is concluded from employing lemma 2.6.

2.5 Generic Super-twisting

Employing the results of section 2.2.3, a generic super-twisting based family of controllers is presented. In a similar spirit as per section 2.2.2 (generic twisting), the algorithm comprise linear sum of two different homogeneity super-twisting based algorithms. By means of strict and locally-Lipschitz Lyapunov functions, the family of algorithms are shown to exhibit the properties of their individual components while at the same time, singularity is avoided in its time derivative.

2.5.1 System description

Consider the following generic super-twisting based dynamics:

$$\dot{z}_{1} = -k_{3}|z_{1}|^{p}\operatorname{sign}(z_{1}) - k_{3n}|z_{1}|^{q}\operatorname{sign}(z_{1}) + z_{2} + d_{1},$$

$$\dot{z}_{2} = -k_{1}|z_{1}|^{2p-1}\operatorname{sign}(z_{1}) - k_{1n}|z_{1}|^{2q-1}\operatorname{sign}(z_{1}) + d_{2}$$
(2 - 23)

where $z_1, z_2 \in \mathbb{R}$, are the scalar state variables, k_1, k_{1n}, k_3, k_{3n} are positive constants, $0.5 \le p \le 1$, and $1 \le q$. Also, d_1 and d_2 are time-varying and/or nonlinear term of uncertainties bounded by:

$$|d_1| \le M_1 |z_1|^p + M_2 |z_1|^q + M_3, \text{ and } |d_2| \le M_4 |z_1|^{2p-1} + M_5 |z_1|^{2q-1} + M_6,$$

with $|d_1| := \sup\{|\delta| : \delta \in \mathbf{K}[d_1]\}$, and $|d_2| := \sup\{|\delta| : \delta \in \mathbf{K}[d_2]\}$,

where $M_1 \ge 0$, $M_2 \ge 0$, $M_3 \ge 0$, $M_4 \ge 0$, $M_5 \ge 0$, $M_6 \ge 0$ with the same p and q as that in (2 - 23). Note that for the case of p = 0.5, the algorithm contains discontinuity and the uncertainty d_2 is upper bounded by nonvanishing constant.

2.5.2 Stability analysis

The system above is essentially a summation of two different degree of homogeneity of the super-twisting-based algorithm considered in section 2.2.3. Hence, a summation of two different degree of homogeneity Lyapunov function based on section 2.2.3 is proposed for system (2 - 23), namely

$$V(z_{1}, z_{2}) = \left(\frac{1}{2}z_{2}^{2} + \frac{k_{1}}{2p}|z_{1}|^{2p} + \frac{k_{1n}}{2q}|z_{1}|^{2q}\right)^{2} - r|z_{1}||z_{2}|^{4-\frac{1}{p}}\operatorname{sign}(z_{1}z_{2})$$
$$-r|z_{1}||z_{2}|^{4-\frac{1}{q}}\operatorname{sign}(z_{1}z_{2})$$

where *r* is a positive constant scalar. Note that the above function is locally Lipschitz and regular. It is differentiable everywhere for p > 0.5, and not differentiable on $z_1 = 0$ for p = 0.5.

Note that the Lyapunov function, V can be lower bounded by,

$$V(z_{1}, z_{2}) \geq \frac{1}{4} |z_{2}|^{4} + \frac{k_{1}^{2}}{4p^{2}} |z_{1}|^{4p} + \frac{k_{1n}^{2}}{4q^{2}} |z_{1}|^{4q} + \frac{k_{1}}{2p} |z_{1}|^{2p} |z_{2}|^{2} + \frac{k_{1n}}{2q} |z_{1}|^{2q} |z_{2}|^{2} + \frac{k_{1}k_{1n}}{2pq} |z_{1}|^{2p+2q} - r|z_{1}||z_{2}|^{4-\frac{1}{p}} - r|z_{1}||z_{2}|^{4-\frac{1}{q}}$$

Using lemma 2.1,

$$\begin{aligned} |z_1|^{4p} + |z_2|^4 &\geq (4p)^{\frac{1}{4p}} |z_1| \left(\frac{4p}{4p-1}\right)^{\frac{4p-1}{4p}} |z_2|^{4-\frac{1}{p}}, \\ |z_1|^{4q} + |z_2|^4 &\geq (4q)^{\frac{1}{4q}} |z_1| \left(\frac{4q}{4q-1}\right)^{\frac{4q-1}{4q}} |z_2|^{4-\frac{1}{q}}, \end{aligned}$$

Hence, for

$$\min\left\{ \left(\frac{k_{1}^{2}}{2p}\right)^{\frac{1}{4p}} \left(\frac{4p}{4p-1}\frac{1}{12}\right)^{\frac{4p-1}{4p}}, \left(\frac{k_{1n}^{2}}{2q}\right)^{\frac{1}{4q}} \left(\frac{4q}{4q-1}\frac{1}{12}\right)^{\frac{4q-1}{4q}}\right\} > r, \qquad (2-24)$$

$$V(z_{1}, z_{2}) \ge \underline{\pi}_{1} \left(|z_{1}|^{4p} + |z_{1}|^{4q} + |z_{2}|^{4}\right)$$
where $\underline{\pi}_{1} = \min\left\{\frac{k_{1}^{2}}{8p^{2}}, \frac{k_{1n}^{2}}{8q^{2}}, \frac{1}{12}\right\}.$

Also, its upper bound can be obtained as,

$$V(z_{1}, z_{2}) \leq \frac{1}{4} |z_{2}|^{4} + \frac{k_{1}^{2}}{4p^{2}} |z_{1}|^{4p} + \frac{k_{1n}^{2}}{4q^{2}} |z_{1}|^{4q} + \frac{k_{1}}{2p} |z_{1}|^{2p} |z_{2}|^{2} + \frac{k_{1n}}{2q} |z_{1}|^{2q} |z_{2}|^{2} + \frac{k_{1n}}{2pq} |z_{1}|^{2p+2q} + r|z_{1}||z_{2}|^{4-\frac{1}{p}} + r|z_{1}||z_{2}|^{4-\frac{1}{q}}$$

Using lemma 2.1, we have,

$$\begin{aligned} \frac{k_1}{2p} |z_1|^{2p} |z_2|^2 &\leq \frac{k_1}{4p} |z_1|^{4p} + \frac{k_1}{4p} |z_2|^4, \\ \frac{k_{1n}}{2q} |z_1|^{2q} |z_2|^2 &\leq \frac{k_{1n}}{4q} |z_1|^{4q} + \frac{k_{1n}}{4q} |z_2|^4, \\ \frac{k_1 k_{1n}}{2pq} |z_1|^{2p+2q} &\leq \frac{k_1 k_{1n}}{4pq} |z_1|^{4p} + \frac{k_1 k_{1n}}{4pq} |z_1|^{4q}, \\ r |z_1| |z_2|^{4-\frac{1}{p}} &\leq \frac{r}{4p} |z_1|^{4p} + \frac{r(4p-1)}{4p} |z_2|^4, \end{aligned}$$

$$r|z_1||z_2|^{4-\frac{1}{q}} \le \frac{r}{4q}|z_1|^{4q} + \frac{r(4q-1)}{4q}|z_2|^4$$

Then,

$$V(z_1, z_2) \le \overline{\pi}_1 \left(|z_1|^{4p} + |z_1|^{4q} + |z_2|^4 \right)$$

where

$$\overline{\pi}_{1} = \max\left\{ \left(\frac{k_{1}^{2}}{4p^{2}} + \frac{k_{1}}{4p} + \frac{r}{4p} + \frac{k_{1}k_{1n}}{4pq} \right), \left(\frac{k_{1n}^{2}}{4q^{2}} + \frac{k_{1n}}{4q} + \frac{r}{4q} + \frac{k_{1}k_{1n}}{4pq} \right), \left(\frac{1}{4} + \frac{k_{1}}{4p} + \frac{k_{1n}}{4q} + \frac{r(4p-1)}{4p} + \frac{r(4q-1)}{4q} \right) \right\}.$$

Thus, *V* is positive definite and radially unbounded. Since (2 - 23) is a differential equation that has discontinuous right-hand side, i.e. when p = 0.5 and since no continuity assumption is made on d_1 , and d_2 , its solution are understood in the sense of Filippov (see definition 2.1). According to lemma 2.4, the time derivative of non-smooth Lyapunov function exists almost everywhere

$$\frac{d}{dt}V(\mathbf{z}(t),t) \stackrel{a.e.}{\in} \dot{\tilde{V}}(\mathbf{z}(t),t) = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^{T} K\begin{bmatrix} \dot{z}_{1} \\ \dot{z}_{2} \end{bmatrix} (\mathbf{z},t).$$

Since the controller (2 - 23) is discontinuous when p = 0.5, for ease of presentation, the analysis is separated for two different cases of p > 0.5 and p = 0.5.

a) Case 1: For $0.5 , and <math>l \le q$

Note that for p > 0.5, V is continuously differentiable, hence

$$\begin{split} \dot{\vec{V}}(z_{1},z_{2}) &= \bigcap_{\xi \in \partial^{r}(\mathbf{x}(t),t)} \xi^{T} K \begin{bmatrix} \dot{z}_{1} \\ \dot{z}_{2} \end{bmatrix} = \nabla V^{T} K \begin{bmatrix} \dot{z}_{1} \\ \dot{z}_{2} \end{bmatrix} \\ &\subset -k_{1}k_{3}|z_{1}|^{3p-1}|z_{2}|^{2} - k_{1}k_{3n}|z_{1}|^{2p+q-1}|z_{2}|^{2} - k_{1n}k_{3}|z_{1}|^{p+2q-1}|z_{2}|^{2} \\ &- k_{1n}k_{3n}|z_{1}|^{3q-1}|z_{2}|^{2} - \frac{k_{1}}{p}k_{1}k_{3}|z_{1}|^{5p-1} - \frac{k_{1}}{p}k_{1}k_{3n}|z_{1}|^{4p+q-1} \\ &- \frac{k_{1}}{p}k_{1n}k_{3}|z_{1}|^{3p+2q-1} - \frac{k_{1}}{p}k_{1n}k_{3n}|z_{1}|^{2p+3q-1} - \frac{k_{1n}}{q}k_{1}k_{3n}|z_{1}|^{2q+3p-1} \\ &- \frac{k_{1n}}{q}k_{1}k_{3n}|z_{1}|^{2p+3q-1} - \frac{k_{1n}}{q}k_{1n}k_{3}|z_{1}|^{p+4q-1} - \frac{k_{1n}}{q}k_{1n}k_{3n}|z_{1}|^{2q+3p-1} \\ &- \frac{k_{1n}}{q}k_{1}k_{3n}|z_{1}|^{2p+3q-1} - \frac{k_{1n}}{q}k_{1n}k_{3}|z_{1}|^{p+4q-1} - \frac{k_{1n}}{q}k_{1n}k_{3n}|z_{1}|^{5q-1} \\ &- r|z_{2}|^{5-\frac{1}{p}} - r|z_{2}|^{5-\frac{1}{q}} + rk_{3}|z_{1}|^{p}|z_{2}|^{4-\frac{1}{p}}sign(z_{1}z_{2}) \\ &+ rk_{3n}|z_{1}|^{q}|z_{2}|^{4-\frac{1}{p}}sign(z_{1}z_{2}) + r\left(4 - \frac{1}{p}\right)k_{1}|z_{1}|^{2p}|z_{2}|^{3-\frac{1}{p}} \\ &+ r\left(4 - \frac{1}{p}\right)k_{1n}|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{p}} + rk_{3}|z_{1}|^{p}|z_{2}|^{4-\frac{1}{q}}sign(z_{1}z_{2}) \\ &+ rk_{3n}|z_{1}|^{q}|z_{2}|^{4-\frac{1}{q}}sign(z_{1}z_{2}) + r\left(4 - \frac{1}{q}\right)k_{1}|z_{1}|^{2p}|z_{2}|^{3-\frac{1}{q}} \\ &+ r\left(4 - \frac{1}{q}\right)k_{1n}|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{p}} \\ &+ r\left(4 - \frac{1}{q}\right)k_{1n}|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{q}} \\ &+ r\left(4 - \frac{1}{q}\right)k_{1n}|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{q}} \\ &+ r\left(4 - \frac{1}{q}\right)k_{1n}|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{q}} \\ &+ r\left(4 - \frac{1}{q}\right)k_{1n}|z_{1}|^{2q}|z_{2}|^{2} + \frac{k_{1}^{2}}{p}|z_{1}|^{4p-1} + \frac{k_{1}k_{1n}}{q}|z_{1}|^{2p+2q-1}\right)sign(z_{1}) \\ \\ &+ K\left[d_{1}\left(-r|z_{2}|^{4-\frac{1}{p}}sign(z_{2}) - r|z_{2}|^{4-\frac{1}{p}}sign(z_{2}) \\ &+ \left(k_{1n}|z_{1}|^{2q-1}|z_{2}|^{2} + \frac{k_{1n}}{p}|z_{1}|^{2q}z_{2} - r\left(4 - \frac{1}{p}\right)z_{1}||z_{2}|^{3-\frac{1}{p}}sign(z_{1}) \\ \\ &+ K\left[d_{2}\left(z_{2}^{3} + \frac{k_{1}}{p}|z_{1}|^{2p}z_{2} + \frac{k_{1n}}{q}|z_{1}|^{2q}z_{2} - r\left(4 - \frac{1}{p}\right)z_{1}||z_{2}|^{3-\frac{1}{p}}sign(z_{1}) \\ \\ &+ K\left[d_{2}\left(z_{2}^{3} + \frac{k_{1}}{p}|z_{1}|z_{2}|^{3-\frac{1}{q}}sign(z_{1}) \\ \\ &- r\left(4 - \frac{1}{q}\right)z_{1}||z_{2}|^{3-$$

After rearrangement:

$$\begin{split} \dot{\tilde{V}} &\leq -k_{1}\left(k_{3}-M_{1}\right)\left|z_{1}\right|^{3p-1}\left|z_{2}\right|^{2}-k_{1n}\left(k_{3n}-M_{2}\right)\left|z_{1}\right|^{3q-1}\left|z_{2}\right|^{2} \\ &-k_{1}\left(k_{3n}-M_{2}\right)\left|z_{1}\right|^{q+2p-1}\left|z_{2}\right|^{2}-k_{1n}\left(k_{3}-M_{1}\right)\left|z_{1}\right|^{p+2q-1}\left|z_{2}\right|^{2} \\ &-k_{1}k_{1n}\left(k_{3}-M_{1}\right)\left(\frac{1}{p}+\frac{1}{q}\right)\left|z_{1}\right|^{3p+2q-1}-k_{1}k_{1n}\left(k_{3n}-M_{2}\right)\left(\frac{1}{q}+\frac{1}{p}\right)\left|z_{1}\right|^{2p+3q-1} \\ &-\frac{k_{1}^{2}}{p}\left(k_{3n}-M_{2}\right)\left|z_{1}\right|^{q+4p-1}-\frac{k_{1n}^{2}}{q}\left(k_{3}-M_{1}\right)\left|z_{1}\right|^{p+4q-1} \\ &-\frac{k_{1}^{2}}{2p}\left(k_{3}-M_{1}\right)\left|z_{1}\right|^{5p-1}-\frac{k_{1n}^{2}}{2q}\left(k_{3n}-M_{2}\right)\left|z_{1}\right|^{5q-1}-\frac{r}{2}\left|z_{2}\right|^{5-\frac{1}{p}}-\frac{r}{2}\left|z_{2}\right|^{5-\frac{1}{q}}+\dot{V}_{1}\right| \\ &+M_{3}\left(\frac{k_{1}^{2}}{p}\left|z_{1}\right|^{4p-1}+\frac{k_{1n}^{2}}{q}\left|z_{1}\right|^{4q-1}+k_{1}\left|z_{1}\right|^{2p-1}\left|z_{2}\right|^{2}+k_{1n}\left|z_{1}\right|^{2q-1}\left|z_{2}\right|^{2}\right) \\ &+M_{6}\left(\left|z_{2}\right|^{3}+\frac{k_{1}}{p}\left|z_{1}\right|^{2p}\left|z_{2}\right|+\frac{k_{1n}}{q}\left|z_{1}\right|^{2q}\left|z_{2}\right| \\ &+r\left(4-\frac{1}{p}\right)\left|z_{1}\right|\left|z_{2}\right|^{3-\frac{1}{p}}+r\left(4-\frac{1}{q}\right)\left|z_{1}\right|\left|z_{2}\right|^{3-\frac{1}{q}}\right)\right) \end{split}$$

$$\begin{split} \dot{V_{1}} &= -\frac{k_{1}^{2}}{2p} (k_{3} - M_{1}) |z_{1}|^{5p-1} - \frac{k_{1n}^{2}}{2q} (k_{3n} - M_{2}) |z_{1}|^{5q-1} - \frac{r}{2} |z_{2}|^{5-\frac{1}{p}} - \frac{r}{2} |z_{2}|^{5-\frac{1}{p}} \\ &+ r (k_{3} + M_{1}) |z_{1}|^{p} |z_{2}|^{4-\frac{1}{p}} + r (k_{3n} + M_{2}) |z_{1}|^{q} |z_{2}|^{4-\frac{1}{p}} \\ &+ r \left(4 - \frac{1}{p} \right) (k_{1} + M_{4}) |z_{1}|^{2p} |z_{2}|^{3-\frac{1}{p}} + r \left(4 - \frac{1}{p} \right) (k_{1n} + M_{5}) |z_{1}|^{2q} |z_{2}|^{3-\frac{1}{p}} \\ &+ r (k_{3} + M_{1}) |z_{1}|^{p} |z_{2}|^{4-\frac{1}{q}} + r (k_{3n} + M_{2}) |z_{1}|^{q} |z_{2}|^{4-\frac{1}{q}} \\ &+ r \left(4 - \frac{1}{q} \right) (k_{1} + M_{4}) |z_{1}|^{2p} |z_{2}|^{3-\frac{1}{q}} + r \left(4 - \frac{1}{q} \right) (k_{1n} + M_{5}) |z_{1}|^{2q} |z_{2}|^{3-\frac{1}{q}} \\ &+ M_{4} |z_{1}|^{2p-1} |z_{2}|^{3} + M_{4} \frac{k_{1}}{p} |z_{1}|^{4p-1} |z_{2}| + \left(M_{4} \frac{k_{1n}}{q} + M_{5} \frac{k_{1}}{p} \right) |z_{1}|^{2p+2q-1} |z_{2}| \\ &+ M_{5} |z_{1}|^{2q-1} |z_{2}|^{3} + M_{5} \frac{k_{1n}}{q} |z_{1}|^{4q-1} |z_{2}| \end{split}$$

Applying lemma 2.1,

$$\begin{split} -|z_{1}|^{5p-1} -|z_{2}|^{5\frac{1}{p}} &\leq -\left(\frac{5p-1}{p}\right)^{\frac{p}{5p-1}} |z_{1}|^{p} \left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}} |z_{2}|^{4\frac{1}{p}}, \\ -|z_{1}|^{5p-1} -|z_{2}|^{5\frac{1}{p}} &\leq -\left(\frac{5p-1}{2p}\right)^{\frac{2p}{5p-1}} |z_{1}|^{2p} \left(\frac{5p-1}{3p-1}\right)^{\frac{3p-1}{5p-1}} |z_{2}|^{3\frac{1}{p}}, \\ -|z_{1}|^{5p-1} -|z_{2}|^{5\frac{1}{p}} &\leq -\left(\frac{5p-1}{2p-1}\right)^{\frac{2p-1}{5p-1}} |z_{1}|^{2p-1} \left(\frac{5p-1}{3p}\right)^{\frac{3p}{5p-1}} |z_{2}|^{3}, \\ -|z_{1}|^{5p-1} -|z_{2}|^{5\frac{1}{p}} &\leq -\left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}} |z_{1}|^{4p-1} \left(\frac{5p-1}{p}\right)^{\frac{3p}{5p-1}} |z_{2}|^{3}, \\ -|z_{1}|^{5p-1} -|z_{2}|^{5\frac{1}{p}} &\leq -\left(\frac{5q-1}{q}\right)^{\frac{q}{5p-1}} |z_{1}|^{4p-1} \left(\frac{5p-1}{4q-1}\right)^{\frac{3q-1}{5p-1}} |z_{2}|^{4\frac{1}{q}}, \\ -|z_{1}|^{5q-1} -|z_{2}|^{5\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q}\right)^{\frac{2q}{5q-1}} |z_{1}|^{2q} \left(\frac{5q-1}{3q-1}\right)^{\frac{3q-1}{5q-1}} |z_{2}|^{3\frac{1}{q}}, \\ -|z_{1}|^{5q-1} -|z_{2}|^{5\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q-1}\right)^{\frac{2q-1}{5q-1}} |z_{1}|^{2q-1} \left(\frac{5q-1}{3q-1}\right)^{\frac{3q-1}{5q-1}} |z_{2}|^{3\frac{1}{q}}, \\ -|z_{1}|^{5q-1} -|z_{2}|^{5\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q-1}\right)^{\frac{2q-1}{5q-1}} |z_{1}|^{2q-1} \left(\frac{5q-1}{3q}\right)^{\frac{3q}{5q-1}} |z_{2}|^{3\frac{1}{q}}, \\ -|z_{1}|^{5q-1} -|z_{2}|^{5\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q-1}\right)^{\frac{5q-1}{5q-1}} |z_{1}|^{4q-1} \left(\frac{5q-1}{3q}\right)^{\frac{3q}{5q-1}} |z_{2}|^{3}, \\ -|z_{1}|^{5p-1} -|z_{1}|^{5q-1} -|z_{2}|^{5\frac{1}{q}} &\leq -\left(\frac{5q-1}{4q-1}\right)^{\frac{5q-1}{5q-1}} |z_{1}|^{4q-1} \left(\frac{5q-1}{q}\right)^{\frac{3q}{5q-1}} |z_{2}|^{3}, \\ -|z_{1}|^{5p-1} -|z_{1}|^{5q-1} -|z_{2}|^{5\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q-1}\right)^{\frac{5q-1}{5q-1}} |z_{1}|^{4q-1} \left(\frac{5q-1}{q}\right)^{\frac{3q}{5q-1}} |z_{2}|^{4\frac{1}{p}}, \\ &\leq -\left(\frac{5p-1}{p}\right)^{\frac{5p-1}{5p-1}} |z_{1}|^{q} \left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}} |z_{2}|^{4\frac{1}{p}}, \end{aligned}$$

$$\begin{split} -|z_{1}|^{5p-1} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{p}} &\leq -|z_{1}|^{5q-\frac{q}{p}} - |z_{2}|^{5-\frac{1}{p}} \\ &\leq -\left(\frac{5p-1}{2p}\right)^{\frac{2p}{5p-1}} |z_{1}|^{2q} \left(\frac{5p-1}{3p-1}\right)^{\frac{3p-1}{5p-1}} |z_{2}|^{3-\frac{1}{p}}, \\ -|z_{1}|^{5p-1} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{p}} &\leq -|z_{1}|^{(2p+2q-1)} \left(\frac{5p-1}{4p-1}\right) - |z_{2}|^{5-\frac{1}{p}} \\ &\leq -\left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}} |z_{1}|^{(2p+2q-1)} \left(\frac{5p-1}{p}\right)^{\frac{p}{5p-1}} |z_{2}|, \\ -|z_{1}|^{5p-1} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -|z_{1}|^{5p-\frac{p}{q}} - |z_{2}|^{5-\frac{1}{q}} \leq -\left(\frac{5q-1}{q}\right)^{\frac{q}{5q-1}} |z_{1}|^{p} \left(\frac{5q-1}{4q-1}\right)^{\frac{4q-1}{5q-1}} |z_{2}|^{4-\frac{1}{q}}, \\ -|z_{1}|^{5p-1} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} \leq -|z_{1}|^{5p-\frac{p}{q}} - |z_{2}|^{5-\frac{1}{q}} \leq -\left(\frac{5q-1}{2q}\right)^{\frac{2q}{5q-1}} |z_{1}|^{2p} \left(\frac{5q-1}{4q-1}\right)^{\frac{3q-1}{5q-1}} |z_{2}|^{3-\frac{1}{q}}, \\ -|z_{1}|^{5p-1} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} \leq -|z_{1}|^{5p-\frac{p}{q}} - |z_{2}|^{5-\frac{1}{q}} \leq -\left(\frac{5q-1}{2q}\right)^{\frac{2q}{5q-1}} |z_{1}|^{2p} \left(\frac{5q-1}{3q-1}\right)^{\frac{3q-1}{5q-1}} |z_{2}|^{3-\frac{1}{q}}, \end{split}$$

where lemma 2.3 has been employed in the last five inequalities since

$$5p-1 \le (2p+2q-1)\left(\frac{5p-1}{4p-1}\right) \le 5q-1, \quad 5p-1 \le 5q-\frac{q}{p} \le 5q-1,$$

$$5p-1 \le 5p-\frac{p}{q} \le 5q-1.$$

Thus, if the following inequalities

$$\min \left\{ \begin{array}{l} \lambda_{1} \frac{k_{1}^{2}(k_{3}-M_{1})}{(k_{3}+M_{1})^{\frac{5p-1}{p}}}, \\ \lambda_{2} \frac{k_{1}^{2}(k_{3}-M_{1})}{(k_{1}+M_{4})^{\frac{5p-1}{2p}}}, \\ \lambda_{3} \frac{k_{1n}^{2}(k_{3n}-M_{2})}{(k_{3n}+M_{2})^{\frac{5q-1}{q}}}, \\ \lambda_{4} \frac{k_{1n}^{2}(k_{3n}-M_{2})}{(k_{1n}+M_{5})^{\frac{5q-1}{2q}}}, \\ \lambda_{5} \frac{\omega_{1}}{(k_{3n}+M_{2})^{\frac{5p-1}{p}}}, \\ \lambda_{6} \frac{\omega_{1}}{(k_{1n}+M_{5})^{\frac{5p-1}{2p}}}, \\ \lambda_{7} \frac{\omega_{1}}{(k_{3}+M_{1})^{\frac{5p-1}{2p}}}, \\ \lambda_{8} \frac{\omega_{1}}{(k_{1}+M_{4})^{\frac{5q-1}{2q}}}, \\ \lambda_{8} \frac{\omega_{1}}{(k_{1}+M_{4})^{\frac{5q-1}{2q}}}, \end{array} \right\} > r > \max \left\{ \begin{array}{l} \lambda_{9} \frac{M_{4}^{\frac{5p-1}{3p}}}{k_{1}^{\frac{4p-2}{3p}}(k_{3}-M_{1})^{\frac{2p-1}{3p}}}, \\ \lambda_{10} \frac{M_{4}^{\frac{5p-1}{p}}}{k_{1}^{\frac{3p-1}{p}}(k_{3}-M_{1})^{\frac{4p-1}{p}}}, \\ \lambda_{10} \frac{M_{5}^{\frac{5q-1}{2q}}}{k_{1n}^{\frac{4q-2}{2}}(k_{3}-M_{1})^{\frac{4p-1}{p}}}, \\ \lambda_{10} \frac{M_{5}^{\frac{5q-1}{3q}}}{k_{1n}^{\frac{3q-1}{q}}(k_{3n}-M_{2})^{\frac{4q-1}{q}}}, \\ \lambda_{11} \frac{M_{5}^{\frac{5q-1}{q}}}{k_{1n}^{\frac{3q-1}{q}}(k_{3n}-M_{2})^{\frac{4q-1}{q}}}, \\ \lambda_{13} \frac{\left(M_{4}\frac{k_{1n}}{q}+M_{5}\frac{k_{1}}{p}\right)^{\frac{5p-1}{p}}}{\omega_{1}^{\frac{4p-1}{p}}}}\right\}$$

$$(2 - 25)$$
$$\begin{split} \lambda_{1} &= \left(\frac{5p-1}{18p^{2}}\right) \left(\frac{5p-1}{56p-14}\right)^{\frac{4p-1}{p}}, \ \lambda_{2} &= \left(\frac{5p-1}{36p^{2}}\right) \left(\frac{5p-1}{42p-14}\right)^{\frac{3p-1}{2p}} \left(\frac{p}{4p-1}\right)^{\frac{5p-1}{2p}}, \\ \lambda_{3} &= \left(\frac{5q-1}{18q^{2}}\right) \left(\frac{5q-1}{48q-12}\right)^{\frac{4q-1}{q}}, \ \lambda_{4} &= \left(\frac{5q-1}{36q^{2}}\right) \left(\frac{5q-1}{36q-12}\right)^{\frac{3q-1}{2q}} \left(\frac{q}{4q-1}\right)^{\frac{5q-1}{2q}}, \\ \lambda_{5} &= \left(\frac{5p-1}{p}\right) \left(\frac{5p-1}{56p-14}\right)^{\frac{4p-1}{p}}, \ \lambda_{6} &= \left(\frac{5p-1}{2p}\right) \left(\frac{5p-1}{42p-14}\right)^{\frac{3p-1}{2p}} \left(\frac{p}{4p-1}\right)^{\frac{5p-1}{2p}}, \\ \lambda_{7} &= \left(\frac{5q-1}{q}\right) \left(\frac{5q-1}{48q-12}\right)^{\frac{4q-1}{q}}, \ \lambda_{8} &= \left(\frac{5q-1}{2q}\right) \left(\frac{5q-1}{36q-12}\right)^{\frac{3q-1}{2q}} \left(\frac{q}{4q-1}\right)^{\frac{5q-1}{2p}}, \\ \lambda_{9} &= \left(\frac{36p^{2}-18p}{5p-1}\right)^{\frac{2p-1}{3p}} \left(\frac{42p}{5p-1}\right), \ \lambda_{10} &= \left(\frac{72p^{2}-18p}{5p-1}\right)^{\frac{4p-1}{p}} \left(\frac{14p}{5p-1}\right) \left(\frac{1}{p}\right)^{\frac{5q-1}{p}}, \\ \lambda_{11} &= \left(\frac{36q^{2}-18q}{5q-1}\right)^{\frac{2q-1}{3q}} \left(\frac{36q}{5q-1}\right), \ \lambda_{12} &= \left(\frac{72q^{2}-18q}{5q-1}\right)^{\frac{4q-1}{q}} \left(\frac{12q}{5q-1}\right) \left(\frac{1}{q}\right)^{\frac{5q-1}{q}}, \\ \lambda_{13} &= \left(\frac{4p-1}{5p-1}\right)^{\frac{4p-1}{p}} \left(\frac{14p}{5p-1}\right), \ \omega_{1} &= \min\left\{\frac{k_{1}^{2}}{18p}(k_{3}-M_{1}), \ k_{1n}^{2}} \left(k_{3n}-M_{2}\right)\right\} \end{split}$$

hold then the function \dot{V}_1 is negative definite. Then,

$$\begin{split} \dot{\tilde{V}} &\leq -\frac{k_{1}^{2}}{2p} (k_{3} - M_{1}) |z_{1}|^{5p-1} - \frac{k_{1n}^{2}}{2q} (k_{3n} - M_{2}) |z_{1}|^{5q-1} - \frac{r}{2} |z_{2}|^{5-\frac{1}{p}} - \frac{r}{2} |z_{2}|^{5-\frac{1}{q}} \\ &+ M_{3} \left(\frac{k_{1}^{2}}{p} |z_{1}|^{4p-1} + \frac{k_{1n}^{2}}{q} |z_{1}|^{4q-1} + k_{1} |z_{1}|^{2p-1} |z_{2}|^{2} + k_{1n} |z_{1}|^{2q-1} |z_{2}|^{2} \\ &+ k_{1} k_{1n} \left(\frac{1}{p} + \frac{1}{q} \right) |z_{1}|^{2p+2q-1} + r |z_{2}|^{4-\frac{1}{p}} + r |z_{2}|^{4-\frac{1}{q}} \\ &+ M_{6} \left(\frac{|z_{2}|^{3} + \frac{k_{1}}{p} |z_{1}|^{2p} |z_{2}| + \frac{k_{1n}}{q} |z_{1}|^{2q} |z_{2}| + r \left(4 - \frac{1}{p} \right) |z_{1}| |z_{2}|^{3-\frac{1}{p}} \\ &+ r \left(4 - \frac{1}{q} \right) |z_{1}| |z_{2}|^{3-\frac{1}{q}} \end{split} \end{split}$$

Applying lemma 2.1,

$$\begin{split} |z_{1}|^{2p-1}|z_{2}|^{2} &\leq \left(\frac{2p-1}{4p-1}\right)|z_{1}|^{4p-1} + \left(\frac{2p}{4p-1}\right)|z_{2}|^{\frac{4p-1}{p}}, \\ |z_{1}|^{2q-1}|z_{2}|^{2} &\leq \left(\frac{2q-1}{4q-1}\right)|z_{1}|^{4q-1} + \left(\frac{2q}{4q-1}\right)|z_{2}|^{\frac{4q-1}{q}}, \\ |z_{1}|^{2p+2q-1} &\leq \frac{1}{2}|z_{1}|^{4p-1} + \frac{1}{2}|z_{1}|^{4q-1}, \\ |z_{1}|^{2p}|z_{2}| &\leq \frac{2}{3}|z_{1}|^{3p} + \frac{1}{3}|z_{2}|^{3}, \\ |z_{1}|^{2q}|z_{2}| &\leq \frac{2}{3}|z_{1}|^{3q} + \frac{1}{3}|z_{2}|^{3}, \\ |z_{1}||z_{2}|^{3\frac{1}{p}} &\leq \left(\frac{1}{3p}\right)|z_{1}|^{3p} + \left(\frac{3p-1}{3p}\right)|z_{2}|^{3}, \\ |z_{1}||z_{2}|^{3\frac{1}{q}} &\leq \left(\frac{1}{3q}\right)|z_{1}|^{3q} + \left(\frac{3q-1}{3q}\right)|z_{2}|^{3}, \end{split}$$

and lemma 2.3,

$$-|z_1|^{5p-1} - |z_1|^{5q-1} \le -|z_1|^{5p-\frac{p}{q}}, \text{ and } -|z_1|^{5p-1} - |z_1|^{5q-1} \le -|z_1|^{5q-\frac{q}{p}},$$

together with lemma 2.2, and the bounds of the Lyapunov function, we have:

$$\begin{split} \dot{\widetilde{V}} &\leq -\omega_{2} \bigg(\left| z_{1} \right|^{5p-1} + \left| z_{1} \right|^{5q-\frac{q}{p}} + \left| z_{2} \right|^{5-\frac{1}{p}} \bigg) - \omega_{3} \bigg(\left| z_{1} \right|^{5q-1} + \left| z_{1} \right|^{5p-\frac{p}{q}} + \left| z_{2} \right|^{5-\frac{1}{q}} \bigg) \\ &+ M_{3} \omega_{4} \bigg(\left| z_{1} \right|^{4p-1} + \left| z_{1} \right|^{4q-1} + \left| z_{2} \right|^{\frac{4p-1}{p}} + \left| z_{2} \right|^{\frac{4q-1}{q}} \bigg) + M_{6} \omega_{5} \bigg(\left| z_{1} \right|^{3p} + \left| z_{1} \right|^{3q} + \left| z_{2} \right|^{3} \bigg) \\ &= -\omega_{2} \bigg(\bigg(\left| z_{1} \right|^{4p} \bigg)^{\frac{5p-1}{4p}} + \bigg(\left| z_{1} \right|^{4q} \bigg)^{\frac{5p-1}{4p}} + \bigg(\left| z_{2} \right|^{4} \bigg)^{\frac{5q-1}{4p}} \bigg) \\ &- \omega_{3} \bigg(\bigg(\left| z_{1} \right|^{4p} \bigg)^{\frac{5q-1}{4q}} + \bigg(\left| z_{1} \right|^{4p} \bigg)^{\frac{5q-1}{4q}} + \bigg(\left| z_{2} \right|^{4} \bigg)^{\frac{5q-1}{4q}} \bigg) \\ &+ M_{3} \omega_{4} \bigg(\bigg(\left| z_{1} \right|^{4p} \bigg)^{\frac{4p-1}{4p}} + \bigg(\left| z_{1} \right|^{4q} \bigg)^{\frac{4q-1}{4q}} + \bigg(\left| z_{2} \right|^{4} \bigg)^{\frac{4p-1}{4p}} + \bigg(\left| z_{2} \right|^{4} \bigg)^{\frac{4q-1}{4q}} \bigg) \\ &+ M_{6} \omega_{5} \bigg(\bigg(\left| z_{2} \right|^{4} \bigg)^{\frac{3}{4}} + \bigg(\left| z_{1} \right|^{4p} \bigg)^{\frac{3}{4}} + \bigg(\left| z_{1} \right|^{4q} \bigg)^{\frac{3}{4}} \bigg) \end{split}$$

$$\leq -\left(\frac{\omega_{2}}{\overline{\pi}_{1}^{\frac{5p-1}{4p}}}\right)V^{\frac{5p-1}{4p}} - \left(\frac{\omega_{3}}{\frac{q-1}{3}^{\frac{q-1}{4q}}\overline{\pi}_{1}^{\frac{5q-1}{4q}}}\right)V^{\frac{5q-1}{4q}} + M_{3}\left(\frac{2^{\frac{1}{4p}}\omega_{4}}{\underline{\pi}_{1}^{\frac{4p-1}{4p}}}\right)V^{\frac{4p-1}{4p}} + M_{3}\left(\frac{2^{\frac{1}{4p}}\omega_{4}}{\underline{\pi}_{1}^{\frac{4p-1}{4p}}}\right)V^{\frac{4p-1}{4p}} + M_{3}\left(\frac{3^{\frac{1}{4}}\omega_{5}}{\underline{\pi}_{1}^{\frac{3}{4}}}\right)V^{\frac{3}{4}}$$

$$+ M_{3}\left(\frac{2^{\frac{1}{4q}}\omega_{4}}{\underline{\pi}_{1}^{\frac{4q-1}{4q}}}\right)V^{\frac{4q-1}{4q}} + M_{6}\left(\frac{3^{\frac{1}{4}}\omega_{5}}{\underline{\pi}_{1}^{\frac{3}{4}}}\right)V^{\frac{3}{4}}$$

$$(2 - 26)$$

where

$$\begin{split} \omega_{2} &= \min\left\{\frac{7k_{1}^{2}}{18p}(k_{3}-M_{1}), \, \omega_{1}, \, \frac{r}{2}\right\}, \, \omega_{3} = \min\left\{\frac{7k_{1n}^{2}}{18q}(k_{3n}-M_{2}), \, \omega_{1}, \, \frac{r}{2}\right\} \\ \omega_{4} &= \max\left\{\frac{\left(\frac{k_{1}^{2}}{p}+k_{1}\left(\frac{2p-1}{4p-1}\right)+\frac{k_{1}k_{1n}}{2}\left(\frac{1}{p}+\frac{1}{q}\right)\right), \\ \left(\frac{k_{1n}^{2}}{q}+k_{1n}\left(\frac{2q-1}{4q-1}\right)+\frac{k_{1}k_{1n}}{2}\left(\frac{1}{p}+\frac{1}{q}\right)\right), \\ \left(r+k_{1}\left(\frac{2p}{4p-1}\right)\right), \, \left(r+k_{1n}\left(\frac{2q}{4q-1}\right)\right) \\ \omega_{5} &= \max\left\{\frac{\left(\frac{2k_{1}}{3p}+r\left(4-\frac{1}{p}\right)\left(\frac{1}{3p}\right)\right), \left(\frac{2k_{1n}}{3q}+r\left(4-\frac{1}{q}\right)\left(\frac{1}{3q}\right)\right), \\ \left(1+\frac{k_{1}}{3p}+\frac{k_{1n}}{3q}+r\left(4-\frac{1}{p}\right)\left(\frac{3p-1}{3p}\right)+r\left(4-\frac{1}{q}\right)\left(\frac{3q-1}{3q}\right)\right) \\ \end{split}\right\}$$

Remark 2.21. Note that the nonlinear inequalities (2 - 24) and (2 - 25) is feasible for sufficiently large k_1 , k_{1n} , k_3 , k_{3n} . Thus, an r > 0 always exists. Consider some particular cases.

- 1. Note that (2 25) is feasible with respect to k_1 , k_{1n} , k_3 , k_{3n} for any $M_1 \ge 0$, $M_2 \ge 0$, M_4 ≥ 0 , and $M_5 \ge 0$.
- 2. If $M_4 = M_5 = 0$, then (2 25) is satisfied for any $k_3 > M_1$ and $k_{3n} > M_2$. This coincides with the conditions obtained through the weak Lyapunov function, i.e. the energy

function
$$E(z_1, z_2) = \frac{1}{2} z_2^2 + \frac{k_1}{2p} |z_1|^{2p} + \frac{k_{1n}}{2q} |z_1|^{2q}$$

Remark 2.22. Hence, when $M_3 = M_6 = 0$, i.e. without persistent perturbations, from lemma 2.5-3,

1. For 0 , and <math>q > 1 the system exhibit uniform asymptotical convergence, where

$$T_{\max}(\mu) = \left(\frac{3^{\frac{q-1}{4q}} \bar{\pi}_1 \frac{5q-1}{4q}}{\omega_3}\right) \left(\frac{4q}{q-1}\right) \left(\frac{1}{\mu^{\frac{q-1}{4q}}}\right),$$

is the time at which the trajectories reach the surface level $V = \mu$. At the same time, the system (2 - 23) will have finite time convergence, in particular after reaching the surface level $V = \mu$, from lemma 2.5-3, the settling time estimate,

$$T(\mu) \leq \left(\frac{\overline{\pi}_1^{\frac{5p-1}{4p}}}{\omega_2}\right) \left(\frac{4p}{1-p}\right) [\mu]^{\frac{1-p}{4p}},$$

where the initial starting states is changed to $V = \mu$. Hence, the total time to reach

the origin can be estimated as
$$T_{total}(\mu) = \left(\frac{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}{\omega_3}\right) \left(\frac{4q}{q-1}\right) \left(\frac{1}{\mu^{\frac{q-1}{4q}}}\right) + \left(\frac{\overline{\pi}_1 \frac{5p-1}{4p}}{\omega_2}\right) \left(\frac{4p}{1-p}\right) [\mu]^{\frac{1-p}{4p}}.$$

The minimum of this function can be found at $\mu = \left(\frac{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}{\omega_3}\right)^{\frac{4pq}{q-p}} \left(\frac{\omega_2}{\overline{\pi}_1 \frac{5p-1}{4p}}\right)^{\frac{4pq}{q-p}}.$

Substituting into the function, a finite settling time independent of initial conditions is obtained,

$$\begin{split} T_{total} = & \left(\frac{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}{\omega_3}\right) \left(\frac{4q}{q-1}\right) \left(\frac{\omega_3}{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}\right)^{\frac{p(q-1)}{q-p}} \left(\frac{\overline{\pi}_1 \frac{5p-1}{4p}}{\omega_2}\right)^{\frac{p(q-1)}{q-p}} \\ &+ \left(\frac{\overline{\pi}_1 \frac{5p-1}{4p}}{\omega_2}\right) \left(\frac{4p}{1-p}\right) \left(\frac{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}{\omega_3}\right)^{\frac{q(1-p)}{q-p}} \left(\frac{\omega_2}{\overline{\pi}_1 \frac{5p-1}{4p}}\right)^{\frac{q(1-p)}{q-p}} \end{split}$$

- 2. For 0 , and <math>q = 1, finite time convergence can be concluded.
- 3. For p = 1, and $q \ge 1$, exponential convergence can be concluded.

The above results are possible due to the negative definiteness of the time derivative of the Lyapunov function, i.e. strict Lyapunov function.

Remark 2.23. While if persistent perturbations occur on the system, $M_3 \neq 0$ and/or $M_6 \neq 0$, from (2 - 26),

$$\begin{split} \hat{\tilde{V}} &\leq -\frac{1}{2} \left(\frac{\omega_2}{\overline{\pi}_1^{\frac{5p-1}{4p}}} \right) V^{\frac{5p-1}{4p}} - \frac{1}{3} \left(\frac{\omega_3}{\frac{q^{-1}}{3^{\frac{4q}{\pi}_1} \overline{\pi}_1^{\frac{5q-1}{4q}}}} \right) V^{\frac{5q-1}{4q}} - V^{\frac{4p-1}{4p}} \left(\frac{1}{2} \left(\frac{\omega_2}{\overline{\pi}_1^{\frac{5p-1}{4p}}} \right) V^{\frac{1}{4}} - M_3 \left(\frac{2^{\frac{1}{4p}} \omega_4}{\underline{\pi}_1^{\frac{4p-1}{4p}}} \right) \right) \\ &- V^{\frac{4q-1}{4q}} \left(\frac{1}{3} \left(\frac{\omega_3}{\frac{q^{-1}}{3^{\frac{4q}{4q}} \overline{\pi}_1^{\frac{5q-1}{4q}}}} \right) V^{\frac{1}{4}} - M_3 \left(\frac{2^{\frac{1}{4q}} \omega_4}{\underline{\pi}_1^{\frac{4q-1}{4q}}} \right) \right) - V^{\frac{3}{4}} \left(\frac{1}{3} \left(\frac{\omega_3}{\frac{q^{-1}}{3^{\frac{4q}{4q}} \overline{\pi}_1^{\frac{5q-1}{4p}}}} \right) V^{\frac{5q-1}{4q}} - M_6 \left(\frac{3^{\frac{1}{4}} \omega_5}{\underline{\pi}_1^{\frac{3}{4}}} \right) \right) \\ &\leq -\frac{1}{2} \left(\frac{\omega_2}{\overline{\pi}_1^{\frac{5p-1}{4p}}} \right) V^{\frac{5p-1}{4p}} - \frac{1}{3} \left(\frac{\omega_3}{\frac{q^{-1}}{3^{\frac{4q}{4q}} \overline{\pi}_1^{\frac{5q-1}{4q}}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_2}{\overline{\pi}_1^{\frac{5p-1}{4p}}} \right) V^{\frac{5p-1}{4p}} - \frac{1}{3} \left(\frac{\omega_3}{\frac{q^{-1}}{3^{\frac{4q}{4q}} \overline{\pi}_1^{\frac{5q-1}{4q}}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_2}{\overline{\pi}_1^{\frac{5p-1}{4p}}} \right) V^{\frac{5p-1}{4q}} - \frac{1}{3} \left(\frac{2^{\frac{1}{4p}} \omega_3}{\frac{q^{-1}}{3^{\frac{4q}{4q}} \overline{\pi}_1^{\frac{5q-1}{4q}}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{3} \left(\frac{2M_3 \left(\frac{\overline{\pi}_1^{\frac{5p-1}{4p}}}{\overline{\omega}_2} \right) \left(2^{\frac{1}{4p}} \omega_4}{\frac{\overline{\pi}_1^{\frac{4p-1}{4p}}}{\overline{\omega}_1^{\frac{4p-1}{4p}}}} \right) \right)^4 , \\ &\int (3M_3 \left(\frac{3M_3 \left(\frac{3^{\frac{4q}{4q}} \overline{\pi}_1^{\frac{5q-1}{4q}}}{\overline{\omega}_3} \right) \left(2^{\frac{1}{4q}} \omega_4}{\frac{\overline{\pi}_1^{\frac{4q-1}{4q}}}{\overline{\omega}_3} \right) \right)^4 , \\ &\int (3M_6 \left(\frac{3M_6 \left(\frac{3^{\frac{4q}{4q}} \overline{\pi}_1^{\frac{5q-1}{4q}}}{\overline{\omega}_3} \right) \left(\frac{3^{\frac{4}{4}} \omega_5}{\underline{\pi}_1^{\frac{3}{4}}} \right) \right)^{\frac{4q}{2q-1}} \right) V^{\frac{4q}{2q-1}} \right) V^{\frac{4q}{2q-1}} \right) \right)^4 \\ &= 0 \quad (M_6 \left(\frac{3M_6 \left(\frac{3^{\frac{4q}{4q}} \overline{\pi}_1^{\frac{5q-1}{4q}}}}{\overline{\omega}_3} \right) \left(\frac{3^{\frac{4}{4}} \omega_5}{\underline{\pi}_1^{\frac{3}{4}}} \right) \right)^{\frac{4q}{2q-1}} \right) V^{\frac{4q}{2q-1}} \right) V^{\frac{4q}{2q-1}} \right) V^{\frac{4q}{2q-1}}$$

thus, uniform ultimate boundedness is achieved by applying lemma 2.6.

b) Case 2: For p = 0.5, and $1 \le q$

For p = 0.5, V is not differentiable on $z_1 = 0$, hence

$$\dot{\widetilde{V}}(z_1, z_2) = \bigcap_{\xi \in \partial V(\mathbf{z}(t), t)} \xi^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix} \subset \dot{\widetilde{V}}_1 + \dot{\widetilde{V}}_2$$

where

$$\dot{\widetilde{V}}_1 = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^T K[f](z_1, z_2), \quad \dot{\widetilde{V}}_2 = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^T K\begin{bmatrix}d_1\\d_2\end{bmatrix},$$

$$K[f](z_1, z_2) \in \begin{bmatrix} -k_3 |z_1|^{0.5} \operatorname{sign}(z_1) - k_{3n} |z_1|^q \operatorname{sign}(z_1) + z_2 \\ -k_1 \operatorname{SGN}(z_1) - k_{1n} |z_1|^{2q-1} \operatorname{sign}(z_1) \end{bmatrix}$$

$$\begin{split} \partial V &= \mathbf{K} [\nabla V] = \mathbf{K} \left[\frac{\partial V}{\partial z_1} \right] \\ &\subset \left[\mathbf{K} \left[\frac{\partial V}{\partial z_1} \right] \right] \\ &\mathbf{K} \left[\frac{\partial V}{\partial z_2} \right] \right] \\ &= \begin{cases} \forall z_1 \neq 0, z_2 \in \mathbb{R} : \\ \left[\left(z_2^2 + 2k_1 |z_1| + \frac{k_{1n}}{q} |z_1|^{2q} \right) (k_1 + k_{1n} |z_1|^{2q-1}) \operatorname{sign}(z_1) - r |z_2|^2 \operatorname{sign}(z_2) - r |z_2|^{4-\frac{1}{q}} \operatorname{sign}(z_2) \right] \\ &\left(z_2^2 + 2k_1 |z_1| + \frac{k_{1n}}{q} |z_1|^{2q} \right) (z_2) - 2r z_1 |z_2| - r \left(4 - \frac{1}{q} \right) z_1 |z_2|^{3-\frac{1}{q}} \\ \forall z_1 = 0, z_2 \in \mathbb{R} : \\ &\left[z_2^2 k_1 [-1,1] - r |z_2|^2 \operatorname{sign}(z_2) - r |z_2|^{4-\frac{1}{q}} \operatorname{sign}(z_2) \right] \end{cases} \end{split}$$

Let us define:

$$\left|\frac{\partial V}{\partial z_1}\right| := \sup\left\{\left|\xi_1\right| : \xi_1 \in \mathbf{K}\left[\frac{\partial V}{\partial z_1}\right]\right\}, \text{ and } \left|\frac{\partial V}{\partial z_2}\right| := \sup\left\{\left|\xi_2\right| : \xi_2 \in \mathbf{K}\left[\frac{\partial V}{\partial z_2}\right]\right\},\$$

with

$$\left|\frac{\partial V}{\partial z_1}\right| \le 2k_1^2 |z_1| + k_1 k_{1n} \left(\frac{1}{q} + 2\right) |z_1|^{2q} + \frac{k_{1n}^2}{q} |z_1|^{4q-1} + k_{1n} |z_1|^{2q-1} |z_2|^2 + k_1 |z_2|^2 + r|z_2|^2 + r|z_2|^{4-\frac{1}{q}},$$

and
$$\left|\frac{\partial V}{\partial z_2}\right| \le |z_2|^3 + 2k_1|z_1||z_2| + \frac{k_{1n}}{q}|z_1|^{2q}|z_2| + 2r|z_1||z_2| + r\left(4 - \frac{1}{q}\right)|z_1||z_2|^{3-\frac{1}{q}}$$

Thus, the term

$$\begin{split} \dot{\tilde{V}}_{2} &= \bigcap_{(\xi_{1},\xi_{2})^{T} \in \partial V(\mathbf{z}(t),t)} \xi_{1}K[d_{1}] + \xi_{2}K[d_{2}] \\ &\leq \left| \frac{\partial V}{\partial z_{1}} \right| d_{1} | + \left| \frac{\partial V}{\partial z_{2}} \right| d_{2} | \\ &\leq \left(M_{1} |z_{1}|^{0.5} + M_{2} |z_{1}|^{q} + M_{3} \left(2k_{1}^{2} |z_{1}| + k_{1}k_{1n} \left(\frac{1}{q} + 2 \right) |z_{1}|^{2q} + \frac{k_{1n}^{2}}{q} |z_{1}|^{4q-1} + k_{1n} |z_{1}|^{2q-1} |z_{2}|^{2} \right) \\ &+ \left(M_{4} + M_{5} |z_{1}|^{2q-1} + M_{6} \left(|z_{2}|^{3} + 2k_{1} |z_{1}| |z_{2}| + \frac{k_{1n}}{q} |z_{1}|^{2q} |z_{2}| + 2r |z_{1}| |z_{2}| + r \left(4 - \frac{1}{q} \right) |z_{1}| |z_{2}|^{3-\frac{1}{q}} \right) \end{split}$$

Computing $\dot{\tilde{V}}_1$ for each case, we have

For $z_1 \neq 0, z_2 \in \mathbb{R}$:

$$\begin{split} \dot{\tilde{V}_{1}} &= -k_{1}k_{3}|z_{1}|^{0.5}|z_{2}|^{2} - k_{1}k_{3n}|z_{1}|^{q}|z_{2}|^{2} - k_{1n}k_{3}|z_{1}|^{2q-0.5}|z_{2}|^{2} - k_{1n}k_{3n}|z_{1}|^{3q-1}|z_{2}|^{2} - 2k_{1}^{2}k_{3}|z_{1}|^{1.5} \\ &- 2k_{1}^{2}k_{3n}|z_{1}|^{1+q} - k_{1}k_{1n}k_{3}\left(2 + \frac{1}{q}\right)|z_{1}|^{0.5+2q} - 2k_{1}k_{1n}k_{3n}|z_{1}|^{3q} - \frac{k_{1n}k_{1}k_{3n}}{q}|z_{1}|^{3q} \\ &- \frac{k_{1n}k_{1n}k_{3}}{q}|z_{1}|^{4q-0.5} - \frac{k_{1n}k_{1n}k_{3n}}{q}|z_{1}|^{5q-1} - r|z_{2}|^{3} - r|z_{2}|^{5-\frac{1}{q}} + rk_{3}|z_{1}|^{0.5}|z_{2}|^{2}\operatorname{sign}(z_{1}z_{2}) \\ &+ rk_{3n}|z_{1}|^{q}|z_{2}|^{2}\operatorname{sign}(z_{1}z_{2}) + r2k_{1}|z_{1}||z_{2}| + r2k_{1n}|z_{1}|^{2q}|z_{2}| \\ &+ rk_{3}|z_{1}|^{0.5}|z_{2}|^{4-\frac{1}{q}}\operatorname{sign}(z_{1}z_{2}) + rk_{3n}|z_{1}|^{q}|z_{2}|^{4-\frac{1}{q}}\operatorname{sign}(z_{1}z_{2}) \\ &+ r\left(4 - \frac{1}{q}\right)k_{1}|z_{1}||z_{2}|^{3-\frac{1}{q}} + r\left(4 - \frac{1}{q}\right)k_{1n}|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{q}} \end{split}$$

For $z_1 = 0, z_2 \in \mathbb{R}$:

Let
$$\begin{bmatrix} z_2^{2}k_1\xi_2 - r|z_2|^2\operatorname{sign}(z_2) - r|z_2|^{4-\frac{1}{q}}\operatorname{sign}(z_2) \\ z_2^{3} \end{bmatrix}$$
 with $\xi_2 \in [-1, 1]$ be an arbitrary element of

 $\partial V(z_1, z_2)$, then

$$\xi^{\mathrm{T}}K[f](z_{1},z_{2}) = \begin{bmatrix} \left(z_{2}^{2}k_{1}\xi_{2}-r|z_{2}|^{2}\operatorname{sign}(z_{2})-r|z_{2}|^{4-\frac{1}{q}}\operatorname{sign}(z_{2})\right) \\ z_{2}^{3} \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} z_{2} \\ -k_{1}[-1,1] \end{bmatrix}$$
$$= [\xi_{2}-1,\xi_{2}+1]k_{1}z_{2}^{3}-r|z_{2}|^{3}-r|z_{2}|^{5-\frac{1}{q}}$$

hence,

$$\dot{\tilde{V}}_{1} = \bigcap_{\xi_{2} \in [-1, 1]} [\xi_{2} - 1, \xi_{2} + 1] k_{1} z_{2}^{3} - r |z_{2}|^{3} - r |z_{2}|^{5 - \frac{1}{q}} = -r |z_{2}|^{3} - r |z_{2}|^{5 - \frac{1}{q}}$$

Thus, for all $(z_1, z_2) \in \mathbb{R}^2$:

$$\begin{split} \dot{\tilde{V}}_{1} &= -k_{1}k_{3}|z_{1}|^{0.5}|z_{2}|^{2} - k_{1}k_{3n}|z_{1}|^{q}|z_{2}|^{2} - k_{1n}k_{3}|z_{1}|^{2q-0.5}|z_{2}|^{2} - k_{1n}k_{3n}|z_{1}|^{3q-1}|z_{2}|^{2} - 2k_{1}^{2}k_{3}|z_{1}|^{1.5} \\ &- 2k_{1}^{2}k_{3n}|z_{1}|^{1+q} - k_{1}k_{1n}k_{3}\left(2 + \frac{1}{q}\right)|z_{1}|^{0.5+2q} - 2k_{1}k_{1n}k_{3n}|z_{1}|^{3q} - \frac{k_{1n}k_{1}k_{3n}}{q}|z_{1}|^{3q} \\ &- \frac{k_{1n}k_{1n}k_{3}}{q}|z_{1}|^{4q-0.5} - \frac{k_{1n}k_{1n}k_{3n}}{q}|z_{1}|^{5q-1} - r|z_{2}|^{3} - r|z_{2}|^{5-\frac{1}{q}} + rk_{3}|z_{1}|^{0.5}|z_{2}|^{2}\operatorname{sign}(z_{1}z_{2}) \\ &+ rk_{3n}|z_{1}|^{q}|z_{2}|^{2}\operatorname{sign}(z_{1}z_{2}) + r2k_{1}|z_{1}||z_{2}| + r2k_{1n}|z_{1}|^{2q}|z_{2}| + rk_{3}|z_{1}|^{0.5}|z_{2}|^{4-\frac{1}{q}}\operatorname{sign}(z_{1}z_{2}) \\ &+ rk_{3n}|z_{1}|^{q}|z_{2}|^{4-\frac{1}{q}}\operatorname{sign}(z_{1}z_{2}) + r\left(4 - \frac{1}{q}\right)k_{1}|z_{1}||z_{2}|^{3-\frac{1}{q}} + r\left(4 - \frac{1}{q}\right)k_{1n}|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{q}} \end{split}$$

Hence, the generalized time derivative of the Lyapunov function, after rearrangement:

$$\begin{split} \dot{\tilde{V}} &= \dot{\tilde{V}_{1}} + \dot{\tilde{V}_{2}} \\ &\leq -k_{1n} \left(k_{3} - M_{1}\right) \left|z_{1}\right|^{2q-0.5} \left|z_{2}\right|^{2} - k_{1n} \left(k_{3n} - M_{2}\right) \left|z_{1}\right|^{3q-1} \left|z_{2}\right|^{2} - k_{1} k_{1n} \left(2 + \frac{1}{q}\right) \left(k_{3} - M_{1}\right) \left|z_{1}\right|^{2q+0.5} \\ &- 2k_{1}^{2} \left(k_{3n} - M_{2}\right) \left|z_{1}\right|^{q+1} - \frac{k_{1n}^{2}}{q} \left(k_{3} - M_{1}\right) \left|z_{1}\right|^{4q-0.5} - \frac{k_{1} k_{1n}}{2} \left(2 + \frac{1}{q}\right) \left(k_{3n} - M_{2}\right) \left|z_{1}\right|^{3q} \\ &- k_{1}^{2} \left(k_{3} - M_{1}\right) \left|z_{1}\right|^{1.5} - \frac{k_{1n}^{2}}{2q} \left(k_{3n} - M_{2}\right) \left|z_{1}\right|^{5q-1} - \frac{r}{2} \left|z_{2}\right|^{3} - \frac{r}{2} \left|z_{2}\right|^{5-\frac{1}{q}} + \dot{V}_{2} \\ &+ M_{3} \left(2k_{1}^{2} \left|z_{1}\right| + k_{1} k_{1n} \left(\frac{1}{q} + 2\right) \left|z_{1}\right|^{2q} + \frac{k_{1n}^{2}}{q} \left|z_{1}\right|^{4q-1} + k_{1n} \left|z_{1}\right|^{2q-1} \left|z_{2}\right|^{2} + \left(k_{1} + r\right) \left|z_{2}\right|^{2} + r \left|z_{2}\right|^{4-\frac{1}{q}}\right) \\ &+ M_{6} \left(\left|z_{2}\right|^{3} + \frac{k_{1n}}{q} \left|z_{1}\right|^{2q} \left|z_{2}\right| + 2\left(k_{1} + r\right) \left|z_{1}\right| \left|z_{2}\right| + r \left(4 - \frac{1}{q}\right) \left|z_{1}\right| \left|z_{2}\right|^{3-\frac{1}{q}}\right) \end{split}$$

where

$$\begin{split} \dot{V}_{2} &= -\frac{k_{1}k_{1n}}{2} \bigg(2 + \frac{1}{q}\bigg) (k_{3n} - M_{2}) |z_{1}|^{3q} - k_{1}^{2} (k_{3} - M_{1}) |z_{1}|^{1.5} - \frac{k_{1n}^{2}}{2q} (k_{3n} - M_{2}) |z_{1}|^{5q-1} \\ &- \frac{r}{4} |z_{2}|^{3} - \frac{r}{2} |z_{2}|^{5 - \frac{1}{q}} - (k_{1}(k_{3} - M_{1}) - r(k_{3} + M_{1})) |z_{1}|^{0.5} |z_{2}|^{2} \\ &- (k_{1}(k_{3n} - M_{2}) - r(k_{3n} + M_{2})) |z_{1}|^{q} |z_{2}|^{2} - \bigg(\frac{r}{4} - M_{4}\bigg) |z_{2}|^{3} \\ &+ 2r(k_{1} + M_{4}) |z_{1}| |z_{2}| + 2r(k_{1n} + M_{5}) |z_{1}|^{2q} |z_{2}| + r(k_{3} + M_{1}) |z_{1}|^{0.5} |z_{2}|^{4 - \frac{1}{q}} \\ &+ r(k_{3n} + M_{2}) |z_{1}|^{q} |z_{2}|^{4 - \frac{1}{q}} + r\bigg(4 - \frac{1}{q}\bigg) (k_{1n} + M_{5}) |z_{1}|^{2q} |z_{2}|^{3 - \frac{1}{q}} \\ &+ r\bigg(4 - \frac{1}{q}\bigg) (k_{1} + M_{4}) |z_{1}| |z_{2}|^{3 - \frac{1}{q}} + 2M_{4}k_{1} |z_{1}| |z_{2}| \\ &+ \bigg(\frac{k_{1n}}{q} M_{4} + 2k_{1} M_{5}\bigg) |z_{1}|^{2q} |z_{2}| + M_{5} |z_{1}|^{2q-1} |z_{2}|^{3} + M_{5} \frac{k_{1n}}{q} |z_{1}|^{4q-1} |z_{2}| \end{split}$$

Applying lemma 2.1,

$$-|z_1|^{1.5} - |z_2|^3 \le -\left(\frac{3}{2}\right)^{\frac{2}{3}} |z_1| (3)^{\frac{1}{3}} |z_2|,$$

$$\begin{split} -|z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{q}\right)^{\frac{q}{5q-1}} |z_{1}|^{q} \left(\frac{5q-1}{4q-1}\right)^{\frac{4q-1}{3q-1}} |z_{2}|^{4-\frac{1}{q}}, \\ -|z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q}\right)^{\frac{2q}{5q-1}} |z_{1}|^{2q} \left(\frac{5q-1}{3q-1}\right)^{\frac{3q-1}{5q-1}} |z_{2}|^{3-\frac{1}{q}}, \\ -|z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q-1}\right)^{\frac{2q-1}{5q-1}} |z_{1}|^{2q-1} \left(\frac{5q-1}{3q}\right)^{\frac{5q}{5q-1}} |z_{2}|^{3}, \\ -|z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{4q-1}\right)^{\frac{4q-1}{5q-1}} |z_{1}|^{4q-1} \left(\frac{5q-1}{3q}\right)^{\frac{5q}{5q-1}} |z_{2}|, \\ -|z_{1}|^{5q-1} - |z_{2}|^{3} &\leq -\left(\frac{5q}{2}\right)^{\frac{2}{3}} |z_{1}|^{2q} (3)^{\frac{1}{3}} |z_{2}|, \\ -|z_{1}|^{3q} - |z_{2}|^{3} &\leq -\left(\frac{3}{2}\right)^{\frac{2}{3}} |z_{1}|^{2q} (3)^{\frac{1}{3}} |z_{2}|, \\ -|z_{1}|^{1.5} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -|z_{1}|^{\frac{5q-1}{2q}} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{q}\right)^{\frac{q}{5q-1}} |z_{1}|^{0.5} \left(\frac{5q-1}{4q-1}\right)^{\frac{4q-1}{5q-1}} |z_{2}|^{4-\frac{1}{q}}, \\ -|z_{1}|^{1.5} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -|z_{1}|^{\frac{5q-1}{2q}} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{q}\right)^{\frac{2q}{5q-1}} |z_{1}|^{0.5} \left(\frac{5q-1}{4q-1}\right)^{\frac{4q-1}{5q-1}} |z_{2}|^{4-\frac{1}{q}}, \\ -|z_{1}|^{1.5} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -|z_{1}|^{\frac{5q-1}{2q}} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q}\right)^{\frac{2q}{5q-1}} |z_{1}|\left(\frac{5q-1}{3q-1}\right)^{\frac{3q-1}{5q-1}} |z_{2}|^{4-\frac{1}{q}}, \\ -|z_{1}|^{1.5} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -|z_{1}|^{\frac{5q-1}{2q}} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q}\right)^{\frac{2q}{5q-1}} |z_{1}|\left(\frac{5q-1}{3q-1}\right)^{\frac{3q-1}{5q-1}} |z_{2}|^{3-\frac{1}{q}}, \\ -|z_{1}|^{1.5} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -|z_{1}|^{\frac{5q-1}{2q}} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q}\right)^{\frac{2q}{5q-1}} |z_{1}|\left(\frac{5q-1}{3q-1}\right)^{\frac{3q-1}{5q-1}} |z_{2}|^{3-\frac{1}{q}}, \\ -|z_{1}|^{1.5} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -|z_{1}|^{\frac{5q-1}{2q}} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q}\right)^{\frac{2q}{5q-1}} |z_{1}|^{\frac{5q-1}{2q-1}} |z_{2}|^{3-\frac{1}{q}}, \\ -|z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -|z_{1}|^{\frac{5q-1}{2q}} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q}\right)^{\frac{5q-$$

where lemma 2.3 has been employed in the last two inequalities since

$$1.5 \le \frac{5q-1}{2q} \le 5q-1,$$

Thus, if the following inequalities

$$\min \left\{ \frac{k_{1}(k_{3}-M_{1})}{(k_{3}+M_{1})}, \frac{k_{1}(k_{3n}-M_{2})}{(k_{3n}+M_{2})}, \\ \lambda_{14}\frac{k_{1}^{2}(k_{3}-M_{1})}{(k_{1}+M_{4})^{\frac{3}{2}}}, \lambda_{15}\frac{k_{1n}^{2}(k_{3n}-M_{2})}{(k_{3n}+M_{2})^{\frac{5q-1}{q}}}, \\ \lambda_{16}\frac{k_{1n}^{2}(k_{3n}-M_{2})}{(k_{1n}+M_{5})^{\frac{5q-1}{2q}}}, \lambda_{17}\frac{k_{1}k_{1n}(k_{3n}-M_{2})}{(k_{1n}+M_{5})^{\frac{3}{2}}}, \\ \lambda_{18}\frac{\omega_{6}}{(k_{3}+M_{1})^{\frac{5q-1}{q}}}, \lambda_{19}\frac{\omega_{6}}{(k_{1}+M_{4})^{\frac{5q-1}{2q}}}, \\ \lambda_{23}\frac{\left(\frac{k_{1n}}{q}M_{4}+2k_{1}M_{5}\right)^{\frac{3}{2}}}{k_{1n}^{2}k_{1n}^{2}(k_{3n}-M_{2})^{2}}, \\ \lambda_{23}\frac{\left(\frac{k_{1n}}{q}M_{4}+2k_{1}M_{5}\right)^{\frac{3}{2}}}{k_{12}^{2}k_{1n}^{2}(k_{3n}-M_{2})^{2}}, \\ \end{array} \right\}$$
(2 - 27)

where

$$\begin{split} \lambda_{14} &= \left(\frac{3}{8}\right) \left(\frac{3}{16}\right)^{\frac{1}{2}} \frac{1}{2^{\frac{3}{2}}}, \ \lambda_{15} = \left(\frac{5q-1}{12q^2}\right) \left(\frac{5q-1}{48q-12}\right)^{\frac{4q-1}{q}}, \\ \lambda_{16} &= \left(\frac{5q-1}{24q^2}\right) \left(\frac{5q-1}{36q-12}\right)^{\frac{3q-1}{2q}} \left(\frac{q}{4q-1}\right)^{\frac{5q-1}{2q}}, \ \lambda_{17} = \left(\frac{3}{8}\right) \left(\frac{2q+1}{q}\right) \left(\frac{3}{16}\right)^{\frac{1}{2}} \left(\frac{1}{2}\right)^{\frac{3}{2}}, \\ \lambda_{18} &= \left(\frac{5q-1}{q}\right) \left(\frac{5q-1}{48q-12}\right)^{\frac{4q-1}{q}}, \ \lambda_{19} = \left(\frac{5q-1}{2q}\right) \left(\frac{5q-1}{36q-12}\right)^{\frac{3q-1}{2q}} \left(\frac{q}{4q-1}\right)^{\frac{5q-1}{2q}}, \\ \lambda_{20} &= \left(\frac{8}{3}\right)^2 \left(\frac{16}{3}\right) 2^3, \ \lambda_{21} = \left(\frac{24q^2-12q}{5q-1}\right)^{\frac{2q-1}{3q}} \left(\frac{36q}{5q-1}\right), \\ \lambda_{22} &= \left(\frac{48q^2-12q}{5q-1}\right)^{\frac{4q-1}{q}} \left(\frac{12q}{5q-1}\right) \left(\frac{1}{q}\right)^{\frac{5q-1}{q}}, \ \lambda_{23} &= \left(\frac{8}{3}\right)^2 \left(\frac{q}{2q+1}\right)^2 \left(\frac{16}{3}\right), \\ \omega_6 &= \min\left\{\frac{k_1^2(k_3-M_1)}{4}, -\frac{k_{1n}^2(k_{3n}-M_2)}{12q}\right\} \end{split}$$

hold then the function \dot{V}_2 is negative definite. Then,

$$\begin{split} \dot{\tilde{V}} &\leq -\frac{k_{1}k_{1n}}{2} \left(2 + \frac{1}{q}\right) (k_{3n} - M_{2}) |z_{1}|^{3q} - k_{1}^{2} (k_{3} - M_{1}) |z_{1}|^{1.5} - \frac{k_{1n}^{2}}{2q} (k_{3n} - M_{2}) |z_{1}|^{5q-1} \\ &\quad -\frac{r}{2} |z_{2}|^{3} - \frac{r}{2} |z_{2}|^{5-\frac{1}{q}} \\ &\quad + M_{3} \left(2k_{1}^{2} |z_{1}| + k_{1}k_{1n} \left(\frac{1}{q} + 2\right) |z_{1}|^{2q} + \frac{k_{1n}^{2}}{q} |z_{1}|^{4q-1} + k_{1n} |z_{1}|^{2q-1} |z_{2}|^{2} \\ &\quad + M_{3} \left(2k_{1}^{2} |z_{1}| + k_{1}k_{1n} \left(\frac{1}{q} + 2\right) |z_{1}|^{2q} + \frac{k_{1n}^{2}}{q} |z_{1}|^{4q-1} + k_{1n} |z_{1}|^{2q-1} |z_{2}|^{2} \\ &\quad + M_{6} \left(|z_{2}|^{3} + \frac{k_{1n}}{q} |z_{1}|^{2q} |z_{2}| + 2(k_{1} + r) |z_{1}| |z_{2}| + r \left(4 - \frac{1}{q}\right) |z_{1}| |z_{2}|^{3-\frac{1}{q}} \right) \end{split}$$

Applying lemma 2.1,

$$\begin{split} |z_{1}|^{2q-1} |z_{2}|^{2} &\leq \left(\frac{2q-1}{4q-1}\right) |z_{1}|^{4q-1} + \left(\frac{2q}{4q-1}\right) |z_{2}|^{\frac{4q-1}{q}}, \\ |z_{1}|^{2q} &= |z_{1}|^{0.5} |z_{1}|^{2q-0.5} \leq \frac{1}{2} |z_{1}| + \frac{1}{2} |z_{1}|^{4q-1}, \\ |z_{1}|^{2q} |z_{2}| &\leq \frac{2}{3} |z_{1}|^{3q} + \frac{1}{3} |z_{2}|^{3}, \\ |z_{1}||z_{2}| &\leq \frac{2}{3} |z_{1}|^{\frac{3}{2}} + \frac{1}{3} |z_{2}|^{3}, \\ |z_{1}||z_{2}|^{3\frac{1}{q}} \leq \left(\frac{1}{3q}\right) |z_{1}|^{3q} + \left(\frac{3q-1}{3q}\right) |z_{2}|^{3}, \end{split}$$

and lemma 2.3

$$-|z_1|^{1.5} - |z_1|^{5q-1} \le -|z_1|^{\frac{5q-1}{2q}}$$

together with lemma 2.2, and the bounds of the Lyapunov function, we have:

$$\begin{split} \dot{\tilde{V}} &\leq -\omega_7 \left(\left| z_1 \right|^{1.5} + \left| z_1 \right|^{3q} + \left| z_2 \right|^3 \right) - \omega_8 \left(\left| z_1 \right|^{5q-1} + \left| z_1 \right|^{\frac{5q-1}{2q}} + \left| z_2 \right|^{5-\frac{1}{q}} \right) \\ &+ M_3 \omega_9 \left(\left| z_1 \right| + \left| z_2 \right|^2 + \left| z_1 \right|^{4q-1} + \left| z_2 \right|^{4-\frac{1}{q}} \right) + M_6 \omega_{10} \left(\left| z_1 \right|^{\frac{3}{2}} + \left| z_1 \right|^{3q} + \left| z_2 \right|^3 \right) \\ &= -\omega_7 \left(\left(\left| z_1 \right|^2 \right)^{\frac{3}{4}} + \left(\left| z_1 \right|^{4q} \right)^{\frac{3}{4}} + \left(\left| z_2 \right|^4 \right)^{\frac{3}{4}} \right) - \omega_8 \left(\left(\left| z_1 \right|^{4q} \right)^{\frac{5q-1}{4q}} + \left(\left| z_2 \right|^4 \right)^{\frac{5q-1}{4q}} + \left| z_2 \right|^4 \right)^{\frac{5q-1}{4q}} \right) \\ &+ M_3 \omega_9 \left(\left(\left| z_1 \right|^2 \right)^{\frac{1}{2}} + \left(\left| z_2 \right|^4 \right)^{\frac{1}{2}} + \left(\left| z_1 \right|^{4q} \right)^{\frac{4q-1}{4q}} + \left| z_2 \right|^4 \right)^{\frac{4q-1}{4q}} \right) \\ &+ M_6 \omega_{10} \left(\left(\left| z_1 \right|^2 \right)^{\frac{3}{4}} + \left(\left| z_1 \right|^{4q} \right)^{\frac{3}{4}} + \left(\left| z_2 \right|^4 \right)^{\frac{3}{4}} \right) \end{split}$$

$$\leq -\left(\frac{\omega_{7}}{\overline{\pi}_{1}^{\frac{3}{4}}}\right) V^{\frac{3}{4}} - \left(\frac{\omega_{8}}{3^{\frac{q-1}{4q}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}\right) V^{\frac{5q-1}{4q}} + M_{3} \left(\frac{2^{\frac{1}{2}} \omega_{9}}{\underline{\pi}_{1}^{\frac{1}{2}}}\right) V^{\frac{1}{2}} + M_{3} \left(\frac{2^{\frac{1}{4q}} \omega_{9}}{\underline{\pi}_{1}^{\frac{1}{4}}}\right) V^{\frac{4q-1}{4q}} + M_{6} \left(\frac{3^{\frac{1}{4}} \omega_{10}}{\underline{\pi}_{1}^{\frac{3}{4}}}\right) V^{\frac{3}{4}}$$

$$(2 - 28)$$

where

$$\begin{split} \omega_{7} &= \min\left\{\frac{k_{1}^{2}(k_{3}-M_{1})}{4}, \frac{k_{1}k_{1n}(k_{3n}-M_{2})}{2}\left(2+\frac{1}{q}\right), \frac{r}{2}\right\},\\ \omega_{8} &= \min\left\{\frac{5k_{1n}^{2}(k_{3n}-M_{2})}{12q}, \omega_{6}, \frac{r}{2}\right\},\\ \omega_{9} &= \max\left\{\left(2k_{1}^{2}+\frac{k_{1}k_{1n}}{2}\left(\frac{2q+1}{q}\right)\right), (k_{1}+r),\\ \left(\frac{k_{1n}^{2}}{q}+k_{1n}\left(\frac{2q-1}{4q-1}\right)+\frac{k_{1}k_{1n}}{2}\left(\frac{2q+1}{q}\right)\right), \left(k_{1n}\left(\frac{2q}{4q-1}\right)+r\right)\right\},\\ \omega_{10} &= \max\left\{\frac{4(k_{1}+r)}{3}, \left(\frac{k_{1n}}{q}\frac{2}{3}+r\left(4-\frac{1}{q}\right)\left(\frac{1}{3q}\right)\right), \left(1+\frac{k_{1n}}{3q}+\frac{2(k_{1}+r)}{3}+r\left(4-\frac{1}{q}\right)\left(\frac{3q-1}{3q}\right)\right)\right\}. \end{split}$$

Remark 2.24. Note that the nonlinear inequalities (2 - 24) and (2 - 27) is feasible with respect to k_1 , k_{1n} , k_3 , k_{3n} for any $M_1 \ge 0$, $M_2 \ge 0$, $M_4 \ge 0$, and $M_5 \ge 0$. As such an r > 0 always exists for sufficiently large k_1 , k_{1n} , k_3 , k_{3n} . In particular, consider the case where $M_4 = M_5 = 0$, inequalities (2 - 27) can be easily satisfied for any $k_3 > M_1$ and $k_{3n} > M_2$. This coincides with the conditions obtained through the weak Lyapunov function, i.e. the energy function

$$E(z_1, z_2) = \frac{1}{2} z_2^2 + \frac{k_1}{2p} |z_1| + \frac{k_{1n}}{2q} |z_1|^2 \text{ (similar to that observed when } p > 0\text{)}.$$

Remark 2.25. Hence, when $M_3 = M_6 = 0$, it is not difficult to show that the system achieved finite time convergence for $q \ge 1$. In particular for the case of q > 1, following similar arguments in section 2.2.4(a), finite convergence time independent of initial conditions for q > 1,

$$\begin{split} T_{total} = & \left(\frac{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}{\omega_8}\right) \left(\frac{4q}{q-1}\right) \left(\frac{\omega_8}{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}\right)^{\frac{0.5(q-1)}{q-0.5}} \left(\frac{\overline{\pi}_1 \frac{3}{4}}{\omega_7}\right)^{\frac{0.5(q-1)}{q-0.5}} \\ & + \left(\frac{4\overline{\pi}_1 \frac{3}{4}}{\omega_7}\right) \left(\frac{3^{\frac{q-1}{4q}} \overline{\pi}_1 \frac{5q-1}{4q}}{\omega_8}\right)^{\frac{0.5q}{q-0.5}} \left(\frac{\omega_7}{\overline{\pi}_1 \frac{3}{4}}\right)^{\frac{0.5q}{q-0.5}} \end{split}$$

and when $M_3 \neq 0$ or $M_6 \neq 0$, from (2 - 28),

$$\begin{split} \dot{\vec{V}} &\leq -\frac{1}{2} \left(\frac{\omega_{7}}{\overline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{\omega_{8}}{\frac{q^{-1}}{3^{\frac{4}{q}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}} \right) V^{\frac{5q-1}{4q}} - V^{\frac{1}{2}} \left(\frac{1}{2} \left(\frac{\omega_{7}}{\overline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{1}{4}} - M_{3} \left(\frac{2^{\frac{1}{2}} \omega_{9}}{\underline{\pi}_{1}^{\frac{1}{2}}} \right) \right) \\ &- V^{\frac{4q-1}{4q}} \left(\frac{1}{3} \left(\frac{\omega_{8}}{\frac{q^{-1}}{3^{\frac{4}{q}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}} \right) V^{\frac{1}{4}} - M_{3} \left(\frac{2^{\frac{1}{4}} \omega_{9}}{\underline{\pi}_{1}^{\frac{4q-1}{4q}}} \right) \right) \\ &- V^{\frac{3}{4}} \left(\frac{1}{3} \left(\frac{\omega_{8}}{\frac{q^{-1}}{3^{\frac{4}{4}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}} \right) V^{\frac{2q-1}{4q}} - M_{6} \left(\frac{3^{\frac{1}{4}} \omega_{10}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right) \\ &\leq -\frac{1}{2} \left(\frac{\omega_{7}}{\overline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{\omega_{8}}{\frac{q^{-1}}{3^{\frac{4}{4}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}} \right) V^{\frac{5q-1}{4q}} - M_{6} \left(\frac{3^{\frac{1}{4}} \omega_{10}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right) \\ &\leq -\frac{1}{2} \left(\frac{\omega_{7}}{\overline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{\omega_{8}}{\frac{q^{-1}}{3^{\frac{4}{4}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}} \right) V^{\frac{5q-1}{4q}} \right) \\ &\leq -\frac{1}{2} \left(\frac{\omega_{7}}{\overline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{2^{\frac{1}{2}} \omega_{9}}{\frac{q^{-1}}{\pi_{1}^{\frac{5}{4}} \overline{\pi}_{1}^{\frac{1}{4}} \overline{\pi}_{1}}} \right) V^{\frac{5q-1}{4q}} \right) \\ &\leq -\frac{1}{2} \left(\frac{\omega_{7}}{\overline{\pi}_{1}^{\frac{3}{4}}}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{2^{\frac{1}{2}} \omega_{9}}{\frac{q^{-1}}{\pi_{1}^{\frac{5}{4}} \overline{\pi}_{1}^{\frac{5q-1}{4q}}}} \right) V^{\frac{5q-1}{4q}} \right) \\ &\leq -\frac{1}{2} \left(\frac{\omega_{7}}{\overline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{2^{\frac{1}{2}} \omega_{9}}{\frac{q^{-1}}{\pi_{1}^{\frac{5}{4}} \overline{\pi}_{1}^{\frac{5q-1}{4}}}} \right) V^{\frac{5q-1}{4q}} \right) \\ &\leq -\frac{1}{2} \left(\frac{2M_{3}}{3} \left(\frac{\frac{q^{-1}{3}}{\overline{\pi}_{1}^{\frac{5q-1}{4}}} \right) \left(\frac{2^{\frac{1}{2}} \omega_{9}}{\frac{q^{-1}}{\pi_{1}^{\frac{5}{4}} \overline{\pi}_{1}^{\frac{5q-1}{4}}}} \right) \right)^{\frac{1}{4}} \right) \\ &= \left(\frac{3M_{3}}{3} \left(\frac{\frac{q^{-1}}{4q} \overline{\pi}_{1}^{\frac{5q-1}{4q}}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{9}}{\overline{\pi}_{1}^{\frac{4q-1}{4}} \overline{\pi}_{1}^{\frac{4q-1}{4}}} \right) \right)^{\frac{1}{4}} \right) \\ &= \left(\frac{3M_{3}}{3} \left(\frac{2^{\frac{1}{4}} \overline{\pi}_{1}^{\frac{5q-1}{4}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{9}} \overline{\pi}_{1}^{\frac{5q-1}{4}} \right) \right)^{\frac{1}{4}} \right) \\ \\ &= \left(\frac{3M_{3}}{3} \left(\frac{2^{\frac{1}{4}} \overline{\pi}_{1}^{\frac{5q-1}{4}} \overline{\pi}_{1}^{\frac{5q-1}{4}}} \right) \left(\frac{2^{\frac{1}{4}}$$

thus, uniform ultimate boundedness is implied from lemma 2.6.

Remark 2.26. Note that even in the case of $M_3 = M_6 = 0$, the system is able to be exactly robust with respect to persistent perturbations in d_2 with an upper bound of M_4 . This interesting feature is possible due to the discontinuous nature of the algorithm when p = 0.5.

Remark 2.27. It is worth mentioning that the system presented here is able to achieve the same type of convergence and robustness properties as the one in [17], in which a second order

system also based on the super-twisting algorithm is studied. Of significance is in [17], an additional term in the z_2 dynamics that has an exponent value p + q + 1, which satisfy $2p-1 \le p+q+1 \le 2q-1$, for $0.5 \le p \le 1$, and $q \ge 1$. The function of this additional term is not clearly stated there, besides that of a notational simplification in the stability analysis. Indeed if that is the case, then it is clearly not needed in our development, lemma 2.3 is available in our development. In addition, there, it requires two different Lyapunov function structure in order to extract different stability properties of the system. In particular, for ascertaining the finite time property, their strict Lyapunov function exhibit singularity in its time derivative, due to the non Lipschitzness of the said Lyapunov function. This is not desirable, as mentioned in [141], if such a system is to be applied for further applications. On the other hand, the results presented here do not have such a problem as the proposed strict Lyapunov function is locally Lipschitz.

2.6 Generic second order algorithm

Due to the similarities of their mechanical energy function in forming the respective strict Lyapunov function for twisting and super-twisting based algorithms, in this section, a generic second order algorithm, which consists of linear sum of the generic twisting algorithm (2 - 11) and generic super-twisting algorithm (2 - 23), is presented. Leveraging the results of previous sections, a strict and locally-Lipschitz is presented for the algorithm to study its stability and robustness properties.

2.6.1 System description

Consider the following generic second order algorithm dynamics:

$$\dot{z}_{1} = -k_{3}|z_{1}|^{p}\operatorname{sign}(z_{1}) - k_{3n}|z_{1}|^{q}\operatorname{sign}(z_{1}) + z_{2} + d_{1},$$

$$\dot{z}_{2} = -k_{1}|z_{1}|^{2p-1}\operatorname{sign}(z_{1}) - k_{1n}|z_{1}|^{2q-1}\operatorname{sign}(z_{1}) - k_{2}|z_{2}|^{\frac{2p-1}{p}}\operatorname{sign}(z_{2}) - k_{2n}|z_{2}|^{\frac{2q-1}{q}}\operatorname{sign}(z_{2}) + d_{2}$$
(2 - 29)

where $z_1, z_2 \in \mathbb{R}$ are the scalar state variables, $k_1, k_{1n}, k_2, k_{2n}, k_3, k_{3n}$ are positive constants, 0.5 $\leq p \leq 1$, and $1 \leq q$. While d_1 and d_2 are time-varying and/or nonlinear terms of uncertainty bounded by:

$$|d_1| \le M_1 |z_1|^p + M_2 |z_1|^q + M_3$$
, and

$$|d_2| \le M_4 |z_1|^{2p-1} + M_5 |z_1|^{2q-1} + M_6 |z_2|^{2-\frac{1}{p}} + M_7 |z_2|^{2-\frac{1}{q}} + M_8,$$

with
$$|d_1| := \sup\{|\delta| : \delta \in \mathbf{K}[d_1]\}$$
, and $|d_2| := \sup\{|\delta| : \delta \in \mathbf{K}[d_2]\}$,

where $M_1 \ge 0$, $M_2 \ge 0$, $M_3 \ge 0$, $M_4 \ge 0$, $M_5 \ge 0$, $M_6 \ge 0$, $M_7 \ge 0$, and $M_8 \ge 0$ with the same p and q as that in (2 - 29). Note that for the case of p = 0.5, the algorithm contains discontinuity and the uncertainty is upper bounded by nonvanishing constant.

2.6.2 Stability analysis

The following Lyapunov function is proposed for the system (2 - 29),

$$V(z_1, z_2) = \left(\frac{1}{2}z_2^2 + \frac{k_1}{2p}|z_1|^{2p} + \frac{k_{1n}}{2q}|z_1|^{2q}\right)^2$$

Remark 2.28. The system (2 - 29) can be viewed as the combination of the generic twisting algorithm and the generic super-twisting algorithm considered in section 2.2.2 and 2.2.4 respectively. Observed that the structure of Lyapunov functions proposed in the previous sections are motivated by the mechanical energy of the system [66], which by itself is a weak Lyapunov function, i.e. only negative semidefinite can be attained in its time derivative taken along the solutions of the system. To ensure negative definiteness, a cross term, that contains both state variables, is added accordingly. In particular, for the case of twisting based algorithm (Section 2.2.1 and 2.2.2), the cross term can only have a z_2 with unity exponent, in order to obtain a negative definite z_1 term in the time derivative of the Lyapunov function. While the exponent of the z_1 part of the cross term, it is selected as to maintain the homogeneity of the energy function. However, in using the original energy function, the resulting cross term is non-Lipschitz in order to maintain homogeneity of the function. Hence, to overcome it, the energy function is simply squared, consequently a locally Lipschitz and strict Lyapunov function is produced. The same arguments applied to super-twisting based algorithm as well. Of interest, is that both twisting and super-twisting based algorithm embodies the same mechanical energy term, moreover in this section, the algorithm (2 - 29) has the same energy function as well. As will be shown in the following development, the energy function is complete by itself without the need for cross term, to generate a negative definite time derivative.

Note that the Lyapunov function, V can be lower and upper bounded by

$$\underline{\pi}_{1}^{2} \left(\left| z_{1} \right|^{2p} + \left| z_{1} \right|^{2q} + \left| z_{2} \right|^{2} \right)^{2} \leq V(z_{1}, z_{2}) \leq \overline{\pi}_{1}^{2} \left(\left| z_{1} \right|^{2p} + \left| z_{1} \right|^{2q} + \left| z_{2} \right|^{2} \right)^{2}$$

where $\overline{\pi}_{1} = \max\left\{ \frac{1}{2}, \frac{k_{1}}{2p}, \frac{k_{1n}}{2q} \right\}$, and $\underline{\pi}_{1} = \min\left\{ \frac{1}{2}, \frac{k_{1}}{2p}, \frac{k_{1n}}{2q} \right\}$

Thus, *V* is positive definite and radially unbounded. Since (2 - 29) is a differential equation that has discontinuous right-hand side, i.e. when p = 0.5 and since no continuity assumption is made on both d_1 and d_2 , its solution are understood in the sense of Filippov (see definition 2.1). According to lemma 2.4, the time derivative of non-smooth Lyapunov function exists almost everywhere,

$$\frac{d}{dt}V(\mathbf{z}(t),t) \stackrel{a.e.}{\in} \dot{\widetilde{V}}(\mathbf{z}(t),t) = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^T K\begin{bmatrix} \dot{z}_1\\ \dot{z}_2 \end{bmatrix} (\mathbf{z},t),$$

In accordance to the analysis performed in previous sections, the case of p > 0.5 and p = 0.5 are analyzed separately, due to discontinuity of the controller (2 - 29) when p = 0.5.

a) Case 1: For $0.5 , and <math>1 \le q$

Note that for p > 0.5, V is continuously differentiable, hence

$$\dot{\widetilde{V}}(z_1, z_2) = \bigcap_{\xi \in \partial V(\mathbf{z}(t), t)} \xi^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix} = \nabla V^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix}$$

$$\begin{split} & \subset -k_1 k_3 |z_1|^{3p-1} |z_2|^2 - k_{1n} k_{3n} |z_1|^{3q-1} |z_2|^2 - k_1 k_{3n} |z_1|^{2p+q-1} |z_2|^2 \\ & -k_{1n} k_3 |z_1|^{2q+p-1} |z_2|^2 - \frac{k_1}{p} k_2 |z_1|^{2p} |z_2|^{\frac{3p-1}{p}} - \frac{k_1}{p} k_{2n} |z_1|^{2p} |z_2|^{\frac{3q-1}{q}} \\ & -\frac{k_{1n}}{q} k_2 |z_1|^{2q} |z_2|^{\frac{3p-1}{p}} - \frac{k_{1n}}{q} k_{2n} |z_1|^{2q} |z_2|^{\frac{3q-1}{q}} \\ & -k_1 k_{1n} k_3 \left(\frac{1}{p} + \frac{1}{q}\right) |z_1|^{3p+2q-1} - k_1 k_{1n} k_{3n} \left(\frac{1}{p} + \frac{1}{q}\right) |z_1|^{2p+3q-1} \\ & -\frac{k_1}{p} k_1 k_{3n} |z_1|^{4p+q-1} - \frac{k_{1n}}{q} k_{1n} k_3 |z_1|^{4q+p-1} \\ & -\frac{k_1}{p} k_1 k_3 |z_1|^{5p-1} - \frac{k_{1n}}{q} k_{1n} k_{3n} |z_1|^{5q-1} - k_2 |z_2|^{\frac{5p-1}{p}} - k_{2n} |z_2|^{\frac{5q-1}{q}} \\ & + K [d_1 \left(\frac{k_1 |z_1|^{2p-1} |z_2|^2 \operatorname{sign}(z_1) + \frac{k_1^2}{p} |z_1|^{4p-1} \operatorname{sign}(z_1) \right) \\ & + K [d_1 \left(\frac{k_1 k_{1n}}{q} |z_1|^{2p+2q-1} \operatorname{sign}(z_1) + \frac{k_{1n}^2}{q} |z_1|^{4q-1} \operatorname{sign}(z_1) \right) \\ & + K [d_2 \left(z_2^3 + \frac{k_1}{p} |z_1|^{2p} z_2 + \frac{k_{1n}}{q} |z_1|^{2q} z_2 \right) \end{split}$$

After rearrangement:

$$\begin{split} \dot{\vec{V}} &\leq -k_{1}(k_{3}-M_{1})|z_{1}|^{3p-1}|z_{2}|^{2}-k_{1n}(k_{3n}-M_{2})|z_{1}|^{3q-1}|z_{2}|^{2} \\ &-k_{1}(k_{3n}-M_{2})|z_{1}|^{2p+q-1}|z_{2}|^{2}-k_{1n}(k_{3}-M_{1})|z_{1}|^{p+2q-1}|z_{2}|^{2} \\ &-\frac{k_{1}}{p}(k_{2}-M_{6})|z_{1}|^{2p}|z_{2}|^{3-\frac{1}{p}}-\frac{k_{1}}{p}(k_{2n}-M_{7})|z_{1}|^{2p}|z_{2}|^{3-\frac{1}{q}} \\ &-\frac{k_{1n}}{q}(k_{2}-M_{6})|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{p}}-\frac{k_{1n}}{q}(k_{2n}-M_{7})|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{q}} \\ &-k_{1}k_{1n}\left(\frac{1}{p}+\frac{1}{q}\right)(k_{3}-M_{1})|z_{1}|^{2q+3p-1}-k_{1}k_{1n}\left(\frac{1}{p}+\frac{1}{q}\right)(k_{3n}-M_{2})|z_{1}|^{3q+2p-1} \\ &-\frac{k_{1}^{2}}{p}(k_{3n}-M_{2})|z_{1}|^{4p+q-1}-\frac{k_{1n}^{2}}{q}(k_{3}-M_{1})|z_{1}|^{p+4q-1}-\frac{k_{1}^{2}}{2p}(k_{3}-M_{1})|z_{1}|^{5p-1} \\ &-\frac{k_{1n}^{2}}{2q}(k_{3n}-M_{2})|z_{1}|^{4p-1}-\frac{(k_{2}-M_{6})}{2}|z_{2}|^{5-\frac{1}{p}}-\frac{(k_{2n}-M_{7})}{2}|z_{2}|^{5-\frac{1}{q}}+\dot{V}_{1} \\ &+M_{3}\left(\frac{k_{1}^{2}}{p}|z_{1}|^{4p-1}+\frac{k_{1n}^{2}}{q}|z_{1}|^{4q-1}+k_{1}|z_{1}|^{2p-1}|z_{2}|^{2}+k_{1n}|z_{1}|^{2q-1}|z_{2}|^{2} \\ &+M_{8}\left(|z_{2}|^{3}+\frac{k_{1}}{p}|z_{1}|^{2p}|z_{2}|+\frac{k_{1n}}{q}|z_{1}|^{2q}|z_{2}|\right) \end{split}$$

where

$$\begin{split} \dot{V_1} &= -\frac{k_1^{2}}{2p} (k_3 - M_1) |z_1|^{5p-1} - \frac{k_{1n}^{2}}{2q} (k_{3n} - M_2) |z_1|^{5q-1} - \frac{(k_2 - M_6)}{2} |z_2|^{5-\frac{1}{p}} \\ &- \frac{(k_{2n} - M_7)}{2} |z_2|^{5-\frac{1}{q}} + M_4 |z_1|^{2p-1} |z_2|^3 + M_4 \frac{k_1}{p} |z_1|^{4p-1} |z_2| + M_5 |z_1|^{2q-1} |z_2|^3 \\ &+ \frac{k_{1n}}{q} M_5 |z_1|^{4q-1} |z_2| + \left(\frac{k_{1n}}{q} M_4 + \frac{k_1}{p} M_5\right) |z_1|^{2p+2q-1} |z_2| \end{split}$$

Applying lemma 2.1,

$$-|z_1|^{5p-1}-|z_2|^{5-\frac{1}{p}} \le -\left(\frac{5p-1}{2p-1}\right)^{\frac{2p-1}{5p-1}}|z_1|^{2p-1}\left(\frac{5p-1}{3p}\right)^{\frac{3p}{5p-1}}|z_2|^3,$$

$$\begin{split} -|z_{1}|^{5p-1} - |z_{2}|^{5-\frac{1}{p}} &\leq -\left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}} |z_{1}|^{4p-1} \left(\frac{5p-1}{p}\right)^{\frac{p}{5p-1}} |z_{2}|, \\ -|z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{2q-1}\right)^{\frac{2q-1}{5q-1}} |z_{1}|^{2q-1} \left(\frac{5q-1}{3q}\right)^{\frac{3q}{5q-1}} |z_{2}|^{3}, \\ -|z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{q}} &\leq -\left(\frac{5q-1}{4q-1}\right)^{\frac{4q-1}{5q-1}} |z_{1}|^{4q-1} \left(\frac{5q-1}{q}\right)^{\frac{q}{5q-1}} |z_{2}|, \\ -|z_{1}|^{5p-1} - |z_{1}|^{5q-1} - |z_{2}|^{5-\frac{1}{p}} &\leq -|z_{1}|^{(2p+2q-1)} \left(\frac{5p-1}{4p-1}\right) - |z_{2}|^{5-\frac{1}{p}} \\ &\leq -\left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}} |z_{1}|^{(2p+2q-1)} \left(\frac{5p-1}{p}\right)^{\frac{p}{5p-1}} |z_{2}|, \end{split}$$

where lemma 2.3 has been employed in the last inequality since

$$5p-1 \le (2p+2q-1)\left(\frac{5p-1}{4p-1}\right) \le 5q-1.$$

Thus, if the following inequalities

$$\min \begin{cases} \lambda_{1}k_{1}^{\frac{4p-2}{5p-1}}(k_{3}-M_{1})^{\frac{2p-1}{5p-1}}(k_{2}-M_{6})^{\frac{3p}{5p-1}}, \\ \lambda_{2}k_{1}^{\frac{3p-1}{5p-1}}(k_{3}-M_{1})^{\frac{4p-1}{5p-1}}(k_{2}-M_{6})^{\frac{p}{5p-1}}, \\ \lambda_{3}k_{1n}^{\frac{4q-2}{5q-1}}(k_{3n}-M_{2})^{\frac{2q-1}{5q-1}}(k_{2n}-M_{7})^{\frac{3q}{5q-1}}, \\ \lambda_{4}k_{1n}^{\frac{3q-1}{5q-1}}(k_{3n}-M_{2})^{\frac{4q-1}{5q-1}}(k_{2n}-M_{7})^{\frac{q}{5q-1}}, \\ \lambda_{5}\left(\frac{pq}{pk_{1n}+qk_{1}}\right)\omega_{1}^{\frac{4p-1}{5p-1}}(k_{2}-M_{6})^{\frac{p}{5p-1}} \end{cases} > \max \{M_{4}, M_{5}\}$$

$$(2 - 30)$$

where

$$\begin{split} \lambda_{1} &= \left(\frac{5p-1}{12p^{2}-6p}\right)^{\frac{2p-1}{5p-1}} \left(\frac{5p-1}{18p}\right)^{\frac{3p}{5p-1}}, \, \lambda_{2} = p \left(\frac{5p-1}{24p^{2}-6p}\right)^{\frac{4p-1}{5p-1}} \left(\frac{5p-1}{6p}\right)^{\frac{p}{5p-1}}, \\ \lambda_{3} &= \left(\frac{5q-1}{12q^{2}-6q}\right)^{\frac{2q-1}{5q-1}} \left(\frac{5q-1}{12q}\right)^{\frac{3q}{5q-1}}, \, \lambda_{4} = q \left(\frac{5q-1}{24q^{2}-6q}\right)^{\frac{4q-1}{5q-1}} \left(\frac{5q-1}{4q}\right)^{\frac{q}{5q-1}}, \\ \lambda_{5} &= \left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}} \left(\frac{5p-1}{6p}\right)^{\frac{p}{5p-1}}, \, \omega_{1} = \min\left\{\frac{k_{1}^{2}(k_{3}-M_{1})}{6p}, \frac{k_{1n}^{2}(k_{3n}-M_{2})}{6q}\right\} \end{split}$$

hold then the function \dot{V}_1 is negative definite. Then,

$$\begin{split} \tilde{V} &\leq -\frac{k_{1}^{2}}{2p} (k_{3} - M_{1}) |z_{1}|^{5p-1} - \frac{k_{1n}^{2}}{2q} (k_{3n} - M_{2}) |z_{1}|^{5q-1} \\ &- \frac{(k_{2} - M_{6})}{2} |z_{2}|^{5-\frac{1}{p}} - \frac{(k_{2n} - M_{7})}{2} |z_{2}|^{5-\frac{1}{q}} \\ &+ M_{3} \left(\frac{k_{1}^{2}}{p} |z_{1}|^{4p-1} + \frac{k_{1n}^{2}}{q} |z_{1}|^{4q-1} + k_{1} |z_{1}|^{2p-1} |z_{2}|^{2} \\ &+ k_{1n} |z_{1}|^{2q-1} |z_{2}|^{2} + k_{1} k_{1n} \left(\frac{1}{p} + \frac{1}{q} \right) |z_{1}|^{2p+2q-1} \right) \\ &+ M_{8} \left(|z_{2}|^{3} + \frac{k_{1}}{p} |z_{1}|^{2p} |z_{2}| + \frac{k_{1n}}{q} |z_{1}|^{2q} |z_{2}| \right) \end{split}$$

Applying lemma 2.1,

$$\begin{split} \left|z_{1}\right|^{2p-1}\left|z_{2}\right|^{2} &\leq \left(\frac{2p-1}{4p-1}\right)\left|z_{1}\right|^{4p-1} + \left(\frac{2p}{4p-1}\right)\left|z_{2}\right|^{\frac{4p-1}{p}},\\ \left|z_{1}\right|^{2q-1}\left|z_{2}\right|^{2} &\leq \left(\frac{2q-1}{4q-1}\right)\left|z_{1}\right|^{4q-1} + \left(\frac{2q}{4q-1}\right)\left|z_{2}\right|^{\frac{4q-1}{q}},\\ \left|z_{1}\right|^{2q+2p-1} &\leq \frac{1}{2}\left|z_{1}\right|^{4q-1} + \frac{1}{2}\left|z_{1}\right|^{4p-1},\\ \left|z_{1}\right|^{2p}\left|z_{2}\right| &\leq \left(\frac{2}{3}\right)\left|z_{1}\right|^{3p} + \left(\frac{1}{3}\right)\left|z_{2}\right|^{3}, \end{split}$$

$$|z_1|^{2q}|z_2| \le \left(\frac{2}{3}\right)|z_1|^{3q} + \left(\frac{1}{3}\right)|z_2|^3,$$

and lemma 2.3,

$$-|z_1|^{5p-1} - |z_1|^{5q-1} \le -|z_1|^{5q-\frac{q}{p}} , -|z_1|^{5p-1} - |z_1|^{5q-1} \le -|z_1|^{5p-\frac{p}{q}} ,$$

since $5p - 1 \le 5q - \frac{q}{p} \le 5q - 1$, and $5p - 1 \le 5p - \frac{p}{q} \le 5q - 1$,

together with lemma 2.2, and the bounds of the Lyapunov function, we have:

$$\begin{split} \dot{\tilde{V}} &\leq -\omega_{2} \left(\left| z_{1} \right|^{5p-1} + \left| z_{1} \right|^{5q-\frac{q}{p}} + \left| z_{2} \right|^{5-\frac{1}{p}} \right) - \omega_{3} \left(\left| z_{1} \right|^{5q-1} + \omega_{1} \left| z_{1} \right|^{5p-\frac{p}{q}} + \left| z_{2} \right|^{5-\frac{1}{q}} \right) \\ &+ M_{3} \omega_{4} \left(\left| z_{1} \right|^{4p-1} + \left| z_{2} \right|^{\frac{4p-1}{p}} + \left| z_{1} \right|^{4q-1} + \left| z_{2} \right|^{\frac{4q-1}{q}} \right) + M_{8} \omega_{5} \left(\left| z_{1} \right|^{3p} + \left| z_{1} \right|^{3q} + \left| z_{2} \right|^{3} \right) \\ &= -\omega_{2} \left(\left(\left| z_{1} \right|^{2p} \right)^{\frac{5p-1}{2p}} + \left(\left| z_{1} \right|^{2q} \right)^{\frac{5p-1}{2p}} + \left(\left| z_{2} \right|^{2} \right)^{\frac{5p-1}{2p}} \right) \\ &- \omega_{3} \left(\left(\left| z_{1} \right|^{2p} \right)^{\frac{5q-1}{2q}} + \left(\left| z_{2} \right|^{2} \right)^{\frac{5q-1}{2q}} + \left(\left| z_{2} \right|^{2} \right)^{\frac{5q-1}{2q}} \right) \\ &+ M_{3} \omega_{4} \left(\left(\left| z_{1} \right|^{2p} \right)^{\frac{4p-1}{2p}} + \left(\left| z_{2} \right|^{2} \right)^{\frac{4p-1}{2p}} + \left(\left| z_{2} \right|^{2} \right)^{\frac{4q-1}{2q}} + \left(\left| z_{2} \right|^{2} \right)^{\frac{4q-1}{2q}} \right) \\ &+ M_{8} \omega_{5} \left(\left(\left| z_{1} \right|^{2p} \right)^{\frac{3}{2}} + \left(\left| z_{1} \right|^{2q} \right)^{\frac{3}{2}} + \left(\left| z_{2} \right|^{2} \right)^{\frac{3}{2}} \right) \end{split}$$

$$\leq -\left(\frac{\omega_{2}}{3^{\frac{3p-1}{2p}}\bar{\pi}_{1}\frac{5p-1}{2p}}\right)V^{\frac{5p-1}{4p}} - \left(\frac{\omega_{3}}{3^{\frac{3q-1}{2q}}\bar{\pi}_{1}\frac{5q-1}{2q}}\right)V^{\frac{5q-1}{4q}} + M_{3}\left(\frac{\omega_{4}}{\underline{\pi}_{1}\frac{4p-1}{2p}}\right)V^{\frac{4p-1}{4p}} + M_{3}\left(\frac{\omega_{4}}{\underline{\pi}_{1}\frac{4q-1}{2q}}\right)V^{\frac{4q-1}{4p}} + M_{8}\left(\frac{\omega_{5}}{\underline{\pi}_{1}\frac{3}{2}}\right)V^{\frac{3}{4}}$$

$$(2 - 31)$$

where

$$\begin{split} \omega_{2} &= \min\left\{\frac{k_{1}^{2}(k_{3}-M_{1})}{6p}, \, \omega_{1}, \frac{(k_{2}-M_{6})}{2}\right\}, \, \omega_{3} = \min\left\{\frac{k_{1n}^{2}(k_{3n}-M_{2})}{6q}, \, \omega_{1}, \frac{(k_{2n}-M_{7})}{2}\right\}, \\ \omega_{4} &= \max\left\{\frac{\left(\frac{k_{1}^{2}}{p} + k_{1}\left(\frac{2p-1}{4p-1}\right) + \frac{k_{1}k_{1n}}{2}\left(\frac{1}{p} + \frac{1}{q}\right)\right), \\ \left(\frac{k_{1n}^{2}}{q} + k_{1n}\left(\frac{2q-1}{4q-1}\right) + \frac{k_{1}k_{1n}}{2}\left(\frac{1}{p} + \frac{1}{q}\right)\right), \, k_{1}\left(\frac{2p}{4p-1}\right), k_{1n}\left(\frac{2q}{4q-1}\right)\right\}, \\ \omega_{5} &= \max\left\{\frac{2k_{1}}{3p}, \frac{2k_{1n}}{3q}, \left(1 + \frac{k_{1}}{3p} + \frac{k_{1n}}{3q}\right)\right\} \end{split}$$

Remark 2.29. Note that the nonlinear inequalities (2 - 30) is feasible with respect to k_1 , k_{1n} , k_2 , k_{2n} , k_3 , k_{3n} for any $M_1 \ge 0$, $M_2 \ge 0$, $M_4 \ge 0$, $M_5 \ge 0$, $M_6 \ge 0$, and $M_7 \ge 0$. If $M_4 = M_5 = 0$, (2 - 30) is satisfied for any $k_2 > M_6$, $k_{2n} > M_7$, $k_3 > M_1$, $k_{3n} > M_2$ and any $k_1 > 0$, $k_{1n} > 0$. This coincides with the conditions obtained through energy function for the generic twisting and generic super-twisting algorithm studied in the previous sections.

Remark 2.30. Hence, when $M_3 = M_8 = 0$, i.e. without persistent perturbations, from lemma 2.5-3,

1. For 0 , and <math>q > 1 the system exhibit uniform asymptotical convergence, where

$$T_{\max}(\mu) = \left(\frac{3^{\frac{3q-1}{2q}} \bar{\pi}_1 \frac{5q-1}{2q}}{\omega_3}\right) \left(\frac{4q}{q-1}\right) \left(\frac{1}{\mu^{\frac{q-1}{4q}}}\right),$$

is the time at which the trajectories reach the surface level $V = \mu$. At the same time,

the system (2 - 29) will have finite time convergence, in particular after reaching the surface level $V=\mu$, from lemma 2.5-3, the settling time estimate,

$$T(\mu) \leq \left(\frac{3^{\frac{3p-1}{2p}} \overline{\pi}_1^{\frac{5p-1}{2p}}}{\omega_2}\right) \left(\frac{4p}{1-p}\right) [\mu]^{\frac{1-p}{4p}},$$

where the initial starting states is changed to $V = \mu$. Hence, the total time to reach the origin can be estimated as

$$T_{total}(\mu) = \left(\frac{3^{\frac{3q-1}{2q}} \bar{\pi}_1^{\frac{5q-1}{2q}}}{\omega_3}\right) \left(\frac{4q}{q-1}\right) \left(\frac{1}{\mu^{\frac{q-1}{4q}}}\right) + \left(\frac{3^{\frac{3p-1}{2p}} \bar{\pi}_1^{\frac{5p-1}{2p}}}{\omega_2}\right) \left(\frac{4p}{1-p}\right) [\mu]^{\frac{1-p}{4p}}.$$

The minimum of this function can be found at

$$\mu = \left(\frac{3^{\frac{3q-1}{2q}} \overline{\pi}_1 \frac{5q-1}{2q}}{\omega_3}\right)^{\frac{4pq}{q-p}} \left(\frac{\omega_2}{3^{\frac{3p-1}{2p}} \overline{\pi}_1 \frac{5p-1}{2p}}\right)^{\frac{4pq}{q-p}}.$$

Substituting into the function, a finite settling time independent of initial conditions is obtained,

$$\begin{split} T_{total} = & \left(\frac{3^{\frac{3q-1}{2q}} \overline{\pi}_1^{\frac{5q-1}{2q}}}{\omega_3}\right) \left(\frac{4q}{q-1}\right) \left(\frac{\omega_3}{3^{\frac{3q-1}{2q}} \overline{\pi}_1^{\frac{5q-1}{2q}}}\right)^{\frac{p(q-1)}{q-p}} \left(\frac{3^{\frac{3p-1}{2p}} \overline{\pi}_1^{\frac{5p-1}{2p}}}{\omega_2}\right)^{\frac{p(q-1)}{q-p}} \\ & + \left(\frac{3^{\frac{3p-1}{2p}} \overline{\pi}_1^{\frac{5p-1}{2p}}}{\omega_2}\right) \left(\frac{4p}{1-p}\right) \left(\frac{3^{\frac{3q-1}{2q}} \overline{\pi}_1^{\frac{5q-1}{2q}}}{\omega_3}\right)^{\frac{q(1-p)}{q-p}} \left(\frac{\omega_2}{3^{\frac{3p-1}{2p}} \overline{\pi}_1^{\frac{5p-1}{2p}}}\right)^{\frac{q(1-p)}{q-p}} \end{split}$$

2. For 0 , and <math>q = 1, finite time convergence can be concluded.

3. For p = 1, and $q \ge 1$, exponential convergence can be concluded.

The above results are possible due to the negative definiteness of the time derivative of the Lyapunov function, i.e. strict Lyapunov function.

Remark 2.31. While if persistent perturbations occur on the system, $M_3 \neq 0$ and/or $M_8 \neq 0$, from (2 - 31),

$$\begin{split} \dot{\tilde{V}} &\leq -\frac{1}{2} \left(\frac{\omega_2}{3^{\frac{3p-1}{2p}} \overline{\pi}_1^{\frac{5p-1}{2p}}} \right) V^{\frac{5p-1}{4p}} - \frac{1}{3} \left(\frac{\omega_3}{3^{\frac{3q-1}{2q}} \overline{\pi}_1^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} \\ &- V^{\frac{4p-1}{4p}} \left(\frac{1}{2} \left(\frac{\omega_2}{3^{\frac{3p-1}{2p}} \overline{\pi}_1^{\frac{5p-1}{2p}}} \right) V^{\frac{1}{4}} - M_3 \left(\frac{\omega_4}{\overline{\pi}_1^{\frac{4p-1}{2p}}} \right) \right) \\ &- V^{\frac{4q-1}{4q}} \left(\frac{1}{3} \left(\frac{\omega_3}{3^{\frac{3q-1}{2q}} \overline{\pi}_1^{\frac{5q-1}{2q}}} \right) V^{\frac{1}{4}} - M_3 \left(\frac{\omega_4}{\overline{\pi}_1^{\frac{4p-1}{2p}}} \right) \right) \\ &- V^{\frac{3}{4}} \left(\frac{1}{3} \left(\frac{\omega_3}{3^{\frac{3q-1}{2q}} \overline{\pi}_1^{\frac{5q-1}{2q}}} \right) V^{\frac{1}{4}} - M_3 \left(\frac{\omega_5}{\overline{\pi}_1^{\frac{3}{2}}} \right) \right) \end{split}$$

$$\leq -\frac{1}{2} \left(\frac{\omega_2}{3^{\frac{3p-1}{2p}} \overline{\pi}_1^{\frac{5p-1}{2p}}} \right) V^{\frac{5p-1}{4p}} - \frac{1}{3} \left(\frac{\omega_3}{3^{\frac{3q-1}{2q}} \overline{\pi}_1^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}},$$

$$for V \geq \max \left\{ \begin{cases} 2M_3 \left(\frac{3^{\frac{3p-1}{2p}} \overline{\pi}_1^{\frac{5p-1}{2p}}}{\omega_2} \right) \left(\frac{\omega_4}{\underline{\pi}_1^{\frac{4p-1}{2p}}} \right) \right)^4, \left(3M_3 \left(\frac{3^{\frac{3q-1}{2q}} \overline{\pi}_1^{\frac{5q-1}{2q}}}{\omega_3} \right) \left(\frac{\omega_4}{\underline{\pi}_1^{\frac{4q-1}{2p}}} \right) \right)^4, \\ \left(3M_8 \left(\frac{3^{\frac{3q-1}{2q}} \overline{\pi}_1^{\frac{5q-1}{2q}}}{\omega_3} \right) \left(\frac{\omega_5}{\underline{\pi}_1^{\frac{3}{2}}} \right) \right)^{\frac{4q}{2q-1}} \end{cases} \right\}$$

uniform ultimate boundedness is concluded by using lemma 2.6.

b) Case 2: For p = 0.5, and $1 \le q$

For p = 0.5, *V* is not differentiable on $z_1 = 0$, hence

$$\dot{\widetilde{V}}(z_1, z_2) = \bigcap_{\xi \in \partial V(\mathbf{z}(t), t)} \xi^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix} \subset \dot{\widetilde{V}}_1 + \dot{\widetilde{V}}_2$$

where

$$\dot{\vec{V}}_{1} = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^{T} K[f](z_{1}, z_{2}), \quad \dot{\vec{V}}_{2} = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^{T} K\begin{bmatrix}d_{1}\\d_{2}\end{bmatrix},$$

$$K[f](z_{1}, z_{2}) \in \begin{bmatrix} -k_{3}|z_{1}|^{0.5} \operatorname{sign}(z_{1}) - k_{3n}|z_{1}|^{q} \operatorname{sign}(z_{1}) + z_{2}\\ -k_{1} \operatorname{SGN}(z_{1}) - k_{1n}|z_{1}|^{2q-1} \operatorname{sign}(z_{1}) - k_{2} \operatorname{SGN}(z_{2}) - k_{2n}|z_{2}|^{\frac{2q-1}{q}} \operatorname{sign}(z_{2}) \end{bmatrix}$$

where

$$SGN(x) = \begin{cases} -1 & x < 0 \\ [-1, 1] & x = 0 \\ 1 & x > 0 \end{cases}$$
$$\partial V = \mathbf{K} [\nabla V] = \mathbf{K} \begin{bmatrix} \frac{\partial V}{\partial z_1} \\ \frac{\partial V}{\partial z_2} \end{bmatrix} \subset \begin{bmatrix} \mathbf{K} \begin{bmatrix} \frac{\partial V}{\partial z_1} \\ \end{bmatrix} \\ \mathbf{K} \begin{bmatrix} \frac{\partial V}{\partial z_2} \end{bmatrix} \end{bmatrix}$$
$$= \begin{cases} \forall z_1 \neq 0, z_2 \in \mathbb{R} : \\ \left[\left(z_2^2 + 2k_1 |z_1| + \frac{k_{1n}}{q} |z_1|^{2q} \right) (k_1 + k_{1n} |z_1|^{2q-1}) sign(z_1) \\ (z_2^2 + 2k_1 |z_1| + \frac{k_{1n}}{q} |z_1|^{2q}) (z_2) \end{cases}$$
$$\forall z_1 = 0, z_2 \in \mathbb{R} : \\ \left[z_2^2 k_1 [-1,1] \\ z_2^3 \end{bmatrix} \end{cases}$$

Let us define:

$$\begin{aligned} \left| \frac{\partial V}{\partial z_1} \right| &:= \sup \left\{ \left| \xi_1 \right| : \xi_1 \in \mathbf{K} \left[\frac{\partial V}{\partial z_1} \right] \right\}, \text{ and } \left| \frac{\partial V}{\partial z_2} \right| := \sup \left\{ \left| \xi_2 \right| : \xi_2 \in \mathbf{K} \left[\frac{\partial V}{\partial z_2} \right] \right\}, \\ \text{with } \left| \frac{\partial V}{\partial z_1} \right| &\leq k_1 \left| z_2 \right|^2 + 2k_1^2 \left| z_1 \right| + \frac{k_1 k_{1n}}{q} \left| z_1 \right|^{2q} + k_{1n} \left| z_1 \right|^{2q-1} \left| z_2 \right|^2 + 2k_1 k_{1n} \left| z_1 \right|^{2q} + \frac{k_{1n}^2}{q} \left| z_1 \right|^{4q-1}, \\ \text{and } \left| \frac{\partial V}{\partial z_2} \right| &\leq \left| z_2 \right|^3 + 2k_1 \left| z_1 \right| \left| z_2 \right| + \frac{k_{1n}}{q} \left| z_1 \right|^{2q} \left| z_2 \right|. \end{aligned}$$

Thus, the term

$$\begin{split} \dot{\tilde{V}}_{2} &= \bigcap_{(\xi_{1},\xi_{2})^{T} \in \partial V(\mathbf{z}(t),t)} \xi_{1} K[d_{1}] + \xi_{2} K[d_{2}] \\ &\leq \left| \frac{\partial V}{\partial z_{1}} \right| d_{1} | + \left| \frac{\partial V}{\partial z_{2}} \right| d_{2} | \\ &\leq \left(M_{1} |z_{1}|^{0.5} + M_{2} |z_{1}|^{q} + M_{3} \right) \begin{pmatrix} k_{1} |z_{2}|^{2} + 2k_{1}^{2} |z_{1}| + \frac{k_{1}k_{1n}}{q} |z_{1}|^{2q} + k_{1n} |z_{1}|^{2q-1} |z_{2}|^{2} \\ &+ 2k_{1}k_{1n} |z_{1}|^{2q} + \frac{k_{1n}^{2}}{q} |z_{1}|^{4q-1} \\ &+ \left(M + M_{5} |z_{1}|^{2q-1} + M_{7} |z_{2}|^{2-\frac{1}{q}} + M_{8} \right) \left(|z_{2}|^{3} + 2k_{1} |z_{1}| |z_{2}| + \frac{k_{1n}}{q} |z_{1}|^{2q} |z_{2}| \right) \end{split}$$

where *M* be defined as, $M := M_4 + M_6$, since for p = 0.5, $|d_1| \le M_1 |z_1|^{0.5} + M_2 |z_1|^q + M_3$, and

$$|d_2| \le M_4 + M_5 |z_1|^{2q-1} + M_6 + M_7 |z_2|^{2-\frac{1}{q}} + M_8$$

Computing $\dot{\tilde{V}}_1$ for each case, we have

For $z_1 \neq 0$, $z_2 \neq 0$:

$$\begin{split} \dot{\vec{V}_{1}} &= -k_{1}k_{3}|z_{1}|^{0.5}|z_{2}|^{2} - k_{1}k_{3n}|z_{1}|^{q}|z_{2}|^{2} - k_{1n}k_{3}|z_{1}|^{2q-0.5}|z_{2}|^{2} - k_{1n}k_{3n}|z_{1}|^{3q-1}|z_{2}|^{2} \\ &- 2k_{1}k_{2}|z_{1}||z_{2}| - \frac{k_{1n}}{q}k_{2}|z_{1}|^{2q}|z_{2}| - 2k_{1}k_{2n}|z_{1}||z_{2}|^{\frac{3q-1}{q}} - \frac{k_{1n}}{q}k_{2n}|z_{1}|^{2q}|z_{2}|^{\frac{3q-1}{q}} \\ &- 2k_{1}^{2}k_{3n}|z_{1}|^{1+q} - k_{1}k_{1n}k_{3}\left(2 + \frac{1}{q}\right)|z_{1}|^{2q+0.5} - k_{1}k_{1n}k_{3n}\left(2 + \frac{1}{q}\right)|z_{1}|^{3q} - \frac{k_{1n}^{2}k_{3}}{q}|z_{1}|^{4q-0.5} \\ &- 2k_{1}^{2}k_{3}|z_{1}|^{1.5} - \frac{k_{1n}^{2}k_{3n}}{q}|z_{1}|^{5q-1} - k_{2}|z_{2}|^{3} - k_{2n}|z_{2}|^{\frac{5q-1}{q}} \end{split}$$

For $z_1 = 0, z_2 \neq 0$:

Let $(\xi_2 k_1 z_2^2, z_2^3)^T$ with $\xi_2 \in [-1, 1]$ be an arbitrary element of $\partial V(z_1, z_2)$, then

$$\xi^{\mathrm{T}}K[f](z_{1},z_{2}) = \begin{bmatrix} \xi_{2}k_{1}z_{2}^{2} \\ z_{2}^{3} \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} z_{2} \\ -[-1,1]k_{1} - k_{2}\mathrm{sign}(z_{2}) - k_{2n}|z_{2}|^{\frac{2q-1}{q}}\mathrm{sign}(z_{2}) \end{bmatrix}$$
$$= [\xi_{2} - 1, \xi_{2} + 1]k_{1}z_{2}^{3} - k_{2}|z_{2}|^{3} - k_{2n}|z_{2}|^{\frac{5q-1}{q}}$$

hence

$$\dot{\widetilde{V}}_{1} = \bigcap_{\xi_{2} \in [-1, 1]} \left[\xi_{2} - 1, \xi_{2} + 1 \right] k_{1} z_{2}^{3} - k_{2} |z_{2}|^{3} - k_{2n} |z_{2}|^{\frac{5q-1}{q}} = -k_{2} |z_{2}|^{3} - k_{2n} |z_{2}|^{\frac{5q-1}{q}}$$

For $z_1 \neq 0, z_2 = 0$:

$$\dot{\widetilde{V}}_{1} = -2k_{1}^{2}k_{3n}|z_{1}|^{1+q} - k_{1}k_{1n}k_{3}\left(2\frac{1}{q}\right)|z_{1}|^{2q+0.5} - k_{1}k_{1n}k_{3n}\left(2+\frac{1}{q}\right)|z_{1}|^{3q} - \frac{k_{1n}^{2}k_{3}}{q}|z_{1}|^{4q-0.5} - 2k_{1}^{2}k_{3}|z_{1}|^{1.5} - \frac{k_{1n}^{2}k_{3n}}{q}|z_{1}|^{5q-1}$$

For $z_1 = 0$, $z_2 = 0$: $\dot{V}_1 = 0$.

Thus, for all $(z_1, z_2) \in \mathbb{R}^2$:

$$\begin{split} \dot{\vec{V}}_{1} &= -k_{1}k_{3}|z_{1}|^{0.5}|z_{2}|^{2} - k_{1}k_{3n}|z_{1}|^{q}|z_{2}|^{2} - k_{1n}k_{3}|z_{1}|^{2q-0.5}|z_{2}|^{2} - k_{1n}k_{3n}|z_{1}|^{3q-1}|z_{2}|^{2} \\ &- 2k_{1}k_{2}|z_{1}||z_{2}| - \frac{k_{1n}}{q}k_{2}|z_{1}|^{2q}|z_{2}| - 2k_{1}k_{2n}|z_{1}||z_{2}|^{\frac{3q-1}{q}} - \frac{k_{1n}}{q}k_{2n}|z_{1}|^{2q}|z_{2}|^{\frac{3q-1}{q}} \\ &- 2k_{1}^{2}k_{3n}|z_{1}|^{1+q} - k_{1}k_{1n}k_{3}\left(2 + \frac{1}{q}\right)|z_{1}|^{2q+0.5} - k_{1}k_{1n}k_{3n}\left(2 + \frac{1}{q}\right)|z_{1}|^{3q} - \frac{k_{1n}^{2}k_{3}}{q}|z_{1}|^{4q-0.5} \\ &- 2k_{1}^{2}k_{3}|z_{1}|^{1.5} - \frac{k_{1n}^{2}k_{3n}}{q}|z_{1}|^{5q-1} - k_{2}|z_{2}|^{3} - k_{2n}|z_{2}|^{\frac{5q-1}{q}} \end{split}$$

Hence, the generalized time derivative of the Lyapunov function, after rearrangement:

$$\begin{split} \dot{\vec{V}} &= \dot{\vec{V}_{1}} + \dot{\vec{V}_{2}} \\ &\leq -k_{1}(k_{3} - M_{1})|z_{1}|^{0.5}|z_{2}|^{2} - k_{1n}(k_{3n} - M_{2})|z_{1}|^{3q-1}|z_{2}|^{2} - k_{1}(k_{3n} - M_{2})|z_{1}|^{q}|z_{2}|^{2} \\ &- k_{1n}(k_{3} - M_{1})|z_{1}|^{2q-0.5}|z_{2}|^{2} - 2k_{1}(k_{2} - M)|z_{1}||z_{2}| - 2k_{1}(k_{2n} - M_{7})|z_{1}||z_{2}|^{3-\frac{1}{q}} \\ &- \frac{k_{1n}}{q}(k_{2} - M)|z_{1}|^{2q}|z_{2}| - \frac{k_{1n}}{q}(k_{2n} - M_{7})|z_{1}|^{2q}|z_{2}|^{3-\frac{1}{q}} - k_{1}k_{1n}\left(2 + \frac{1}{q}\right)(k_{3} - M_{1})|z_{1}|^{2q+0.5} \\ &- \frac{k_{1}k_{1n}}{2}\left(2 + \frac{1}{q}\right)(k_{3n} - M_{2})|z_{1}|^{3q} - 2k_{1}^{2}(k_{3n} - M_{2})|z_{1}|^{q+1} \\ &- \frac{k_{1n}^{2}}{q}(k_{3} - M_{1})|z_{1}|^{4q-0.5} - 2k_{1}^{2}(k_{3} - M_{1})|z_{1}|^{1.5} - \frac{k_{1n}^{2}}{2q}(k_{3n} - M_{2})|z_{1}|^{5q-1} \\ &- \frac{(k_{2} - M)}{2}|z_{2}|^{3} - \frac{(k_{2n} - M_{7})}{2}|z_{2}|^{5-\frac{1}{q}} + \dot{V}_{2} \\ &+ M_{3}\left(k_{1}|z_{2}|^{2} + 2k_{1}^{2}|z_{1}| + k_{1}k_{1n}\left(\frac{1}{q} + 2\right)|z_{1}|^{2q} + k_{1n}|z_{1}|^{2q-1}|z_{2}|^{2} + \frac{k_{1n}^{2}}{q}|z_{1}|^{4q-1}\right) \\ &+ M_{8}\left(|z_{2}|^{3} + 2k_{1}|z_{1}||z_{2}| + \frac{k_{1n}}{q}|z_{1}|^{2q}|z_{2}|\right) \end{split}$$

where

$$\dot{V}_{2} = -\frac{k_{1}k_{1n}}{2} \left(2 + \frac{1}{q}\right) \left(k_{3n} - M_{2}\right) |z_{1}|^{3q} - \frac{k_{1n}^{2}}{2q} \left(k_{3n} - M_{2}\right) |z_{1}|^{5q-1} - \frac{\left(k_{2n} - M_{7}\right)}{2} |z_{2}|^{5-\frac{1}{q}} - \frac{\left(k_{2} - M\right)}{2} |z_{2}|^{3} + M_{5} |z_{1}|^{2q-1} |z_{2}|^{3} + 2k_{1}M_{5} |z_{1}|^{2q} |z_{2}| + \frac{k_{1n}}{q} M_{5} |z_{1}|^{4q-1} |z_{2}|$$

Applying lemma 2.1,

$$\begin{split} &-\left|z_{1}\right|^{5q-1}-\left|z_{2}\right|^{5-\frac{1}{q}}\leq-\left(\frac{5q-1}{2q-1}\right)^{\frac{2q-1}{5q-1}}\left|z_{1}\right|^{2q-1}\left(\frac{5q-1}{3q}\right)^{\frac{3q}{5q-1}}\left|z_{2}\right|^{3},\\ &-\left|z_{1}\right|^{5q-1}-\left|z_{2}\right|^{5-\frac{1}{q}}\leq-\left(\frac{5q-1}{4q-1}\right)^{\frac{4q-1}{5q-1}}\left|z_{1}\right|^{4q-1}\left(\frac{5q-1}{q}\right)^{\frac{q}{5q-1}}\left|z_{2}\right|, \end{split}$$

$$-|z_1|^{3q} - |z_2|^3 \le -\left(\frac{3}{2}\right)^{\frac{2}{3}} |z_1|^{2q} (3)^{\frac{1}{3}} |z_2|,$$

Thus, if the following inequalities

$$\min \left\{ \begin{array}{l} \lambda_{6}k_{1n}^{\frac{4q-2}{5q-1}}(k_{3n}-M_{2})^{\frac{2q-1}{5q-1}}(k_{2n}-M_{7})^{\frac{3q}{5q-1}}, \\ \lambda_{7}k_{1n}^{\frac{3q-1}{5q-1}}(k_{3n}-M_{2})^{\frac{4q-1}{5q-1}}(k_{2n}-M_{7})^{\frac{q}{5q-1}}, \\ \lambda_{8}\frac{k_{1n}^{\frac{2}{3}}}{k_{1}^{\frac{1}{3}}}(k_{3n}-M_{2})^{\frac{2}{3}}(k_{2}-M)^{\frac{1}{3}} \end{array} \right\} > M_{5} \qquad (2-32)$$

where

$$\begin{split} \lambda_6 = & \left(\frac{5q-1}{8q^2-4q}\right)^{\frac{2q-1}{5q-1}} \left(\frac{5q-1}{12q}\right)^{\frac{3q}{5q-1}}, \ \lambda_7 = q \left(\frac{5q-1}{16q^2-4q}\right)^{\frac{4q-1}{5q-1}} \left(\frac{5q-1}{4q}\right)^{\frac{q}{5q-1}}, \\ \lambda_8 = & \left(\frac{1}{2}\right) \left(\frac{3}{2}\right)^{\frac{1}{3}} \left(\frac{6q+3}{4q}\right)^{\frac{2}{3}}, \end{split}$$

hold then the function \dot{V}_2 is negative definite. Then,

$$\begin{split} \dot{\tilde{V}} &\leq -2k_1^{\ 2} \left(k_3 - M_1\right) |z_1|^{1.5} - \frac{k_1 k_{1n}}{2} \left(2 + \frac{1}{q}\right) \left(k_{3n} - M_2\right) |z_1|^{3q} - \frac{\left(k_2 - M\right)}{2} |z_2|^3 \\ &\quad - \frac{k_{1n}^{\ 2}}{2q} \left(k_{3n} - M_2\right) |z_1|^{5q-1} - \frac{\left(k_{2n} - M_7\right)}{2} |z_2|^{5-\frac{1}{q}} \\ &\quad + M_3 \left(k_1 |z_2|^2 + 2k_1^{\ 2} |z_1| + k_1 k_{1n} \left(\frac{1}{q} + 2\right) |z_1|^{2q} + k_{1n} |z_1|^{2q-1} |z_2|^2 + \frac{k_{1n}^{\ 2}}{q} |z_1|^{4q-1}\right) \\ &\quad + M_8 \left(|z_2|^3 + 2k_1 |z_1| |z_2| + \frac{k_{1n}}{q} |z_1|^{2q} |z_2|\right) \end{split}$$

Applying lemma 2.1,

$$\begin{split} |z_{1}|^{2q-1} |z_{2}|^{2} &\leq \left(\frac{2q-1}{4q-1}\right) |z_{1}|^{4q-1} + \left(\frac{2q}{4q-1}\right) |z_{2}|^{\frac{4q-1}{q}}, \\ |z_{1}|^{2q} &= |z_{1}|^{0.5} |z_{1}|^{2q-0.5} \leq \frac{1}{2} |z_{1}| + \frac{1}{2} |z_{1}|^{4q-1}, \\ |z_{1}| |z_{2}| &\leq \left(\frac{2}{3}\right) |z_{1}|^{\frac{3}{2}} + \left(\frac{1}{3}\right) |z_{2}|^{3}, \\ |z_{1}|^{2q} |z_{2}| &\leq \left(\frac{2}{3}\right) |z_{1}|^{3q} + \left(\frac{1}{3}\right) |z_{2}|^{3}, \end{split}$$

and lemma 2.3,

$$-|z_1|^{1.5} - |z_1|^{5q-1} \le -|z_1|^{\frac{5q-1}{2q}} \text{ since } 1.5 \le \frac{5q-1}{2q} \le 5q-1,$$

together with lemma 2.2, and the bounds of the Lyapunov function, we have:

$$\begin{split} \dot{\tilde{V}} &\leq -\omega_{6} \left(\left| z_{1} \right|^{1.5} + \left| z_{1} \right|^{3q} + \left| z_{2} \right|^{3} \right) - \omega_{7} \left(\left| z_{1} \right|^{5q-1} + \left| z_{1} \right|^{\frac{5q-1}{2q}} + \left| z_{2} \right|^{5-\frac{1}{q}} \right) \\ &+ M_{3} \omega_{8} \left(\left| z_{1} \right| + \left| z_{2} \right|^{2} + \left| z_{1} \right|^{4q-1} + \left| z_{2} \right|^{\frac{4q-1}{q}} \right) + M_{8} \omega_{9} \left(\left| z_{1} \right|^{\frac{3}{2}} + \left| z_{1} \right|^{3q} + \left| z_{2} \right|^{3} \right) \\ &= -\omega_{6} \left(\left| z_{1} \right|^{1.5} + \left(\left| z_{1} \right|^{2q} \right)^{1.5} + \left(\left| z_{2} \right|^{2} \right)^{1.5} \right) - \omega_{7} \left(\left(\left| z_{1} \right|^{2q} \right)^{\frac{5q-1}{2q}} + \left| z_{1} \right|^{\frac{5q-1}{2q}} + \left(\left| z_{2} \right|^{2} \right)^{\frac{5q-1}{2q}} \right) \\ &+ M_{3} \omega_{8} \left(\left| z_{1} \right| + \left| z_{2} \right|^{2} + \left(\left| z_{1} \right|^{2q} \right)^{\frac{4q-1}{2q}} + \left(\left| z_{2} \right|^{2} \right)^{\frac{4q-1}{2q}} \right) + M_{8} \omega_{9} \left(\left| z_{1} \right|^{\frac{3}{2}} + \left(\left| z_{1} \right|^{2q} \right)^{\frac{3}{2}} + \left(\left| z_{2} \right|^{2} \right)^{\frac{3}{2}} \right) \end{split}$$

$$\leq -\left(\frac{\omega_{6}}{3^{0.5}\bar{\pi}_{1}^{1.5}}\right)V^{\frac{3}{4}} - \left(\frac{\omega_{7}}{3^{\frac{3q-1}{2q}}\bar{\pi}_{1}^{\frac{5q-1}{2q}}}\right)V^{\frac{5q-1}{4q}} + M_{3}\left(\frac{\omega_{8}}{\underline{\pi}_{1}}\right)V^{\frac{1}{2}} + M_{3}\left(\frac{\omega_{8}}{\underline{\pi}_{1}^{\frac{4q-1}{2q}}}\right)V^{\frac{4q-1}{4q}} + M_{8}\left(\frac{\omega_{9}}{\underline{\pi}_{1}^{\frac{3}{2}}}\right)V^{\frac{3}{4}}$$

$$(2 - 33)$$
where

$$\begin{split} \omega_{6} &= \min\left\{k_{1}^{2}(k_{3}-M_{1}), \frac{k_{1}k_{1n}(k_{3n}-M_{2})}{2}\left(2+\frac{1}{q}\right), \frac{(k_{2}-M)}{2}\right\},\\ \omega_{7} &= \min\left\{\frac{k_{1n}^{2}}{4q}(k_{3n}-M_{2}), \min\left\{k_{1}^{2}(k_{3}-M_{1}), \frac{k_{1n}^{2}}{4q}(k_{3n}-M_{2})\right\}, \frac{(k_{2n}-M_{7})}{2}\right\},\\ \omega_{8} &= \max\left\{\left(2k_{1}^{2}+\frac{k_{1}k_{1n}}{2}\left(\frac{1}{q}+2\right)\right), k_{1},\\ \left(\frac{k_{1}k_{1n}}{2}\left(\frac{1}{q}+2\right)+k_{1n}\left(\frac{2q-1}{4q-1}\right)+\frac{k_{1n}^{2}}{q}\right), k_{1n}\left(\frac{2q}{4q-1}\right)\right\},\\ \omega_{9} &= \max\left\{\left(\frac{4k_{1}}{3}\right), \frac{2k_{1n}}{3q}, \left(1+\frac{2k_{1}}{3}+\frac{k_{1n}}{3q}\right)\right\}\end{split}$$

Remark 2.32. Note that the nonlinear inequalities (2 - 32) is feasible with respect to k_1 , k_{1n} , k_2 , k_{2n} , and k_{3n} for any $M \ge 0$, $M_2 \ge 0$, $M_5 \ge 0$ and $M_7 \ge 0$. In particular, consider the case where $M_5 = 0$, inequalities (2 - 32) can be easily satisfied for any $k_2 > M$, $k_{2n} > M_7$, $k_{3n} > M_2$ and any $k_1 > 0$, $k_{1n} > 0$.

Remark 2.33. Hence, when $M_3 = M_8 = 0$, together with $k_3 > M_1$, it is not difficult to show that the system achieved finite time convergence for $q \ge 1$. In particular for the case of q > 1, following similar arguments in section 2.6.1, finite convergence time independent of initial conditions can be found to be,

$$T_{total} = \left(\frac{3^{\frac{3q-1}{2q}} \bar{\pi}_1 \frac{5q-1}{2q}}{\omega_7}\right) \left(\frac{4q}{q-1}\right) \left(\frac{\omega_7}{3^{\frac{3q-1}{2q}} \bar{\pi}_1 \frac{5q-1}{2q}}\right)^{\frac{0.5(q-1)}{q-0.5}} \left(\frac{3^{0.5} \bar{\pi}_1 \frac{1.5}{\omega_6}}{\omega_6}\right)^{\frac{0.5(q-1)}{q-0.5}} + \left(\frac{3^{0.5} \bar{\pi}_1 \frac{1.5}{\omega_6}}{\omega_6}\right) \left(4\right) \left(\frac{3^{\frac{2q-1}{2q}} \bar{\pi}_1 \frac{5q-1}{2q}}{\omega_7}\right)^{\frac{0.5q}{q-0.5}} \left(\frac{\omega_6}{3^{0.5} \bar{\pi}_1 \frac{1.5}{1.5}}\right)^{\frac{0.5q}{q-0.5}}$$

and when $M_3 \neq 0$ and/or $M_8 \neq 0$, from (2 - 33),

$$\begin{split} \hat{\vec{V}} &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{0.5} \,\overline{\pi}_{1}^{1.5}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{\omega_{7}}{3^{\frac{3q-1}{2q}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} - V^{\frac{1}{2}} \left(\frac{1}{2} \left(\frac{\omega_{6}}{3^{0.5} \,\overline{\pi}_{1}^{1.5}} \right) V^{\frac{1}{4}} - M_{3} \left(\frac{\omega_{8}}{\underline{\pi}_{1}} \right) \right) \\ &- V^{\frac{4q-1}{4q}} \left(\frac{1}{3} \left(\frac{\omega_{7}}{3^{\frac{3q-1}{2q}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{1}{4}} - M_{3} \left(\frac{\omega_{8}}{\underline{\pi}_{1}^{\frac{4q-1}{2q}}} \right) \right) - V^{\frac{3}{4}} \left(\frac{1}{3} \left(\frac{\omega_{7}}{3^{\frac{3q-1}{2q}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{2q-1}{4q}} - M_{8} \left(\frac{\omega_{9}}{\underline{\pi}_{1}^{\frac{3}{2}}} \right) \right) \right) \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{0.5} \,\overline{\pi}_{1}^{1.5}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{\omega_{7}}{3^{\frac{3q-1}{2q}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{0.5} \,\overline{\pi}_{1}^{1.5}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{\omega_{7}}{3^{\frac{3q-1}{2q}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{0.5} \,\overline{\pi}_{1}^{1.5}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{\omega_{7}}{3^{\frac{3q-1}{2q}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{0.5} \,\overline{\pi}_{1}^{1.5}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{\omega_{7}}{3^{\frac{3q-1}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{0.5} \,\overline{\pi}_{1}^{1.5}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{\omega_{7}}{3^{\frac{3q-1}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{0.5} \,\overline{\pi}_{1}^{1.5}} \right) V^{\frac{3}{4}} - \frac{1}{3} \left(\frac{\omega_{7}}{3^{\frac{3q-1}{2q}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{\frac{3q-1}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{\frac{3q-1}{2q} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{\frac{3q-1}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{\frac{3q-1}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{\frac{5q-1}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{\frac{5q-1}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \right) V^{\frac{5q-1}{4q}} , \\ &\leq -\frac{1}{2} \left(\frac{\omega_{6}}{3^{\frac{5q-1}} \,\overline{\pi}_{1}^{\frac{5q-1}{2q}}} \,\overline{\pi}_{1}^{\frac{5q-1$$

uniform ultimate boundedness is implied from lemma 2.6.

Remark 2.34. Note that even in the case of $M_3 = M_8 = 0$, the system is able to be exactly

robust with respect to persistent perturbations with an upper bound of *M*. This interesting feature is possible due to the discontinuous nature of the algorithm when p = 0.5.

2.7 Numerical Simulations

In this section, numerical simulations pertaining to the algorithms discussed in the previous sections are presented. The simulation setups for each algorithm are described. Discussion and analysis of the results are presented accordingly.

2.7.1 Simulation Setup

1) Twisting based algorithm: Recall from (2 - 6), the dynamics of the algorithm are,

$$\dot{z}_1 = z_2,$$

$$\dot{z}_2 = -k_1 |z_1|^b \operatorname{sign}(z_1) - k_2 |z_2|^{\frac{2b}{1+b}} \operatorname{sign}(z_2) + d,$$
 with $|d| \le M_1 |z_1|^b + M_2 |z_2|^{\frac{2b}{1+b}} + M_3.$

The parameter values of the dynamics are $k_1 = 1$, $k_2 = 2$, and $d = |z_2|^{\frac{2b}{1+b}} \operatorname{sign}(z_1) \operatorname{sin}(t)$. The simulations are performed for three different values of $b = \{0.6, 0.7, 1.2\}$. The initial conditions are $z_1(0) = -2$ and $z_2(0) = -2$. Note that the disturbance satisfy the upper bound of the system description of section 2.2 with $M_1 = M_3 = 0$, and $M_2 = 1$. The type of disturbance chosen here is typical in the literature (see [18], [37], [38], [46]). The specific

nature of the disturbance is selected to show that past results on this algorithm are conservative. Similarly, the values of b are to demonstrate that the present results are less conservative.

Then, the another set of parameter values, $b = 0, k_1 = 4, k_2 = 2$, and d = 1, is performed to show the exact robustness property of the algorithm under persistent disturbance.

2) Super-twisting based algorithm: Recall from (2 - 17), the dynamics of the algorithm are,

$$\dot{z}_1 = -k_3 |z_1|^p \operatorname{sign}(z_1) + z_2 + d_1,$$

 $\dot{z}_2 = -k_1 |z_1|^{2p-1} \operatorname{sign}(z_1) + d_2$ with $|d_1| \le M_1 |z_1|^p + M_2$, and $|d_2| \le M_3 |z_1|^{2p-1} + M_4$.

The parameter values of the dynamics are $k_3 = 2$, $k_1 = 0.3$, and $d_1 = |z_1|^p \operatorname{sign}(z_2).\operatorname{sin}(t)$. The simulations are performed for three different values of $p = \{0.5, 0.7, 1.2\}$. The initial conditions are $z_1(0) = -2$ and $z_2(0) = -2$. Note that the disturbances satisfy the upper bound of the system description of section 2.4 with $M_2 = M_3 = M_4 = 0$, and $M_1 = 1$. The type of disturbance chosen here is typical in the literature (see [26], [27], [68]). The disturbance and values of p are selected as such to show the conservativeness of previous results.

Another simulation is performed with the parameter values of $k_3 = 1$, $k_1 = 1$, and d = 0 with p = 0.5 for the unperturbed case.

3) Generic super-twisting based algorithm: Recall from (2 - 23), the dynamics of the algorithm are,

$$\dot{z}_{1} = -k_{3}|z_{1}|^{p}\operatorname{sign}(z_{1}) - k_{3n}|z_{1}|^{q}\operatorname{sign}(z_{1}) + z_{2} + d_{1}, \quad \text{with} \quad \begin{vmatrix} d_{1} \\ \leq M_{1}|z_{1}|^{p} + M_{2}|z_{1}|^{q} + M_{3} \\ d_{2}| \leq M_{4}|z_{1}|^{2p-1} \operatorname{sign}(z_{1}) - k_{1n}|z_{1}|^{2q-1}\operatorname{sign}(z_{1}) + d_{2} \end{vmatrix} \quad \text{with} \quad \begin{vmatrix} d_{1} \\ \leq M_{1}|z_{1}|^{p} + M_{2}|z_{1}|^{q} + M_{3} \\ d_{2}| \leq M_{4}|z_{1}|^{2p-1} + M_{5}|z_{1}|^{2q-1} + M_{6} \end{vmatrix}.$$

The parameter values of the dynamics are $k_1 = k_{1n} = 4$, $k_3 = k_{3n} = 2$, $p = \frac{1}{1.4}$, and $q = \frac{1}{0.4}$ under perturbations $d_1 = \operatorname{sign}(z_2) \operatorname{sin}(t) (|z_1|^p + |z_1|^q)$ and $d_2 = \operatorname{sign}(z_2) \operatorname{sin}(t) (|z_1|^{2p-1} + |z_1|^{2q-1})$. The simulations are performed with three different values of initial conditions

 $(z_1(0), z_2(0)) = \{(-0.1, -0.1), (-1, -1), (-10, -10)\}$. Note that the disturbances satisfy the upper bound of the system description of section 2.5 with $M_1 = M_2 = 1, M_3 = 0, M_4 = M_5 = 1$, and $M_6 = 0$. The type of disturbances chosen are typical in the literature (see [46], [17]). The three different initial conditions, each being an order of magnitude greater than the previous one, are selected as such to show the uniform finite-time capabilities of the algorithm.

4) Generic twisting based algorithm: Recall from (2 - 11), the dynamics of the algorithm are,

$$\dot{z}_{1} = z_{2},$$

$$\dot{z}_{2} = -k_{1}|z_{1}|^{2p-1}\operatorname{sign}(z_{1}) - k_{1n}|z_{1}|^{2q-1}\operatorname{sign}(z_{1}) - k_{2}|z_{2}|^{2-\frac{1}{p}}\operatorname{sign}(z_{2}) - k_{2n}|z_{2}|^{2-\frac{1}{q}}\operatorname{sign}(z_{2}) + d$$
with $|d| \le M_{1}|z_{1}|^{2p-1} + M_{2}|z_{1}|^{2q-1} + M_{3}|z_{2}|^{2-\frac{1}{p}} + M_{4}|z_{2}|^{2-\frac{1}{q}} + M_{5}.$

The parameter values of the dynamics are $k_1 = k_{1n} = 4$, $k_2 = k_{2n} = 2$, $p = \frac{1}{1.4}$, and $q = \frac{1}{0.4}$

under perturbations $d = \operatorname{sign}(z_2) \cdot \operatorname{sin}(t) (|z_1|^{2p-1} + |z_1|^{2q-1}) + \operatorname{sign}(z_1) \cdot \operatorname{sin}(t) (|z_2|^{2-\frac{1}{p}} + |z_2|^{2-\frac{1}{q}}).$ The simulation is performed with the initial conditions $(z_1(0), z_2(0)) = (-3, -3)$. Note that the disturbances satisfy the upper bound of the system description of section 2.3 with $M_1 = M_2 = 1, M_3 = M_4 = 1$, and $M_5 = 0$. The type of disturbances chosen is based on the sum of disturbances considered in twisting-based simulations that are of different exponent.

5) Generic second-order based algorithm: Recall from (2 - 29), the dynamics of the algorithm are,

$$\dot{z}_{1} = -k_{3}|z_{1}|^{p}\operatorname{sign}(z_{1}) - k_{3n}|z_{1}|^{q}\operatorname{sign}(z_{1}) + z_{2} + d_{1},$$

$$\dot{z}_{2} = -k_{1}|z_{1}|^{2p-1}\operatorname{sign}(z_{1}) - k_{1n}|z_{1}|^{2q-1}\operatorname{sign}(z_{1}) - k_{2}|z_{2}|^{\frac{2p-1}{p}}\operatorname{sign}(z_{2}) - k_{2n}|z_{2}|^{\frac{2q-1}{q}}\operatorname{sign}(z_{2}) + d_{2}$$

with

$$|d_1| \le M_1 |z_1|^p + M_2 |z_1|^q + M_3$$
 and

$$|d_2| \le M_4 |z_1|^{2p-1} + M_5 |z_1|^{2q-1} + M_6 |z_2|^{2-\frac{1}{p}} + M_7 |z_2|^{2-\frac{1}{q}} + M_8.$$

The parameter values of the dynamics are

$$k_1 = k_{1n} = k_2 = k_{2n} = 4, k_3 = k_{3n} = 2, p = \frac{1}{1.4}, \text{ and } q = \frac{1}{0.4}$$

under perturbations $d_1 = \operatorname{sign}(z_2) \cdot \operatorname{sin}(t) (|z_1|^p + |z_1|^q)$, and $d_2 = \operatorname{sign}(z_2) \cdot \operatorname{sin}(t) (|z_1|^{2p-1} + |z_1|^{2q-1})$ + $\operatorname{sign}(z_1) \cdot \operatorname{sin}(t) (|z_2|^{2-\frac{1}{p}} + |z_2|^{2-\frac{1}{q}})$. The simulation is performed with the initial conditions of $(z_1(0), z_2(0)) = (-3, -3)$. Note that the disturbances satisfy the upper bound of the system description of section 2.6 with $M_1 = M_2 = 1$, $M_3 = 0$, $M_4 = M_5 = M_6 = M_7 = 1$, and $M_8 = 0$. The type of disturbances chosen is based on the sum of disturbances considered in generic super-twisting and generic twisting-based simulations that are of different exponent.

2.7.2 Results and Discussions

For better visualization of the plots, some figures are shown in two windows; each with different time intervals.



Figure 2.1 Twisting based algorithm. States of (2 - 6) with $k_1 = 1$, $k_2 = 2$, and three different values of $b = \{0.6, 0.7, 1.2\}$.



Figure 2.2 Twisting based algorithm. Disturbance $d = |z_2|^{\frac{2b}{1+b}} \operatorname{sign}(z_1) \cdot \operatorname{sin}(t)$ for $b = \{0.6, 0.7, 1.2\}.$



Figure 2.3 Twisting based algorithm. States of (2 - 6) for b = 0 with $k_1 = 4$, and $k_2 = 2$ under persistent disturbance d = 1.



Figure 2.4 Super-twisting based algorithm. States of (2 - 17) with $k_3 = 2$, $k_1 = 0.3$, and three different values of $p = \{0.5, 0.7, 1.2\}$.







Figure 2.5 Super-twisting based algorithm. Disturbance $d_1 = |z_1|^p \operatorname{sign}(z_2) \cdot \operatorname{sin}(t)$ for $p = \{0.5, 0.7, 1.2\}.$



Figure 2.6 Super-twisting based algorithm. States of (2 - 17) for p = 0.5 with $k_3 = 1$ and $k_1 = 1$ without perturbation.



Figure 2.7 Generic super-twisting based algorithm. States of (2 - 23) with $k_1 = k_{1n} = 4$, and $k_3 = k_{3n} = 2$ under three values of initial conditions, $(z_1(0), z_2(0)) = \{(-0.1, -0.1), (-1, -1), (-10, -10)\}.$



(a) Disturbance d_1 for time t = [0, 10] s.

(b) Disturbance d_2 for time t = [0, 10] s.

Figure 2.8 Generic super-twisting based algorithm. Disturbance $d_1 = \operatorname{sign}(z_2) \cdot \operatorname{sin}(t) (|z_1|^p + |z_1|^q)$ and $d_2 = \operatorname{sign}(z_2) \cdot \operatorname{sin}(t) (|z_1|^{2p-1} + |z_1|^{2q-1})$ for three different values of initial conditions, $(z_1(0), z_2(0)) = \{(-0.1, -0.1), (-1, -1), (-10, -10)\}$.



Figure 2.9 Generic twisting based algorithm. States of (2 - 11) with $p = \frac{1}{1.4}, q = \frac{1}{0.4}$,

$$k_1 = k_{1n} = 4$$
, and $k_2 = k_{2n} = 2$ under the disturbance,

$$d = \operatorname{sign}(z_2) \cdot \sin(t) \left(|z_1|^{2p-1} + |z_1|^{2q-1} \right) + \operatorname{sign}(z_1) \cdot \sin(t) \cdot \left(|z_2|^{2-\frac{1}{p}} + |z_2|^{2-\frac{1}{q}} \right).$$



Figure 2.10 Generic second order based algorithm. States of (2 - 29) with $p = \frac{1}{1.4}, q = \frac{1}{0.4}$,

$$k_1 = k_{1n} = k_2 = k_{2n} = 4$$
, and $k_3 = k_{3n} = 2$ under the disturbances,
 $d_1 = \operatorname{sign}(z_2) \cdot \sin(t) (|z_1|^p + |z_1|^q)$ and
 $d_2 = \operatorname{sign}(z_2) \cdot \sin(t) (|z_1|^{2p-1} + |z_1|^{2q-1}) + \operatorname{sign}(z_1) \cdot \sin(t) \cdot (|z_2|^{2-\frac{1}{p}} + |z_2|^{2-\frac{1}{q}}).$

1) Twisting based algorithm:

Figure 2.1 shows the convergence of the states, z_1 and z_2 in finite time for b = 0.6 and 0.7. For b = 1.2 the states are converging to the origin asymptotically. Figure 2.2 shows the non-Lipschitz disturbances acting on the system. For the system (2 - 6), with parameters as stated in the simulation setup section, i.e. if $M_1 = M_3 = 0$, as per weak Lyapunov function [18], the requirements for finite time stability are $k_2 > M_2$ with any $k_1 > 0$. The gains of the simulation are selected to satisfy these conditions.

The following are the conditions on the gains, of the same dynamics under the same perturbations, obtained from the literature for comparison purposes.

1. In the results of [37] (see Theorem 2 of [37]), the conditions on the gains are

$$k_1(k_2 - M_2) > \frac{6}{(1+\alpha)(2-\alpha)}, \quad k_2 > \max\left\{M_2, M_2 + \frac{3}{2(1+\alpha)}\right\},$$

$$k_1 > 0, \quad \frac{k_1}{2} > \frac{(k_2 - M_2)}{1+\alpha} > 0$$

with $\alpha = \frac{2b}{1+b}$ for $b \in [0,1)$.

2. In Theorem 2 and 3 of [38], the conditions on gains are $k_1 > 0$, $k_2 > \max\left\{M_2, M_2 + \frac{3}{2(1+\alpha)}\right\}$, $\frac{k_1}{2} > \frac{(k_2 - M_2)}{1+\alpha} > 0$, $k_1(k_2 - M_2) > \frac{6}{(1+\alpha)(2-\alpha)}$, $k_1^3 > 1$ and $k_2 > M_2 + \frac{3(1-\alpha)}{2}$, $k_1 > \frac{1}{1+\alpha} \max\{M_2, \alpha\}$, $k_1(k_2 - M_2) > \frac{3}{2}\left(\frac{1+\alpha}{(2-\alpha)}\right)$ with $\alpha = \frac{2b}{1+b}$ for $b \in [0,1)$. 3. In [44] (see Theorem 1 of [44]) and [46] (see theorem 3.2 of [46]), the conditions

are
$$k_1 - M_2 > k_2 > M_2 > 0$$
 with are need only applicable for $\frac{2b}{1+b} \in \left(\frac{2}{3}, 1\right)$

4. In [45] only for $\frac{2b}{1+b} \in [0,1)$ it requires

$$k_1 > (k_2 + M_2) \frac{(2 - \alpha)}{2}$$
 and $k_2 > M_2$

Clearly, the simulation choice of $k_1 = 1$, $k_2 = 2$ with $M_2 = 1$ does not satisfy all of the above conditions. Also, the simulation parameters of $b = \{0.6, 0.7, 1.2\}$ do not belong to the range considered there. However, as per remark 2.4, the strict Lyapunov function presented in section 2.2 allows the conditions imposed on the simulations, showing different convergence properties dependent on the parameter *b*. Thus, the results show that conservativeness of previous results mentioned above. In particular, the Lyapunov results of [18], [37] and [38] are not able to extend to exponent greater than 1. It is also worth mentioning that the types of disturbances are also extended, specifically, in [18], only global asymptotic stability is achieved for 0 < b < 1 under disturbances upper bounded by one state variable only, which in part due to the weak Lyapunov function employed. In contrast, the strict Lyapunov function of section 2.2 allows disturbances upper bounded by the sum of both states.

Figure 2.3 shows the finite time convergence of system (2 - 6) under persistent disturbance d = 1 for b = 0 with $k_1 = 4$, and $k_2 = 2$. From remark 2.7, finite time convergence is achieved when for any $k_1 > k_2 + M$, and $k_2 > M$, which coincides with the well-known result [20]. On the other hand, the results of [37] and [38], while applicable to b = 0.5, it is unclear on how to

obtain this conditions.

As a result, the strict Lyapunov function presented here is able to fully characterize the stability of the twisting based algorithm for any $b \ge 0$, which essentially fills the gaps in the literature, in the sense that those prior results are only applicable to certain range of *b* while at the same time requiring conservative conditions on gains with respect to disturbances.

2) Super-twisting based algorithm:

Figure 2.4 shows the convergence of the states, z_1 and z_2 in finite time for p = 0.5 and 0.7. For p = 1.2 the states are converging to the origin asymptotically. Figure 2.5 shows the non-Lipschitz disturbances acting on the system. For the system (2 - 17), with parameters as stated in the simulation setup section, i.e. if $M_2 = M_3 = M_4 = 0$, as per energy based Lyapunov function (see remark 2.15), the requirements for stability (finite time if p = [0,1), exponential if p = 1, and asymptotical if p > 1) are $k_3 > M_1$ with any $k_1 > 0$. The gains of the simulation are selected to satisfy these conditions.

In [26], the Lyapunov function presented is for the special case of p = 0.5 of the supertwisting based algorithm studied here. Under the same disturbances as per the simulation, the conditions given for finite time stability are

$$k_1 > k_3 \frac{5M_1k_3 + 4\left(\frac{M_1}{k_3}\right)^2}{2(k_3 - 2M_1)}$$
 and $k_3 > 2M_1$

which is clearly not satisfied by the simulation parameter values for the gains, which shows the conservativeness of prior results.

Meanwhile, the Lyapunov function for the super-twisting algorithm presented by [26] and [27] (for p = 0.5) will cause singularity issue when used in further application such as with certainty-equivalence method in [67]. This issue is due to the non-Lipschitz nature of their proposed Lyapunov functions, in which case, is not a issue with our proposed Lyapunov function which is locally Lipschitz and strict (see section 2.4). Pertaining to the issue of solving singularity, a continuously differentiable Lyapunov function for the super-twisting algorithm (p = 0.5) is presented in [68], albeit, it comes with conservative gain conditions, namely, for the unperturbed system, $k_3^2 > 2k_1 > 0$, while in the present results, as per remark 2.18 and (2 -21), for the unperturbed system, finite time is achieved with any $k_1 > 0$ and $k_3 > 0$, which is shown in Figure 2.6 with $k_3 = 1$ and $k_1 = 1$. Also worth mentioning is that the mentioned prior results is only applicable to the case of p = 0.5, while the results shown here is for any $p \ge 0.5$ as per results shown in Figure 2.4.

3) Generic super-twisting based algorithm:

Figure 2.7 shows the uniform finite time convergence of the states, z_1 and z_2 under three different values of initial conditions that differ by an order of magnitude. In fact, for all three different initial conditions cases, all of them reach the origin in time, t less than 5 seconds even though the furthest initial conditions are of two order of magnitudes difference than the closest one. This strong convergence feature is due to the strong control terms that have exponent greater than 1, namely, $|z_1|^q \operatorname{sign}(z_1)$ and $|z_1|^{2q-1} \operatorname{sign}(z_1)$.

Figure 2.8 shows the non-Lipschitz disturbances acting on the system. Similar forms of disturbances are also studied in [46] and [17]. Particularly, the Lyapunov function proposed in [17] is able to show uniform finite time convergence as well under disturbances that are upper-bounded in a form similar to that in the simulation. However, these prior results require two distinct Lyapunov functions to ascertain different convergence properties of the system, i.e. uniform and finite time convergence, in which case for the finite time convergence, their Lyapunov function suffer the same singularity issue mentioned in the super-twisting based results above. In addition, due to the two different structure of Lyapunov functions employed there, the control requires an additional term in the \dot{z}_2 dynamics, i.e. $|z_1|^{p+q-1} \operatorname{sign}(z_1)$, to ensure strictness of Lyapunov functions. The proposed strict and locally-Lipschitz Lyapunov function (see section 2.5) is able to overcome these shortcomings.

4) Generic twisting based algorithm:

Figure 2.9 shows the uniform finite time convergence of the states, z_1 and z_2 under the influence of non-Lipschitz disturbances. The disturbances, which comprise both states of the system, considered in this simulation satisfy the upper bound of the system description in section 2.3. In fact, as per remark 2.9, since p = 1/1.4 < 1 and q = 1/0.4 > 1, uniform finite time convergence is guaranteed. The generic twisting algorithm is based on the results of twisting based algorithm by combining control term of different homogeneity with the intent of combining different stability properties within an algorithm

5) Generic second order based algorithm:

Figure 2.10 shows the uniform finite time convergence of the states, z_1 and z_2 under the influence of non-Lipschitz disturbances. The disturbances, which comprise both states of the system, considered in this simulation satisfy the upper bound of the system description in section 2.6. In fact, as per remark 2.30, since p = 1/1.4 < 1 and q = 1/0.4 > 1, uniform finite time convergence is guaranteed. The generic second order algorithm is based on the results of combining the generic twisting and generic super-twisting based algorithm. Through such combination, the inherent energy based function of the system (which is found to be a weak Lyapunov function when applied on either twisting or super-twisting based algorithm alone), is a strict and locally Lipschitz Lyapunov function as per remark 2.28.

Remark 2.35. The disturbances considered here (see Figure 2.2, 2.5, 2.8, 2.9(c), and 2.10(c)) comprise bounded discontinuity, which corroborates with the results of the stability analysis which employ the generalized Lyapunov theorem (see section 2.1).

2.8 Summary

Two twisting-based family of algorithms, two super-twisting based family of algorithms and a generic second order algorithm that combines the super-twisting and twisting algorithms are developed. In each case, strict Lyapunov functions have been introduced, that can fully characterize different stability properties of a parameterized family of algorithms. For the twisting based algorithm, it generalize the well-known twisting algorithm, continuous finite-time second order system, linear PD control, and uniform convergence algorithm. While for the super-twisting based algorithm, it generalize the super-twisting algorithm, homogeneous and linear PI control, and uniform convergence algorithm. Then we show that the linear combination of two homogeneous algorithms with different degree of homogeneity can indeed produce a system that has the characteristics of its individual component. Due to the availability of strict Lyapunov functions, settling time for finite time convergence can also be obtained. Finally, the robustness to different classes of perturbations can be easily considered as well.

Chapter 3: ROBOT MANIPULATOR CONTROL: FULL STATE FEEDBACK APPROACH

In this chapter, a family of controllers is developed for the trajectory tracking of robot manipulator. Based on the twisting based family of algorithms presented in the previous chapter, the proposed controller is able to generalised PID control to a homogeneous PID-like control. Semiglobal practical tracking stability is achieved despite uncertainty and additive disturbances in the robot dynamics. For the special case of regulation problem, the proposed controller is able to achieve finite-time or exponential convergence, depending on the chosen parameters. The stability analysis allows selection of control gains based on desired performance instead of gains tuning. Numerical simulations using two-link robot manipulator demonstrate the performance of the proposed controller.

3.1 System Description

A nonlinear mechanical system with *n*-degree of freedom in closed loop with a nonlinear controller that generalised the PID control is considered.

3.1.1 Manipulator Dynamics

A class of rigid, fully actuated, unconstrained mechanical systems which can be modelled by the Euler-Lagrange principle that results in a class of nonlinear systems modelled by a set of highly coupled nonlinear differential equations is considered. The dynamics of n-joint serial rigid robotic manipulators can be described by the following differential equation [117]

$$M(q)\ddot{q} + C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D(q,\dot{q},t) = \tau, \qquad (3-1)$$

where $q \in \mathbb{R}^n$ is the vector of generalized joint coordinates, $M(q) \in \mathbb{R}^{n \times n}$ is the inertia matrix, and $C(q, \dot{q}) \dot{q}$, $F\dot{q}$, G(q), $D(q, \dot{q}, t)$, $\tau \in \mathbb{R}^n$ are, respectively, the vectors of Coriolis and centrifugal, viscous friction, gravity, disturbances, and input generalized forces, with F being a constant, positive definite, diagonal (viscous friction coefficient) matrix and $D(q, \dot{q}, t)$ being a locally bounded disturbances. The terms of the robot manipulator dynamics satisfy some well-known properties (see for instance [117], [142]), in which several are recollected here.

Property 3.1: The inertia matrix M(q) is a positive definite symmetric matrix satisfying $\underline{m} \leq ||M(q)|| \leq \overline{m}, \forall q \in \mathbb{R}^n$, for some positive constants $\underline{m} \leq \overline{m}$.

Property 3.2: The Coriolis matrix $C(q, \dot{q})$ satisfies:

3.2.1. $x^{T}\left[\frac{1}{2}\dot{M}(q,\dot{q})-C(q,\dot{q})\right]x=0, \forall x,q,\dot{q}\in\mathbb{R}^{n};$

3.2.2.
$$\dot{M}(q,\dot{q}) = C(q,\dot{q}) + C^T(q,\dot{q}), \forall q,\dot{q} \in \mathbb{R}^n;$$

- 3.2.3. $C(w, x + y)z = C(w, x)z + C(w, y)z, \quad \forall w, x, y, z \in \mathbb{R}^{n};$
- 3.2.4. $C(x, y)z = C(x, z)y, \quad \forall x, y, z \in \mathbb{R}^n;$
- 3.2.5. $||C(x, y)z|| \le C_m ||y|| ||z||, \forall x, y, z \in \mathbb{R}^n$, for some constant $C_m \ge 0$.

Property 3.3: The gravitational torques vector for robots having only revolute joints satisfies (see [142] page 101):

3.1. $||G(q)|| \le G_m$, $\forall q \in \mathbb{R}^n$, for some constant $G_m > 0$.

3.2.
$$||G(x) - G(y)|| \le k_g ||x - y||, \forall x, y \in \mathbb{R}^n$$
, for some constant $k_g > 0$.

Property 3.4: The viscous friction coefficient matrix satisfies $\underline{f} \leq ||F|| \leq \overline{f}$, where $0 < \underline{f} := \min_{i} \{f_i\} \leq \max_{i} \{f_i\} := \overline{f}$.

In this chapter, it is assume that both joint positions and velocities are available from measurement, i.e. full state feedback is viable. The control objective here is to design a robust full state feedback controller that ensures the robot configuration vector q tracks a desired trajectory vector, q_d (t) with an ultimately bounded error that can be made as small as required, from any initial conditions that belong to an arbitrarily large compact set.

The desired trajectory vector, $q_d(t)$ is assumed to be twice continuously differentiable vector-function such that $||q_d(t)||$, $||\dot{q}_d(t)||$, and $||\ddot{q}_d(t)||$ are bounded by *a priori* known constants. This is a standard assumption in the trajectory tracking control of robot manipulator (see for instance [103], [131], [113]).

Note that no continuity assumption is made on $D(q, \dot{q}, t)$, so it may have discontinuity, such as Coulomb friction. In particular, the form of $D(q, \dot{q}, t)$ considered here is assumed to be upper bounded by the function

$$\|D\| \le p_0 + p_1 \|e_1\| + p_2 \|\dot{q}\| + p_3 \|e_1\|^2 + p_4 \|\dot{q}\|^2, \text{ with } \|D\| := \sup\{\|\zeta\| : \zeta \in K[D]\}$$

where p_0 , p_1 , p_2 , p_3 , and p_4 are some nonnegative constants, while $e_1 := q - q_d \in \mathbb{R}^n$, and $e_2 = \dot{q} - \dot{q}_d \in \mathbb{R}^n$.

3.1.2 Control Development

The following notions, which were used in [36] and [7], are introduced for simplicity of notation and will be used in the analysis and design of the controller.

$$sig(x)^a = \left[|x_1|^a \operatorname{sign}(x_1), \dots, |x_n|^a \operatorname{sign}(x_n) \right]^T, \forall x \in \mathbb{R}^n$$

The controller proposed is given by

$$\tau = -K \operatorname{sig}(s)^a, \qquad (3-2)$$

where *K* is a positive definite diagonal matrix, i.e. $K = \text{diag}\{k_i\}_{i=1}^n$, with $k_i > 0, \forall i = 1, ..., n, a$ ≥ 0 constant, and $s \in \mathbb{R}^n$ is the desired error dynamics defined as $s = e_2 + \sigma$, with

$$\dot{\sigma} = K_2 \operatorname{sig}(e_2)^{\frac{2b}{1+b}} + K_1 \operatorname{sig}(e_1)^b,$$
 (3 - 3)

where K_1 and K_2 are positive definite diagonal matrices, i.e. $K_1 = \text{diag}\{k_{1i}\}_{i=1}^n$, with $k_{1i} > 0$, $K_2 = \text{diag}\{k_{2i}\}_{i=1}^n$, with $k_{2i} > 0 \quad \forall i = 1, ..., n, b \ge 0$ constant.

3.2 Stability Analysis

The closed-loop system of (3 - 1), (3 - 2), and (3 - 3) can be written as

$$\begin{split} \dot{\sigma} &= K_2 \operatorname{sig}(e_2)^{\frac{2b}{1+b}} + K_1 \operatorname{sig}(e_1)^b, \\ \dot{e}_1 &= e_2, \\ \dot{e}_2 &= -M^{-1}(q) K \operatorname{sig}(s)^a - M^{-1}(q) (C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D) - \ddot{q}_d, \end{split}$$

To rewrite the closed-loop system in a form more convenient for analysis, let us define the change of variable $z_1 = e_1$, and $z_2 = -\sigma$, and we obtain the following form of closed-loop system,

$$\dot{z}_{1} = z_{2} + s,$$

$$\dot{z}_{2} = -K_{2} \operatorname{sig}(z_{2} + s)^{\frac{2b}{1+b}} - K_{1} \operatorname{sig}(z_{1})^{b},$$

$$\dot{s} = -M^{-1}(z_{1} + q_{d}) \operatorname{Ksig}(s)^{a} + \Delta(\cdot),$$

(3 - 4)

where

$$\Delta(\cdot) = -M^{-1}(z_1 + q_d)C(z_1 + q_d, z_2 + s + \dot{q}_d)(z_2 + s + \dot{q}_d) - M^{-1}(z_1 + q_d)F(z_2 + s + \dot{q}_d) - M^{-1}(z_1 + q_d)G(z_1 + q_d) - M^{-1}(z_1 + q_d)D - \ddot{q}_d + K_2 \operatorname{sig}(z_2 + s)^{\frac{2b}{1+b}} + K_1 \operatorname{sig}(z_1)^b$$

3.2.1 Construction of Lyapunov Function

From the closed-loop dynamics (3 - 4), the structure (z_1, z_2) is essentially the desired error dynamics injected by the controller through (3 - 3). Essentially, for any $i \in 1,...,n$, when $s_i = 0$, the dynamics of the subsystem (z_{1i}, z_{2i}) is identical to that of the twisting-based family of algorithm studied in section 2.2 of Chapter 2. In other words, s_i can be viewed as a perturbations on the (z_{1i}, z_{2i}) dynamics. While in the *s*-dynamics, its structure is akin to that of sliding mode control.

Since the differential equations (3 - 4) have discontinuous right-hand side, i.e. when a = 0 and/or b = 0, or D, and since no continuity assumption is made on D, its solutions are understood in the sense of Filippov (see definition 2.1).

The following Lyapunov functions will be used in the analysis:

for
$$i = 1, ..., n$$
,
 $V_{zi}(z_{1i}, z_{2i}) = \frac{k_{1i}^{2}}{(1+b)^{2}} |z_{1i}|^{2+2b} + \frac{1}{4} |z_{2i}|^{4} + r_{zi} |z_{1i}|^{\frac{3+3b}{2}} \operatorname{sign}(z_{1i}) z_{2i} + \frac{k_{1i}}{(1+b)} |z_{1i}|^{1+b} |z_{2i}|^{2}$,
and
 $V_{s}(s,q) = \frac{1}{2} s^{T} M(q) s$

Remark 3.1. Note that the Lyapunov function for the (z_{1i}, z_2) -subsystem is a strict Lyapunov function proposed for the twisting-based family of algorithms (section 2.2.2) of Chapter 2, where $\forall q \in \mathbb{R}^n$:

$$\underline{\pi}_{1i} \left(|z_{1i}|^{2+2b} |z_{2i}|^4 \right) \le V_{zi} \left(|z_{1i}|, |z_{2i}|^2 \right) \le \overline{\pi}_{1i} \left(|z_{1i}|^{2+2b} + |z_{2i}|^4 \right)$$

with

$$\underline{\pi}_{1i} := \min\left\{\frac{1}{8}, \frac{1}{2}\left(\frac{k_{1i}}{1+b}\right)^2\right\}, \ \overline{\pi}_{1i} := \max\left\{\frac{3r_{zi}}{4} + \left(\frac{k_{1i}}{1+b}\right)^2 + \left(\frac{k_{1i}}{2+2b}\right), \ \left(\frac{k_{1i}}{2+2b}\right) + \frac{r_{zi}}{4} + \frac{1}{4}\right\},$$

while for V_s ,

$$\frac{1}{2}\underline{m}\|s\|^2 \le V_s \le \frac{1}{2}\overline{m}\|s\|^2.$$

Let us define the following sets,

$$\Omega_{z}(R) = \{ (z_{1}, z_{2}) \in \mathbb{R}^{2n} : V_{zi}(z_{1i}, z_{2i}) \le \rho_{1i}(R), \dots, V_{zn}(z_{1n}, z_{2n}) \le \rho_{1i}(R), \text{ for } i = 1, \dots, n \},\$$

$$\Omega_{s}(R) = \{ s \in \mathbb{R}^{n} : V_{s} \le R^{2}/2 \},\$$

$$\Omega(R) = \{ (z_1, z_2, s) \in \mathbb{R}^{3n} : \Omega_z(R), \ \Omega_s(R) \},\$$

where

$$\begin{split} & \text{for } i = 1, \dots, n \\ & \rho_{1i}(R) = \begin{cases} \left(\frac{4\overline{\pi}_{1i} \frac{3+5b}{4+4b}}{\omega_{4i}} \right)^4 \left(\frac{2^{\frac{1}{2+2b}} \omega_{2i}}{\underline{\pi}_{1i} \frac{1+2b}{2+2b}} \right)^4, \\ & \left(\frac{4\overline{\pi}_{1i} \frac{3+5b}{4+4b}}{\omega_{4i}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{3i}}{\underline{\pi}_{1i} \frac{3}{4}} \right) \right)^{\frac{2+2b}{b}} \end{cases} \times \left(\frac{R}{\sqrt{\underline{m}}} \right)^4, \quad \text{for } b > 0, \\ & \left(\frac{4\overline{\pi}_{1i} \frac{3+5b}{4+4b}}{\omega_{4i}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{3i}}{\underline{\pi}_{1i} \frac{3}{4}} \right) \right)^{\frac{2+2b}{b}} \end{cases} \times \left(\frac{R}{\sqrt{\underline{m}}} \right)^4, \quad \text{for } b > 0, \\ & \left(\frac{1}{2} \frac{4\overline{\pi}_{1i} \frac{3+5b}{4+4b}}{\omega_{4i}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{3i}}{\underline{\pi}_{1i} \frac{3}{4}} \right) \right)^{\frac{2+2b}{b}} \end{cases} \\ & \left(\frac{1}{2} \frac{$$

bd
$$\phi_{3i}\left(\frac{R}{\sqrt{\underline{m}}}\right)$$
 is the boundary of the region

$$\phi_{3i}\left(\frac{R}{\sqrt{\underline{m}}}\right) = \left\{ \left(z_{1i}, z_{2i}\right) \in \mathbb{R}^2 : \left|z_{1i}\right| \le a_{3i}\left(\frac{R}{\sqrt{\underline{m}}}\right)^2, \left|z_{2i}\right| \le \frac{2k_{1i}}{k_{2i}}\left(\frac{R}{\sqrt{\underline{m}}}\right) \right\},\$$

which is a compact set, with the constant $a_{3i} > 0$ (see Appendix A.1).

Let also $\gamma_1(R) = \operatorname{diag} \{ \gamma_{1i}(R) \}_{i=1}^n$ with $\gamma_{1i}(R) > 0, \forall i = 1, ..., n$, such that $\forall (z_1, z_2, s) \in \Omega(R)$:

$$s^{T}M(\cdot)\boldsymbol{K}[\Delta(\cdot)] + \frac{1}{2}s^{T}\dot{M}(\cdot)s \leq \sum_{i=1}^{n}\gamma_{1i}(\boldsymbol{R})|s_{i}|,$$

Remark 3.2. Note that such an upper bound always exists for any given compact set $\Omega(R)$ since *M*, *C*, *F*, *G*, *D*, and the desired trajectories are locally bounded, it implies that $\Delta(.)$ is

locally bounded as well (i.e. it compose of summation of locally bounded function). From [143], the multi-valued function $K[\Delta(.)]$ is locally bounded as well. Also, using the skew-symmetry property 3.2.2, the function, $\frac{1}{2}s^T\dot{M}(\cdot)s = s^TC(\cdot)s$ is locally bounded as well. Hence, within a compact set, an upper bound on the above function exists.

3.2.2 Stability Criterion Determination

The time derivatives of the Lyapunov functions, in accordance to lemma 2.4, $\forall (z_1, z_2, s) \in \Omega(R)$ of the closed-loop system satisfy the following inequalities:

Differential inequalities for the z-subsystem (see Appendix A.1):

$$\forall i = 1, ..., n,$$

for b > 0:

$$\begin{split} \dot{V}_{zi}(z_{1i}, z_{2i}) &\stackrel{a.e.}{\leq} -\frac{1}{2} \left(\frac{\omega_{4i}}{\overline{\pi}_{1i} \frac{3+5b}{4+4b}} \right) V_{zi} \frac{3+5b}{4+4b} - \frac{1}{4} \left(\frac{\omega_{4i}}{\overline{\pi}_{1i} \frac{3+5b}{4+4b}} \right) V_{zi} \frac{1+2b}{2+2b} \left(V_{zi} \frac{1}{4} - |s_i| \left(\frac{4\overline{\pi}_{1i} \frac{3+5b}{4+4b}}{\omega_{4i}} \right) \left(\frac{2^{\frac{1}{2+2b}} \omega_{2i}}{\underline{\pi}_{1i} \frac{1+2b}{2+2b}} \right) \right) \\ &- \frac{1}{4} \left(\frac{\omega_{4i}}{\overline{\pi}_{1i} \frac{3+5b}{4+4b}} \right) V_{zi} \frac{3}{4} \left(V_{zi} \frac{b}{2+2b} - a_{1i} |s_i| \frac{2b}{1+b} \left(\frac{4\overline{\pi}_{1i} \frac{3+5b}{4+4b}}{\omega_{4i}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{3i}}{\underline{\pi}_{1i} \frac{3}{4}} \right) \right) \end{split}$$
(3 - 5)

for b = 0:

$$\begin{split} \dot{V}_{zi}(z_{1i}, z_{2i}) &\stackrel{\text{a.e.}}{\in} \dot{\widetilde{V}}_{zi} \leq 2k_{1i}^{2} |z_{1i}| |s_{i}| + \frac{3}{2} r_{i} |z_{1i}|^{\frac{1}{2}} |z_{2i}| |s_{i}| + k_{1i} |z_{2i}|^{2} |s_{i}| + \frac{3}{2} r_{i} |z_{1i}|^{\frac{1}{2}} |z_{2i}|^{2} \\ &- 2k_{1i} k_{2i} |z_{1i}| z_{2i} \text{SGN}(z_{2i} + s_{i}) \\ &- k_{2i} |z_{2i}|^{3} \operatorname{sign}(z_{2i}) \text{SGN}(z_{2i} + s_{i}) - r_{i} (k_{1i} - k_{2i}) |z_{1i}|^{\frac{3}{2}} \\ < 0 \quad \text{for} \quad V_{zi} \geq \max_{(z_{1i}, z_{2i}) \in \operatorname{bd} \phi_{3i}(|s_{i}|)} V_{zi}(z_{1i}, z_{2i}) \end{split}$$
(3 - 6)

Differential inequality for the *s*-subsystem:

$$\begin{split} \dot{V}_{s} \stackrel{a.e}{\in} \dot{\widetilde{V}}_{s} &= \bigcap_{\xi \in \partial V_{s}} \xi^{\mathrm{T}} \left(\begin{array}{c} \boldsymbol{K}[f](s) \\ 1 \end{array} \right) = \nabla V_{s}^{\mathrm{T}} \boldsymbol{K}[f](s) + \frac{1}{2} s^{\mathrm{T}} \dot{M}(\cdot) s \\ & \subset s^{\mathrm{T}} M(\cdot) \boldsymbol{K} \Big[-M^{-1}(\cdot) K \mathrm{sig}(s)^{a} + \Delta(\cdot) \Big] + \frac{1}{2} s^{\mathrm{T}} \dot{M}(\cdot) s \\ & = -s^{\mathrm{T}} \boldsymbol{K} \Big[K \mathrm{sig}(s)^{a} \Big] + s^{\mathrm{T}} M(\cdot) \boldsymbol{K} [\Delta(\cdot)] + \frac{1}{2} s^{\mathrm{T}} \dot{M}(\cdot) s \\ & = -\sum_{i=1}^{n} k_{i} |s_{i}|^{1+a} + s^{\mathrm{T}} M(\cdot) \boldsymbol{K} [\Delta(\cdot)] + \frac{1}{2} s^{\mathrm{T}} \dot{M}(\cdot) s \\ & \leq -\sum_{i=1}^{n} k_{i} |s_{i}|^{1+a} + \sum_{i=1}^{n} \gamma_{1i}(R) |s_{i}| \end{split}$$

Remark 3.3. Note that the above analysis for the s-subsystem also apply to the case of a = 0. In particular, when a = 0, using the following notation,

$$N^{+}(s) = \{i \in \{1, ..., n\} : |s_i| \neq 0\}, \quad N^{0}(s) = \{i \in \{1, ..., n\} : s_i = 0\},\$$

if $N^+(s) \neq \emptyset$, then for at least one index $i \in \{i, ..., n\}, |s_i| \neq 0$, observe that

$$-s^{\mathrm{T}}KK[\operatorname{sign}(s)] = -\sum_{i \in N^{+}(s)} k_{i}|s_{i}| - \sum_{i \in N^{0}(s)} k_{i}(0) \times [-1, +1] = -\sum_{i \in N^{+}(s)} k_{i}|s_{i}| - 0 = -\sum_{i=i}^{n} k_{i}|s_{i}|,$$

if
$$N^+(s) = \emptyset \Longrightarrow s_i = 0, \forall i \in \{i, \dots, n\}$$
, hence, $-s^{\mathsf{T}} K \mathbf{K}[\operatorname{sign}(s)] = 0$.

Thus, when a = 0, $\forall s_i \in \mathbb{R}$, $\forall i \in \{i, ..., n\}$: $-s^{\mathsf{T}} K \mathbf{K}[\operatorname{sign}(s)] = -\sum_{i=i}^{n} k_i |s_i|$

Hence,

for a = 0:

$$\begin{split} \dot{\vec{V}}_{s} &\leq -\sum_{i=1}^{n} |s_{i}| (k_{i} - \gamma_{1i}(R)) \leq -\min_{i} (k_{i} - \gamma_{1i}(R)) \sum_{i=1}^{n} (|s_{i}|^{2})^{\frac{1}{2}} \\ &\leq -\min_{i} (k_{i} - \gamma_{1i}(R)) (\sum_{i=1}^{n} |s_{i}|^{2})^{\frac{1}{2}}, \quad \text{using lemma 2.2 of chapter 2,} \\ &= -\min_{i} (k_{i} - \gamma_{1i}(R)) ||s|| \end{split}$$
(3 - 7)

for *a* > 0:

$$\begin{split} \hat{\tilde{V}}_{s} &\leq -\lambda_{\min}(K) \sum_{i=1}^{n} |s_{i}|^{1+a} + \sum_{i=1}^{n} \gamma_{1i}(R) |s_{i}| \\ &\leq -\frac{\lambda_{\min}(K)}{n^{a}} \left(\sum_{i=1}^{n} |s_{i}| \right)^{1+a} + \lambda_{\max}(\gamma_{1i}(R)) \sum_{i=1}^{n} |s_{i}|, \quad \text{using lemma 2.2 of chapter 2,} \\ &= -\frac{\lambda_{\min}(K)}{n^{a}} \left(\sum_{i=1}^{n} (|s_{i}|^{2})^{\frac{1}{2}} \right)^{1+a} + \lambda_{\max}(\gamma_{1i}(R)) \sum_{i=1}^{n} (|s_{i}|^{2})^{\frac{1}{2}} \\ &\leq -\frac{\lambda_{\min}(K)}{n^{a}} \left(\left(\sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}} \right)^{1+a} + \lambda_{\max}(\gamma_{1i}(R)) \sqrt{n} \left(\sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}}, \quad \text{using lemma 2.2 of chapter 2,} \\ &= -\frac{\lambda_{\min}(K)}{n^{a}} \|s\|^{1+a} + \lambda_{\max}(\gamma_{1i}(R)) \sqrt{n} \|s\| \end{split}$$

$$(3 - 8)$$

Remark 3.4. For ease of presentation, note that the above differential inequalities are stated for two cases for each subsystems. This is due to the fact that the subsystems are discontinuous when b = 0 and a = 0 respectively.

Theorem 3.1: For any given $K_1, K_2 > 0$, suppose that the initial conditions for the closed-loop system (3 - 4) belong to a given compact set, there always exists a c > 0 such that initially, $(z_1, z_2, s) \in \Omega(c)$. Depending on the value of *a*:

For the case of *a* > 0, a μ > 0 can be selected such that *c* > μ > 0 and by selecting *K* such that

$$\lambda_{\min}(K) > \frac{\lambda_{\max}(\gamma_1(c))n^{a+0.5}\overline{m}^{\frac{a}{2}}}{\mu^a},$$

all the trajectories will enter the compact set $\Omega(\mu)$, in finite time, and stay there for all future times.

2. For the case of a = 0, by selecting K such that

$$k_i > \gamma_{1i}(c), \forall i = 1,...,n, ...$$

all the trajectories will enter the compact set $\Omega_z(c) \times \Omega_s(0)$ in finite time, and stay there for all future times. Additionally, the states (z_1, z_2) , have finite time convergence for $0 \le b \le 1$, exponential convergence for b = 1, or for asymptotical convergence for $b \ge 1$. Proof of Theorem 3.1: The stability analysis proceeds in two steps.

1. Obviously, for a given compact set of initial conditions, there exists c > 0 such that initially (z_1, z_2, s) belong to some compact set strictly inside $\Omega(c)$. A trajectory may leave the set $\Omega(c)$ only through one of the boundaries:

$$V_{zi}(z_{1i}, z_{2i}) = \rho_{1i}(c), \forall i = 1, ..., n, \text{ or } V_s = c^2/2.$$

Let us show that it is impossible.

(a) For the s-subsystem, $V_s = \frac{1}{2} s^{\mathrm{T}} M(q) s = \frac{1}{2} c^2$ implies $||s|| \ge \frac{c}{\sqrt{m}}$.

Hence, for a = 0, from (3 - 7), $\dot{V}_s \le -\min_i (k_i - \gamma_{1i}(c)) \|s\|$

$$\therefore \dot{\tilde{V}}_s \leq -c_1, \text{ where } c_1 = \min_i \left(k_i - \gamma_{1i}(c) \right) \left(\frac{c}{\sqrt{\overline{m}}} \right),$$

For *a* > 0, from (3 - 8)

$$\dot{V}_{s} \stackrel{a.e.}{\leq} -\frac{\lambda_{\min}(K)}{n^{a}} \|s\|^{1+a} + \lambda_{\max}(\gamma_{1i}(c))\sqrt{n}\|s\|$$
$$= -\|s\|\left[\frac{\lambda_{\min}(K)}{n^{a}}\|s\|^{a} - \lambda_{\max}(\gamma_{1}(c))\sqrt{n}\right]$$
$$\therefore \dot{V}_{s} \stackrel{a.e.}{\leq} - c_{2}$$

where
$$c_2 = \frac{c}{\sqrt{\overline{m}}} \left[\frac{\lambda_{\min}(K)}{n^a} \left| \frac{c}{\sqrt{\overline{m}}} \right|^a - \lambda_{\max}(\gamma_1(c))\sqrt{n} \right]$$
 (3 - 9)

Hence, for k_i sufficiently big, c_1 is positive for a = 0, or $\lambda_{\min}(K)$ sufficiently big, c_2 is positive for a > 0, such that, $\dot{V}_s < 0$ almost everywhere. If the gain K satisfy the

conditions of the Theorem 3.1, then, V_s is a decreasing function of t, so s stays in $\Omega_s(c)$ and

$$||s|| \le \frac{c}{\sqrt{\underline{m}}}$$
, together with the fact that $|s_i| \le ||s||$, for $i = 1, ..., n$, we have $|s_i| \le \frac{c}{\sqrt{\underline{m}}}$, for $i = 1, ..., n$.

(b) For the *z*-subsystem, for b > 0, at the boundary, it implies

$$\forall i = 1, ..., n :$$

$$V_{zi} = \rho_{1i}(c) = \max\left\{ \left(\frac{4\overline{\pi}_{1i}}{\omega_{4i}}^{\frac{3+5b}{1+4b}}}{\omega_{4i}} \right)^4 \left(\frac{2^{\frac{1}{2+2b}}}{\underline{\pi}_{1i}}^{\frac{1+2b}{2+2b}}}{\underline{\pi}_{1i}}^4, \left(a_{1i} \left(\frac{4\overline{\pi}_{1i}}{\omega_{4i}}^{\frac{3+5b}{4+4b}}}{\underline{\sigma}_{4i}} \right) \left(\frac{2^{\frac{1}{4}}}{\underline{\pi}_{1i}}^{\frac{3}{4}}}{\underline{\pi}_{1i}}^3 \right) \right)^{\frac{2+2b}{b}} \right\} \times \left(\frac{c}{\sqrt{\underline{m}}} \right)^4$$

$$\ge \max\left\{ \left(\frac{4\overline{\pi}_{1i}}{\omega_{4i}}^{\frac{3+5b}{4+4b}}}{\underline{\sigma}_{4i}} \right)^4 \left(\frac{2^{\frac{1}{2+2b}}}{\underline{\pi}_{1i}}^{\frac{1+2b}{2+2b}}}{\underline{\pi}_{1i}}^4, \left(a_{1i} \left(\frac{4\overline{\pi}_{1i}}{\omega_{4i}}^{\frac{3+5b}{4+4b}}}{\underline{\sigma}_{4i}} \right) \left(\frac{2^{\frac{1}{4}}}{\underline{\pi}_{1i}}^{\frac{3}{4}}}{\underline{\pi}_{1i}}^3 \right) \right)^{\frac{2+2b}{b}} \right\} \times \left| s_i \right|^4$$

Thus, we have from (3 - 5),

$$\dot{V}_{zi} \stackrel{a.e.}{\leq} -\frac{1}{2} \left(\frac{\omega_{4i}}{\overline{\pi}_{1i}} \right) V_{zi}^{\frac{3+5b}{4+4b}} < 0$$

So, \dot{V}_{zi} is a decreasing function of *t* for all i = 1, ..., n and (z_1, z_2) stay in $\Omega_z(c)$.

While, for b = 0, at the boundary,

 $V_{zi} = \rho_{1i}(c), \quad \forall i = 1,...,n, \text{ it implies}$

$$\left(\frac{c}{\sqrt{\underline{m}}}\right) \ge \|s\| \ge |s_i| \Rightarrow \phi_{3i}\left(\frac{c}{\sqrt{\underline{m}}}\right) \supset \phi_{3i}\left(|s_i|\right), \text{ (see the above definition of the region } \phi_{3i}\left(\cdot\right))$$
$$\Rightarrow V_{zi} = \rho_{1i}(c) \ge \rho_{1i}\left(\sqrt{\underline{m}}|s_i|\right), \text{ for } i = 1, \dots, n$$

since $\max_{(z_{1i}, z_{2i}) \in bd \ \phi_{3i}(|s_i|)} V_{zi}(z_{1i}, z_{2i}) = \rho_{1i}(\sqrt{\underline{m}}|s_i|)$ (see the above definition of function $\rho_{1i}(.)$)

Hence, from (3 - 6), it implies $\dot{V}_{zi} \stackrel{a.e.}{<} 0, \forall i = 1,...,n$.

So, \dot{V}_{zi} is a decreasing function of *t* for all i = 1, ..., n and (z_1, z_2) stay in $\Omega_z(c)$ for the case of b = 0.

As a result, the set $\Omega(c)$ is positively invariant, i.e. the trajectories of (3 - 4) stay in it once they have entered it.

- 2. Now we have shown boundedness, next is to show convergence to a smaller compact set $\Omega(\mu)$ with $c > \mu > 0$.
 - (a) For the *s*-subsystem, for a = 0, from (3 7)

$$\dot{\tilde{V}}_{s} \leq -\min_{i}(k_{i} - \gamma_{1i}(c)) \|s\|$$

Hence, for the case of case of a = 0, $\forall i = 1, ..., n$, if $k_i > \gamma_{1i}(c)$, $\tilde{V}_s < 0$ for $s \neq 0$, which implies that the *s* trajectory will converge to zero in finite time and stay there for all future times, i.e. all $(z_1, z_2, s) \in \Omega_z(c) \times \Omega_s(c)$ will enter the set $\Omega_z(c) \times \Omega_s(0)$ in finite time.
While, for the case of or a > 0, observed that $(z_1, z_2, s) \in \Omega(c) \setminus \Omega(\mu)$ implies that

$$\|s\| \ge \frac{\mu}{\sqrt{m}}, \text{ hence from } (3 - 8)$$

$$\dot{V}_{s}^{a.e.} \le -\frac{\lambda_{\min}(K)}{n^{a}} \|s\|^{1+a} + \lambda_{\max}(\gamma_{1i}(c))\sqrt{n}\|s\|$$

$$= -\|s\| \left(\frac{\lambda_{\min}(K)}{n^{a}} \|s\|^{a} - \lambda_{\max}(\gamma_{1i}(c))\sqrt{n}\right)$$

$$\therefore \dot{V}_{s}^{a.e.} \le -c_{3} \qquad (3 - 10)$$

where
$$c_3 = \frac{\mu}{\sqrt{m}} \left[\frac{\lambda_{\min}(K)}{n^a} \left| \frac{\mu}{\sqrt{m}} \right|^a - \lambda_{\max}(\gamma_1(c))\sqrt{n} \right]$$
 (3 - 11)

Thus, for the case of a > 0, with $\lambda_{\min}(K)$ sufficiently big, c_3 is positive, such that, \dot{V}_s < 0 almost everywhere. Then, V_s is a decreasing function of t, so s enters the set $\Omega_s(\mu)$ in finite time and stays in it for all future times, in particular $||s|| \le \frac{\mu}{\sqrt{m}}$ together with the fact that $|s_i| \le ||s||$, for i = 1, ..., n, we have $|s_i| \le \frac{\mu}{\sqrt{m}}$, for i = 1, ..., n.

(b) For the *z*-subsystem, for all $b \ge 0$, if a = 0, once the *s* reaches zero (inside the set $\Omega_z(c) \times \Omega_s(0)$), the *z*-subsystem becomes:

$$\dot{z}_1 = z_2,$$

 $\dot{z}_2 = -K_2 \operatorname{sig}(z_2)^{\frac{2b}{1+b}} - K_1 \operatorname{sig}(z_1)^b,$

which is a multi-dimensional version of the planar homogeneous twisting-based control proposed in Chapter 2 (see section 2.2.1). Its convergence properties is based

on the *b* parameters, where for $0 \le b < 1$ finite time convergence can be achieved, for b = 1 will provide exponential convergence, while for b > 1 the system exhibits asymptotical convergence.

While, for $b \ge 0$, if a > 0, following the same arguments presented in point (1) - (b), it is not hard to see that, once s(t) is inside the set $\Omega_s(\mu)$, (z_1, z_2) will enter the set $\Omega_z(\mu)$ in finite time and stay in it for all future times. Hence, the set $\Omega_s(\mu) \times \Omega_z(\mu)$ is positively invariant and attracting for all trajectories of the system (3 - 4) originating inside the set $\Omega_s(c) \times \Omega_z(c)$.

Remark 3.5. Note that the results above show that the control law achieves semiglobal practical stability for the case when a > 0. Its region of attraction can be given as the set $\Omega_s(c) \times \Omega_z(c)$, while its ultimate invariant set given by $\Omega_s(\mu) \times \Omega_z(\mu)$. The semiglobal nature of the control law can be seen where the estimate of region of attraction for each set of initial states can be increased by choosing a sufficiently big gain *K*. While the practical stability nature, pertaining to the set where the solutions converge is stable and may be reduced at will, can be achieved also by tuning the gain *K* (see the conditions on *K* in Theorem 3.1). On the other hand, when a = 0, the control law achieves semiglobal stability in which its convergence is dependent on the parameter *b* (i.e. finite time with $0 \le b < 1$, exponential for b = 1, and asymptotical for b > 1). Similarly, its region of attraction is $\Omega_s(c) \times \Omega_z(c)$.

Remark 3.6. Note that for the special case of a = b = 1, the proposed controller (3 - 2) becomes the well-known PID control, indeed when written in original coordinates

$$\tau = -K_{d}e_{2} - K_{p}e_{1} - K_{i}\int_{t_{0}}^{t}e_{1}(\varsigma)d\varsigma + K_{i}e_{1}(t_{0}),$$

where $K_d = K$, $K_p = KK_2$, and $K_i = KK_1$. Thus, based on the stability analysis above, one can select the gains of the conventional PID control based on the desired error dynamics (K_1 , K_2) and the desired region of attraction with respect to the ultimate bound (K). This is a great simplification to the heuristic PID gain tuning method, where tuning a particular gain will affect the tuning of all the other gains of the system [97]. It is worth mentioning that in [99], a similar gain selection method is presented, but due to their specific formulation, the bounds on the inertia matrix are required to compute the PID gains. The need for the inertia matrix is echoed by TDE and UDE based PID approach as well [90] [88]. On the other hand, our proposed controller is model-free. Another issue in [99], is its stability formulation requires the time derivative of the robot dynamics, in which case is not directly applicable if discontinuity is to be considered in the type of disturbances affecting the system. Additionally, in the work of [103], semiglobal PID control results is achieved which is modelfree without requiring time derivative of the robot dynamics, however, the gains selection only pertains to the desired region of attraction and ultimate bound, the transient performance or the desired error dynamics is not clearly specified with respect to the PID gains. **Theorem 3.2:** In addition to the conditions in Theorem 3.1, consider the special case of regulation problem, where the desired trajectory of the robot dynamics (3 - 1) is a constant value, q_0 , (i.e. $q_d = q_0$, $\dot{q}_d = \ddot{q}_d = 0$), the control (3 - 2) with the parameter *b* restricted to 0 <

 $b \le 1$, and parameter *a* selected as $a = \frac{2b}{1+b}$, while the additive disturbance is upper bounded by $||D|| \le p_1||e_1|| + p_2||\dot{q}|| + p_3||e_1||^2 + p_4||\dot{q}||^2$, i.e. vanishing perturbation. Then, for b = 1, semiglobal exponential regulation is guaranteed, provided that *K* is large enough with respect to initial error conditions. While, for $0 \le b \le 1$, semiglobal finite-time regulation is assured, provided that *K* is large enough with respect to initial error conditions, and the gravity vector at the constant desired position, $G(q_d)$ is zero.

Proof of Theorem 3.2: For this section, the exponent of the control law (3 - 2), *a* is selected as $a = \frac{2b}{1+b}$ for $0 < b \le 1$. Let us define the following variables,

$$\overline{z}_1 = 0, \quad \overline{z}_2 = -\overline{s}, \quad \overline{s} = -\operatorname{sig}(K^{-1}G(q_0))^{\frac{1+\rho}{2b}}, \\ \widetilde{z}_1 = z_1 - \overline{z}_1, \quad \widetilde{z}_2 = z_2 - \overline{z}_2, \quad \widetilde{s} = s - \overline{s},$$

where $\bar{s}_i = -|k_i^{-1}g_i(q_0)|^{\frac{1+b}{2b}}\operatorname{sign}(k_i^{-1}g_i(q_0))$ and $g_i(q_0)$ is the *i* - th element of the vector $G(q_0)$.

Remark 3.7. Note that q_0 is a constant vector, and as a result the vector, \overline{s} is a constant vector as well since the matrix *K* comprises constants as well. Also, \overline{s} is a constant that is defined for stability analyses only, its actual value, which require knowledge of gravity vector, G(q), is

not required in the control law. Also note that *D* comprise vanishing perturbations only, i.e. $p_0 = 0$ (note that this is the general assumption on regulation problem , however, if constant perturbation do exists, the \overline{s} can be redefined to accommodate this extra constant term).

Hence the closed-loop system (3 - 4) could be rewritten as

$$\begin{aligned} \dot{\tilde{z}}_1 &= \tilde{z}_2 + \tilde{s}, \\ \dot{\tilde{z}}_2 &= -K_2 \operatorname{sig}(\tilde{z}_2 + \tilde{s})^{\frac{2b}{1+b}} - K_1 \operatorname{sig}(\tilde{z}_1)^b \\ \dot{\tilde{s}} &= -M^{-1}(\tilde{z}_1 + q_d) K \operatorname{sig}(\tilde{s})^{\frac{2b}{1+b}} + M^{-1}(\tilde{z}_1 + q_d) K \left(\operatorname{sig}(\tilde{s})^{\frac{2b}{1+b}} - \operatorname{sig}(s)^{\frac{2b}{1+b}} + \operatorname{sig}(\bar{s})^{\frac{2b}{1+b}} \right) + \tilde{\Delta}(\cdot) \end{aligned}$$

where

$$\widetilde{\Delta}(\cdot) = -M^{-1}(\widetilde{z}_1 + q_d)C(\widetilde{z}_1 + q_d, \widetilde{z}_2 + \widetilde{s} + \dot{q}_d)(\widetilde{z}_2 + \widetilde{s} + \dot{q}_d) - M^{-1}(\widetilde{z}_1 + q_d)F(\widetilde{z}_2 + \widetilde{s} + \dot{q}_d) - M^{-1}(\widetilde{z}_1 + q_d)G(\widetilde{z}_1 + q_d) - G(q_0) - M^{-1}(\widetilde{z}_1 + q_d)D - \ddot{q}_d + K_2\operatorname{sig}(\widetilde{z}_2 + \widetilde{s})^{\frac{2b}{1+b}} + K_1\operatorname{sig}(\widetilde{z}_1)^b$$

Remark 3.8. Note that the fact of $sig(\bar{s})^{\frac{2b}{1+b}} = -K^{-1}G(q_0)$ has been employed based on the above definition. Also, observe that the \tilde{s} and \tilde{z} subsystem have the same form as the s and z subsystem discussed in Theorem 3.1 (i.e. by replacing (s, z_1, z_2) with $(\tilde{s}, \tilde{z}_1, \tilde{z}_2)$).

Hence using a similar Lyapunov function structure, consider the following Lyapunov function

$$V(\widetilde{s},\widetilde{z}_1,\widetilde{z}_2) = [V_{\widetilde{s}}(\widetilde{s})]^2 + V_{\widetilde{z}}(\widetilde{z}_1,\widetilde{z}_2),$$

with
$$V_{\widetilde{s}}(\widetilde{s}) = \frac{1}{2} \widetilde{s}^{\mathrm{T}} M \widetilde{s}$$
, and $V_{\widetilde{z}}(\widetilde{z}_1, \widetilde{z}_2) = \sum_{i=1}^n V_{\widetilde{z}_i}(\widetilde{z}_{1i}, \widetilde{z}_{2i})$,

where $V_{\tilde{z}i}(\tilde{z}_{1i},\tilde{z}_{2i}) = \frac{k_{1i}^{2}}{(1+b)^{2}} |\tilde{z}_{1i}|^{2+2b} + \frac{1}{4} |\tilde{z}_{2i}|^{4} + r_{\tilde{z}i} |\tilde{z}_{1i}|^{\frac{3+3b}{2}} \operatorname{sign}(\tilde{z}_{1i})\tilde{z}_{2i} + \frac{k_{1i}}{(1+b)} |\tilde{z}_{1i}|^{1+b} |\tilde{z}_{2i}|^{2}$

Note the following properties of the Lyapunov functions:

$$\frac{1}{2}\underline{m}\|\widetilde{s}\|^{2} \leq V_{\widetilde{s}}(\widetilde{s}) \leq \frac{1}{2}\overline{m}\|\widetilde{s}\|^{2} \Leftrightarrow \frac{\sqrt{2}}{\sqrt{\overline{m}}} [V_{\widetilde{s}}(\widetilde{s})]^{\frac{1}{2}} \leq \|\widetilde{s}\| \leq \frac{\sqrt{2}}{\sqrt{\underline{m}}} [V_{\widetilde{s}}(\widetilde{s})]^{\frac{1}{2}}$$

Also, from Appendix A.1,

$$V_{z}(\tilde{z}_{1}, \tilde{z}_{2}) \geq \sum_{i=1}^{n} \underline{\pi}_{1i} \left(\left| \widetilde{z}_{1i} \right|^{2+2b} + \left| \widetilde{z}_{2i} \right|^{4} \right)$$

$$\geq \underline{\pi}_{1} \left(\sum_{i=1}^{n} \left| \widetilde{z}_{1i} \right|^{2+2b} + \sum_{i=1}^{n} \left| \widetilde{z}_{2i} \right|^{4} \right), \text{ where } \underline{\pi}_{1} \coloneqq \min_{i} \left\{ \underline{\pi}_{1i} \right\},$$

$$\geq \underline{\pi}_{1} \left(\frac{1}{n^{b}} \left(\sum_{i=1}^{n} \left(\left| \widetilde{z}_{1i} \right|^{2} \right) \right)^{\frac{2+2b}{2}} + \frac{1}{n} \left(\sum_{i=1}^{n} \left(\left| \widetilde{z}_{2i} \right|^{2} \right) \right)^{\frac{4}{2}} \right), \text{ using lemma 2.2 of chapter 2,}$$

$$= \underline{\pi}_{1} \left(\frac{1}{n^{b}} \left\| \widetilde{z}_{1} \right\|^{2+2b} + \frac{1}{n} \left\| \widetilde{z}_{2} \right\|^{4} \right)$$

$$\geq \frac{\underline{\pi}_{1}}{n} \left(\left\| \widetilde{z}_{1} \right\|^{2+2b} + \left\| \widetilde{z}_{2} \right\|^{4} \right), \text{ since } \min\left\{ \frac{1}{n^{b}}, \frac{1}{n} \right\} = \frac{1}{n} \text{ for } 0 < b \le 1 \text{ and } n \ge 1$$

and the term,

$$\begin{split} \|\widetilde{z}_{1}\|^{b} + \|\widetilde{z}_{2}\|^{\frac{2b}{1+b}} &= \left(\|\widetilde{z}_{1}\|^{2+2b} \right)^{\frac{b}{2+2b}} + \left(\|\widetilde{z}_{2}\|^{4} \right)^{\frac{b}{2+2b}} \\ &\leq n^{\frac{2+b}{2+2b}} \left(\|\widetilde{z}_{1}\|^{2+2b} + \|\widetilde{z}_{2}\|^{4} \right)^{\frac{b}{2+2b}}, \text{ using lemma 2.2 of chapter 2,} \\ &\leq n^{\frac{2+b}{2+2b}} \left(\frac{n}{\underline{\pi}_{i}} V_{\widetilde{z}}(\widetilde{z}_{1}, \widetilde{z}_{2}) \right)^{\frac{b}{2+2b}} \\ &= \left(\frac{n}{\underline{\pi}_{i}^{\frac{b}{2+2b}}} \right) [V_{\widetilde{z}}(\widetilde{z}_{1}, \widetilde{z}_{2})]^{\frac{b}{2+2b}} . \end{split}$$

Consider the time derivative of the Lyapunov function $V_{\tilde{s}}$ along the solutions of the system for the \tilde{s} subsystem:

$$\begin{split} \dot{V}_{\widetilde{s}} \stackrel{a.e.}{\in} \tilde{V}_{\widetilde{s}} \\ &= \bigcap_{\xi \in \partial V_{\widetilde{s}}} \xi^{\mathrm{T}} \left(\begin{matrix} \mathbf{K}[f](\widetilde{s}) \\ 1 \end{matrix} \right) \\ &= \nabla V_{\widetilde{s}}^{\mathrm{T}} \mathbf{K}[f](\widetilde{s}) + \frac{1}{2} \widetilde{s}^{\mathrm{T}} \dot{M}(q) \widetilde{s} \\ &\subset \widetilde{s}^{\mathrm{T}} M \left(\begin{matrix} -M^{-1}(\widetilde{z}_{1} + q_{d}) K \mathrm{sig}(\widetilde{s}) \frac{2b}{1+b} \\ +M^{-1}(\widetilde{z}_{1} + q_{d}) K \left(\mathrm{sig}(\widetilde{s}) \frac{2b}{1+b} - \mathrm{sig}(s) \frac{2b}{1+b} + \mathrm{sig}(\widetilde{s}) \frac{2b}{1+b} \right) + \mathbf{K}[\widetilde{\Delta}(\cdot)] \right) + \frac{1}{2} \widetilde{s}^{\mathrm{T}} \dot{M} \widetilde{s} \qquad (3 - 12) \end{split}$$

Remark 3.9. Note that the following three properties that will simplify (3 - 12):

1. From Theorem 3.1, the states (s, z_1, z_2) will reach and stay inside the compact set, $\Omega_s(\mu) \times \Omega_z(\mu)$ in finite time, and note that $\overline{s}, \overline{z_1}, \overline{z_2}$ are constants, we have, for $0 < b \le 1$, the following upper bounds (Note that upper bound of the perturbation, ||D|| has been expressed in terms of state variables of the closed-loop system):

$$\begin{split} \widetilde{s}^{\mathrm{T}} M(\cdot) \mathbf{K} \Big[\widetilde{\Delta}(\cdot) \Big] + \frac{1}{2} \widetilde{s}^{\mathrm{T}} \dot{M}(\cdot) \widetilde{s} \leq -\underline{f} \| \widetilde{s} \|^{2} + \overline{f} \| \widetilde{s} \| \| \widetilde{z}_{2} \| + \overline{f} \| \widetilde{s} \| \| \dot{q}_{d} \| + \| \widetilde{s} \| C_{m} \| \widetilde{z}_{2} \|^{2} \\ &+ C_{m} \| \widetilde{z}_{2} \| \| \widetilde{s} \|^{2} + C_{m} \| \dot{q}_{d} \| \| \widetilde{s} \|^{2} + \| \widetilde{s} \| C_{m} \| \dot{q}_{d} \|^{2} \\ &+ 2 \| \widetilde{s} \| C_{m} \| \dot{q}_{d} \| \| \widetilde{z}_{2} \| + \| \widetilde{s} \| k_{g} \| \widetilde{z}_{1} \| + \| \widetilde{s} \| k_{g} \| q_{d} - q_{0} \| \\ &+ \| \widetilde{s} \| p_{1} \| \widetilde{z}_{1} \| + \| \widetilde{s} \| p_{2} \| \widetilde{z}_{2} \| + p_{2} \| \widetilde{s} \|^{2} + \| \widetilde{s} \| p_{2} \| \dot{q}_{d} \| \\ &+ \| \widetilde{s} \| p_{3} \| \widetilde{z}_{1} \|^{2} + \| \widetilde{s} \| p_{4} \| \widetilde{z}_{2} \|^{2} + p_{4} \| \widetilde{s} \|^{3} + \| \widetilde{s} \| p_{4} \| \dot{q}_{d} \|^{2} \\ &+ 2p_{4} \| \widetilde{z}_{2} \| \| \widetilde{s} \|^{2} + 2 \| \widetilde{s} \| p_{4} \| \widetilde{z}_{2} \|^{2} \| \dot{q}_{d} \| + 2p_{4} \| \widetilde{s} \|^{2} \| \dot{q}_{d} \| \\ &+ \overline{m} \| \widetilde{s} \| \| \ddot{q}_{d} \| + \overline{m} \overline{k}_{1} \| \widetilde{s} \| n^{\frac{1-b}{2}} \| \widetilde{z}_{1} \|^{b} + \overline{m} \overline{k}_{2} \| \widetilde{s} \| n^{\frac{1-b}{2+2b}} \| \widetilde{z}_{2} \|^{\frac{2b}{1+b}} \\ &+ \overline{m} \overline{k}_{2} \| \widetilde{s} \| n^{\frac{1-b}{2+2b}} \| \widetilde{s} \|^{\frac{2b}{1+b}} \\ &\leq -\underline{f} \| \widetilde{s} \|^{2} + \gamma_{2} (\mu) \| \widetilde{s} \|^{\frac{1+3b}{1+b}} + \gamma_{3} (\mu) \| \widetilde{s} \| \left(\| \widetilde{z}_{1} \|^{b} + \| \widetilde{z}_{2} \|^{\frac{2b}{1+b}} \right) \\ &+ \alpha (\| \eta(t) \|) \gamma_{4} (\mu) \| \widetilde{s} \| \end{split}$$

where

$$\begin{split} \bar{k}_{1} &= \lambda_{\max} \left(K_{1} \right), \quad \bar{k}_{2} = \lambda_{\max} \left(K_{2} \right), \\ \left(p_{2} \| \tilde{s} \|^{2} + 3p_{4} \| \tilde{s} \|^{3} + \overline{m} \bar{k}_{2} n^{\frac{1-b}{2+2b}} \| \tilde{s} \| \| \tilde{s} \|^{\frac{2b}{1+b}} \right) \leq \gamma_{2} (\mu) \| \tilde{s} \|^{\frac{1+3b}{1+b}} \\ \left(\overline{f} \| \tilde{s} \| \| \tilde{z}_{2} \| + \overline{m} \bar{k}_{1} n^{\frac{1-b}{2}} \| \tilde{s} \| \| \tilde{z}_{1} \|^{b} + \overline{m} \bar{k}_{2} n^{\frac{1-b}{2+2b}} \| \tilde{s} \| \| \tilde{z}_{2} \|^{\frac{2b}{1+b}} \\ + k_{g} \| \tilde{s} \| \| \tilde{z}_{1} \| + p_{1} \| \tilde{s} \| \| \tilde{z}_{1} \| + p_{2} \| \tilde{s} \| \| \tilde{z}_{2} \| \\ + 2C_{m} \| \tilde{s} \| \| \tilde{z}_{2} \|^{2} + 3p_{4} \| \tilde{s} \| \| \tilde{z}_{2} \|^{2} + p_{3} \| \tilde{s} \| \| \tilde{z}_{1} \|^{2} + C_{m} \| \tilde{z}_{2} \| \| \tilde{s} \|^{2} \end{split} \leq \gamma_{3} (\mu) \| \tilde{s} \| \left(\| \tilde{z}_{1} \|^{b} + \| \tilde{z}_{2} \|^{\frac{2b}{1+b}} \right) \\ \left(k_{g} \| \tilde{s} \| \| q_{d} - q_{0} \| + C_{m} \| \dot{q}_{d} \| \| \tilde{s} \|^{2} + 3p_{4} \| \tilde{s} \| \| \dot{q}_{d} \|^{2} + p_{2} \| \tilde{s} \| \| \dot{q}_{d} \| \\ + 2C_{m} \| \tilde{s} \| \| \dot{q}_{d} \|^{2} + \overline{f} \| \tilde{s} \| \| \dot{q}_{d} \| + \overline{m} \| \tilde{s} \| \| \ddot{q}_{d} \| \end{split}$$

with $\gamma_2(.)$, $\gamma_3(.)$, $\gamma_4(.)$ are positive functions and $\alpha(.)$ is a class \mathcal{K} function, and the

vector $\eta(t)$ is defined as:

$$\eta^{\mathrm{T}}(t) = \left[(q_{d}(t) - q_{0})^{\mathrm{T}}, \dot{q}_{d}^{\mathrm{T}}(t), \ddot{q}_{d}^{\mathrm{T}}(t) \right] \in \mathbb{R}^{3n}$$

2. Next, let us define the diagonal matrix,

$$\Lambda := \operatorname{diag}\left(\operatorname{sig}\left(\frac{\widetilde{s}_{i}}{\overline{s}_{i}}\right)^{\frac{2b}{1+b}} - \operatorname{sig}\left(\frac{\widetilde{s}_{i}}{\overline{s}_{i}} + 1\right)^{\frac{2b}{1+b}} + 1\right), \text{ and since, } \operatorname{sig}(\overline{s})^{\frac{2b}{1+b}} = -K^{-1}G(q_{0}),$$

then,
$$\sum_{i=1}^{n} \widetilde{s}_{i} k_{i} \left(\left| \widetilde{s}_{i} \right|^{\frac{2b}{1+b}} \operatorname{sign}(\widetilde{s}_{i}) - \left| s_{i} \right|^{\frac{2b}{1+b}} \operatorname{sign}(s_{i}) + \left| \overline{s}_{i} \right|^{\frac{2b}{1+b}} \operatorname{sign}(\overline{s}_{i}) \right) = -\widetilde{s}^{\mathsf{T}} \Lambda G(q_{0}) \quad .$$

3. Next, using lemma 2.2,

$$\sum_{i=1}^{n} \left|\widetilde{s}_{i}\right|^{\frac{1+3b}{1+b}} \geq \frac{1}{n^{\frac{2b}{1+b}}} \left\|\widetilde{s}\right\|^{\frac{1+3b}{1+b}},$$

Substituting the properties of remark 3.9 into (3 - 12), $\dot{V}_{\tilde{s}}$:

$$\begin{split} \dot{\widetilde{V}}_{\widetilde{s}} &\leq -\frac{\underline{k}}{n^{\frac{2b}{1+b}}} \|\widetilde{s}\|^{\frac{1+3b}{1+b}} + \|\widetilde{s}\| \|\Lambda G(q_0)\| - \underline{f} \|\widetilde{s}\|^2 + \gamma_2(\mu) \|\widetilde{s}\|^{\frac{1+3b}{1+b}} \\ &+ \gamma_3(\mu) \|\widetilde{s}\| \left(\|\widetilde{z}_1\|^b + \|\widetilde{z}_2\|^{\frac{2b}{1+b}} \right) + \alpha(\|\eta(t)\|) \gamma_4(\mu) \|\widetilde{s}\| \end{split}$$

where $\underline{k} = \lambda_{\min}(K)$.

Employing the bounds on the Lyapunov functions and lemma 2.2, one obtains,

$$\begin{split} \dot{\widetilde{V}}_{\widetilde{s}} &\leq -\left(\frac{\underline{k}}{n^{1+b}}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}} V_{\widetilde{s}}^{\frac{1+3b}{2+2b}} - \underline{f}\left(\frac{2}{\overline{m}}\right) V_{\widetilde{s}} + \gamma_{2}\left(\mu\right) \left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right)^{\frac{1+3b}{1+b}} V_{\widetilde{s}}^{\frac{1+3b}{2+2b}} \\ &+ \gamma_{3}\left(\mu\right) \left(\frac{\sqrt{2}n}{\sqrt{\underline{m}} \, \underline{\pi}_{i}^{\frac{b}{2+2b}}}\right) V_{\widetilde{s}}^{\frac{1}{2}} V_{\widetilde{z}}^{\frac{b}{2+2b}} + \left[\alpha(\|\eta(t)\|)\gamma_{4}(\mu) + \|\Lambda G(q_{0})\| \left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right) V_{\widetilde{s}}^{\frac{1}{2}} \end{split}$$
(3 - 13)

Next, we compute the time derivative of the Lyapunov function for the \tilde{z} subsystem, recall that:

$$V_{\widetilde{z}}(\widetilde{z}_1,\widetilde{z}_2) = \sum_{i=1}^n V_{\widetilde{z}i}(\widetilde{z}_{1i},\widetilde{z}_{2i})$$

From Appendix A.1, we have

$$\dot{\widetilde{V}}_{\widetilde{z}}(\widetilde{z}_{1},\widetilde{z}_{2}) \leq -d_{0} \sum_{i=1}^{n} V_{\widetilde{z}_{i}}^{\frac{3+5b}{4+4b}} + d_{1} \sum_{i=1}^{n} |\widetilde{s}_{i}| V_{\widetilde{z}_{i}}^{\frac{1+2b}{2+2b}} + d_{2} \sum_{i=1}^{n} |\widetilde{s}_{i}|^{\frac{2b}{1+b}} V_{\widetilde{z}_{i}}^{\frac{3}{4}}$$

where

$$d_{0} = \min_{i} \left(\frac{\omega_{4i}}{\overline{\pi}_{1i}^{\frac{3+5b}{4+4b}}} \right), \qquad d_{1} = \max_{i} \left(\frac{2^{\frac{1}{2+2b}} \omega_{2i}}{\underline{\pi}_{1i}^{\frac{1+2b}{2+2b}}} \right), \quad \text{and} \quad d_{2} = \max_{i} a_{1i} \left(\frac{2^{\frac{1}{4}} \omega_{3i}}{\underline{\pi}_{1i}^{\frac{3}{4}}} \right).$$

Using lemma 2.2,

$$-d_{0}\sum_{i=1}^{n}V_{\widetilde{z}i}^{\frac{3+5b}{4+4b}} \leq -d_{0}\left(\sum_{i=1}^{n}V_{\widetilde{z}i}\right)^{\frac{3+5b}{4+4b}},$$
$$d_{1}\sum_{i=1}^{n}V_{\widetilde{z}i}^{\frac{1+2b}{2+2b}} \leq n^{\frac{1}{2+2b}}d_{1}\left(\sum_{i=1}^{n}V_{\widetilde{z}i}\right)^{\frac{1+2b}{2+2b}},$$

$$d_{2}\sum_{i=1}^{n}V_{\tilde{z}i}^{\frac{3}{4}} \leq n^{\frac{1}{4}}d_{2}\left(\sum_{i=1}^{n}V_{\tilde{z}i}\right)^{\frac{3}{4}}$$

and the fact that $|x_i| \le ||x||, \forall i = 1, ..., n, x \in \mathbb{R}^n$, it follows that

$$\dot{V}_{\tilde{z}}(\tilde{z}_{1},\tilde{z}_{2})^{a.e.} \leq -d_{0}V_{\tilde{z}}^{\frac{3+5b}{4+4b}} + n^{\frac{1}{2+2b}}d_{1}\|\tilde{s}\|V_{\tilde{z}}^{\frac{1+2b}{2+2b}} + n^{\frac{1}{4}}d_{2}\|\tilde{s}\|^{\frac{2b}{1+b}}V_{\tilde{z}}^{\frac{3}{4}}$$

Employing the bounds on the Lyapunov functions and lemma 2.2, one obtains,

$$\dot{V}_{\tilde{z}}(\tilde{z}_{1},\tilde{z}_{2})^{a.e.} \leq -d_{0}V_{\tilde{z}}^{\frac{3+5b}{4+4b}} + n^{\frac{1}{2+2b}}d_{1}\left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right)V_{\tilde{s}}^{\frac{1}{2}}V_{\tilde{z}}^{\frac{1+2b}{2+2b}} + n^{\frac{1}{4}}d_{2}\left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right)^{\frac{2b}{1+b}}V_{\tilde{s}}^{\frac{b}{1+b}}V_{\tilde{z}}^{\frac{3}{4}}$$
(3 - 14)

With the above results, we are now in a position to find the time derivative of the Lyapunov function for the closed-loop system, recall that

$$V(\widetilde{s},\widetilde{z}_1,\widetilde{z}_2) = [V_{\widetilde{s}}(\widetilde{s})]^2 + V_{\widetilde{z}}(\widetilde{z}_1,\widetilde{z}_2)$$

hence

$$\dot{V}(\widetilde{s},\widetilde{z}_1,\widetilde{z}_2) = 2V_{\widetilde{s}}\dot{V}_{\widetilde{s}} + \dot{V}_{\widetilde{z}},$$

Thus, substituting results from (3 - 13) and (3 - 14), after rearrangement,

$$\begin{split} \dot{V} \stackrel{a.e.}{\leq} - \left(\frac{3}{2}\right) \left(\frac{\underline{k}}{\frac{2b}{n^{1+b}}}\right) \left(\frac{\sqrt{2}}{\sqrt{\overline{m}}}\right)^{\frac{1+3b}{1+b}} V_{\tilde{s}}^{\frac{3+5b}{2+2b}} - \underline{f}\left(\frac{4}{\overline{m}}\right) V_{\tilde{s}}^{2} - \left(\frac{3}{4}\right) d_{0} V_{\tilde{z}}^{\frac{3+5b}{4+4b}} + \dot{V}_{1} \\ + 2\left[\alpha(\|\eta(t)\|)\gamma_{4}(\mu) + \|\Lambda G(q_{0})\|\left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right) V_{\tilde{s}}^{\frac{3}{2}} \end{split}$$

where

$$\begin{split} \dot{V_1} &= -\left(\frac{1}{4}\right) \left(\frac{k}{n^{\frac{2b}{1+b}}}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}} V_{\tilde{s}}^{\frac{3+5b}{2+2b}} - \frac{d_0}{4} V_{\tilde{z}}^{\frac{3+5b}{4+4b}} \\ &- \left(\left(\frac{1}{4}\right) \left(\frac{k}{n^{\frac{2b}{1+b}}}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}} - 2\gamma_2(\mu) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}}\right) V_{\tilde{s}}^{\frac{3+5b}{2+2b}} \\ &+ 2\gamma_3(\mu) \left(\frac{\sqrt{2n}}{\sqrt{m} \pi_i^{\frac{b}{2+2b}}}\right) V_{\tilde{s}}^{\frac{3}{2}} V_{\tilde{z}}^{\frac{b}{2+2b}} + n^{\frac{1}{2+2b}} d_1\left(\frac{\sqrt{2}}{\sqrt{m}}\right) V_{\tilde{s}}^{\frac{1}{2}} V_{\tilde{z}}^{\frac{1+2b}{2+2b}} \\ &+ n^{\frac{1}{4}} d_2\left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{2b}{1+b}} V_{\tilde{s}}^{\frac{b}{1+b}} V_{\tilde{z}}^{\frac{3}{4}} \end{split}$$

For \dot{V}_1 , applying lemma 2.1,

$$\begin{split} &-V_{\widetilde{s}}^{\frac{3+5b}{2+2b}} - V_{\widetilde{z}}^{\frac{3+5b}{4+4b}} \leq -\left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{3+5b}} V_{\widetilde{s}}^{\frac{3}{2}} \left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} V_{\widetilde{z}}^{\frac{b}{2+2b}}, \\ &-V_{\widetilde{s}}^{\frac{3+5b}{2+2b}} - V_{\widetilde{z}}^{\frac{3+5b}{4+4b}} \leq -\left(\frac{3+5b}{1+b}\right)^{\frac{1+b}{3+5b}} V_{\widetilde{s}}^{\frac{1}{2}} \left(\frac{3+5b}{2+4b}\right)^{\frac{2+4b}{3+5b}} V_{\widetilde{z}}^{\frac{1+2b}{2+2b}}, \\ &-V_{\widetilde{s}}^{\frac{3+5b}{2+2b}} - V_{\widetilde{z}}^{\frac{3+5b}{4+4b}} \leq -\left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} V_{\widetilde{s}}^{\frac{b}{1+b}} \left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{3+5b}} V_{\widetilde{z}}^{\frac{3}{4}}, \end{split}$$

thus if the following inequalities

$$\underline{k} > \max\left\{\lambda_4 \gamma_3(\mu), \quad \lambda_5 [\gamma_4(\mu)]^{\frac{3+5b}{3+3b}}, \quad \lambda_6, \quad \lambda_7\right\},$$

where

$$\begin{split} \lambda_{4} &= 8n^{\frac{2b}{1+b}} \left(\frac{\sqrt{m}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}}, \\ \lambda_{5} &= 2^{\frac{3+5b}{3+3b}} \left(\frac{36+36b}{3+5b}\right) \left(\frac{24b}{d_{0}(3+5b)}\right)^{\frac{2b}{3+3b}} n^{\frac{2b}{1+b}} \left(\frac{\sqrt{2}n}{\sqrt{\underline{m}} \, \underline{\pi}_{i}^{\frac{b}{2+2b}}}\right)^{\frac{3+5b}{3+3b}} \left(\frac{\sqrt{\overline{m}}}{\sqrt{2}}\right)^{\frac{1+3b}{1+b}}, \\ \lambda_{6} &= \left(\frac{12+12b}{3+5b}\right) \left(\frac{24+48b}{d_{0}(3+5b)}\right)^{\frac{2+4b}{1+b}} n^{\frac{2b}{1+b}} \left(n^{\frac{1}{2+2b}} d_{1} \left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right)^{\frac{3+5b}{1+b}} \left(\frac{\sqrt{\overline{m}}}{\sqrt{2}}\right)^{\frac{1+3b}{1+b}}, \\ \lambda_{7} &= \left(\frac{24b}{3+5b}\right) \left(\frac{36+36b}{d_{0}(3+5b)}\right)^{\frac{3+3b}{2b}} n^{\frac{2b}{1+b}} \left(n^{\frac{1}{4}} d_{2} \left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right)^{\frac{2b}{1+b}}\right)^{\frac{3+5b}{2b}} \left(\frac{\sqrt{\overline{m}}}{\sqrt{2}}\right)^{\frac{1+3b}{1+b}}, \end{split}$$

hold then the function \dot{V}_1 is negative definite. Then, we have,

$$\begin{split} \dot{\mathcal{V}}^{a.e.} &= \left(\frac{3}{2}\right) \left(\frac{k}{n^{\frac{2b}{1+b}}}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}} \mathcal{V}_{\overline{s}}^{\frac{3+5b}{2+2b}} - \underline{f}\left(\frac{4}{\overline{m}}\right) \mathcal{V}_{\overline{s}}^{2} - \left(\frac{3}{4}\right) d_{0} \mathcal{V}_{\overline{s}}^{\frac{3+5b}{4+4b}} \\ &+ 2\left[\alpha(\|\eta(t)\|)\mathcal{Y}_{4}(\mu) + \|\Lambda G(q_{0})\| \left(\frac{\sqrt{2}}{\sqrt{m}}\right) \mathcal{V}_{\overline{s}}^{\frac{3}{2}} \\ &\leq -\pi_{4}\left(\mathcal{V}_{\overline{s}}^{\frac{3+5b}{2+2b}} + \mathcal{V}_{\overline{s}}^{\frac{3+5b}{4+4b}}\right) - \underline{f}\left(\frac{4}{\overline{m}}\right) \mathcal{V}_{\overline{s}}^{2} + 2\left[\alpha(\|\eta(t)\|)\mathcal{Y}_{4}(\mu) + \|\Lambda G(q_{0})\| \left(\frac{\sqrt{2}}{\sqrt{m}}\right) \mathcal{V}_{\overline{s}}^{\frac{3}{2}} \\ &= -\pi_{4}\left(\left(\mathcal{V}_{\overline{s}}^{2}\right)^{\frac{3+5b}{4+4b}} + \mathcal{V}_{\overline{s}}^{\frac{3+5b}{4+4b}}\right) - \underline{f}\left(\frac{4}{\overline{m}}\right) \mathcal{V}_{\overline{s}}^{2} + 2\left[\alpha(\|\eta(t)\|)\mathcal{Y}_{4}(\mu) + \|\Lambda G(q_{0})\| \left(\frac{\sqrt{2}}{\sqrt{m}}\right) \mathcal{V}_{\overline{s}}^{2}\right)^{\frac{3}{4}} \\ &\leq -\pi_{4}\left(\mathcal{V}_{\overline{s}}^{2} + \mathcal{V}_{\overline{s}}\right)^{\frac{3+5b}{4+4b}} + 2\left[\alpha(\|\eta(t)\|)\mathcal{Y}_{4}(\mu) + \|\Lambda G(q_{0})\| \left(\frac{\sqrt{2}}{\sqrt{m}}\right) \mathcal{V}_{\overline{s}}^{2}\right)^{\frac{3}{4}} \\ &= -\pi_{4}\mathcal{V}^{\frac{3+5b}{4+4b}} + 2\left[\alpha(\|\eta(t)\|)\mathcal{Y}_{4}(\mu) + \|\Lambda G(q_{0})\| \left(\frac{\sqrt{2}}{\sqrt{m}}\right) \mathcal{V}_{\overline{s}}^{2}\right)^{\frac{3}{4}} \end{split}$$

$$(3 - 15)$$

where

$$\pi_4 := \min\left\{ \left(\frac{3}{2} \right) \left(\frac{\underline{k}}{n^{\frac{2b}{1+b}}}\right) \left(\frac{\sqrt{2}}{\sqrt{\overline{m}}}\right)^{\frac{1+3b}{1+b}}, \left(\frac{3}{4}\right) d_0, \right\},\$$

Note that lemma 2.2 with $\frac{3+5b}{4+4b} \le 1$ and the bounds of the Lyapunov functions $V = [V_{\tilde{s}}(\tilde{s})]^2 + V_{\tilde{z}}(\tilde{z}_1, \tilde{z}_2) \ge [V_{\tilde{s}}(\tilde{s})]^2$ have been employed in the above inequality.

Consider the case of b = 1, recall that the term from remark 3.9,

$$\Lambda G(q_0) = \operatorname{diag}\left(\operatorname{sig}\left(\frac{\widetilde{s}_i}{\overline{s}_i}\right)^{\frac{2b}{1+b}} - \operatorname{sig}\left(\frac{\widetilde{s}_i}{\overline{s}_i} + 1\right)^{\frac{2b}{1+b}} + 1\right) G(q_0) = \operatorname{diag}(-1+1)G(q_0) = 0,$$

while for the case of 0 < b < 1, consider the case when the final desired position corresponds to the rest position of the manipulator where $G(q_0) = 0$, or the gravitational torque of the manipulator dynamics is absent (i.e. in space where gravity is absent or in a planar horizontal configuration) where G(q) = 0 for $\forall q \in \mathbb{R}^n$, the term $\Lambda G(q_0)$ vanishes to zero. From (3 - 15),

$$\dot{V}(\widetilde{s},\widetilde{z}_{1},\widetilde{z}_{2})^{a.e.} \leq -\pi_{4}V^{\frac{3+5b}{4+4b}} + 2\left[\alpha(\|\eta(t)\|)\gamma_{4}(\mu)\right]\left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right)V^{\frac{3}{4}}$$

Thus, for $||\eta(t)|| \to 0$ as $t \to \infty$ (where the desired trajectory approaches a constant final desired position), then we have $V(\tilde{s}, \tilde{z}_1, \tilde{z}_2) \to 0$ as $t \to \infty$ as well. Obviously, if $||\eta(t)|| \equiv 0$ (regulation problem), when b = 1, the system will converge to the equilibrium exponentially, while for 0 < b < 1, the system achieves finite-time regulation, in which the states reach the origin in finite time with the estimate of reaching time as,

$$T(\widetilde{s}_0, \widetilde{z}_{10}, \widetilde{z}_{20}) \leq \left(\frac{4+4b}{\pi_4(1-b)}\right) \left[V(\widetilde{s}_0, \widetilde{z}_{10}, \widetilde{z}_{20})\right]^{\frac{1-b}{4+4b}}$$

where $(\tilde{s}_0, \tilde{z}_{10}, \tilde{z}_{20})$ are the states of the system when it first enters the region $\Omega_s(\mu) \times \Omega_z(\mu)$.

Remark 3.10. The stability analysis presented above is akin to the conventional sliding mode one, in which the sliding variable is forced to zero or to be made as small as possible followed by the desired error dynamics, i.e. sliding manifold, being perturbed by that value of sliding variable. There is a major difference, in which the conventional sliding mode has a first order sliding manifold, and hence there is a reaching phase. While in our proposed controller, due to its integral nature, it has the same properties of an integral sliding mode, i.e. the ability to eliminate or reducing the effect of the reaching phase. In particular, the initial condition of the integrator can always be selected such that initially s(0) = 0. From the above analysis, having s(0) = 0 implies initially the states are inside $\Omega_s(\mu) \times \Omega_z(c)$ and will remain in it for all future times. Thus, from initial time, the system will behave as per the desired error dynamics under the influence of a bounded perturbations of $||s|| \le \mu/\sqrt{m}$.

3.3 Numerical Simulations

In this section, numerical simulations on a two-link robot manipulator were carried out to illustrate the results discussed in this chapter. The setups for each simulation are described. Discussion and analysis of the results are presented accordingly.

3.3.1 Simulation Setup

1) Simulation 1:

A two-link rigid robot manipulator is adopted in the simulation. The dynamics of robot manipulator (3 - 1) with the following parameter values (the dynamic parameters are from [144]):

$$M(q) = \begin{bmatrix} 3.511 + 0.191\cos(q_2) & 0.072 + 0.096\cos(q_2) \\ 0.072 + 0.096\cos(q_2) & 0.072 \end{bmatrix},$$

$$C(q, \dot{q}) = \begin{bmatrix} -0.096\dot{q}_2 \sin(q_2) & -0.096(\dot{q}_1 + \dot{q}_2)\sin(q_2) \\ 0.096\dot{q}_1 \sin(q_2) & 0 \end{bmatrix},$$

$$G(q) = \begin{bmatrix} 40.888\sin(q_1) + 2.079\sin(q_1 + q_2) \\ 2.079\sin(q_1 + q_2) \end{bmatrix},$$

$$F = \begin{bmatrix} 0.764 & 0 \\ 0 & 0.328 \end{bmatrix},$$

Note that Properties 3.1, 3.2, 3.3 and 3.4 are satisfied. The desired trajectory vector and the additive disturbances Coulomb friction vector was defined as,

$$q_{d}(t) = \begin{bmatrix} \pi + 0.5 \sin(t) \\ 0.5 \cos(t) \end{bmatrix}, \quad D(\dot{q}) = \begin{bmatrix} 0.7 \operatorname{sign}(\dot{q}_{1}) \\ 0.3 \operatorname{sign}(\dot{q}_{2}) \end{bmatrix},$$

The initial conditions were,

$$q(0) = \begin{bmatrix} 0 \\ 0 \end{bmatrix}, \dot{q}(0) = \begin{bmatrix} 0 \\ 0 \end{bmatrix},$$
$$q_d(0) = \begin{bmatrix} \pi \\ 0.5 \end{bmatrix}, \dot{q}_d(0) = \begin{bmatrix} 0.5 \\ 0 \end{bmatrix},$$

The control (3 - 2) gains were selected as follows,

$$K = \begin{bmatrix} 150 & 0 \\ 0 & 15 \end{bmatrix}, K_1 = \begin{bmatrix} 18 & 0 \\ 0 & 18 \end{bmatrix}, K_2 = \begin{bmatrix} 11 & 0 \\ 0 & 11 \end{bmatrix}, b = 0.6$$

where the simulations were performed for each value of $a = \{0.7, 0.8, 0.9\}$ to examine its effect on the maximal position error. The initial conditions of the vector σ were selected as

$$\sigma(0) = -e_2(0) = \dot{q}_d(0) - \dot{q}(0) = \begin{bmatrix} 0.5\\0 \end{bmatrix} - \begin{bmatrix} 0\\0 \end{bmatrix} = \begin{bmatrix} 0.5\\0 \end{bmatrix}$$

The $\sigma(0)$ is selected as such so that the state s(0) is zero initially, i.e. $s(0) = e_2(0) + \sigma(0)$. Hence, by theorem 3.1, the state *s* will stay inside the region $\Omega_s(\mu)$ as per remark 3.10.

2) Simulation 2:

The setup of Simulation 2 is exactly the same as that of Simulation 1. In this simulation, the value of the parameter, a was fixed at a = 0.9, while the simulations were performed for each control gains of K

$$K = \left\{ \begin{bmatrix} 150 & 0 \\ 0 & 15 \end{bmatrix}, \begin{bmatrix} 225 & 0 \\ 0 & 22.5 \end{bmatrix}, \begin{bmatrix} 300 & 0 \\ 0 & 30 \end{bmatrix} \right\},\$$

for examining its effect on the maximal position error.

3.3.2 Results and Discussions

For better visualization of the plots, some figures are shown in two windows; each with different time intervals.

1) Simulation 1:

Figure 3.1 shows the tracking errors under control (3 - 2) with three different values of $a = \{0.7, 0.8, 0.9\}$. Particularly, the smaller the value of a, the smaller the ultimate bound on the position errors. Indeed, recall from the stability analysis (3 - 11), for c_3 to have a positive value,

$$\mu > \sqrt{\overline{m}} \left(\frac{\lambda_{\max}(\gamma_1(c)) n^{a+0.5}}{\lambda_{\min}(K)} \right)^{\frac{1}{a}}.$$

Thus, by satisfying the conditions of Theorem 3.1, the term within the bracket is less than unity, while for 0 < a < 1, the exponent 1/a is greater than unity, which implies the ultimate bound on *s* can be reduced significantly by lowering *a* while maintaining the same control gain *K*. As can be seen in figure 3.2, reducing the value of *a* does not alter the control effort by much but the precision gained are more significant in comparison.



(a) Trajectory, q_1 for time t = [0, 15] s.



(b) Tracking error, e_{11} for time t = [5, 15] s.



(c) Trajectory, q_2 for time t = [0, 15] s.

(d) Tracking error, e_{12} for time t = [5, 15] s.

Figure 3.1 Simulation 1. Tracking errors using control (3 - 2) for three different values of $a = \{0.7, 0.8, 0.9\}.$



Figure 3.2 Simulation 1. Control input of joint 1 and joint 2.



(a) Trajectory, q_1 for time t = [0, 15] s.



(b) Tracking error, e_{11} for time t = [5, 15] s.



Figure 3.3 Simulation 2. Tracking errors using control (3 - 2) with a = 0.9 with three different values of control gains, *K*.



Figure 3.4 Simulation 2. Control input of joint 1 and joint 2.

2) Simulation 2:

Figure 3.3 shows the tracking errors under control (3 - 2) with *a* fixed at the value of 0.9 while changing the gain *K*. In essence, larger value of *K* results in smaller ultimate bound on the position error. Following (3 - 11), the ultimate bound μ indeed is smaller for larger *K* but

not as dramatic as lowering the value a as shown in simulation 1. It is worth mentioning that in figure 3.4, the plots with larger control gain K exhibit larger slope initially. This high rate of change is not desirable as it may excite unmodelled dynamics in the system.

3) Comparative Study:

Comparative results are difficult to obtain because any comparative result can be dangerously biased. Besides, it is difficult to quantitatively compare controllers that are structurally different. Therefore, the comparative discussions presented here will be limited to PID controllers [90], [99], [103], [104], and [105] because the controller proposed here have a PID structure when a = b = 1 (see remark 3.6). Particularly, the discussions will focus on the analysis aspects of the controller, which governs the gains selection through the corresponding stability analysis.

- 1. In [90] the PID control is analysed to be a combination of feedback linearization term plus uncertainty disturbance estimation term. The structure allows injection of desired error dynamics (similar to the proposed controller) and an estimation term to cancel the effect of uncertainty. However, the existence of the estimation term is not shown and it is mentioned that the compensation does not exist for systems that have discontinuous disturbances and uncertainties.
- 2. Similar to [90], in [99] the PID control is analysed into a modelling error estimator and a desired error dynamics function. However, since the analysis requires taking the time derivative of the robot manipulator dynamics, the stability results is not applicable for discontinuous disturbances and uncertainties. In addition, the analysis

showed that it is necessary to have an accurate estimate, \overline{M} of the inertia matrix of the robot manipulator (requires $||I - M^{-1}\overline{M}|| < 1$) to ensure negative definiteness of the time derivative of Lyapunov function there. However, as remarked by the author, even when such condition is violated, stability is still observed in their simulations. Hence, it clearly shows the conservativeness of the results notwithstanding the aforementioned shortcomings.

- 3. In [103] a strict Lyapunov function is given that allows the selection of gains based on region of attraction and ultimate bound. Also, the analysis does not require taking the time derivative of the robot dynamics. However, it lacks the simple structure of desired error dynamics selection and modelling error compensation found in point 1 and 2. Particularly, although the author claimed the analysis is applicable for tracking control, the proof shown for regulation is not directly clear on how it can be extended for the tracking purposes. In fact, the analysis requires the time derivative of the gravity vector $G(q_d)$ in its auxiliary variable, which is zero for regulation problem(since q_d is a constant vector in regulation problem) but for tracking q_d is no longer a constant(it requires the time derivative of gravity vector).
- 4. For regulation, in [104] and [105], the procedure for PID tuning is extracted from the stability analysis which allows gain selection for stability to specified arbitrary domain. Nevertheless, its extension to tracking and the abilities to inject desired error dynamics are unclear.

Note that the stability analyses mentioned above are all for the same PID control structure for the control of robotic manipulators. The present results of this chapter are able to overcome the shortcomings mentioned. Particularly, recall from remark 3.6, for the special case of a = b = 1, the proposed controller (3 - 2) becomes the well-known PID control, indeed when written in original coordinates

$$\tau = -K_{d}e_{2} - K_{p}e_{1} - K_{i}\int_{t_{0}}^{t}e_{1}(\varsigma)d\varsigma + K_{i}e_{1}(t_{0}),$$

where $K_d = K$, $K_p = KK_2$, and $K_i = KK_1$. For the injection of desired error dynamics, it is done through the selection of gain K_1 and K_2 . As per Theorem 3.1, after selecting K_1 and K_2 , control gain K is selected to ensure stability and ultimate boundedness. Since the vector σ can be selected to have an initial value such that s(0) is zero, from the stability analysis, desired error dynamics, under bounded s, will begin without reaching phase. Thus, the gain selections method is similar to that of point 1 and 2; albeit without requiring an estimate of inertia matrix and time derivative of robot dynamics (As such discontinuous disturbances are allowed in the analysis through generalized Lyapunov theorem). Additionally, the results pertains to semiglobal trajectory tracking(regulation is a special case of tracking), in which case the region of attraction and ultimate bound can be ascertained through the strict Lyapunov functions in section 3.2.1. namely the regions $\Omega(c)$ (which can be arbitrarily enlarged by increasing gain K)and $\Omega(\mu)$. Besides, the stability analysis presented here allows the extension of the results to allow non-Lipschitz desired error dynamics through strict Lyapunov functions of Chapter 2, which allows finite time regulation as per Theorem 3.2.

3.4 Summary

In this chapter, the trajectory tracking control of robot manipulator is developed. In particular, semiglobal practical stability is assured where the ultimate bound of the states can be made arbitrarily small and the region of attraction arbitrarily large by tuning a single parameter. Also, the stability analysis permits the disturbances to have discontinuity, i.e. hence controller is robust to disturbances such as Coulomb friction. Of interest is the ability of the proposed controller in generalizing the well-known PID control. From the stability analysis, the PID gains selection is transform into the selection of desired error dynamics and the selection of acceptable precision of error. For the special case of position regulation problem, sufficient conditions on the gains are obtained to ensure either finite-time or exponential convergence of the system towards the regulation point. In addition, due to integral nature of the controller, it is possible for the system to behave as per the desired error dynamics from the onset of control even in the presence of disturbances.

Chapter 4: ROBOT MANIPULATOR CONTROL: OUTPUT FEEDBACK APPROACH

This chapter considers the tracking control design of robot manipulator when joint velocity measurement is not available. Building on previous results in Chapters 2 and 3, an observer inspired from the super-twisting based family of algorithms is proposed to achieve semiglobal practical stability in the presence of unknown robotic model parameters and additive bounded disturbances. By adding a linear velocity observation error correction term into the proposed observer, the observer gains for the non-Lipschitz terms can be reduced without affecting the region of attractions. For the special case of regulation problem, the controller-observer structure is able to achieve finite-time or exponential convergence depending on parameter of the structure.

4.1 Observer Dynamics

Motivated by the results of section 2.4 and so-called "dirty-derivative" filter found in the literature, an observer dynamics is presented here. Essentially, it comprises a linear combination of the super-twisting based algorithm of Chapter 2 with a linear damping term. Its stability analysis that supports the main results of this chapter is described in this section.

4.1.1 System Description

Consider the super-twisting based dynamics,

$$\begin{aligned} \dot{\widetilde{e}}_1 &= -L_1 \operatorname{sig}(\widetilde{e}_1)^p + \widetilde{e}_2 ,\\ \dot{\widetilde{e}}_2 &= -L_2 \operatorname{sig}(\widetilde{e}_1)^{2p-1} - L_3 \widetilde{e}_2 + d \end{aligned}$$

where $\dot{\tilde{e}}_1, \dot{\tilde{e}}_2 \in \mathbb{R}^n$, are the vector state variables, $L_1, L_2, L_3 \in \mathbb{R}^{n \times n}$ are positive definite diagonal matrices, $0.5 \le p \le 1$, and $d \mathbb{R}^n$ is time-varying and/or nonlinear vector of bounded uncertainty and disturbances.

Consider element wise, $\forall i = 1,...,n$,

$$\dot{\widetilde{e}}_{1i} = -l_{1i} \operatorname{sig}(\widetilde{e}_{1i})^p + \widetilde{e}_{2i} ,$$

$$\dot{\widetilde{e}}_{2i} = -l_{2i} \operatorname{sig}(\widetilde{e}_{1i})^{2p-1} - l_{3i} \widetilde{e}_{2i} + d_i$$

$$(4 - 1)$$

with $|d_i| := \sup \{ \varepsilon_i | : \varepsilon_i \in \mathbf{K}[d_i] \}$.

4.1.2 Stability Analysis

Consider the following Lyapunov function:

$$V_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) = \frac{1}{2}\tilde{e}_{2i}^{2} + \frac{1}{2p}\left|\tilde{e}_{1i}\right|^{2p}$$

Note that the above function is locally Lipschitz and regular. It is differentiable everywhere for p > 0.5, and not differentiable on $\tilde{e}_{1i} = 0$ for p = 0.5.

It can be bounded by

$$\min\left\{\frac{1}{2},\frac{1}{2p}\right\}\left(\left|\widetilde{e}_{2i}\right|^{2}+\left|\widetilde{e}_{1i}\right|^{2p}\right)\leq V_{\widetilde{e}i}\leq \max\left\{\frac{1}{2},\frac{1}{2p}\right\}\left(\left|\widetilde{e}_{2i}\right|^{2}+\left|\widetilde{e}_{1i}\right|^{2p}\right),$$

thus, $V_{\tilde{e}i}$ is positive definite and radially unbounded. Since (4 - 1) is a differential equation that has discontinuous right-hand side, i.e. when p = 0.5 and since no continuity assumption is made on d_i , its solutions are understood in the sense of Filippov (see definition 2.1). According to lemma 2.4, the time derivative of non-smooth Lyapunov function exists almost everywhere

$$\frac{d}{dt}V_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \stackrel{a.e.}{\in} \dot{\tilde{V}}_{\tilde{e}i} := \bigcap_{\xi \in \partial V_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i})} \xi^{\mathrm{T}} \mathbf{K} \begin{bmatrix} \dot{\tilde{e}}_{1i} \\ \dot{\tilde{e}}_{2i} \end{bmatrix}$$

Note that for $0.5 , <math>V_{\tilde{e}i}$ is continuously differentiable:

$$\begin{split} \dot{\widetilde{V}}_{\widetilde{e}i} &= \bigcap_{\xi \in \partial V_{\widetilde{e}i}(\widetilde{e}_{1i},\widetilde{e}_{2i})} \xi^{\mathrm{T}} \mathbf{K} \begin{bmatrix} \dot{\widetilde{e}}_{1i} \\ \dot{\widetilde{e}}_{2i} \end{bmatrix} = \nabla V_{\widetilde{e}i}^{\mathrm{T}} K \begin{bmatrix} \dot{\widetilde{e}}_{1i} \\ \dot{\widetilde{e}}_{2i} \end{bmatrix} \\ & \subset -l_{3i} |\widetilde{e}_{2i}|^2 - l_{1i} |\widetilde{e}_{1i}|^{3p-1} + (1 - l_{2i}) |\widetilde{e}_{1i}|^{2p-1} |\widetilde{e}_{2i}| \mathrm{sign}(\widetilde{e}_{1i}\widetilde{e}_{2i}) + \mathbf{K} [d_i] \widetilde{e}_{2i} \end{split}$$

For p = 0.5, $V_{\tilde{e}i}$ is not differentiable on $\tilde{e}_{1i} = 0$:

$$\begin{bmatrix} \dot{\widetilde{e}}_{1i} \\ \dot{\widetilde{e}}_{2i} \end{bmatrix} \in \begin{bmatrix} -l_{1i} \operatorname{sig}(\widetilde{e}_{1i})^{\frac{1}{2}} + \widetilde{e}_{2i} \\ -l_{2i} \mathbf{K}[\operatorname{sign}(\widetilde{e}_{1i})] - l_{3i} \widetilde{e}_{2i} + \mathbf{K}[d_i] \end{bmatrix}, \quad \partial V_{\widetilde{e}i}(\widetilde{e}_{1i}, \widetilde{e}_{2i}) = \begin{bmatrix} \operatorname{SGN}(\widetilde{e}_{1i}) \\ \widetilde{e}_{2i} \end{bmatrix},$$

for $\tilde{e}_{1i} \neq 0$, $\forall \tilde{e}_{2i} \in \mathbb{R}$:

$$\dot{\widetilde{V}}_{\widetilde{e}i} = -l_{3i} |\widetilde{e}_{2i}|^2 - l_{1i} |\widetilde{e}_{1i}|^{3p-1} + (1 - l_{2i}) |\widetilde{e}_{1i}|^{2p-1} |\widetilde{e}_{2i}| \operatorname{sign}(\widetilde{e}_{1i}\widetilde{e}_{2i}) + K[d_i]\widetilde{e}_{2i}|\widetilde{e}_{2i}| \widetilde{e}_{2i}|\widetilde{e}_{2i}| \widetilde{e}_{2i}|\widetilde{e}_{2i}| \widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}| \widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|\widetilde{e}_{2i}|$$

for $\tilde{e}_{1i} = 0, \forall \tilde{e}_{2i} \in \mathbb{R}$:

Let $[\xi_2, \tilde{e}_{2i}]^T$ with $\xi_2 \in [-1, 1]$ be an arbitrary element of $\partial V_{\tilde{e}i}$, then

$$\xi^{\mathrm{T}} \mathbf{K} \begin{bmatrix} \dot{\tilde{e}}_{1i} \\ \dot{\tilde{e}}_{2i} \end{bmatrix} = [\xi_{2}, \tilde{e}_{2i}] \begin{bmatrix} \tilde{e}_{2i} \\ -l_{2i}[-1,1] - l_{3i}\tilde{e}_{2i} + \mathbf{K}[d_{i}] \end{bmatrix} = ([\xi_{2} - l_{2i}, \xi_{2} + l_{2i}])\tilde{e}_{2i} - l_{3i}|\tilde{e}_{2i}|^{2} + \mathbf{K}[d_{i}]\tilde{e}_{2i}$$
implies
$$\dot{\tilde{V}}_{\tilde{e}i} = \bigcap_{\xi_{2} \in [-1,1]} ([\xi_{2} - l_{2i}, \xi_{2} + l_{2i}])\tilde{e}_{2i} - l_{3i}|\tilde{e}_{2i}|^{2} + \mathbf{K}[d_{i}]\tilde{e}_{2i} \leq |1 - l_{2i}||\tilde{e}_{2i}| - l_{3i}|\tilde{e}_{2i}|^{2} + |d_{i}||\tilde{e}_{2i}|$$

$$(0 \quad \text{for } l_{i} = 1)$$

Note that $\bigcap_{\xi_2 \in [-1,1]} ([\xi_2 - l_{2i}, \xi_2 + l_{2i}]) = \begin{cases} 0, & \text{for } l_{2i} = 1, \\ [1 - l_{2i}, l_{2i} - 1], & \text{for } l_{2i} > 1, \\ \emptyset, & \text{for } 0 < l_{2i} < 1 \end{cases}$

where the convention for the empty set of max $\dot{\tilde{V}} = -\infty$, if $\dot{\tilde{V}} = \emptyset$ is employed.(see [74])

Thus, for all $(\tilde{e}_{1i}, \tilde{e}_{2i}) \in \mathbb{R}^n$ and $\forall 0.5 \le p \le 1$, we have

$$\dot{V}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i})^{a.e.} \le -l_{3i}|\tilde{e}_{2i}|^2 - l_{1i}|\tilde{e}_{1i}|^{3p-1} + |1 - l_{2i}||\tilde{e}_{1i}|^{2p-1}|\tilde{e}_{2i}| + |d_i||\tilde{e}_{2i}|$$
(4 - 2)

For the ease of analysis, the state space is divided into the following three regions:

$$\begin{split} \psi_1(|\zeta_i|, l_{3i}) &= \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : \widetilde{e}_{1i} \in \mathbb{R}, |\widetilde{e}_{2i}| \ge \beta_2(|\zeta_i|, l_{3i}) \}, \\ \psi_2(|\zeta_i|, l_{3i}) &= \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \ge \beta_1(|\zeta_i|, l_{3i}), |\widetilde{e}_{2i}| \le \beta_2(|\zeta_i|, l_{3i}) \}, \\ \psi_3(|\zeta_i|, l_{3i}) &= \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \le \beta_1(|\zeta_i|, l_{3i}), |\widetilde{e}_{2i}| \le \beta_2(|\zeta_i|, l_{3i}) \}, \end{split}$$

where $|\zeta_i| := \max\{|1 - l_{2i}|, |d_i|\}$ and the functions β_1 and β_2 are of class- \mathcal{KL} (see [145])

defined by:

$$\beta_1(|\zeta_i|, l_{3i}) \coloneqq \max\left\{ \begin{pmatrix} \frac{3\beta_2(|\zeta_i|, l_{3i})|\zeta_i|}{l_{1i}} \end{pmatrix}^{\frac{1}{p}}, \\ \begin{pmatrix} \frac{3\beta_2(|\zeta_i|, l_{3i})|\zeta_i|}{l_{1i}} \end{pmatrix}^{\frac{1}{3p-1}} \\ \end{pmatrix}, \end{cases} \right\}$$

$$\beta_{2}(|\zeta_{i}|, l_{3i}) := \begin{cases} \frac{3|\zeta_{i}|}{l_{3i}}, & \text{for } p = 1, \\ \max\left\{\frac{3|\zeta_{i}|}{l_{3i}}, \lambda_{1} \frac{|\zeta_{i}|^{\frac{3p-1}{1-p}}}{l_{1i}^{\frac{2p-1}{1-p}} l_{3i}^{\frac{p}{1-p}}}\right\}, & \text{for } 0.5$$

with
$$l_{1i}l_{3i} > \left(\frac{4p-2}{3p-1}\right)\left(\frac{3p}{3p-1}\right)\left|1-l_{2i}\right|^2$$
, for $p=1$, and $\lambda_1 = \left(\frac{4p-2}{3p-1}\right)^{\frac{2p-1}{1-p}}\left(\frac{3p}{3p-1}\right)^{\frac{p}{1-p}}$

In the region: $\psi_1(\zeta_i|, l_{3i}) = \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : \widetilde{e}_{1i} \in \mathbb{R}, |\widetilde{e}_{2i}| \ge \beta_2(|\zeta_i|, l_{3i}) \}$

Applying lemma 2.1 (for 0.5):

$$-\frac{l_{1i}}{2}\left|\widetilde{e}_{1i}\right|^{3p-1} - \frac{l_{3i}}{3}\left|\widetilde{e}_{2i}\right|^{2} \le -\left(\left(\frac{3p-1}{2p-1}\right)\frac{l_{1i}}{2}\right)^{\frac{2p-1}{3p-1}}\left|\widetilde{e}_{1i}\right|^{2p-1}\left(\left(\frac{3p-1}{p}\right)\frac{l_{3i}}{3}\right)^{\frac{p}{3p-1}}\left|\widetilde{e}_{2i}\right|^{\frac{2p}{3p-1}}\right)$$

From (4 - 2), we have

$$\begin{split} \dot{V}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) &\leq -\frac{l_{1i}}{2} |\tilde{e}_{1i}|^{3p-1} - \frac{l_{3i}}{3} |\tilde{e}_{2i}|^2 - |\tilde{e}_{2i} \left(\frac{l_{3i}}{3} |\tilde{e}_{2i}| - |d_i| \right) \\ &- |\tilde{e}_{1i}|^{2p-1} |\tilde{e}_{2i} \left(\left(\frac{3p-1}{2p-1} \right) \frac{l_{1i}}{2} \right)^{\frac{2p-1}{3p-1}} \left(\left(\frac{3p-1}{p} \right) \frac{l_{3i}}{3} \right)^{\frac{p}{3p-1}} |\tilde{e}_{2i}|^{\frac{1-p}{3p-1}} - |1 - l_{2i}| \right) \end{split}$$

While for p = 0.5, from (4 - 2)

$$\dot{V}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \stackrel{a.e.}{\leq} -\frac{l_{3i}}{2} |\tilde{e}_{2i}|^2 - l_{1i} |\tilde{e}_{1i}|^{0.5} - |\tilde{e}_{2i}| \left(\frac{l_{3i}}{2} |\tilde{e}_{2i}| - (|1 - l_{2i}| + |d_i|)\right)$$

If the following inequalities hold, for different cases of *p*:

for *p* = 1:

$$|\tilde{e}_{2i}| \ge \frac{3|d_i|}{l_{3i}}$$
, with $l_{1i}l_{3i} \ge \left(\frac{4p-2}{3p-1}\right)\left(\frac{3p}{3p-1}\right)\left(1-l_{2i}\right)^2$, for $p=1$,

or

for 0.5 < *p* < 1:

$$\left|\widetilde{e}_{2i}\right| \ge \max\left\{\frac{3|d_i|}{l_{3i}}, \lambda_1 \frac{\left|1 - l_{2i}\right|^{\frac{3p-1}{1-p}}}{l_{1i}^{\frac{2p-1}{1-p}}l_{3i}^{\frac{p}{1-p}}}\right\} \text{ where } \lambda_1 = \left(\frac{4p-2}{3p-1}\right)^{\frac{2p-1}{1-p}} \left(\frac{3p}{3p-1}\right)^{\frac{p}{1-p}}$$

or for p = 0.5:

$$|\tilde{e}_{2i}| \ge \frac{2(|1 - l_{2i}| + |d_i|)}{l_{3i}}$$

then $\dot{V}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) < 0$, which is sufficiently satisfied by the states in this region.

Next, consider the region: $\psi_2(\zeta_i|, l_{3i}) = \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \ge \beta_1(\zeta_i|, l_{3i}), |\widetilde{e}_{2i}| \le \beta_2(\zeta_i|, l_{3i}) \}$ Thus, from (4 - 2),

$$\begin{split} \dot{V}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i})^{a.e.} &= l_{3i}|\tilde{e}_{2i}|^{2} - l_{1i}|\tilde{e}_{1i}|^{3p-1} + |1 - l_{2i}||\tilde{e}_{1i}|^{2p-1}\beta_{2}(|\zeta_{i}|,l_{3i}) + |d_{i}|\beta_{2}(|\zeta_{i}|,l_{3i}) \\ &= -l_{3i}|\tilde{e}_{2i}|^{2} - \frac{l_{1i}}{3}|\tilde{e}_{1i}|^{3p-1} - |\tilde{e}_{1i}|^{2p-1}\left(\frac{l_{1i}}{3}|\tilde{e}_{1i}|^{p} - |1 - l_{2i}|\beta_{2}(|\zeta_{i}|,l_{3i})\right) \\ &- \left(\frac{l_{1i}}{3}|\tilde{e}_{1i}|^{3p-1} - |d_{i}|\beta_{2}(|\zeta_{i}|,l_{3i})\right) \\ &< 0 \end{split}$$

if
$$|\widetilde{e}_{1i}| \ge \max\left\{ \left(\frac{3\beta_2 (|\zeta_i|, l_{3i})|1 - l_{2i}|}{l_{1i}} \right)^{\frac{1}{p}}, \left(\frac{3\beta_2 (|\zeta_i|, l_{3i})|d_i|}{l_{1i}} \right)^{\frac{1}{3p-1}} \right\}$$

which is sufficiently satisfied for the states in this region.

Next consider the compact set:

$$\psi_{3}(|\zeta_{i}|, l_{3i}) = \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^{2} : |\widetilde{e}_{1i}| \leq \beta_{1}(|\zeta_{i}|, l_{3i}), |\widetilde{e}_{2i}| \leq \beta_{2}(|\zeta_{i}|, l_{3i}) \}$$

Note that, $\dot{V}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \stackrel{a.e.}{\leq} 0$ for $(\tilde{e}_{1i},\tilde{e}_{2i}) \notin \psi_3(|\zeta_i|,l_{3i})$.

Next, define a Lyapunov level set $\sum_{\tilde{e}i} (\zeta_i |, I_{3i}) = \{ (\tilde{e}_{1i}, \tilde{e}_{2i}) \in \mathbb{R}^2 : V_{\tilde{e}i} \leq \rho_{3i} (|\zeta_i|, I_{3i}) \}$, where the class \mathcal{KL} function ρ_{3i} is defined as,

$$\begin{split} \rho_{3i}(|\zeta_{i}|, l_{3i}) &= \max_{(\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in bd \ \psi_{3}(|\zeta_{i}|, l_{3i})} V_{\widetilde{e}i}(\widetilde{e}_{1i}, \widetilde{e}_{2i}) \\ &= \max_{(\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in bd \ \psi_{3}(|\zeta_{i}|, l_{3i})} \left(\frac{1}{2} \widetilde{e}_{2i}^{2} + \frac{1}{2p} |\widetilde{e}_{1i}|^{2p}\right) \\ &= \left(\frac{1}{2} [\beta_{2}(|\zeta_{i}|, l_{3i})]^{2} + \frac{1}{2p} [\beta_{1}(|\zeta_{i}|, l_{3i})]^{2p}\right) \end{split}$$

which exists since the boundary of the set is compact and $V_{\tilde{e}i}$ is continuous. Then we observe that $\psi_3(|\zeta_i|, l_{3i}) \subset \Sigma_{\tilde{e}i}(|\zeta_i|, l_{3i})$. As a result, we have

$$\frac{d}{dt}V_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \stackrel{a.e.}{\in} \dot{\widetilde{V}}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) < 0 \quad for \quad V_{\tilde{e}i} \ge \rho_{3i}(|\zeta_i|,l_{3i})$$

which implies that each of the trajectories for the *i*-th planar system will enter their respective compact level set $\sum_{\tilde{e}i} (|\zeta_i|, l_{3i})$ in finite time and stay in it once entered.

Remark 4.1. Note that for any given l_{1i} , $l_{2i} > 0$, and a bounded $|\zeta_i|$, ρ_{3i} ($|\zeta_i|$, l_{3i}) can be made arbitrarily small by increasing $l_{3i} > 0$. Hence, it can be observed that for $0.5 \le p \le 1$, the increase of observer gain L_3 will result in a smaller upper bound on the observation errors.

Remark 4.2. The finite time property of the observer is not shown here yet. The homogeneity of $0.5 \le p < 1$ will be utilised for showing the finite time capability of the observer in the following sections of this chapter, as the controller is developed.

4.2 Manipulator Dynamics

Recall the dynamics of an *n*-joint serial rigid robotic manipulators as in (3 - 1) of Chapter 3

$$M(q)\ddot{q} + C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D(q,\dot{q},t) = \tau$$
(4-3)

where $q \in \mathbb{R}^n$ is the vector of generalized joint coordinates, $M(q) \in \mathbb{R}^{n \times n}$ is the inertia matrix, and $C(q, \dot{q}) \dot{q}$, $F\dot{q}$, G(q), $D(q, \dot{q}, t)$, $\tau \in \mathbb{R}^n$ are, respectively, the vectors of Coriolis and centrifugal, viscous friction, gravity, disturbances, and input generalized forces, with *F* being a constant, positive definite, diagonal (viscous friction coefficient) matrix and $D(q, \dot{q}, t)$ being a locally bounded disturbances. The robot manipulator satisfies the same properties as that in Chapter 3 and will not be repeated here. Also, similar assumptions on the bound of $D(q, \dot{q}, t)$ and twice differentiability of the desired trajectory $q_d(t)$ apply here. The only exception is that only joint positions measurement is available. No velocity information from the system is assumed in this chapter. As such, the following development will follow an output feedback approach.

4.2.1 Control Development

The proposed controller has the following form,

$$\tau = -K \operatorname{sig}(\hat{s})^a, \qquad (4-4)$$

where *K* is a positive definite diagonal matrix, i.e. $K = \text{diag}\{k_i\}_{i=1}^n$, with $k_i > 0, \forall i = 1, ..., n, a$ ≥ 0 constant, and $\hat{s} \in \mathbb{R}^n$ is the velocity-estimate-based desired error dynamics defined as $\hat{s} = \hat{e}_2 + \sigma$, with

$$\dot{\sigma} = K_2 \operatorname{sig}(\hat{e}_2)^{\frac{2b}{1+b}} + K_1 \operatorname{sig}(e_1)^b$$
 (4 - 5)

where K_1 and K_2 are positive definite diagonal matrices, i.e. $K_1 = \text{diag}\{k_{1i}\}_{i=1}^n$, with $k_{1i} > 0$, $K_2 = \text{diag}\{k_{2i}\}_{i=1}^n$, with $k_{2i} > 0 \quad \forall i = 1, ..., n, b \ge 0$ constant, and $e_1 := q - q_d \in \mathbb{R}^n$, $\hat{e}_2 \in \mathbb{R}^n$ is the output of the observer defined as,

$$\dot{\hat{e}}_{1} = -L_{1} \operatorname{sig}(\tilde{e}_{1})^{p} + \hat{e}_{2} ,$$

$$\dot{w} = -L_{2} \operatorname{sig}(\tilde{e}_{1})^{2p-1} - L_{3}\hat{e}_{2}$$

$$\hat{e}_{2} = w + L_{3}e_{1}$$
(4 - 6)

where $\hat{e}_1, \hat{e}_2 \in \mathbb{R}^n$, L_1, L_2 , and L_3 are positive definite diagonal matrices, i.e. $L_1 = \text{diag}\{l_{1i}\}_{i=1}^n$, with $l_{1i} > 0$, $L_2 = \text{diag}\{l_{2i}\}_{i=1}^n$, with $l_{2i} > 0$, and $L_3 = \text{diag}\{l_{3i}\}_{i=1}^n$, with $l_{3i} > 0$, $\forall i = 1, ..., n$, and $0.5 \le p \le 1$. Let us define $e_2 = \dot{q} - \dot{q}_d \in \mathbb{R}^n$, $\tilde{e}_1 = \hat{e}_1 - e_1 \in \mathbb{R}^n$, and $\tilde{e}_2 = \hat{e}_2 - e_2 \in \mathbb{R}^n$, then the closed-loop system of (4 - 3), (4 - 4), (4 - 5) and (4 - 6) can be written as

$$\begin{split} \dot{\sigma} &= K_2 \operatorname{sig}(\hat{e}_2)^{\frac{2b}{1+b}} + K_1 \operatorname{sig}(e_1)^b \\ \dot{e}_1 &= e_2 \\ \dot{e}_2 &= M^{-1}(q)\tau - M^{-1}(q)(C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D) - \ddot{q}_d \\ \dot{\tilde{e}}_1 &= -L_1 \operatorname{sig}(\widetilde{e}_1)^p + \widetilde{e}_2 \quad , \\ \dot{\tilde{e}}_2 &= -L_2 \operatorname{sig}(\widetilde{e}_1)^{2p-1} - L_3 \widetilde{e}_2 - (M^{-1}(q)\tau - M^{-1}(q)(C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D) - \ddot{q}_d) \end{split}$$

To rewrite the closed-loop system in a form convenient for analysis, let us define $z_1 = e_1$, $z_2 = -\sigma$, with $s = e_2 + \sigma = \hat{s} - (\hat{e}_2 - e_2) = \hat{s} - \tilde{e}_2$, and we obtain the following form of closedloop system,
$$\begin{aligned} \dot{z}_{1} &= z_{2} + s, \\ \dot{z}_{2} &= -K_{2} \operatorname{sig}(z_{2})^{\frac{2b}{1+b}} - K_{1} \operatorname{sig}(z_{1})^{b} - K_{2} \left(\operatorname{sig}(z_{2} + s + \widetilde{e}_{2})^{\frac{2b}{1+b}} - \operatorname{sig}(z_{2})^{\frac{2b}{1+b}} \right) \\ \dot{s} &= -M^{-1} (z_{1} + q_{d}) K \operatorname{sig}(s)^{a} + \Delta_{1} (\cdot) - M^{-1} (z_{1} + q_{d}) K \left(\operatorname{sig}(s + \widetilde{e}_{2})^{a} - \operatorname{sig}(s)^{a} \right) \\ &+ K_{2} \left(\operatorname{sig}(z_{2} + s + \widetilde{e}_{2})^{\frac{2b}{1+b}} - \operatorname{sig}(z_{2} + s)^{\frac{2b}{1+b}} \right) \\ \dot{\tilde{e}}_{1} &= -L_{1} \operatorname{sig}(\widetilde{e}_{1})^{p} + \widetilde{e}_{2} \quad , \\ \dot{\tilde{e}}_{2} &= -L_{2} \operatorname{sig}(\widetilde{e}_{1})^{2p-1} - L_{3} \widetilde{e}_{2} + \Delta_{2} (\cdot) \end{aligned}$$

$$(4 - 7)$$

where

$$\Delta_{1}(\cdot) = -M^{-1}(z_{1}+q_{d})C(z_{1}+q_{d},z_{2}+s+\dot{q}_{d})(z_{2}+s+\dot{q}_{d}) - M^{-1}(z_{1}+q_{d})F(z_{2}+s+\dot{q}_{d}) -M^{-1}(z_{1}+q_{d})G(z_{1}+q_{d}) - M^{-1}(z_{1}+q_{d})D - \ddot{q}_{d} + K_{2}\mathrm{sig}(z_{2}+s)^{\frac{2b}{1+b}} + K_{1}\mathrm{sig}(z_{1})^{b},$$

$$\Delta_{2}(\cdot) = M^{-1}(z_{1} + q_{d})K(\operatorname{sig}(s + \tilde{e}_{2})^{a}) + M^{-1}(z_{1} + q_{d})C(z_{1} + q_{d}, z_{2} + s + \dot{q}_{d})(z_{2} + s + \dot{q}_{d}) + M^{-1}(z_{1} + q_{d})F(z_{2} + s + \dot{q}_{d}) + M^{-1}(z_{1} + q_{d})G(z_{1} + q_{d}) + M^{-1}(z_{1} + q_{d})D + \ddot{q}_{d},$$

4.2.2 Stability Analysis

From the closed-loop dynamics (4 - 7), the structure (z_1, z_2) is essentially the desired error dynamics injected by the controller through (4 - 5). Essentially, for any $i \in 1, ..., n$, when $s_i = \tilde{e}_{2i} = 0$, the dynamics of the subsystem (z_{1i}, z_{2i}) is identical to that of the twisting-based family of algorithm studied in section 2.2 of Chapter 2. In other words, s_i and \tilde{e}_{2i} can be viewed as a perturbations on the (z_{1i}, z_{2i}) dynamics. While in the *s*-dynamics, its structure is akin to that of sliding mode control with \tilde{e}_{2i} as perturbations. Since the differential equations (4 - 7) have discontinuous right-hand side, i.e. when a = 0 or b = 0 or p = 0.5, and since no continuity assumption is made on *D*, its solutions are understood in the sense of Filippov (see definition 2.1).

The following Lyapunov functions will be used in the analysis:

$$V_{zi}(z_{1i}, z_{2i}) = \frac{k_{1i}^{2}}{(1+b)^{2}} |z_{1i}|^{2+2b} + \frac{1}{4} |z_{2i}|^{4} + r_{zi} |z_{1i}|^{\frac{3+3b}{2}} \operatorname{sign}(z_{1i}) z_{2i}$$
$$+ \frac{k_{1i}}{(1+b)} |z_{1i}|^{1+b} |z_{2i}|^{2}, \qquad \text{for } i = 1, \dots, n,$$

$$V_s = \frac{1}{2} s^T M(q) s,$$

$$V_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) = \frac{1}{2}\tilde{e}_{2i}^{2} + \frac{1}{2p}|\tilde{e}_{1i}|^{2p}$$
, for $i = 1,...,n$.

Note that the Lyapunov function for the (z_{1i}, z_2) -subsystem is a strict Lyapunov function proposed for the twisting-based family of algorithms in Chapter 2, where $\forall q \in \mathbb{R}^n$:

$$\underline{\pi}_{1i}\left(|z_{1i}|^{2+2b}|z_{2i}|^{4}\right) \leq V_{zi}\left(z_{1i}, z_{2i}\right) \leq \overline{\pi}_{1i}\left(|z_{1i}|^{2+2b}+|z_{2i}|^{4}\right),$$

where

$$\underline{\pi}_{1i} := \min\left\{\frac{1}{8}, \frac{1}{2}\left(\frac{k_{1i}}{1+b}\right)^2\right\}, \ \overline{\pi}_{1i} := \max\left\{\frac{3r_{zi}}{4} + \left(\frac{k_{1i}}{1+b}\right)^2 + \left(\frac{k_{1i}}{2+2b}\right), \ \left(\frac{k_{1i}}{2+2b}\right) + \frac{r_{zi}}{4} + \frac{1}{4}\right\},$$

while for V_s ,

$$\frac{1}{2}\underline{m}\|s\|^2 \le V_s \le \frac{1}{2}\overline{m}\|s\|^2,$$

and for $V_{\tilde{e}i}$

$$\min\left\{\frac{1}{2},\frac{1}{2p}\right\}\left(\left|\widetilde{e}_{2i}\right|^{2}+\left|\widetilde{e}_{1i}\right|^{2p}\right)\leq V_{\widetilde{e}i}\leq \max\left\{\frac{1}{2},\frac{1}{2p}\right\}\left(\left|\widetilde{e}_{2i}\right|^{2}+\left|\widetilde{e}_{1i}\right|^{2p}\right)$$

Let us define the following sets:

$$\Omega_{z}(R) = \left\{ (z_{1}, z_{2}) \in \mathbb{R}^{2n} : V_{zi}(z_{1i}, z_{2i}) \leq \rho_{1i}(R), \dots, V_{zn}(z_{1n}, z_{2n}) \leq \rho_{1i}(R), \text{ for } i = 1, \dots, n \right\},$$

$$\Omega_{s}(R) = \left\{ s \in \mathbb{R}^{n} : V_{s} \leq \rho_{2} \frac{R^{2}}{2} \right\},$$

$$\Omega_{\tilde{e}}(R) = \left\{ (\tilde{e}_{1}, \tilde{e}_{2}) \in \mathbb{R}^{2n} : V_{\tilde{e}i} \leq \frac{R^{2}}{2}, \dots, V_{\tilde{e}n} \leq \frac{R^{2}}{2}, \text{ for } i = 1, \dots, n \right\},$$

$$\Omega(R) = \left\{ (z_{1}, z_{2}, s, \tilde{e}_{1}, \tilde{e}_{2}) \in \mathbb{R}^{5n} : \Omega_{z}(R), \ \Omega_{s}(R), \ \Omega_{\tilde{e}}(R) \right\}$$

where

$$\begin{aligned} & \text{for } i = 1, \dots, n \\ & \rho_{1i}(R) = \begin{cases} \left(\rho_2 \right)^2 \left(\frac{4\overline{\pi}_{1i} \frac{3+5b}{a+4b}}{\omega_{4i}} \right)^4 \left(\frac{2^{\frac{1}{2+2b}} \omega_{2i}}{\underline{\pi}_{1i} \frac{1+2b}{2+2b}} \right)^4, \\ & \left(\sqrt{\rho_2} + \sqrt{n \underline{m}} \right)^4 \left(a_{1i} \left(\frac{4\overline{\pi}_{1i} \frac{3+5b}{a+4b}}{\omega_{4i}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{3i}}{\underline{\pi}_{1i} \frac{3}{4}} \right) \right)^{\frac{2+2b}{b}} \right\} \times \left(\frac{R}{\sqrt{\underline{m}}} \right)^4, \quad \text{for } b > 0, \\ & \left(\sqrt{\rho_2} + \sqrt{n \underline{m}} \right)^4 \left(a_{1i} \left(\frac{4\overline{\pi}_{1i} \frac{3+5b}{a+4b}}{\omega_{4i}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{3i}}{\underline{\pi}_{1i} \frac{3}{4}} \right) \right)^{\frac{2+2b}{b}} \right\} \times \left(\frac{R}{\sqrt{\underline{m}}} \right)^4, \quad \text{for } b > 0, \\ & \left(\max_{(z_{1i}, z_{2i}) \in bd \ \phi_{3i} \left(\left(\frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} + \sqrt{n} \right) R \right) V_{zi}(z_{1i}, z_{2i}), \end{cases} \quad \text{for } b = 0, \end{aligned}$$

$$\rho_{2} = \begin{cases} \overline{m} \left(\frac{2a_{4}}{(1-a_{5})} \frac{\lambda_{\max}(K)}{\lambda_{\min}(K)} n^{\frac{3a+1}{2}} \right)^{\frac{2}{a}}, & \text{for } a > 0, \\ \\ \overline{m} \left(4n \frac{\lambda_{\max}(K)}{\lambda_{\min}(K)} \right)^{2}, & \text{for } a = 0, \end{cases}$$

bd
$$\phi_{3i}\left(\left(\frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} + \sqrt{n}\right)R\right)$$
 is the boundary of the region

$$\phi_{3i}\left(\left(\frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} + \sqrt{n}\right)R\right)$$
$$= \left\{ \left(z_{1i}, z_{2i}\right) \in \mathbb{R}^2 : |z_{1i}| \le a_{3i} \left(\left(\frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} + \sqrt{n}\right)R\right)^2, |z_{2i}| \le \frac{2k_{1i}}{k_{2i}} \left(\left(\frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} + \sqrt{n}\right)R\right)\right\},$$

which is a compact set, with the constant, $a_{3i} > 0$ (see Appendix A.1), while the positive constants a_4 and a_5 defined as

$$a_4 := \frac{a_5 + 2}{\left(\left(a_5 + 1\right)^{\frac{1}{a}} - 1\right)^a} > 0$$
, for any $0 < a_5 < 1$

Remark 4.3. By selecting $\left(\frac{\lambda_{\max}(K)}{\lambda_{\min}(K)}\right)$ to be a constant ratio, ρ_2 is a constant.

Define the positive diagonal matrix, $\gamma_1(R) = \text{diag}\{\gamma_{1i}(R)\}_{i=1}^n \text{ with, } \gamma_{1i}(R) > 0, \forall i = 1, ..., n, \text{ and}$ the following positive vector

$$\gamma_{2}(R) = (\gamma_{2i}(R), \dots, \gamma_{2n}(R))^{\mathrm{T}} \text{ with } \gamma_{2i}(R) > 0, \quad \forall i = 1, \dots, n, \text{ such that } \forall (z_{1}, z_{2}, s, \widetilde{e}_{1}, \widetilde{e}_{2}) \in \Omega(R):$$

$$s^{\mathrm{T}}M(\cdot)K[\Delta_{1}(\cdot)] + \frac{1}{2}s^{\mathrm{T}}\dot{M}(\cdot)s + s^{\mathrm{T}}M(\cdot)K_{2}\left(\operatorname{sig}(z_{2} + s + \widetilde{e}_{2})^{\frac{2b}{1+b}} - \operatorname{sig}(z_{2} + s)^{\frac{2b}{1+b}}\right) \leq \sum_{i=1}^{n} \gamma_{1i}(R)|s_{i}|,$$

$$K[\Delta_{2}(\cdot)] \leq \gamma_{2}(R),$$

where such an upper bound always exist since M, C, F, G, D and desired trajectories are locally bounded, it implies that $\Delta_1(.)$, and $\Delta_2(.)$ are locally bounded as well (i.e. it compose of summation of locally bounded functions). From [143], the multi-valued function $K[\Delta_1(.)]$ and $K[\Delta_2(.)]$ are locally bounded as well. Also, using the skew-symmetry property 3.2.2, we have

$$\frac{1}{2}s^T \dot{M}(\cdot)s = s^T C(\cdot)s$$

which is locally bounded as well. Hence, within a compact set, an upper bound on the above function exists.

The time derivatives of the Lyapunov functions, in accordance to lemma 2.4, $\forall (z_1, z_2, s, \tilde{e}_1, \tilde{e}_2) \in \Omega(R)$ of the closed-loop system satisfy the following inequalities:

Differential inequalities for the z-subsystem: (see Appendix A.1)

 $\forall i = 1, \dots, n$:

for b > 0:

$$\begin{split} \dot{V}_{zi}(z_{1i}, z_{2i}) & \stackrel{a.e.}{\leq} -\frac{1}{2} \left(\frac{\omega_{4i}}{\overline{\pi}_{1i}} \right) V_{zi}^{\frac{3+5b}{4+4b}} - \frac{1}{4} \left(\frac{\omega_{4i}}{\overline{\pi}_{1i}^{\frac{3+5b}{4+4b}}} \right) V_{zi}^{\frac{1+2b}{2+2b}} \left(V_{zi}^{\frac{1}{4}} - |s_i| \left(\frac{4\overline{\pi}_{1i}}{\omega_{4i}} \right) \left(\frac{2^{\frac{1}{2+2b}}}{\overline{\pi}_{1i}^{\frac{1+2b}{2+2b}}} \right) \right) \\ & -\frac{1}{4} \left(\frac{\omega_{4i}}{\overline{\pi}_{1i}^{\frac{3+5b}{4+4b}}} \right) V_{zi}^{\frac{3}{4}} \left(V_{zi}^{\frac{b}{2+2b}} - a_{1i} |s_i + \widetilde{e}_{2i}|^{\frac{2b}{1+b}} \left(\frac{4\overline{\pi}_{1i}}{\omega_{4i}} \right) \left(\frac{2^{\frac{1}{4}}}{\overline{\pi}_{1i}^{\frac{3}{4}}} \right) \right) \end{split}$$

for b = 0:

$$\begin{split} \dot{V}_{zi}(z_{1i}, z_{2i}) &\stackrel{a.e.}{\leq} 2k_{1i}^{-2} |z_{1i}| |s_i| + \frac{3}{2} r_i |z_{1i}|^{\frac{1}{2}} |z_{2i}| |s_i| + k_{1i} |z_{2i}|^2 |s_i| + \frac{3}{2} r_i |z_{1i}|^{\frac{1}{2}} |z_{2i}|^2 \\ &- 2k_{1i} k_{2i} |z_{1i}| z_{2i} \text{SGN}(z_{2i} + s_i + \widetilde{e}_{2i}) \\ &- k_{2i} |z_{2i}|^3 \operatorname{sign}(z_{2i}) \text{SGN}(z_{2i} + s_i + \widetilde{e}_{2i}) - r_i (k_{1i} - k_{2i}) |z_{1i}|^{\frac{3}{2}} \\ &< 0, \text{ for } V_{zi} \geq \max_{(z_{1i}, z_{2i}) \in \operatorname{bd} \phi_{3i}(\max_{\{|s_i|, |s_i + \widetilde{e}_{2i}|\}})} V_{zi}(z_{1i}, z_{2i}) \end{split}$$

Differential inequalities for the *s*-subsystem:

$$\dot{V}_{s} \stackrel{a.e}{\in} \dot{\widetilde{V}} =_{s} \bigcap_{\xi \in \partial V_{s}} \xi^{\mathsf{T}} \begin{pmatrix} \boldsymbol{K}[f](s) \\ 1 \end{pmatrix} = \nabla V_{s}^{\mathsf{T}} \boldsymbol{K}[f](s) + \frac{1}{2} s^{\mathsf{T}} \dot{M}(\cdot) s$$

For *a* = 0:

Using the following notation:

 $N^+(s) = \{i \in \{1, ..., n\} : s_i \neq 0\}, N^0(s) = \{i \in \{1, ..., n\} : s_i = 0\}, \text{ observe that}$

$$-s^{\mathrm{T}}KK[\operatorname{sign}(s)] = -\sum_{i \in N^{+}(s)} k_{i} |s_{i}| - \sum_{i \in N^{0}(s)} k_{i}(0) \times [-1, +1] = -\sum_{i \in N^{+}(s)} k_{i} |s_{i}| - 0 = -\sum_{i=i}^{n} k_{i} |s_{i}|,$$

Also, note that, $K[(\operatorname{sign}(s_i + \widetilde{e}_{2i}) - \operatorname{sign}(s_i))] \le 2$, and observe that for $|s_i| > |\widetilde{e}_{2i}| > 0$,

$$\boldsymbol{K}[(\operatorname{sign}(s_i + \widetilde{e}_{2i}) - \operatorname{sign}(s_i))] = (\operatorname{sign}(s_i + \widetilde{e}_{2i}) - \operatorname{sign}(s_i)) = \operatorname{sign}(s_i) - \operatorname{sign}(s_i) = 0$$

$$\therefore |s_i k_i \boldsymbol{K}[(\operatorname{sign}(s_i + \widetilde{e}_{2i}) - \operatorname{sign}(s_i))] \le 2k_i |\widetilde{e}_{2i}|$$

Hence,

$$\begin{split} \dot{V}_{s} \stackrel{a.e.}{\subseteq} s^{\mathrm{T}} M(\cdot) \mathbf{K} \begin{bmatrix} -M^{-1}(\cdot) K \mathrm{sign}(s) + \Delta_{1}(\cdot) - M^{-1}(\cdot) K (\mathrm{sign}(s + \tilde{e}_{2}) - \mathrm{sign}(s)) \\ + K_{2} \left(\mathrm{sig}(z_{2} + s + \tilde{e}_{2})^{\frac{2b}{1+b}} - \mathrm{sig}(z_{2} + s)^{\frac{2b}{1+b}} \right) \end{bmatrix} + \frac{s^{\mathrm{T}} \dot{M}(\cdot) s}{2} \\ \leq -\sum_{i=1}^{n} k_{i} |s_{i}| + s^{\mathrm{T}} M(\cdot) \mathbf{K} [\Delta_{1}](\cdot) + \frac{1}{2} s^{\mathrm{T}} \dot{M}(\cdot) s - s^{\mathrm{T}} K \mathbf{K} [(\mathrm{sign}(s + \tilde{e}_{2}) - \mathrm{sign}(s))] \\ + s^{\mathrm{T}} M(\cdot) K_{2} \left(\mathrm{sig}(z_{2} + s + \tilde{e}_{2})^{\frac{2b}{1+b}} - \mathrm{sig}(z_{2} + s)^{\frac{2b}{1+b}} \right) \\ \leq -\sum_{i=1}^{n} k_{i} |s_{i}| + \sum_{i=1}^{n} \gamma_{1i}(R) |s_{i}| - \sum_{i=1}^{n} 2k_{i} |\tilde{e}_{2i}| \\ = -\frac{1}{2} \sum_{i=1}^{n} k_{i} |s_{i}| + \sum_{i=1}^{n} 2k_{i} |\tilde{e}_{2i}| - \sum_{i=1}^{n} |s_{i}| \left(\frac{k_{i}}{2} - \gamma_{1i}(R) \right) \\ \leq -\frac{\lambda_{\min}(K)}{2} \sum_{i=1}^{n} |s_{i}| + 2\lambda_{\max}(K) \sum_{i=1}^{n} |\tilde{e}_{2i}| - \min\left(\frac{k_{i}}{2} - \gamma_{1i}(R) \right) \sum_{i=1}^{n} |s_{i}| \\ = -\frac{\lambda_{\min}(K)}{2} \sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}} + 2\lambda_{\max}(K) \sum_{i=1}^{n} |\tilde{e}_{2i}|^{2} \right)^{\frac{1}{2}} - \min\left(\frac{k_{i}}{2} - \gamma_{1i}(R) \right) \sum_{i=1}^{n} |s_{i}| \\ \leq -\frac{\lambda_{\min}(K)}{2} \|s\| + 2\sqrt{n} \lambda_{\max}(K) \|\tilde{e}_{2}\| - \min\left(\frac{k_{i}}{2} - \gamma_{1i}(R) \right) \|s\|, \quad (\text{using lemma 2 of chapter 2), \end{split}$$

where lemma 2 of chapter 2 has been employed in the last inequality.

For *a* > 0:

From Appendix A.2-proposition 1, one obtains

$$-s^{T}K(\operatorname{sig}(s+\widetilde{e}_{2})^{a}-\operatorname{sig}(s)^{a}) = s^{T}K(\operatorname{sig}(s)^{a}-\operatorname{sig}(s+\widetilde{e}_{2})^{a})$$

$$= \sum_{i=1}^{n} s_{i}k_{i}(|s_{i}|^{a}\operatorname{sign}(s_{i})-|s_{i}+\widetilde{e}_{2i}|^{a}\operatorname{sign}(s_{i}+\widetilde{e}_{2i}))$$

$$\leq \sum_{i=1}^{n} |s_{i}|k_{i}||s_{i}|^{a}\operatorname{sign}(s_{i})-|s_{i}+\widetilde{e}_{2i}|^{a}\operatorname{sign}(s_{i}+\widetilde{e}_{2i})|$$

$$\leq \sum_{i=1}^{n} |s_{i}|k_{i}(a_{4}|\widetilde{e}_{2i}|^{a}+a_{5}|s_{i}|^{a})$$

$$= a_{4}\sum_{i=1}^{n} |s_{i}|k_{i}|\widetilde{e}_{2i}|^{a}+a_{5}\sum_{i=1}^{n} k_{i}|s_{i}|^{1+a}$$

where positive constants $a_4, a_5 \in \mathbb{R}^+$ are as defined earlier.

Then,

$$\begin{split} \dot{V}_{s} \stackrel{a.e.}{=} s^{\mathrm{T}} M(\cdot) \mathbf{K} \begin{bmatrix} -M^{-1}(\cdot) \mathrm{Ksig}(s)^{a} + \Delta_{1}(\cdot) - M^{-1}(\cdot) \mathrm{K} \left(\mathrm{sig}(s + \widetilde{e}_{2})^{a} - \mathrm{sig}(s)^{a} \right) \\ + K_{2} \left(\mathrm{sig}(z_{2} + s + \widetilde{e}_{2})^{\frac{2b}{1+b}} - \mathrm{sig}(z_{2} + s)^{\frac{2b}{1+b}} \right) \end{bmatrix} + \frac{s^{\mathrm{T}} \dot{M}(\cdot) s}{2} \\ \leq -\sum_{i=1}^{n} k_{i} |s_{i}|^{1+a} + s^{\mathrm{T}} M(\cdot) \mathbf{K} [\Delta_{1}](\cdot) + \frac{1}{2} s^{\mathrm{T}} \dot{M}(\cdot) s - s^{\mathrm{T}} \mathrm{K} \left(\mathrm{sig}(s + \widetilde{e}_{2})^{a} - \mathrm{sig}(s)^{a} \right) \\ + s^{\mathrm{T}} M(\cdot) K_{2} \left(\mathrm{sig}(z_{2} + s + \widetilde{e}_{2})^{\frac{2b}{1+b}} - \mathrm{sig}(z_{2} + s)^{\frac{2b}{1+b}} \right) \\ \leq -\sum_{i=1}^{n} k_{i} |s_{i}|^{1+a} + \sum_{i=1}^{n} \gamma_{1i}(R) |s_{i}| + a_{4} \sum_{i=1}^{n} |s_{i}| k_{i} |\widetilde{e}_{2i}|^{a} + a_{5} \sum_{i=1}^{n} k_{i} |s_{i}|^{1+a} \quad \text{with } a_{5} < 1, \\ = -(1 - a_{5}) \sum_{i=1}^{n} k_{i} |s_{i}|^{1+a} + a_{4} \sum_{i=1}^{n} |s_{i}| k_{i} |\widetilde{e}_{2i}|^{a} + \sum_{i=1}^{n} \gamma_{1i}(R) |s_{i}| \\ \leq - \left(1 - a_{5}) \lambda_{\min}(K) \sum_{i=1}^{n} |s_{i}|^{1+a} + a_{4} \lambda_{\max}(K) ||\widetilde{e}_{2}||^{a} \sum_{i=1}^{n} |s_{i}| + \lambda_{\max}(\gamma_{1i}(R)) \sum_{i=1}^{n} |s_{i}| \\ \leq - \frac{(1 - a_{5}) \lambda_{\min}(K)}{n^{a}} \left(\sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}} \right)^{1+a} + a_{4} \lambda_{\max}(K) ||\widetilde{e}_{2}||^{a} \sum_{i=1}^{n} |s_{i}| + \lambda_{\max}(\gamma_{1i}(R)) \sum_{i=1}^{n} |s_{i}| \\ = - \frac{(1 - a_{5}) \lambda_{\min}(K)}{n^{a}} \left(\sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}} \right)^{1+a} + a_{4} \lambda_{\max}(K) ||\widetilde{e}_{2}||^{a} \sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}} + \lambda_{\max}(\gamma_{1i}(R)) \sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}} \\ \leq - \frac{(1 - a_{5}) \lambda_{\min}(K)}{n^{a}} \left(\sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}} \right)^{1+a} + a_{4} \lambda_{\max}(K) ||\widetilde{e}_{2}||^{a} \sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}} + \lambda_{\max}(\gamma_{1i}(R)) \sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}} \\ \leq - \frac{(1 - a_{5}) \lambda_{\min}(K)}{n^{a}} \left(\sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}} \right)^{1+a} + a_{4} \lambda_{\max}(\gamma_{1i}(R)) \sqrt{n} \left(\sum_{i=1}^{n} |s_{i}|^{2} \right)^{\frac{1}{2}} \\ = - ||s| \left(\frac{(1 - a_{5}) \lambda_{\min}(K)}{2n^{a}}} \right) ||s||^{a} - a_{4} \lambda_{\max}(\gamma_{1i}(R)) \sqrt{n} \right)$$

Note that the property of $|\tilde{e}_{2i}| \leq ||\tilde{e}_2||$, $\forall i$ and lemma 2 of chapter 2 have been employed.

Differential inequalities for the \tilde{e} -subsystem: (see Section 4.1)

 $\forall i = 1, \dots, n$:

$$\begin{split} \dot{V}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) &\leq -l_{3i}|\tilde{e}_{2i}|^2 - l_{1i}|\tilde{e}_{1i}|^{3p-1} + |1 - l_{2i}||\tilde{e}_{1i}|^{2p-1}|\tilde{e}_{2i}| + \gamma_{2i}(R)|\tilde{e}_{2i}| \\ &< 0, \quad \text{for} \quad V_{\tilde{e}i} \geq \rho_{3i}(|\zeta_i|, l_{3i}), \end{split}$$

where $|\zeta_i| := \max\{|1 - l_{2i}|, \gamma_{2i}(R)\}$, the class \mathcal{KL} function,

$$\rho_{3i}(\zeta_i|, l_{3i}) \coloneqq \max_{(\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathrm{bd} \, \psi_{3i}(|\zeta_i|, l_{3i})} V_{\widetilde{e}i}(\widetilde{e}_{1i}, \widetilde{e}_{2i}),$$

bd $\psi_{3i}(\zeta_i|, l_{3i})$ is the boundary of the region

$$\psi_{3}(|\zeta_{i}|, l_{3i}) = \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^{2} : |\widetilde{e}_{1i}| \leq \beta_{1}(|\zeta_{i}|, l_{3i}), |\widetilde{e}_{2i}| \leq \beta_{2}(|\zeta_{i}|, l_{3i}) \}$$

which is a compact set, with functions $\beta_1(.)$ and $\beta_2(.)$ are of class- \mathcal{KL} .

Theorem 4.1: By redefining the control gain *K*, as $K = \hat{k} \times \text{diag} \{ \tilde{k}_i \}_{i=1}^n \in \mathbb{R}^{n \times n}$, where $\hat{k} > 0$ is the control gain, \tilde{k}_i is positive constant selected *a priori* for all i = 1, ..., n. Then, the ratio,

$$\left(\frac{\lambda_{\max}(K)}{\lambda_{\min}(K)}\right) = \frac{\max_{i} k_{i}}{\min_{i} k_{i}} = \frac{\max_{i} k_{i}}{\min_{i} \tilde{k}_{i}} = \text{constant}$$

can be selected *a priori* independent of the control gain, $\hat{k} > 0$. Thus, for any given K_1, K_2, L_1 , L_2 , and \tilde{k}_i , suppose that the initial conditions for the closed-loop system (4 - 7) belong to a given compact set, there always exists a c > 0 such that initially, $(z_1, z_2, s, \tilde{e}_1, \tilde{e}_2) \in \Omega(c)$. Also, a $\mu > 0$ can be selected such that $c > \mu > 0$. Then, by selecting \hat{k} such that

$$\hat{k} > \begin{cases} \frac{2\gamma_{1i}(c)}{\widetilde{k}_{i}}, & \text{for } a = 0, \\ \left(\frac{2\lambda_{\max}(\gamma_{1i}(c))n^{a+0.5}\overline{m}^{\frac{a}{2}}}{\left(\min_{i}\widetilde{k}_{i}\right)(1-a_{5})\rho_{2}^{\frac{a}{2}}}\right) \times \frac{1}{\mu^{a}}, & \text{for } a > 0, \end{cases}$$

and L_3 such that

$$\frac{\mu^2}{2} \ge \rho_{3i}(|\zeta_i|, l_{3i}), \text{ with } |\zeta_i| := \max\{|1 - l_{2i}|, \gamma_{2i}(c)\}$$

all the trajectories will enter the compact set $\Omega(\mu)$, in finite time, and stay there for all future times.

Proof of Theorem 4.1: The stability analysis proceeds in two steps.

 Obviously, for a given compact set of initial conditions, there exists c > 0 such that initially (z₁, z₂, s, e₁, e₂) belong to some compact set strictly inside Ω(c). A trajectory may leave the set Ω(c) only through one of the boundaries:

$$V_{zi}(z_{1i}, z_{2i}) = \rho_{1i}(c), \quad \forall \ i = 1, ..., n, \qquad V_s = \rho_2 \frac{c^2}{2}, \quad \text{or} \ V_{\tilde{e}i} = \frac{c^2}{2}, \qquad \forall \ i = 1, ..., n$$

Let us show that it is impossible:

(a) For the \tilde{e} -subsystem, on the boundary $\Omega(c)$: $\forall i = 1, ..., n$, $V_{\tilde{e}i} = \frac{c^2}{2}$,

now, in order for $\dot{V}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \stackrel{a.e.}{<} 0$ on the boundary, we need:

$$\frac{c^2}{2} \ge \rho_{3i} \left(\max\{ |1 - l_{2i}|, \gamma_{2i}(c)\}, l_{3i} \right)$$
(4 - 8)

To satisfy the inequality, recall that (see Section 4.1) ρ_{3i} is a class \mathcal{KL} function, thus, it

can be observed that for any given L_1 , L_2 , and c, by increasing the observer gain, L_3 , the value $\rho_{3i} \left(\max\{ |1-l_{2i}|, \gamma_{2i}(c)\}, l_{3i} \right)$ can be arbitrarily reduced such that the inequality (4 - 8) is satisfied. Thus, selecting L_3 as per Theorem 4.1, (4 - 8) is satisfied.

Hence, at the boundary, $V_{\tilde{e}i} = \frac{c^2}{2}$, $\dot{V}_{\tilde{e}i} (\tilde{e}_{1i}, \tilde{e}_{2i})^{a.e.} < 0$, so $V_{\tilde{e}i}$ is a decreasing function of time, and $(\tilde{e}_{1i}, \tilde{e}_{2i})$ stays in $\Omega_{\tilde{e}}(c)$, and

$$\frac{c^2}{2} = V_{\widetilde{e}_i} \ge \frac{1}{2p} |\widetilde{e}_{1i}|^{2p} \Longrightarrow (p)^{\frac{1}{2p}} c^{\frac{1}{p}} \ge |\widetilde{e}_{1i}| \Longrightarrow \sqrt{n} (p)^{\frac{1}{2p}} c^{\frac{1}{p}} \ge \|\widetilde{e}_1\|,$$

and
$$\frac{c^2}{2} = V_{\widetilde{e}_i} \ge \frac{1}{2} |\widetilde{e}_{2i}|^2 \Longrightarrow c \ge |\widetilde{e}_{2i}| \Longrightarrow \sqrt{n} c \ge \|\widetilde{e}_2\|$$

(b) For the s-subsystem, on the boundary $\Omega(c)$: $V_s(\cdot) = \frac{1}{2} s^{\mathrm{T}} M(\cdot) s = \rho_2 \frac{c^2}{2}$ implies

$$||s|| \ge \frac{\sqrt{\rho_2}}{\sqrt{\overline{m}}} c$$
 and $\Rightarrow |\widetilde{e}_{2i}| \le ||\widetilde{e}_2|| \le \sqrt{n} c$

For *a* = 0:

$$\begin{split} \dot{V}_{s} & \stackrel{a.e.}{\leq} -\frac{\lambda_{\min}(K)}{2} \|s\| + 2\sqrt{n} \lambda_{\max}(K) \|\widetilde{e}_{2}\| - \min_{i} \left(\frac{k_{i}}{2} - \gamma_{1i}(c)\right) \|s\| \\ & = -\left(\frac{\lambda_{\min}(K)}{2} \|s\| - 2\sqrt{n} \lambda_{\max}(K) \|\widetilde{e}_{2}\|\right) - \min_{i} \left(\frac{k_{i}}{2} - \gamma_{1i}(c)\right) \|s\| \end{split}$$

Note that on the boundary of the set $\Omega(c)$, the first term of the above inequality is non-positive:

$$-\left(\frac{\lambda_{\min}(K)}{2}\|s\| - 2\sqrt{n}\,\lambda_{\max}(K)\|\widetilde{e}_{2}\|\right) \leq -\left(\frac{\lambda_{\min}(K)}{2}\frac{\sqrt{\rho_{2}}}{\sqrt{\overline{m}}}\,c - 2n\lambda_{\max}(K)c\right) = 0$$

Hence,

$$\therefore \dot{V}_s \stackrel{a.e.}{\leq} -\min_i \left(\frac{k_i}{2} - \gamma_{1i}(c)\right) \|s\| \leq -c_1, \text{ where } c_1 = \min_i \left(\frac{k_i}{2} - \gamma_{1i}(c)\right) \left(\frac{\sqrt{\rho_2}}{\sqrt{m}}c\right),$$

Hence, for k_i sufficiently big, c_1 is positive and, correspondingly, $\dot{V}_s < 0$ almost everywhere. If the gain K satisfy the conditions of the Theorem 4.1, then, V_s is a decreasing function of t, so s stays in $\Omega_s(c)$ and

$$\|s\| \leq \frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} c \; .$$

For *a* > 0:

$$\begin{split} \dot{V}_{s} \stackrel{a.e.}{\leq} &- \|s\| \left(\frac{(1-a_{5})\lambda_{\min}(K)}{2n^{a}} \|s\|^{a} - a_{4} \lambda_{\max}(K)\sqrt{n} \|\widetilde{e}_{2}\|^{a} \right) \\ &- \|s\| \left(\frac{(1-a_{5})\lambda_{\min}(K)}{2n^{a}} \|s\|^{a} - \lambda_{\max}(\gamma_{1i}(c))\sqrt{n} \right) \end{split}$$

Note that on the boundary of the set $\Omega(c)$, the first term of the above inequality is non-

positive:

$$-\left(\frac{(1-a_5)\lambda_{\min}(K)}{2n^a}\|s\|^a - a_4 \lambda_{\max}(K)\sqrt{n}\|\widetilde{e}_2\|^a\right)$$
$$\leq -\left(\frac{(1-a_5)\lambda_{\min}(K)}{2n^a}\left(\frac{\sqrt{\rho_2}}{\sqrt{m}}c\right)^a - a_4 \lambda_{\max}(K)\sqrt{n}\left(\sqrt{n}c\right)^a\right) = 0$$

Hence,

$$\therefore \dot{V}_{s} \stackrel{a.e.}{\leq} - \|s\| \left(\frac{(1-a_{5})\lambda_{\min}(K)}{2n^{a}} \|s\|^{a} - \lambda_{\max}(\gamma_{1i}(c))\sqrt{n} \right) \leq -c_{2}$$

where

$$c_{2} = \left(\frac{\sqrt{\rho_{2}}}{\sqrt{\overline{m}}}c\right) \left(\frac{(1-a_{5})\lambda_{\min}(K)}{2n^{a}} \left(\frac{\sqrt{\rho_{2}}}{\sqrt{\overline{m}}}c\right)^{a} - \lambda_{\max}(\gamma_{1i}(c))\sqrt{n}\right)$$
(4-9)

Hence, for $\lambda_{\min}(K)$ sufficiently big, c_2 is positive and, correspondingly, $\dot{V}_s < 0$ almost everywhere. Then, V_s is a decreasing function of t, so s stays in $\Omega_s(c)$ and

$$\|s\| \leq \frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} c \; .$$

(c) For the *z*-subsystem, on the boundary of the set $\Omega(c)$:

For b > 0: At the boundary, it implies

$$\forall i = 1, \dots, n$$
:

$$V_{zi} = \rho_{1i}(c) = \max \begin{cases} (\rho_2)^2 \left(\frac{4\bar{\pi}_{1i}^{\frac{3+5b}{4+4b}}}{\omega_{4i}} \right)^4 \left(\frac{2^{\frac{1}{2+2b}} \omega_{2i}}{\underline{\pi}_{1i}^{\frac{1+2b}{2+2b}}} \right)^4, \\ \left(\sqrt{\rho_2} + \sqrt{n\underline{m}} \right)^4 \left(a_{1i} \left(\frac{4\bar{\pi}_{1i}^{\frac{3+5b}{4+4b}}}{\omega_{4i}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{3i}}{\underline{\pi}_{1i}^{\frac{3}{4}}} \right) \right)^{\frac{2+2b}{b}} \right\} \times \left(\frac{c}{\sqrt{\underline{m}}} \right)^4 \\ \geq \max \begin{cases} \left(\frac{4\bar{\pi}_{1i}^{\frac{3+5b}{4+4b}}}{\omega_{4i}} \right)^4 \left(\frac{2^{\frac{1}{2+2b}} \omega_{2i}}{\underline{\pi}_{1i}^{\frac{1+2b}{2+2b}}} \right)^4 |s_i|^4, \\ \left(a_{1i} \left(\frac{4\bar{\pi}_{1i}^{\frac{3+5b}{4+4b}}}{\omega_{4i}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{3i}}{\underline{\pi}_{1i}^{\frac{3}{4}}} \right) \right)^{\frac{2+2b}{b}} \left(|s_i| + |\tilde{e}_{2i}| \right)^4 \end{cases} \end{cases}$$

Noting that, $|s_i| + |\tilde{e}_{2i}| \ge |s_i + \tilde{e}_{2i}|$, hence, we have

$$\dot{V}_{zi}(z_{1i}, z_{2i}) \stackrel{a.e.}{\leq} -\frac{1}{2} \left(\frac{\omega_{4i}}{\overline{\pi}_{1i}^{\frac{3+5b}{4+4b}}} \right) V_{zi}^{\frac{3+5b}{4+4b}} < 0, \ \forall i = 1, \dots, n,$$

So, \dot{V}_{zi} is a decreasing function of *t* for all i = 1, ..., n and (z_1, z_2) stay in $\Omega_z(c)$.

For b = 0: At the boundary,

$$V_{zi}(z_{1i}, z_{2i}) = \rho_{1i}(c), \quad \forall \ i = 1, ..., n,$$

it implies

$$\frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} c \ge \|s\| \ge |s_i| \text{ and } \left(\frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} + \sqrt{n}\right) c \ge |s_i| + |\widetilde{e}_{2i}| \ge |s_i + \widetilde{e}_{2i}|,$$

Hence,

$$\Rightarrow \phi_{3i}\left(\left(\frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} + \sqrt{n}\right)c\right) \supset \phi_{3i}\left(|s_i + \widetilde{e}_{2i}|\right), \text{ and } \phi_{3i}\left(\left(\frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} + \sqrt{n}\right)c\right) \supset \phi_{3i}\left(|s_i|\right),$$
$$\Rightarrow V_{zi} = \rho_{1i}(c) \ge \rho_{1i}\left(\frac{|s_i + \widetilde{e}_{2i}|}{\left(\frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} + \sqrt{n}\right)}\right), \text{ and } V_{zi} = \rho_{1i}(c) \ge \rho_{1i}\left(\frac{|s_i|}{\left(\frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}} + \sqrt{n}\right)}\right)$$

Thus, $\dot{V}_{zi}(z_{1i}, z_{2i}) \stackrel{a.e.}{<} 0$, for $\forall i = 1, ..., n$. So, \dot{V}_{zi} is a decreasing function of *t* for all i = 1, ..., n and (z_1, z_2) stay in $\Omega_z(\mathbf{c})$ for the case of b = 0.

In the (b) section for the s-subsystem, the trajectories will stay inside the boundary $V_s(\cdot) = \rho_2 \frac{c^2}{2},$

provided that $\lambda_{\min}(K)$ or k_i is sufficiently high while maintaining the ratio

$$\left(\frac{\lambda_{\max}(K)}{\lambda_{\min}(K)}\right) = \text{constant},$$

such that ρ_2 remains as a constant. The increment of the gain *K* will cause the upper bound of the function $\gamma_2(c)$ to increase as well, since vector $\Delta_2(.)$ consists of term proportional to the gain *K* and $\mathbf{K}[\Delta_2(\cdot)] \leq \gamma_2(c)$

This in turn, implies that the observer gains, elements of the vector L_3 , have to be increased accordingly for the same compact region *c*, such that $\dot{V}_{\tilde{e}i}(\tilde{e}_{1i}, \tilde{e}_{2i})^{a.e.} < 0$.

As a result, the set $\Omega(c)$ is positively invariant, i.e. the trajectories of (4 - 7) stay in it once they have entered it.

- 2. Now we have shown boundedness, next is to show convergence to a smaller compact set $\Omega(\mu)$ with $c > \mu > 0$.
 - (a) For the \tilde{e} -subsystem:

Recall from the above that by increasing the observer gains, one can satisfy the following inequality for a given c:

$$\frac{c^2}{2} \ge \rho_{3i} \left(\max \left\{ \left| 1 - l_{2i} \right|, \gamma_{2i}(c) \right\}, l_{3i} \right),$$

now if the gain L_{3i} of the observer is much higher such that

$$\frac{c^2}{2} > \frac{\mu^2}{2} \ge \rho_{3i} \left(\max\left\{ \left| 1 - l_{2i} \right|, \gamma_{2i}(c) \right\}, l_{3i} \right),$$

all the trajectories starting in $\Omega_{\tilde{e}}(c)$ will enter $\Omega_{\tilde{e}}(\mu)$ within finite time and stay in it for all future times. Then, the upper bound on $(\tilde{e}_1, \tilde{e}_2)$ can be found to be

$$\frac{\mu^2}{2} = V_{\widetilde{e}i} \ge \frac{1}{2p} \left| \widetilde{e}_{1i} \right|^{2p} \Longrightarrow \left(p \right)^{\frac{1}{2p}} \mu^{\frac{1}{p}} \ge \left| \widetilde{e}_{1i} \right| \Longrightarrow \sqrt{n} \left(p \right)^{\frac{1}{2p}} \mu^{\frac{1}{p}} \ge \left\| \widetilde{e}_{1} \right\|,$$

and
$$\frac{\mu^2}{2} = V_{\widetilde{e}_i} \ge \frac{1}{2} |\widetilde{e}_{2i}|^2 \Longrightarrow \mu \ge |\widetilde{e}_{2i}| \Longrightarrow \sqrt{n} \ \mu \ge \|\widetilde{e}_2\|.$$

(b) For the s-subsystem, consider the states $(z_1, z_2, s) \in \Omega_z(c) \times \Omega_s(c) \setminus \Omega_z(\mu) \times \Omega_s(\mu)$ and

$$(\widetilde{e}_1, \widetilde{e}_2) \in \Omega_{\widetilde{e}}(\mu)$$
, it implies that, $||s|| \ge \frac{\sqrt{\rho_2}}{\sqrt{\overline{m}}} \mu$ and $||\widetilde{e}_2|| \le \sqrt{n} \mu$.

Hence, for a = 0:

$$\dot{V}_{s} \stackrel{a.e.}{\leq} - \left(\frac{\lambda_{\min}(K)}{2} \|s\| - 2\sqrt{n} \lambda_{\max}(K) \|\widetilde{e}_{2}\|\right) - \min_{i} \left(\frac{k_{i}}{2} - \gamma_{1i}(c)\right) \|s\|$$

Note that the first term of the above inequality is non-positive,

$$\therefore \dot{V}_{s} \stackrel{a.e.}{\leq} -\min_{i} \left(\frac{k_{i}}{2} - \gamma_{1i}(c) \right) \|s\| \leq -c_{3}, \text{ for } \|s\| \geq \frac{\sqrt{\rho_{2}}}{\sqrt{m}} \mu$$

where $c_{3} = \min_{i} \left(\frac{k_{i}}{2} - \gamma_{1i}(c) \right) \left(\frac{\sqrt{\rho_{2}}}{\sqrt{m}} \mu \right)$ which is positive for $k_{i}, > 2\gamma_{1i}(c), \forall i = 1, ..., n$,

which implies that the *s* trajectory will enter the set $\Omega_s(\mu)$ in finite time and stay there for all future times.

For *a* > 0:

$$\begin{split} \dot{V_{s}} & \stackrel{a.e.}{\leq} - \|s\| \bigg(\frac{(1-a_{5})\lambda_{\min}(K)}{2n^{a}} \|s\|^{a} - a_{4} \lambda_{\max}(K)\sqrt{n} \|\widetilde{e}_{2}\|^{a} \bigg) \\ & - \|s\| \bigg(\frac{(1-a_{5})\lambda_{\min}(K)}{2n^{a}} \|s\|^{a} - \lambda_{\max}(\gamma_{1i}(c))\sqrt{n} \bigg) \end{split}$$

Note that the first term of the above inequality is non-positive

$$\therefore \dot{V}_{s} \stackrel{a.e.}{\leq} - \|s\| \left(\frac{(1-a_{s})\lambda_{\min}(K)}{2n^{a}} \|s\|^{a} - \lambda_{\max}(\gamma_{1i}(c))\sqrt{n} \right) \leq -c_{4}, \quad \forall \|s\| \geq \frac{\sqrt{\rho_{2}}}{\sqrt{m}} \mu$$

where
$$c_4 = \left(\frac{\sqrt{\rho_2}}{\sqrt{\overline{m}}}\mu\right) \left(\frac{(1-a_5)\lambda_{\min}(K)}{2n^a}\left(\frac{\sqrt{\rho_2}}{\sqrt{\overline{m}}}\mu\right)^a - \lambda_{\max}(\gamma_{1i}(c))\sqrt{n}\right)$$

Note that by choosing sufficiently big *K*, as per the conditions of the Theorem 4.1 (while maintaining the ratio $\left(\frac{\lambda_{\max}(K)}{\lambda_{\min}(K)}\right) = \text{constant}$), c_4 is positive.

Thus, for all $a \ge 0$, the states $(z_1, z_2, s) \in \Omega_z(c) \times \Omega_s(c) \setminus \Omega_z(\mu) \times \Omega_s(\mu)$ and $(\tilde{e}_1, \tilde{e}_2) \in \Omega_{\tilde{e}}(\mu)$

will enter the set $(z_1, z_2, s, \tilde{e}_1, \tilde{e}_2) \in \Omega_z(c) \times \Omega_s(\mu) \times \Omega_{\tilde{e}}(\mu)$ in finite time and in it for all future times, in particular

$$||s|| \le \mu \frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}}$$
 together with the fact that $|s_i| \le ||s||$, for $i = 1, ..., n$, we have $|s_i| \le \mu \frac{\sqrt{\rho_2}}{\sqrt{\underline{m}}}$, for $i = 1, ..., n$

(c) For the z-subsystem, consider the states $(z_1, z_2) \in \Omega_z(c) \setminus \Omega_z(\mu)$ and $(s, \tilde{e}_1, \tilde{e}_2) \in \Omega_s(\mu) \times \Omega_{\tilde{e}}(\mu)$. Following the same arguments presented in point (1)-(c), it is not hard to see that, once $(\tilde{e}_1(t), \tilde{e}_2(t))$ enter the set $\Omega_{\tilde{e}}(\mu)$, and s(t) is inside the set $\Omega_s(\mu), (z_1, z_2)$ will enter the set $\Omega_z(\mu)$ in finite time and stay in it for all future times.

Hence, the set $\Omega(\mu) = \Omega_{\tilde{e}}(\mu) \times \Omega_s(\mu) \times \Omega_z(\mu)$ is positively invariant and attracting for all trajectories of the system (4 - 7) originating inside the set $\Omega(c) = \Omega_{\tilde{e}}(c) \times \Omega_s(c) \times \Omega_z(c)$ with $\mu < c$.

Remark 4.4. Note that the results above show that the control law achieves semiglobal practical stability. Its region of attraction can be given as the set $\Omega_{\hat{e}}(c) \times \Omega_s(c) \times \Omega_z(c)$, while its ultimate invariant set given by $\Omega_{\hat{e}}(\mu) \times \Omega_s(\mu) \times \Omega_z(\mu)$. The semiglobal nature of the control law can be seen where the estimate of region of attraction for each set of initial states can be increased by choosing a sufficiently big gains *K*, and *L*₃, for any *K*₁, *K*₂, and *L*₁, *L*₂. While the practical stability nature, pertaining to the set where the solutions converge is stable and may be reduced at will, can be achieved as well by tuning the gains *K* and *L*₃, (see the conditions on *K* and *L*₃ in Theorem 4.1).

Remark 4.5. The observation error dynamics comprise of the homogeneous super-twisting based observer plus a linear damping term. Note that the linear term depends on the velocity observation error, which is not available. However, it is not necessary in implementation by formulating as per (4 - 6).

Next, consider the special case of Theorem 4.1 with the observer parameter p = 0.5. In particular, when p = 0.5, the non-Lipschitz terms of the observer (4 - 7) will contain discontinuity and becomes a super-twisting observer with a linear damping term. From the results of Chapter 2 (see section 2.4.2), it has been shown that when p = 0.5, the super-twisting algorithm is able to be exactly robust with respect to persistent, non-vanishing, additive disturbances. Thus, these properties will be examined in the following.

Theorem 4.2: Consider the special case of the observer with p = 0.5. Using Theorem 4.1 with the following additional inequality,

$$\min\left\{l_{2i}, \left(\frac{3}{4}\right)\left(\frac{3}{8}\right)^{\frac{1}{2}}\left(\frac{1}{2^{\frac{3}{2}}}\right)\frac{l_{1i}l_{2i}^{2}}{(\gamma_{2i}(\mu)+l_{2i})^{\frac{3}{2}}}\right\} > \max\left\{4\gamma_{2i}(\mu), \left(\frac{4}{3}\right)^{2}\left(\frac{8}{3}\right)2^{3}\frac{\gamma_{2i}(\mu)^{3}}{l_{1i}^{2}l_{2i}}\right\}$$

which can be satisfied for sufficiently large L_1 and L_2 , the observation error, $(\tilde{e}_1, \tilde{e}_2)$ will converge to zero in finite time and stay there for all future times.

Proof of Theorem 4.2: The inequality of Theorem 4.2 can be satisfied by sufficiently large L_1 and L_2 . Then, using similar arguments as in the proofs of Theorem 4.1, for any compact set of initial conditions, there always exists a c > 0 such that initially, $(z_1, z_2, s, \tilde{e_1}, \tilde{e_2}) \in \Omega(c)$. Also, a $\mu > 0$ can be selected such that $c > \mu > 0$. Then, by selecting \bar{k} and L_3 as per Theorem 4.1, all the trajectories starting in $\Omega(c)$ will enter the compact set $\Omega(\mu)$ in finite time and stay in it for all future times.

Next, recall that the observer error dynamics for p = 0.5,

$$\begin{aligned} \dot{\widetilde{e}}_1 &= -L_1 \operatorname{sig}(\widetilde{e}_1)^{0.5} + \widetilde{e}_2 \quad ,\\ \dot{\widetilde{e}}_2 &= -L_2 \operatorname{sig}(\widetilde{e}_1)^0 - L_3 \widetilde{e}_2 + \Delta_2(\cdot) \end{aligned}$$

with element wise, $\forall i = 1,...,n$,

$$\dot{\widetilde{e}}_{1i} = -l_{1i} \operatorname{sig}(\widetilde{e}_{1i})^{0.5} + \widetilde{e}_{2i} ,$$

$$\dot{\widetilde{e}}_{2i} = -l_{2i} \operatorname{sign}(\widetilde{e}_{1i}) - l_{3i} \widetilde{e}_{2i} + \Delta_{2i}(\cdot)$$
(4 - 10)

Note that inside $\Omega(\mu)$, we have the upper bound

$$\boldsymbol{K}[\Delta_2(\cdot)] \leq \boldsymbol{\gamma}_2(\boldsymbol{\mu}),$$

Consider the Lyapunov function,

$$W_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) = \left(\frac{1}{2}\tilde{e}_{2i}^{2} + l_{2i}|\tilde{e}_{1i}|\right)^{2} - r_{\tilde{e}}|\tilde{e}_{1i}||\tilde{e}_{2i}|^{2}\operatorname{sign}(\tilde{e}_{1i}\tilde{e}_{2i})$$

(Note that this Lyapunov function has appeared as a strict Lyapunov function for the supertwisting based algorithm in Section 2.4.2)

where
$$\underline{\pi}_{2}\left(\left|\widetilde{e}_{2i}\right|^{4}+\left|\widetilde{e}_{1i}\right|^{2}\right) \leq W_{\widetilde{e}i} \leq \overline{\pi}_{2}\left(\left|\widetilde{e}_{2i}\right|^{4}+\left|\widetilde{e}_{1i}\right|^{2}\right)$$

$$\underline{\pi}_{2} := \min\left\{\frac{1}{8}, \frac{l_{2i}^{2}}{2}\right\}, \ \overline{\pi}_{2} := \max\left\{l_{2i}^{2} + \frac{l_{2i}}{2} + \frac{r_{\tilde{e}}}{2}, \ \frac{1}{4} + \frac{l_{2i}}{2} + \frac{r_{\tilde{e}}}{2}\right\}$$

with (see section 2.4.2) $\forall i = 1,...,n$,

$$\left(\frac{1}{4}\right)^{\frac{1}{2}}l_{2i} > r_{\tilde{e}}$$

Thus, $W_{\tilde{e}i}$ is positive definite and radially unbounded (see section 2.4.2). Since (4 - 10) is a differential equation that has discontinuous right-hand side, its solutions are understood in the sense of Filippov (see definition 2.1). According to lemma 2.4, the time derivative of non-smooth Lyapunov function exists almost everywhere

$$\frac{d}{dt}W_{\widetilde{e}i}(\widetilde{e}_{1i},\widetilde{e}_{2i})\stackrel{a.e.}{\in}\stackrel{i}{\widetilde{W}_{\widetilde{e}i}}=\bigcap_{\xi\in\partial W_{\widetilde{e}i}(\widetilde{e}_{1i},\widetilde{e}_{2i})}\xi^{\mathrm{T}}\boldsymbol{K}\begin{bmatrix}\dot{\widetilde{e}}_{1i}\\\dot{\widetilde{e}}_{2i}\end{bmatrix}$$

Note that for p = 0.5, $W_{\tilde{e}i}$ is not differentiable on $\tilde{e}_{1i} = 0$:

$$\begin{bmatrix} \dot{\tilde{e}}_{1i} \\ \dot{\tilde{e}}_{2i} \end{bmatrix} \in \begin{bmatrix} -l_{1i} \operatorname{sig}(\tilde{e}_{1i})^{\frac{1}{2}} + \tilde{e}_{2i} \\ -l_{2i} \mathbf{K}[\operatorname{sign}(\tilde{e}_{1i})] - l_{3i} \tilde{e}_{2i} + \mathbf{K}[\Delta_{2i}(\cdot)] \end{bmatrix},$$

and $\partial W_{\tilde{e}i}(\tilde{e}_{1i}, \tilde{e}_{2i}) = \begin{bmatrix} (\tilde{e}_{2i}^{2} + 2l_{2i}|\tilde{e}_{1i}|) (l_{2i} \operatorname{SGN}(\tilde{e}_{1i}) - r_{\tilde{e}}|\tilde{e}_{2i}|^{2} \operatorname{sign}(\tilde{e}_{2i})) \\ (\tilde{e}_{2i}^{2} + 2l_{2i}|\tilde{e}_{1i}|) (\tilde{e}_{2i}) - 2r_{\tilde{e}} \tilde{e}_{1i}|\tilde{e}_{2i}| \end{bmatrix}$

For $\tilde{e}_{1i} \neq 0$, $\forall \tilde{e}_{2i} \in \mathbb{R}$:

$$\begin{split} \dot{\widetilde{W}}_{\widetilde{e}i} &= -2l_{1i}l_{2i}^{2} |\widetilde{e}_{1i}|^{\frac{3}{2}} - l_{3i} |\widetilde{e}_{2i}|^{4} - l_{1i}l_{2i} |\widetilde{e}_{1i}|^{\frac{1}{2}} |\widetilde{e}_{2i}|^{2} - 2l_{2i}l_{3i} |\widetilde{e}_{1i}| |\widetilde{e}_{2i}|^{2} \\ &+ r_{\widetilde{e}}l_{1i} |\widetilde{e}_{1i}|^{\frac{1}{2}} |\widetilde{e}_{2i}|^{2} \operatorname{sign}(\widetilde{e}_{1i}\widetilde{e}_{2i}) - r_{\widetilde{e}} |\widetilde{e}_{2i}|^{3} + 2r_{\widetilde{e}}l_{2i} |\widetilde{e}_{1i}| |\widetilde{e}_{2i}| \\ &+ 2r_{\widetilde{e}}l_{3i}\widetilde{e}_{1i} |\widetilde{e}_{2i}|^{2} \operatorname{sign}(\widetilde{e}_{2i}) + K[\Delta_{2i}(\cdot)](2l_{2i} |\widetilde{e}_{1i}|\widetilde{e}_{2i} + \widetilde{e}_{2i}^{3} - 2r_{\widetilde{e}}\widetilde{e}_{1i}|\widetilde{e}_{2i}|) \end{split}$$

For $\tilde{e}_{1i} = 0$, $\forall \tilde{e}_{2i} \in \mathbb{R}$:

Let

 $\left[\left(l_{2i}\widetilde{e}_{2i}^{2}\right)\xi_{2}-r_{\widetilde{e}}\left|\widetilde{e}_{2i}\right|^{2}\operatorname{sign}(\widetilde{e}_{2i}), \widetilde{e}_{2i}^{3}\right]^{T} \text{ with } \xi_{2} \in [-1, 1] \text{ be an arbitrary element of } \partial W_{\widetilde{e}i}, \text{ then }$

$$\xi^{\mathrm{T}} \mathbf{K} \begin{bmatrix} \dot{\vec{e}}_{1i} \\ \dot{\vec{e}}_{2i} \end{bmatrix} = \left[\left(l_{2i} \tilde{e}_{2i}^{2} \right) \xi_{2} - r_{\tilde{e}} |\tilde{e}_{2i}|^{2} \operatorname{sign}(\tilde{e}_{2i}), \tilde{e}_{2i}^{3} \right] \begin{bmatrix} \tilde{e}_{2i} \\ -l_{2i} [-1,1] - l_{3i} \tilde{e}_{2i} + \mathbf{K} [\Delta_{2i}(\cdot)] \end{bmatrix}$$
$$= \left([\xi_{2} - 1, \xi_{2} + 1] \right) l_{2i} \tilde{e}_{2i}^{3} - r_{\tilde{e}} |\tilde{e}_{2i}|^{3} - l_{3i} |\tilde{e}_{2i}|^{4} + \mathbf{K} [\Delta_{2i}(\cdot)] \tilde{e}_{2i}^{3}$$

implies

$$\begin{aligned} \hat{W}_{\tilde{e}i} &= \bigcap_{\xi_{2} \in [-1,1]} \left(\left[\xi_{2} - 1, \xi_{2} + 1 \right] \right) l_{2i} \tilde{e}_{2i}^{3} - r_{\tilde{e}} \left| \tilde{e}_{2i} \right|^{3} - l_{3i} \left| \tilde{e}_{2i} \right|^{4} + K \left[\Delta_{2i} \left(\cdot \right) \right] \tilde{e}_{2i}^{3} \\ &= -r_{\tilde{e}} \left| \tilde{e}_{2i} \right|^{3} - l_{3i} \left| \tilde{e}_{2i} \right|^{4} + K \left[\Delta_{2i} \left(\cdot \right) \right] \tilde{e}_{2i}^{3} \end{aligned}$$

Thus, for all $(\tilde{e}_{1i}, \tilde{e}_{2i}) \in \mathbb{R}^n$, after rearrangement:

$$\dot{W}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i})^{a.e.} \leq -l_{1i}l_{2i}^{2}|\tilde{e}_{1i}|^{1.5} - \frac{r_{\tilde{e}}}{2}|\tilde{e}_{2i}|^{3} - l_{3i}|\tilde{e}_{2i}|^{4} + \dot{V}_{1}$$

where

$$\dot{V}_{1} = -l_{1i}l_{2i}^{2} |\widetilde{e}_{1i}|^{1.5} - \frac{r_{\tilde{e}}}{4} |\widetilde{e}_{2i}|^{3} - \left(\frac{r_{\tilde{e}}}{4} - \gamma_{2i}(\mu)\right) |\widetilde{e}_{2i}|^{3} - l_{1i}(l_{2i} - r_{\tilde{e}}) |\widetilde{e}_{1i}|^{0.5} |\widetilde{e}_{2i}|^{2} + 2l_{2i}\gamma_{2i}(\mu) |\widetilde{e}_{1i}| |\widetilde{e}_{2i}| + 2r_{\tilde{e}}(\gamma_{2i}(\mu) + l_{2i}) |\widetilde{e}_{1i}| |\widetilde{e}_{2i}| - 2l_{3i}(l_{2i} - r_{\tilde{e}}) |\widetilde{e}_{1i}| |\widetilde{e}_{2i}|^{2}$$

Applying lemma 2.1:

$$-\left|\widetilde{e}_{1i}\right|^{1.5}-\left|\widetilde{e}_{2i}\right|^{3}\leq-(1.5)^{\frac{1}{1.5}}\left|\widetilde{e}_{1i}\right|(3)^{\frac{1}{3}}\left|\widetilde{e}_{2i}\right|$$

Thus, if the following inequalities

$$\min\left\{l_{2i}, \left(\frac{3}{4}\right)\left(\frac{3}{8}\right)^{\frac{1}{2}}\left(\frac{1}{2^{\frac{3}{2}}}\right)\frac{l_{1i}l_{2i}^{2}}{\left(\gamma_{2i}(\mu)+l_{2i}\right)^{\frac{3}{2}}}\right\} > r_{\tilde{e}} > \max\left\{4\gamma_{2i}(\mu), \left(\frac{4}{3}\right)^{2}\left(\frac{8}{3}\right)2^{3}\frac{\gamma_{2i}(\mu)^{3}}{l_{1i}^{2}l_{2i}}\right\}$$

hold then the function \dot{V}_1 is negative definite. Then,

$$\dot{W}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \stackrel{a.e.}{\leq} - l_{1i} l_{2i}^{2} |\tilde{e}_{1i}|^{1.5} - \frac{r_{\tilde{e}}}{2} |\tilde{e}_{2i}|^{3}$$

Employing lemma 2.2, and the bounds of the Lyapunov function, we have:

$$\begin{split} \dot{W}_{\widetilde{e}i}(\widetilde{e}_{1i},\widetilde{e}_{2i})^{a.e.} & = -\omega_5 \left(\left| \widetilde{e}_{1i} \right|^{1.5} + \left| \widetilde{e}_{2i} \right|^3 \right) = -\omega_5 \left(\left| \left| \widetilde{e}_{1i} \right|^2 \right)^{\frac{3}{4}} + \left| \left| \widetilde{e}_{2i} \right|^4 \right)^{\frac{3}{4}} \right) \\ & \leq -\omega_5 \left(\left| \widetilde{e}_{1i} \right|^2 + \left| \widetilde{e}_{2i} \right|^4 \right)^{\frac{3}{4}} \\ & \leq - \left(\frac{\omega_5}{\overline{\pi}_2^{\frac{3}{4}}} \right) W_{\widetilde{e}i}^{\frac{3}{4}} \end{split}$$

where $\omega_5 = \min\left\{l_{1i}l_{2i}^2, \frac{r_{\tilde{e}}}{2}\right\}$

Hence, the states $(\tilde{e}_{1i}, \tilde{e}_{2i})$ will converge to the origin in finite time.

Remark 4.6. The addition of a linear damping term to the observer, L_3 has the benefit of reducing the gain of the super-twisting part of the observer (i.e. when p = 0.5) in achieving exact robustness property with respect to persistent, non-vanishing disturbances. Essentially, from Theorem 4.1, the L_3 term is instrumental for all the trajectories starting in $\Omega(c)$ to enter a smaller compact set $\Omega(\mu)$. In particular, note that the term $\gamma_2(\mu) < \gamma_2(c)$ (see definition in (4 - 7) and section 4.3), thus, from the condition of Theorem 4.2, a smaller gain pair of L_1 , L_2 is required to dominate $\gamma_2(\mu)$ as opposed to that of $\gamma_2(c)$.

Remark 4.7. Once the observation error converge to zero in finite time as per Theorem 4.2, the control law (4 - 4) will be identical to that of (3 - 2) presented in Chapter 3 for full-state feedback.

Theorem 4.3: In addition to the conditions in Theorem 4.1, consider the special case of regulation problem, where the desired trajectory is a constant value, q_0 , of the robot dynamics (4 - 3) and control (4 - 4) and observer (4 - 6) with the parameter *b* restricted to $0 < b \le 1$, and *a* and *p* selected as

$$a = \frac{2b}{1+b}, \ p = \frac{1+b}{2}$$

and the disturbance is upper bounded by $||D|| \le p_1 ||e_1|| + p_2 ||\dot{q}|| + p_3 ||e_1||^2 + p_4 ||\dot{q}||^2$, i.e. vanishing perturbation. Then, for b = 1, semiglobal exponential regulation is guaranteed, provided that K and L_3 are large enough with respect to initial error conditions. While, for 0 < b < 1, semiglobal finite-time regulation is assured, provided that K and L_3 are large enough with respect to initial error conditions, and the gravity vector at the constant desired position, $G(q_d)$ is zero.

Proof of Theorem 4.3: For this section, power of the control law a is selected as

$$a = \frac{2b}{1+b}, \ p = \frac{1+b}{2} \text{ for } 0 < b \le 1.$$

Let,

$$ar{z}_1 = 0, \quad ar{z}_2 = -ar{s}, \quad ar{s} = -\operatorname{sig}(K^{-1}G(q_0))^{\frac{1+b}{2b}},$$

 $\widetilde{z}_1 = z_1 - ar{z}_1, \quad \widetilde{z}_2 = z_2 - ar{z}_2, \quad ar{s} = s - ar{s},$

where

$$\overline{s}_i = -\left|k_i^{-1}g_i(q_0)\right|^{\frac{1+b}{2b}}\operatorname{sign}\left(k_i^{-1}g_i(q_0)\right) \text{ and } g_i(q_0) \text{ is the } i \text{ - th element of the vector } G(q_0).$$

Note that q_0 is a constant vector, and as a result the \overline{s} vector is a constant vector since the matrix *K* comprises constants as well. Also, \overline{s} is a constant that is defined for stability analyses only, its actual value, which require knowledge of gravity vector, G(q) is not required in the control law. Also note that *D* comprise of vanishing perturbations only, i.e. $p_0 = 0$ (note that this is the general assumption on regulation problem, however, if constant perturbation do

exists, the constant vector, \overline{s} can be redefined to accommodate this extra constant term).

Hence the closed loop system (4 - 7) could be rewritten as

$$\begin{split} \dot{\tilde{z}}_{1} &= \tilde{z}_{2} + \tilde{s}, \\ \dot{\tilde{z}}_{2} &= -K_{2} \operatorname{sig}(\tilde{z}_{2} + \tilde{s} + \tilde{e}_{2})^{\frac{2b}{1+b}} - K_{1} \operatorname{sig}(\tilde{z}_{1})^{b} \\ \dot{\tilde{s}} &= -M^{-1}(\tilde{z}_{1} + q_{d}) K \operatorname{sig}(\tilde{s})^{\frac{2b}{1+b}} + M^{-1}(\tilde{z}_{1} + q_{d}) K \left(\operatorname{sig}(\tilde{s})^{\frac{2b}{1+b}} - \operatorname{sig}(s)^{\frac{2b}{1+b}} + \operatorname{sig}(\bar{s})^{\frac{2b}{1+b}} \right) \\ &- M^{-1}(\tilde{z}_{1} + q_{d}) K \left(\operatorname{sig}(s + \tilde{e}_{2})^{\frac{2b}{1+b}} - \operatorname{sig}(s)^{\frac{2b}{1+b}} \right) + \tilde{\Delta}_{1}(\cdot) \\ \dot{\tilde{e}}_{1} &= -L_{1} \operatorname{sig}(\tilde{e}_{1})^{\frac{1+b}{2}} + \tilde{e}_{2} \quad , \\ \dot{\tilde{e}}_{2} &= -L_{2} \operatorname{sig}(\tilde{e}_{1})^{b} - L_{3}\tilde{e}_{2} + \tilde{\Delta}_{2}(\cdot) \end{split}$$

$$(4 - 11)$$

where

$$\begin{split} \widetilde{\Delta}_{1}(\cdot) &= -M^{-1}(\widetilde{z}_{1}+q_{d})C(\widetilde{z}_{1}+q_{d},\widetilde{z}_{2}+\widetilde{s}+\dot{q}_{d})(\widetilde{z}_{2}+\widetilde{s}+\dot{q}_{d}) - M^{-1}(\widetilde{z}_{1}+q_{d})F(\widetilde{z}_{2}+\widetilde{s}+\dot{q}_{d}) \\ &-M^{-1}(\widetilde{z}_{1}+q_{d})(G(\widetilde{z}_{1}+q_{d})-G(q_{0})) - M^{-1}(\widetilde{z}_{1}+q_{d})D - \ddot{q}_{d} \\ &+K_{2}\mathrm{sig}(\widetilde{z}_{2}+\widetilde{s}+\widetilde{e}_{2})^{\frac{2b}{1+b}} + K_{1}\mathrm{sig}(\widetilde{z}_{1})^{b} \end{split}$$

$$\begin{split} \widetilde{\Delta}_{2}(\cdot) &= -M^{-1}(\widetilde{z}_{1}+q_{d}) K \operatorname{sig}(\overline{s})^{\frac{2b}{1+b}} + M^{-1}(\widetilde{z}_{1}+q_{d}) K \operatorname{sig}(s+\widetilde{e}_{2})^{\frac{2b}{1+b}} \\ &+ M^{-1}(\widetilde{z}_{1}+q_{d}) C(\widetilde{z}_{1}+q_{d}, \widetilde{z}_{2}+\widetilde{s}+\dot{q}_{d}) (\widetilde{z}_{2}+\widetilde{s}+\dot{q}_{d}) \\ &+ M^{-1}(\widetilde{z}_{1}+q_{d}) F(\widetilde{z}_{2}+\widetilde{s}+\dot{q}_{d}) + M^{-1}(\widetilde{z}_{1}+q_{d}) (G(\widetilde{z}_{1}+q_{d})-G(q_{0})) \\ &+ M^{-1}(\widetilde{z}_{1}+q_{d}) D + \ddot{q}_{d} \end{split}$$

Note that the fact of $sig(\bar{s})^{\frac{2b}{1+b}} = -K^{-1}G(q_0)$ has been employed based on the above definition.

Consider the Lyapunov function

$$V(\widetilde{s}, \widetilde{z}_1, \widetilde{z}_2, \widetilde{e}_1, \widetilde{e}_2) = [V_{\widetilde{s}}(\widetilde{s})]^2 + V_{\widetilde{z}}(\widetilde{z}_1, \widetilde{z}_2) + W_{\widetilde{e}}(\widetilde{e}_1, \widetilde{e}_2)$$

with
$$V_{\widetilde{s}}(\widetilde{s}) = \frac{1}{2} \widetilde{s}^{\mathrm{T}} M \widetilde{s}$$
, $V_{\widetilde{z}}(\widetilde{z}_{1}, \widetilde{z}_{2}) = \sum_{i=1}^{n} V_{\widetilde{z}i}(\widetilde{z}_{1i}, \widetilde{z}_{2i})$, $W_{\widetilde{e}}(\widetilde{e}_{1}, \widetilde{e}_{2}) = \sum_{i=1}^{n} W_{\widetilde{e}i}(\widetilde{e}_{1i}, \widetilde{e}_{2i})$.

Note that the \tilde{s} and \tilde{z} subsystem have the same form as the s and z subsystem discussed above in Theorem 4.1 (i.e. by replacing (s, z_1, z_2) with $(\tilde{s}, \tilde{z}_1, \tilde{z}_2)$. Hence, a similar Lyapunov function structure is employed here, where

$$V_{\tilde{z}i}(\tilde{z}_{1i},\tilde{z}_{2i}) = \frac{k_{1i}^{2}}{(1+b)^{2}} |\tilde{z}_{1i}|^{2+2b} + \frac{1}{4} |\tilde{z}_{2i}|^{4} + r_{\tilde{z}i} |\tilde{z}_{1i}|^{\frac{3+3b}{2}} \operatorname{sign}(\tilde{z}_{1i})\tilde{z}_{2i} + \frac{k_{1i}}{(1+b)} |\tilde{z}_{1i}|^{1+b} |\tilde{z}_{2i}|^{2}$$

and $W_{\tilde{e}i}(\tilde{e}_{1i}, \tilde{e}_{2i}) = \left(\frac{1}{2}\tilde{e}_{2i}^{2} + \frac{l_{2i}}{1+b}|\tilde{e}_{1i}|^{1+b}\right)^{2} - r_{\tilde{e}}|\tilde{e}_{1i}||\tilde{e}_{2i}|^{\frac{2+4b}{1+b}}\operatorname{sign}(\tilde{e}_{1i}\tilde{e}_{2i})$

Note that $r_{\tilde{z}i} > 0$ always exists for any $k_{1i} > 0$ and $k_{2i} > 0$ such that $V_{\tilde{z}i} > 0$. While

$$\therefore W_{r\widetilde{e}}(\widetilde{e}_1, \widetilde{e}_2) \ge 0 \quad \text{for } \forall i = 1, \dots, n, \ \left(\frac{1+b}{8b+4}\right)^{\frac{1+2b}{2+2b}} \left(\frac{1}{1+b}\right)^{\frac{1}{2+2b}} l_{2i}^{\frac{1}{1+b}} > r_{\widetilde{e}}$$

where $r_{\tilde{e}} > 0$ is a constant . This Lyapunov function is positive definite (see Section 2.4.2 in Chapter 2).

Remark 4.8. Note that an additional term is included for the observer Lyapunov function (see Section 4.1) to reflect its finite time nature, which will be shown in the following development.

Note the following properties of the Lyapunov functions:

$$\frac{1}{2}\underline{m}\|\widetilde{s}\|^{2} \leq V_{\widetilde{s}}(\widetilde{s}) \leq \frac{1}{2}\overline{m}\|\widetilde{s}\|^{2} \Leftrightarrow \frac{\sqrt{2}}{\sqrt{\overline{m}}} \left[V_{\widetilde{s}}(\widetilde{s})\right]^{\frac{1}{2}} \leq \|\widetilde{s}\| \leq \frac{\sqrt{2}}{\sqrt{\underline{m}}} \left[V_{\widetilde{s}}(\widetilde{s})\right]^{\frac{1}{2}}$$

Also, (see Appendix A.1)

$$V_{z}(\tilde{z}_{1}, \tilde{z}_{2}) \geq \sum_{i=1}^{n} \underline{\pi}_{1i} \left\| \tilde{z}_{1i} \right\|^{2+2b} + \left\| \tilde{z}_{2i} \right\|^{4} \right)$$

$$\geq \underline{\pi}_{1} \left(\sum_{i=1}^{n} |\tilde{z}_{1i}|^{2+2b} + \sum_{i=1}^{n} |\tilde{z}_{2i}|^{4} \right), \text{ where } \underline{\pi}_{1} \coloneqq \min_{i} \{ \underline{\pi}_{1i} \},$$

$$\geq \underline{\pi}_{1} \left(\frac{1}{n^{b}} \left(\sum_{i=1}^{n} \left\| \tilde{z}_{1i} \right\|^{2} \right) \right)^{\frac{2+2b}{2}} + \frac{1}{n} \left(\sum_{i=1}^{n} \left\| \tilde{z}_{2i} \right\|^{2} \right)^{\frac{4}{2}} \right), \text{ using lemma 2 of chapter 2,}$$

$$= \underline{\pi}_{1} \left(\frac{1}{n^{b}} \left\| \widetilde{z}_{1} \right\|^{2+2b} + \frac{1}{n} \left\| \widetilde{z}_{2} \right\|^{4} \right)$$

$$\geq \frac{\underline{\pi}_{1}}{n} \left(\left\| \widetilde{z}_{1} \right\|^{2+2b} + \left\| \widetilde{z}_{2} \right\|^{4} \right), \text{ since } \min_{i=1}^{n} \frac{1}{n^{b}}, \frac{1}{n} \right) = \frac{1}{n} \text{ for } 0 < b \le 1 \text{ and } n \ge 1,$$

and the term,

$$\begin{split} \|\widetilde{z}_1\|^b + \|\widetilde{z}_2\|^{\frac{2b}{1+b}} &= \left(\|\widetilde{z}_1\|^{2+2b} \right)^{\frac{b}{2+2b}} + \left(\|\widetilde{z}_2\|^4 \right)^{\frac{b}{2+2b}} \\ &\leq n^{\frac{2+b}{2+2b}} \left(\|\widetilde{z}_1\|^{2+2b} + \|\widetilde{z}_2\|^4 \right)^{\frac{b}{2+2b}}, \text{ using lemma 2 of chapter 2,} \\ &\leq n^{\frac{2+b}{2+2b}} \left(\frac{n}{\underline{\pi}_i} V_{\widetilde{z}}(\widetilde{z}_1, \widetilde{z}_2) \right)^{\frac{b}{2+2b}} \\ &= \left(\frac{n}{\underline{\pi}_i^{\frac{b}{2+2b}}} \right) [V_{\widetilde{z}}(\widetilde{z}_1, \widetilde{z}_2)]^{\frac{b}{2+2b}} \end{split}$$

Consider the time derivative of the Lyapunov function $V_{\tilde{s}}$ along the solutions of the system for the \tilde{s} -subsystem:

$$\dot{V}_{\tilde{s}} \stackrel{a.e}{\in} \dot{\widetilde{V}}_{\tilde{s}} = \nabla V_{\tilde{s}}^{\mathrm{T}} \boldsymbol{K}[f](\tilde{s}) + \frac{1}{2} s^{\mathrm{T}} \dot{M}(\cdot) s$$

$$\subset \tilde{s}^{\mathrm{T}} M(\cdot) \begin{pmatrix} -M^{-1}(\tilde{z}_{1} + q_{d}) K \mathrm{sig}(\tilde{s})^{\frac{2b}{1+b}} \\ +M^{-1}(\tilde{z}_{1} + q_{d}) K \mathrm{sig}(\tilde{s})^{\frac{2b}{1+b}} - \mathrm{sig}(s)^{\frac{2b}{1+b}} + \mathrm{sig}(\bar{s})^{\frac{2b}{1+b}} \mathrm{sig}(\bar{s})^{\frac{2b}{1+b}} \\ -M^{-1}(\tilde{z}_{1} + q_{d}) K \mathrm{sig}(s + \tilde{e}_{2})^{\frac{2b}{1+b}} - \mathrm{sig}(s)^{\frac{2b}{1+b}} \mathrm{sig}(\bar{s})^{\frac{2b}{1+b}} \mathrm{sig$$

Note the following four properties that will simplify the above expression:

From Theorem 4.1, the states (s, z₁, z₂) will reach and stay inside the compact set, Ω_e(μ) × Ω_s(μ) × Ω_z(μ), and s̄, z̄₁, z̄₂ are constants, we have, for 0 < b ≤ 1, the following upper bounds (Note that the upper bound of the disturbance, ||D|| has been expressed in terms of the state variables of the closed-loop system):

$$\begin{split} \widetilde{s}^{\mathrm{T}} M(\cdot) \mathbf{K} \Big[\widetilde{\Delta}_{1}(\cdot) \Big] + \frac{1}{2} \widetilde{s}^{\mathrm{T}} \dot{M}(\cdot) \widetilde{s} &\leq -\underline{f} \| \widetilde{s} \|^{2} + \overline{f} \| \widetilde{s} \| \| \widetilde{z}_{2} \| + \overline{f} \| \widetilde{s} \| \| \dot{q}_{d} \| + \| \widetilde{s} \| C_{m} \| \widetilde{z}_{2} \|^{2} + C_{m} \| \widetilde{z}_{2} \| \| \widetilde{s} \|^{2} \\ &+ C_{m} \| \dot{q}_{d} \| \| \widetilde{s} \|^{2} + \| \widetilde{s} \| C_{m} \| \dot{q}_{d} \|^{2} + 2 \| \widetilde{s} \| C_{m} \| \dot{q}_{d} \| \| \widetilde{z}_{2} \| \\ &+ \| \widetilde{s} \| k_{g} \| \widetilde{z}_{1} \| + \| \widetilde{s} \| k_{g} \| q_{d} - q_{0} \| + \| \widetilde{s} \| p_{1} \| \| \widetilde{z}_{1} \| + \| \widetilde{s} \| p_{2} \| \| \widetilde{z}_{2} \| \\ &+ p_{2} \| \widetilde{s} \|^{2} + \| \widetilde{s} \| p_{2} \| \dot{q}_{d} \| + \| \widetilde{s} \| p_{3} \| \| \widetilde{z}_{1} \|^{2} + \| \widetilde{s} \| p_{4} \| \| \widetilde{z}_{2} \|^{2} \\ &+ p_{4} \| \widetilde{s} \|^{3} + \| \widetilde{s} \| p_{4} \| \dot{q}_{d} \|^{2} + 2 p_{4} \| \widetilde{z}_{2} \| \| \widetilde{s} \|^{2} + 2 \| \widetilde{s} \| p_{4} \| \| \widetilde{z}_{2} \| \| \dot{q}_{d} \| \\ &+ 2 p_{4} \| \widetilde{s} \|^{2} \| \dot{q}_{d} \| + \overline{m} \| \widetilde{s} \| \| \| \tilde{q}_{d} \| + \overline{m} \overline{k}_{1} \| \widetilde{s} \| n^{\frac{1-b}{2}} \| \widetilde{z}_{1} \|^{b} \\ &+ \overline{m} \overline{k}_{2} \| \widetilde{s} \| n^{\frac{1-b}{2+2b}} \| \widetilde{z}_{2} \|^{\frac{2b}{1+b}} + \overline{m} \overline{k}_{2} \| \widetilde{s} \| n^{\frac{1-b}{2+2b}} \| \widetilde{s} \| \frac{2b}{1+b} \\ &+ \overline{m} \overline{k}_{2} \| \widetilde{s} \| n^{\frac{1-b}{2+2b}} \| \widetilde{z}_{2} \|^{\frac{2b}{1+b}} \\ &\leq -\underline{f} \| \widetilde{s} \|^{2} + \gamma_{3} (\mu) \| \widetilde{s} \|^{\frac{1+3b}{1+b}} + \gamma_{4} (\mu) \| \widetilde{s} \| \Big(\| \widetilde{z}_{1} \|^{b} + \| \widetilde{z}_{2} \|^{\frac{2b}{1+b}} \Big) \\ &+ \overline{m} \overline{k}_{2} n^{\frac{1-b}{2+2b}} \| \widetilde{s} \| \| \widetilde{e}_{2} \|^{\frac{2b}{1+b}} + \alpha (\| \eta(t) \|) \gamma_{5} (\mu) \| \widetilde{s} \| \end{split}$$

where

$$\begin{split} \bar{k}_{1} &= \lambda_{\max} \left(K_{1} \right), \quad \bar{k}_{2} = \lambda_{\max} \left(K_{2} \right), \\ \left(p_{2} \left\| \tilde{s} \right\|^{2} + 3p_{4} \left\| \tilde{s} \right\|^{3} + \overline{m} \bar{k}_{2} n^{\frac{1-b}{2+2b}} \left\| \tilde{s} \right\| \left\| \tilde{s} \right\|^{\frac{2b}{1+b}} \right) \leq \gamma_{3} (\mu) \left\| \tilde{s} \right\|^{\frac{1+3b}{1+b}} \\ \left(\overline{f} \left\| \tilde{s} \right\| \left\| \tilde{z}_{2} \right\| + \overline{m} \bar{k}_{1} n^{\frac{1-b}{2}} \left\| \tilde{s} \right\| \left\| \tilde{z}_{1} \right\|^{b} + \overline{m} \bar{k}_{2} n^{\frac{1-b}{2+2b}} \left\| \tilde{s} \right\| \left\| \tilde{z}_{2} \right\|^{\frac{2b}{1+b}} \\ + k_{g} \left\| \tilde{s} \right\| \left\| \tilde{z}_{1} \right\| + p_{1} \left\| \tilde{s} \right\| \left\| \tilde{z}_{1} \right\| + p_{2} \left\| \tilde{s} \right\| \left\| \tilde{z}_{2} \right\| \\ + 2C_{m} \left\| \tilde{s} \right\| \left\| \tilde{z}_{2} \right\|^{2} + 3p_{4} \left\| \tilde{s} \right\| \left\| \tilde{z}_{2} \right\|^{2} + p_{3} \left\| \tilde{s} \right\| \left\| \tilde{z}_{1} \right\|^{2} + C_{m} \left\| \tilde{z}_{2} \right\| \left\| \tilde{s} \right\|^{2} \\ \left(k_{g} \left\| \tilde{s} \right\| \left\| q_{d} - q_{0} \right\| + C_{m} \left\| \dot{q}_{d} \right\| \left\| \tilde{s} \right\|^{2} + 3p_{4} \left\| \tilde{s} \right\| \left\| \dot{q}_{d} \right\|^{2} + p_{2} \left\| \tilde{s} \right\| \left\| \dot{q}_{d} \right\| \\ + 2C_{m} \left\| \tilde{s} \right\| \left\| \dot{q}_{d} \right\|^{2} + \overline{f} \left\| \tilde{s} \right\| \left\| \dot{q}_{d} \right\| + \overline{m} \left\| \tilde{s} \right\| \left\| \ddot{q}_{d} \right\| \\ \end{split} \right) \leq \alpha \left(\left\| \eta(t) \right\| \right) \gamma_{5}(\mu) \left\| \tilde{s} \right\| \end{aligned}$$

with $\gamma_3(.)$, $\gamma_4(.)$, $\gamma_5(.)$ are positive functions and $\alpha(.)$ is a class \mathcal{K} function, and the vector

 $\eta(t)$ is defined as:

$$\eta^{\mathrm{T}}(t) = \left[\left(q_{d}(t) - q_{0} \right)^{\mathrm{T}}, \dot{q}_{d}^{\mathrm{T}}(t), \ddot{q}_{d}^{\mathrm{T}}(t) \right] \in \mathbb{R}^{3n}.$$

2. Let, us define the diagonal matrix $\Lambda := \operatorname{diag}\left(-\left|\operatorname{sig}\left(\frac{\widetilde{s}_i}{\overline{s}_i}\right)^{\frac{2b}{1+b}} - \operatorname{sig}\left(\frac{\widetilde{s}_i}{\overline{s}_i} + 1\right)^{\frac{2b}{1+b}}\right| + 1\right)$, and

since,
$$\operatorname{sig}(\overline{s})^{\frac{2b}{1+b}} = -K^{-1}G(q_0)$$
,
then, $\sum_{i=1}^{n} \widetilde{s}_i k_i \left(\left| \widetilde{s}_i \right|^{\frac{2b}{1+b}} \operatorname{sign}(\widetilde{s}_i) - \left| s_i \right|^{\frac{2b}{1+b}} \operatorname{sign}(s_i) + \left| \overline{s}_i \right|^{\frac{2b}{1+b}} \operatorname{sign}(\overline{s}_i) \right) = -\widetilde{s}^{\mathrm{T}} \Lambda G(q_0)$.

3. From lemma 2.2:

$$\sum_{i=1}^{n} \left|\widetilde{s}_{i}\right|^{\frac{1+3b}{1+b}} \geq \frac{1}{n^{\frac{2b}{1+b}}} \left\|\widetilde{s}\right\|^{\frac{1+3b}{1+b}}, \text{ and } \sum_{i=1}^{n} \left|\widetilde{s}_{i}\right| \leq \sqrt{n} \left\|\widetilde{s}\right\|,$$

4. From Appendix A.3 - Proposition 2,

$$\left\| s_i + \widetilde{e}_{2i} \right\|^{\frac{2b}{1+b}} \operatorname{sign}\left(s_i + \widetilde{e}_{2i}\right) - \left|s_i\right|^{\frac{2b}{1+b}} \operatorname{sign}\left(s_i\right) \le 2^{\frac{1-b}{1+b}} \left|\widetilde{e}_{2i}\right|^{\frac{2b}{1+b}},$$

and the fact that $\left|\widetilde{e}_{2i}\right| \le \left\|\widetilde{e}_{2i}\right\|, \forall i \in 1, \dots, n$.

Substituting these properties, $\dot{\tilde{V}_{s}}$ becomes,

$$\begin{split} \dot{V}_{\widetilde{s}} &\leq -\sum_{i=1}^{n} k_{i} |\widetilde{s}_{i}|^{\frac{1+3b}{1+b}} + \sum_{i=1}^{n} \widetilde{s}_{i} k_{i} \left(|\widetilde{s}_{i}|^{\frac{2b}{1+b}} \operatorname{sign}(\widetilde{s}_{i}) - |s_{i}|^{\frac{2b}{1+b}} \operatorname{sign}(s_{i}) + |\overline{s}_{i}|^{\frac{2b}{1+b}} \operatorname{sign}(\overline{s}_{i}) \right) \\ &- \sum_{i=1}^{n} \widetilde{s}_{i} \left(k_{i} |s_{i} + \widetilde{e}_{2i}|^{\frac{2b}{1+b}} \operatorname{sign}(s_{i} + \widetilde{e}_{2i}) - k_{i} |s_{i}|^{\frac{2b}{1+b}} \operatorname{sign}(s_{i}) \right) \\ &- \underline{f} \|\widetilde{s}\|^{2} + \gamma_{3}(\mu) \|\widetilde{s}\|^{\frac{1+3b}{1+b}} + \gamma_{4}(\mu) \|\widetilde{s}\| \left(\|\widetilde{z}_{1}\|^{b} + \|\widetilde{z}_{2}\|^{\frac{2b}{1+b}} \right) \\ &+ \overline{m} \overline{k}_{2} n^{\frac{1-b}{2+2b}} \|\widetilde{s}\| \|\widetilde{e}_{2}\|^{\frac{2b}{1+b}} + \alpha (\|\eta(t)\|) \gamma_{5}(\mu) \|\widetilde{s}\| \\ &\leq - \frac{k}{n^{\frac{2b}{1+b}}} \|s\|^{\frac{1+3b}{1+b}} - \underline{f}\|\widetilde{s}\|^{2} + \gamma_{3}(\mu) \|\widetilde{s}\|^{\frac{1+3b}{1+b}} + \gamma_{4}(\mu) \|\widetilde{s}\| \left(\|\widetilde{z}_{1}\|^{b} + \|\widetilde{z}_{2}\|^{\frac{2b}{1+b}} \right) \\ &+ \alpha (\|\eta(t)\|) \gamma_{5}(\mu) \|\widetilde{s}\| + \|\widetilde{s}\| \|\Lambda G(q_{0})\| + \left(\overline{m} \overline{k}_{2} n^{\frac{1-b}{2+2b}} + 2^{\frac{1-b}{1+b}} \overline{k} \sqrt{n} \right) \|\widetilde{s}\| \|\widetilde{e}_{2}\|^{\frac{2b}{1+b}} \end{split}$$

where

$$\overline{k} = \lambda_{\max}(K), \quad \underline{k} = \lambda_{\min}(K).$$

Employing the bounds on the Lyapunov functions and lemma 2.2, one obtains,

$$\begin{split} \dot{\tilde{V}}_{\tilde{s}} &\leq -\left(\frac{k}{n^{\frac{2b}{1+b}}}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}} V_{\tilde{s}}^{\frac{1+3b}{2+2b}} - \underline{f}\left(\frac{2}{\overline{m}}\right) V_{\tilde{s}} + \gamma_{3}\left(\mu\right) \left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right)^{\frac{1+3b}{1+b}} V_{\tilde{s}}^{\frac{1+3b}{2+2b}} \\ &+ \gamma_{4}\left(\mu\right) \left(\frac{\sqrt{2}n}{\sqrt{\underline{m}} \, \underline{\pi}_{i}^{\frac{2}{2+2b}}}\right) V_{\tilde{s}}^{\frac{1}{2}} V_{\tilde{z}}^{\frac{b}{2+2b}} + \left[\alpha(\|\eta(t)\|)\gamma_{5}(\mu) + \|\Lambda G(q_{0})\| \left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right) V_{\tilde{s}}^{\frac{1}{2}} \\ &+ \left(\overline{m}\bar{k}_{2}n^{\frac{1-b}{2+2b}} + 2^{\frac{1-b}{1+b}}\bar{k}\sqrt{n}\right) \left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right) V_{\tilde{s}}^{\frac{1}{2}} \|\tilde{e}_{2}\|^{\frac{2b}{1+b}} \end{split}$$
(4 - 12)

Similarly, we compute the time derivative of the Lyapunov function for the \tilde{z} -subsystem, (see the appendix A.1):

$$V_{\widetilde{z}}(\widetilde{z}_1,\widetilde{z}_2) = \sum_{i=1}^n V_{\widetilde{z}_i}(\widetilde{z}_{1i},\widetilde{z}_{2i})$$

Hence,

$$\dot{\widetilde{V}}_{\widetilde{z}}(\widetilde{z}_{1},\widetilde{z}_{2}) \leq -d_{0}\sum_{i=1}^{n}V_{\widetilde{z}i}^{\frac{3+5b}{4+4b}} + d_{1}\sum_{i=1}^{n}|\widetilde{s}_{i}| V_{\widetilde{z}i}^{\frac{1+2b}{2+2b}} + d_{2}\sum_{i=1}^{n}|\widetilde{s}_{i}| + \widetilde{e}_{2i}|^{\frac{2b}{1+b}}V_{\widetilde{z}i}^{\frac{3}{4}}$$

where

$$d_{0} = \min_{i} \left(\frac{\omega_{4i}}{\overline{\pi}_{1i}^{\frac{3+5b}{4+4b}}} \right), \qquad d_{1} = \max_{i} \left(\frac{2^{\frac{1}{2+2b}} \omega_{2i}}{\underline{\pi}_{1i}^{\frac{1+2b}{2+2b}}} \right), \qquad d_{2} = \max_{i} a_{1i} \left(\frac{2^{\frac{1}{4}} \omega_{3i}}{\underline{\pi}_{1i}^{\frac{3}{4}}} \right),$$

Using lemma 2.2:

$$-d_{0}\sum_{i=1}^{n}V_{\tilde{z}i}^{\frac{3+5b}{4+4b}} \leq -d_{0}\left(\sum_{i=1}^{n}V_{\tilde{z}i}\right)^{\frac{3+5b}{4+4b}},$$

$$d_{1}\sum_{i=1}^{n} V_{\tilde{z}i}^{\frac{1+2b}{2+2b}} \leq n^{\frac{1}{2+2b}} d_{1} \left(\sum_{i=1}^{n} V_{\tilde{z}i}^{\frac{1+2b}{2+2b}}\right)^{\frac{1+2b}{2+2b}},$$
$$d_{2}\sum_{i=1}^{n} V_{\tilde{z}i}^{\frac{3}{4}} \leq n^{\frac{1}{4}} d_{2} \left(\sum_{i=1}^{n} V_{\tilde{z}i}^{\frac{3}{4}}\right)^{\frac{3}{4}}$$

and the fact that $|x_i| \le ||x||, \forall i = 1, ..., n, x \in \mathbb{R}^n$, thus it follows that

$$\begin{split} \dot{V}_{\tilde{z}}(\tilde{z}_{1},\tilde{z}_{2}) &\leq -d_{0}V_{\tilde{z}}^{\frac{3+5b}{4+4b}} + \|\tilde{s}\|n^{\frac{1}{2+2b}}d_{1}V_{\tilde{z}}^{\frac{1+2b}{2+2b}} + \|\tilde{s}+\tilde{e}_{2}\|^{\frac{2b}{1+b}}n^{\frac{1}{4}}d_{2}V_{\tilde{z}}^{\frac{3}{4}} \\ &\leq -d_{0}V_{\tilde{z}}^{\frac{3+5b}{4+4b}} + n^{\frac{1}{2+2b}}d_{1}\|\tilde{s}\|V_{\tilde{z}}^{\frac{1+2b}{2+2b}} + n^{\frac{1}{4}}d_{2}\|\tilde{s}\|^{\frac{2b}{1+b}}V_{\tilde{z}}^{\frac{3}{4}} + n^{\frac{1}{4}}d_{2}\|\tilde{e}_{2}\|^{\frac{2b}{1+b}}V_{\tilde{z}}^{\frac{3}{4}} \end{split}$$

where in the last inequality, we have employed the following inequality:

$$\begin{split} \|\widetilde{s} + \widetilde{e}_2\|^{\frac{2b}{1+b}} &\leq \left(\|\widetilde{s}\| + \|\widetilde{e}_2\|\right)^{\frac{2b}{1+b}}, \text{ using triangle inequality} \\ &\leq \|\widetilde{s}\|^{\frac{2b}{1+b}} + \|\widetilde{e}_2\|^{\frac{2b}{1+b}}, \text{ for } 0 < b \leq 1, \text{ using lemma 2 of chapter 2} \end{split}$$

Employing the bounds on the Lyapunov functions and lemma 2.2, one obtains,

$$\dot{V}_{\tilde{z}}(\tilde{z}_{1},\tilde{z}_{2})^{a.e.} - d_{0}V_{\tilde{z}}^{\frac{3+5b}{4+4b}} + n^{\frac{1}{2+2b}}d_{1}\left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right)V_{\tilde{s}}^{\frac{1}{2}}V_{\tilde{z}}^{\frac{1+2b}{2+2b}} + n^{\frac{1}{4}}d_{2}\left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right)^{\frac{2b}{1+b}}V_{\tilde{s}}^{\frac{1}{4}+b}V_{\tilde{z}}^{\frac{3}{4}} + n^{\frac{1}{4}}d_{2}\left\|\tilde{e}_{2}\right\|^{\frac{2b}{1+b}}V_{\tilde{z}}^{\frac{3}{4}}$$

$$(4 - 13)$$

Next for the \tilde{e} -subsystem, the derivative of the Lyapunov function is:

$$\dot{W}_{\widetilde{e}} \stackrel{a.e}{\in} \overset{\dot{\widetilde{W}}_{\widetilde{e}}}{\in} = \bigcap_{\xi \in \partial W_{\widetilde{e}}} \xi^{\mathrm{T}} \mathbf{K} \begin{bmatrix} \dot{\widetilde{e}}_{1} \\ \dot{\widetilde{e}}_{2} \end{bmatrix} (\widetilde{e}_{1}, \widetilde{e}_{2})$$

$$\begin{split} \dot{\widetilde{W}}_{\widetilde{e}} &= \nabla W_{\widetilde{e}}^{\mathrm{T}} \mathbf{K} \begin{bmatrix} \dot{\widetilde{e}}_{1} \\ \dot{\widetilde{e}}_{2} \end{bmatrix} (\widetilde{e}_{1}, \widetilde{e}_{2}) \\ &= \sum_{i=1}^{n} \begin{pmatrix} -\frac{2l_{1i}l_{2i}^{-2}}{1+b} |\widetilde{e}_{1i}|^{\frac{3+5b}{2}} - l_{3i} |\widetilde{e}_{2i}|^{4} - l_{1i}l_{2i} |\widetilde{e}_{1i}|^{\frac{1+3b}{2}} |\widetilde{e}_{2i}|^{2} - \frac{2l_{2i}l_{3i}}{1+b} |\widetilde{e}_{1i}|^{1+b} |\widetilde{e}_{2i}|^{2} - r_{\widetilde{e}} |\widetilde{e}_{2i}|^{\frac{3+5b}{1+b}} \\ &+ r_{\widetilde{e}}l_{1i} |\widetilde{e}_{1i}|^{\frac{1+b}{2}} |\widetilde{e}_{2i}|^{\frac{2+4b}{1+b}} \operatorname{sign}(\widetilde{e}_{1i}\widetilde{e}_{2i}) + r_{\widetilde{e}} \left(\frac{2+4b}{1+b}\right) l_{2i} |\widetilde{e}_{1i}|^{1+b} |\widetilde{e}_{2i}|^{\frac{1+3b}{1+b}} \\ &+ r_{\widetilde{e}} \left(\frac{2+4b}{1+b}\right) l_{3i} |\widetilde{e}_{1i}| |\widetilde{e}_{2i}|^{\frac{2+4b}{1+b}} \operatorname{sign}(\widetilde{e}_{1i}\widetilde{e}_{2i}) \\ &+ \sum_{i=1}^{n} \mathbf{K} \left[\widetilde{\Delta}_{2i}(\cdot) \right] \left(\frac{2l_{2i}}{1+b} |\widetilde{e}_{1i}|^{1+b} |\widetilde{e}_{2i}| + |\widetilde{e}_{2i}|^{3} - r_{\widetilde{e}} \left(\frac{2+4b}{1+b}\right) \widetilde{e}_{1i} |\widetilde{e}_{2i}|^{\frac{1+3b}{1+b}} \right) \end{split}$$

Define $\|\tilde{\Delta}_2(.)\| \coloneqq \sup\{\|\varepsilon\|: \varepsilon \in K[\tilde{\Delta}_2(.)]\}\)$, then, we obtain:

$$\begin{split} \dot{\widetilde{W}}_{\widetilde{e}} &\leq -\sum_{i=1}^{n} \left(\frac{l_{1i}l_{2i}}{1+b} |\widetilde{e}_{1i}|^{\frac{3+5b}{2}} \right) - \frac{r_{\widetilde{e}}}{2} \sum_{i=1}^{n} |\widetilde{e}_{2i}|^{\frac{3+5b}{1+b}} + \dot{W}_{1} \\ &+ \left\| \widetilde{\Delta}_{2}(\cdot) \right\| \sum_{i=1}^{n} \left(\frac{2l_{2i}}{1+b} |\widetilde{e}_{1i}|^{1+b} |\widetilde{e}_{2i}| + |\widetilde{e}_{2i}|^{3} + r_{\widetilde{e}} \left(\frac{2+4b}{1+b} \right) |\widetilde{e}_{1i}| |\widetilde{e}_{2i}|^{\frac{1+3b}{1+b}} \right) \end{split}$$

where

$$\dot{W_{1}} = \sum_{i=1}^{n} \left(-\frac{l_{1i}l_{2i}^{2}}{1+b} |\widetilde{e}_{1i}|^{\frac{3+5b}{2}} - l_{3i}|\widetilde{e}_{2i}|^{4} - l_{1i}l_{2i}|\widetilde{e}_{1i}|^{\frac{1+3b}{2}} |\widetilde{e}_{2i}|^{2} - \frac{2l_{2i}l_{3i}}{1+b} |\widetilde{e}_{1i}|^{1+b}|\widetilde{e}_{2i}|^{2} - \frac{r_{\tilde{e}}}{2} |\widetilde{e}_{2i}|^{\frac{3+5b}{1+b}} \right) \\ + r_{\tilde{e}}l_{1i}|\widetilde{e}_{1i}|^{\frac{1+b}{2}} |\widetilde{e}_{2i}|^{\frac{2+4b}{1+b}} + r_{\tilde{e}}\left(\frac{2+4b}{1+b}\right) l_{2i}|\widetilde{e}_{1i}|^{1+b}|\widetilde{e}_{2i}|^{\frac{1+3b}{1+b}} + r_{\tilde{e}}\left(\frac{2+4b}{1+b}\right) l_{3i}|\widetilde{e}_{1i}||\widetilde{e}_{2i}|^{\frac{2+4b}{1+b}} \right)$$

Applying lemma 2.1 we have $\forall i = 1, ..., n$:

$$-\left|\widetilde{e}_{1i}\right|^{\frac{3+5b}{2}} - \left|\widetilde{e}_{2i}\right|^{\frac{3+5b}{1+b}} \le -\left(\frac{3+5b}{1+b}\right)^{\frac{1+b}{3+5b}} \left|\widetilde{e}_{1i}\right|^{\frac{1+b}{2}} \left(\frac{3+5b}{2+4b}\right)^{\frac{2+4b}{3+5b}} \left|\widetilde{e}_{2i}\right|^{\frac{2+4b}{1+b}},$$

$$-\left|\widetilde{e}_{1i}\right|^{\frac{3+5b}{2}} - \left|\widetilde{e}_{2i}\right|^{\frac{3+5b}{1+b}} \le -\left(\frac{3+5b}{2+2b}\right)^{\frac{2+2b}{3+5b}} \left|\widetilde{e}_{1i}\right|^{1+b} \left(\frac{3+5b}{1+3b}\right)^{\frac{1+3b}{3+5b}} \left|\widetilde{e}_{2i}\right|^{\frac{1+3b}{1+b}},$$

$$-\left|\widetilde{e}_{2i}\right|^{4}-\left|\widetilde{e}_{1i}\right|^{1+b}\left|\widetilde{e}_{2i}\right|^{2}\leq-\left(\frac{1+b}{b}\right)^{\frac{b}{1+b}}\left|\widetilde{e}_{2i}\right|^{\frac{4b}{1+b}}(1+b)^{\frac{1}{1+b}}\left|\widetilde{e}_{1i}\right|\left|\widetilde{e}_{2i}\right|^{\frac{2}{1+b}}$$

Thus, if the following inequalities

$$\min\left\{\lambda_{1}\left(\frac{l_{2i}^{2}}{l_{1i}^{\frac{2+4b}{1+b}}}\right), \lambda_{2}\left(l_{1i}l_{2i}^{\frac{1-b}{2+2b}}\right), \lambda_{3}l_{2i}^{\frac{1}{1+b}}\right\} \geq r_{\tilde{e}}$$

where

$$\lambda_{1} = \left(\frac{1}{6}\right)^{\frac{3+5b}{1+b}} \left(\frac{6+10b}{(1+b)^{2}}\right) \left(\frac{3+5b}{2+4b}\right)^{\frac{2+4b}{1+b}},$$

$$\lambda_{2} = \left(\frac{(1+b)}{12+24b}\right)^{\frac{3+5b}{2+2b}} \left(\frac{3+5b}{(1+b)^{2}}\right) \left(\frac{3+5b}{1+3b}\right)^{\frac{1+3b}{2+2b}}, \quad \lambda_{3} = \left(\frac{1+b}{b}\right)^{\frac{b}{1+b}} \left(\frac{1+b}{2+4b}\right)^{\frac{1}{1+b}},$$

hold then the function \dot{W}_1 is negative definite. Note that such an $r_{\tilde{e}i} > 0$ always exists for any $l_{1i} > 0$, $l_{2i} > 0$.

Then,

$$\begin{split} \dot{W}_{\widetilde{e}} \stackrel{a.e.}{\leq} &- \sum_{i=1}^{n} \left(\frac{l_{1i} l_{2i}^{2}}{1+b} |\widetilde{e}_{1i}|^{\frac{3+5b}{2}} \right) - \frac{r_{\widetilde{e}}}{2} \sum_{i=1}^{n} |\widetilde{e}_{2i}|^{\frac{3+5b}{1+b}} \\ &+ \left\| \widetilde{\Delta}_{2}(\cdot) \right\| \sum_{i=1}^{n} \left(\frac{2l_{2i}}{1+b} |\widetilde{e}_{1i}|^{1+b} |\widetilde{e}_{2i}| + |\widetilde{e}_{2i}|^{3} + r_{\widetilde{e}} \left(\frac{2+4b}{1+b} \right) |\widetilde{e}_{1i}| |\widetilde{e}_{2i}|^{\frac{1+3b}{1+b}} \end{split}$$
(4 - 14)

Note the following properties that will aid in simplifying the above expression:

1. Note the above definition of the diagonal matrix, Λ , we have

$$\left(K\operatorname{sig}(\widetilde{s})^{\frac{2b}{1+b}} - K\operatorname{sig}(s)^{\frac{2b}{1+b}} + K\operatorname{sig}(\overline{s})^{\frac{2b}{1+b}}\right) = -\Lambda G(q_0),$$
2. While from Appendix A.3-Proposition 2,

$$\left| K \operatorname{sig}(s + \widetilde{e}_2)^{\frac{2b}{1+b}} - K \operatorname{sig}(s)^{\frac{2b}{1+b}} \right| \le 2^{\frac{1-b}{1+b}} K \left| \widetilde{e}_2 \right|^{\frac{2b}{1+b}}, \text{ where } \left| \widetilde{e}_2 \right| = \left(\left| \widetilde{e}_{21} \right|, \dots, \left| \widetilde{e}_{2n} \right| \right)^{\mathrm{T}},$$

- 3. From the property 1 of the manipulator dynamics, $0 < \underline{m_I} \le ||M^{-1}|| \le \overline{m_I}$,
- 4. From lemma 2 of chapter 2,

$$\begin{aligned} \left\| \operatorname{sig}(\widetilde{s})^{\frac{2b}{1+b}} \right\| &= \left(\sum_{i=1}^{n} \left(\left| \widetilde{s}_{i} \right|^{\frac{2b}{1+b}} \right)^{2} \right)^{\frac{1}{2}} = \left(\sum_{i=1}^{n} \left(\left| \widetilde{s}_{i} \right|^{2} \right)^{\frac{2b}{1+b}} \right)^{\frac{1}{2}} \\ &\leq \left(n^{\frac{1-b}{1+b}} \left(\sum_{i=1}^{n} \left| \widetilde{s}_{i} \right|^{2} \right)^{\frac{2b}{1+b}} \right)^{\frac{1}{2}}, \quad \text{using lemma 2 of chapter 2,} \\ &= n^{\frac{1-b}{2+2b}} \left(\left(\sum_{i=1}^{n} \left| \widetilde{s}_{i} \right|^{2} \right)^{\frac{1}{2}} \right)^{\frac{2b}{1+b}} = n^{\frac{1-b}{2+2b}} \left\| \widetilde{s} \right\|^{\frac{2b}{1+b}} \end{aligned}$$

and similar arguments we obtained $\left\| \left| \widetilde{e}_2 \right|^{\frac{2b}{1+b}} \right\| \le n^{\frac{1-b}{2+2b}} \left\| \widetilde{e}_2 \right\|^{\frac{2b}{1+b}}$,

5. From lemma 2.1 of chapter 2,

$$2\left\|\widetilde{z}_{2}\right\|\left\|\widetilde{s}\right\|+2\left\|\widetilde{z}_{2}\right\|\left\|\dot{q}_{d}\right\|+2\left\|\widetilde{s}\right\|\left\|\dot{q}_{d}\right\|\leq2\left\|\widetilde{z}_{2}\right\|^{2}+2\left\|\widetilde{s}\right\|^{2}+2\left\|\dot{q}_{d}\right\|^{2},$$

6. From Theorem 4.1, the states (s, z_1, z_2) will reach and stay inside the compact set, $\Omega_{\hat{e}}(\mu) \times \Omega_s(\mu) \times \Omega_z(\mu)$, and since $\overline{s}, \overline{z}_1$, and \overline{z}_2 are constants, for $0 < b \le 1$, the following functions can be upper bounded by,

$$\begin{pmatrix} \overline{m}_{I} \ \overline{k}n^{\frac{1-b}{2+2b}} \|\widetilde{s}\|^{\frac{2b}{1+b}} + 3\overline{m}_{I}C_{m} \|\widetilde{s}\|^{2} + \overline{m}_{I}\overline{f}\|\widetilde{s}\| + \overline{m}_{I}p_{2} \|\widetilde{s}\| + 3\overline{m}_{I}p_{4} \|\widetilde{s}\|^{2} \end{pmatrix} \leq \gamma_{6}(\mu) \|\widetilde{s}\|^{\frac{2b}{1+b}} \\ \begin{pmatrix} \overline{m}_{I}k_{g} \|\widetilde{z}_{1}\| + \overline{m}_{I}p_{1} \|\widetilde{z}_{1}\| + \overline{m}_{I}p_{3} \|\widetilde{z}_{1}\|^{2} + \overline{m}_{I}p_{2} \|\widetilde{z}_{2}\| \\ + \overline{m}_{I}\overline{f}\|\widetilde{z}_{2}\| + 3\overline{m}_{I}p_{4} \|\widetilde{z}_{2}\|^{2} + 3\overline{m}_{I}C_{m} \|\widetilde{z}_{2}\|^{2} \end{pmatrix} \leq \gamma_{7}(\mu) \left(\|\widetilde{z}_{1}\|^{b} + \|\widetilde{z}_{2}\|^{\frac{2b}{1+b}} \right) \\ \begin{pmatrix} \overline{m}_{I}(p_{2} + \overline{f}) \|\dot{q}_{d}\| + 3\overline{m}_{I}(p_{4} + C_{m}) \|\dot{q}_{d}\|^{2} \\ + \overline{m}_{I}k_{g} \|q_{d} - q_{0}\| + \|\ddot{q}_{d}\| + \overline{m}_{I} \|\Lambda G(q_{0})\| \end{pmatrix} \leq \alpha_{2}(\|\eta(t)\|) + \overline{m}_{I} \|\Lambda G(q_{0})\|$$

with $\gamma_6(.)$, and $\gamma_7(.)$, are positive functions and $\alpha_2(.)$ is a class K function, and the

vector $\eta(t)$ is defined as: $\eta^{\mathrm{T}}(t) = \left[\left(q_d(t) - q_0 \right)^{\mathrm{T}}, \dot{q}_d^{\mathrm{T}}(t), \ddot{q}_d^{\mathrm{T}}(t) \right] \in \mathbb{R}^{3n}$

7. Employing the above properties, after algebraic rearrangement,

$$\begin{split} \left\| \widetilde{\Delta}_{2}(\cdot) \right\| &\leq \gamma_{6}(\mu) \left\| \widetilde{s} \right\|^{\frac{2b}{1+b}} + \left(\overline{m}_{I} \,\overline{k} \, 2^{\frac{1-b}{1+b}} n^{\frac{1-b}{2+2b}} \left\| \widetilde{e}_{2} \right\|^{\frac{2b}{1+b}} \right) + \gamma_{7}(\mu) \left(\left\| \widetilde{z}_{1} \right\|^{b} + \left\| \widetilde{z}_{2} \right\|^{\frac{2b}{1+b}} \right) \\ &+ \alpha_{2} \left(\left\| \eta(t) \right\| \right) + \overline{m}_{I} \left\| \Lambda G(q_{0}) \right\| \end{split}$$

Note that the upper bound of the disturbance, ||D|| has been expressed in terms of the state variables of the closed-loop system in obtaining the above inequality.

8. From lemma 2.2 of chapter 2, the following can be obtained,

$$-\sum_{i=1}^{n} \left|\widetilde{e}_{1i}\right|^{\frac{3+5b}{2}} \leq -\frac{1}{n^{\frac{1+5b}{2}}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{2}}, -\sum_{i=1}^{n} \left|\widetilde{e}_{2i}\right|^{\frac{3+5b}{1+b}} \leq -\frac{1}{n^{\frac{2+4b}{1+b}}} \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}}, \sum_{i=1}^{n} \left|\widetilde{e}_{2i}\right|^{3} \leq n^{\frac{3}{2}} \left\|\widetilde{e}_{2}\right\|^{3},$$

$$\sum_{i=1}^{n} |\widetilde{e}_{1i}|^{1+b} |\widetilde{e}_{2i}| \leq ||\widetilde{e}_{1}||^{1+b} \sum_{i=1}^{n} |\widetilde{e}_{2i}|, \quad \text{since} ||\widetilde{e}_{1}|| \geq |\widetilde{e}_{1i}|,$$
$$\leq \sqrt{n} ||\widetilde{e}_{1}||^{1+b} ||\widetilde{e}_{2}||$$

$$\sum_{i=1}^{n} |\widetilde{e}_{1i}| \|\widetilde{e}_{2i}|^{\frac{1+3b}{1+b}} \le \|\widetilde{e}_{2}\|^{\frac{1+3b}{1+b}} \sum_{i=1}^{n} |\widetilde{e}_{1i}|, \text{ since } \|\widetilde{e}_{2}\| \ge |\widetilde{e}_{2i}|,$$
$$\le \sqrt{n} \|\widetilde{e}_{1}\| \|\widetilde{e}_{2}\|^{\frac{1+3b}{1+b}}$$

Substituting the above properties into (4 - 14), one obtains,

$$\begin{split} \dot{W}_{\tilde{e}} \stackrel{a.e.}{\leq} &- \left(\frac{l_{1}}{1+b} l_{2}^{2}}{1+b} \right) \left(\frac{1}{n^{\frac{1+5b}{2}}} \right) \|\widetilde{e}_{1}\|^{\frac{3+5b}{2}} - \left(\frac{r_{\tilde{e}}}{2} \right) \left(\frac{1}{n^{\frac{2+4b}{1+b}}} \right) \|\widetilde{e}_{2}\|^{\frac{3+5b}{1+b}} \\ &+ \left(\gamma_{6}(\mu) \|\widetilde{s}\|^{\frac{2b}{1+b}} + \overline{m}_{I} \,\overline{k} \, 2^{\frac{1-b}{1+b}} n^{\frac{1-b}{2+2b}} \|\widetilde{e}_{2}\|^{\frac{2b}{1+b}} \\ &+ \gamma_{7}(\mu) \left(\|\widetilde{z}_{1}\|^{b} + \|\widetilde{z}_{2}\|^{\frac{2b}{1+b}} \right) + \alpha_{2} \left(\|\eta(t)\|) + \overline{m}_{I} \|\Lambda G(q_{0})\|\right) \times \left(\frac{2\overline{l_{2}}}{1+b} \sqrt{n} \|\widetilde{e}_{1}\|^{1+b} \|\widetilde{e}_{2}\| + n^{\frac{3}{2}} \|\widetilde{e}_{2}\|^{3} \\ &+ r_{\tilde{e}} \left(\frac{2+4b}{1+b}\right) \sqrt{n} \|\widetilde{e}_{1}\| \|\widetilde{e}_{2}\|^{\frac{1+3b}{1+b}} \right) \end{split}$$

where

$$\lambda_{\min}(L_1) = \underline{l_1}, \quad \lambda_{\min}(L_2) = \underline{l_2}, \quad \lambda_{\max}(L_2) = \overline{l_2},$$

Employing the bounds on the Lyapunov functions and lemma 2.2, one obtains,

$$\begin{split} \vec{W}_{\vec{e}} \stackrel{a.c.}{\leq} & - \left(\frac{l_{1}}{1+b} \int_{\alpha}^{2} \frac{1}{n^{2}}\right) \left\| \vec{e}_{1} \right\|^{\frac{3+5b}{2}} - \left(\frac{r_{\vec{e}}}{2} \int_{\alpha}^{2} \frac{1}{n^{\frac{2+4b}{1+b}}} \right) \left\| \vec{e}_{2} \right\|^{\frac{3+5b}{1+b}} \\ & + \overline{l_{2}} \gamma_{6} \left(\mu \int_{\alpha} \frac{\sqrt{2}}{\sqrt{m}} \right)^{\frac{2b}{1+b}} \left(\frac{2\sqrt{n}}{1+b} \right) V_{\vec{s}}^{\frac{b}{1+b}} \left\| \vec{e}_{1} \right\|^{1+b} \left\| \vec{e}_{2} \right\| + \gamma_{6} \left(\mu \right) n^{\frac{3}{2}} \left(\frac{\sqrt{2}}{\sqrt{m}} \right)^{\frac{2b}{1+b}} V_{\vec{s}}^{\frac{b}{1+b}} \left\| \vec{e}_{2} \right\|^{3} \\ & + r_{\vec{e}} \gamma_{6} \left(\mu \right) \sqrt{n} \left(\frac{\sqrt{2}}{\sqrt{m}} \right)^{\frac{2b}{1+b}} \left(\frac{2+4b}{1+b} \right) V_{\vec{s}}^{\frac{b}{1+b}} \left\| \vec{e}_{1} \right\| \left\| \vec{e}_{2} \right\|^{\frac{1+3b}{1+b}} \\ & + \overline{l_{2}} \left(\frac{2\sqrt{n}}{1+b} \right) 2^{\frac{1-b}{1+b}} \overline{m}_{1} \overline{k} n^{\frac{1-b}{2+2b}} \left\| \vec{e}_{1} \right\|^{1+b} \left\| \vec{e}_{2} \right\|^{\frac{1+3b}{1+b}} + 2^{\frac{1-b}{1+b}} \overline{m}_{1} \overline{k} n^{\frac{4+2b}{2+2b}} \left\| \vec{e}_{2} \right\|^{\frac{3+5b}{1+b}} \\ & + r_{\vec{e}} \left(\frac{2+4b}{1+b} \right) 2^{\frac{1-b}{1+b}} \overline{m}_{1} \overline{k} n^{\frac{1-b}{2+2b}} \left\| \vec{e}_{1} \right\| \left\| \vec{e}_{2} \right\|^{\frac{1+5b}{1+b}} \\ & + \overline{l_{2}} \gamma_{7} \left(\mu \int_{\alpha} \left(\frac{2\sqrt{n}}{1+b} \right) \left(\frac{n}{\underline{\pi}_{1}^{\frac{b}{2+2b}}} \right) V_{\vec{z}}^{\frac{b}{2+2b}} \left\| \vec{e}_{1} \right\|^{1+b} \left\| \vec{e}_{2} \right\| \\ & + \gamma_{7} \left(\mu \int_{\alpha} \left(\frac{n^{\frac{5}{2}}}{\underline{\pi}_{1}^{\frac{b}{2+2b}}} \right) V_{\vec{z}}^{\frac{b}{2+2b}} \left\| \vec{e}_{1} \right\|^{1+b} \left\| \vec{e}_{2} \right\|^{1+b} \left\| \vec{e}_{2} \right\| \\ & + \left[\alpha_{2} \left(\left\| \eta(t) \right\| \right) + \overline{m}_{1} \right\| \Lambda G(q_{0}) \right\| \right] \times \left(\frac{2\overline{l_{2}}}{1+b} \sqrt{n} \left\| \vec{e}_{1} \right\|^{1+b} \left\| \vec{e}_{2} \right\|^{\frac{1+3b}{1+b}} \right) \right)$$

$$(4 - 15)$$

With the above results, we are now in a position to find the time derivative of the Lyapunov function for the closed-loop system, namely:

$$V(\tilde{s}, \tilde{z}_1, \tilde{z}_2, \tilde{e}_1, \tilde{e}_2) = [V_{\tilde{s}}(\tilde{s})]^2 + V_{\tilde{z}}(\tilde{z}_1, \tilde{z}_2) + W_{\tilde{e}}(\tilde{e}_1, \tilde{e}_2) \text{ and hence } \dot{V} = 2V_{\tilde{s}}\dot{V}_{\tilde{s}} + \dot{V}_{\tilde{z}} + \dot{W}_{\tilde{e}} .$$

Thus, substituting results (4 - 12), (4 - 13) and (4 - 15), after rearrangement,

$$\begin{split} \dot{V} \stackrel{a.e.}{\leq} &- \left(\frac{k}{n^{\frac{2b}{1+b}}}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}} V_{\tilde{s}}^{\frac{3+5b}{2+2b}} - \underline{f}\left(\frac{4}{m}\right) V_{\tilde{s}}^{2} - \frac{d_{0}}{2} V_{\tilde{z}}^{\frac{3+5b}{4+4b}} - \left(\frac{1}{2}\right) \left(\frac{l_{1}}{2} \frac{l_{2}}{1+b}\right) \left(\frac{1}{n^{\frac{1+5b}{2}}}\right) \|\widetilde{e}_{1}\|^{\frac{3+5b}{2}} \\ &- \left(\frac{r_{\tilde{e}}}{4}\right) \left(\frac{1}{n^{\frac{2+4b}{1+b}}}\right) \|\widetilde{e}_{2}\|^{\frac{3+5b}{1+b}} + \dot{V}_{1} + \dot{V}_{2} + 2\left[\alpha(\|\eta(t)\|)\gamma_{5}(\mu) + \|\Lambda G(q_{0})\|\left(\frac{\sqrt{2}}{\sqrt{m}}\right) V_{\tilde{s}}^{\frac{3}{2}} \\ &+ \left[\alpha_{2}(\|\eta(t)\|) + \overline{m}_{I}\|\Lambda G(q_{0})\|\right] \times \left(\frac{2l_{2}}{1+b} \sqrt{n}\|\widetilde{e}_{1}\|^{1+b}\|\widetilde{e}_{2}\| + n^{\frac{3}{2}}\|\widetilde{e}_{2}\|^{3} \\ &+ r_{\tilde{e}}\left(\frac{2+4b}{1+b}\right) \sqrt{n}\|\widetilde{e}_{1}\|\|\widetilde{e}_{2}\|^{\frac{1+3b}{1+b}} \end{split}$$

where

$$\begin{split} \dot{V}_{1} &= -\left(\frac{1}{4}\right) \left(\frac{k}{n^{\frac{2b}{1+b}}}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}} V_{\tilde{s}}^{\frac{3+5b}{2+2b}} - \frac{d_{0}}{4} V_{\tilde{z}}^{\frac{3+5b}{4+4b}} \\ &- \left(\left(\frac{1}{4}\right) \left(\frac{k}{n^{\frac{2b}{1+b}}}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}} - 2\gamma_{3}(\mu) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}}\right) V_{\tilde{s}}^{\frac{3+5b}{2+2b}} \\ &+ 2\gamma_{4}(\mu) \left(\frac{\sqrt{2}n}{\sqrt{m} \pi_{i}^{\frac{b}{2+2b}}}\right) V_{\tilde{s}}^{\frac{3}{2}} V_{\tilde{z}}^{\frac{b}{2+2b}} + n^{\frac{1}{2+2b}} d_{1}\left(\frac{\sqrt{2}}{\sqrt{m}}\right) V_{\tilde{s}}^{\frac{1}{2}} V_{\tilde{z}}^{\frac{1+2b}{2+2b}} \\ &+ n^{\frac{1}{4}} d_{2} \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{2b}{1+b}} V_{\tilde{s}}^{\frac{b}{1+b}} V_{\tilde{z}}^{\frac{3}{4}} \end{split}$$

$$\begin{split} \dot{V}_{2} &= -\left(\frac{1}{2}\right) \left(\frac{k}{2^{h}}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}} V_{3}^{\frac{3+5b}{2+2b}} - \frac{d_{0}}{4} V_{z}^{\frac{3+5b}{4+4b}} - \left(\frac{1}{2}\right) \left(\frac{l}{1+b}\right) \left(\frac{1}{1+b}\right) \left(\frac{1}{n^{\frac{1}{2}}}\right) \|\widetilde{e}_{1}\|^{\frac{3+5b}{2}} \\ &- \left(\frac{r_{\widetilde{e}}}{8}\right) \left(\frac{1}{n^{\frac{2+4b}{1+b}}}\right) \|\widetilde{e}_{2}\|^{\frac{3+5b}{1+b}} - \left(\left(\frac{r_{\widetilde{e}}}{8}\right) \left(\frac{1}{n^{\frac{2+4b}{1+b}}}\right) - 2^{\frac{1-b}{1+b}} \overline{m}_{I} \overline{k} n^{\frac{4+2b}{2+2b}}\right) \|\widetilde{e}_{2}\|^{\frac{3+5b}{1+b}} \\ &+ n^{\frac{1}{4}} d_{2} \|\widetilde{e}_{2}\|^{\frac{2b}{1+b}} V_{\overline{z}}^{\frac{3}{4}} + 2\left(\overline{m} \overline{k}_{2} n^{\frac{1-b}{2+2b}} + 2^{\frac{1-b}{1+b}} \overline{k} \sqrt{n}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right) V_{\overline{s}}^{\frac{3}{2}} \|\widetilde{e}_{2}\|^{\frac{2b}{1+b}} \\ &+ n^{\frac{1}{4}} d_{2} \|\widetilde{e}_{2}\|^{\frac{2b}{1+b}} V_{\overline{z}}^{\frac{3}{4}} + 2\left(\overline{m} \overline{k}_{2} n^{\frac{1-b}{2+2b}} + 2^{\frac{1-b}{1+b}} \overline{k} \sqrt{n}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right) V_{\overline{s}}^{\frac{3}{2}} \|\widetilde{e}_{2}\|^{\frac{2b}{1+b}} \\ &+ \overline{l}_{2} \gamma_{6} \left(\mu \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{2b}{1+b}} \left(\frac{2\sqrt{n}}{1+b}\right) V_{\overline{s}}^{\frac{b}{1+b}} \|\widetilde{e}_{1}\|^{1+b} \|\widetilde{e}_{2}\| + \gamma_{6} (\mu) n^{\frac{3}{2}} \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{2b}{1+b}} V_{\overline{s}}^{\frac{b}{1+b}} \|\widetilde{e}_{2}\| \\ &+ r_{\overline{e}} \gamma_{6} \left(\mu \right) \sqrt{n} \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{2b}{1+b}} \left(\frac{2+4b}{1+b}\right) V_{\overline{s}}^{\frac{b}{1+b}} \|\widetilde{e}_{1}\| \|\widetilde{e}_{2}\|^{\frac{1+3b}{1+b}} \\ &+ \overline{l}_{2} \left(\frac{2\sqrt{n}}{1+b}\right) 2^{\frac{1-b}{1+b}} \overline{m}_{I} \overline{k} n^{\frac{1-b}{2+2b}} \|\widetilde{e}_{1}\|^{1+b} \|\widetilde{e}_{2}\|^{\frac{1+3b}{1+b}} + r_{\overline{e}} \left(\frac{2+4b}{1+b}\right) 2^{\frac{1-b}{1+b}} \overline{m}_{I} \overline{k} n^{\frac{1}{1+b}} \|\widetilde{e}_{1}\| \|\widetilde{e}_{2}\|^{\frac{1+5b}{1+b}} \\ &+ \overline{l}_{2} \gamma_{7} \left(\mu \left(\frac{2\sqrt{n}}{1+b}\right) \left(\frac{n}{\pi}\frac{b}{2+2b}\right) V_{\overline{z}}^{\frac{b}{2+2b}} \|\widetilde{e}_{1}\|^{1+b} \|\widetilde{e}_{2}\| \\ &+ \gamma_{7} \left(\mu \left(\frac{n^{\frac{5}{2}}}{\frac{\pi}{2+2b}}\right) V_{\overline{z}}^{\frac{b}{2+2b}} \|\widetilde{e}_{2}\|^{3} + r_{\overline{e}} \gamma_{7} \left(\mu \left(\frac{2+4b}{1+b}\right) \left(\frac{n^{\frac{3}{2}}}{\frac{\pi}{2+2b}}\right) V_{\overline{z}}^{\frac{b}{2+2b}} \|\widetilde{e}_{1}\|^{\frac{1+3b}{1+b}} \\ &+ \gamma_{7} \left(\mu \left(\frac{n^{\frac{5}{2}}}{\frac{\pi}{2+2b}}\right) V_{\overline{z}}^{\frac{b}{2+2b}} \|\widetilde{e}_{2}\|^{3} + r_{\overline{e}} \gamma_{7} \left(\mu \left(\frac{2+4b}{1+b}\right) \left(\frac{n^{\frac{3}{2}}}{\frac{\pi}{2+2b}}\right) V_{\overline{z}}^{\frac{b}{2+2b}} \|\widetilde{e}_{1}\|^{\frac{1+3b}{1+b}} \\ &+ \frac{1-2}{2} \left(\frac{n}{2} \left(\frac{n}{2}\right) \left(\frac{n}{2}\right) \left(\frac{n}{2} \left(\frac{n}{2}\right) \left(\frac{n}{2}\right) \left(\frac{n}{2}\right) \left(\frac{n}{2}\right) \left(\frac{n}{2}\right) \left(\frac{n}{2}\right) \left(\frac{n}{2$$

For \dot{V}_1 , applying lemma 2.1

$$\begin{split} &-V_{\tilde{s}}^{\frac{3+5b}{2+2b}} - V_{\tilde{z}}^{\frac{3+5b}{4+4b}} \leq -\left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{3+5b}} V_{\tilde{s}}^{\frac{3}{2}} \left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} V_{\tilde{z}}^{\frac{b}{2+2b}}, \\ &-V_{\tilde{s}}^{\frac{3+5b}{2+2b}} - V_{\tilde{z}}^{\frac{3+5b}{4+4b}} \leq -\left(\frac{3+5b}{1+b}\right)^{\frac{1+b}{3+5b}} V_{\tilde{s}}^{\frac{1}{2}} \left(\frac{3+5b}{2+4b}\right)^{\frac{2+4b}{3+5b}} V_{\tilde{z}}^{\frac{1+2b}{2+2b}}, \\ &-V_{\tilde{s}}^{\frac{3+5b}{2+2b}} - V_{\tilde{z}}^{\frac{3+5b}{4+4b}} \leq -\left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} V_{\tilde{s}}^{\frac{b}{1+b}} \left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{3+5b}} V_{\tilde{z}}^{\frac{3}{4}}, \end{split}$$

thus if the following inequalities

$$\underline{k} > \max\left\{\lambda_4 \gamma_3(\mu), \quad \lambda_5 [\gamma_4(\mu)]^{\frac{3+5b}{3+3b}}, \quad \lambda_6, \quad \lambda_7\right\},$$

where

$$\begin{split} \lambda_{4} &= 8n^{\frac{2b}{1+b}} \left(\frac{\sqrt{m}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}}, \\ \lambda_{5} &= 2^{\frac{3+5b}{3+3b}} \left(\frac{36+36b}{3+5b}\right) \left(\frac{24b}{d_{0}(3+5b)}\right)^{\frac{2b}{3+3b}} n^{\frac{2b}{1+b}} \left(\frac{\sqrt{2}n}{\sqrt{m} \ \underline{\pi}_{i}^{\frac{b}{2+2b}}}\right)^{\frac{3+5b}{3+3b}} \left(\frac{\sqrt{m}}{\sqrt{2}}\right)^{\frac{1+3b}{1+b}}, \\ \lambda_{6} &= \left(\frac{12+12b}{3+5b}\right) \left(\frac{24+48b}{d_{0}(3+5b)}\right)^{\frac{2+4b}{1+b}} n^{\frac{2b}{1+b}} \left(n^{\frac{1}{2+2b}} d_{1} \left(\frac{\sqrt{2}}{\sqrt{m}}\right)\right)^{\frac{3+5b}{1+b}} \left(\frac{\sqrt{m}}{\sqrt{2}}\right)^{\frac{1+3b}{1+b}}, \\ \lambda_{7} &= \left(\frac{24b}{3+5b}\right) \left(\frac{36+36b}{d_{0}(3+5b)}\right)^{\frac{3+3b}{2b}} n^{\frac{2b}{1+b}} \left(n^{\frac{1}{4}} d_{2} \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{2b}{1+b}}\right)^{\frac{3+5b}{2b}} \left(\frac{\sqrt{m}}{\sqrt{2}}\right)^{\frac{1+3b}{1+b}}, \end{split}$$

hold then the function \dot{V}_1 is negative definite. The above inequalities can be satisfied by a sufficiently large gain *K*.

For \dot{V}_2 , applying lemma 2.1,

$$\begin{split} &- \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} - V_{\widetilde{z}}^{-\frac{3+5b}{4+4b}} \leq -\left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} \left\|\widetilde{e}_{2}\right\|^{\frac{2b}{1+b}} \left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{3+5b}} V_{\widetilde{z}}^{-\frac{3}{4}}, \\ &- \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} - V_{\widetilde{s}}^{-\frac{3+5b}{2+2b}} \leq -\left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} \left\|\widetilde{e}_{2}\right\|^{\frac{2b}{1+b}} \left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{3+5b}} V_{\widetilde{s}}^{-\frac{3}{2}}, \end{split}$$

$$\begin{split} &-\left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} - V_{\overline{s}}^{\frac{3+5b}{2+2b}} \leq -\left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{3+5b}} \left\|\widetilde{e}_{2}\right\|^{3} \left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} V_{\overline{s}}^{\frac{b}{1+b}}, \\ &-\left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{2}} - \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} \leq -\left(\frac{3+5b}{2+2b}\right)^{\frac{2+2b}{3+5b}} \left\|\widetilde{e}_{1}\right\|^{1+b} \left(\frac{3+5b}{1+3b}\right)^{\frac{1+3b}{3+5b}} \left\|\widetilde{e}_{2}\right\|^{\frac{1+3b}{1+b}}, \\ &-\left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{2}} - \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} \leq -\left(\frac{3+5b}{2}\right)^{\frac{2}{3+5b}} \left\|\widetilde{e}_{1}\right\| \left(\frac{3+5b}{1+5b}\right)^{\frac{1+5b}{3+5b}} \left\|\widetilde{e}_{2}\right\|^{\frac{1+5b}{1+b}}, \\ &-\left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{2}} - V_{\overline{z}}^{\frac{3+5b}{4+4b}} \leq -\left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{3+5b}} \left\|\widetilde{e}_{2}\right\|^{3} \left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} V_{\overline{z}}^{\frac{b}{2+2b}}, \\ &-\left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{2}} - \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} - V_{\overline{s}}^{\frac{3+5b}{2+2b}} \leq -\left(\frac{3}{2}\right)^{\frac{2}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left(3\right)^{\frac{1}{3}} \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{3+3b}} - V_{\overline{s}}^{\frac{3+5b}{2+2b}} \\ &-\left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{2}} - \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} - V_{\overline{s}}^{\frac{3+5b}{2+2b}} \leq -\left(\frac{3}{2}\right)^{\frac{2}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left(3\right)^{\frac{1}{3}} \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{3+3b}} - V_{\overline{s}}^{\frac{3+5b}{2+2b}} \\ &-\left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{2}} - \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} - V_{\overline{s}}^{\frac{3+5b}{2+2b}} \leq -\left(\frac{3}{2}\right)^{\frac{2}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left(3\right)^{\frac{1}{3}} \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{3+3b}} - V_{\overline{s}}^{\frac{3+5b}{2+2b}} \\ &-\left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{2}} - \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} - \left(\frac{3+5b}{3}\right)^{\frac{2}{3+5b}} \leq -\left(\frac{3}{2}\right)^{\frac{2}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left(3\right)^{\frac{1}{3}} \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{3+5b}} - V_{\overline{s}}^{\frac{3+5b}{2+2b}} \\ &-\left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{2}} - \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} - \left(\frac{3+5b}{3}\right)^{\frac{2}{3+5b}} \leq -\left(\frac{3}{2}\right)^{\frac{2}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left(3\right)^{\frac{1}{3}} \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{3+5b}} - \left(\frac{3+5b}{3}\right)^{\frac{3+5b}{3+5b}} \right\|^{\frac{3+5b}{3}} \right\|^{\frac{3+5b}{3+5b}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{3+5b}} \right\|^{\frac{3+5b}{3}} \right\|^{\frac{3+5b}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3}} \left\|\widetilde$$

$$\leq -\left(\left(\frac{3+5b}{3+3b}\right)\left(\frac{3}{2}\right)^{\frac{2}{3}}\left(3\right)^{\frac{1}{3}}\right)^{\frac{3+3b}{3+5b}} \|\widetilde{e}_{1}\|^{1+b} \|\widetilde{e}_{2}\|\left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} V_{\widetilde{s}}^{\frac{b}{1+b}},$$

$$- \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{2}} - \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} - V_{\widetilde{s}}^{\frac{3+5b}{2+2b}} \leq -\left(\frac{3+3b}{2}\right)^{\frac{2}{3+3b}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3+3b}} \left(\frac{3+3b}{1+3b}\right)^{\frac{1+3b}{3+3b}} \left\|\widetilde{e}_{2}\right\|^{\left(\frac{(3+5b)(1+3b)}{(1+b)(3+3b)}\right)} - V_{\widetilde{s}}^{\frac{3+5b}{2+2b}} \\ \leq -\left(\left(\frac{3+5b}{3+3b}\right)\left(\frac{3+3b}{2}\right)^{\frac{2}{3+3b}} \left(\frac{3+3b}{1+3b}\right)^{\frac{1+3b}{3+3b}}\right)^{\frac{3+3b}{3+5b}} \left\|\widetilde{e}_{1}\right\| \left\|\widetilde{e}_{2}\right\|^{\frac{1+3b}{1+b}} \left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} V_{\widetilde{s}}^{\frac{b}{1+b}},$$

$$- \|\widetilde{e}_{1}\|^{\frac{3+5b}{2}} - \|\widetilde{e}_{2}\|^{\frac{3+5b}{1+b}} - V_{\widetilde{z}}^{\frac{3+5b}{4+4b}} \leq -\left(\frac{3}{2}\right)^{\frac{2}{3}} \|\widetilde{e}_{1}\|^{\frac{3+5b}{3}} (3)^{\frac{1}{3}}\|\widetilde{e}_{2}\|^{\frac{3+5b}{3+3b}} - V_{\widetilde{z}}^{\frac{3+5b}{4+4b}} \\ \leq -\left(\left(\frac{3+5b}{3+3b}\right)\left(\frac{3}{2}\right)^{\frac{2}{3}} (3)^{\frac{1}{3}}\right)^{\frac{3+3b}{3+5b}} \|\widetilde{e}_{1}\|^{1+b}\|\widetilde{e}_{2}\|\left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} V_{\widetilde{z}}^{\frac{b}{2+2b}}$$

$$- \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{2}} - \left\|\widetilde{e}_{2}\right\|^{\frac{3+5b}{1+b}} - V_{z}^{\frac{3+5b}{4+4b}} \leq -\left(\frac{3+3b}{2}\right)^{\frac{2}{3+3b}} \left\|\widetilde{e}_{1}\right\|^{\frac{3+5b}{3+3b}} \left(\frac{3+3b}{1+3b}\right)^{\frac{1+3b}{3+3b}} \left\|\widetilde{e}_{2}\right\|^{\left(\frac{(3+5b)(1+3b)}{(1+b)(3+3b)}\right)} - V_{z}^{\frac{3+5b}{4+4b}} \leq -\left(\left(\frac{3+5b}{3+3b}\right)^{\frac{2}{3+3b}} \left(\frac{3+3b}{2}\right)^{\frac{2}{3+3b}} \left(\frac{3+3b}{1+3b}\right)^{\frac{1+3b}{3+5b}} \left\|\widetilde{e}_{1}\right\| \left\|\widetilde{e}_{2}\right\|^{\frac{1+3b}{1+b}} \left(\frac{3+5b}{2b}\right)^{\frac{2b}{3+5b}} V_{z}^{\frac{b}{2+2b}}$$

thus if the following inequalities

$$\min \begin{cases} \lambda_{8} \underline{l_{1}} \ \underline{l_{2}}^{2}, \\ \lambda_{9} \ \underline{l_{1}}^{\frac{1}{1+b}} \ \underline{l_{2}}^{\frac{2}{1+b}}, \\ \overline{l_{7}(\mu)}^{\frac{3+5b}{2+2b}}, \\ \lambda_{10} \ \underline{l_{1}}^{\frac{1}{1+b}} \ \underline{l_{2}}^{\frac{2}{1+b}}, \\ \lambda_{10} \ \underline{l_{1}}^{\frac{1}{1+b}} \ \underline{l_{2}}^{\frac{2}{1+b}}, \\ \overline{l_{7}(\mu)}^{\frac{3+5b}{2+2b}}, \\ \lambda_{17} \ \underline{l_{1}}^{\frac{2+2b}{1+3b}} \ \underline{l_{2}}^{\frac{1-b}{1+3b}} \\ \underline{l_{1}}^{\frac{1-b}{2}} \ \underline{l_{2}}^{\frac{1-b}{1+3b}} \\ \underline{l_{1}}^{\frac{2+2b}{1+b}} \ \underline{l_{2}}^{\frac{1-b}{1+3b}}, \\ \lambda_{16} \ \underline{l_{7}(\mu)}^{\frac{3+5b}{3+3b}}, \\ \lambda_{17} \ \underline{l_{7}(\mu)}^{\frac{3+5b}{1+b}} \ \underline{l_{2}}^{\frac{1-b}{1+b}} \\ \underline{l_{1}}^{\frac{2}{2} \ \underline{l_{2}}^{\frac{1-b}{1+b}}} \\ \underline{l_{1}}^{\frac{2}{2} \ \underline{l_{2}}^{\frac{1-b}{1+b}}}, \\ \lambda_{18} \ \underline{l_{1}}^{\frac{2+2b}{1+b}} \ \underline{l_{2}}^{\frac{1-b}{1+b}} \\ \underline{l_{2}}^{\frac{1-b}{1+b}} \\ \underline{l_{2}}^{\frac{1-b}{1+b}} \ \underline{l_{2}}^{\frac{1-b}{1+b}} \\ \underline{l_{2}}^{\frac{1-$$

where

$$\lambda_8 = \left(\frac{3+5b}{24+24b}\right) \left(\frac{1}{n^{\frac{1+5b}{2}}}\right) \left(\frac{3+5b}{80+400b} \left(\frac{1}{n^{\frac{2+4b}{1+b}}}\right)\right)^{\frac{1+5b}{2}} \left(\frac{1+b}{2^{\frac{1-b}{1+b}} (2+4b)\overline{m}_I \ \bar{k} \ n^{\frac{1}{1+b}}}\right)^{\frac{3+5b}{2}},$$

$$\begin{split} \lambda_{9} &= \left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{2+2b}} \left(\frac{1}{8n^{\frac{1+5b}{2}}}\right)^{\frac{1}{1+b}} \left(\frac{3+3b}{(80+240b)n^{\frac{2+4b}{1+b}}}\right)^{\frac{1+3b}{2+2b}} \\ &\times \left(\left(\frac{(3+5b)k}{16bn^{\frac{2b}{1+b}}}\right) \left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{1+3b}{1+b}}\right)^{\frac{1+3b}{1+b}} \left(\frac{\sqrt{m}}{\sqrt{2}}\right)^{\frac{2b}{1+b}} \left(\frac{1+b}{\sqrt{n}(2+4b)}\right)^{\frac{3+5b}{2+2b}}, \end{split}$$

$$\begin{split} \lambda_{10} = & \left(\frac{3+5b}{3+3b}\right)^{\frac{3+3b}{2+2b}} \left(\frac{1}{8n^{\frac{1+5b}{2}}}\right)^{\frac{2}{2+2b}} \left(\left(\frac{3+3b}{80+240b}\right) \left(\frac{1}{n^{\frac{2+4b}{1+b}}}\right)\right)^{\frac{1+3b}{2+2b}} \\ & \times \left(\frac{3+5b}{2b}\frac{d_0}{16}\right)^{\frac{2b}{2+2b}} \left(\frac{1+b}{2+4b}\right)^{\frac{3+5b}{2+2b}} \left(\frac{\underline{\pi}_i^{\frac{b}{2+2b}}}{n^{\frac{3}{2}}}\right)^{\frac{3+5b}{2+2b}}, \end{split}$$

$$\lambda_{11} = 2^{\frac{4+2b}{1+b}} \overline{m}_I \,\overline{k} \, n^{\frac{4+2b}{2+2b}} n^{\frac{2+4b}{1+b}}, \qquad \lambda_{12} = 80 \left(\frac{2b}{3+5b}\right) \left(\frac{48+48b}{(3+5b)d_0}\right)^{\frac{3+3b}{2b}} n^{\frac{2+4b}{1+b}} n^{\frac{3+5b}{8b}} d_2^{\frac{3+5b}{2b}},$$

$$\begin{split} \lambda_{13} &= 80 \Biggl(2^{\frac{3+5b}{2b}} n^{\frac{2+4b}{1+b}} \Biggr) \Biggl(\frac{2b}{3+5b} \Biggr) \Biggl(\Biggl(\frac{24+24b}{3+5b} \Biggr) \Biggl(\frac{n^{\frac{2b}{1+b}}}{\underline{k}} \Biggr) \Biggl(\frac{\sqrt{m}}{\sqrt{2}} \Biggr)^{\frac{1+3b}{1+b}} \Biggr)^{\frac{3+3b}{2b}} \\ &\times \Biggl(\overline{m} \overline{k}_2 n^{\frac{1-b}{2+2b}} + 2^{\frac{1-b}{1+b}} \overline{k} \sqrt{n} \Biggr)^{\frac{3+5b}{2b}} \Biggl(\frac{\sqrt{2}}{\sqrt{\underline{m}}} \Biggr)^{\frac{3+5b}{2b}}, \end{split}$$

$$\lambda_{14} = 80 \left(\frac{3+3b}{3+5b}\right) \left(\left(\frac{16b}{3+5b}\right) \left(\frac{n^{\frac{2b}{1+b}}}{\underline{k}}\right) \left(\frac{\sqrt{\overline{m}}}{\sqrt{2}}\right)^{\frac{1+3b}{1+b}} \right)^{\frac{2b}{3+3b}} n^{\frac{2+4b}{1+b}} \left(n^{\frac{3}{2}} \left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right)^{\frac{2b}{1+b}}\right)^{\frac{3+5b}{3+3b}},$$

$$\lambda_{15} = 80 \left(\frac{1+3b}{3+5b}\right) n^{\frac{2+4b}{1+b}} \left(\left(\frac{24+24b}{3+5b}\right) (1+b) \left(n^{\frac{1+5b}{2}}\right) \right)^{\frac{2+2b}{1+3b}} \left(\left(\frac{2\sqrt{n}}{1+b}\right) 2^{\frac{1-b}{1+b}} \overline{m}_I \ \bar{k} \ n^{\frac{1-b}{2+2b}} \right)^{\frac{3+5b}{1+3b}},$$

$$\lambda_{16} = 80n^{\frac{2+4b}{1+b}} \left(\frac{3+3b}{3+5b}\right) \left(\frac{32b}{d_0(3+5b)}\right)^{\frac{2b}{3+3b}} \left(\frac{n^{\frac{5}{2}}}{\underline{\pi}_i^{\frac{b}{2+2b}}}\right)^{\frac{3+5b}{3+3b}},$$

$$\begin{split} \lambda_{17} &= \left(\frac{24+24b}{3}\right)^2 \left(\frac{80}{3}\right) n^{\frac{2+4b}{1+b}} n^{1+5b} \left(\frac{3+3b}{3+5b}\right)^3 \\ &\times \left(\left(\frac{16b}{3+5b}\right) \left(\frac{n^{\frac{2b}{1+b}}}{\underline{k}}\right) \left(\frac{\sqrt{m}}{\sqrt{2}}\right)^{\frac{1+3b}{1+b}}\right)^{\frac{2b}{1+b}} \left(\left(\frac{\sqrt{2}}{\sqrt{m}}\right)^{\frac{2b}{1+b}} \left(\frac{2\sqrt{n}}{1+b}\right)\right)^{\frac{3+5b}{1+b}}, \\ \lambda_{18} &= \left(\frac{80}{3}\right) \left(\frac{32b}{(3+5b)d_0}\right)^{\frac{2b}{1+b}} \left(\frac{3+3b}{3+5b}\right)^{\frac{3+3b}{1+b}} n^{\frac{2+4b}{1+b}} \left((8+8b)n^{\frac{1+5b}{2}}\right)^{\frac{2+2b}{1+b}} \left(\frac{2\sqrt{n}}{1+b}\right)^{\frac{3+5b}{1+b}} \left(\frac{n}{\underline{\pi}_i^{\frac{b}{2+2b}}}\right)^{\frac{3+5b}{1+b}}, \end{split}$$

hold then the function \dot{V}_2 is negative definite. Note that the inequalities (4 - 16) is feasible with respect to l_1, l_2 . Thus, for a given K_1, K_2 , and K, there exists a sufficiently large L_1, L_2 , with the ratio,

$$\left(\frac{\bar{l}_2}{\bar{l}_2}\right) = \text{constant}$$

remaining constant, such that a positive $r_{\tilde{e}} > 0$ exists that satisfy all the above conditions.

Then, we have,

$$\begin{split} \dot{V} \stackrel{a.e.}{\leq} &- \left(\frac{\underline{k}}{n^{\frac{2b}{1+b}}}\right) \left(\frac{\sqrt{2}}{\sqrt{\overline{m}}}\right)^{\frac{1+3b}{1+b}} V_{\tilde{s}}^{\frac{3+5b}{2+2b}} - \underline{f} \left(\frac{4}{\overline{m}}\right) V_{\tilde{s}}^{2} - \frac{d_{0}}{2} V_{\tilde{z}}^{\frac{3+5b}{4+4b}} - \left(\frac{1}{2}\right) \left(\frac{\underline{l}_{1}}{1+b}\right) \left(\frac{1}{n^{\frac{1+5b}{2}}}\right) \|\widetilde{e}_{1}\|^{\frac{3+5b}{2}} \\ &- \left(\frac{r_{\tilde{e}}}{4}\right) \left(\frac{1}{n^{\frac{2+4b}{1+b}}}\right) \|\widetilde{e}_{2}\|^{\frac{3+5b}{1+b}} + 2\left[\alpha(\|\eta(t)\|)\gamma_{5}(\mu) + \|\Lambda G(q_{0})\|\right] \left(\frac{\sqrt{2}}{\sqrt{\underline{m}}}\right) V_{\tilde{s}}^{\frac{3}{2}} \\ &+ \left[\alpha_{2}(\|\eta(t)\|) + \overline{m}_{I}\|\Lambda G(q_{0})\|\right] \times \left(\frac{2\overline{l}_{2}}{1+b}\sqrt{n}\|\widetilde{e}_{1}\|^{1+b}\|\widetilde{e}_{2}\| + n^{\frac{3}{2}}\|\widetilde{e}_{2}\|^{3} + r_{\tilde{e}}\left(\frac{2+4b}{1+b}\right)\sqrt{n}\|\widetilde{e}_{1}\|\|\widetilde{e}_{2}\|^{\frac{1+3b}{1+b}} \right) \end{split}$$

Note that from lemma 2.1, we have

$$\left(\frac{2\sqrt{n}}{1+b}\right)\overline{l_2}\left\|\widetilde{e}_1\right\|^{1+b}\left\|\widetilde{e}_2\right\|_{*}+n^{\frac{3}{2}}\left\|\widetilde{e}_2\right\|^{3}+r_{\widetilde{e}}\left(\frac{2+4b}{1+b}\right)\sqrt{n}\left\|\widetilde{e}_1\right\|\left\|\widetilde{e}_2\right\|^{\frac{1+3b}{1+b}}\leq\pi_3\left(\left\|\widetilde{e}_1\right\|^{\frac{3+3b}{2}}+\left\|\widetilde{e}_2\right\|^{3}\right)$$

where

$$\pi_{3} = \max \begin{cases} \left(\left(\frac{2\sqrt{n}\,\overline{l_{2}}}{1+b}\right) \left(\frac{2}{3}\right) + r_{\tilde{e}}\,\sqrt{n} \left(\frac{2+4b}{1+b}\right) \left(\frac{2}{3+3b}\right) \right), \\ \left(\left(\frac{2\sqrt{n}\,\overline{l_{2}}}{1+b}\right) \left(\frac{1}{3}\right) + n^{\frac{3}{2}} + r_{\tilde{e}}\,\sqrt{n} \left(\frac{2+4b}{1+b}\right) \left(\frac{1+3b}{3+3b}\right) \right) \end{cases}$$

Next, note that (see Section 2.4 of chapter 2) we have the following bounds on the Lyapunov function, $W_{\tilde{e}}$,

$$\underline{\pi}_{2} \sum_{i=1}^{n} \left(\left| \widetilde{e}_{1i} \right|^{2+2b} + \left| \widetilde{e}_{2i} \right|^{4} \right) \leq W_{\widetilde{e}} \left(\widetilde{e}_{1}, \widetilde{e}_{2} \right) \leq \overline{\pi}_{2} \sum_{i=1}^{n} \left(\left| \widetilde{e}_{1i} \right|^{2+2b} + \left| \widetilde{e}_{2i} \right|^{4} \right)$$

where

$$\underline{\pi}_2 = \min\left\{\frac{1}{8}, \frac{1}{2}\left(\frac{l_2}{1+b}\right)^2\right\},\$$

$$\overline{\pi}_2 = \max\left\{\left(\frac{\overline{l_2}}{1+b}\right)^2 + \left(\frac{l_{2i}}{2+2b}\right) + \frac{r_{\tilde{e}}}{2+2b}, \frac{1}{4} + \left(\frac{\overline{l_2}}{2+2b}\right) + r_{\tilde{e}}\left(\frac{1+2b}{2+2b}\right)\right\},\$$

Using lemma 2.2,

$$\sum_{i=1}^{n} |\widetilde{e}_{1i}|^{2+2b} + \sum_{i=1}^{n} |\widetilde{e}_{2i}|^{4} = \sum_{i=1}^{n} \left(|\widetilde{e}_{1i}|^{2} \right)^{\frac{2+2b}{2}} + \sum_{i=1}^{n} \left(|\widetilde{e}_{2i}|^{2} \right)^{\frac{4}{2}}$$

$$\geq \frac{1}{n^{b}} \left(\sum_{i=1}^{n} \left(|\widetilde{e}_{1i}|^{2} \right) \right)^{\frac{2+2b}{2}} + \frac{1}{n} \left(\sum_{i=1}^{n} \left(|\widetilde{e}_{2i}|^{2} \right) \right)^{\frac{4}{2}}, \quad \text{using lemma 2.2,}$$

$$= \frac{1}{n^{b}} ||\widetilde{e}_{1}||^{2+2b} + \frac{1}{n} ||\widetilde{e}_{2}||^{4}$$

$$\geq \frac{1}{n} \left(||\widetilde{e}_{1}||^{2+2b} + ||\widetilde{e}_{2}||^{4} \right), \text{ since } \min \left\{ \frac{1}{n^{b}}, \frac{1}{n} \right\} = \frac{1}{n} \text{ for } 0 < b \le 1 \text{ and } n \ge 1,$$

and

$$\sum_{i=1}^{n} |\widetilde{e}_{1i}|^{2+2b} + \sum_{i=1}^{n} |\widetilde{e}_{2i}|^{4} = \sum_{i=1}^{n} \left(|\widetilde{e}_{1i}|^{2} \right)^{\frac{2+2b}{2}} + \sum_{i=1}^{n} \left(|\widetilde{e}_{2i}|^{2} \right)^{\frac{4}{2}}$$
$$\leq \left(\sum_{i=1}^{n} |\widetilde{e}_{1i}|^{2} \right)^{\frac{2+2b}{2}} + \left(\sum_{i=1}^{n} |\widetilde{e}_{2i}|^{2} \right)^{\frac{4}{2}}, \text{ using lemma 2.2,}$$
$$= \left\| \widetilde{e}_{1} \right\|^{2+2b} + \left\| \widetilde{e}_{2} \right\|^{4}$$

Hence,

$$\frac{\underline{\pi}_{2}}{n} \left(\left\| \widetilde{e}_{1} \right\|^{2+2b} + \left\| \widetilde{e}_{2} \right\|^{4} \right) \leq W_{\widetilde{e}} \left(\widetilde{e}_{1}, \widetilde{e}_{2} \right) \leq \overline{\pi}_{2} \left(\left\| \widetilde{e}_{1} \right\|^{2+2b} + \left\| \widetilde{e}_{2} \right\|^{4} \right)$$

Next, note the following inequalities (using lemma 2.2):

$$\begin{split} \|\widetilde{e}_{1}\|^{\frac{3+3b}{2}} + \|\widetilde{e}_{2}\|^{3} &= \left(\|\widetilde{e}_{1}\|^{2+2b} \right)^{\frac{3}{4}} + \left(\|\widetilde{e}_{2}\|^{4} \right)^{\frac{3}{4}} \\ &\leq 2^{\frac{1}{4}} \left(\|\widetilde{e}_{1}\|^{2+2b} + \|\widetilde{e}_{2}\|^{4} \right)^{\frac{3}{4}} \\ &\leq 2^{\frac{1}{4}} \left(\frac{n}{\underline{\pi}_{2}} \right)^{\frac{3}{4}} W_{\widetilde{e}}^{\frac{3}{4}} \\ \|\widetilde{e}_{1}\|^{\frac{3+5b}{2}} + \|\widetilde{e}_{2}\|^{\frac{3+5b}{1+b}} &= \left(\|\widetilde{e}_{1}\|^{2+2b} \right)^{\frac{3+5b}{4+4b}} + \left(\|\widetilde{e}_{2}\|^{4} \right)^{\frac{3+5b}{4+4b}} \\ &\geq \left(\|\widetilde{e}_{1}\|^{2+2b} + \|\widetilde{e}_{2}\|^{4} \right)^{\frac{3+5b}{4+4b}}, \text{ using lemma 2.2, since } \frac{3+5b}{4+4b} \leq 1, \text{ for } 0 < b \leq 1, \\ &\geq \left(\frac{1}{\overline{\pi}_{2}} \right)^{\frac{3+5b}{4+4b}} W_{\widetilde{e}}^{\frac{3+5b}{4+4b}} \end{split}$$

Thus, the time derivative of the Lyapunov function becomes,

$$\begin{split} \dot{V} \stackrel{\text{a.e.}}{\leq} &- \pi_4 \bigg(V_{\tilde{s}}^{\frac{3+5b}{2+2b}} + V_{\tilde{z}}^{\frac{3+5b}{4+4b}} + W_{\tilde{e}}^{\frac{3+5b}{4+4b}} \bigg) - \underline{f} \bigg(\frac{4}{\overline{m}} \bigg) V_{\tilde{s}}^2 + \alpha_3 \big(\|\eta(t)\| \big) \bigg(V_{\tilde{s}}^{\frac{3}{2}} + W_{\tilde{e}}^{\frac{3}{4}} \bigg) \\ &+ \pi_5 \big\| \Lambda G(q_0) \big\| \bigg(V_{\tilde{s}}^{\frac{3}{2}} + W_{\tilde{e}}^{\frac{3}{4}} \bigg) \\ &= -\pi_4 \bigg((V_{\tilde{s}}^2)^{\frac{3+5b}{4+4b}} + V_{\tilde{z}}^{\frac{3+5b}{4+4b}} + W_{\tilde{e}}^{\frac{3+5b}{4+4b}} \bigg) - \underline{f} \bigg(\frac{4}{\overline{m}} \bigg) V_{\tilde{s}}^2 + \alpha_3 \big(\|\eta(t)\| \bigg) \bigg((V_{\tilde{s}}^2)^{\frac{3}{4}} + W_{\tilde{e}}^{\frac{3}{4}} \bigg) \\ &+ \pi_5 \big\| \Lambda G(q_0) \big\| \bigg((V_{\tilde{s}}^2)^{\frac{3}{4}} + W_{\tilde{e}}^{\frac{3}{4}} \bigg) \bigg) \\ &\leq -\pi_4 \big(V_{\tilde{s}}^2 + V_{\tilde{z}} + W_{\tilde{e}} \bigg)^{\frac{3+5b}{4+4b}} + 2^{\frac{1}{4}} \alpha_3 \big(\|\eta(t)\| \big) \bigg(V_{\tilde{s}}^2 + W_{\tilde{e}}^{\frac{3}{4}} \bigg)^{\frac{3}{4}} \\ &+ 2^{\frac{1}{4}} \pi_5 \big\| \Lambda G(q_0) \big\| \bigg(V_{\tilde{s}}^2 + W_{\tilde{e}}^{\frac{3}{4}} \bigg)^{\frac{3}{4}} \\ &= -\pi_4 V^{\frac{3+5b}{4+4b}} + 2^{\frac{1}{4}} \alpha_3 \big(\|\eta(t)\| \big) V^{\frac{3}{4}} + 2^{\frac{1}{4}} \pi_5 \big\| \Lambda G(q_0) \big\| V^{\frac{3}{4}} \end{split}$$

where

$$\begin{aligned} \pi_{4} &\coloneqq \min \left\{ \begin{pmatrix} \frac{k}{\frac{2b}{n^{1+b}}} \end{pmatrix} \left(\frac{\sqrt{2}}{\sqrt{m}} \right)^{\frac{1+3b}{1+b}}, \frac{d_{0}}{2}, \\ \left(\frac{1}{\overline{\pi}_{2}} \right)^{\frac{3+5b}{4+4b}} \left(\frac{1}{2} \right) \left(\frac{l_{1}}{\frac{l_{2}}{1+b}} \right) \left(\frac{1}{n^{\frac{1+5b}{2}}} \right), \left(\frac{1}{\overline{\pi}_{2}} \right)^{\frac{3+5b}{4+4b}} \left(\frac{r_{\tilde{e}}}{4} \right) \left(\frac{1}{\frac{2^{2+4b}}{n^{1+b}}} \right) \right\}, \\ \pi_{5} &\coloneqq \max \left\{ \left(\frac{2^{\frac{3}{2}}}{\sqrt{\underline{m}}} \right), \overline{m}_{I} \left(\frac{2^{\frac{1}{4}} \pi_{3} n^{\frac{3}{4}}}{\underline{\pi}_{2}^{\frac{3}{4}}} \right) \right\}, \\ \alpha_{3} \left(\left\| \eta(t) \right\| \right) &\coloneqq \max \left\{ \alpha \left(\left\| \eta(t) \right\| \right) \gamma_{5} \left(\mu \right) \left(\frac{2^{\frac{3}{2}}}{\sqrt{\underline{m}}} \right), \alpha_{2} \left(\left\| \eta(t) \right\| \right) \left(\frac{2^{\frac{1}{4}} \pi_{3} n^{\frac{3}{4}}}{\underline{\pi}_{2}^{\frac{3}{4}}} \right) \right\}, \end{aligned}$$

Note that lemma 2.2 with $\frac{3+5b}{4+4b} \le 1$ and the bounds of the Lyapunov functions,

$$V = \left[V_{\widetilde{s}} \left(\widetilde{s} \right) \right]^2 + V_{\widetilde{z}} \left(\widetilde{z}_1, \widetilde{z}_2 \right) + W_{\widetilde{e}} \left(\widetilde{e}_1, \widetilde{e}_2 \right) \ge \left[V_{\widetilde{s}} \left(\widetilde{s} \right) \right]^2 + W_{\widetilde{e}} \ ,$$

have been employed in the above inequality.

Note that when b = 1, recall that the term

$$\Lambda G(q_0) = \operatorname{diag}\left(-\left|\operatorname{sig}\left(\frac{\widetilde{s}_i}{\overline{s}_i}\right)^{\frac{2b}{1+b}} - \operatorname{sig}\left(\frac{\widetilde{s}_i}{\overline{s}_i} + 1\right)^{\frac{2b}{1+b}}\right| + 1\right) G(q_0) = \operatorname{diag}\left(-\left|-1\right| + 1\right) G(q_0) = 0.$$

Also for the case of 0 < b < 1, consider the case when the final desired position vector q_0 corresponds to the rest position of the manipulator where $G(q_0) = (0, ..., 0)^T$, or the gravitational torque of the manipulator dynamics is absent (i.e. in space where gravity is absent or in a planar horizontal configuration) where G(q) = 0 for $\forall q \in \mathbb{R}^n$, the term $\Lambda G(q_0)$ vanishes to zero. We obtain,

$$\dot{V}^{a.e.}_{\leq -\pi_4} V^{\frac{3+5b}{4+4b}} + 2^{\frac{1}{4}} \alpha_3 (\|\eta(t)\|) V^{\frac{3}{4}}$$

Thus, for $\|\eta(t)\| \to 0$ as $t \to \infty$ (where the desired trajectory approaches a constant final desired position), then we have $V(\tilde{s}, \tilde{z}_1, \tilde{z}_2, \tilde{e}_1, \tilde{e}_2) \to 0$ as $t \to \infty$ as well. Obviously, if $\|\eta(t)\| \equiv 0$ (regulation problem), when b = 1, the system will converge to the equilibrium exponentially, while for 0 < b < 1, the system achieves finite-time regulation, in which the states reach the origin in finite time with the estimate of reaching time as,

$$T(\widetilde{s}_0, \widetilde{z}_{10}, \widetilde{z}_{20}, \widetilde{e}_{10}, \widetilde{e}_{20}) \leq \left(\frac{4+4b}{\pi_4(1-b)}\right) \left[V(\widetilde{s}_0, \widetilde{z}_{10}, \widetilde{z}_{20}, \widetilde{e}_{10}, \widetilde{e}_{20})\right]^{\frac{1-b}{4+4b}}$$

where $(\tilde{s}_0, \tilde{z}_{10}, \tilde{z}_{20}, \tilde{e}_{10}, \tilde{e}_{20})$ are the states of the system when it first enters the region $\Omega_{\tilde{e}}(\mu) \times \Omega_s(\mu) \times \Omega_s(\mu)$.

Remark 4.9. Note that a smaller value of μ can help lower the required observer gains L_1 , and L_2 (see (4 - 16)). This is useful for the case of 0 < b < 1, since both gains L_1 , and L_2 corresponds to the non-Lipschitz component of the observer, while μ can be reduced by increasing the linear observer gain L_3 (see Section 4.1).

Remark 4.10. As per remark 4.5 and 4.8, the "dirty-derivative" inspired linear damping term of the observer, $L_3\tilde{e}_2$ has a synergistic effect on the performance of the observer by allowing the reduction on the level of gains of the non-Lipschitz part of the observer while maintaining special properties of the super-twisting algorithm such as finite-time convergence and exact robustness.

4.3 Numerical Simulations

In this section, numerical simulations on a two-link robot manipulator were carried out to illustrate the results discussed in this chapter. The setups for each simulation are described. Discussion and analysis of the results are presented accordingly.

4.3.1 Simulation Setup

1) Simulation 1:

The same two-link rigid robot manipulator considered in section 3.3 is adopted in simulation. The dynamics of robot manipulator (4 - 3) have the same parameter values as that in section 3.3. The desired trajectory vector and the additive disturbances Coulomb friction vector were defined similarly as well.

The control (4 - 4) parameter values were selected as follows,

$$a = 0.9, K = \begin{bmatrix} 150 & 0 \\ 0 & 15 \end{bmatrix}, K_1 = \begin{bmatrix} 18 & 0 \\ 0 & 18 \end{bmatrix}, K_2 = \begin{bmatrix} 11 & 0 \\ 0 & 11 \end{bmatrix}, b = 0.6$$

The initial conditions of the vector σ were selected as

$$\sigma(0) = \begin{bmatrix} 0 \\ 0 \end{bmatrix}$$

since no velocity measurement is available.

The observer (4 - 6) parameter values were selected as follows,

$$L_1 = L_2 = \begin{bmatrix} 2 & 0 \\ 0 & 2 \end{bmatrix}, L_3 = \begin{bmatrix} 150 & 0 \\ 0 & 150 \end{bmatrix}, p = 0.8$$

The initial conditions of the observer were selected as

$$\hat{e}_1(0) = e_1(0),$$

 $w(0) = -L_3 e_1(0)$

since position measurement is available, so that initially, the observation error

$$\widetilde{e}_{1}(0) = 0,$$

 $\hat{e}_{2}(0) = 0,$
 $\widetilde{e}_{2}(0) = -e_{2}(0)$

Note that the initial conditions for the velocity observation error were selected to be non zero to show the convergence of the observer through simulations, by having the initial conditions for the manipulator,

$$q(0) = \begin{bmatrix} 0 \\ 0 \end{bmatrix}, \dot{q}(0) = \begin{bmatrix} 1.2 \\ 0.5 \end{bmatrix},$$
$$q_d(0) = \begin{bmatrix} \pi \\ 0.5 \end{bmatrix}, \dot{q}_d(0) = \begin{bmatrix} 0.5 \\ 0 \end{bmatrix}$$

2) Simulation 2:

The setup of Simulation 2 is exactly the same as that of Simulation 1 except, in this simulation, only the parameter values of the observer (4 - 6) were changed to examine the effect of removing the linear damping term from the observer as follows,

First case:

$$L_1 = L_2 = \begin{bmatrix} 2 & 0 \\ 0 & 2 \end{bmatrix}, L_3 = \begin{bmatrix} 0 & 0 \\ 0 & 0 \end{bmatrix}, p = 0.8,$$

Second case:

$$L_1 = \begin{bmatrix} 125 & 0 \\ 0 & 125 \end{bmatrix}, L_2 = \begin{bmatrix} 800 & 0 \\ 0 & 800 \end{bmatrix}, L_3 = \begin{bmatrix} 0 & 0 \\ 0 & 0 \end{bmatrix}, p = 0.8,$$

3) Simulation 3:

The setup of Simulation 3 is exactly the same as that of Simulation 1 except the observer parameter p was selected as p = 0.5 to demonstrate the observation errors converge to origin in finite time under uncertainties with

$$L_1 = L_2 = \begin{bmatrix} 2 & 0 \\ 0 & 2 \end{bmatrix}, L_3 = \begin{bmatrix} 150 & 0 \\ 0 & 150 \end{bmatrix}, p = 0.5.$$

For comparison purposes, under the same setup, a simulation with the full-state feedback control (3 - 2) of Chapter 3 were performed with

 $\sigma(0) = -e_2(0)$, so that initially s(0) = 0, i.e. since full-state is available

4.3.2 Results and Discussions

For better visualization of the plots, some figures are shown in two windows; each with different time intervals.



Figure 4.1 Simulation 1. Tracking errors and control input of joint 1 and joint 2.



(a) Position estimate, \hat{e}_{11} for time t = [0, 15] s.

(b) Position observation error, \tilde{e}_{11} for time t =[0, 15] s.



(c) Position estimate, \hat{e}_{12} for time t = [0, 15] s. (d) Position observation error, \tilde{e}_{12} for time t =[0, 15] s.

Figure 4.2 Simulation 1. Position observation errors using observer (4 - 6) of joint 1 and joint

2.



Figure 4.3 Simulation 1. Velocity observation errors using observer (4 - 6) of joint 1 and joint

2.





(a) Position estimate, \hat{e}_{11} for time t = [0, 2] s.

(b) Position estimate, \hat{e}_{12} for time t = [0, 2] s.



(c) Velocity estimate, \hat{e}_{21} for time t = [0, 2] s. (d) Velocity estimate, \hat{e}_{22} for time t = [0, 2] s.

Figure 4.4 Simulation 2. Observation errors without L_3 , small L_1 and L_2 .



Figure 4.5 Simulation 2. Observation errors without L_3 , large L_1 and L_2 .



Figure 4.6 Simulation 3. Observation errors with p = 0.5.



Figure 4.7 Simulation 3. Tracking errors using control (3 - 2), control (4 - 4) with p = 0.5, and control input of joint 1 and 2.

1) Simulation 1:

Figure 4.1 shows the tracking errors and control inputs under the output-feedback control (4 - 4) and the observer (4 - 6). As per Theorem 4.1, for initial conditions inside the region $\Omega(c)$ will enter the region $\Omega(\mu)$ infinite time. In fact, in Figure 4.2 and Figure 4.3, the observation errors converge to a bounded region in about 2.5 seconds. Once the observation errors are sufficiently bounded, boundedness of the other states will follow provided gain *K* satisfy the sufficient conditions of Theorem 4.1. Particularly, the stability is highly dependent on the boundedness of the observation errors. In step 2 of the proof of Theorem 4.1 shows that the boundedness of observation error \tilde{e}_2 affects the ultimate bound of the tracking errors. In order to achieve better observer convergence, a linear damping term $L_3 \tilde{e}_2$ is added. This term is instrumental to achieve semiglobal stability. As a matter of fact, from remark 4.1, increase in L_3 will reduce the ultimate bound on the observation error while at the same time the region $\Omega_{\tilde{e}}(c)$ remains unchanged due to its Lyapunov region $V_{\tilde{e}i}$ that is independent of L_3 .

2) Simulation 2:

To study the impact of the linear damping term $L_3 \tilde{e}_2$, Simulation 2 were performed under the same setup as Simulation 1 by removing the damping term from the observer (4 - 6). Figure 4.4 shows the observer estimates with respect to its actual values with the same small gains of L_1 and L_2 as per Simulation 1. The simulation was stopped at 2 seconds due to instability of the systems. The observation errors were too large for such a small gains to compensate as per the inequality in Theorem 4.2. In fact, to ensure stability the L_1 and L_2 gains have to be increased to a very large value to compensate for the nonlinearities as shown in Figure 4.5. This is in part due to the non-Lipschitz terms of the observer $sig(\tilde{e}_{1i})^p$ and $sig(\tilde{e}_{1i})^{2p-1}$ which are slow to grow when the observation error is from the origin due to $0.5 \le p$ < 1. However, there are some interesting features of the observer, particularly when p = 0.5that enable the observer to be exactly robust to persistent disturbances and ensures convergence in finite time (see section 2.4). This feature will be discussed in the next simulation.

3) Simulation 3:

After showing the importance of $L_3\tilde{e}_2$ in the previous discussions, the exact robustness of the observer when p = 0.5 is shown in Figure 4.6. Note that the observation errors were able to converge in finite time with such small non-Lipschitz gains of 2, is due to the synergistic effect of the linear damping term. As per remark 4.6, after the convergence of the states of the system into the region $\Omega(\mu)$, the nonlinearities and state-dependent disturbances will be smaller. Thus, uncertainties that were initially unable to cope with by the non-Lipschitz gains (see Figure 4.4) are now much smaller when the states are inside $\Omega(\mu)$ thanks to the term, L_3 \tilde{e}_2 as per Theorem 4.1. In fact, when the observation errors converge to zero and stay in it in all future times, the control (4 - 4) becomes that of the full-state feedback of (3 - 2). In fact, from Figure 4.7, it can be observed that the ultimate bound on the tracking errors were the same for both controllers.

4.4 Summary

Trajectory tracking control of robot manipulator without velocity measurement in a semiglobal practical manner is achieved with the results of this chapter. Effectively, the proposed controller is an output feedback version of that develops in Chapter 3. The velocity measurement is substituted with the output of an observer that combines the super-twisting based algorithm develop in Chapter 2 with a damping term that is termed in the literature as "dirty-derivative". Through the non-Lipschitzness of the super-twisting based part of the observer, several finite time properties is produced, such as finite time exact robustness to achieve finite time convergence of observation error and finite time regulation. While the added linear damping term aids in reducing the level of gains of the non-Lipschitz part of the observer while maintaining performance.

Chapter 5: ROBOT MANIPULATOR CONTROL: BOUNDED CONTROL APPROACH WITH FULL-STATE FEEDBACK

In this chapter, a bounded controller is developed for the tracking control of robot manipulator. The proposed controller is based on the non-saturated results in chapter 3. By having an integral of nonlinear function in the controller, a conditional integrator-like behaviour is attained, and is able to achieve global practical stability results for trajectory tracking despite bounded uncertainties and disturbances. Similar performance as the non-saturated controller is obtained when the bounded controller is not saturated.

5.1 Nonlinear Robot Dynamics

The dynamic model of a rigid *n*-link serial robot manipulator, described in Section 3.1,

$$M(q)\ddot{q} + C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D(q,\dot{q},t) = \tau$$
(5 - 1)

where $q, \dot{q}, \ddot{q} \in \mathbb{R}^n$ are, respectively, the position (generalized coordinates), velocity and acceleration vectors, $M(q) \in \mathbb{R}^{n \times n}$ is the inertia matrix, and $C(q, \dot{q})\dot{q}, F\dot{q}, G(q), \tau \in \mathbb{R}^n$ are, respectively, the vectors of Coriolis and centrifugal, viscous friction, gravity, and torque input, with *F* being a constant, positive definite, diagonal (viscous friction coefficient) matrix and $D(q, \dot{q}, t)$ being a bounded disturbances. The properties that the dynamics satisfied can be found in Section 3.1.

In this chapter, let us suppose that the absolute value of each joint input τ_i is constrained to be smaller than a given saturation bound $T_{i, \max} > 0$, i.e., $|\tau_i| \le T_{i, \max}$, $\forall i = 1, ..., n$. The control objective here is to design a robust full state feedback saturated control that ensures the robot configuration vector q tracks a desired trajectory vector, $q_d(t)$ with an ultimately bounded error that can be made as small as required globally even under the presence of bounded disturbances.

The desired trajectory vector, $q_d(t)$ is assumed to be twice continuously differentiable vector-function such that $||q_d(t)||$, $||\dot{q}_d(t)||$, and $||\ddot{q}_d(t)||$ are bounded by *a priori* known constants. While the type of disturbances considered here has to be upper bounded by a constant, due to the limited actuation, (Note that no continuity assumption is made so that discontinuous models of friction may be used in *D*), i.e.

$$||D|| \le \operatorname{sat}_{p_5} \left[p_0 + p_1 ||e_1|| + p_2 ||\dot{q}|| + p_3 ||e_1||^2 + p_4 ||\dot{q}||^2 \right]$$

where $||D|| := \sup \{ ||\varsigma|| : \varsigma \in K[D] \}$, p_0 , p_1 , p_2 , p_3 , p_4 , p_5 are some nonnegative constants, $e_1 = q - q_d \in \mathbb{R}^n$, and $e_2 = \dot{q} - \dot{q}_d \in \mathbb{R}^n$.

5.1.1 Control Development

Let us define the following scalar saturation function as

$$\operatorname{sat}_{\varepsilon}[x] = \begin{cases} x, & \text{if } |x| \leq \varepsilon, \\ \varepsilon \operatorname{sign}[x], & \text{if } |x| > \varepsilon, \end{cases} \quad \forall x \in \mathbb{R},$$

if $\varepsilon_i = 1$, it will become the standard saturation function (see p.19 of [71]) and the subscript will be omitted, while the vector saturation function, given a set of positive number $\varepsilon \in \{\varepsilon_l, ..., \varepsilon_n\}$, define

$$\operatorname{sat}_{\varepsilon}[x] = [\operatorname{sat}_{\varepsilon_1}[x_1], \ldots, \operatorname{sat}_{\varepsilon_n}[x_n]], \forall x \in \mathbb{R}^n$$

and recall the following notation (see Section 3.1.2)

$$\operatorname{sig}(x)^{a} = \left[\left| x_{1} \right|^{a} \operatorname{sign}(x_{1}), \dots, \left| x_{n} \right|^{a} \operatorname{sign}(x_{n}) \right]^{T}, \forall x \in \mathbb{R}^{n}.$$

Under the above assumptions, the proposed control law is of the form

$$\tau = -K \operatorname{sat}\left[\operatorname{sig}(\mu^{-1}s)^{a}\right], \ s \in \mathbb{R}^{n},$$
(5 - 2)

where K and μ are positive definite diagonal matrix, i.e. $K = \text{diag}\{k_i\}_{i=1}^n$, with $k_i > 0$, $\mu = \text{diag}\{\mu_i\}_{i=1}^n$, with $\mu_i > 0$, $\forall i = 1, ..., n \ a \ge 0$ constant, and $s \in \mathbb{R}^n$

i.e.
$$\tau_i = -k_i \operatorname{sat}\left[\frac{|s_i|^a}{\mu_i^a}\operatorname{sign}(s_i)\right]$$
 for $i = 1 \dots n$.

When a = 0, the control becomes a discontinuous control law,

$$\tau = -K \operatorname{sign}(s), \ s \in \mathbb{R}^n$$
.

Note that the control is bounded, i.e. such that $|\tau_i(t)| \le k_i$, for $i = 1, ..., n, \forall t \ge 0$.

The *s* is the desired error dynamics defined as $s = e_2 + \sigma$, with

$$\dot{\sigma} = -K_a \sigma + K_a \mu \operatorname{sat}\left[\mu^{-1}s\right] - K_a \operatorname{sat}_{\left(\frac{\varepsilon_2}{k_2}\right)^{\frac{1+b}{2b}}}\left[e_2\right] + \operatorname{sat}_{\varepsilon_2}\left[K_2 \operatorname{sig}(e_2)^{\frac{2b}{1+b}}\right] + \operatorname{sat}_{\varepsilon_1}\left[K_1 \operatorname{sig}(e_1)^b\right] \quad (5-3)$$

where K_a , K_1 , and K_2 are positive definite diagonal matrices, i.e. $K_a = \text{diag}\{k_{ai}\}_{i=1}^n$, with

$$k_{ai} > 0, K_{1} = \operatorname{diag}\{k_{1i}\}_{i=1}^{n}, \text{ with } k_{1i} > 0, K_{2} = \operatorname{diag}\{k_{2i}\}_{i=1}^{n}, \text{ with } k_{2i} > 0, k_{2} \in \{k_{21}, \dots, k_{2n}\},\$$

$$\varepsilon_{1} \in \{\varepsilon_{11}, \dots, \varepsilon_{1n}\}, \text{ with } \varepsilon_{1i} > 0, \quad \varepsilon_{2} \in \{\varepsilon_{21}, \dots, \varepsilon_{2n}\}, \text{ with } \varepsilon_{2i} > 0, \forall i = 1, \dots, n, \text{ and } b \ge 0$$

constant, with
$$\varepsilon_{1i} > k_{ai} \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}} + \varepsilon_{2i}$$
,

i.e.

$$\dot{\sigma}_{i} = -k_{ai}\sigma_{i} + k_{ai}\mu_{i}\operatorname{sat}\left[\frac{s_{i}}{\mu_{i}}\right] - k_{ai}\operatorname{sat}\left[\frac{\varepsilon_{2i}}{k_{2i}}\right]^{\frac{1+b}{2b}}\left[e_{2i}\right]$$
$$+ \operatorname{sat}_{\varepsilon_{2i}}\left[k_{2i}|e_{2i}|^{\frac{2b}{1+b}}\operatorname{sign}(e_{2i})\right] + \operatorname{sat}_{\varepsilon_{1}}\left[k_{1i}|e_{1i}|^{b}\operatorname{sign}(e_{1i})\right]$$

for i = 1 ... n.

Note that when b = 0, let the element of ε_1 equal to that of K_1 , and ε_2 to that of K_2 , i.e. $\varepsilon_{1n} = k_{1n}, \forall i \in N$ and $\varepsilon_{2n} = k_{2n}, \forall i \in N$, i.e.

$$\dot{\sigma}_i = -k_{ai}\sigma_i + k_{ai}\mu_i \operatorname{sat}\left[\frac{s_i}{\mu_i}\right] - k_{ai}\operatorname{sat}\left[e_{2i}\right] + k_{2i}\operatorname{sign}(e_{2i}) + k_{1i}\operatorname{sign}(e_{1i})$$

Then, the closed-loop system of (5 - 1), (5 - 2), and (5 - 3) can be written as

$$\dot{\sigma} = -K_a \sigma + K_a \mu \operatorname{sat}[\mu^{-1}s] - K_a \operatorname{sat}_{\left(\frac{\varepsilon_2}{k_2}\right)^{\frac{1+b}{2b}}} [e_2] + \operatorname{sat}_{\varepsilon_2} \left[K_2 \operatorname{sig}(e_2)^{\frac{2b}{1+b}}\right] + \operatorname{sat}_{\varepsilon_1} \left[K_1 \operatorname{sig}(e_1)^b\right],$$

$$\dot{e}_1 = e_2,$$

$$\dot{e}_2 = -M^{-1}(q)K \operatorname{sat}\left[\operatorname{sig}(\mu^{-1}s)^a\right] - M^{-1}(q)(C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D) - \ddot{q}_d,$$

To rewrite the closed-loop system in a form more convenient for analysis, let us define the change of variable $v_1 = e_1$, and $v_2 = -\sigma$, and we obtain the following form of closed-loop system,

$$\dot{v}_{1} = v_{2} + s,$$

$$\dot{v}_{2} = -K_{a}v_{2} - K_{a}\mu \operatorname{sat}[\mu^{-1}s] + K_{a} \operatorname{sat}_{\left(\frac{e_{2}}{k_{2}}\right)^{\frac{1+b}{2b}}}[v_{2} + s]$$

$$-\operatorname{sat}_{\varepsilon_{2}}\left[K_{2}\operatorname{sig}(v_{2} + s)^{\frac{2b}{1+b}}\right] - \operatorname{sat}_{\varepsilon_{1}}\left[K_{1}\operatorname{sig}(v_{1})^{b}\right],$$

$$\dot{s} = -M^{-1}(z_{1} + q_{d})K \operatorname{sat}\left[\operatorname{sig}(\mu^{-1}s)^{a}\right] - M^{-1}(q)(C(q, \dot{q})\dot{q} + F\dot{q} + G(q) + D) - \ddot{q}_{d} + \dot{\sigma},$$

(5 - 4)

5.1.2 Stability Analysis

Theorem 5.1: Consider the robot dynamics (5 - 1), with the bounded controller given by (5 - 2), global practical trajectory tracking of the desired trajectory q_d can be assured, provided that the gain, *K* is sufficiently large up to the saturation bound $T_{i, \max}$, $\forall i = 1,...,n$, and the desired trajectory sufficiently slow.

Proof of Theorem 5.1: The stability analysis proceeds in three steps.

1. First we will show the boundedness of σ , by the following Lyapunov function

$$V_{\sigma} = \frac{1}{2}\sigma_i^2, \quad \forall i = 1, \dots, n,$$

the time derivative of the Lyapunov function, along its solution of (5 - 3), gives rise to:

$$\begin{split} \dot{V}_{\sigma} &= -k_{ai}\sigma_{i}^{2} + \sigma_{i}k_{ai}\mu_{i}\operatorname{sat}\left(\frac{s_{i}}{\mu_{i}}\right) - \sigma_{i}k_{ai}\operatorname{sat}_{\left(\frac{s_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}}(e_{2i}) \\ &+ \sigma_{i}\operatorname{sat}_{\varepsilon_{2i}}\left(k_{2i}|e_{2i}|^{\frac{2b}{1+b}}\operatorname{sign}(e_{2i})\right) + \sigma_{i}\operatorname{sat}_{\varepsilon_{1i}}\left(k_{1i}|e_{1i}|^{b}\operatorname{sign}(e_{1i})\right) \\ &\leq -\frac{k_{ai}}{2}\sigma_{i}^{2} - \frac{k_{ai}}{2}\sigma_{i}^{2} + |\sigma_{i}\left(k_{ai}\mu_{i} + k_{ai}\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}} + \varepsilon_{2i} + \varepsilon_{1i}\right) \\ &\therefore |\sigma_{i}| \leq 2\left(\mu_{i} + \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}} + \frac{\varepsilon_{2i}}{k_{ai}} + \frac{\varepsilon_{1i}}{k_{ai}}\right), \end{split}$$

Since μ_i , ε_{1i} , $\varepsilon_{2i} > 0$ constants, $|\sigma_i|$ is bounded. Also, from (5 - 3), we can see that $|\sigma_i|$ is bounded as well.

2. Now, we proceed to show the boundedness of *s* by the following Lyapunov function

$$V_s = \frac{1}{2} s^{\mathrm{T}} M(.) s \, .$$

Note that $\dot{q} = s + (\dot{q}_d - \sigma)$,

The *s* dynamics is described by,

$$\dot{s} = -M^{-1}(q)Ksat\left[sig(\mu^{-1}s)^{a}\right] - M^{-1}(q)(C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D) - \ddot{q}_{d} + \dot{\sigma}$$
(5 - 5)

Since the differential equation (5 - 5) has discontinuous right-hand side, i.e. when a = 0 or since no continuity assumption is made on *D*, its solutions are understood in the sense of Filippov (see definition 2.1), and in accordance to lemma 2.4, the time derivative of the Lyapunov function, V_s along the dynamics (5 - 5) for all $a \ge 0$:

$$\begin{split} \dot{V}_{s} \stackrel{a.e}{\in} \dot{\tilde{V}}_{s} \\ \dot{\tilde{V}}_{s} \stackrel{a.e}{\in} \dot{\tilde{V}}_{s} \\ \dot{\tilde{V}}_{s} &= \bigcap_{\xi \in \partial V_{s}} \xi^{\mathrm{T}} \binom{\boldsymbol{K}[f](s)}{1} = \nabla V_{s}^{\mathrm{T}} \boldsymbol{K}[f](s) + \frac{1}{2} s^{\mathrm{T}} \dot{M}(\cdot) s \\ &\subset \left[s^{\mathrm{T}} M(\cdot)\right] \boldsymbol{K} \left[M^{-1}(q) \tau - M^{-1}(q) (C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D) - \ddot{q}_{d} + \dot{\sigma}\right] + \frac{1}{2} s^{\mathrm{T}} \dot{M}(\cdot) s \\ &= s^{\mathrm{T}} \boldsymbol{K}[\tau] + s^{\mathrm{T}} (-G(q) - M(q) \ddot{q}_{d} + M(q) \dot{\sigma}) - s^{\mathrm{T}} \boldsymbol{K}[D] - s^{\mathrm{T}} F \dot{q} - s^{\mathrm{T}} C(q, \dot{q}) \dot{q} \\ &+ s^{\mathrm{T}} C(q, \dot{q}) s \\ &= s^{\mathrm{T}} \boldsymbol{K}[\tau] + s^{\mathrm{T}} (-G(q) - M(q) \ddot{q}_{d} + M(q) \dot{\sigma}) - s^{\mathrm{T}} \boldsymbol{K}[D] - s^{\mathrm{T}} F (\dot{q}_{d} - \sigma) \\ &- s^{\mathrm{T}} [C(q, (\dot{q}_{d} - \sigma))] (\dot{q}_{d} - \sigma) - s^{\mathrm{T}} F s - s^{\mathrm{T}} [C(q, (\dot{q}_{d} - \sigma))]] s \\ &\leq s^{\mathrm{T}} \boldsymbol{K}[\tau] + s^{\mathrm{T}} \Delta - \|s\|^{2} (\underline{f} - C_{m}\|(\dot{q}_{d} - \sigma)\|), \end{split}$$

with the vector

$$\Delta = -G(q) - M(q)\ddot{q}_d + M(q)\dot{\sigma} - K[D] - F(\dot{q}_d - \sigma) - C(q, (\dot{q}_d - \sigma))(\dot{q}_d - \sigma),$$

Remark 5.1. Observe that for a > 0, $s^{T} \mathbf{K}[\tau] = -s^{T} K \operatorname{sat}[\mu^{-1} \operatorname{sig}(s)^{a}]$, while for a = 0, using the following notation:

$$N^{+}(s) = \{i \in \{1, ..., n\} : s_{i} \neq 0\}, N^{0}(s) = \{i \in \{1, ..., n\} : s_{i} = 0\},\$$

$$s^{T} \mathbf{K}[\tau] = -s^{T} K \mathbf{K}[sign(s)]$$

$$= -\sum_{i \in N^{+}(s)} k_{i} |s_{i}| - \sum_{i \in N^{0}(s)} k_{i}(0) \times [-1, +1]$$

$$= -\sum_{i \in N^{+}(s)} k_{i} |s_{i}| - 0$$

$$= -\sum_{i = i}^{n} k_{i} |s_{i}|$$
Also, let us define the set of real number, $N \in \{1, ..., n\}$, and $|\Delta_i| := \sup \{ \varepsilon_i | : \varepsilon_i \in \mathbf{K}[\Delta_i] \}, \forall i \in N$, with Δ_i as elements of Δ , where

$$\boldsymbol{K}[\Delta] \subset \begin{bmatrix} \boldsymbol{K}[\Delta_1] \\ \vdots \\ \boldsymbol{K}[\Delta_n] \end{bmatrix}.$$

Since \dot{q}_d , \ddot{q}_d , are bounded from the assumptions and σ , σ are bounded (see step (1)), the vector Δ is bounded as well. Hence from [143], the multi-valued function $K[\Delta]$ is bounded as well. Thus, if

$$\underline{f} > C_m \| (\dot{q}_d - \sigma) \|,$$

which is possible by selecting appropriate desired trajectory and the desired error dynamics, we have:

For a = 0:

$$\dot{V}_{s} \stackrel{a.e.}{\leq} -\sum_{i=i}^{n} k_{i} |s_{i}| + s^{\mathrm{T}} \Delta$$

$$\leq -\sum_{i=i}^{n} k_{i} |s_{i}| + \sum_{i=1}^{n} |s_{i}| |\Delta_{i}|$$

$$= -\sum_{i=i}^{n} |s_{i}| (k_{i} - |\Delta_{i}|)$$

$$\therefore \dot{V}_{s} \stackrel{a.e.}{\leq} 0 \quad \text{for } \forall |s_{i}| \neq 0$$

if the control gain is selected as $k_i > |\Delta_i|$. Thus, the state *s* will reach zero in finite time and stay there for all future times.

While for the case of a > 0:

$$\dot{V}_s \stackrel{a.e.}{\leq} - s^{\mathrm{T}} K \mathrm{sat} \bigg[\mathrm{sig} (\mu^{-1} s)^a \bigg] + s^{\mathrm{T}} \Delta$$

and it can be rewritten as

$$\dot{V_s} \stackrel{a.e.}{\leq} \sum_{i \in N} -k_i |s_i| \operatorname{sat}\left[\frac{|s_i|^a}{\mu_i^a}\right] + |\Delta_i| |s_i|$$
$$= -k_j |s_j| \operatorname{sat}\left[\frac{|s_j|^a}{\mu_j^a}\right] + |\Delta_j| |s_j| + \left(\sum_{i \in N \setminus j} |s_i| \left(-k_i \operatorname{sat}\left[\frac{|s_i|^a}{\mu_i^a}\right] + |\Delta_i|\right)\right)\right)$$

Remark 5.2. Note that the last equality has single out one of its element, $j \in N$ from the summation for ease of analysis.

Due to the saturated nature of the control, it is necessary for stability that $k_i > |\Delta_i|, \forall i \in N$. Then the maximum of the last term in the above inequality occurs when $|s_i| < \mu_i$. In particular, by taking the derivative of the term w.r.t. $|s_i|$ and equating it to zero, the maximum is found to occur at

$$|s_i| = \left| \frac{|\Delta_i|}{(1+a)k_i} \right|^{\frac{1}{a}} \mu_i, \quad \forall i \in N \setminus j$$

and the corresponding maximum,

$$\left(\frac{a}{\left(1+a\right)^{\frac{1+a}{a}}}\right)\left(\frac{\left|\Delta_{i}\right|}{k_{i}}\right)^{\frac{1}{a}}\mu_{i}\left|\Delta_{i}\right|, \ \forall i \in N \setminus j$$

Hence, one obtains,

$$\begin{split} \dot{V_s} \stackrel{a.e.}{\leq} &-k_j \left| s_j \left| \operatorname{sat} \left[\frac{\left| s_j \right|^a}{\mu_j^a} \right] + \left| \Delta_j \left\| s_j \right| + \left(\frac{a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in \mathbb{N} \setminus j} \left(\frac{\left| \Delta_i \right|}{k_i} \right)^{\frac{1}{a}} \mu_i \left| \Delta_i \right| \\ &= -(1-\theta)k_j \left| s_j \left| \operatorname{sat} \left[\frac{\left| s_j \right|^a}{\mu_j^a} \right] - \frac{\theta}{2} k_j \left| s_j \left| \operatorname{sat} \left[\frac{\left| s_j \right|^a}{\mu_j^a} \right] + \left| \Delta_j \left\| s_j \right| - \frac{\theta}{2} k_j \left| s_j \right| \operatorname{sat} \left[\frac{\left| s_j \right|^a}{\mu_j^a} \right] \\ &+ \left(\frac{a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in \mathbb{N} \setminus j} \left(\frac{\left| \Delta_i \right|}{k_i} \right)^{\frac{1}{a}} \mu_i \left| \Delta_i \right| \end{split}$$

with any constant $0 < \theta < 1$,

$$\dot{V}_{s} \stackrel{a.e.}{\leq} -(1-\theta)k_{j}|s_{j}|\operatorname{sat}\left[\frac{|s_{j}|^{a}}{\mu_{j}^{a}}\right], \text{ for } |s_{j}| \geq \pi_{j}, \text{ and } |s_{i}| \in \mathbb{R}, \forall i \in N \setminus j$$

with

$$\pi_{j} = \max\left\{ \left(\frac{2}{\theta} \frac{|\Delta_{j}|}{k_{j}}\right)^{\frac{1}{a}} \mu_{j}, \left(\frac{2\mu_{j}^{a}}{\theta k_{j}}\right)^{\frac{1}{1+a}} \left(\frac{a}{(1+a)^{\frac{1+a}{a}}}\right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} \left(\frac{|\Delta_{i}|}{k_{i}}\right)^{\frac{1}{a}} \mu_{i} |\Delta_{i}|\right)^{\frac{1}{1+a}}\right\},$$

and
$$k_j > \max\left\{\frac{2|\Delta_j|}{\theta}, \frac{2}{\theta\mu_j}\left(\frac{a}{(1+a)^{\frac{1+a}{a}}}\right)\sum_{i\in N\setminus j}\left(\frac{|\Delta_i|}{k_i}\right)^{\frac{1}{a}}\mu_i|\Delta_i|\right\}$$
 (5 - 6)

Note that with the above conditions, we have $\pi_j < \mu_j$. Thus, selecting each k_j , $\forall j \in$

N that satisfy (5 - 6), we have

$$\dot{V}_{s}^{a.e.} < -c_{1}, \forall s \notin \Omega_{\pi}$$

where the compact set is defined as: $\Omega_{\pi} = \left\{ s \in \mathbb{R}^{n} : \left| s_{i} \right| \le \pi_{i}, \forall i \in N \right\}$

with

$$c_1 = \min_i \left\{ (1 - \theta) k_i \pi_i \left(\frac{\pi_i}{\mu_j} \right)^a \right\}, \forall i \in N,$$

Note that this compact set can be made arbitrarily small by increasing k_i up to the maximal allowable control bound, $T_{i, \text{ max}}$. In particular note that by having large k_i implies a smaller π_i since

$$\pi_{j} = \max\left\{ \left(\frac{2}{\theta} \frac{\left|\Delta_{j}\right|}{k_{j}}\right)^{\frac{1}{a}} \mu_{j}, \left(\frac{2\mu_{j}^{a}}{\theta k_{j}}\right)^{\frac{1}{1+a}} \left(\frac{a}{\left(1+a\right)^{\frac{1+a}{a}}}\right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} \left(\frac{\left|\Delta_{i}\right|}{k_{i}}\right)^{\frac{1}{a}} \mu_{i}\left|\Delta_{i}\right|\right)^{\frac{1}{1+a}}\right\}, \forall j \in N$$

which further implies that a smaller compact set Ω_{π} .

Now with each given μ_i , let c_1 be a positive constant $0 < c_1 < 1$, consider the following compact set

$$\Omega_{\mu} = \left\{ s \in \mathbb{R}^{n} : \left| s_{i} \right| \leq c_{1} \mu_{i}, \forall i \in N \right\},$$
$$\Sigma_{\mu} = \left\{ s \in \mathbb{R}^{n} : V_{s} \leq k_{\mu} \right\},$$

where $k_{\mu} > 0$ is defined as $k_{\mu} = \min_{s \in bd \ \Omega_{\mu}} V_s$,

which exists since the boundary, bd Ω_{μ} is a compact set. Note that $\Sigma_{\mu} \subset \Omega_{\mu}$. Since Σ_{μ} is a Lyapunov level set, if the states, *s* can be confined within this set, all *s_i*'s are not saturated. Hence, if the states *s* stay inside the Lyapunov level set Σ_{μ} , all *s_i*'s are not saturated. To achieve this, we simply need the set $\Omega_{\pi} \subset \Sigma_{\mu}$, which can be attained when *k_i* is large enough, such that \dot{V}_s being negative the outside of the set Σ_{μ} ,

$$\dot{V}_s \stackrel{a.e.}{<} 0$$
, for $V_s \ge k_{\mu}$

which implies that the trajectories of *s* will enter the set Σ_{μ} in finite time and stay in it once entered. Hence, we have

$$|s_i| \le c_1 \mu_i < \mu_i \Leftrightarrow |s_i| < \mu_i, \ \forall i \in N$$

In particular, a sufficient condition on k_i such that each s_i is unsaturated can be shown as following:

Note that:

$$k_{\mu} = \min_{s \in \mathrm{bd} \ \Omega_{\mu}} V_{s} \geq \min_{s \in \mathrm{bd} \ \Omega_{\mu}} \frac{1}{2} \underline{m} \|s\|^{2} = \frac{1}{2} \underline{m} \left(\min_{i} c_{1} \mu_{i} \right)^{2}$$

Also,

$$\begin{split} \max_{s \in \mathrm{bd} \,\Omega_{\pi}} V_{s} &\leq \max_{s \in \mathrm{bd} \,\Omega_{\pi}} \frac{1}{2} \overline{m} \left\| s \right\|^{2} \\ &= \frac{1}{2} \overline{m} \left(\max_{s \in \mathrm{bd} \,\Omega_{\pi}} \left\| s \right\| \right)^{2} = \frac{1}{2} \overline{m} \left(\left(\sum_{i}^{n} \left| \pi_{i} \right|^{2} \right)^{\frac{1}{2}} \right)^{2} = \frac{1}{2} \overline{m} \left(\sum_{i}^{n} \left| \pi_{i} \right|^{2} \right) \\ &\leq \frac{1}{2} \overline{m} \left(\sum_{i}^{n} \left| \pi_{i} \right| \right)^{2}, \quad \text{from lemma 2,} \\ &= \frac{1}{2} \overline{m} \left(\sum_{j}^{n} \max \left\{ \frac{\left(\frac{2}{2} \frac{\left| \Delta_{j} \right|}{\theta \, k_{j}} \right)^{\frac{1}{n}} \mu_{j}, \\ &\left(\frac{2\mu_{j}^{a}}{\theta \, k_{j}} \right)^{\frac{1}{n}} \left(\frac{a}{(1+a)^{\frac{1+a}{a}}} \right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} \left(\frac{\left| \Delta_{i} \right|}{k_{i}} \right)^{\frac{1}{a}} \mu_{i} \left| \Delta_{i} \right| \right)^{\frac{1}{1+a}} \right\} \\ &\leq \frac{1}{2} \overline{m} \left(\frac{1}{\underline{k}} \right)^{\frac{2}{a}} \left\{ \sum_{j}^{n} \max \left\{ \frac{\left(\frac{2\left| \Delta_{j} \right|}{\theta \, k_{j}} \right)^{\frac{1}{a}} \mu_{j}, \\ &\left(\frac{2\mu_{j}^{a}}{\theta \, k_{j}} \right)^{\frac{1}{n}} \left(\frac{2}{(1+a)^{\frac{1+a}{a}}} \right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} \left| \Delta_{i} \right|^{\frac{1}{a}} \mu_{i} \left| \Delta_{i} \right| \right)^{\frac{1}{1+a}} \right\} \right\} \\ & = \frac{1}{2} \overline{m} \left(\frac{1}{\underline{k}} \right)^{\frac{2}{a}} \left\{ \sum_{j}^{n} \max \left\{ \frac{\left(\frac{2\left| \Delta_{j} \right|}{\theta \, k_{j}} \right)^{\frac{1}{a}} \mu_{j}, \\ &\left(\frac{2\mu_{j}^{a}}{\theta \, k_{j}} \right)^{\frac{1}{1+a}} \left(\frac{a}{(1+a)^{\frac{1+a}{a}}} \right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} \left| \Delta_{i} \right|^{\frac{1}{a}} \mu_{i} \left| \Delta_{i} \right| \right)^{\frac{1}{1+a}} \right\} \right\} \right\}$$

where $\underline{k} = \min_{i} k_{i}$

Hence, to satisfy the condition $\Omega_{\pi} \subset \Sigma_{\mu}$, it is sufficient for:

$$\begin{split} k_{\mu} &\geq \max_{s \in \mathrm{bd} \ \Omega_{\pi}} V_{s} \\ \Rightarrow &\frac{1}{2} \underline{m} \left(\min_{i} c_{1} \mu_{i} \right)^{2} \\ &\geq \frac{1}{2} \overline{m} \left(\frac{1}{\underline{k}} \right)^{\frac{2}{a}} \left(\sum_{j}^{n} \max \left\{ \left(\frac{2|\Delta_{j}|}{\theta} \right)^{\frac{1}{a}} \mu_{j}, \\ &\left(\frac{2|\Delta_{j}|}{\theta} \right)^{\frac{1}{a}} \left(\frac{2|\Delta_{j}|}{\theta} \right)^{\frac{1}{a}} \left(\frac{2|\Delta_{j}|}{\theta} \right)^{\frac{1}{a}} \mu_{j}, \\ &\left(\frac{2|\Delta_{j}|}{\theta} \right)^{\frac{1}{1+a}} \left(\frac{a}{(1+a)^{\frac{1+a}{a}}} \right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} |\Delta_{i}|^{\frac{1}{a}} \mu_{i}|\Delta_{i}| \right)^{\frac{1}{1+a}} \right\} \end{split}$$

$$\Leftrightarrow \underline{k} \geq \underbrace{\left(\underbrace{\frac{2|\Delta_{j}|}{\theta}}_{j} \right)^{\frac{1}{a}} \mu_{j}, }_{\left(\frac{2\mu_{j}^{a}}{\theta}\right)^{\frac{1}{1+a}} \left(\frac{a}{(1+a)^{\frac{1+a}{a}}}\right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} |\Delta_{i}|^{\frac{1}{a}} \mu_{i}|\Delta_{i}|\right)^{\frac{1}{1+a}} \right)^{\frac{1}{1+a}} }_{\left(\frac{\underline{m}}{\overline{m}}\right)^{\frac{a}{2}} \left(\min_{i} c_{1} \mu_{i}\right)^{a}}$$
(5 - 7)

It is not difficult to show that selecting *K* that satisfies (5 - 7) implies that (5 - 6) is sufficiently satisfied as well. Thus, for the case of a > 0 the control will be unsaturated in finite time and remain so thereafter.

3. Having shown the boundedness of σ in step 1 and *s* being unsaturated and bounded or zero through step 2, depending on the parameter *a*, we are going to show the stability analysis of the desired error dynamics (5 - 3).

For the case of b > 0:

For a = 0, we have, since *s* will reach zero in finite time, from (5 - 4) the (v_1 , v_2) dynamics

$$\dot{v}_{1} = v_{2},$$

$$\dot{v}_{2} = -K_{a}v_{2} + K_{a} \operatorname{sat}_{\left(\frac{\varepsilon_{2}}{k_{2}}\right)^{\frac{1+b}{2b}}} \left[v_{2}\right] - \operatorname{sat}_{\varepsilon_{2}} \left[K_{2}\operatorname{sig}(v_{2})^{\frac{2b}{1+b}}\right] - \operatorname{sat}_{\varepsilon_{1}} \left[K_{1}\operatorname{sig}(v_{1})^{b}\right]$$

where element-wise, we have

$$\dot{v}_{1i} = v_{2i}$$

$$\dot{v}_{2i} = -k_{ai} \left[(v_{2i}) - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} [v_{2i}] \right] - \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} |v_{2i}|^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i}) \right]$$

$$- \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |v_{1i}|^{b} \operatorname{sign}(v_{1i}) \right]$$

While for a > 0:

Thus, we have, since the control will be unsaturated in finite time, $sat[\mu^{-1}s] = \mu^{-1}s$, from (5 - 4) the (v_1 , v_2) dynamics

$$\dot{v}_{1} = v_{2} + s,$$

$$\dot{v}_{2} = -K_{a}v_{2} - K_{a}s + K_{a} \operatorname{sat}_{\left(\frac{s_{2}}{k_{2}}\right)^{\frac{1+b}{2b}}} \left[v_{2} + s\right] - \operatorname{sat}_{s_{2}} \left[K_{2}\operatorname{sig}(v_{2} + s)^{\frac{2b}{1+b}}\right]$$
$$-\operatorname{sat}_{s_{1}} \left[K_{1}\operatorname{sig}(v_{1})^{b}\right]$$

where element-wise,

$$\dot{v}_{1i} = v_{2i} + s_i$$

$$\dot{v}_{2i} = -k_{ai} \left((v_{2i} + s_i) - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} [v_{2i} + s_i] \right) - \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} |v_{2i} + s_i|^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i} + s_i) \right] \quad (5 - 8)$$

$$- \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |v_{1i}|^b \operatorname{sign}(v_{1i}) \right]$$

Remark 5.3. Note that the desired error dynamics, after the convergence of *s*, for the case of a = 0 is identical to that of the case of a > 0 when s = 0.

From Appendix B.1, with the Lyapunov function

$$V = \left(\frac{1}{2}v_{2i}^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |r|^{b} \operatorname{sign}(r)\right] dr\right)^{2} + r_{1} \left(\frac{3+3b}{2}\right) \left(\int_{0}^{v_{1i}} |r|^{\frac{1}{2}} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \left[|r|^{\frac{3b}{2}}\right] dr\right) v_{2i}$$
$$+ r_{1} (1+b) k_{ai} \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}} \left(\int_{0}^{v_{1i}} |r|^{\frac{3}{2}} \operatorname{sign}(r) - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \left[|r|^{\frac{3}{2}} \operatorname{sign}(r)\right] dr\right)$$

where $r_1 > 0$ is a constant, it is shown that

$$\dot{\widetilde{V}} < 0$$
, for $V \ge \rho_{1i}(|s_i|)$,

where the class \mathcal{K} function, ρ_{1i} is defined as, $\rho_{1i}(|s_i|) = \max_{(v_{1i}, v_{2i}) \in bd |\Psi_{3i}(|s_i|)} V$,

with bd $\Psi_{3i}(|s_i|)$ as the boundary of the compact set

$$\Psi_{3i}(|s_i|) = \{(v_{1i}, v_{2i}) \in \mathbb{R}^2 : |v_{1i}| \le \alpha_1(|s_i|), |v_{2i}| \le \alpha_2(|s_i|)\}, \text{ and}$$

 α_1 , α_2 are class \mathcal{K} functions defined in Appendix B.1. Hence, invoking lemma 2.6

together with the boundedness of *s* (see step 2), the states (v_1, v_2) are uniformly ultimately bounded.

For the special case of b = 0:

Let the elements of ε_1 equal to that of K_1 , and ε_2 to that of K_2 , i.e. $\varepsilon_{1n} = k_{1n}, \forall i \in N$ and $\varepsilon_{2n} = k_{2n}, \forall i \in N$, hence the desired dynamics becomes

$$\dot{v}_{1i} = v_{2i} + s_i \dot{v}_{2i} = -k_{ai} ((v_{2i} + s_i) - \text{sat}[v_{2i} + s_i]) - k_{2i} \text{sign}(v_{2i} + s_i) - k_{1i} \text{sign}(v_{1i})$$
(5 - 9)

From Appendix B.1, with the Lyapunov function

$$V = \left(\frac{1}{2}v_{2i}^{2} + k_{1i}\int_{0}^{v_{1i}}\operatorname{sign}(r) dr\right)^{2} + r_{1}|v_{1i}|^{\frac{3}{2}}\operatorname{sign}(v_{1i})v_{2i} + r_{1}\left(\frac{2}{5}\right)k_{ai}|v_{1i}|^{\frac{5}{2}} - r_{1}k_{ai}\int_{0}^{v_{1i}}\left[\operatorname{sat}_{1}\left[\left|r\right|^{\frac{3}{2}}\right]\right]\operatorname{sign}(r) dr$$

where $r_1 > 0$ is a constant, it is shown that

$$\widetilde{V} < 0$$
, for $V \ge \rho_{2i}(|s_i|)$,

where the class \mathcal{K} function, ρ_{2i} is defined as $\rho_{2i}(|s_i|) = \max_{(v_{1i}, v_{2i}) \in bd \ \phi_{3i}(|s_i|)} V$,

with bd $\phi_{3i}(|s_i|)$ as the boundary of the compact set

$$\phi_{3i}(|s_i|) = \{(v_{1i}, v_{2i}) \in \mathbb{R}^2 : |v_{1i}| \le \alpha_{11}(|s_i|), |v_{2i}| \le \alpha_{12}(|s_i|)\} \text{ and }$$

 α_{11} , α_{12} are class \mathcal{K} functions defined in Appendix B.1. Hence, invoking lemma 2.6

together with the boundedness of *s* (see step 2 0), the states (v_1, v_2) are uniformly ultimately bounded.

Remark 5.4. Hence, for *K* satisfying (5 - 7), the trajectories of the closed-loop system (5 - 4), are globally stable, with the ultimate bound being reduced as desired up to the saturation limit of the actuator, $T_{i, \max} > 0$, $\forall i = 1,...,n$. Another way to view the stability results is, for a given bounds on the actuator limit, and hence *K*, if the bounds on the uncertainties,

$$\Delta = -G(q) - M(q)\ddot{q}_d + M(q)\dot{\sigma} - \mathbf{K}[D] - F(\dot{q}_d - \sigma) - C(q, (\dot{q}_d - \sigma))(\dot{q}_d - \sigma)$$

are sufficiently small to satisfy (5 - 7), the global practical stability results are still assured. Essentially, besides the bounds on the parameters of the robot manipulator dynamics and disturbances, the term Δ is also dependent on the desired acceleration, velocity and the desired error dynamics through the term σ and σ . In which case, a slower desired trajectory or slower desired error dynamics can, in effect, produce a smaller upper bound of $|\Delta|$.

Remark 5.5. It is desired for the controller (5 - 2) to behave as per its non-bounded counterpart (Chapter 3) when the controller is not saturated so that it exhibits the same properties such as the ability to inject desired error dynamics, and desired performance. Indeed from stability proof above (step 2), the control (5 - 2) will be unsaturated in finite time and stay so in all future times. Particularly, if the upper bound of $|s_i|$ is sufficiently small (by selecting gain *K* satisfying (5 - 7) up to the allowable control bound, $T_{i, \max}$), from step 3 above, the states (v_1 , v_2) of the dynamics (5 - 8) and (5 - 9) will become unsaturated as well (see Appendix B.1),

for b > 0,

$$\dot{v}_{1i} = v_{2i} + s_i$$

$$\dot{v}_{2i} = -k_{1i} |v_{1i}|^b \operatorname{sign}(v_{1i}) - k_{2i} |v_{2i} + s_i|^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i} + s_i)$$

for b = 0,

$$\dot{v}_{1i} = v_{2i} + s_i$$

 $\dot{v}_{2i} = -k_{2i} \operatorname{sign}(v_{2i} + s_i) - k_{1i} \operatorname{sign}(v_{1i})$

which is similar to the *z*-subsystem (3 - 4) considered in Chapter 3, and hence will exhibit the same properties, since the states will remain unsaturated for all future times, provided $|s_i|$ is sufficiently small.

Remark 5.6. Note that the desired error dynamics injected by the controller through (5 - 3) is a modified version of (3 - 3) in Chapter 3 to account for the bounded control structure considered in this chapter. Specifically, it is an integral of saturation functions that provides boundedness of σ as per step 1 of the proof of Theorem 5.1. This in turn, helps to dominate the effects of Coriolis term as shown in step 2 above, which is essential for global practical stability by accounting the inherent viscous friction of the robot manipulator.

Remark 5.7. Observe that in step 2 above, it is necessary for the viscous friction term to satisfy the inequality $\underline{f} > C_m \| (\dot{q}_d - \sigma) \|$. This condition that is commonly found in the literature of global stability of robot manipulator with bounded control (see, for instance, [118] and [113]). This condition can be met by selecting the desired trajectories (through the desired velocity vector) and desired error dynamics (through (5 - 3)) appropriately. This condition restricts the desired velocity and acceleration (i.e. through the desired error dynamics) vectors but not the location of the desired task (i.e. q_d). Essentially, the desired trajectory may be defined anywhere within the workspace of the robotic manipulator provided it has sufficiently slow motions.

Remark 5.8. It is worth mentioning that the desired error dynamics injection term have an inherent anti-windup structure, rewriting (5 - 3) by adding and subtracting $K_a s$,

$$\dot{\sigma} = -K_a \sigma + K_a s - \underbrace{K_a \left(s - \mu \operatorname{sat}\left[\mu^{-1} s\right]\right)}_{anti-windup} - K_a \operatorname{sat}_{\left(\frac{\varepsilon_2}{k_2}\right)^{\frac{1+b}{2b}}} \left[e_2\right] + \operatorname{sat}_{\varepsilon_2} \left[K_2 \operatorname{sig}(e_2)^{\frac{2b}{1+b}}\right] + \operatorname{sat}_{\varepsilon_1} \left[K_1 \operatorname{sig}(e_1)^b\right],$$

hence when the control is saturated, the above feedback signal tries to drive the "control input error" $(s - \mu \operatorname{sat}[\mu^{-1}s])$ to zero. Anti-windup prevents control loop from severe stability and performance degradation induced by integral action winding [117].

5.2 Numerical Simulations

In this section, numerical simulations on a two-link robot manipulator were carried out to illustrate the results discussed in this chapter. The setups for each simulation are described. Discussion and analysis of the results are presented accordingly.

5.2.1 Simulation Setup

1) Simulation 1:

The same two-link rigid robot manipulator considered in section 3.3 is adopted in simulation. The dynamics of robot manipulator (5 - 1) have the same parameter values as that in section 3.3. The desired trajectory vector and the additive disturbances Coulomb friction

vector were defined similarly as well. The initial conditions of the robot manipulator were selected as,

$$q(0) = \begin{bmatrix} 0 \\ 0 \end{bmatrix}, \dot{q}(0) = \begin{bmatrix} 10 \\ 16 \end{bmatrix}, q_d(0) = \begin{bmatrix} \pi \\ 0.5 \end{bmatrix}, \dot{q}_d(0) = \begin{bmatrix} 0.5 \\ 0 \end{bmatrix}$$

The control (5-2) parameter values were selected as follows,

$$a = 0.9, \ K = \begin{bmatrix} 150 & 0 \\ 0 & 15 \end{bmatrix}, \ K_1 = \begin{bmatrix} 18 & 0 \\ 0 & 18 \end{bmatrix}, \ K_2 = \begin{bmatrix} 11 & 0 \\ 0 & 11 \end{bmatrix}, \ b = 0.6$$
$$\mu = \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix}, \ \varepsilon_{11} = \varepsilon_{12} = 18, \ \varepsilon_{21} = \varepsilon_{22} = 11, \ K_a = \begin{bmatrix} 10 & 0 \\ 0 & 10 \end{bmatrix},$$

The initial conditions of the vector σ were selected as

$$\sigma(0) = -e_2(0) = \begin{bmatrix} 0.5\\0 \end{bmatrix} - \begin{bmatrix} 10\\16 \end{bmatrix} = \begin{bmatrix} -9.5\\-16 \end{bmatrix}$$

Note that the initial velocity is purposefully made to be far from the origin in order to bring the controller to saturation.

2) Simulation 2:

The setup of Simulation 2 is exactly the same as that of Simulation 1. The simulation is repeated using the unbounded full-state feedback control (3 - 2), to examine its maximal tracking errors compared to that of control (5 - 2), with the parameter values selected as

$$a = 0.9, K = \begin{bmatrix} 150 & 0 \\ 0 & 15 \end{bmatrix}, K_1 = \begin{bmatrix} 18 & 0 \\ 0 & 18 \end{bmatrix}, K_2 = \begin{bmatrix} 11 & 0 \\ 0 & 11 \end{bmatrix}, b = 0.6$$

Note that the same parameter values are attained when the control (5 - 2) vectors of Simulation 1 are unsaturated.

5.2.2 Results and Discussions

For better visualization of the plots, some figures are shown in two windows; each with different time intervals.







(c) Tracking error, e_{12} for time t = [0, 15] s.

time (s)



(d) Tracking error, e_{12} for time t = [5, 15] s.

Figure 5.2 Simulation 2. Tracking errors using control (3 - 2), control (5 - 2).

1) Simulation 1:

As can be observed Figure 5.1(e), the large initial conditions caused the control (5 - 2) to saturate. From Theorem 5.1, the state σ will converge to a bounded region in finite time, due

to the saturated nature of the integral (see (5 - 3)), which is clearly shown in Figure 5.1(c)-(d). According to the stability proof, once the state σ converge to a bounded region and stay in it for all future times, with control gain *K* satisfying (5 - 6), the control will be unsaturated in finite time and stay in it for all future times. In fact, from Figure 5.1(e)-(f) the control inputs for both joints of the robot manipulator remained unsaturated after the initial saturation.

2) Simulation 2:

Figure 5.2 shows that both controllers (3 - 2) and (5 - 2) have similar ultimate bound on the tracking errors. This is expected as when the controller (5 - 2) becomes unsaturated, it has an identical structure of the unbounded control (3 - 2). However, a more important issue is how the control (5 - 2) can remain unsaturated once it becomes unsaturated. The controller (5 - 2) solve this issue in two-fold. Firstly, through its saturated integral structure (5 - 3), it ensures the boundedness of the states of σ . This in turns, provide boundedness of the vector Δ (including the Coriolis term), together with the inherent viscous friction of the dynamics, the global boundedness of both σ and *s* are achieved. Secondly, through sufficiently high saturation levels of the control, i.e. such that for a = 0, $k_i > |\Delta_i|$ while for a > 0, need (5 - 6), by means of strict Lyapunov functions, the state *s* will become unsaturated and remain so for all future times.

Hence, due to the specific design (5 - 3) of control (5 -2), a saturated controller that inherits interesting properties (such as gain selection based on desired error dynamics injection and

performance bound) of its unbounded counterpart is shown.

5.3 Summary

A saturated controller is developed for the trajectory tracking or robot manipulators under the influence of bounded disturbances. Global stability is assured by taking into account of the viscous friction and the proposed nonlinear integrator. Effectively, the controller allows the injection of desired error dynamics similar to the unbounded control of Chapter 3, hence allowing simple gain selection as well. When the saturation level is sufficiently high for the user-defined speed of desired trajectory, global finite-time and exponential regulation can be achieved.

Chapter 6: ROBOT MANIPULATOR CONTROL: OUTPUT FEEDBACK BOUNDED CONTROL APPROACH

Leveraging the results of Chapter 4 and 5, this chapter explore an output feedback bounded tracking control of robot manipulators with bounded disturbances. Lyapunov based stability analysis is provided to show global practical tracking results, while global finite time or exponential convergence can be obtained as well for regulation problem. Simulation results are provided to display the control performance.

6.1 Observer Dynamics

Inspired by the results of Chapter 4, the observer that exhibits properties of super-twisting algorithms with the addition of a linear damping term will be employed. Essentially, the same observer dynamics of section 4.1 will be considered in this chapter for the robot manipulator under bounded control approach.

6.1.1 System Description

Recall from section 4.1, the observer dynamics,

$$\begin{aligned} \dot{\widetilde{e}}_1 &= -L_1 \operatorname{sig}(\widetilde{e}_1)^p + \widetilde{e}_2 , \\ \dot{\widetilde{e}}_2 &= -L_2 \operatorname{sig}(\widetilde{e}_1)^{2p-1} - L_3 \widetilde{e}_2 + d \end{aligned}$$

where $\dot{\tilde{e}}_1, \dot{\tilde{e}}_2 \in \mathbb{R}^n$, are the vector state variables, $L_1, L_2, L_3 \in \mathbb{R}^{n \times n}$ are positive definite diagonal matrices, $0.5 \le p \le 1$, and *d* comprise elements of bounded perturbations.

Consider element wise, $\forall i = 1, ..., n$,

$$\dot{\widetilde{e}}_{1i} = -l_{1i} \operatorname{sig}(\widetilde{e}_{1i})^p + \widetilde{e}_{2i} ,$$

$$\dot{\widetilde{e}}_{2i} = -l_{2i} \operatorname{sig}(\widetilde{e}_{1i})^{2p-1} - l_{3i} \widetilde{e}_{2i} + d_i$$
(6 - 1)

with $|d_i| := \sup \{ \varepsilon_i \mid : \varepsilon_i \in \mathbf{K}[d_i] \}$.

The stability analysis that supports the main results of this chapter is described in the following.

6.1.2 Stability Analysis

Consider the following Lyapunov function:

$$V_{\widetilde{e}i}(\widetilde{e}_{1i},\widetilde{e}_{2i}) = \left(\frac{1}{2}\widetilde{e}_{2i}^{2} + \frac{l_{2i}}{2p}|\widetilde{e}_{1i}|^{2p}\right)^{2} - r_{\widetilde{e}}|\widetilde{e}_{1i}||\widetilde{e}_{2i}|^{\frac{4p-1}{p}}\operatorname{sign}(\widetilde{e}_{1i}\widetilde{e}_{2i})$$

(Note that this Lyapunov function is presented for the super-twisting based algorithms in section 2.4.2)

where
$$\underline{\pi}_{2i} \left\| \widetilde{e}_{2i} \right\|^4 + \left| \widetilde{e}_{1i} \right|^{4p} \le V_{\widetilde{e}i} \le \overline{\pi}_{2i} \left\| \widetilde{e}_{2i} \right\|^4 + \left| \widetilde{e}_{1i} \right|^{4p} \right)$$

$$\underline{\pi}_{2i} \coloneqq \min\left\{ \frac{1}{8}, \frac{1}{2} \left(\frac{l_{2i}}{2p} \right)^2 \right\}, \quad \overline{\pi}_{2i} \coloneqq \max\left\{ \left(\frac{l_{2i}}{2p} \right)^2 + \frac{l_{2i}}{4p} + \frac{r_{\widetilde{e}}}{4p}, \quad \frac{1}{4} + \frac{l_{2i}}{4p} + \frac{r_{\widetilde{e}}(4p-1)}{4p} \right\},$$

thus, $V_{\tilde{e}i}$ is positive definite and radially unbounded. Since (6 - 1) is a differential equation that has discontinuous right-hand side, i.e. when p = 0.5 and since no continuity assumption is made on d_i , its solutions are understood in the sense of Filippov (see definition 2.1). *Remark* 6.1. The stability analysis of (6 - 1) will be considered with a different Lyapunov function as in section 4.1. This is due to the fact that in Chapter 4, the results pertain to semiglobal stability, while on a bounded control approach of Chapter 5, global stability is achieved. Since in this chapter, the observer (6 - 1) will be utilised under a bounded control approach in the subsequent sections, to show global stability, the observer dynamics analysis is repeated with a different Lyapunov function.

According to lemma 2.4, the time derivative of non-smooth Lyapunov function exists almost everywhere

$$\frac{d}{dt}V_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \stackrel{a.e.}{\in} \dot{\tilde{V}}_{\tilde{e}i} \coloneqq \bigcap_{\xi \in \partial V_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i})} \xi^{\mathrm{T}} \mathbf{K} \begin{bmatrix} \dot{\tilde{e}}_{1i} \\ \dot{\tilde{e}}_{2i} \end{bmatrix}$$

Since (6 - 1) is discontinuous when p = 0.5, for ease of presentation, the analysis is separated for the case of 0.5 and <math>p = 0.5.

a) Case 1: For
$$0.5 :$$

Note that for $0.5 , <math>V_{\tilde{e}i}$ is continuously differentiable:

$$\dot{\widetilde{V}}_{\widetilde{e}i} = \bigcap_{\xi \in \partial V_{\widetilde{e}i}(\widetilde{e}_{1i},\widetilde{e}_{2i})} \xi^{\mathrm{T}} \boldsymbol{K} \begin{bmatrix} \dot{\widetilde{e}}_{1i} \\ \dot{\widetilde{e}}_{2i} \end{bmatrix}$$
$$= \nabla V_{\widetilde{e}i}^{\mathrm{T}} \boldsymbol{K} \begin{bmatrix} \dot{\widetilde{e}}_{1i} \\ \dot{\widetilde{e}}_{2i} \end{bmatrix}$$

$$\begin{split} & \subset -\frac{l_{1i}l_{2i}^{2}}{p} |\widetilde{e}_{1i}|^{5p-1} - l_{3i} |\widetilde{e}_{2i}|^{4} - l_{1i}l_{2i} |\widetilde{e}_{1i}|^{3p-1} |\widetilde{e}_{2i}|^{2} - \frac{l_{2i}l_{3i}}{p} |\widetilde{e}_{1i}|^{2p} |\widetilde{e}_{2i}|^{2} - r_{\widetilde{e}} |\widetilde{e}_{2i}|^{\frac{5p-1}{p}} \\ & + r_{\widetilde{e}}l_{1i} |\widetilde{e}_{1i}|^{p} |\widetilde{e}_{2i}|^{\frac{4p-1}{p}} \operatorname{sign}(\widetilde{e}_{1i}\widetilde{e}_{2i}) + r_{\widetilde{e}}l_{2i} \left(\frac{4p-1}{p}\right) |\widetilde{e}_{1i}|^{2p} |\widetilde{e}_{2i}|^{\frac{3p-1}{p}} \\ & + r_{\widetilde{e}}l_{3i} \left(\frac{4p-1}{p}\right) |\widetilde{e}_{1i}| |\widetilde{e}_{2i}|^{\frac{4p-1}{p}} \operatorname{sign}(\widetilde{e}_{1i}\widetilde{e}_{2i}) \\ & + K \left[d_{i} \left(\frac{l_{2i}}{p} |\widetilde{e}_{1i}|^{2p} \widetilde{e}_{2i} + \widetilde{e}_{2i}^{3} - r_{\widetilde{e}} \left(\frac{4p-1}{p}\right) \widetilde{e}_{1i} |\widetilde{e}_{2i}|^{\frac{3p-1}{p}} \right) \end{split}$$

After rearrangement,

$$\dot{V}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \stackrel{a.e.}{\leq} -\frac{l_{1i}l_{2i}^{2}}{3p} |\tilde{e}_{1i}|^{5p-1} -\frac{r_{\tilde{e}}}{2} |\tilde{e}_{2i}|^{\frac{5p-1}{p}} +\dot{V}_{1} +\dot{V}_{2}$$

where

$$\begin{split} \dot{V_{1}} &= -\frac{l_{1i}l_{2i}^{2}}{3p} \left| \widetilde{e}_{1i} \right|^{5p-1} - \frac{l_{3i}}{2} \left| \widetilde{e}_{2i} \right|^{4} - \frac{l_{2i}l_{3i}}{2p} \left| \widetilde{e}_{1i} \right|^{2p} \left| \widetilde{e}_{2i} \right|^{2} - l_{1i}l_{2i} \left| \widetilde{e}_{1i} \right|^{3p-1} \left| \widetilde{e}_{2i} \right|^{2} \\ &- \frac{r_{\tilde{e}}}{2} \left| \widetilde{e}_{2i} \right|^{\frac{5p-1}{p}} + r_{\tilde{e}}l_{1i} \left| \widetilde{e}_{1i} \right|^{p} \left| \widetilde{e}_{2i} \right|^{\frac{4p-1}{p}} \\ &+ r_{\tilde{e}}l_{2i} \left(\frac{4p-1}{p} \right) \widetilde{e}_{1i} \left|^{2p} \left| \widetilde{e}_{2i} \right|^{\frac{3p-1}{p}} + r_{\tilde{e}}l_{3i} \left(\frac{4p-1}{p} \right) \widetilde{e}_{1i} \left| \widetilde{e}_{2i} \right|^{\frac{4p-1}{p}} \end{split}$$

$$\begin{split} \dot{V_2} &= -\frac{l_{1i}l_{2i}}{3p} |\widetilde{e}_{1i}|^{5p-1} - \frac{l_{3i}}{2} |\widetilde{e}_{2i}|^4 - \frac{l_{2i}l_{3i}}{2p} |\widetilde{e}_{1i}|^{2p} |\widetilde{e}_{2i}|^2 \\ &+ |d_i| \left(\frac{l_{2i}}{p} |\widetilde{e}_{1i}|^{2p} |\widetilde{e}_{2i}| + |\widetilde{e}_{2i}|^3 + r_{\widetilde{e}} \left(\frac{4p-1}{p}\right) |\widetilde{e}_{1i}| |\widetilde{e}_{2i}|^{\frac{3p-1}{p}} \end{split}$$

For \dot{V}_1 :

Applying lemma 2.1,

$$\begin{split} -\left|\widetilde{e}_{1i}\right|^{5p-1} - \left|\widetilde{e}_{2i}\right|^{\frac{5p-1}{p}} &\leq -\left(\frac{5p-1}{p}\right)^{\frac{p}{5p-1}} \left|\widetilde{e}_{1i}\right|^{p} \left(\frac{5p-1}{4p-1}\right)^{\frac{4p-1}{5p-1}} \left|\widetilde{e}_{2i}\right|^{\frac{4p-1}{p}}, \\ -\left|\widetilde{e}_{1i}\right|^{5p-1} - \left|\widetilde{e}_{2i}\right|^{\frac{5p-1}{p}} &\leq -\left(\frac{5p-1}{2p}\right)^{\frac{2p}{5p-1}} \left|\widetilde{e}_{1i}\right|^{2p} \left(\frac{5p-1}{3p-1}\right)^{\frac{3p-1}{5p-1}} \left|\widetilde{e}_{2i}\right|^{\frac{3p-1}{p}}, \\ -\left|\widetilde{e}_{2i}\right|^{4} - \left|\widetilde{e}_{1i}\right|^{2p} \left|\widetilde{e}_{2i}\right|^{2} &\leq -\left(2p\right)^{\frac{1}{2p}} \left|\widetilde{e}_{1i}\right| \left|\widetilde{e}_{2i}\right|^{\frac{1}{p}} \left(\frac{2p}{2p-1}\right)^{\frac{2p-1}{2p}} \left|\widetilde{e}_{2i}\right|^{\frac{4p-2}{p}}, \end{split}$$

Thus, if the following inequalities

$$\min\left\{\lambda_{1}\frac{l_{2i}^{2}}{l_{1i}\frac{4p-1}{p}}, \ \lambda_{2}l_{1i}l_{2i}\frac{1-p}{2p}, \ \lambda_{3}l_{2i}\frac{1}{2p}\right\} > r_{\tilde{e}} > 0$$

where

$$\begin{split} \lambda_{1} &= \left(\frac{5p-1}{6p^{2}}\right) \left(\frac{5p-1}{16p-4}\right)^{\frac{4p-1}{p}}, \ \lambda_{2} = \left(\frac{5p-1}{12p^{2}}\right) \left(\frac{5p-1}{12p-4}\right)^{\frac{3p-1}{2p}} \left(\frac{p}{4p-1}\right)^{\frac{5p-1}{2p}}, \\ \lambda_{3} &= \left(\frac{p}{2p-1}\right)^{\frac{2p-1}{2p}} \left(\frac{p}{4p-1}\right), \end{split}$$

hold then the function \dot{V}_1 is negative definite. Note that such an $r_{\tilde{e}} > 0$ always exists for any $l_{1i} > 0$, $l_{2i} > 0$.

For \dot{V}_2 :

Applying lemma 2.1,

$$\left|\widetilde{e}_{1i}\right|\left|\widetilde{e}_{2i}\right|^{\frac{3p-1}{p}} = \left|\widetilde{e}_{1i}\right|\left|\widetilde{e}_{2i}\right|^{\frac{1}{2p}}\left|\widetilde{e}_{2i}\right|^{\frac{6p-3}{2p}} \le \left(\frac{1}{2p}\right)\left|\widetilde{e}_{1i}\right|^{2p}\left|\widetilde{e}_{2i}\right| + \left(\frac{2p-1}{2p}\right)\left|\widetilde{e}_{2i}\right|^{3}$$

Then,

$$\begin{split} \dot{V_{2}} &\leq -\frac{l_{1i}l_{2i}^{2}}{3p} |\widetilde{e}_{1i}|^{5p-1} - \frac{l_{3i}}{2} |\widetilde{e}_{2i}|^{4} - \frac{l_{2i}l_{3i}}{2p} |\widetilde{e}_{1i}|^{2p} |\widetilde{e}_{2i}|^{2} \\ &+ |d_{i} \left(\left(\frac{l_{2i}}{p} + r_{\widetilde{e}} \left(\frac{4p-1}{2p^{2}} \right) \right) |\widetilde{e}_{1i}|^{2p} |\widetilde{e}_{2i}| + \left(1 + r_{\widetilde{e}} \left(\frac{4p-1}{p} \right) \left(\frac{2p-1}{2p} \right) \right) |\widetilde{e}_{2i}|^{3} \right) \end{split}$$

The state space is divided into the following three regions for ease of analysis:

$$\begin{split} \phi_{1i}(|d_i|, l_{3i}) &= \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : \widetilde{e}_{1i} \in \mathbb{R}, |\widetilde{e}_{2i}| \ge \beta_2(|d_i|, l_{3i}) \}, \\ \phi_{2i}(|d_i|, l_{3i}) &= \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \ge \beta_1(|d_i|, l_{3i}), |\widetilde{e}_{2i}| \le \beta_2(|d_i|, l_{3i}) \}, \\ \phi_{3i}(|d_i|, l_{3i}) &= \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \le \beta_1(|d_i|, l_{3i}), |\widetilde{e}_{2i}| \le \beta_2(|d_i|, l_{3i}) \}, \end{split}$$

where the functions β_1 and β_2 are of class- \mathcal{KL} defined by:

$$\beta_{1}(|d_{i}|, l_{3i}) \coloneqq \max \begin{cases} \left(\frac{\beta_{2}(|d_{i}|, l_{3i})|d_{i}|}{l_{1i}l_{2i}^{2}} \left(6l_{2i} + r_{\tilde{e}}\left(\frac{12p-3}{p}\right)\right)\right)^{\frac{1}{3p-1}}, \\ \left(\frac{|d_{i}|[\beta_{2}(|d_{i}|, l_{3i})]^{3}}{l_{1i}l_{2i}^{2}} \left(6p + r_{\tilde{e}}\left(\frac{4p-1}{p}\right)(6p-3)\right)\right)^{\frac{1}{5p-1}} \right\}, \\ \beta_{2}(|d_{i}|, l_{3i}) \coloneqq \frac{|d_{i}|}{l_{3i}} \times \max\left\{ \left(2 + r_{\tilde{e}}\left(\frac{4p-1}{p}\right)\left(\frac{2p-1}{p}\right)\right), \left(2 + \frac{r_{\tilde{e}}}{l_{2i}}\left(\frac{4p-1}{p}\right)\right) \right\} \end{cases}$$

Consider the region: $\phi_{1i}(|d_i|, l_{3i}) = \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : \widetilde{e}_{1i} \in \mathbb{R}, |\widetilde{e}_{2i}| \ge \beta_2(|d_i|, l_{3i}) \}$

Thus,

$$\begin{split} \dot{V_2} &= -\frac{l_{1i}l_{2i}^{2}}{3p} |\widetilde{e_{1i}}|^{5p-1} - |\widetilde{e_{2i}}|^3 \left(\frac{l_{3i}}{2} |\widetilde{e_{2i}}| - |d_i| \left(1 + r_{\widetilde{e}} \left(\frac{4p-1}{p}\right) \left(\frac{2p-1}{2p}\right)\right)\right) \\ &- |\widetilde{e_{1i}}|^{2p} |\widetilde{e_{2i}}| \left(\frac{l_{2i}l_{3i}}{2p} |\widetilde{e_{2i}}| - |d_i| \left(\frac{l_{2i}}{p} + r_{\widetilde{e}} \left(\frac{4p-1}{2p^2}\right)\right)\right) \\ &\leq 0 \end{split}$$

for the states in this region.

Next, consider the region: $\phi_{2i}(|d_i|, l_{3i}) = \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \ge \beta_1(|d_i|, l_{3i}), |\widetilde{e}_{2i}| \le \beta_2(|d_i|, l_{3i}) \}$ Thus,

$$\begin{split} \dot{V}_{2} &\leq -\frac{l_{1i}l_{2i}^{2}}{3p} |\widetilde{e}_{1i}|^{5p-1} - \frac{l_{3i}}{2} |\widetilde{e}_{2i}|^{4} - \frac{l_{2i}l_{3i}}{2p} |\widetilde{e}_{1i}|^{2p} |\widetilde{e}_{2i}|^{2} \\ &+ |d_{i} \left[\left(\frac{l_{2i}}{p} + r_{\tilde{e}} \left(\frac{4p-1}{2p^{2}} \right) \right) \widetilde{e}_{1i}|^{2p} \beta_{2} \left(|d_{i}|, l_{3i} \right) + \left(1 + r_{\tilde{e}} \left(\frac{4p-1}{p} \right) \left(\frac{2p-1}{2p} \right) \right) \left[\beta_{2} \left(|d_{i}|, l_{3i} \right) \right]^{3} \right) \\ &= - |\widetilde{e}_{1i}|^{2p} \left(\frac{l_{1i}l_{2i}^{2}}{6p} |\widetilde{e}_{1i}|^{3p-1} - |d_{i} \left(\frac{l_{2i}}{p} + r_{\tilde{e}} \left(\frac{4p-1}{2p^{2}} \right) \right) \beta_{2} \left(|d_{i}|, l_{3i} \right) \right) \\ &- \left(\frac{l_{1i}l_{2i}^{2}}{6p} |\widetilde{e}_{1i}|^{5p-1} - |d_{i} \left(1 + r_{\tilde{e}} \left(\frac{4p-1}{p} \right) \left(\frac{2p-1}{2p} \right) \right) \left[\beta_{2} \left(|d_{i}|, l_{3i} \right) \right]^{3} \right) \end{split}$$

where $\dot{V}_2 \le 0$ for

$$|\tilde{e}_{1i}| \ge \max\left\{ \begin{cases} \left(\frac{\beta_2(|d_i|, l_{3i})|d_i|}{l_{1i}l_{2i}^2} \left(6l_{2i} + r_{\tilde{e}}\left(\frac{12p - 3}{p}\right)\right)\right)^{\frac{1}{3p - 1}}, \\ \left(\frac{|d_i|[\beta_2(|d_i|, l_{3i})]^3}{l_{1i}l_{2i}^2} \left(6p + r_{\tilde{e}}\left(\frac{4p - 1}{p}\right)(6p - 3)\right)\right)^{\frac{1}{5p - 1}} \end{cases} \right\}$$

which is sufficiently satisfied for the states in this region.

Next consider the compact set:

$$\phi_{3i}(|d_i|, l_{3i}) = \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \leq \beta_1(|d_i|, l_{3i}), |\widetilde{e}_{2i}| \leq \beta_2(|d_i|, l_{3i}) \}$$

Note that, $\dot{V}_2 \leq 0$ for $(\tilde{e}_{1i}, \tilde{e}_{2i}) \notin \phi_{3i}(|d_i|, l_{3i})$.

Thus,

$$\dot{V}_{\tilde{e}i}\left(\widetilde{e}_{1i},\widetilde{e}_{2i}\right)^{a.e.} \leq -\frac{l_{1i}l_{2i}^{2}}{3p} \left|\widetilde{e}_{1i}\right|^{5p-1} - \frac{r_{\tilde{e}}}{2} \left|\widetilde{e}_{2i}\right|^{\frac{5p-1}{p}} < 0, \text{ for } \left(\widetilde{e}_{1i},\widetilde{e}_{2i}\right) \notin \phi_{3i}\left(\left|d_{i}\right|,l_{3i}\right)$$

Next, define a Lyapunov level set:

$$\Sigma_{\widetilde{e}i}(|d_i|, l_{3i}) = \left\{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : V_{\widetilde{e}i} \leq \rho_{1i}(|d_i|, l_{3i}) \right\}$$

where the class \mathcal{KL} function ρ_{1i} ($|d_i|$, l_{3i}) is defined as follows:

$$\rho_{1i}(d_i|, l_{3i}) = \max_{(\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathrm{bd} \ \phi_3(d_i|, l_{3i})} V_{\widetilde{e}i}(\widetilde{e}_{1i}, \widetilde{e}_{2i})$$

which exists since the boundary of the set is compact and $V_{\tilde{e}i}$ is continuous. Then we observe that $\phi_{3i}(|d_i|, l_{3i}) \subset \Sigma_{\tilde{e}i}(|d_i|, l_{3i})$. As a result, we have

$$\frac{d}{dt}V_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \stackrel{a.e.}{\in} \dot{\tilde{V}}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \leq -\frac{l_{1i}l_{2i}^{2}}{3p} |\tilde{e}_{1i}|^{5p-1} - \frac{r_{\tilde{e}}}{2} |\tilde{e}_{2i}|^{\frac{5p-1}{p}}, \text{ for } V_{\tilde{e}i} \geq \rho_{1i}(|d_i|,l_{3i})$$

which implies that each of the trajectories for the *i*-th planar system will enter their respective compact level set $\sum_{\tilde{e}i} (|d_i|, l_{3i})$ in finite time and stay in it once entered.

b) Case 2: For
$$p = 0.5$$

Note that or p = 0.5, $V_{\tilde{e}_i}$ is not differentiable on $\tilde{e}_{1i} = 0$:

$$\begin{bmatrix} \vec{e}_{1i} \\ \vec{e}_{2i} \end{bmatrix} \in \begin{bmatrix} -l_{1i} \operatorname{sig}(\vec{e}_{1i})^{\frac{1}{2}} + \vec{e}_{2i} \\ -l_{2i} \mathbf{K} [\operatorname{sign}(\vec{e}_{1i})] - l_{3i} \vec{e}_{2i} + \mathbf{K} [d_i] \end{bmatrix}$$
$$\partial V_{\tilde{e}i} (\vec{e}_{1i}, \vec{e}_{2i}) = \begin{bmatrix} (\vec{e}_{2i}^2 + 2l_{2i} |\vec{e}_{1i}|) (l_{2i} \operatorname{SGN}(\vec{e}_{1i}) - r_{\tilde{e}} |\vec{e}_{2i}|^2 \operatorname{sign}(\vec{e}_{2i})) \\ (\vec{e}_{2i}^2 + 2l_{2i} |\vec{e}_{1i}|) (\vec{e}_{2i}) - 2r_{\tilde{e}} \vec{e}_{1i} |\vec{e}_{2i}| \end{bmatrix}$$

for $\tilde{e}_{1i} \neq 0, \forall \tilde{e}_{2i} \in \mathbb{R}$:

$$\dot{\widetilde{V}}_{\widetilde{e}i} = -2l_{1i}l_{2i}^{2}|\widetilde{e}_{1i}|^{\frac{3}{2}} - l_{3i}|\widetilde{e}_{2i}|^{4} - l_{1i}l_{2i}|\widetilde{e}_{1i}|^{\frac{1}{2}}|\widetilde{e}_{2i}|^{2} - 2l_{2i}l_{3i}|\widetilde{e}_{1i}||\widetilde{e}_{2i}|^{2} + r_{\widetilde{e}}l_{1i}|\widetilde{e}_{1i}|^{\frac{1}{2}}|\widetilde{e}_{2i}|^{2}\operatorname{sign}(\widetilde{e}_{1i}\widetilde{e}_{2i}) - r_{\widetilde{e}}|\widetilde{e}_{2i}|^{3} + 2r_{\widetilde{e}}l_{2i}|\widetilde{e}_{1i}||\widetilde{e}_{2i}| + 2r_{\widetilde{e}}l_{3i}\widetilde{e}_{1i}|\widetilde{e}_{2i}|^{2}\operatorname{sign}(\widetilde{e}_{2i}) + \mathbf{K}[d_{i}](2l_{2i}|\widetilde{e}_{1i}|\widetilde{e}_{2i} + \widetilde{e}_{2i}^{3} - 2r_{\widetilde{e}}\widetilde{e}_{1i}|\widetilde{e}_{2i}|)$$

for $\tilde{e}_{1i} = 0, \forall \tilde{e}_{2i} \in \mathbb{R}$:

Let

$$\left[\left(l_{2i}\widetilde{e}_{2i}^{2}\right)\xi_{2}-r_{\widetilde{e}}\left|\widetilde{e}_{2i}\right|^{2}\operatorname{sign}(\widetilde{e}_{2i}),\widetilde{e}_{2i}^{3}\right]^{T} \text{ with } \xi_{2} \in [-1, 1] \text{ be an arbitrary element of } \partial V_{\widetilde{e}i}, \text{ then}$$

$$\xi^{\mathrm{T}} \mathbf{K} \begin{bmatrix} \dot{\vec{e}}_{1i} \\ \dot{\vec{e}}_{2i} \end{bmatrix} = \left[\left(l_{2i} \tilde{\vec{e}}_{2i}^{2} \right) \xi_{2} - r_{\tilde{e}} |\tilde{\vec{e}}_{2i}|^{2} \operatorname{sign}(\tilde{\vec{e}}_{2i}), \tilde{\vec{e}}_{2i}^{3} \right] \begin{bmatrix} \tilde{\vec{e}}_{2i} \\ -l_{2i} [-1,1] - l_{3i} \tilde{\vec{e}}_{2i} + \mathbf{K} [d_{i}] \end{bmatrix}$$
$$= \left([\xi_{2} - 1, \xi_{2} + 1] \right) l_{2i} \tilde{\vec{e}}_{2i}^{3} - r_{\tilde{e}} |\tilde{\vec{e}}_{2i}|^{3} - l_{3i} |\tilde{\vec{e}}_{2i}|^{4} + \mathbf{K} [d_{i}] \tilde{\vec{e}}_{2i}^{3}$$

implies

$$\dot{\widetilde{V}}_{\widetilde{e}i} = \bigcap_{\xi_2 \in [-1,1]} ([\xi_2 - 1, \xi_2 + 1]) l_{2i} \widetilde{e}_{2i}^3 - r_{\widetilde{e}} |\widetilde{e}_{2i}|^3 - l_{3i} |\widetilde{e}_{2i}|^4 + \mathbf{K} [d_i] \widetilde{e}_{2i}^3 = -r_{\widetilde{e}} |\widetilde{e}_{2i}|^3 - l_{3i} |\widetilde{e}_{2i}|^4 + \mathbf{K} [d_i] \widetilde{e}_{2i}^3$$

Thus, for all $(\tilde{e}_{1i}, \tilde{e}_{2i}) \in \mathbb{R}^n$, after rearrangement:

$$\dot{V}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i})^{a.e.} \le -\frac{2l_{1i}l_{2i}^{2}}{3} |\tilde{e}_{1i}|^{1.5} - \frac{r_{\tilde{e}}}{2} |\tilde{e}_{2i}|^{3} + \dot{V}_{3} + \dot{V}_{4}$$

where

$$\dot{V}_{3} = -\frac{2l_{1i}l_{2i}^{2}}{3} |\widetilde{e}_{1i}|^{1.5} - \frac{r_{\tilde{e}}}{2} |\widetilde{e}_{2i}|^{3} - l_{3i}(l_{2i} - 2r_{\tilde{e}}) |\widetilde{e}_{1i}| |\widetilde{e}_{2i}|^{2} - l_{1i}(l_{2i} - r_{\tilde{e}}) |\widetilde{e}_{1i}|^{0.5} |\widetilde{e}_{2i}|^{2} + 2r_{\tilde{e}}l_{2i} |\widetilde{e}_{1i}| |\widetilde{e}_{2i}|,$$

$$\dot{V}_{4} = -\frac{2l_{1i}l_{2i}^{2}}{3} |\widetilde{e}_{1i}|^{1.5} - l_{3i}|\widetilde{e}_{2i}|^{4} - l_{2i}l_{3i} |\widetilde{e}_{1i}| |\widetilde{e}_{2i}|^{2} + |d_{i}|(2(l_{2i} + r_{\tilde{e}}))\widetilde{e}_{1i}| |\widetilde{e}_{2i}| + |\widetilde{e}_{2i}|^{3})$$

For \dot{V}_3 :

Applying lemma 2.1,

$$-|\widetilde{e}_{1i}|^{1.5}-|\widetilde{e}_{2i}|^{3} \leq -\left(\frac{3}{2}\right)^{\frac{2}{3}}|\widetilde{e}_{1i}|(3)^{\frac{1}{3}}|\widetilde{e}_{2i}|$$

Thus, if the following inequalities

$$\min\left\{l_{2i}, \frac{l_{2i}}{2}, \left(\frac{3^{\frac{1}{2}}}{2^{2}}\right)l_{1i}l_{2i}^{\frac{1}{2}}\right\} > r_{\tilde{e}}$$

hold then the function \dot{V}_3 is negative definite. Note that such an $r_{\tilde{e}} > 0$ always exists for any $l_{1i} > 0$, $l_{2i} > 0$.

For
$$\dot{V}_4$$
:

The state space is divided into the following three regions for ease of analysis:

$$\begin{split} \phi_{1i}(|d_i|, l_{3i}) &= \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : \widetilde{e}_{1i} \in \mathbb{R}, |\widetilde{e}_{2i}| \ge \beta_4(|d_i|, l_{3i}) \}, \\ \phi_{2i}(|d_i|, l_{3i}) &= \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \ge \beta_3(|d_i|, l_{3i}), |\widetilde{e}_{2i}| \le \beta_4(|d_i|, l_{3i}) \}, \\ \phi_{3i}(|d_i|, l_{3i}) &= \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \le \beta_3(|d_i|, l_{3i}), |\widetilde{e}_{2i}| \le \beta_4(|d_i|, l_{3i}) \}, \end{split}$$

where the functions $\beta_1(.)$ and $\beta_2(.)$ are class κ functions defined by:

$$\beta_{3}(|d_{i}|, l_{3i}) \coloneqq \max\left\{ \left(\frac{\beta_{4}(|d_{i}|, l_{3i})|d_{i}|}{l_{1i}l_{2i}^{2}} 3(l_{2i} + r_{\tilde{e}}) \right)^{2}, \left(\frac{3|d_{i}|[\beta_{4}(|d_{i}|, l_{3i})]^{3}}{l_{1i}l_{2i}^{2}} \right)^{\frac{2}{3}} \right\}, \\ \beta_{4}(|d_{i}|, l_{3i}) \coloneqq \frac{|d_{i}|}{l_{3i}} \times \max\left\{ 1, 2 + \frac{2r_{\tilde{e}}}{l_{2i}} \right\}$$

Consider the region: $\phi_{1i}(|d_i|, l_{3i}) = \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : \widetilde{e}_{1i} \in \mathbb{R}, |\widetilde{e}_{2i}| \ge \beta_4(|d_i|, l_{3i}) \}$

Thus,

$$\dot{V}_{4} = -\frac{2l_{1i}l_{2i}^{2}}{3} |\tilde{e}_{1i}|^{1.5} - |\tilde{e}_{2i}|^{3} (l_{3i}|\tilde{e}_{2i}| - |d_{i}|) - |\tilde{e}_{1i}||\tilde{e}_{2i}| (l_{2i}l_{3i}|\tilde{e}_{2i}| - 2|d_{i}|(l_{2i} + r_{\tilde{e}})) \le 0$$

for the states in this region.

Next, consider the region: $\phi_{2i}(|d_i|, l_{3i}) = \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \ge \beta_3(|d_i|, l_{3i}), |\widetilde{e}_{2i}| \le \beta_4(|d_i|, l_{3i}) \}$ Thus,

$$\begin{split} \dot{V}_{4} &\leq -\frac{2l_{1i}l_{2i}^{2}}{3} \left| \widetilde{e}_{1i} \right|^{1.5} - l_{3i} \left| \widetilde{e}_{2i} \right|^{4} - l_{2i}l_{3i} \left| \widetilde{e}_{1i} \right| \left| \widetilde{e}_{2i} \right|^{2} + \left| d_{i} \right| \left(2(l_{2i} + r_{\widetilde{e}}) \widetilde{e}_{1i} \right) \beta_{4} \left(\left| d_{i} \right|, l_{3i} \right) + \left[\beta_{4} \left(\left| d_{i} \right|, l_{3i} \right) \right]^{3} \right) \\ &\leq - \left| \widetilde{e}_{1i} \left(\frac{l_{1i}l_{2i}^{2}}{3} \left| \widetilde{e}_{1i} \right|^{0.5} - \left| d_{i} \right| \left(l_{2i} + r_{\widetilde{e}}) \beta_{4} \left(\left| d_{i} \right|, l_{3i} \right) \right) - \left(\frac{l_{1i}l_{2i}^{2}}{3} \left| \widetilde{e}_{1i} \right|^{1.5} - \left| d_{i} \right| \left[\beta_{4} \left(\left| d_{i} \right|, l_{3i} \right) \right]^{3} \right) \end{split}$$

where $\dot{V}_4 \leq 0$ for

$$\left|\widetilde{e}_{1i}\right| \geq \max\left\{ \left(\frac{\beta_4(|d_i|, l_{3i})|d_i|}{l_{1i}l_{2i}^2} 3(l_{2i} + r_{\widetilde{e}}) \right)^2, \left(\frac{3|d_i| \left[\beta_4(|d_i|, l_{3i})\right]^3}{l_{1i}l_{2i}^2} \right)^{\frac{2}{3}} \right\}$$

which is sufficiently satisfied for the states in this region.

Next consider the compact set:

$$\phi_{3i}(|d_i|, l_{3i}) = \{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \leq \beta_3(|d_i|, l_{3i}), |\widetilde{e}_{2i}| \leq \beta_4(|d_i|, l_{3i}) \}$$

Note that, $\dot{V}_4 \leq 0$ for $(\tilde{e}_{1i}, \tilde{e}_{2i}) \notin \phi_3(|d_i|, l_{3i})$.

Thus,

$$\dot{V}_{\tilde{e}i}\left(\widetilde{e}_{1i},\widetilde{e}_{2i}\right)^{a.e.} \leq -\frac{2l_{1i}l_{2i}^{2}}{3}\left|\widetilde{e}_{1i}\right|^{1.5} - \frac{r_{\tilde{e}}}{2}\left|\widetilde{e}_{2i}\right|^{3} < 0 \text{ for } \left(\widetilde{e}_{1i},\widetilde{e}_{2i}\right) \notin \phi_{3i}\left(\left|d_{i}\right|,l_{3i}\right)$$

Next, define a Lyapunov level set:

$$\Sigma_{\widetilde{e}i}(|d_i|, l_{3i}) = \left\{ (\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathbb{R}^2 : V_{\widetilde{e}i} \le \rho_{2i}(|d_i|, l_{3i}) \right\}$$

where the positive definite function ρ_{2i} ($|d_i|$, l_{3i}) is defined as follows:

$$\rho_{2i}(d_i|, l_{3i}) = \max_{(\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in bd \ \phi_3(|d_i|, l_{3i})} V_{\widetilde{e}i}(\widetilde{e}_{1i}, \widetilde{e}_{2i})$$

which exists since the boundary of the set is compact and $V_{\tilde{e}i}$ is continuous. Then we observe that $\phi_3(|d_i|, l_{3i}) \subset \Sigma_{\tilde{e}i}(|d_i|, l_{3i})$ As a result, we have

$$\frac{d}{dt}V_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \stackrel{a.e.}{\in} \dot{\tilde{V}}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \leq -\frac{2l_{1i}l_{2i}^{2}}{3}|\tilde{e}_{1i}|^{1.5} - \frac{r_{\tilde{e}}}{2}|\tilde{e}_{2i}|^{3}, \text{ for } V_{\tilde{e}i} \geq \rho_{2i}(|d_i|,l_{3i})$$

which implies that each of the trajectories for the *i*-th planar system will enter their respective compact level set $\sum_{\tilde{e}i} (|d_i|, l_{3i})$ in finite time and stay in it once entered.

6.2 Manipulator Dynamics

The dynamic model of a rigid *n*-link serial nonredundant robot manipulator, with all actuated revolute joints described in joint coordinates, is given as follows:

$$M(q)\ddot{q} + C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D(q,\dot{q},t) = \tau$$
(6-2)

where $q, \dot{q}, \ddot{q} \in \mathbb{R}^n$ are, respectively, the position (generalized coordinates), velocity and acceleration vectors, $M(q) \in \mathbb{R}^{n \times n}$ is the inertia matrix, and $C(q, \dot{q})\dot{q}, F\dot{q}, G(q), \tau \in \mathbb{R}^n$ are, respectively, the vectors of Coriolis and centrifugal, viscous friction, gravity, and torque input, with *F* being a constant, positive definite, diagonal (viscous friction coefficient) matrix. $D(q, \dot{q}, t)$ is a an additional bounded uncertainty or perturbation term. (Note that no continuity assumption is made so that discontinuous models of friction may be used in *D*). The robot manipulator satisfies the same properties as that in Chapter 3 and will not be repeated here. In this chapter, only joint positions measurement is available. No velocity information from the system is assumed in this chapter. As such, the following development will follow an output feedback approach. In addition, let us suppose that the absolute value of each joint input τ_i is constrained to be smaller than a given saturation bound $T_{i, \max} > 0$, i.e., $|\tau_i| \leq T_{i, \max}, \forall i = 1, ..., n$.

The control objective here is to design a robust output feedback saturated control that ensures the robot configuration vector q tracks a desired trajectory vector, $q_d(t)$ with an ultimately bounded error that can be made as small as required globally even under the presence of bounded disturbances.

The desired trajectory vector, $q_d(t)$ is assumed to be twice continuously differentiable

vector-function such that $||q_d(t)||$, $||\dot{q}_d(t)||$, and $||\ddot{q}_d(t)||$ are bounded by *a priori* known constants. While the type of disturbances considered here has to be upper bounded by a constant, due to the limited actuation (Note that no continuity assumption is made so that discontinuous models of friction may be used in *D*), i.e.

$$||D|| \le \operatorname{sat}_{p_5} \left[p_0 + p_1 ||e_1|| + p_2 ||\dot{q}|| + p_3 ||e_1||^2 + p_4 ||\dot{q}||^2 \right]$$

where $||D|| := \sup\{||\varsigma|| : \varsigma \in K[D]\}$, p_0, p_1, p_2, p_3, p_4 , p_5 are some nonnegative constants, $e_1 = q - q_d \in \mathbb{R}^n$, and $e_2 = \dot{q} - \dot{q}_d \in \mathbb{R}^n$.

6.2.1 Control Development

Under these constraints, the following controller is proposed,

$$\tau = -K \operatorname{sat}\left[\operatorname{sig}(\mu^{-1}\hat{s})^{a}\right], \qquad (6-3)$$

where K and μ are positive definite diagonal matrices, i.e. $K = \text{diag}\{k_i\}_{i=1}^n$, with $k_i > 0$, $\mu = \text{diag}\{\mu_i\}_{i=1}^n$, with $\mu_i > 0$, $a \ge 0$ constant, and $\hat{s} \in \mathbb{R}^n$

i.e.
$$\tau_i = -k_i \operatorname{sat}\left(\frac{|\hat{s}_i|^a}{\mu_i^a}\operatorname{sign}(\hat{s}_i)\right)$$
 for $i = 1 \dots n$.

When a = 0, the control becomes a discontinuous control law,

$$\tau = -K \operatorname{sign}(\hat{s}), \ \hat{s} \in \mathbb{R}^n$$

Note that the control is bounded, i.e. such that $|\tau_i(t)| \le k_i$, for i = 1, ..., n, $\forall t \ge 0$. The \hat{s} is the velocity-estimate-based desired error dynamics defined as $\hat{s} = \hat{e}_2 + \sigma$, with

$$\dot{\sigma} = -K_a \sigma + K_a \mu \operatorname{sat}\left[\mu^{-1} \hat{s}\right] - K_a \operatorname{sat}_{\left(\frac{\varepsilon_2}{k_2}\right)^{\frac{1+b}{2b}}} \left[\hat{e}_2\right] + \operatorname{sat}_{\varepsilon_2}\left[K_2 \operatorname{sig}(\hat{e}_2)^{\frac{2b}{1+b}}\right] + \operatorname{sat}_{\varepsilon_1}\left[K_1 \operatorname{sig}(e_1)^b\right] \quad (6-4)$$

where K_a , K_1 , and K_2 are positive definite diagonal matrices, i.e. $K_a = \operatorname{diag}\{k_{ai}\}_{i=1}^n$, with $k_{ai} > 0$, $K_1 = \operatorname{diag}\{k_{1i}\}_{i=1}^n$, with $k_{1i} > 0$, $K_2 = \operatorname{diag}\{k_{2i}\}_{i=1}^n$, with $k_{2i} > 0$, $k_2 \in \{k_{21}, \dots, k_{2n}\}$, $\varepsilon_1 \in \{\varepsilon_{11}, \dots, \varepsilon_{1n}\}$, with $\varepsilon_{1i} > 0$, $\varepsilon_2 \in \{\varepsilon_{21}, \dots, \varepsilon_{2n}\}$, with $\varepsilon_{2i} > 0$, $\forall i = 1, \dots, n$, and $b \ge 0$

constant, with $\varepsilon_{1i} > k_{ai}\mu_i + k_{ai}\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}} + \varepsilon_{2i}$, and $\hat{e}_2 \in \mathbb{R}^n$,

i.e.

$$\dot{\sigma}_{i} = -k_{ai}\sigma_{i} + k_{ai}\mu_{i}\operatorname{sat}\left[\frac{\hat{s}_{i}}{\mu_{i}}\right] - k_{ai}\operatorname{sat}_{\left(\frac{\hat{s}_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}}\left[\hat{e}_{2i}\right] + \operatorname{sat}_{\varepsilon_{2i}}\left[k_{2i}|\hat{e}_{2i}|^{\frac{2b}{1+b}}\operatorname{sign}(\hat{e}_{2i})\right] + \operatorname{sat}_{\varepsilon_{1}}\left[k_{1i}|e_{1i}|^{b}\operatorname{sign}(e_{1i})\right]$$

for i = 1 ... n.

Note that when b = 0, let the element of ε_1 equal to that of K_1 , and ε_2 to that of K_2 , i.e. $\varepsilon_{1n} = k_{1n}, \forall i \in N \text{ and } \varepsilon_{2n} = k_{2n}, \forall i \in N, \text{ i.e.}$

$$\dot{\sigma}_i = -k_{ai}\sigma_i + k_{ai}\mu_i \operatorname{sat}\left[\frac{\hat{s}_i}{\mu_i}\right] - k_{ai}\operatorname{sat}\left[\hat{e}_{2i}\right] + k_{2i}\operatorname{sign}(\hat{e}_{2i}) + k_{1i}\operatorname{sign}(e_{1i})$$

for i = 1 ... n.

Note that \hat{e}_2 is the output of the observer defined as:

$$\dot{\hat{e}}_{1} = -L_{1} \operatorname{sig}(\tilde{e}_{1})^{p} + \hat{e}_{2} ,$$

$$\dot{w} = -L_{2} \operatorname{sig}(\tilde{e}_{1})^{2p-1} - L_{3}\hat{e}_{2}$$

$$\hat{e}_{2} = w + L_{3}e_{1}$$
(6 - 5)

where $\hat{e}_1, \hat{e}_2 \in \mathbb{R}^n$, L_1, L_2 , and L_3 are positive definite diagonal matrices, i.e. $L_1 = \text{diag}\{l_{1i}\}_{i=1}^n$, with $l_{1i} > 0$, $L_2 = \text{diag}\{l_{2i}\}_{i=1}^n$, with $l_{2i} > 0$, and $L_3 = \text{diag}\{l_{3i}\}_{i=1}^n$, with $l_{3i} > 0$, $\forall i = 1, ..., n$, and $0.5 \le p \le 1$. Let us define $\tilde{e}_1 = \hat{e}_1 - e_1 \in \mathbb{R}^n$, and $\tilde{e}_2 = \hat{e}_2 - e_2 \in \mathbb{R}^n$, then, the closed-loop system (6 - 2), (6 - 3), (6 - 4) and (6 - 5) can be written as

$$\begin{split} \dot{\sigma} &= -K_a \sigma + K_a \mu \operatorname{sat} \left[\mu^{-1} \hat{s} \right] - K_a \operatorname{sat}_{\left(\frac{\delta^2}{k_2} \right)^{\frac{1+\delta}{2b}}} \left[\hat{e}_2 \right] \\ &+ \operatorname{sat}_{\varepsilon_2} \left[K_2 \operatorname{sig}(\hat{e}_2)^{\frac{2b}{1+b}} \right] + \operatorname{sat}_{\varepsilon_1} \left[K_1 \operatorname{sig}(e_1)^b \right] \\ \dot{e}_1 &= e_2, \\ \dot{e}_2 &= -M^{-1}(q) K \operatorname{sat} \left[\operatorname{sig}(\mu^{-1} \hat{s})^a \right] - M^{-1}(q) (C(q, \dot{q}) \dot{q} + F \dot{q} + G(q) + D) - \ddot{q}_d, \\ \dot{\tilde{e}}_1 &= -L_1 \operatorname{sig}(\tilde{e}_1)^p + \tilde{e}_2 , \\ \dot{\tilde{e}}_2 &= -L_2 \operatorname{sig}(\tilde{e}_1)^{2p-1} - L_3 \tilde{e}_2 \\ &- \left(-M^{-1}(q) K \operatorname{sat} \left[\operatorname{sig}(\mu^{-1} \hat{s})^a \right] - M^{-1}(q) (C(q, \dot{q}) \dot{q} + F \dot{q} + G(q) + D) - \ddot{q}_d \right) \end{split}$$

To rewrite the closed-loop system in a form convenient for analysis, let us define $v_1 = e_1$, $v_2 = -\sigma$, with $s = e_2 + \sigma = \hat{s} - (\hat{e}_2 - e_2) = \hat{s} - \tilde{e}_2$, and we obtain the following form of closedloop system,

$$\begin{split} \dot{v}_{1} &= v_{2} + s, \\ \dot{v}_{2} &= -K_{a}v_{2} - K_{a}\mu \operatorname{sat}\left[\mu^{-1}(s+\tilde{e}_{2})\right] + K_{a}\operatorname{sat}_{\left(\frac{e_{2}}{k_{2}}\right)^{\frac{1+b}{2b}}}\left[v_{2} + s + \tilde{e}_{2}\right] \\ &- \operatorname{sat}_{\varepsilon_{2}}\left[K_{2}\operatorname{sig}(v_{2} + s + \tilde{e}_{2})^{\frac{2b}{1+b}}\right] - \operatorname{sat}_{\varepsilon_{1}}\left[K_{1}\operatorname{sig}(v_{1})^{b}\right], \\ \dot{s} &= -M^{-1}(z_{1} + q_{d})K\operatorname{sat}\left[\operatorname{sig}(\mu^{-1}\hat{s})^{a}\right] - M^{-1}(q)(C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D) - \ddot{q}_{d} + \dot{\sigma}, \quad (6-6) \\ \dot{\tilde{e}}_{1} &= -L_{1}\operatorname{sig}(\tilde{e}_{1})^{p} + \tilde{e}_{2}, \\ \dot{\tilde{e}}_{2} &= -L_{2}\operatorname{sig}(\tilde{e}_{1})^{2p-1} - L_{3}\tilde{e}_{2} \\ &- \left(-M^{-1}(q)K\operatorname{sat}\left[\operatorname{sig}(\mu^{-1}\hat{s})^{a}\right] - M^{-1}(q)(C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D) - \ddot{q}_{d}\right) \end{split}$$

6.2.2 Stability Analysis

Theorem 6.1: Consider the robot dynamics (6 - 2) with the bounded controller given by (6 - 3), and the observer (6 - 5), global practical trajectory tracking of the desired trajectory, q_d can be assured, even under bounded disturbances and without velocity measurement, provided that the gain *K* is sufficiently large up to the saturation bound $T_{i,\max}$, $\forall i = 1,...,n$, and the observer gain L_3 sufficiently large and the desired trajectory sufficiently slow.

Proof of Theorem 6.1: The stability analysis will proceeds in three steps.

- 1. First we will show that σ and *s* are bounded globally, and then the observer is perturbed by bounded disturbances only.
 - a. First we will show the boundedness of σ , by the following Lyapunov function

$$V_{\sigma} = \frac{1}{2}\sigma_i^2, \quad \forall i = 1, \dots, n.$$
Its time derivative along its solution (6 - 4), gives rise to

$$\begin{split} \dot{V}_{\sigma} &= -k_{ai}\sigma_{i}^{2} + \sigma_{i}k_{ai}\mu_{i}\operatorname{sat}\left(\frac{\hat{s}_{i}}{\mu_{i}}\right) - \sigma_{i}k_{ai}\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}}(\hat{e}_{2i}) + \sigma_{i}\operatorname{sat}_{\varepsilon_{2i}}\left(k_{2i}|\hat{e}_{2i}|^{\frac{2b}{1+b}}\operatorname{sign}(\hat{e}_{2i})\right) \\ &+ \sigma_{i}\operatorname{sat}_{\varepsilon_{1i}}\left(k_{1i}|e_{1i}|^{b}\operatorname{sign}(e_{1i})\right) \\ &\leq -\frac{k_{ai}}{2}\sigma_{i}^{2} - \frac{k_{ai}}{2}\sigma_{i}^{2} + |\sigma_{i}\left(k_{ai}\mu_{i} + k_{ai}\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}} + \varepsilon_{2i} + \varepsilon_{1i}\right) \\ &\therefore |\sigma_{i}| \leq 2\left(\mu_{i} + \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}} + \frac{\varepsilon_{2i}}{k_{ai}} + \frac{\varepsilon_{1i}}{k_{ai}}\right), \end{split}$$

Since μ_i , ε_{1i} , $\varepsilon_{2i} > 0$ are constants, $|\sigma_i|$ is bounded. Also, from (6 - 4), we can see that $|\sigma_i|$ is bounded as well.

b. Next, we proceed to show the boundedness of s by the following Lyapunov function

$$V_{s} = \frac{1}{2} s^{\mathrm{T}} M(.) s, \text{ with } \frac{1}{2} \underline{m} \|s\|^{2} \leq V_{s} \leq \frac{1}{2} \overline{m} \|s\|^{2},$$

Note that $\dot{q} = s + (\dot{q}_d - \sigma)$,

The *s* dynamics is described by,

$$\dot{s} = -M^{-1}(q)K\operatorname{sat}\left[\operatorname{sig}(\mu^{-1}\hat{s})^{a}\right] - M^{-1}(q)(C(q,\dot{q})\dot{q} + F\dot{q} + G(q) + D) - \ddot{q}_{d} + \dot{\sigma}$$
(6 - 7)

Since the differential equation (6 - 7) has discontinuous right-hand side, i.e. when a = 0 or since no continuity assumption is made on *D*, its solutions are understood in the sense of Filippov (see definition 2.1), and in accordance to lemma 2.4, the time derivative of the Lyapunov function V_s along the dynamics (6 - 7) for all $a \ge 0$:

$$\begin{split} \dot{V}_{s}^{a,e} &\in \dot{\tilde{V}}_{s} \\ \dot{\tilde{V}}_{s} &= \bigcap_{\xi \in \partial V_{s}} \xi^{\mathrm{T}} \binom{K[f](s)}{1} = \nabla V_{s}^{\mathrm{T}} K[f](s) + \frac{1}{2} s^{\mathrm{T}} \dot{M}(\cdot) s \\ &\subset \left[s^{\mathrm{T}} M(\cdot) \right] K \begin{bmatrix} -M^{-1}(q) K \mathrm{sat} \left[\mathrm{sig}(\mu^{-1} \hat{s})^{a} \right] \\ -M^{-1}(q) (C(q,\dot{q}) \dot{q} + F \dot{q} + G(q) + D) - \ddot{q}_{d} + \dot{\sigma} \end{bmatrix} + \frac{1}{2} s^{\mathrm{T}} \dot{M}(\cdot) s \\ &= -s^{\mathrm{T}} K K \left[\mathrm{sat} \left[\mathrm{sig}(\mu^{-1} \hat{s})^{a} \right] \right] + s^{\mathrm{T}} (-G(q) - M(q) \ddot{q}_{d} + M(q) \dot{\sigma}) - s^{\mathrm{T}} K[D] - s^{\mathrm{T}} F \dot{q} \\ -s^{\mathrm{T}} C(q, \dot{q}) \dot{q} + s^{\mathrm{T}} C(q, \dot{q}) s \end{split}$$

$$= -s^{\mathrm{T}} K K \left[\mathrm{sat} \left[\mathrm{sig}(\mu^{-1} \hat{s})^{a} \right] \right] + s^{\mathrm{T}} (-G(q) - M(q) \ddot{q}_{d} + M(q) \dot{\sigma}) - s^{\mathrm{T}} K[D] \\ -s^{\mathrm{T}} F(\dot{q}_{d} - \sigma) - s^{\mathrm{T}} [C(q, (\dot{q}_{d} - \sigma))] (\dot{q}_{d} - \sigma) - s^{\mathrm{T}} F s - s^{\mathrm{T}} [C(q, (\dot{q}_{d} - \sigma))] s \\ \leq -s^{\mathrm{T}} K K \left[\mathrm{sat} \left[\mathrm{sig}(\mu^{-1} \hat{s})^{a} \right] \right] + s^{\mathrm{T}} \Delta_{1}(\cdot) - \| s \|^{2} (\underline{f} - C_{m} \| (\dot{q}_{d} - \sigma) \|) \end{split}$$

$$\leq \|s\|(\|K\| + \|\Delta_1(\cdot)\|) - \|s\|^2 (\underline{f} - C_m\|(\dot{q}_d - \sigma)\|)$$
(6-8)

with $\|\Delta_1\| \coloneqq \sup\{\|\varepsilon\|: \varepsilon \in \mathbf{K}[\Delta_1(.)]\}\)$, and the vector

$$\Delta_{1}(\cdot) = -G(q) - M(q)\ddot{q}_{d} + M(q)\dot{\sigma} - \mathbf{K}[D] - F(\dot{q}_{d} - \sigma) - C(q, (\dot{q}_{d} - \sigma))(\dot{q}_{d} - \sigma),$$

and $\left\|\mathbf{K}\left[\operatorname{sat}\left[\operatorname{sig}(\mu^{-1}\hat{s})^{a}\right]\right]\right\| \leq 1.$

Note that the upper bound of the vector $\Delta_1(.)$ is determined by the physical properties of the robot dynamics, desired trajectory, and the desired error dynamics. Since \dot{q}_d , \ddot{q}_d , are bounded from the assumptions and σ , σ are bounded (see step (1-a)), the vector Δ_1 is bounded as well. Hence from [143], the multi-valued function $K[\Delta_1]$ is bounded as well. Thus, if

$$\underline{f} > C_m \| (\dot{q}_d - \sigma) \|$$

Define $k_c := \underline{f} - C_m \| (\dot{q}_d - \sigma) \| > 0$, which is possible by selecting appropriate desired trajectory and the desired error dynamics. Then we have,

$$\begin{split} \dot{V}_{s} \stackrel{a.e.}{\leq} &- \left(\frac{k_{c}}{2}\right) \|s\|^{2} - \left(\frac{k_{c}}{2}\right) \|s\| \left(\|s\| - \frac{2\left(\|K\| + \|\Delta_{1}\|\right)}{k_{c}}\right) \\ \dot{V}_{s} \stackrel{a.e.}{<} 0, \text{ for } \|s\| \geq \left(\frac{2\left(\|K\| + \|\Delta_{1}\|\right)}{k_{c}}\right) \Longrightarrow \text{ for } V_{s} \geq \frac{\overline{m}}{2} \left(\frac{2\left(\|K\| + \|\Delta_{1}\|\right)}{k_{c}}\right)^{2} \end{split}$$

Hence, ||s|| is bounded by

$$\Rightarrow \|s\| \leq \frac{\sqrt{\overline{m}}}{\sqrt{\underline{m}}} \left(\frac{2(\|K\| + \|\Delta_1\|)}{k_c} \right)$$

Note that from above:

$$\dot{q} = s + (\dot{q}_d - \sigma),$$

$$\therefore \|\dot{q}\| = \|s + (\dot{q}_d - \sigma)\| \le \|s\| + \|\dot{q}_d\| + \|\sigma\|$$

where the velocity is bounded globally since the right hand side of the inequality is bounded, i.e. *s* is upper bounded by the $\Delta_1(.)$ and the control gain *K*, while desired trajectory and desired error dynamics are upper bounded by design.

c. The observer dynamics, from (6 - 6),

For ∀ *i*=1, ..., *n*:

$$\begin{split} \widetilde{e}_{1i} &= -l_{1i} \operatorname{sig}(\widetilde{e}_{1i})^p + \widetilde{e}_{2i} \quad ,\\ \dot{\widetilde{e}}_{2i} &= l_{2i} \operatorname{sig}(\widetilde{e}_{1i})^{2p-1} - l_{3i} \widetilde{e}_{2i} + \Delta_{2i}(\cdot) \end{split}$$

where

$$\Delta_{2}(\cdot) = (\Delta_{2i}(\cdot), \dots, \Delta_{2n}(\cdot))^{\mathrm{T}}, \quad \forall i = 1, \dots, n, \text{ and}$$

$$\Delta_{2}(\cdot) = M^{-1}(q)K \text{sat} \Big[\mu^{-1} \text{sig}(\hat{s})^{a} \Big] + M^{-1}(q)C(q, \dot{q})\dot{q} + M^{-1}(q)F\dot{q} + M^{-1}(q)G(q) + M^{-1}(q)D + \ddot{q}_{d}, \qquad (6 - 9)$$

Let us define $|\Delta_{2i}| := \sup \{ |\varepsilon_i| : \varepsilon_i \in \mathbf{K}[\Delta_{2i}(\cdot)] \}.$

Remark 6.2. Note that $\Delta_2(.)$ is upper bounded by the physical properties of the robot dynamics, desired trajectory, control gain *K*, and velocity of the robot manipulator. Since \dot{q}_d , \ddot{q}_d , are bounded from the assumptions, σ , \dot{q} and *s* are bounded (see step (1-a to 1-b)), the vector Δ_2 is bounded as well. Hence from [143], the multi-valued function $K[\Delta_{2i}]$ is bounded as well.

To show the boundedness of $(\tilde{e}_1, \tilde{e}_2)$, consider the following Lyapunov function (see

section 6.1),
$$V_{\tilde{e}i}(\tilde{e}_{1i}, \tilde{e}_{2i}) = \left(\frac{1}{2}\tilde{e}_{2i}^{2} + \frac{l_{2i}}{2p}|\tilde{e}_{1i}|^{2p}\right)^{2} - r_{\tilde{e}}|\tilde{e}_{1i}||\tilde{e}_{2i}|^{\frac{4p-1}{p}}\operatorname{sign}(\tilde{e}_{1i}\tilde{e}_{2i})$$

The differential inequalities for the $(\tilde{e}_1, \tilde{e}_2)$ -subsystem satisfy (see section 6.1):

$$\dot{V}_{\tilde{e}i}(\tilde{e}_{1i},\tilde{e}_{2i}) \stackrel{a.e.}{\leq} \begin{cases} -\frac{l_{1i}l_{2i}^{2}}{3p} |\tilde{e}_{1i}|^{5p-1} - \frac{r_{\tilde{e}}}{2} |\tilde{e}_{2i}|^{\frac{5p-1}{p}} & \text{for } V_{\tilde{e}i} \geq \rho_{1i}(|\Delta_{2i}|,l_{3i}) \text{ and } 0.5$$

where the functions ho_{1i} and ho_{2i} are of class- \mathcal{KL} ,

$$\rho_{1i}(|\Delta_{2i}|, l_{3i}) \coloneqq \max_{(\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathrm{bd} \ \phi_3(|\Delta_{2i}|, l_{3i})} V_{\widetilde{e}i}(\widetilde{e}_{1i}, \widetilde{e}_{2i}),$$

$$\rho_{2i}(|\Delta_{2i}|, l_{3i}) = \max_{(\widetilde{e}_{1i}, \widetilde{e}_{2i}) \in \mathrm{bd} \ \phi_3(|\Delta_{2i}|, l_{3i})} V_{\widetilde{e}i}(\widetilde{e}_{1i}, \widetilde{e}_{2i}),$$

bd $\phi_{3i}(|\Delta_{2i}|, l_{3i})$ is the boundary of the region

$$\begin{split} \phi_{3i}(\Delta_{2i}|,l_{3i}) \\ &= \begin{cases} \{ (\widetilde{e}_{1i},\widetilde{e}_{2i}) \in \mathbb{R}^2 : |\widetilde{e}_{1i}| \le \beta_1(|\Delta_{2i}|,l_{3i}), |\widetilde{e}_{2i}| \le \beta_2(|\Delta_{2i}|,l_{3i}) \}, \text{ for } 0.5$$

which is a compact set, with functions β_1 , β_2 , β_3 , and β_4 are of class- \mathcal{KL} . (See section

Define the following Lyapunov level set,

$$\Sigma_{\tilde{e}i}(|\Delta_{2i}|, l_{3i}) = \begin{cases} \langle (\tilde{e}_{1i}, \tilde{e}_{2i}) \in \mathbb{R}^2 : V_{\tilde{e}i} \leq \rho_{1i}(|\Delta_{2i}|, l_{3i}) \rangle, & \text{for } 0.5$$

Thus, it implies that each of the trajectories for the *i*-th planar system will enter their respective compact level set $\sum_{\tilde{e}i}(|d_i|, l_{3i})$ in finite time and stay in it once entered.

In particular, once inside the compact set $\sum_{\tilde{e}i}(|d_i|, l_{3i})$ and since $\underline{\pi}_{2i}\left(|\tilde{e}_{2i}|^4 + |\tilde{e}_{1i}|^{4p}\right) \leq V_{\tilde{e}i}$,

it implies

$$\begin{aligned} \forall i = 1, \dots, n: \\ |\widetilde{e}_{2i}| &\leq \begin{cases} \left(\frac{\rho_{1i}(|\Delta_{2i}|, l_{3i})}{\underline{\pi}_{2i}}\right)^{\frac{1}{4}}, \text{ for } 0.5
$$(6 - 10)$$$$

Hence, it can be observed that for $0.5 \le p \le 1$, the increase of observer gain L_3 will result in a smaller upper bound on the observation errors, since the functions ρ_{1i} and ρ_{2i} are of class- \mathcal{KL} .

Remark 6.3. It is worth mentioning that increasing the control gain *K*, the upper bound of the vector $\Delta_2(.)$ will increase as well (see (6 - 9)), requiring a larger observer gains L_3 to obtain the same upper bounds on the observation errors.

2. Now recall the analysis for *s*-dynamics, with this new-found bound on observation error, we are going to show that the bound on *s* can be made arbitrarily small; in particular the

control will be unsaturated.

Recall that from (6 - 8), since $k_c := F_m - C_m ||(\dot{q}_d - \sigma)|| > 0$

$$\dot{V}_{s} \stackrel{a.e.}{\leq} - s^{\mathrm{T}} K \mathbf{K} \left[\operatorname{sat} \left[\operatorname{sig} \left(\mu^{-1} \hat{s} \right)^{a} \right] \right] + s^{\mathrm{T}} \Delta_{1} (\cdot)$$

and $s = \hat{s} - \tilde{e}_2$,

a. For a = 0:

Then we can rewrite the above inequality as:

$$\dot{V}_{s} \stackrel{a.e.}{\leq} -s^{\mathrm{T}} K \mathbf{K}[\operatorname{sign}(s+\widetilde{e}_{2})] + s^{\mathrm{T}} \Delta_{1}(\cdot) = -s^{\mathrm{T}} K \mathbf{K}[\operatorname{sign}(s)] + s^{\mathrm{T}} \Delta_{1}(\cdot) - s^{\mathrm{T}} K \mathbf{K}[\operatorname{sign}(s+\widetilde{e}_{2})] + s^{\mathrm{T}} K \mathbf{K}[\operatorname{sign}(s)]$$

Using the following notation:

$$N^{+}(s) = \{i \in \{1,...,n\}: s_{i} \neq 0\}, N^{0}(s) = \{i \in \{1,...,n\}: s_{i} = 0\},$$

 $-s^{T} KK[sign(s)] = -\sum_{i \in N^{+}(s)} k_{i}|s_{i}| - \sum_{i \in N^{0}(s)} k_{i}(0) \times [-1,+1]$
 $= -\sum_{i \in N^{+}(s)} k_{i}|s_{i}| - 0$
 $= -\sum_{i=i}^{n} k_{i}|s_{i}|$

Thus,

$$\dot{V}_{s} \stackrel{a.e.}{\leq} -\sum_{i=i}^{n} k_{i} |s_{i}| + s^{\mathrm{T}} \Delta_{1}(\cdot) - s^{\mathrm{T}} K \left(\mathbf{K} [\operatorname{sign}(s + \widetilde{e}_{2})] - \mathbf{K} [\operatorname{sign}(s)] \right)$$

$$\leq -\sum_{i=i}^{n} k_{i} |s_{i}| + s^{\mathrm{T}} \Delta_{1}(\cdot) - \sum_{i=1}^{n} s_{i} k_{i} \left(\mathbf{K} [\operatorname{sign}(s_{i} + \widetilde{e}_{2i})] - \mathbf{K} [\operatorname{sign}(s_{i})] \right)$$

Note that:

 $\boldsymbol{K}[\operatorname{sign}(s_i + \widetilde{e}_{2i})] - \boldsymbol{K}[\operatorname{sign}(s_i)] \leq 2,$

Also, for
$$|s_i| > |\widetilde{e}_{2i}| > 0$$
:
 $K[\operatorname{sign}(s_i + \widetilde{e}_{2i})] - K[\operatorname{sign}(s_i)] = (\operatorname{sign}(s_i + \widetilde{e}_{2i}) - \operatorname{sign}(s_i))$
 $= \operatorname{sign}(s_i) - \operatorname{sign}(s_i)$
 $= 0$
 $\therefore |s_i k_i K[(\operatorname{sign}(s_i + \widetilde{e}_{2i}) - \operatorname{sign}(s_i))] \le 2k_i |\widetilde{e}_{2i}|$

Hence we have:

$$\begin{split} \dot{V}_{s} \stackrel{a.e.}{\leq} &- \sum_{i=i}^{n} k_{i} |s_{i}| + \sum_{i=1}^{n} |s_{i}| |\Delta_{1i}| + \sum_{i=1}^{n} 2k_{i} |\widetilde{e}_{2i}| \\ &\leq -(1-\theta) \sum_{i=i}^{n} k_{i} |s_{i}| + \sum_{i=1}^{n} 2k_{i} |\widetilde{e}_{2i}| - \sum_{i=i}^{n} \theta k_{i} |s_{i}| + \sum_{i=1}^{n} |s_{i}| |\Delta_{1i}| \\ &\leq -(1-\theta) \lambda_{\min}(K) ||s|| + 2\sqrt{n} \lambda_{\max}(K) ||\widetilde{e}_{2}|| - \sum_{i=i}^{n} |s_{i}| (\theta k_{i} - |\Delta_{1i}|) \\ &\leq -(1-\theta) \lambda_{\min}(K) \frac{\sqrt{2}}{\sqrt{m}} V_{s}^{\frac{1}{2}} + 2\sqrt{n} \lambda_{\max}(K) ||\widetilde{e}_{2}|| - \sum_{i=i}^{n} |s_{i}| (\theta k_{i} - |\Delta_{1i}|) \end{split}$$

where the following property has been employed, $\|x\|_{2} \leq \|x\|_{1} \leq \sqrt{n} \|x\|_{2}, \forall x \in \mathbb{R}^{n}$

$$\therefore \dot{V}_{s} \stackrel{a.e.}{\leq} -\sum_{i=i}^{n} |s_{i}| (\theta k_{i} - |\Delta_{1i}|) < 0, \quad \text{for} \quad V_{s} \ge \left(\frac{\overline{m}}{2}\right) \left(\left(\frac{2\sqrt{n}}{(1-\theta)}\right) \left(\frac{\lambda_{\max}(K)}{\lambda_{\min}(K)}\right) \|\widetilde{e}_{2}\|\right)^{2}, \text{ with}$$

$$k_{i} > \frac{|\Delta_{1i}|}{\theta}, \ \forall i \in N.$$

Remark 6.4. Note that for a given desired trajectory and given desired error dynamics (hence an upper bound on the vector $\Delta_1(.)$), there exist k_i such that the above condition

on control gains *K* is satisfied. Accordingly, for a given control gain *K*, desired trajectory, and desired error dynamics (hence an upper bound on the vector $\Delta_2(.)$), there exist observer gains L_3 such that the observation velocity error \tilde{e}_2 is bounded.

Since \tilde{e}_2 is bounded, the state *s* will reach the above Lyapunov level set in finite time and stay in it for all future times. In particular, the ultimate bound on *s* can be obtained as:

$$\|s\| \le \left(\frac{\overline{m}}{\underline{m}}\right)^{\frac{1}{2}} \left(\frac{2\sqrt{n}}{(1-\theta)}\right) \left(\frac{\lambda_{\max}(K)}{\lambda_{\min}(K)}\right) \|\widetilde{e}_{2}\|.$$
(6 - 11)

Note that the desired dynamics (6 - 4) comprise a term of the form:

$$k_{ai}\mu_i \operatorname{sat}\left(\frac{\hat{s}_i}{\mu_i}\right) = k_{ai}\mu_i \operatorname{sat}\left(\frac{s_i + \widetilde{e}_{2i}}{\mu_i}\right),$$

For this term to be unsaturated, it is sufficient for

$$\left|s_{i}\right|+\left|\widetilde{e}_{2i}\right|\leq\mu_{i},$$

and since

$$\left|s_{i}+\widetilde{e}_{2i}\right| \leq \left|s_{i}\right|+\left|\widetilde{e}_{2i}\right|,$$

while the ultimate bound on s is dependent on that of the observation error for a given K, recall from (6 - 10) and (6 - 11),

$$\begin{split} |\widetilde{e}_{2i}| &\leq \left(\frac{\rho_i\left(|\Delta_{2i}|, l_{3i}\right)}{\underline{\pi}_{2i}}\right)^{\frac{1}{4}}, \text{ where } \rho_i\left(|\Delta_{2i}|, l_{3i}\right) \coloneqq \begin{cases} \rho_{1i}\left(|\Delta_{2i}|, l_{3i}\right), \text{ for } 0.5$$

where the following property has been employed: $||x||_{\infty} \le ||x||_{2} \le \sqrt{n} ||x||_{\infty}, \forall x \in \mathbb{R}^{n}$, then the sufficient condition such that no saturation occur, is

$$\max_{i} \left\{ \left(\frac{\rho_{i} \left(|\Delta_{2i}|, l_{3i} \right)}{\underline{\pi}_{2i}} \right)^{\frac{1}{4}} \right\} \times \left(1 + \left(\frac{\overline{m}}{\underline{m}} \right)^{\frac{1}{2}} \left(\frac{2n}{(1-\theta)} \right) \left(\frac{\lambda_{\max}(K)}{\lambda_{\min}(K)} \right) \right) \le \mu_{i}, \qquad (6-12)$$

Thus, it is possible for the term $sat(s_i + \tilde{e}_{2i})$ to be unsaturated in finite time, provided a sufficiently big observer gain L_3 .

b. For a > 0:

Then we can rewrite the above inequality as:

$$\begin{split} \dot{V}_{s} \stackrel{a.e.}{\leq} &- s^{\mathrm{T}} K \mathrm{sat} \Big[\mathrm{sig} \big(\mu^{-1} s \big)^{a} \Big] + s^{\mathrm{T}} \Delta_{1} (\cdot) - s^{\mathrm{T}} K \Big(\mathrm{sat} \Big[\mathrm{sig} \big(\mu^{-1} \hat{s} \big)^{a} \Big] - \mathrm{sat} \Big[\mathrm{sig} \big(\mu^{-1} s \big)^{a} \Big] \Big) \\ &\leq - s^{\mathrm{T}} K \mathrm{sat} \Big[\mathrm{sig} \big(\mu^{-1} s \big)^{a} \Big] + s^{\mathrm{T}} \Delta_{1} (\cdot) - s^{\mathrm{T}} K \Big(\mathrm{sat} \Big[\mathrm{sig} \big(\mu^{-1} \big(s + \widetilde{e}_{2} \big) \big)^{a} \Big] - \mathrm{sat} \Big[\mathrm{sig} \big(\mu^{-1} s \big)^{a} \Big] \Big) \\ &\leq \sum_{i=1}^{n} - k_{i} \Big| s_{i} \Big| \mathrm{sat} \left[\frac{|s_{i}|^{a}}{\mu_{i}^{a}} \right] + \Big| s_{i} \Big| |\Delta_{1i} \Big| \\ &+ \Big| s_{i} \Big| k_{i} \Big| \mathrm{sat} \left[\frac{|s_{i} + \widetilde{e}_{2i}|^{a} \operatorname{sign} \big(s_{i} + \widetilde{e}_{2i} \big) \Big] - \mathrm{sat} \left[\frac{|s_{i}|^{a} \operatorname{sign} \big(s_{i} \big) \Big] \\ &\mu_{i}^{a} \Big] \end{split}$$

where

$$\Delta_1(\cdot) = (\Delta_{1i}(\cdot), \ldots, \Delta_{1n}(\cdot))^{\mathrm{T}}, \quad \forall i = 1, \ldots, n,$$

Note that from Appendix B.2-proposition 1, the following inequality is satisfied

$$\left|\operatorname{sat}\left[\frac{\left|s_{i}+\widetilde{e}_{2i}\right|^{a}\operatorname{sign}(s_{i}+\widetilde{e}_{2i})}{\mu_{i}^{a}}\right]-\operatorname{sat}\left[\frac{\left|s_{i}\right|^{a}\operatorname{sign}(s_{i})}{\mu_{i}^{a}}\right]\right|\leq\operatorname{sat}_{2}\left[\beta(\widetilde{e}_{2i}|)\right]$$

where β is a class \mathcal{K} function.

Now, let us define the set of real number $N \in \{1,...,n\}$ and $|\Delta_{1i}| := \sup \{ \varepsilon_i | : \varepsilon_i \in \mathbf{K}[\Delta_{1i}] \}, \forall i \in N$, then \dot{V}_s becomes

Note that the last equality has single out one of its element, $j \in N$ from the summation and rearranged to separate terms due to observation errors and uncertainties. To show sign definiteness, the two terms are considered separately:

<u>For *ω*_{s1}:</u>

$$\omega_{s1} = -\frac{k_j}{2} \left| s_j \right| \operatorname{sat} \left[\frac{\left| s_j \right|^a}{\mu_j^a} \right] + \left| s_j \right| \left| \Delta_{1j} \right| + \left(\sum_{i \in N \setminus j} \left| s_i \right| \left(-\frac{k_i}{2} \operatorname{sat} \left[\frac{\left| s_i \right|^a}{\mu_i^a} \right] + \left| \Delta_{1i} \right| \right) \right)$$

Note that it is necessary that $0.5k_i > |\Delta_i|, \forall i \in N$, then the maximum of the last term in the above equation occurs when $|s_i| < \mu_i$. In particular, by taking the derivative of the term w.r.t. $|s_i|$ and equating it to zero, the maximum is found to occur at

$$|s_i| = \left|\frac{2|\Delta_{1i}|}{(1+a)k_i}\right|^{\frac{1}{a}} \mu_i, \quad \forall i \in N \setminus j$$

and the corresponding maximum,

$$\left(\frac{a}{\left(1+a\right)^{\frac{1+a}{a}}}\right)\left(\frac{2|\Delta_{1i}|}{k_i}\right)^{\frac{1}{a}}\mu_i|\Delta_{1i}|, \ \forall i \in N \setminus j$$

Hence, one obtains,

$$\begin{split} \omega_{s1} &\leq -\frac{k_{j}}{2} \left| s_{j} \right| \operatorname{sat} \left[\frac{\left| s_{j} \right|^{a}}{\mu_{j}^{a}} \right] + \left| s_{j} \right| \left| \Delta_{1j} \right| + \left(\frac{a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in N \setminus j} \left(\frac{2|\Delta_{1i}|}{k_{i}} \right)^{\frac{1}{a}} \mu_{i} \left| \Delta_{1i} \right| \\ &= -\left(1 - \theta \right) \frac{k_{j}}{2} \left| s_{j} \right| \operatorname{sat} \left[\frac{\left| s_{j} \right|^{a}}{\mu_{j}^{a}} \right] - \frac{\theta}{4} k_{j} \left| s_{j} \right| \operatorname{sat} \left[\frac{\left| s_{j} \right|^{a}}{\mu_{j}^{a}} \right] + \left| s_{j} \right| \left| \Delta_{1j} \right| - \frac{\theta}{4} k_{j} \left| s_{j} \right| \operatorname{sat} \left[\frac{\left| s_{j} \right|^{a}}{\mu_{j}^{a}} \right] \\ &+ \left(\frac{a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in N \setminus j} \left(\frac{2|\Delta_{1i}|}{k_{i}} \right)^{\frac{1}{a}} \mu_{i} \left| \Delta_{1i} \right|, \\ &\leq -\left(1 - \theta \right) \frac{k_{j}}{2} \left| s_{j} \right| \operatorname{sat} \left[\frac{\left| s_{j} \right|^{a}}{\mu_{j}^{a}} \right], \text{ for } \left| s_{j} \right| \geq \pi_{1j}, \text{ and } \left| s_{i} \right| \in \mathbb{R}, \forall i \in N \setminus j \end{split}$$

with

$$\pi_{1j} = \max\left\{ \left(\frac{4}{\theta} \frac{\left|\Delta_{1j}\right|}{k_j}\right)^{\frac{1}{a}} \mu_j, \left(\frac{4\mu_j^a a}{\theta k_j (1+a)^{\frac{1+a}{a}}}\right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} \left(\frac{2|\Delta_{1i}|}{k_i}\right)^{\frac{1}{a}} \mu_i |\Delta_{1i}|\right)^{\frac{1}{1+a}}\right\},$$

and
$$k_j > \max\left\{\frac{4\left|\Delta_{1j}\right|}{\theta}, \frac{4}{\theta\mu_j}\left(\frac{a}{(1+a)^{\frac{1+a}{a}}}\right)\sum_{i\in N\setminus j}\left(\frac{2\left|\Delta_{1i}\right|}{k_i}\right)^{\frac{1}{a}}\mu_i\left|\Delta_{1i}\right|\right\},$$
 (6 - 14)

where $0 < \theta < 1$ constant. Note that with the above conditions, we have $\pi_{1j} < \mu_j$.

<u>For *ω*_{s2}:</u>

$$\begin{split} \omega_{s2} &= -\frac{k_j}{2} |s_j| \operatorname{sat} \left[\frac{|s_j|^a}{\mu_j^a} \right] + |s_j| k_j \operatorname{sat}_2 \left[\beta \left(\widetilde{e}_{2j} \right) \right] \\ &+ \left(\sum_{i \in N \setminus j} k_i |s_i| \left(-\frac{1}{2} \operatorname{sat} \left[\frac{|s_i|^a}{\mu_i^a} \right] + \operatorname{sat}_2 \left[\beta \left(|\widetilde{e}_{2j}| \right) \right] \right) \right), \end{split}$$

Note that it is necessary that

$$\operatorname{sat}_{2}\left[\beta\left(\left|\widetilde{e}_{2j}\right|\right)\right] \leq 0.5, \quad \forall i = 1, \dots, n$$

such that the maximum of the last term occurs when $|s_i| < \mu_i$.

This can be achieved by having sufficiently high observer gain, L_3 for a given control gain *K* (see step (1-c)). In particular, by taking the derivative of the term w.r.t. $|s_i|$ and equating it to zero, the maximum is found to occur at

$$|s_i| = \left(\frac{2\text{sat}_2[\beta(\widetilde{e}_{2j}|)]}{(1+a)}\right)^{\frac{1}{a}} \mu_i, \ \forall i \in N \setminus j$$

and the corresponding maximum,

$$\left(\frac{2^{\frac{1}{a}}a}{(1+a)^{\frac{1+a}{a}}}\right)k_{i}\mu_{i}\left(\operatorname{sat}_{2}\left[\beta\left(\widetilde{e}_{2j}\right)\right)\right)^{\frac{1+a}{a}}, \quad \forall i \in N \setminus j$$

Hence, one obtains,

$$\begin{split} \omega_{s2} &\leq -\frac{k_{j}}{2} |s_{j}| \operatorname{sat} \left[\frac{|s_{j}|^{a}}{\mu_{j}^{a}} \right] + |s_{j}| k_{j} \operatorname{sat}_{2} \left[\beta \left(\left| \widetilde{e}_{2j} \right| \right) \right] + \left(\frac{2^{\frac{1}{a}} a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in \mathbb{N} \setminus j} k_{i} \mu_{i} \left(\operatorname{sat}_{2} \left[\beta \left(\left| \widetilde{e}_{2j} \right| \right) \right] \right)^{\frac{1+a}{a}} \\ &= -(1-\theta) \frac{k_{j}}{2} |s_{j}| \operatorname{sat} \left[\frac{|s_{j}|^{a}}{\mu_{j}^{a}} \right] - \frac{\theta}{4} k_{j} |s_{j}| \operatorname{sat} \left[\frac{|s_{j}|^{a}}{\mu_{j}^{a}} \right] + |s_{j}| k_{j} \operatorname{sat}_{2} \left[\beta \left(\left| \widetilde{e}_{2j} \right| \right) \right] \right)^{\frac{1+a}{a}} \\ &- \frac{\theta}{4} k_{j} |s_{j}| \operatorname{sat} \left[\frac{|s_{j}|^{a}}{\mu_{j}^{a}} \right] + \left(\frac{2^{\frac{1}{a}} a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in \mathbb{N} \setminus j} k_{i} \mu_{i} \left(\operatorname{sat}_{2} \left[\beta \left(\left| \widetilde{e}_{2j} \right| \right) \right) \right)^{\frac{1+a}{a}} \\ &\leq -(1-\theta) \frac{k_{j}}{2} |s_{j}| \operatorname{sat} \left[\frac{|s_{j}|^{a}}{\mu_{j}^{a}} \right], \quad \text{for} |s_{j}| \geq \pi_{2j}, \text{ and } |s_{i}| \in \mathbb{R}, \forall i \in \mathbb{N} \setminus j, \end{split}$$

with

$$\pi_{2j} = \max\left\{ \begin{cases} \left(\frac{4}{\theta} \operatorname{sat}_{2}\left[\beta\left(\left|\widetilde{e}_{2j}\right|\right)\right]\right)^{\frac{1}{a}} \mu_{j}, \\ \left(\frac{4\mu_{j}^{a}}{\theta k_{j}} \left(\frac{2^{\frac{1}{a}}a}{(1+a)^{\frac{1+a}{a}}}\right) \sum_{i \in N \setminus j} k_{i} \mu_{i} \left(\operatorname{sat}_{2}\left[\beta\left(\left|\widetilde{e}_{2j}\right|\right)\right)^{\frac{1+a}{a}}\right)^{\frac{1}{1+a}} \right), \end{cases} \right\}$$

$$\operatorname{sat}_{2}\left[\beta\left(\left|\widetilde{e}_{2j}\right|\right)\right] \leq \min\left\{\frac{\theta}{4}, 0.5\right\}, \quad \forall i = 1, \dots, n,$$

$$\left\|\operatorname{sat}_{2}\left[\beta\left(\left|\widetilde{e}_{2}\right|\right)\right)\right\| \leq \left(\frac{\theta}{4(n-1)}\left(\frac{k_{j}}{\lambda_{\max}(K)}\right)\left(\frac{\mu_{j}}{\lambda_{\max}(\mu)}\right)\left(\frac{(1+a)^{\frac{1+a}{a}}}{2^{\frac{1}{a}}a}\right)\right)^{\frac{a}{1+a}}, \quad (6-15)$$

where $\operatorname{sat}_{2}\left[\beta\left(|\widetilde{e}_{2}|\right)\right] = \left(\operatorname{sat}_{2}\left[\beta\left(|\widetilde{e}_{21}|\right), \ldots, \operatorname{sat}_{2}\left[\beta\left(|\widetilde{e}_{2n}|\right)\right]\right)^{\mathrm{T}}\right)$, and $0 < \theta < 1$.

Note that the above conditions can be satisfied by having sufficiently high observer gain L_3 for a given control gain *K* (see step (1-c)). With the above conditions, we have $\pi_{2j} < \mu_j$.

Now, from (6 - 13) we have:

$$\dot{V}_{s} \stackrel{a.e.}{\leq} \omega_{s1} + \omega_{s2},$$

and by selecting each k_j , $\forall j \in N$ that satisfy (6 - 14), and having the observer gain, L_3 sufficiently high that satisfy (6 - 15), we have

$$\omega_{s1} \leq -c_{s1}, \forall s \notin \Omega_{\pi 1}$$
$$\omega_{s2} \leq -c_{s2}, \forall s \notin \Omega_{\pi 2}$$

where the compact sets are defined as: $\Omega_{\pi 1} = \{s \in \mathbb{R}^n : |s_i| \le \pi_{1i}, \forall i \in N\}$, and

$$\Omega_{\pi^2} = \left\{ s \in \mathbb{R}^n : \left| s_i \right| \le \pi_{2i}, \forall i \in N \right\},\$$

with positive constants,

$$c_{s1} = \min_{i} \left\{ (1-\theta) \frac{k_{i}}{2} \pi_{1i} \left(\frac{\pi_{1i}}{\mu_{i}} \right)^{a} \right\}, \text{ and } c_{s2} = \min_{i} \left\{ (1-\theta) \frac{k_{i}}{2} \pi_{2i} \left(\frac{\pi_{2i}}{\mu_{i}} \right)^{a} \right\}, \forall i \in \mathbb{N},$$

Define $\max_{j} \{ \pi_{1j}, \pi_{2j} \} = \pi_M$, $\forall j \in N$, then, we have:

$$\dot{V}_{s}^{a.e.} < 0, \ \forall s \notin \Omega_{\pi M}$$

where $\Omega_{\pi M} = \{ s \in \mathbb{R}^n : |s_i| \le \pi_M, \forall i \in N \}$, since $\Omega_{\pi M} \supset (\Omega_{\pi 1} \cup \Omega_{\pi 2})$.

Note that this compact set can be made arbitrarily small by increasing k_i up to the

maximal allowable control bound $T_{i,\max}$, to reduce π_{1i} and the increase of the observer gain, L_3 (for a given control gain *K*) to reduce π_{2i} . Now with each given μ_i , let c_1 be a positive constant $0 < c_1 < 1$, consider the following compact set

$$\Omega_{\mu} = \left\{ s \in \mathbb{R}^{n} : \left| s_{i} \right| \le c_{1} \mu_{i}, \forall i \in N \right\} \text{ and } \Sigma_{\mu} = \left\{ s \in \mathbb{R}^{n} : V_{s} \le k_{\mu} \right\},\$$

where $k_{\mu} > 0$ is defined as

$$k_{\mu} = \min_{s \in bd \ \Omega_{\mu}} V_s$$

which exists since the boundary, bd Ω_{μ} is a compact set. Note that $\Sigma_{\mu} \subset \Omega_{\mu}$. Note that if the states, *s* can be confined within Σ_{μ} , which is a Lyapunov level set, $|s_i| \leq c_1 \mu_i$. To achieve this, we simply need the set $\Omega_{\pi M} \subset \Sigma_{\mu}$, which can be attained when *K*, and *L*₃ are large enough, such that \dot{V}_s being negative the outside of the set Σ_{μ} ,

$$\dot{V}_s \stackrel{a.e.}{<} 0$$
, for $V_s \ge k_{\mu}$

which implies that the trajectories of *s* will enter the set Σ_{μ} in finite time and stay in it once entered. Hence, we have

$$|s_i| \le c_1 \mu_i < \mu_i \Leftrightarrow |s_i| < \mu_i, \ \forall i \in N$$

In particular, a sufficient condition on the control gain, k_i and observer gain, l_i such that the control is unsaturated can be obtained as follows,

Firstly note that:

$$k_{\mu} = \min_{s \in \mathrm{bd} \ \Omega_{\mu}} V_s \geq \min_{s \in \mathrm{bd} \ \Omega_{\mu}} \frac{1}{2} \underline{m} \|s\|^2 = \frac{1}{2} \underline{m} \left(\min_{i} c_1 \mu_i \right)^2$$

Next, observed that for a given K, the observer gain, L_3 can be chosen to be sufficiently high such that

 $\pi_{1j} > \pi_{2j}, \ \forall j \in N$, to that effect, recall that

$$\pi_{2j} \leq \left\| \left(\operatorname{sat}_{2} \left[\beta \left(\left| \widetilde{e}_{2} \right| \right) \right) \right\|^{\frac{1}{a}} \times \max \left\{ \left(\frac{4}{\theta} \right)^{\frac{1}{a}} \mu_{j}, \left(\frac{4 \mu_{j}^{a}}{\theta k_{j}} \left(\frac{2^{\frac{1}{a}} a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in N \setminus j} k_{i} \mu_{i} \right)^{\frac{1}{1+a}} \right\}$$

Thus, to satisfy $\pi_{1j} > \pi_{2j}$, $\forall j \in N$, we need

$$\max \begin{cases} \left(\frac{4}{\theta} \frac{|\Delta_{1j}|}{k_j}\right)^{\frac{1}{a}} \mu_j, \\ \left(\frac{4\mu_j^a a}{\theta k_j (1+a)^{\frac{1+a}{a}}}\right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} \left(\frac{2|\Delta_{1i}|}{k_i}\right)^{\frac{1}{a}} \mu_i |\Delta_{1i}|\right)^{\frac{1}{1+a}} \right) \\ > \left\| \left(\operatorname{sat}_2 \left[\beta(|\widetilde{e}_2|)\right] \right) \right\|^{\frac{1}{a}} \times \max \begin{cases} \left(\frac{4}{\theta}\right)^{\frac{1}{a}} \mu_j, \\ \left(\frac{4\mu_j^a}{\theta k_j} \left(\frac{2^{\frac{1}{a}} a}{(1+a)^{\frac{1+a}{a}}}\right) \sum_{i \in N \setminus j} k_i \mu_i\right)^{\frac{1}{1+a}} \right) \end{cases}$$

$$\Leftrightarrow \left\| \left(\operatorname{sat}_{2} \left[\beta\left(\left[\widetilde{e}_{2} \right] \right) \right) \right\|^{\frac{1}{a}} \right)^{\frac{1}{a}} \mu_{j}, \left(\frac{4\mu_{j}^{a}a}{\theta k_{j}(1+a)^{\frac{1+a}{a}}} \right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} \left(\frac{2|\Delta_{1i}|}{k_{i}} \right)^{\frac{1}{a}} \mu_{i} |\Delta_{1i}| \right)^{\frac{1}{1+a}} \right) \right\|$$

$$< \frac{1}{1+a} \left(\frac{4}{\theta} \int_{a}^{\frac{1}{a}} \mu_{j}, \left(\frac{4\mu_{j}^{a}}{\theta k_{j}} \left(\frac{2^{\frac{1}{a}}a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in N \setminus j} k_{i} \mu_{i} \right)^{\frac{1}{1+a}} \right) \right\|$$

$$(6 - 16)$$

Then, we have

$$\pi_{M} = \max_{j} \{ \pi_{1j}, \pi_{2j} \} = \max_{j} \{ \pi_{1j} \}, \ \forall j \in N$$

Hence,

$$\begin{split} \max_{s \in bd \ \Omega_{adv}} V_s &\leq \max_{s \in bd \ \Omega_{adv}} \frac{1}{2} \overline{m} \left\| s \right\|^2 \\ &= \frac{1}{2} \overline{m} \left(\max_{s \in bd \ \Omega_{adv}} \left\| s \right\| \right)^2 = \frac{1}{2} \overline{m} \left(\left(\sum_i^n \left| \pi_M \right|^2 \right)^{\frac{1}{2}} \right)^2 = \frac{n}{2} \overline{m} \left| \pi_M \right|^2 \\ &= \frac{n}{2} \overline{m} \left(\max_j^n \left\{ \frac{4}{\theta} \frac{\left| \Delta_{1j} \right|}{\theta} \right\}^{\frac{1}{\theta}} \mu_j, \\ &\left(\frac{4\mu_j^n a}{\theta k_j (1+a)^{\frac{1+a}{a}}} \right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} \left(\frac{2\left| \Delta_{1i} \right|}{k_i} \right)^{\frac{1}{a}} \mu_i \left| \Delta_{1i} \right| \right)^{\frac{1}{1+a}} \right)^n, \text{ for } j \in N, \\ &\leq \frac{n}{2} \overline{m} \left(\frac{1}{k} \right)^{\frac{2}{a}} \left(\max_j^n \left(\frac{4\left| \Delta_{1j} \right|}{\theta} \right)^{\frac{1}{a}} \mu_j, \\ &\left(\frac{4\mu_j^n a}{\theta (1+a)^{\frac{1+a}{a}}} \right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} \left(2\left| \Delta_{1i} \right| \right)^{\frac{1}{a}} \mu_i \left| \Delta_{1i} \right| \right)^{\frac{1}{1+a}} \right)^2 \end{split}$$

where $\underline{k} = \min_{i} k_{i}$

Hence, to satisfy the condition $\Omega_{\pi M} \subset \Sigma_{\mu}$, it is sufficient for:

$$k_{\mu} \geq \max_{s \in bd \ \Omega_{ad}} V_{s}$$

$$\Rightarrow \frac{1}{2} \underline{m} \left(\min_{i} c_{1} \mu_{i} \right)^{2} \geq \frac{n}{2} \overline{m} \left(\frac{1}{\underline{k}} \right)^{\frac{2}{a}} \left(\max_{j} \left\{ \left(\frac{4 |\Delta_{1j}|}{\theta} \right)^{\frac{1}{a}} \mu_{j}, \left(\frac{4 |\Delta_{1j}|}{\theta (1+a)^{\frac{1+a}{a}}} \right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} (2 |\Delta_{1i}|)^{\frac{1}{a}} \mu_{i} |\Delta_{1i}| \right)^{\frac{1}{1+a}} \right\} \right)^{2}$$

$$\Leftrightarrow \underline{k} \geq \frac{\left(\max_{j} \left\{ \left(\frac{4 |\Delta_{1j}|}{\theta} \right)^{\frac{1}{a}} \mu_{j}, \left(\frac{4 \mu_{j}^{\ a} a}{\theta (1+a)^{\frac{1+a}{a}}} \right)^{\frac{1}{1+a}} \left(\sum_{i \in N \setminus j} (2 |\Delta_{1i}|)^{\frac{1}{a}} \mu_{i} |\Delta_{1i}| \right)^{\frac{1}{1+a}} \right)^{a}}{\left(\frac{m}{n \overline{m}} \right)^{\frac{2}{a}} \left(\min_{i} c_{1} \mu_{i} \right)^{a}}$$

$$(6 - 17)$$

It is not difficult to show that selecting *K* that satisfies (6 - 17) implies that (6 - 14) is sufficiently satisfied as well. Thus, for the case of a > 0 the control will be unsaturated in finite time and remain so thereafter.

Note that for any $0 \le c_1 \le 1$, there always exist a positive constant c_2 such that $0 \le c_2 \le (1 - c_1)$.

Thus, with $|s_i| \le c_1 \mu_i$, $\forall i \in N$, a sufficient condition for the control to be unsaturated is, for, $|\tilde{e}_{2i}| \le c_2 \mu_i$ such that,

$$\begin{aligned} |\hat{s}_i| &= |s_i + \widetilde{e}_{2i}| \le |s_i| + |\widetilde{e}_{2i}| \\ &\le c_1 \,\mu_i + c_2 \mu_i \\ &< \mu_i, \text{ since } c_1 + c_2 < 1, \end{aligned}$$

Together with the above conditions, it is sufficient for the observer gains to satisfy the following (for a given control gain *K*) by combining (6 - 15), (6 - 16) and $|\tilde{e}_{2i}| \le c_2 \mu_i$:

$$\max \left\| \widetilde{e}_{2i} \right\| \left\| \operatorname{sat}_{2} \left[\beta \left(\left| \widetilde{e}_{2} \right| \right) \right] \right\|$$

$$< \min \left\{ \begin{cases} c_{2} \mu_{i}, \frac{\theta}{4}, 0.5, \left(\frac{\theta}{4(n-1)} \left(\frac{k_{j}}{\lambda_{\max}(K)} \right) \left(\frac{\mu_{j}}{\lambda_{\max}(\mu)} \right) \left(\frac{(1+a)^{\frac{1+a}{a}}}{2^{\frac{1}{a}} a} \right) \right)^{\frac{a}{1+a}}, \\ \left(\frac{4}{\theta} \left| \frac{4}{\theta} \right|_{j} \right)^{\frac{1}{a}} \mu_{j}, \\ \max \left\{ \left(\frac{4\mu_{j}^{a} a}{\theta k_{j} (1+a)^{\frac{1+a}{a}}} \right)^{\frac{1}{1+a}} \left(\sum_{i \in \mathbb{N} \setminus j} \left(\frac{2|\Delta_{1i}|}{k_{i}} \right)^{\frac{1}{a}} \mu_{i} |\Delta_{1i}| \right)^{\frac{1}{1+a}} \right) \\ \max \left\{ \left(\frac{4}{\theta} \right)^{\frac{1}{a}} \mu_{j}, \left(\frac{4\mu_{j}^{a}}{\theta k_{j}} \left(\frac{2^{\frac{1}{a}} a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in \mathbb{N} \setminus j} k_{i} \mu_{i} \right)^{\frac{1}{1+a}} \right\} \\ \left(\frac{1}{\theta} \right)^{\frac{1}{a}} \mu_{j}, \left(\frac{4\mu_{j}^{a}}{\theta k_{j}} \left(\frac{2^{\frac{1}{a}} a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in \mathbb{N} \setminus j} k_{i} \mu_{i} \right)^{\frac{1}{1+a}} \right\} \\ \left(\frac{1}{\theta} \right)^{\frac{1}{a}} \mu_{i}, \left(\frac{4\mu_{j}^{a}}{\theta k_{j}} \left(\frac{2^{\frac{1}{a}} a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in \mathbb{N} \setminus j} k_{i} \mu_{i} \right)^{\frac{1}{1+a}} \right\} \\ \left(\frac{1}{\theta} \right)^{\frac{1}{a}} \mu_{i}, \left(\frac{4\mu_{j}^{a}}{\theta k_{j}} \left(\frac{2^{\frac{1}{a}} a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in \mathbb{N} \setminus j} k_{i} \mu_{i} \right)^{\frac{1}{a}} \right)^{\frac{1}{a}} \right\} \\ \left(\frac{1}{\theta} \right)^{\frac{1}{a}} \mu_{i}, \left(\frac{4\mu_{j}^{a}}{\theta k_{j}} \left(\frac{2^{\frac{1}{a}} a}{(1+a)^{\frac{1+a}{a}}} \right) \sum_{i \in \mathbb{N} \setminus j} k_{i} \mu_{i} \mu_{i} \right)^{\frac{1}{a}} \right)^{\frac{1}{a}} \right\}$$

and the control will be unsaturated in finite time and remain so thereafter.

3. Having shown the boundedness of $(\tilde{e}_{1i}, \tilde{e}_{2i})$, and \hat{s} being unsaturated, we are going to show the boundedness property of the desired error dynamics (6 - 4).

From step 2, we have $|s_i| < c_1 \mu_i$, $|\tilde{e}_{2i}| \le c_2 \mu_i$, $\forall i \in N$, for a > 0, and

 $|s_i| + |\tilde{e}_{2i}| \le \mu_i$, $\forall i \in N$, for a = 0, then, in this region, the following is equivalent:

$$k_{ai}\mu_{i}\operatorname{sat}\left[\frac{\left(s_{i}+\widetilde{e}_{2i}\right)}{\mu_{i}}\right] = k_{ai}\left(s_{i}+\widetilde{e}_{2i}\right) = k_{ai}s_{i} + k_{ai}\operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i}\right], \text{ and}$$
$$v_{2i}+s_{i}+\widetilde{e}_{2i}=v_{2i}+s_{i}+\operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i}\right], \text{ since } |\widetilde{e}_{2i}| < \mu_{i},$$

For the case of b > 0:

Then, the desired error dynamics can be written element-wise as (6 - 6), $\forall i = 1,...,n$,

$$\dot{v}_{1i} = v_{2i} + s_{i}$$

$$\dot{v}_{2i} = -k_{ai} \left(v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} \left[\widetilde{e}_{2i} \right] \right) + k_{ai} \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{1+b}} \left[v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} \left[\widetilde{e}_{2i} \right] \right]$$

$$- \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} \left[\widetilde{e}_{2i} \right] \right]^{\frac{2b}{1+b}} \operatorname{sign} \left(v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} \left[\widetilde{e}_{2i} \right] \right) \right]$$

$$- \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} | v_{1i} |^{b} \operatorname{sign} \left(v_{1i} \right) \right]$$
(6 - 19)

From Appendix B.1, with the Lyapunov function,

$$V = \left(\frac{1}{2}v_{2i}^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |r|^{b} \operatorname{sign}(r)\right] dr\right)^{2} + r_{1} \left(\frac{3+3b}{2}\right) \left(\int_{0}^{v_{1i}} |r|^{\frac{1}{2}} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \left[|r|^{\frac{3b}{2}}\right] dr\right) v_{2i}$$
$$+ r_{1} (1+b) k_{ai} \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}} \left(\int_{0}^{v_{1i}} |r|^{\frac{3}{2}} \operatorname{sign}(r) - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}b}} \left[|r|^{\frac{3}{2}} \operatorname{sign}(r)\right] dr\right)$$

where $r_1 > 0$ is a constant, it is shown that

 $\dot{\widetilde{V}} < 0$, for $V \ge \rho_{1i} (|\zeta_i|)$,

where $|\zeta_i|$ is defined as $|\zeta_i| = \max\{|s_i|, \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}|]\}$, and the class \mathcal{K} function, ρ_{1i} is

defined as, $\rho_{1i}(|\zeta_i|) = \max_{(v_{1i}, v_{2i}) \in bd \ \Psi_3(|\zeta_i|)} V$

with bd $\Psi_{3i}(\zeta_i)$ as the boundary of the compact set

$$\Psi_{3i}(|\zeta_i|) = \{ (v_{1i}, v_{2i}) \in \mathbb{R}^2 : |v_{1i}| \le \alpha_1(|\zeta_i|), |v_{2i}| \le \alpha_2(|\zeta_i|) \}, \text{ and}$$

 α_1 , α_2 are class \mathcal{K} functions defined in Appendix B.1. Hence, invoking lemma 2.6,

the states (v_1, v_2) are uniformly ultimately bounded.

For the special case of b = 0:

Let the elements of ε_1 equal to that of K_1 , and ε_2 to that of K_2 , i.e. $\varepsilon_{1n} = k_{1n}, \forall i \in N$ and $\varepsilon_{2n} = k_{2n}, \forall i \in N$, hence the desired dynamics becomes

$$\dot{v}_{1i} = v_{2i} + s_i$$

$$\dot{v}_{2i} = -k_{ai} \left(v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\tilde{e}_{2i}] \right) + k_{ai} \operatorname{sat} \left[v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\tilde{e}_{2i}] \right]$$

$$-k_{2i} \operatorname{sign} \left(v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\tilde{e}_{2i}] \right) - k_{1i} \operatorname{sign} (v_{1i})$$

(6 - 20)

From Appendix B.1, with the Lyapunov function

$$V = \left(\frac{1}{2}{v_{2i}}^{2} + k_{1i}\int_{0}^{v_{1i}}\operatorname{sign}(r) dr\right)^{2} + r_{1}|v_{1i}|^{\frac{3}{2}}\operatorname{sign}(v_{1i}) v_{2i} + r_{1}\left(\frac{2}{5}\right)k_{ai}|v_{1i}|^{\frac{5}{2}} - r_{1}k_{ai}\int_{0}^{v_{1i}} \left[\operatorname{sat}_{1}\left[\left|r\right|^{\frac{3}{2}}\right]\right]\operatorname{sign}(r) dr$$

where $r_1 > 0$ is a constant, it is shown that

 $\dot{\widetilde{V}} < 0$, for $V \ge \rho_{2i} (|\zeta_i|)$,

where $|\zeta_i|$ is defined as $|\zeta_i| = \max\{s_i|, \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}|]\}$, and the class \mathcal{K} function, ρ_{2i} is

defined as $\rho_{2i}(|\zeta_i|) = \max_{(v_{1i}, v_{2i}) \in \mathrm{bd} \ \phi_{3i}(|\zeta_i|)} V$, with $\mathrm{bd} \ \phi_{3i}(|\zeta_i|)$ as the boundary of the compact set $\phi_{3i}(|s_i|) = \{(v_{1i}, v_{2i}) \in \mathbb{R}^2 : |v_{1i}| \le \alpha_{11}(|s_i|), |v_{2i}| \le \alpha_{12}(|s_i|)\}$

and α_{11} , α_{12} are class K functions defined in Appendix B.1. Hence, invoking lemma

2.6, the states (v_1, v_2) are uniformly ultimately bounded. Hence, invoking lemma 2.6, the states (v_1, v_2) are uniformly ultimately bounded.

Remark 6.5. Hence, for *K* satisfying (6 - 17), and observer gain L_3 sufficiently high such that (6 - 18) is satisfied through (6 - 10), the trajectories of the closed-loop system (6 - 6), are globally stable, with the ultimate bound being reduced as desired up to the saturation limit of the actuator, $T_{i, \text{max}} > 0$, $\forall i = 1,...,n$, and correspondingly that of the observer L_3 . Another way to view the stability results is for a given bounds on the actuator limit, and hence *K*, if the bounds on the uncertainties,

$$\Delta_1(\cdot) = -G(q) - M(q)\ddot{q}_d + M(q)\dot{\sigma} - \mathbf{K}[D] - F(\dot{q}_d - \sigma) - C(q, (\dot{q}_d - \sigma))(\dot{q}_d - \sigma)$$

are sufficiently small to satisfy (6 - 17), and observer gain L_3 sufficiently large for (6 - 18) the global practical stability results are still assured. Essentially, besides the bounds on the parameters of the robot manipulator dynamics and disturbances, the term Δ_1 is also dependent on the desired acceleration, velocity and the desired error dynamics through the term σ and σ . In which case, a slower desired trajectory or slower desired error dynamics can in effect produce a smaller upper bound of $|\Delta_1|$. *Remark* 6.6. It is desired for the controller (6 - 3) to behave as per its non-bounded counterpart (Chapter 4) when the controller is not saturated so that it exhibits the same properties such as the ability to inject desired error dynamics, and desired performance. Indeed from stability proof above (step 2), the control (6 - 3) will be unsaturated in finite time and stay so in all future times. Particularly, if the upper bound of $|\zeta_i|$ is sufficiently small (by selecting gain *K* satisfying (6 - 17) up to the allowable control bound, $T_{i, \max}$ and observer gain L_3 satisfying (6 - 18) through (6 - 10)), from step 3 above, the states (v_1, v_2) of the dynamics (6 - 19) and (6 - 20), will become unsaturated as well (see Appendix B.1),

for b > 0,

$$\dot{v}_{1i} = v_{2i} + s_i$$

$$\dot{v}_{2i} = -k_{1i} |v_{1i}|^b \operatorname{sign}(v_{1i}) - k_{2i} |v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}])$$

for b = 0,

$$v_{1i} = v_{2i} + s_i
\dot{v}_{2i} = -k_{2i} \operatorname{sign}(v_{2i} + s_i + \operatorname{sat}_{\mu_i}[\tilde{e}_{2i}]) - k_{1i} \operatorname{sign}(v_{1i})$$

which is similar to the *z*-subsystem (4 - 7) considered in Chapter 4, and hence will exhibit the same properties, since the states will remain unsaturated for all future times, provided $|\zeta_i|$ is sufficiently small.

Remark 6.7. Since the same observer structure of that in Chapter 4 is used here, for the special

case of p = 0.5, if the gains of the observer, L_1 and L_2 are sufficiently large to satisfy conditions of Theorem 4.2, the observation error $(\tilde{e}_{l_1}, \tilde{e}_{2_1})$ will converge to zero in finite time and stay there for all future times (refer to step 1-c of proof of Theorem 6.1). In that case, the closed-loop system (6 - 6) will be identical to the bounded full-state feedback control of (5 -4) considered in the previous chapter, Chapter 5.

Remark 6.8. Since the control (6 - 3) follows a similar bounded approach as Chapter 5, it has some similar inherent properties as that of (5 - 2) such as the use of integral of saturation functions that give rise to the boundedness of σ (see step 1-a of the proof of Theorem 6.1), the need to satisfy inequality $\underline{f} > C_m \|(\dot{q}_d - \sigma)\|$ (see step 1-b of the proof of Theorem 6.1) by having sufficiently slow desired motions, and the anti-windup structure (recall from (6 - 4), by adding and subtracting $K_a \hat{s}$),

$$\dot{\sigma} = -K_a \sigma + K_a \hat{s} - \underbrace{K_a \left(\hat{s} - \mu \operatorname{sat}\left[\mu^{-1} \hat{s}\right]\right)}_{anti-windup} - K_a \operatorname{sat}_{\left(\frac{\varepsilon_2}{k_2}\right)^{\frac{1+b}{2b}}} \left[\hat{e}_2\right] + \operatorname{sat}_{\varepsilon_2} \left[K_2 \operatorname{sig}(\hat{e}_2)^{\frac{2b}{1+b}}\right] + \operatorname{sat}_{\varepsilon_1} \left[K_1 \operatorname{sig}(e_1)^b\right].$$

6.3 Numerical Simulations

In this section, numerical simulations on a two-link robot manipulator were carried out to illustrate the results discussed in this chapter. The setups for each simulation are described. Discussion and analysis of the results are presented accordingly.

6.3.1 Simulation Setups

1) Simulation 1:

The same two-link rigid robot manipulator considered in section 3.3 is adopted in simulation. The dynamics of robot manipulator (6 - 2) have the same parameter values as that in section 3.3. The desired trajectory vector and the additive disturbances Coulomb friction vector were defined similarly as well. The initial conditions of the robot manipulator were selected as,

$$q(0) = \begin{bmatrix} 0 \\ 0 \end{bmatrix}, \dot{q}(0) = \begin{bmatrix} 10 \\ 16 \end{bmatrix}, q_d(0) = \begin{bmatrix} \pi \\ 0.5 \end{bmatrix}, \dot{q}_d(0) = \begin{bmatrix} 0.5 \\ 0 \end{bmatrix}$$

The control (6 - 3) parameter values were selected as follows,

$$a = 0.9, \ K = \begin{bmatrix} 150 & 0 \\ 0 & 15 \end{bmatrix}, \ K_1 = \begin{bmatrix} 18 & 0 \\ 0 & 18 \end{bmatrix}, \ K_2 = \begin{bmatrix} 11 & 0 \\ 0 & 11 \end{bmatrix}, \ b = 0.6$$
$$\mu = \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix}, \ \varepsilon_{11} = \varepsilon_{12} = 18, \ \varepsilon_{21} = \varepsilon_{22} = 11, \ K_a = \begin{bmatrix} 10 & 0 \\ 0 & 10 \end{bmatrix},$$

The initial conditions for the vector σ were selected as

$$\sigma(0) = \begin{bmatrix} 0 \\ 0 \end{bmatrix}$$

The observer (6 - 5) parameter values were selected as,

$$L_1 = L_2 = \begin{bmatrix} 2 & 0 \\ 0 & 2 \end{bmatrix}, L_3 = \begin{bmatrix} 150 & 0 \\ 0 & 150 \end{bmatrix}, p = 0.8,$$

with the following initial conditions,

$$\hat{e}_1(0) = e_1(0), w(0) = -L_3 e_1(0)$$

since position measurement is available.

2) Simulation 2:

The setup of Simulation 2 is exactly the same as that of Simulation 1. The simulation is repeated using the unbounded output-state feedback control (4 - 4), to examine its maximal tracking errors compared to that of control (6 - 3), with the parameter values selected as

$$a = 0.9, K = \begin{bmatrix} 150 & 0 \\ 0 & 15 \end{bmatrix}, K_1 = \begin{bmatrix} 18 & 0 \\ 0 & 18 \end{bmatrix}, K_2 = \begin{bmatrix} 11 & 0 \\ 0 & 11 \end{bmatrix}, b = 0.6$$

Note that when the control (6 - 3) vectors of Simulation 1 are unsaturated, it has the same parameter values as those employed in Simulation 2.

6.3.2 Results and Discussions

For better visualization of the plots, some figures are shown in two windows; each with different time intervals.









(b) Position observation error, \tilde{e}_{12} for time t = [0, 15] s.



(c) Velocity observation error, \tilde{e}_{21} for time t = (d) Velocity observation error, \tilde{e}_{22} for time t = [0, 15] s. [0, 15] s.

Figure 6.2 Simulation 1. Observation errors for joint 1 and joint 2.



(a) Tracking error, e_{11} for time t = [0, 15] s.







(b) Tracking error, e_{11} for time t = [5, 15] s.



(d) Tracking error, e_{12} for time t = [5, 15] s.

Figure 6.3 Simulation 2. Tracking errors using control (4 - 4), control (6 - 3).

1) Simulation 1:

As can be observed Figure 6.1(e)-(f), the large initial conditions caused the control (6 - 3) to saturate. From Theorem 6.1, the state σ will converge to a bounded region in finite time, due

to the saturated nature of the integral (see (6 - 4)), which is clearly shown in Figure 6.1(c)-(d). According to the stability proof, once the state σ converge to a bounded region and stay in it for all future times, with control gain *K* satisfying (6 - 14), the control will be unsaturated in finite time and stay in it for all future times, provided the observation errors \tilde{e}_2 are sufficiently small, which in turn are governed by the observer linear damping gain L_3 . In fact, from Figure 6.1(e)-(f) the control inputs for both joints of the robot manipulator remained unsaturated after the initial saturation. The performance of the observer, as shown in Figure 6.2, is very good in terms of estimation speed and residual error. This is due to the synergistic combination of the super-twisting based algorithm and a linear damping term.

2) Simulation 2:

Figure 6.3 shows that both controllers (4 - 4) and (6 - 3) have similar ultimate bound on the tracking errors, which is expected since the unsaturated (6 - 3) is identical to (4 - 4). Similar to the argument of the bounded control of Chapter 5, the vital point of the control (6 - 3) is for it to become unsaturated and remain so for all future times. The controller (6 - 3) is structurally identical to that of Chapter 5 (5 - 2), hence the design of the saturated integral (6 - 4) has the same properties as that in Chapter 5. The only difference is that the velocity measurement is considered not available in this chapter, hence the use of the observer introduced in Chapter 4. From the stability proof of Theorem 6.1, the stability of this control (6 - 3) boils down to the boundedness of the velocity observation errors, \tilde{e}_2 ; i.e. if \tilde{e}_2 is sufficiently small, the bounded output-feedback control will be unsaturated and remain so for all future times.

Hence, by combining the observer of Chapter 4 together with the integral of saturated functions designed in Chapter 5, a bounded output-feedback control (6 - 3) that inherits the properties of it unbounded and full-state feedback controller (3 - 2) is shown.

6.4 Summary

In this chapter, globally stabilizing saturated controller for the trajectory tracking of robot manipulators without velocity measurements with additive bounded disturbances were proposed. The velocity is being observed through a super-twisting based plus linear damping observer. Strict Lyapunov functions developed in the previous chapter 2 are being modified to accommodate the bounded nature of the desired error dynamics. Essentially, after the controller forces the states to have unsaturated control, similar properties of their unbounded counterpart can be obtained.

Chapter 7: CONCLUSION AND FUTURE WORK

7.1 Dissertation Summary

The focus of this research is to develop strict Lyapunov functions for algorithms that are based on twisting and super-twisting algorithms and applying them to the control of robot manipulators with real world problem considerations including output feedback without velocity measurements and actuator saturation. Due to the non-Lipschitz nature of the algorithms and the type of disturbances are allowed to exhibit discontinuity, nonsmooth Lyapunov theorem is employed throughout the work to obtain important results from the proposed strict Lyapunov functions such as settling time estimate for the finite-time convergence and robustness to non-Lipschitz disturbances. Real world systems are always affected by nonlinear behaviours that are often not considered; i.e. such as saturation in control channels, lack of velocity measurements, Coulomb friction, the work in this dissertation aims to compensate for these phenomena with practical control designs that can be implemented easily.

Family of algorithms based on second order sliding mode algorithm, namely that of twisting and super-twisting, are introduced in Chapter 2. Due to the non-Lipschitzness of the system, using the generalized solutions in the sense of Filippov, strict Lyapunov functions proposed are analysed using the generalized Lyapunov theorem. In particular, the strict Lyapunov function can fully characterize different stability properties of a parameterized family of controllers. Effectively, the family of controllers generalized the proportional-derivative (PD) control and twisting algorithm, and that of proportional-integral (PI) control and supertwisting algorithm. At the same time the strict Lyapunov functions proposed can similarly generalized the type of stability of these systems, from finite-time convergence to exponential convergence to uniform asymptotic convergence. Leveraging on this results, algorithms that combines the family of controllers with different degree of homogeneity are develop, and the corresponding strict Lyapunov functions are similarly develop by combining the one from the individual family, and the stability properties of each algorithm remains in the new combination. In essence, it is possible to achieve finite-time with uniform convergence, which in effect produces finite time convergence with fixed- settling time that is independent of initial conditions of the system.

Chapter 3 focuses on the trajectory tracking control of robot manipulator. Semiglobal practical stability is assured where the ultimate bound of the states can be made arbitrarily small and the region of attraction arbitrarily large by tuning a single parameter. Due to the generalized Lyapunov theorem and the proposed Lyapunov function, the stability analysis permits the disturbances to have discontinuity, such as Coulomb friction. Of interest is the ability of the proposed controller in generalizing the well-known PID control. In particular, for this special case, the PID gains selection is transform into the selection of desired error dynamics and the selection of acceptable precision of error. This gain tuning simplification to gains selection is important, as it allows the nonlinear robust control of desired error dynamics injection and disturbance compensation into an existing PID control, which is of great benefit since PID control is widely used in industrial robot manipulator. In addition, for the special case of position regulation problem, sufficient conditions on the gains are obtained to ensure either finite-time or exponential convergence of the system towards the regulation

point. In addition, due to integral nature of the controller, it is possible for the system to behave as per the desired error dynamics from the onset of control even in the presence of disturbances.

Trajectory tracking control of robot manipulator without velocity measurement is tackle in Chapter 4. The velocity measurement is substituted with the output of an observer that combines the super-twisting based algorithm develop in Chapter 2 with a damping term that is termed in the literature as "dirty-derivative". The addition of the linear damping term has the benefit of reducing the gain required of the non-Lipschitz part of the observer that is responsible for finite-time convergence. At the same time it allows the definition of the region of attraction to grow with the linear damping term, in which case is not possible if the term is not added and peaking phenomenon will occur if the initial observation error is not small enough. With the addition of this observer structure, the controller can maintain its useful feature from Chapter 3, where effectively, the controller here is an output feedback version of that proposed in Chapter 3.

The problem of saturated control is developed in Chapter 5 for the trajectory tracking or robot manipulators under the influence of bounded disturbances. By taking into account of the viscous friction and the proposed nonlinear integrator that injects a bounded desired error dynamics, global practical stability is achieved. Also, when the saturation level is sufficiently high for the user-defined speed of desired trajectory, the finite-time and exponential regulation of the unbounded control in Chapter 3 is recovered but in a global manner instead of semiglobal.

In the final chapter, the control robot manipulator is assumed to have both constraints; no
velocity measurements available and the control is bounded. For the velocity observation, the observer proposed in Chapter 4 is employed while the framework of bounded control in Chapter 5 is applied. As a result, a globally stabilizing saturated controller for the trajectory tracking of robot manipulators without velocity measurements under the influence of additive bounded disturbances is developed. Essentially, after the controller forces the states to have unsaturated control, similar properties of its unbounded counterpart in Chapter 4, can be obtained.

7.2 Limitations and Future Work

The work in this dissertation complements the SOSM algorithms and control designs for the trajectory tracking of robot manipulator. At the same time, it reveals new information on existing nonlinear systems. Hence in this section, open problems related to the research in this dissertation are presented.

In Chapter 2, the strict Lyapunov proposed are for the algorithms with constant gains. In particular, the mechanical energy of the system is utilized as part of the strict Lyapunov function. However, in doing so, if the gains of the system is allowed to be time varying, the construction of Lyapunov function is not as straightforward. An example of a strict Lyapunov function for super-twisting algorithm can be found in [146] where the structure of the Lyapunov functions dictate the type and form of the variable gains, which is restrictive. Hence, it is of interest if strict Lyapunov function can be developed that allows a full range of variable gains. As the task of finding strict Lyapunov function is not straightforward especially for higher order sliding mode algorithms, it is hoped that the work in this

dissertations help sheds light on through employing the inherent structure of the algorithms into the Lyapunov function.

For chapter 3, the stability result achieved is semiglobal which implies that the gains of the control are selected based on the initial conditions of the closed-loop system. If the gains are allowed to vary in accordance to the initial conditions by means of an adaptive algorithm, for instance, global stability may be attained. Particularly, given the strictness of the Lyapunov function proposed, utilizing certainty equivalence techniques similar to [147], an adaptive version of the controller may be possibly developed.

An output feedback version of the controller in Chapter 3 is presented in Chapter 4, by utilizing a super-twisting based algorithm plus a linear damping term observer. The stability analysis here is also dependent on satisfying sufficient conditions pertaining to the initial conditions of the closed-loop system. Hence, similar arguments as in Chapter 3 applies here, in which case, an adaptive version of the output feedback controller seems plausible through the strict Lyapunov functions of both the observer and controller produced here.

From Chapter 5, global results are achieved for the bounded controller by taking into account of the damping effect of the viscous friction and a bounded desired error dynamics in counteracting the effect of Coriolis and centrifugal terms of the dynamics. However, as a result, the type of desired trajectory, in particular the speed and acceleration of the desired trajectory is affected in order to satisfy the sufficient conditions for stability. Although, it is understood that, given a limit on the actuation, there is a bound on the speed at which the system can operate, it is useful if the limit on the desired trajectory be variable instead of a constant bound, depending on the configuration of the robot manipulator, so that the

bandwidth of the control can be utilised optimally.

Chapter 6 combines the results of the observer based control in Chapter 4 and the bounded control framework of Chapter 5 to produce a global practical stability result on trajectory tracking with the real world considerations of actuator limits and the lack of velocity measurements from the system. All controllers proposed from Chapter 3 to Chapter 6 for robot manipulator can achieved finite time convergence for the regulation problems by satisfying sufficient conditions, including the need for the gravity vector, G(q) to be zero at the point of regulation. However, such a need is not necessary for the exponential convergence for the regulation problem. This is mainly due to the non-Lipschitzness of the control, for finite-time convergence, occur at the origin, when a change of variable is applied to the desired gravity point, the new equilibrium point is locally Lipschitz, hence the control cannot render the convergence in finite time unless the desired gravity exactly compensated or zero.

Appendix A

A.1 **PROOF OF DESIRED DYNAMICS (CH3 & 4)**

Consider the following dynamics:

$$\dot{z}_1 = z_2 + d_1,$$

$$\dot{z}_2 = -k_1 \operatorname{sig}(z_1)^b - k_2 \operatorname{sig}(z_2)^{\frac{2b}{1+b}} + \delta$$
(A1 - 1)

where $z_1, z_2 \in \mathbb{R}$, are the scalar state variables, k_1, k_2 are positive constants, $b \ge 0$ real number with $\delta = k_2 |z_2|^{\frac{2b}{1+b}} \operatorname{sign}(z_2) - k_2 |z_2 + d_2|^{\frac{2b}{1+b}} \operatorname{sign}(z_2 + d_2)$ and $d_1, d_2 \in \mathbb{R}$ are bounded disturbances:

 $|d_1| \coloneqq \sup\{|\varepsilon_1| : \varepsilon_1 \in K[d_1]\}, |d_2| \coloneqq \sup\{|\varepsilon_2| : \varepsilon_2 \in K[d_2]\} \text{ and } |\delta| \coloneqq \sup\{|\varepsilon_3| : \varepsilon_3 \in K[\delta]\}$ (Note that no continuity assumption is made so that discontinuous disturbances may be used in *d*).

The results in this section are applicable to the desired dynamics section of both chapter 3 & 4. In particular, by applying the following change of variable:

For chapter 3:

$$z_1 = z_{1i}, z_2 = z_{2i}, d_1 = d_2 = s_i,$$

For chapter 4:

$$z_1 = z_{1i}, z_2 = z_{2i}, d_1 = d_2 = s_i,$$

the same differential equations are obtained.

Next, we take the Lyapunov function candidate of the form

$$V(z_1, z_2) = \frac{k_1^2}{(1+b)^2} |z_1|^{2+2b} + \frac{1}{4} |z_2|^4 + r|z_1|^{\frac{3+3b}{2}} \operatorname{sign}(z_1) z_2 + \frac{k_1}{(1+b)} |z_1|^{1+b} |z_2|^2$$

where r > 0 is a constant. Note that V is positive definite (with $r_i > 0$ exist, see Chapter 1).

$$\underline{\pi}_{1}\left(\left|z_{1}\right|^{2+2b}\left|z_{2}\right|^{4}\right) \leq V(z_{1}, z_{2}) \leq \overline{\pi}_{1}\left(\left|z_{1}\right|^{2+2b} + \left|z_{2}\right|^{4}\right)$$

where
$$\underline{\pi}_1 := \min\left\{\frac{1}{8}, \frac{1}{2}\left(\frac{k_1}{1+b}\right)^2\right\}, \ \overline{\pi}_1 := \max\left\{\frac{3r}{4} + \left(\frac{k_1}{1+b}\right)^2 + \left(\frac{k_1}{2+2b}\right), \ \left(\frac{k_1}{2+2b}\right) + \frac{r}{4} + \frac{1}{4}\right\},$$

In accordance to lemma 2.4, taking the time derivative of the Lyapunov function along the solutions of the system (A1 - 1) exists almost everywhere:

$$\frac{d}{dt}V(\mathbf{z}(t),t) \stackrel{a.e.}{\in} \overset{\cdot}{\widetilde{V}}(\mathbf{z}(t),t) = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^{T} K\begin{bmatrix} \dot{z}_{1}\\ \dot{z}_{2} \end{bmatrix} (\mathbf{z},t)$$

For ease of presentation, the analysis is separated for the case of b > 0 and b = 0.

For the case of b > 0:

Note that for b > 0, *V* is continuously differentiable:

$$\dot{\widetilde{V}}(z_1, z_2) = \bigcap_{\xi \in \partial V(\mathbf{z}(t), t)} \xi^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix} = \nabla V^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix}$$

$$= \left(-\frac{2k_{1}k_{2}}{(1+b)} |z_{1}|^{1+b} |z_{2}|^{\frac{1+3b}{1+b}} - k_{2}|z_{2}|^{\frac{3+5b}{1+b}} - rk_{1}|z_{1}|^{\frac{3+5b}{2}} + r\left(\frac{3+3b}{2}\right) |z_{1}|^{\frac{1+3b}{2}} |z_{2}|^{2} \right)$$

$$+ K[d_{1}\left[\frac{2k_{1}^{2}}{(1+b)} |z_{1}|^{1+2b} \operatorname{sign}(z_{1}) + r\left(\frac{3+3b}{2}\right) |z_{1}|^{\frac{1+3b}{2}} z_{2} + k_{1}|z_{1}|^{b} |z_{2}|^{2} \operatorname{sign}(z_{1}) \right]$$

$$+ K[\delta]\left[|z_{2}|^{3} \operatorname{sign}(z_{2}) + r|z_{1}|^{\frac{3+3b}{2}} \operatorname{sign}(z_{1}) + \frac{2k_{1}}{(1+b)} |z_{1}|^{1+b} |z_{2}| \operatorname{sign}(z_{2})\right]$$

Also, from the results of the planar system (Appendix A.2-proposition 1):

$$\begin{split} \left|\delta\right| &\leq a_1 \left|d_2\right|^{\frac{2b}{1+b}} + a_2 \left|z_2\right|^{\frac{2b}{1+b}} \quad \text{with} \quad 0 < a_2 < k_2 \text{ and} \\ \\ a_1 &= k_2 \left(\frac{a_2 + 2k_2}{\left(\left(a_2 + k_2\right)^{\frac{1+b}{2b}} - k_2^{\frac{1+b}{2b}}\right)^{\frac{2b}{1+b}}}\right), \end{split}$$

Then, after rearrangement:

$$\begin{split} \dot{\tilde{V}}(z_{1},z_{2}) &\leq -\frac{2k_{1}}{(1+b)}(k_{2}-a_{2})|z_{1}|^{1+b}|z_{2}|^{\frac{1+3b}{1+b}} - \frac{rk_{1}}{2}|z_{1}|^{\frac{3+5b}{2}} - \frac{(k_{2}-a_{2})}{2}|z_{2}|^{\frac{3+5b}{1+b}} + \dot{V}_{1} \\ &+ |d_{1}| \left(\frac{2k_{1}^{2}}{(1+b)}|z_{1}|^{1+2b} + r\left(\frac{3+3b}{2}\right)|z_{1}|^{\frac{1+3b}{2}}|z_{2}| + k_{1}|z_{1}|^{b}|z_{2}|^{2}\right) \\ &+ a_{1}|d_{2}|^{\frac{2b}{1+b}} \left(|z_{2}|^{3} + r|z_{1}|^{\frac{3+3b}{2}} + \frac{2k_{1}}{(1+b)}|z_{1}|^{1+b}|z_{2}|\right) \end{split}$$

where

$$\dot{V}_{1} = -\frac{rk_{1}}{2}|z_{1}|^{\frac{3+5b}{2}} - \frac{(k_{2} - a_{2})}{2}|z_{2}|^{\frac{3+5b}{1+b}} + r\left(\frac{3+3b}{2}\right)|z_{1}|^{\frac{1+3b}{2}}|z_{2}|^{2} + r(k_{2} + a_{2})|z_{1}|^{\frac{3+3b}{2}}|z_{2}|^{\frac{2b}{1+b}}$$

Note that from Section 2.2 (twisting algorithm): if the following inequalities

$$\min\left\{\lambda_{1} \frac{k_{1}^{\frac{3+3b}{2b}}(k_{2}-a_{2})}{(k_{2}+a_{2})^{\frac{3+5b}{2b}}}, \lambda_{2}k_{1}^{\frac{1+3b}{2+2b}}(k_{2}-a_{2})\right\} > r > 0$$

with $\lambda_{1} = \left(\frac{3+5b}{24+24b}\right)^{\frac{3+3b}{2b}} \left(\frac{3+5b}{16b}\right), \lambda_{2} = \left(\frac{3+5b}{8+24b}\right)^{\frac{1+3b}{2+2b}} \left(\frac{3+5b}{16+16b}\right) \left(\frac{2}{3+3b}\right)^{\frac{3+5b}{2+2b}},$

hold, then the function \dot{V}_1 is negative definite. Such an r > 0 always exists for any $k_1 > 0$ and $k_2 > a_2 > 0$. Then,

$$\begin{split} \dot{\tilde{V}}(z_{1}, z_{2}) &\leq -\frac{rk_{1}}{2} |z_{1}|^{\frac{3+5b}{2}} - \frac{(k_{2} - a_{2})}{2} |z_{2}|^{\frac{3+5b}{1+b}} \\ &+ |d_{1}| \left(\frac{2k_{1}^{2}}{(1+b)} |z_{1}|^{1+2b} + r\left(\frac{3+3b}{2}\right) |z_{1}|^{\frac{1+3b}{2}} |z_{2}| + k_{1} |z_{1}|^{b} |z_{2}|^{2} \right) \\ &+ a_{1} |d_{2}|^{\frac{2b}{1+b}} \left(|z_{2}|^{3} + r|z_{1}|^{\frac{3+3b}{2}} + \frac{2k_{1}}{(1+b)} |z_{1}|^{1+b} |z_{2}| \right) \end{split}$$

Applying lemma 2.1,

$$\begin{split} & \left|z_{1}\right|^{1+b}\left|z_{2}\right| \leq \frac{2}{3}\left|z_{1}\right|^{\frac{3+3b}{2}} + \frac{1}{3}\left|z_{2}\right|^{3}, \\ & \left|z_{1}\right|^{\frac{1+3b}{2}}\left|z_{2}\right| \leq \left(\frac{1+3b}{2+4b}\right) \left|z_{1}\right|^{1+2b} + \left(\frac{1+b}{2+4b}\right) \left|z_{2}\right|^{\frac{2+4b}{1+b}}, \\ & \left|z_{1}\right|^{b}\left|z_{2}\right|^{2} \leq \left(\frac{b}{1+2b}\right) \left|z_{1}\right|^{1+2b} + \left(\frac{1+b}{1+2b}\right) \left|z_{2}\right|^{\frac{2+4b}{1+b}}, \end{split}$$

together with lemma 2.2, and the bounds of the Lyapunov function, we have:

$$\begin{split} \dot{\tilde{V}}(z_{1},z_{2}) &\leq -\omega_{1} \bigg(\left| z_{1} \right|^{\frac{3+5b}{2}} + \left| z_{2} \right|^{\frac{3+5b}{1+b}} \bigg) + \left| d_{1} \right| \omega_{2} \bigg(\left| z_{1} \right|^{1+2b} + \left| z_{2} \right|^{\frac{2+4b}{1+b}} \bigg) + a_{1} \left| d_{2} \right|^{\frac{2b}{1+b}} \omega_{3} \bigg(\left| z_{1} \right|^{\frac{3+3b}{2}} + \left| z_{2} \right|^{3} \bigg) \\ &= -\omega_{1} \bigg(\bigg(\left| z_{1} \right|^{2+2b} \bigg)^{\frac{3+5b}{4+4b}} + \left| \left| z_{2} \right|^{4} \bigg)^{\frac{3+5b}{4+4b}} \bigg) + \left| d_{1} \right| \omega_{2} \bigg(\bigg(\left| z_{1} \right|^{2+2b} \bigg)^{\frac{1+2b}{2+2b}} + \left| z_{2} \right|^{4} \bigg)^{\frac{1+2b}{2+2b}} \bigg) \\ &+ a_{1} \left| d_{2} \right|^{\frac{2b}{1+b}} \omega_{3} \bigg(\bigg(\left| z_{1} \right|^{2+2b} \bigg)^{\frac{3}{4}} + \left| \left| z_{2} \right|^{4} \bigg)^{\frac{3}{4}} \bigg) \\ &\leq - \bigg(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \bigg) V^{\frac{3+5b}{4+4b}} + \left| d_{1} \right\| \bigg(\frac{2^{\frac{1}{2+2b}} \omega_{2}}{\underline{\pi}_{1}^{\frac{1+2b}{2+2b}}} \bigg) V^{\frac{1+2b}{2+2b}} + a_{1} \left| d_{2} \right|^{\frac{2b}{1+b}} \bigg(\frac{2^{\frac{1}{4}} \omega_{3}}{\underline{\pi}_{1}^{\frac{3}{4}}} \bigg) V^{\frac{3}{4}} \end{split}$$

where

$$\begin{split} \omega_{1} &= \min\left\{\frac{rk_{1}}{2}, \frac{(k_{2}-l)}{2}\right\}, \ \omega_{2} &= \max\left\{\left(\frac{2k_{1}^{2}}{(1+b)} + r\left(\frac{3+3b}{2}\right)\left(\frac{1+3b}{2+4b}\right) + k_{1}\left(\frac{b}{1+2b}\right)\right), \\ \left(r\left(\frac{3+3b}{2}\right)\left(\frac{1+b}{2+4b}\right) + k_{1}\left(\frac{1+b}{1+2b}\right)\right) \\ \omega_{3} &= \max\left\{\left(r + \frac{4k_{1}}{(3+3b)}\right), \left(1 + \frac{2k_{1}}{(3+3b)}\right)\right\}, \ \omega_{4} &= \begin{cases} \omega_{1} & \text{for } 0 < b \le 1, \\ \frac{\omega_{1}}{2^{\frac{b-1}{4+4b}}} & \text{for } b > 1 \end{cases} \end{split}$$

It can be further arranged into the following form to dominate the positive term,

$$\begin{split} \dot{V}(z_{1},z_{2}) & \leq -\frac{1}{2} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{3+5b}{4+4b}} - \frac{1}{4} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{3+5b}{4+4b}} + \left| d_{1} \right| \left(\frac{2^{\frac{1}{2+2b}}}{\underline{\pi}_{1}^{\frac{1+2b}{2+2b}}} \right) V^{\frac{1+2b}{2+2b}} - \frac{1}{4} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{3+5b}{4+4b}} \\ &+ a_{1} \left| d_{2} \right|^{\frac{2b}{1+b}} \left(\frac{2^{\frac{1}{4}} \omega_{3}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) V^{\frac{3}{4}} \\ &= -\frac{1}{2} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{3+5b}{4+4b}} - \frac{1}{4} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{1+2b}{2+2b}} \left(V^{\frac{1}{4}} - \left| d_{1} \right| \left(\frac{4\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}}{\omega_{4}} \right) \left(\frac{2^{\frac{1}{2+2b}} \omega_{2}}{\underline{\pi}_{1}^{\frac{1+2b}{2+2b}}} \right) \right) \\ &- \frac{1}{4} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{3}{4}} \left(V^{\frac{b}{2+2b}} - a_{1} \left| d_{2} \right|^{\frac{2b}{1+b}} \left(\frac{4\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}}{\omega_{4}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{3}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right) \\ &- \frac{1}{4} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{3}{4}} \left(V^{\frac{b}{2+2b}} - a_{1} \left| d_{2} \right|^{\frac{2b}{1+b}} \left(\frac{4\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}}{\omega_{4}} \right) \left(\frac{2^{\frac{1}{2}} \omega_{3}}{\underline{\pi}_{1}^{\frac{3}{2}}} \right) \right) \\ &\leq -\frac{1}{2} \left(\frac{\omega_{4}}{\overline{\pi}_{1}^{\frac{3+5b}{4+4b}}} \right) V^{\frac{3+5b}{4+4b}}, \quad \forall V(z_{1}, z_{2}) \geq \Delta := \max \left\{ \left| d_{1} \right|^{4} \left(\frac{4\overline{\pi}_{1}^{\frac{3+5b}} \omega_{4}}{\omega_{4}} \right) \left(\frac{2^{\frac{1}{4}} \omega_{3}}{\underline{\pi}_{1}^{\frac{3}{4}}} \right) \right\}^{\frac{2+2b}{b}} \right\} \right\}$$

Hence, invoking lemma 2.6, the states will reach the compact set

$$\Omega_{\Delta} = \left\{ \left(z_1, z_2 \right) \in \mathbb{R}^2 : V(z_1, z_2) \le \Delta \right\} \text{ in finite time. Note that when } |d_1| = |d_2| = 0, \text{ everywhere}$$
$$\dot{V}(z) \le -\left(\frac{\omega_4}{\overline{\pi}_1^{\frac{3+5b}{4+4b}}} \right) V^{\frac{3+5b}{4+4b}}, \ \forall (z_1, z_2) \in \mathbb{R}^2$$

For the case of b = 0:

First we consider the planar case:

Now let us consider the planar dynamics with b = 0 and $k_1 > k_2 > 0$.

$$\dot{z}_1 = z_2 + d_1,$$

 $\dot{z}_2 = -k_1 \operatorname{sign}(z_1) - k_2 \operatorname{sign}(z_2 + d_2)$

$$V(z_1, z_2) = k_1^2 |z_1|^2 + \frac{1}{4} |z_2|^4 + r|z_1|^{\frac{3}{2}} \operatorname{sign}(z_1) z_2 + k_1 |z_1| |z_2|^2$$

Now, for b = 0, the Lyapunov function is not differentiable at $z_1 = 0$ but Lipschitz continuous and the system is described by the differential inclusion:

$$\dot{\widetilde{V}}(\mathbf{z}(t),t) = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^T K \begin{bmatrix} \dot{z}_1 \\ \dot{z}_2 \end{bmatrix} (\mathbf{z},t) \subset \dot{\widetilde{V}}_1 + \dot{\widetilde{V}},$$

where

$$\begin{split} \dot{\vec{V}}_{1} &= \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^{T} K[f](z_{1},z_{2}), \ \dot{\vec{V}}_{2} = \bigcap_{\xi \in \partial V(\mathbf{z}(t),t)} \xi^{T} K\begin{bmatrix}d_{1}\\0\end{bmatrix}, \\ K[f](z_{1},z_{2}) &= \begin{cases} \left\{ \begin{pmatrix} z_{2}, \\ -k_{1} \text{sign}(z_{1}) - k_{2} \text{sign}(z_{2}) \end{pmatrix} \right\} \forall z_{1} \neq 0, |z_{2}| > |d_{2}| \\ \left\{ \begin{pmatrix} z_{2}, \\ -k_{1}[-1, 1] - k_{2} \text{sign}(z_{2}) \end{pmatrix} \right\} \forall z_{1} = 0, |z_{2}| > |d_{2}| \\ \left\{ \begin{pmatrix} z_{2}, \\ -k_{1} \text{sign}(z_{1}) - k_{2} \text{SGN}(z_{2} + d_{2}) \end{pmatrix} \right\} \forall z_{1} \neq 0, |z_{2}| \leq |d_{2}| \\ \left\{ \begin{pmatrix} z_{2}, \\ -k_{1}[-1, 1] - k_{2} \text{SGN}(z_{2} + d_{2}) \end{pmatrix} \right\} \forall z_{1} = 0, |z_{2}| \leq |d_{2}| \\ \left\{ \begin{pmatrix} z_{2}, \\ -k_{1}[-1, 1] - k_{2} \text{SGN}(z_{2} + d_{2}) \end{pmatrix} \right\} \forall z_{1} = 0, |z_{2}| \leq |d_{2}| \end{cases} \end{split}$$

and

$$SGN(x) = \begin{cases} -1 & x < 0\\ [-1, 1] & x = 0\\ 1 & x > 0 \end{cases}$$

$$\partial V = \mathbf{K} \left[\nabla V \right] = \mathbf{K} \left[\frac{\partial V}{\partial z_1} \right]$$

$$\subset \left[\mathbf{K} \left[\frac{\partial V}{\partial z_1} \right] \right] \left\{ \left\{ \left(2k_1^2 z_1 + \frac{3}{2}r|z_1|^{\frac{1}{2}} z_2 + k_1 \operatorname{sign}(z_1)|z_2|^2 \right) \right\} \forall z_1 \neq 0, z_2 \in \mathbb{R} \right\}$$

$$\left[\mathbf{K} \left[\frac{\partial V}{\partial z_2} \right] \right] \left\{ \left\{ \left([-1, 1]k_1 z_2^2 \\ z_2^{-3} + r|z_1|^{\frac{3}{2}} \operatorname{sign}(z_1) + 2k_1|z_1|z_2 \right) \right\} \forall z_1 \neq 0, z_2 \in \mathbb{R} \right\}$$

Let us define:

$$\left| \frac{\partial V}{\partial z_1} \right| \coloneqq \sup \left\{ \left| \xi_1 \right| \colon \xi_1 \in \mathbf{K} \left[\frac{\partial V}{\partial z_1} \right] \right\}, \text{ and } \left| \frac{\partial V}{\partial z_2} \right| \coloneqq \sup \left\{ \left| \xi_2 \right| \colon \xi_2 \in \mathbf{K} \left[\frac{\partial V}{\partial z_2} \right] \right\},$$

$$\text{with } \left| \frac{\partial V}{\partial z_1} \right| \le 2k_1^2 |z_1| + \frac{3}{2}r|z_1|^{\frac{1}{2}}|z_2| + k_1|z_2|^2,$$

Thus, the term

$$\dot{\widetilde{V}}_{2} = \bigcap_{\xi_{1} \in \mathbf{K} \left[\frac{\partial V}{\partial z_{1}}\right]} \xi_{1} K[d_{1}] \leq \left| \frac{\partial V}{\partial z_{1}} \right| |d_{1}| \leq |d_{1} \left(2k_{1}^{2} |z_{1}| + \frac{3}{2}r|z_{1}|^{\frac{1}{2}} |z_{2}| + k_{1} |z_{2}|^{2} \right)$$

Computing $\dot{\widetilde{V}}_1$ for each case, we have

For $z_1 \neq 0$, $|z_2| > |d_2|$:

$$\dot{\tilde{V}}_{1} = \frac{3}{2}r|z_{1}|^{\frac{1}{2}}z_{2}^{2} - 2k_{1}k_{2}|z_{1}||z_{2}| - k_{2}|z_{2}|^{3} - rk_{1}|z_{1}|^{\frac{3}{2}} - rk_{2}|z_{1}|^{\frac{3}{2}}\operatorname{sign}(z_{1}z_{2})$$

For $z_1 = 0, |z_2| > |d_2|$:

Let $(\xi_2 k_1 z_2^2, z_2^3)^T$ with $\xi_2 \in [-1, 1]$ be an arbitrary element of $\partial V(\mathbf{z}, t)$, then

$$\dot{\widetilde{V}}_{1} = \bigcap_{\xi_{2} \in [-1, 1]} [\xi_{2} - 1, \xi_{2} + 1] k_{1} z_{2}^{3} - k_{2} |z_{2}|^{3} = -k_{2} |z_{2}|^{3}$$

For $z_1 \neq 0$, $|z_2| \leq |d_2|$:

$$\dot{\tilde{V}}_{1} = \frac{3}{2}r|z_{1}|^{\frac{1}{2}}|z_{2}|^{2} - 2k_{1}k_{2}|z_{1}|z_{2}SGN(z_{2}+d_{2}) - k_{2}|z_{2}|^{3}sign(z_{2})SGN(z_{2}+d_{2}) -rk_{1}|z_{1}|^{\frac{3}{2}} - rk_{2}|z_{1}|^{\frac{3}{2}}sign(z_{1})SGN(z_{2}+d_{2})$$

For $z_1 = 0, |z_2| \le |d_2|$:

$$\dot{\vec{V}}_{1} = \bigcap_{\xi_{2} \in [-1, 1]} [\xi_{2} - 1, \xi_{2} + 1] k_{1} |z_{2}|^{3} \operatorname{sign}(z_{2}) - k_{2} |z_{2}|^{3} \operatorname{sign}(z_{2}) \operatorname{SGN}(z_{2} + d_{2})$$
$$= -k_{2} |z_{2}|^{3} \operatorname{sign}(z_{2}) \operatorname{SGN}(z_{2} + d_{2})$$

Thus, for $\forall (z_1, z_2) \in \mathbb{R}^2$

$$\dot{\widetilde{V}}_{1} \leq \frac{3}{2}r|z_{1}|^{\frac{1}{2}}|z_{2}|^{2} - 2k_{1}k_{2}|z_{1}|z_{2}\text{SGN}(z_{2}+d_{2}) -k_{2}|z_{2}|^{3}\text{sign}(z_{2})\text{SGN}(z_{2}+d_{2}) - rk_{1}|z_{1}|^{\frac{3}{2}} + rk_{2}|z_{1}|^{\frac{3}{2}}$$

Hence, the generalized time derivative of the Lyapunov function, after rearrangement:

$$\dot{\widetilde{V}} = \dot{\widetilde{V}}_{1} + \dot{\widetilde{V}}_{2} \le 2k_{1}^{2}|z_{1}||d_{1}| + \frac{3}{2}r|z_{1}|^{\frac{1}{2}}|z_{2}||d_{1}| + k_{1}|z_{2}|^{2}|d_{1}| + \frac{3}{2}r|z_{1}|^{\frac{1}{2}}|z_{2}|^{2} - 2k_{1}k_{2}|z_{1}|z_{2}SGN(z_{2} + d_{2}) - k_{2}|z_{2}|^{3}sign(z_{2})SGN(z_{2} + d_{2}) - r(k_{1} - k_{2})|z_{1}|^{\frac{3}{2}}$$

For analysis, the state space is divided into three regions:

$$\begin{split} \phi_{1}(|\zeta|) &= \left\{ (z_{1}, z_{2}) \in \mathbb{R}^{2} : z_{1} \in \mathbb{R}, |z_{2}| \geq \frac{2k_{1}}{k_{2}} |\zeta| \right\}, \\ \phi_{2}(|\zeta|) &= \left\{ (z_{1}, z_{2}) \in \mathbb{R}^{2} : |z_{1}| \geq a_{3} |\zeta|^{2}, |z_{2}| \leq \frac{2k_{1}}{k_{2}} |\zeta| \right\}, \\ \phi_{3}(|\zeta|) &= \left\{ (z_{1}, z_{2}) \in \mathbb{R}^{2} : |z_{1}| \leq a_{3} |\zeta|^{2}, |z_{2}| \leq \frac{2k_{1}}{k_{2}} |\zeta| \right\}, \end{split}$$

where $|\zeta| = \max\{|d_1|, |d_2|\}$ and

$$a_{3} := \max\left\{ \left(\frac{24k_{1}^{2}}{r(k_{1}-k_{2})}\right)^{2}, \frac{\frac{12k_{1}}{k_{2}}\left(1+\left(\frac{2k_{1}}{k_{2}}\right)\right)}{(k_{1}-k_{2})}, \left(\frac{48\left(\frac{k_{1}^{3}}{k_{2}^{2}}\right)}{r(k_{1}-k_{2})}\right)^{\frac{2}{3}}\right\} > 0$$

since $k_1 > k_2$

Consider the region:

$$\phi_1(|\zeta|) = \left\{ (z_1, z_2) \in \mathbb{R}^2 : z_1 \in \mathbb{R}, |z_2| \ge \frac{2k_1}{k_2} |\zeta| \right\}$$

Note that in this region, the following properties apply:

 $z_2 \text{SGN}(z_2 + d_2) = |z_2| \text{sign}(z_2) \text{SGN}(z_2 + d_2) = |z_2|$ since $\frac{2k_1}{k_2} |\zeta_i| \ge |d_2|$, where the equality only occur when $|d_1| = |d_2| = 0$; also when $|d_2| = 0$, the above is trivially satisfied, when $|d_2| \ne 0$, we have $\text{SGN}(z_2 + d_2) = \text{sign}(z_2 + d_2)$, (since $(z_2 + d_2) \ne 0$ in this region) $= \text{sign}(z_2)$ since $|z_2| > |d_2|$ in this region Then, the time derivative of the Lyapunov function becomes:

$$\dot{\widetilde{V}} \leq -k_{1}|z_{1}|(k_{2}|z_{2}|-2k_{1}|d_{1}|)-|z_{2}|^{2}\left(\frac{k_{2}}{2}|z_{2}|-k_{1}|d_{1}|\right)-k_{2}|z_{2}|\left((k_{1}|z_{1}|)^{\frac{1}{2}}-\left(\frac{1}{2}|z_{2}|^{2}\right)^{\frac{1}{2}}\right)^{2}\\-\frac{1}{2}|z_{1}|^{\frac{1}{2}}|z_{2}|^{2}\left(2^{\frac{1}{2}}k_{1}^{\frac{1}{2}}k_{2}-3r\right)-\frac{1}{2}|z_{1}|^{\frac{1}{2}}|z_{2}|\left(2^{\frac{1}{2}}k_{1}^{\frac{1}{2}}k_{2}|z_{2}|-3r|d_{1}|\right)-r(k_{1}-k_{2})|z_{1}|^{\frac{3}{2}}$$

and $\dot{\widetilde{V}} < 0$ for any positive

$$0 < r < \frac{2^{\frac{1}{2}}}{3}k_1^{\frac{1}{2}}k_2$$
, with $k_1 > k_2 > 0$

for all
$$\forall z_1 \in \mathbb{R}$$
, $|z_2| \ge \max\left\{\frac{2k_1|d_1|}{k_2}, \frac{3r}{2^{\frac{1}{2}}k_1^{\frac{1}{2}}k_2}|d_1|\right\} = \frac{2k_1|d_1|}{k_2} > |d_1| \text{ and } |z_2| > |d_2|,$

which is sufficiently satisfied for the states in this region.

Next, consider the region:

$$\phi_{2}(|\zeta|) = \left\{ (z_{1}, z_{2}) \in \mathbb{R}^{2} : |z_{1}| \ge a_{3} |\zeta|^{2}, |z_{2}| \le \frac{2k_{1}}{k_{2}} |\zeta| \right\}$$

Then, the time derivative of the Lyapunov function becomes:

$$\dot{\tilde{V}} \le 2k_1^2 |z_1| |d_1| + \frac{3}{2}r|z_1|^{\frac{1}{2}} |z_2| |d_1| + k_1 |z_2|^2 |d_1| + \frac{3}{2}r|z_1|^{\frac{1}{2}} |z_2|^2 + 2k_1 k_2 |z_1| |z_2| + k_2 |z_2|^3 - r(k_1 - k_2) |z_1|^{\frac{3}{2}}$$

After rearrangement,

$$\dot{\widetilde{V}} \leq |z_1| (2k_1^2 |d_1| + 2k_1k_2 |z_2|) + \frac{3}{2}r|z_1|^{\frac{1}{2}} |z_2| (|d_1| + |z_2|) + k_1 |z_2|^2 |d_1| + k_2 |z_2|^3 - r(k_1 - k_2) |z_1|^{\frac{3}{2}} |z_1|^{\frac{3}{2}} |d_1| + |z_2|^2 |d_1| + k_2 |z_2|^3 - r(k_1 - k_2) |z_1|^{\frac{3}{2}} |z_1|^{\frac{3}{2}} |d_1| + |z_2|^2 |d_1| + k_2 |z_2|^3 - r(k_1 - k_2) |z_1|^{\frac{3}{2}} |d_1| + |z_2|^2 |d_1| + k_2 |z_2|^3 - r(k_1 - k_2) |z_1|^{\frac{3}{2}} |d_1| + |z_2|^2 |d_1| + k_2 |z_2|^3 - r(k_1 - k_2) |z_1|^{\frac{3}{2}} |d_1| + |z_2|^2 |d_1| + k_2 |z_2|^3 - r(k_1 - k_2) |z_1|^{\frac{3}{2}} |d_1| + |z_2|^2 |d_1| + |z_2|$$

Note that in this region, we have:

$$|z_2| \leq \frac{2k_1}{k_2} |\zeta|, \quad \text{also}, |\zeta| \geq |d_1|$$

Hence,

$$\begin{split} \dot{\widetilde{V}} &\leq -|z_1| \left(\frac{r}{4} (k_1 - k_2) |z_1|^{\frac{1}{2}} - 6k_1^{-2} |\zeta_i| \right) - r|z_1|^{\frac{1}{2}} \left(\frac{1}{4} (k_1 - k_2) |z_1| - \frac{3k_1}{k_2} \left(1 + \frac{2k_1}{k_2} \right) |\zeta_i|^2 \right) \\ &- \left(\frac{r}{4} (k_1 - k_2) |z_1|^{\frac{3}{2}} - 12 \left(\frac{k_1^{-3}}{k_2^{-2}} \right) |\zeta_i|^3 \right) - \frac{r}{4} (k_1 - k_2) |z_1|^{\frac{3}{2}} \end{split}$$

where $\dot{\widetilde{V}} < 0$ for any positive r > 0, with $k_1 > k_2 > 0$

for all

$$|z_{1}| \geq \max\left\{ \left(\frac{24k_{1}^{2}}{r(k_{1}-k_{2})}\right)^{2}, \frac{\frac{12k_{1}}{k_{2}}\left(1+\left(\frac{2k_{1}}{k_{2}}\right)\right)}{(k_{1}-k_{2})}, \left(\frac{48\left(\frac{k_{1}^{3}}{k_{2}^{2}}\right)}{r(k_{1}-k_{2})}\right)^{2}\right\} \times |\zeta_{i}|^{2}, \text{ and } |z_{2}| \leq \frac{2k_{1}}{k_{2}}|\zeta_{i}|$$

which is sufficiently satisfied for the states in this region.

Finally, consider the region
$$\phi_3(|\zeta|) = \left\{ (z_1, z_2) \in \mathbb{R}^2 : |z_1| \le a_3 |\zeta|^2, |z_2| \le \frac{2k_1}{k_2} |\zeta| \right\}$$
 which is a

compact set. Hence, with the above results, we have

$$\dot{\widetilde{V}} < 0$$
, for $(z_1, z_2) \notin \phi_3(|\zeta|)$

Now, define a Lyapunov level set:

$$\Omega_{\mu} = \left\{ (z_1, z_2) \in \mathbb{R}^2 : V \le \rho_1 (|\zeta|) \right\}$$

where the positive definite function $\rho_{l}(|\zeta|)$ is defined as follows:

$$\rho_1(|\zeta|) = \max_{(z_1, z_2) \in \mathrm{bd} \ \phi_3(|\zeta|)} V$$

which exists since the boundary of the set is compact and V is continuous. Then we observe that $\phi_3(|\zeta|) \subset \Omega_{\mu}$. As a result, we have

$$\frac{d}{dt}V(z_1, z_2) \stackrel{a.e.}{\in} \dot{\widetilde{V}}(z_1, z_2) < 0 \quad for \quad V \ge \rho_1(|\zeta|)$$

which implies that the trajectories will enter the compact level set Ω_{μ} in finite time and stay in it once entered.

A.2 **PROPOSITION 1**

For every real numbers $k_2, d_2, z_2 \in \mathbb{R}$, b > 0, and $|d_2| \coloneqq \sup\{|\varepsilon_2|: \varepsilon_2 \in K[d_2]\}$ the following inequalities is satisfied

$$\begin{aligned} \left| k_{2} |z_{2}|^{\frac{2b}{1+b}} \operatorname{sign}(z_{2}) - k_{2} |z_{2} + d_{2}|^{\frac{2b}{1+b}} \operatorname{sign}(z_{2} + d_{2}) \right| \\ &\leq \max \begin{cases} k_{2} ||z_{2}| + |d_{2}||^{\frac{2b}{1+b}} - k_{2} ||z_{2}|^{\frac{2b}{1+b}}, \\ k_{2} |z_{2}|^{\frac{2b}{1+b}} - k_{2} ||z_{2}| - |d_{2}||^{\frac{2b}{1+b}} \operatorname{sign}(|z_{2}| - |d_{2}||) \end{cases}, \\ \\ &\left| k_{2} ||z_{2}|^{\frac{2b}{1+b}} \operatorname{sign}(z_{2}) - k_{2} ||z_{2} + d_{2}||^{\frac{2b}{1+b}} \operatorname{sign}(|z_{2} + d_{2}||) \leq a_{1} ||d_{2}|^{\frac{2b}{1+b}} + a_{2} ||z_{2}|^{\frac{2b}{1+b}}, \end{aligned} \end{aligned}$$

with some positive constants a_1 , a_2 define by $0 < a_2 < k_2$ and

$$a_{1} := k_{2} \left(\frac{a_{2} + 2k_{2}}{\left(\left(a_{2} + k_{2} \right)^{\frac{1+b}{2b}} - k_{2}^{\frac{1+b}{2b}} \right)^{\frac{2b}{1+b}}} \right) > 0$$

Proof of Proposition 1: Let $\delta = k_2 |z_2|^{\frac{2b}{1+b}} \operatorname{sign}(z_2) - k_2 |z_2 + d_2|^{\frac{2b}{1+b}} \operatorname{sign}(z_2 + d_2)$ and $|\delta| := \sup\{|\varepsilon_3|: \varepsilon_3 \in K[\delta]\}.$

First we are going to show that $\forall z_2 \in \mathbb{R}$, and any $\varepsilon_2 \in K[d_2]$,

$$|\varepsilon_{3}| \leq \max\left\{k_{2}||z_{2}| + |\varepsilon_{2}||^{\frac{2b}{1+b}} - k_{2}||z_{2}||^{\frac{2b}{1+b}}, k_{2}||z_{2}||^{\frac{2b}{1+b}} - k_{2}||z_{2}| - |\varepsilon_{2}||^{\frac{2b}{1+b}}\operatorname{sign}(||z_{2}|| - |\varepsilon_{2}||)\right\}$$

Let us note that when $|\varepsilon_2| = 0$, the above is trivially satisfied. Hence, for $|\varepsilon_2| > 0$, we consider the two cases of sign $(\varepsilon_2) < 0$ or sign $(\varepsilon_2) > 0$.

For sign(ε_2) > 0:

$$\left|\varepsilon_{3}\right| = \left|k_{2}\left|z_{2}\right|^{\frac{2b}{1+b}}\operatorname{sign}(z_{2})-k_{2}\left|z_{2}+\left|\varepsilon_{2}\right|\right|^{\frac{2b}{1+b}}\operatorname{sign}(z_{2}+\left|\varepsilon_{2}\right|\right)\right|$$

In the region of $z_2 \ge 0$, it becomes:

$$\left|\varepsilon_{3}\right| = \left|k_{2}\left|z_{2}\right|^{\frac{2b}{1+b}} - k_{2}\left\|z_{2}\right| + \left|\varepsilon_{2}\right\|^{\frac{2b}{1+b}}\right| = \left|k_{2}\left\|z_{2}\right| + \left|\varepsilon_{2}\right\|^{\frac{2b}{1+b}} - k_{2}\left|z_{2}\right|^{\frac{2b}{1+b}}\right|$$

In the region of $z_2 \leq 0$, it becomes:

$$\begin{aligned} |\varepsilon_{3}| &= \left| -k_{2} |z_{2}|^{\frac{2b}{1+b}} - k_{2} |-|z_{2}| + |\varepsilon_{2}| \Big|^{\frac{2b}{1+b}} \operatorname{sign}(-|z_{2}| + |\varepsilon_{2}|) \right| \\ &= \left| -k_{2} |z_{2}|^{\frac{2b}{1+b}} + k_{2} ||z_{2}| - |\varepsilon_{2}| \Big|^{\frac{2b}{1+b}} \operatorname{sign}(|z_{2}| - |\varepsilon_{2}|) \right| \\ &= \left| k_{2} |z_{2}|^{\frac{2b}{1+b}} - k_{2} ||z_{2}| - |\varepsilon_{2}| \Big|^{\frac{2b}{1+b}} \operatorname{sign}(|z_{2}| - |\varepsilon_{2}|) \right| \end{aligned}$$

For sign(ε_2) < 0:

$$\left|\varepsilon_{3}\right| = \left|k_{2}\left|z_{2}\right|^{\frac{2b}{1+b}}\operatorname{sign}(z_{2})-k_{2}\left|z_{2}-\left|\varepsilon_{2}\right|\right|^{\frac{2b}{1+b}}\operatorname{sign}(z_{2}-\left|\varepsilon_{2}\right|\right)\right|$$

In the region of $z_2 \ge 0$, it becomes:

$$|\varepsilon_{3}| = |k_{2}|z_{2}|^{\frac{2b}{1+b}} - k_{2}||z_{2}| - |\varepsilon_{2}||^{\frac{2b}{1+b}} \operatorname{sign}(|z_{2}| - |\varepsilon_{2}|)|$$

which is identical to the case of sign(ε_2) > 0 in the region $z_2 \le 0$.

In the region of $z_2 \leq 0$, it becomes:

$$\begin{aligned} |\varepsilon_{3}| &= \left| -k_{2}|z_{2}|^{\frac{2b}{1+b}} - k_{2}| - |z_{2}| - |\varepsilon_{2}| \right|^{\frac{2b}{1+b}} \operatorname{sign}\left(-|z_{2}| - |\varepsilon_{2}| \right) \\ &= \left| -k_{2}|z_{2}|^{\frac{2b}{1+b}} + k_{2}||z_{2}| + |\varepsilon_{2}||^{\frac{2b}{1+b}} \\ &= \left| k_{2}||z_{2}| + |\varepsilon_{2}||^{\frac{2b}{1+b}} - k_{2}|z_{2}|^{\frac{2b}{1+b}} \right| \end{aligned}$$

which is identical to the case of sign(ε_2) > 0 in the region $z_2 \ge 0$.

Hence, for all $|\varepsilon_2| \ge 0$ and $|z_2| \ge 0$:

$$\left|\varepsilon_{3}\right| \leq \max\left\{\left|k_{2}\left|\left|z_{2}\right|+\left|\varepsilon_{2}\right|\right|^{\frac{2b}{1+b}}-k_{2}\left|z_{2}\right|^{\frac{2b}{1+b}}\right|, \quad \left|k_{2}\left|z_{2}\right|^{\frac{2b}{1+b}}-k_{2}\left|\left|z_{2}\right|-\left|\varepsilon_{2}\right|\right|^{\frac{2b}{1+b}}\operatorname{sign}\left(\left|z_{2}\right|-\left|\varepsilon_{2}\right|\right)\right\}\right\}$$
(A2 - 1)

Next, note that the function $f(|\cdot|) = k_2 |\cdot|^{\frac{2b}{1+b}}$ is strictly increasing for $|z_2| \ge 0$ and $||z_2| + |\varepsilon_2|| \ge |z_2|$, we have

$$k_2 \|z_2\| + |\varepsilon_2||^{\frac{2b}{1+b}} \ge k_2 |z_2|^{\frac{2b}{1+b}} \Longrightarrow k_2 \|z_2\| + |\varepsilon_2||^{\frac{2b}{1+b}} - k_2 |z_2|^{\frac{2b}{1+b}} \ge 0$$

Also, note that the following is an increasing function, $k_2|z_2|^{\frac{2b}{1+b}}\operatorname{sign}(z_2)$ and since $|z_2| \ge |z_2| - |\varepsilon_2|$ we have

$$k_{2}|z_{2}|^{\frac{2b}{1+b}}\operatorname{sign}(|z_{2}|) \ge k_{2}||z_{2}| - |\varepsilon_{2}||^{\frac{2b}{1+b}}\operatorname{sign}(|z_{2}| - |\varepsilon_{2}|)$$
$$\implies k_{2}|z_{2}|^{\frac{2b}{1+b}} - k_{2}||z_{2}| - |\varepsilon_{2}||^{\frac{2b}{1+b}}\operatorname{sign}(|z_{2}| - |\varepsilon_{2}|) \ge 0$$

Hence we can write (A2 - 1) as:

$$|\varepsilon_{3}| \leq \max\left\{k_{2}||z_{2}| + |\varepsilon_{2}||^{\frac{2b}{1+b}} - k_{2}|z_{2}|^{\frac{2b}{1+b}}, k_{2}|z_{2}|^{\frac{2b}{1+b}} - k_{2}||z_{2}| - |\varepsilon_{2}||^{\frac{2b}{1+b}}\operatorname{sign}(|z_{2}| - |\varepsilon_{2}|)\right\}$$

Now, since $|d_2| \coloneqq \sup\{|\varepsilon_2|: \varepsilon_2 \in K[d_2]\}$, we have

$$\begin{aligned} |z_2| + |d_2| &\ge |z_2| + |\varepsilon_2| \Longrightarrow k_2 ||z_2| + |d_2||^{\frac{2b}{1+b}} \ge k_2 ||z_2| + |\varepsilon_2||^{\frac{2b}{1+b}} \\ &\implies k_2 ||z_2| + |d_2||^{\frac{2b}{1+b}} - k_2 |z_2|^{\frac{2b}{1+b}} \ge k_2 ||z_2| + |\varepsilon_2||^{\frac{2b}{1+b}} - k_2 |z_2|^{\frac{2b}{1+b}} \end{aligned}$$

also, we have

$$\begin{aligned} |z_{2}| - |\varepsilon_{2}| &\ge |z_{2}| - |d_{2}| \\ \Rightarrow k_{2} ||z_{2}| - |\varepsilon_{2}||^{\frac{2b}{1+b}} \operatorname{sign}(|z_{2}| - |\varepsilon_{2}|) &\ge k_{2} ||z_{2}| - |d_{2}||^{\frac{2b}{1+b}} \operatorname{sign}(|z_{2}| - |d_{2}|) \\ \Rightarrow k_{2} ||z_{2}|^{\frac{2b}{1+b}} - k_{2} ||z_{2}| - |d_{2}||^{\frac{2b}{1+b}} \operatorname{sign}(|z_{2}| - |d_{2}|) &\ge k_{2} ||z_{2}|^{\frac{2b}{1+b}} - k_{2} ||z_{2}| - |\varepsilon_{2}||^{\frac{2b}{1+b}} \operatorname{sign}(|z_{2}| - |\varepsilon_{2}|) \end{aligned}$$

Hence we have $\forall \varepsilon_2 \in K[d_2], \forall |z_2| \ge 0$:

$$|\varepsilon_{3}| \leq |\delta| \leq \max\left\{k_{2}||z_{2}| + |d_{2}||^{\frac{2b}{1+b}} - k_{2}|z_{2}|^{\frac{2b}{1+b}}, k_{2}|z_{2}|^{\frac{2b}{1+b}} - k_{2}||z_{2}| - |d_{2}||^{\frac{2b}{1+b}}\operatorname{sign}(|z_{2}| - |d_{2}|)\right\}$$

Thus, to determine the upper bound of $|\delta|$, it is sufficient to consider two cases only for $|z_2| \ge 0$ 0 where both cases are greater than or equal to zero for $|z_2| \ge 0$ and $|d_2| \ne 0$

Case 1:

Consider the following function:

$$(a_2 + k_2)|z_2|^{\frac{2b}{1+b}}$$
, with $0 < a_2 < k_2$

and for the following inequality to be satisfied

$$k_2 \|z_2\| + |d_2\|^{\frac{2b}{1+b}} \le (a_2 + k_2)|z_2|^{\frac{2b}{1+b}}$$

we need

$$|z_2| \ge C|d_2|$$
, where $C =: \left(\frac{k_2^{\frac{1+b}{2b}}}{(a_2 + k_2)^{\frac{1+b}{2b}} - k_2^{\frac{1+b}{2b}}}\right)$

where C > 0 for any $a_2 > 0$, which implies for $|z_2| \ge C \overline{d}$

$$k_{2} \left\| z_{2} \right\| + \left| d_{2} \right\|^{\frac{2b}{1+b}} - k_{2} \left| z_{2} \right|^{\frac{2b}{1+b}} \le a_{2} \left| z_{2} \right|^{\frac{2b}{1+b}}$$

and, for $|z_2| \le C |d_2|$, we have

$$k_2 \left\| z_2 \right\| + \left| d_2 \right|^{\frac{2b}{1+b}} \le k_2 \left| C \right| d_2 \right| + \left| d_2 \right|^{\frac{2b}{1+b}} = k_2 \left(C + 1 \right)^{\frac{2b}{1+b}} \left| d_2 \right|^{\frac{2b}{1+b}}$$

which also implies

$$k_2 \|z_2\| + |d_2||^{\frac{2b}{1+b}} - k_2|z_2|^{\frac{2b}{1+b}} \le k_2 (C+1)^{\frac{2b}{1+b}} |d_2|^{\frac{2b}{1+b}} - k_2|z_2|^{\frac{2b}{1+b}} = k_2 |z_2|^{\frac{2b}{1+b}} + k_2 |z_2|^{\frac{2b}{1+b}} \le k_2 (C+1)^{\frac{2b}{1+b}} |z_2|^{\frac{2b}{1+b}} = k_2 |z_2|^{\frac{2b}{1+b}} \le k_2 |z_2|^{\frac{2b}{1+b}} \le k_2 (C+1)^{\frac{2b}{1+b}} |z_2|^{\frac{2b}{1+b}} \le k_2 |z_2|^{\frac{2b}{1+b}} \le k_2$$

since $k_2(C+1)^{\frac{2b}{1+b}} |d_2|^{\frac{2b}{1+b}} - k_2 |z_2|^{\frac{2b}{1+b}} \le k_2(C+1)^{\frac{2b}{1+b}} |d_2|^{\frac{2b}{1+b}}$

$$\therefore k_2 \|z_2\| + |d_2||^{\frac{2b}{1+b}} - k_2|z_2|^{\frac{2b}{1+b}} \le k_2 (C+1)^{\frac{2b}{1+b}} |d_2|^{\frac{2b}{1+b}}$$

where the equality holds only when both $|z_2| = |d_2| = 0$.

Hence, for all $|z_2| \ge 0$, we have, $k_2 ||z_2| + |d_2||^{\frac{2b}{1+b}} - k_2 |z_2|^{\frac{2b}{1+b}} \le k_2 (C+1)^{\frac{2b}{1+b}} |d_2|^{\frac{2b}{1+b}} + a_2 |z_2|^{\frac{2b}{1+b}}$

Now consider case 2:

Consider the following function:

$$(a_2 + k_2)||z_2| - |d_2||^{\frac{2b}{1+b}} \operatorname{sign}(|z_2| - |d_2|)$$

and for the following inequality to be satisfied:

$$k_{2}|z_{2}|^{\frac{2b}{1+b}} \le (a_{2}+k_{2})||z_{2}| - |d_{2}||^{\frac{2b}{1+b}}\operatorname{sign}(||z_{2}|| - ||d_{2}||)$$

we need:

$$|z_2| \ge D|d_2|$$
, where $D =: \left(\frac{(a_2 + k_2)^{\frac{1+b}{2b}}}{(a_2 + k_2)^{\frac{1+b}{2b}} - k_2^{\frac{1+b}{2b}}}\right)$

where D > 1 for any $a_2 > 0$, which implies for $|z_2| \ge D |\overline{d}_2|$

$$k_{2}|z_{2}|^{\frac{2b}{1+b}} - k_{2}||z_{2}| - |d_{2}||^{\frac{2b}{1+b}}\operatorname{sign}(|z_{2}| - |d_{2}|) \le a_{2}||z_{2}| - |d_{2}||^{\frac{2b}{1+b}}\operatorname{sign}(|z_{2}| - |d_{2}|)$$

Note that $a_2 |z_2|^{\frac{2b}{1+b}} \ge a_2 ||z_2| - |d_2||^{\frac{2b}{1+b}} \operatorname{sign}(|z_2| - |d_2|)$

$$\Rightarrow k_2 |z_2|^{\frac{2b}{1+b}} - k_2 ||z_2| - |d_2||^{\frac{2b}{1+b}} \operatorname{sign}(|z_2| - |d_2|) \le a_2 |z_2|^{\frac{2b}{1+b}}$$

Also, for $|z_2| \le D |d_2|$, we have

$$\begin{split} k_{2}|z_{2}|^{\frac{2b}{1+b}} - k_{2}||z_{2}| - |d_{2}||^{\frac{2b}{1+b}} \operatorname{sign}(|z_{2}| - |d_{2}|) &\leq k_{2}|z_{2}|^{\frac{2b}{1+b}} + k_{2}||z_{2}| - |d_{2}||^{\frac{2b}{1+b}} \\ &\leq k_{2}D^{\frac{2b}{1+b}}|d_{2}|^{\frac{2b}{1+b}} + k_{2}(D-1)^{\frac{2b}{1+b}}|d_{2}|^{\frac{2b}{1+b}} \\ &= k_{2}\left(D^{\frac{2b}{1+b}} + (D-1)^{\frac{2b}{1+b}}\right)|d_{2}|^{\frac{2b}{1+b}} \end{split}$$

Note that D > 1 from above.

Hence, for all $|z_2| \ge 0$, we have

$$k_{2}|z_{2}|^{\frac{2b}{1+b}} - k_{2}||z_{2}| - |d_{2}||^{\frac{2b}{1+b}}\operatorname{sign}(|z_{2}| - |d_{2}|) \le k_{2}\left(D^{\frac{2b}{1+b}} + (D-1)^{\frac{2b}{1+b}}\right)|d_{2}|^{\frac{2b}{1+b}} + a_{2}|z_{2}|^{\frac{2b}{1+b}}$$

Thus,

$$\begin{split} |\delta| &\leq \max\left\{k_2 \left\|z_2\right| + \left|d_2\right|^{\frac{2b}{1+b}} - k_2 \left|z_2\right|^{\frac{2b}{1+b}}, k_2 \left|z_2\right|^{\frac{2b}{1+b}} - k_2 \left\|z_2\right| - \left|d_2\right|^{\frac{2b}{1+b}} \operatorname{sign}\left(\left|z_2\right| - \left|d_2\right|\right)\right\} \\ &\leq \max\left\{k_2 \left(C+1\right)^{\frac{2b}{1+b}} \left|d_2\right|^{\frac{2b}{1+b}} + a_2 \left|z_2\right|^{\frac{2b}{1+b}}, k_2 \left(D^{\frac{2b}{1+b}} + \left(D-1\right)^{\frac{2b}{1+b}}\right) \left|d_2\right|^{\frac{2b}{1+b}} + a_2 \left|z_2\right|^{\frac{2b}{1+b}}\right\} \\ \Rightarrow \left|\delta\right| &\leq a_1 \left|d_2\right|^{\frac{2b}{1+b}} + a_2 \left|z_2\right|^{\frac{2b}{1+b}} \end{split}$$

where

$$\begin{split} a_{1} &= k_{2} \times \max\left\{ \left(C+1\right)^{\frac{2b}{1+b}}, \left(D^{\frac{2b}{1+b}} + (D-1)^{\frac{2b}{1+b}}\right) \right\} \\ &= k_{2} \times \max\left\{ \frac{a_{2} + k_{2}}{\left(\left(a_{2} + k_{2}\right)^{\frac{1+b}{2b}} - k_{2}^{\frac{1+b}{2b}}\right)^{\frac{2b}{1+b}}}, \frac{a_{2} + 2k_{2}}{\left(\left(a_{2} + k_{2}\right)^{\frac{1+b}{2b}} - k_{2}^{\frac{1+b}{2b}}\right)^{\frac{2b}{1+b}}} \right\} \\ &= k_{2} \left(\frac{a_{2} + 2k_{2}}{\left(\left(a_{2} + k_{2}\right)^{\frac{1+b}{2b}} - k_{2}^{\frac{1+b}{2b}}\right)^{\frac{2b}{1+b}}} \right) \\ \Rightarrow \left|\delta\right| \leq a_{1} \left|d_{2}\right|^{\frac{2b}{1+b}} + a_{2} \left|z_{2}\right|^{\frac{2b}{1+b}} \end{split}$$

where a_2 is chosen as $0 < a_2 < k_2$.

Remark: The reason for having $a_2 < k_2$ will be apparent in the subsequent stability analysis.

A.3 **PROPOSITION 2**

For $x \in \mathbb{R}$, $y \in \mathbb{R}$, $p \ge 1$, the following inequality is satisfied

$$|x+y| \le 2^{1-\frac{1}{p}} |\operatorname{sig}(x)^p + \operatorname{sig}(y)^p|^{\frac{1}{p}}.$$

Proof of proposition 2: (We extend the proof from Lemma 2.3 of [148] where it is originally for integer *p* only.):

It is straightforward to prove that the function

$$f(\lambda) := \left(\operatorname{sig}(\lambda)^p + \operatorname{sig}(1-\lambda)^p \right) 2^{p-1} - 1$$

arrives its minimal value at $\lambda = 1/2$. Thus, $f(\lambda) \ge f(1/2) = 0$. Consequently,

$$\left(\operatorname{sig}(\lambda)^{p} + \operatorname{sig}(1-\lambda)^{p}\right) 2^{p-1} \geq 1, \quad \forall \lambda \in \mathbb{R}$$

In the case when $x + y \neq 0$, set $\lambda = x/(x + y)$. Then, the proposition follows immediately. When x + y = 0, the proposition is trivial. Hence, the proposition is true $\forall x \in \mathbb{R}, y \in \mathbb{R}$.

Appendix B

B.1 PROOF OF DESIRED ERROR DYNAMICS (CH 5 & 6)

For the case of b > 0:

Consider the following dynamics:

$$\dot{v}_{1i} = v_{2i} + s_i$$

$$\dot{v}_{2i} = -k_{ai} \left(v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right) + k_{ai} \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}}} \left[v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right]$$

$$- \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right]^{\frac{2b}{1+b}} \operatorname{sign} \left(v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right) = \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} | v_{1i} |^b \operatorname{sign} \left(v_{1i} \right) \right]$$

where v_{1i} , v_{2i} , s_i , $\tilde{e}_{2i} \in \mathbb{R}$, are the scalar state variables, k_{1i} , k_{2i} , k_{ai} , ε_{1i} , ε_{2i} , μ_i are positive constants, $b \ge 0$ real number

with

$$\varepsilon_{3i} := \begin{cases} \varepsilon_{1i} - k_{ai} \mu_i - k_{ai} \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}} - \varepsilon_{2i}, & \text{for } |\widetilde{e}_{2i}| \neq 0, \\ \\ \varepsilon_{1i} - k_{ai} \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}} - \varepsilon_{2i}, & \text{for } \widetilde{e}_{2i} = 0 \quad (\text{for the case of full state feedback}) \end{cases}$$

The results in this section are applicable to the desired dynamics section of both chapter 5 & 6. In particular, for Chapter 5, let $\tilde{e}_{2i} = 0$, while for chapter 6, no changes required, and the same differential equations are obtained.

Lyapunov function:

$$V = \left(\frac{1}{2}v_{2i}^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |r|^{b} \operatorname{sign}(r)\right] dr\right)^{2} + r_{1} \left(\frac{3+3b}{2}\right) \left(\int_{0}^{v_{1i}} |r|^{\frac{1}{2}} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \left[|r|^{\frac{3b}{2}}\right] dr\right) v_{2i}$$
$$+ r_{1} (1+b) k_{ai} \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}} \left[\int_{0}^{v_{1i}} |r|^{\frac{3}{2}} \operatorname{sign}(r) - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \left[|r|^{\frac{3}{2}} \operatorname{sign}(r)\right] dr\right)$$

where r_1 is a positive constant scalar, will be shown as a strict Lyapunov function. Note that the above function is locally Lipschitz and regular. It is differentiable everywhere for b > 0, and not differentiable on $v_{1i} = 0$ for b = 0. In particular, when none of the terms of the dynamics are saturated, the Lyapunov function is indeed identical to that proposed in Section 2.2 for the twisting-based family of algorithms.

Sign definiteness of V

Note the following properties:

$$\int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |r|^{b} \operatorname{sign}(r) \right] dr = \frac{k_{1i}}{1+b} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}} \left[|v_{1i}|^{1+b} \right] + \varepsilon_{1i} \left[|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left[|v_{1i}| \right] \right]$$

$$r_{1}\left(\frac{3+3b}{2}\right)_{0}^{\nu_{1}}|r|^{\frac{1}{2}}\operatorname{sat}_{\left(\frac{\delta_{1l}}{k_{1l}}\right)^{\frac{3}{2}}}[|r|^{\frac{3b}{2}}] dr$$

$$= \left(r_{1}\operatorname{sat}_{\left(\frac{\delta_{1l}}{k_{1l}}\right)^{\frac{3+3b}{2b}}}\left[|v_{1l}|^{\frac{3+3b}{2}}\right] + r_{1}\left(1+b\right)\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{3}{2}}\left[|v_{1l}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{\delta_{1l}}{k_{1l}}\right)^{\frac{3}{2b}}}\left[|v_{1l}|^{\frac{3}{2}}\right]\right]\right)\operatorname{sign}(v_{1i})$$

$$= \left(r_{1}\operatorname{sat}_{\left(\frac{\delta_{1l}}{k_{1l}}\right)^{\frac{3+3b}{2b}}}\left[|v_{1l}|^{\frac{3+3b}{2}}\right] + r_{1}\left(1+b\right)\operatorname{sat}_{\left(\frac{\delta_{1l}}{k_{1l}}\right)^{\frac{3}{2}}}\left[|v_{1l}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{\delta_{1l}}{k_{1l}}\right)^{\frac{3}{2b}}}\left[|v_{1l}|^{\frac{3}{2}}\right]\right]\right)\operatorname{sign}(v_{1i})$$

$$= \left(r_{1}\left(1+b\right)\operatorname{sat}_{\left(\frac{\delta_{1l}}{k_{1l}}\right)^{\frac{3}{2}}}\left[|v_{1l}|^{\frac{3b}{2}}\right]v_{1l}|^{\frac{3}{2}} - r_{1}b\operatorname{sat}_{\left(\frac{\delta_{1l}}{k_{1l}}\right)^{\frac{3+3b}{2b}}}\left[|v_{1l}|^{\frac{3+3b}{2}}\right]\operatorname{sign}(v_{1i})$$

Using lemma 2.2,

$$\begin{aligned} |v_{1i}|^{\frac{3}{2}} = \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} [|v_{1i}|] + \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} [|v_{1i}|] \right)^{\frac{3}{2}} &\leq 2^{\frac{1}{2}} \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} [|v_{1i}|] \right)^{\frac{3}{2}} + 2^{\frac{1}{2}} \left(\operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} [|v_{1i}|] \right)^{\frac{3}{2}} \\ &= 2^{\frac{1}{2}} \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} [|v_{1i}|] \right)^{\frac{3}{2}} + 2^{\frac{1}{2}} \left(\operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{b}}} [|v_{1i}|] \right)^{\frac{3}{2}} \end{aligned}$$

Note that the term

$$r_{1}(1+b)k_{ai}\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}\left(\int_{0}^{v_{1i}}|r|^{\frac{3}{2}}\operatorname{sign}(r) - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}}\left[|r|^{\frac{3}{2}}\operatorname{sign}(r)\right]dr\right) \geq 0$$

$$r_{1}\left(\frac{3+3b}{2}\right)\left(\int_{0}^{v_{l}}|r|^{\frac{1}{2}}\operatorname{sat}_{\left(\frac{\delta_{l}}{k_{l}}\right)^{\frac{3}{2}}}[|r|^{\frac{3b}{2}}] dr\right)v_{2i} \geq -\left(r_{1}\left(1+b\right)\operatorname{sat}_{\left(\frac{\delta_{l}}{k_{l}}\right)^{\frac{3}{2}}}\left[|v_{li}|^{\frac{3b}{2}}\right]|v_{li}|^{\frac{3}{2}}\right)v_{2i}|$$

$$\geq -\left(r_{1}b\operatorname{sat}_{\left(\frac{\delta_{ll}}{k_{l}}\right)^{\frac{3+3b}{2b}}}\left[|v_{li}|^{\frac{3+3b}{2}}\right] \left||v_{li}|| - \operatorname{sat}_{\left(\frac{\delta_{ll}}{k_{li}}\right)^{\frac{3}{2}}}\left[|v_{li}||\right]\right)^{\frac{3}{2}}\right)|v_{2i}|$$

$$\geq -\left(r_{1}2^{\frac{1}{2}}(1+b)\operatorname{sat}_{\left(\frac{\delta_{ll}}{k_{ll}}\right)^{\frac{3}{2}}}\left[|v_{li}|^{\frac{3+3b}{2}}\right] \left||v_{li}|| - \operatorname{sat}_{\left(\frac{\delta_{ll}}{k_{li}}\right)^{\frac{3}{2}}}\left[|v_{li}|^{\frac{3+3b}{2}}\right]\right)|v_{2i}|$$

$$\geq -\left(r_{1}2^{\frac{1}{2}}(1+b)\left(\frac{\delta_{li}}{k_{li}}\right)^{\frac{3}{2}}\left[|v_{li}| - \operatorname{sat}_{\left(\frac{\delta_{ll}}{k_{li}}\right)^{\frac{3+3b}{2}}}\left[|v_{li}|^{\frac{3+3b}{2}}\right]\right)|v_{2i}|$$

$$\geq -\left(r_{1}2^{\frac{1}{2}}(1+b)\left(\frac{\delta_{li}}{k_{li}}\right)^{\frac{3}{2}}\left[|v_{li}| - \operatorname{sat}_{\left(\frac{\delta_{li}}{k_{li}}\right)^{\frac{3}{2}}}\left[|v_{li}|^{\frac{3+3b}{2}}\right]\right)|v_{2i}|$$

$$\geq -\left(r_{1}2^{\frac{1}{2}}(1+b)\left(\frac{\delta_{li}}{k_{li}}\right)^{\frac{3}{2}}\left[|v_{li}| - \operatorname{sat}_{\left(\frac{\delta_{li}}{k_{li}}\right)^{\frac{3}{2}}}\left[|v_{li}|^{\frac{3+3b}{2}}\right]\right)|v_{2i}|$$

Now, firstly note that the upper bound on the Lyapunov function:

$$V \leq \left(\frac{1}{2}|v_{2i}|^{2} + \frac{k_{1i}}{1+b}\operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}} \|v_{1i}|^{1+b}\right] + \mathcal{E}_{1i}\left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \|v_{1i}|\right)\right)^{2}$$
$$+ r_{1}\operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{3+3b}{2b}}} \left[|v_{1i}|^{\frac{3+3b}{2}}\right] |v_{2i}| + r_{1}\left(1+b\right)\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{3}{2}} \left(|v_{1i}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \left[|v_{1i}|^{\frac{3}{2}}\right]\right) |v_{2i}|$$
$$+ r_{1}\left(1+b\right)k_{ai}\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{3}{2}} \left[\int_{0}^{v_{1i}} |r|^{\frac{3}{2}}\operatorname{sign}(r) - \operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}} \left[|r|^{\frac{3}{2}}\operatorname{sign}(r)\right] dr\right)$$

which is positive definite and radially unbounded.

Next, the lower bound on the Lyapunov function, using the above properties, can be obtained as:

$$\begin{split} V &\geq \left(\frac{1}{2}v_{2i}^{2} + \frac{k_{1i}}{1+b}\operatorname{sat}_{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{1+\delta}{b}}} \left[|v_{1i}|^{1+\delta} \right] + \varepsilon_{1i} \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left[|v_{1i}| \right] \right) \right)^{2} \\ &- r_{1} 2^{\frac{1}{2}} \left(1 + b \right) \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{3}{2}} \left[|v_{1i}| - \operatorname{sat}_{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left[|v_{1i}| \right] \right)^{\frac{3}{2}} |v_{2i}| \\ &- r_{1} \left(2^{\frac{1}{2}} + \left(2^{\frac{1}{2}} - 1 \right) b \right) \operatorname{sat}_{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[|v_{1i}|^{\frac{3+3\delta}{2}} \right] |v_{2i}| \\ &\geq \frac{1}{8} |v_{2i}|^{4} + \frac{1}{2} \left(\frac{k_{1i}}{1+b} \right)^{2} \operatorname{sat}_{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+2\delta}{b}}} \left[|v_{1i}|^{2+2b} \right] + \frac{\varepsilon_{1i}^{2}}{2} \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left[|v_{1i}| \right] \right)^{\frac{3}{2}} \\ &+ \frac{1}{16} |v_{2i}|^{4} + \frac{\varepsilon_{1i}^{2}}{2} \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left[|v_{1i}| \right] \right)^{2} - r_{1} 2^{\frac{1}{2}} \left(1 + b \right) \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{3}{2}} \left[|v_{1i}| \right] \right)^{\frac{3}{2}} |v_{2i}| \\ &+ \frac{1}{16} |v_{2i}|^{4} + \frac{1}{2} \left(\frac{k_{1i}}{1+b} \right)^{2} \operatorname{sat}_{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left[|v_{1i}| \right]^{2} - r_{1} 2^{\frac{1}{2}} \left(1 + b \right) \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{3}{2}} \left[|v_{1i}| - \operatorname{sat}_{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left[|v_{2i}| \right] \\ &+ \frac{1}{16} |v_{2i}|^{4} + \frac{1}{2} \left(\frac{k_{1i}}{1+b} \right)^{2} \operatorname{sat}_{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{2+2\delta}{b}} \left[|v_{1i}|^{2+2\delta} \right] - r_{1} \left(2^{\frac{1}{2}} + \left(2^{\frac{1}{2}} - 1 \right) b \right) \operatorname{sat}_{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2}}} \left[|v_{1i}|^{\frac{3+3\delta}{2}} \right] |v_{2i}| \end{aligned}$$

Using lemma 2.1,

$$\frac{1}{16}|v_{2i}|^{4} + \frac{\varepsilon_{1i}^{2}}{2} \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}}[|v_{1i}|]\right)^{2} \ge \left(\frac{1}{4}\right)^{\frac{1}{4}} |v_{2i}| \left(\frac{4\varepsilon_{1i}^{2}}{6}\right)^{\frac{3}{4}} \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}}[|v_{1i}|]\right)^{\frac{3}{2}}$$

and

$$\frac{1}{16} |v_{2i}|^4 + \frac{1}{2} \left(\frac{k_{1i}}{1+b}\right)^2 \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{2+2b}{b}}} \left\| v_{1i} \right\|^{\frac{2+2b}{b}} \right] \ge \left(\frac{1}{4}\right)^{\frac{1}{4}} |v_{2i}| \left(\frac{2}{3} \left(\frac{k_{1i}}{1+b}\right)^2\right)^{\frac{3}{4}} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{2+2b}{2}}} \left[|v_{1i}|^{\frac{3+3b}{2}} \right]$$

Consequently:

$$V \ge \frac{1}{8} |v_{2i}|^4 + \frac{1}{2} \left(\frac{k_{1i}}{1+b}\right)^2 \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{2+2b}{b}}} \left[|v_{1i}|^{2+2b} \right] + \frac{\varepsilon_{1i}^2}{2} \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left[|v_{1i}| \right] \right)^2$$

with

$$\min\left\{ \left(\frac{2^{\frac{1}{2}}}{6^{\frac{3}{4}}}\right) \left(\frac{k_{1i}^{\frac{3}{2}}}{1+b}\right), \left(\frac{2^{\frac{1}{2}}}{6^{\frac{3}{4}}}\right) \frac{2^{\frac{1}{2}}}{\left(2^{\frac{1}{2}} + \left(2^{\frac{1}{2}} - 1\right)b\right)} \left(\frac{k_{1i}}{1+b}\right)^{\frac{3}{2}}\right\} > r_{1i}$$

Note that such an $r_1 > 0$ always exists for any positive k_{1i} and b. Thus, the Lyapunov function is positive definite and radially unbounded. In accordance to lemma 2.4, the time derivative of the Lyapunov function along the solutions of the system exists almost everywhere:

<u>Time derivative of Lyapunov function b > 0:</u>

Note that for b > 0, *V* is continuously differentiable:

$$\dot{\widetilde{V}} = \bigcap_{\xi \in \partial V(\mathbf{v}(t),t)} \xi^T K \begin{bmatrix} \dot{\mathbf{v}}_{1i} \\ \dot{\mathbf{v}}_{2i} \end{bmatrix} = \nabla V^T K \begin{bmatrix} \dot{\mathbf{v}}_{1i} \\ \dot{\mathbf{v}}_{2i} \end{bmatrix} \subset \dot{V}_{1s} + \dot{V}_{2s} + \dot{V}_{s}$$

with

$$\ddot{V}_{1s} = \begin{pmatrix} -\left(\frac{1}{2}v_{2l}^{2} + \int_{0}^{v_{1}} \operatorname{stat}_{e_{0}}[k_{1l}|r|^{b}\operatorname{sign}(r)]dr \right) \times \left(v_{2l}\operatorname{stat}_{e_{2l}}[k_{2l}|v_{2l}|^{\frac{2b}{1+b}}\operatorname{sign}(v_{2l})]\right) \\ + r_{i}\left(\frac{3+3b}{2}\right) |v_{1l}|^{\frac{1}{2}}\operatorname{stat}_{\left(\frac{e_{1l}}{k_{1l}}\right)^{\frac{3}{2}}}[|v_{1l}|^{\frac{3b}{2}}]v_{2l}(v_{2l}) \\ - r_{i}\operatorname{stat}_{\left(\frac{e_{1l}}{k_{1l}}\right)^{\frac{3+3b}{2}}}\left[|v_{1l}|^{\frac{3+3b}{2}}\right]\operatorname{sign}(v_{1l}) \times \begin{pmatrix} k_{al}v_{2l} - k_{al}\operatorname{stat}_{\left(\frac{e_{2l}}{k_{2l}}\right)^{\frac{3}{2b}}}[v_{2l}] \\ + \operatorname{sat}_{e_{2l}}[k_{2l}|v_{2l}|^{\frac{2b}{1+b}}\operatorname{sign}(v_{2l})] \end{pmatrix} \\ - \frac{r_{i}\operatorname{stat}_{\left(\frac{e_{2l}}{k_{1l}}\right)^{\frac{3+3b}{2}}}{\left(\frac{e_{1l}}{k_{1l}}\right)^{\frac{3}{2}}}\left[|v_{1l}|^{\frac{3+3b}{2}}\right]\operatorname{sat}_{e_{1l}}[k_{1l}|v_{1l}|^{b}] \\ - \frac{r_{i}\left(1+b\right)\left(\frac{e_{1l}}{k_{1l}}\right)^{\frac{3}{2}}\left[|v_{1l}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{e_{2l}}{k_{2l}}\right)^{\frac{3}{2b}}}[v_{2l} + s_{i} + \operatorname{sat}_{\mu_{i}}[\tilde{e}_{2l}]]\right) \\ - \left(\frac{r_{i}(1+b)\left(\frac{e_{1l}}{k_{1l}}\right)^{\frac{3}{2}}\left[|v_{1l}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{e_{2l}}{k_{2l}}\right)^{\frac{3}{2b}}}[v_{2l} + s_{i} + \operatorname{sat}_{\mu_{i}}[\tilde{e}_{2l}]]\right) \\ - \left(\frac{r_{i}(1+b)\left(\frac{e_{1l}}{k_{1l}}\right)^{\frac{3}{2}}\left[|v_{1l}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{e_{2l}}{k_{2l}}\right)^{\frac{3}{2b}}}[v_{2l} + s_{i} + \operatorname{sat}_{\mu_{i}}[\tilde{e}_{2l}]\right]\right) \\ - \left(\frac{r_{i}(1+b)\left(\frac{e_{2l}}{k_{1l}}\right)^{\frac{3}{2}}\left[|v_{1l}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{e_{2l}}{k_{2l}}\right)^{\frac{3}{2b}}}[v_{1l}|^{\frac{3}{2}}\right]}\right]\operatorname{sign}(v_{1l}) \\ - \left(\frac{r_{i}(1+b)\left(\frac{e_{2l}}{k_{1l}}\right)^{\frac{3}{2}}\left[|v_{1l}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{e_{2l}}{k_{2l}}\right)^{\frac{3}{2b}}}\left[|v_{1l}|^{\frac{3}{2}}\right]}\right]\operatorname{sign}(v_{2l} + s_{i} + \operatorname{sat}_{\mu_{i}}[\tilde{e}_{2l}]\right]\right) \\ - \left(\frac{r_{i}(1+b)\left(\frac{e_{2l}}{k_{1l}}\right)^{\frac{3}{2}}\left[|v_{1l}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{e_{2l}}{k_{2l}}\right)^{\frac{3}{2b}}}\left[|v_{1l}|^{\frac{3}{2}}\right]}\right]\operatorname{sign}(v_{1l})\left[\operatorname{sat}_{e_{2l}}[k_{1l}|v_{1l}|^{b}\operatorname{sign}(v_{1l})\right]\right) \\ - \left(\frac{r_{i}(1+b)\left(\frac{e_{2l}}{k_{1l}}\right)^{\frac{3}{2}}\left[|v_{1l}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{e_{2l}}{k_{2l}}\right)^{\frac{3}{2b}}}\left[|v_{1l}|^{\frac{3}{2}}\right]}\right]\operatorname{sign}(v_{1l})\left[\operatorname{sat}_{e_{2l}}[k_{1l}|v_{1l}|^{b}\operatorname{sign}(v_{1l})\right]\right]} \\ - \left(\frac{r_{i}(1+b)\left(\frac{e_{2l}}{k_{2l}}\right)^{\frac{3}{2}}\left[|v_{2l}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{e_{2l}}{k_{2l}}\right)^{\frac{3}{2b}}}\left[|v_{2l}|^$$

$$\dot{V}_{2s} = \begin{bmatrix} -\frac{r_{1}}{2} \operatorname{sat}_{\left(\frac{s_{ll}}{k_{ll}}\right)^{\frac{3+2b}{2b}}} \left[|v_{ll}|^{\frac{3+3b}{2}} \right] \operatorname{sat}_{s_{ll}} \left[k_{ll} |v_{ll}|^{b} \right] \\ - \begin{bmatrix} \frac{r_{1}(1+b)}{2} \left(\frac{\varepsilon_{ll}}{k_{ll}}\right)^{\frac{3}{2}} \left[|v_{ll}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{\varepsilon_{ll}}{k_{ll}}\right)^{\frac{3}{2}b}} \left[|v_{ll}|^{\frac{3}{2}} \right] \right] \operatorname{sign}(v_{ll}) \\ \times \left(k_{al} \operatorname{sat}_{\mu_{l}} \left[\widetilde{e}_{2l} \right] - k_{al} \operatorname{sat}_{\left(\frac{\varepsilon_{2l}}{k_{ll}}\right)^{\frac{3+b}{2b}}} \left[v_{2l} + s_{l} + \operatorname{sat}_{\mu_{l}} \left[\widetilde{e}_{2l} \right] \right] \right] \\ = \begin{bmatrix} \frac{r_{1}(1+b)}{2} \left(\frac{\varepsilon_{ll}}{k_{ll}} \right)^{\frac{3}{2}} \left[|v_{ll}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{\varepsilon_{2l}}{k_{ll}}\right)^{\frac{3+b}{2b}}} \left[|v_{ll}|^{\frac{3}{2}} \right] \right] \operatorname{sign}(v_{ll}) \\ \times \operatorname{sat}_{\varepsilon_{2l}} \left[k_{2l} |v_{2l} + s_{l} + \operatorname{sat}_{\mu_{l}} \left[\widetilde{e}_{2l} \right]^{\frac{2b}{2b}} \left[|v_{ll}|^{\frac{3}{2}} \right] \right] \operatorname{sign}(v_{ll}) \\ - \frac{r_{1}(1+b)}{2} \left(\frac{\varepsilon_{ll}}{k_{ll}} \right)^{\frac{3}{2}} \left[|v_{ll}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{\varepsilon_{ll}}{k_{ll}}\right)^{\frac{3}{2b}}} \left[|v_{ll}|^{\frac{3}{2}} \right] \right] \operatorname{sign}(v_{2l} + s_{l} + \operatorname{sat}_{\mu_{l}} \left[\widetilde{e}_{2l} \right] \right] \right] \\ - \frac{r_{1}(1+b)}{2} \left(\frac{\varepsilon_{ll}}{k_{ll}} \right)^{\frac{3}{2}} \left[|v_{ll}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{\varepsilon_{ll}}{k_{ll}}\right)^{\frac{3}{2b}}} \left[|v_{ll}|^{\frac{3}{2}} \right] \right] \operatorname{sign}(v_{ll}) \left[\operatorname{sat}_{\varepsilon_{ll}} \left[k_{ll} |v_{ll}|^{b} \operatorname{sign}(v_{ll}) \right] \right] \\ - \left(\frac{1}{2} v_{2l}^{2} + \int_{0}^{v_{l}} \operatorname{sat}_{\varepsilon_{ll}} \left[k_{ll} |r|^{b} \operatorname{sign}(r) \right] dr \right] \times \left(v_{2l} \operatorname{sat}_{\varepsilon_{2l}} \left[k_{2l} |v_{2l} |^{\frac{2b}{1+b}} \operatorname{sign}(v_{2l}) \right] \right]$$

$$\begin{split} \vec{V}_{s} &= \begin{pmatrix} 2 \bigg(\frac{1}{2} v_{2i}^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} [k_{1i} | r|^{b} \operatorname{sign}(r)] dr \bigg) \\ &= \left(\begin{array}{c} \left(2 \bigg(\frac{1}{2} v_{2i}^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} [k_{1i} | r|^{b} \operatorname{sign}(r)] dr \bigg) \\ &= \left(\begin{array}{c} \left(\frac{2}{2} v_{2i}^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{2i}} [k_{2i} | v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}]] \\ &= v_{2i} k_{ai} \operatorname{sat}_{\varepsilon_{2i}} [k_{2i} | v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}]] + v_{2i} k_{ai} \operatorname{sat}_{v_{i}} [v_{2i}] \\ &+ v_{2i} \operatorname{sat}_{\varepsilon_{2i}} [k_{2i} | v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}]] + v_{2i} k_{ai} \operatorname{sat}_{v_{i}} [v_{2i}] \bigg) \\ &= \left(\begin{array}{c} v_{2i} \bigg(\frac{s_{2i}}{k_{2i}} \bigg) \bigg) v_{1i} \bigg|^{\frac{1}{2}} \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} \bigg|^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}]) \bigg) \\ &- v_{2i} \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} \bigg|^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i}) \bigg] - s_{i} \operatorname{sat}_{\varepsilon_{1i}} [k_{1i} | v_{1i} \bigg|^{b} \operatorname{sign}(v_{1i})] \\ &- v_{2i} \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} \bigg|^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i}) \bigg] - s_{i} \operatorname{sat}_{\varepsilon_{1i}} [k_{1i} | v_{1i} \bigg|^{b} \operatorname{sign}(v_{1i})] \\ &- v_{2i} \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} \bigg|^{\frac{2b}{1+b}} \operatorname{sign}(v_{1i}) \bigg) \\ &- v_{2i} \operatorname{sat}_{\varepsilon_{2i}} \left[v_{1i} \bigg|^{\frac{3}{2}} \operatorname{sat}_{\varepsilon_{2i}} \left[v_{1i} \bigg|^{\frac{3b}{2}} \right] v_{2i} \left[v_{1i} \bigg|^{\frac{3b}{2}} \right] v_{2i} \left[v_{1i} \bigg|^{\frac{3b}{2}} \right] v_{2i} \left[v_{1i} \bigg|^{\frac{3b}{2}} \right] \\ &- v_{2i} \operatorname{sat}_{\varepsilon_{2i}} \left[v_{2i} \bigg|^{\frac{3+3b}{2}} \right] \operatorname{sign}(v_{1i}) \\ &- \left(\left(\sum_{i=1}^{k} \left[\frac{s_{2i}}{2i} \right] \right) + k_{ai} \operatorname{sat}_{\varepsilon_{2i}} \left[\frac{s_{2i}}{k_{2i}} \right] \right] \\ &- k_{ai} \operatorname{sat}_{\varepsilon_{2i}} \left[v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} \left[\tilde{e}_{2i} \right] \right] \\ &+ \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} \left[\tilde{e}_{2i} \right] \right] \\ &- \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} \right|^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i}) \right] \\ &- \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} \right] \\ &+ v_{2i} \left[v_{2i} | v_{2i} v_{2i} v_{2i} \right] \\ &+ v_{2i} \left[v_{2i} v_{2i} v_{2i} v_{2i} \right] \\ &- v_{2i} \left[v_{2i} v_{2i$$

Note that the following property has been employed:

$$r_{1}\left(\frac{3+3b}{2}\right)_{0}^{\nu_{1i}}|r|^{\frac{1}{2}}\operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}}[|r|^{\frac{3b}{2}}] dr = \begin{pmatrix} r_{1}\operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3+3b}{2b}}}\left[|\nu_{1i}|^{\frac{3+3b}{2}}\right] \\ + r_{1}\left(1+b\right)\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}\left(|\nu_{1i}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}}\left[|\nu_{1i}|^{\frac{3}{2}}\right] \end{pmatrix} \operatorname{sign}(\nu_{1i})$$

Let us define the following terms \dot{V}_1 and \dot{V}_2 :

$$\dot{V}_{1} = \begin{pmatrix} -\left(\frac{1}{2}|v_{2i}|^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} [k_{1i}|r|^{b} \operatorname{sign}(r)] dr \right) \times \left(|v_{2i}| \operatorname{sat}_{\varepsilon_{2i}} [k_{2i}|v_{2i}|^{\frac{2b}{1+b}}] \right) \\ + r_{1} \left(\frac{3+3b}{2}\right) |v_{1i}|^{\frac{1}{2}} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} [|v_{1i}|^{\frac{3b}{2}}] |v_{2i}|^{2} \\ + \left(r_{1} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3+3b}{2b}}} \left[|v_{1i}|^{\frac{3+3b}{2}} \right] \right) \times \left(k_{ai}|v_{2i}| - k_{ai} \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} [|v_{2i}|] + \operatorname{sat}_{\varepsilon_{2i}} [k_{2i}|v_{2i}|^{\frac{2b}{1+b}}] \right) \\ - \frac{r_{1}}{2} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3+3b}{2b}}} \left[|v_{1i}|^{\frac{3+3b}{2}} \right] \operatorname{sat}_{\varepsilon_{1i}} [k_{1i}|v_{1i}|^{b}] \\ - \left(\frac{r_{1}(1+b)}{2}\right) \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}} \varepsilon_{3i} \left(|v_{1i}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}} [|v_{1i}|^{\frac{3}{2}}] \right)$$

$$\dot{V}_{2} = \begin{pmatrix} -\frac{r_{1}}{2} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3+3b}{2b}}} \left[|v_{1i}|^{\frac{3+3b}{2}} \right] \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |v_{1i}|^{b} \right] \\ - \left(\frac{r_{1}(1+b)}{2} \right) \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{3}{2}} \varepsilon_{3i} \left(|v_{1i}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}} \left[|v_{1i}|^{\frac{3}{2}} \right] \right) \\ - \left(\frac{1}{2} |v_{2i}|^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |r|^{b} \operatorname{sign}(r) \right] dr \right) \times \left(|v_{2i}| \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} |v_{2i}|^{\frac{2b}{1+b}} \right] \right) \\ - 2 \left(\frac{1}{2} |v_{2i}|^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |r|^{b} \operatorname{sign}(r) \right] dr \right) \times \left(|v_{2i}| k_{ai} |v_{2i}| - |v_{2i}| k_{ai} \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} \left[|v_{2i}| \right] \right) \end{pmatrix}$$

with

$$\varepsilon_{3i} := \begin{cases} \varepsilon_{1i} - k_{ai} \mu_i - k_{ai} \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}} - \varepsilon_{2i}, & \text{for } |\widetilde{e}_{2i}| \neq 0, \\ \\ \varepsilon_{1i} - k_{ai} \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}} - \varepsilon_{2i}, & \text{for } \widetilde{e}_{2i} = 0 \quad (\text{for the case of full state feedback}) \end{cases}$$

Note that $\dot{V}_{1s} \leq \dot{V}_1$ and $\dot{V}_{2s} \leq \dot{V}_2$, thus,

$$\dot{\widetilde{V}} \leq \dot{V_1} + \dot{V_2} + \dot{V_s}$$

Note that \dot{V}_2 is negative definite for $\varepsilon_{3i} > 0$. \dot{V}_s is considered as perturbation term. We are going to show that an $r_1 > 0$ exists such that the \dot{V}_1 is negative definite. Sign definiteness of \dot{V}_1 :

Let us divide the entire state space into three regions:

$$\Omega_{1} = \left\{ \left(v_{1i}, v_{2i} \right) \in \mathbb{R}^{2} : \left| v_{1i} \right| \in \mathbb{R}, \left| v_{2i} \right| \ge \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \right\}, \\ \Omega_{2} = \left\{ \left(v_{1i}, v_{2i} \right) \in \mathbb{R}^{2} : \left| v_{1i} \right| \ge \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{1}{b}}, \left| v_{2i} \right| \le \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \right\}, \\ \Omega_{3} = \left\{ \left(v_{1i}, v_{2i} \right) \in \mathbb{R}^{2} : v_{1i} \le \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{1}{b}}, v_{2i} \le \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \right\}$$

For region:
$$\Omega_1 = \left\{ \left(v_{1i}, v_{2i} \right) \in \mathbb{R}^2 : \left| v_{1i} \right| \in \mathbb{R}, \left| v_{2i} \right| \ge \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \right\}$$

Note that:

$$\int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |r|^{b} \operatorname{sign}(r) \right] dr = \frac{k_{1i}}{1+b} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}} \left[|v_{1i}|^{1+b} \right] + \varepsilon_{1i} \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left(|v_{1i}| \right) \right)$$

and

$$|v_{2i}|\operatorname{sat}_{\varepsilon_{2i}}\left[k_{2i}|v_{2i}|^{\frac{2b}{1+b}}\right] = \varepsilon_{2i}\left(|v_{2i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}}\left[|v_{2i}|\right]\right) + \operatorname{sat}_{\varepsilon_{2i}}\left[k_{2i}|v_{2i}|^{\frac{2b}{1+b}}\right]\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}}\left[|v_{2i}|\right]$$

and

$$\operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3+3b}{2b}}}\left[\left|v_{1i}\right|^{\frac{3+3b}{2}}\right] = \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}}\left[\left|v_{1i}\right|^{1+b}\right]\operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{2b}}}\left[\left|v_{1i}\right|^{\frac{1+b}{2}}\right] \le \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}}\left[\left|v_{1i}\right|^{1+b}\right]\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{2b}}$$

and using lemma 2 of chapter 2, we have:

$$\begin{aligned} |v_{1i}|^{\frac{1}{2}} &= \left(\left| v_{1i} \right| - \operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left(\left| v_{1i} \right| \right) + \operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left(\left| v_{1i} \right| \right) \right)^{\frac{1}{2}} &\leq \left(\left| v_{1i} \right| - \operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left(\left| v_{1i} \right| \right) \right)^{\frac{1}{2}} + \left(\operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left(\left| v_{1i} \right| \right) \right)^{\frac{1}{2}} \\ &\leq \left(\left| v_{1i} \right| - \operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left(\left| v_{1i} \right| \right) \right)^{\frac{1}{2}} + \left(\frac{\mathcal{E}_{1i}}{k_{1i}} \right)^{\frac{1}{b}} \right)^{\frac{1}{2}} \end{aligned}$$

Then,
$$\begin{split} \dot{V_{1}} &\leq -\left(\frac{1}{2}|v_{2l}| - \varepsilon_{1l}^{\frac{1}{2}} \left(\left|v_{1l}\right| - \operatorname{sat}_{\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1}{b}}}\left(v_{1l}|\right)\right)^{\frac{1}{2}}\right)^{2} \times \left(\left|v_{2l}\right| \operatorname{sat}_{\varepsilon_{2l}}\left[k_{2l}\left|v_{2l}\right|\right|^{\frac{2b}{1+b}}\right]\right) \\ &- \left(\left|v_{1l}\right| - \operatorname{sat}_{\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1}{b}}}\left(v_{1l}\right|\right)\right)^{\frac{1}{2}} \left|v_{2l}\right|^{2} \left(\varepsilon_{1l}^{\frac{1}{2}} \operatorname{sat}_{\varepsilon_{2l}}\left[k_{2l}\left|v_{2l}\right|^{\frac{2b}{1+b}}\right] - r_{1}\left(\frac{3+3b}{2}\right)\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{3}{2}}\right) \\ &- \left|v_{2l}\right|^{2} \left(\frac{1}{4}|v_{2l}| \operatorname{sat}_{\varepsilon_{2l}}\left[k_{2l}\left|v_{2l}\right|^{\frac{2b}{1+b}}\right] - r_{1}\left(\frac{3+3b}{2}\right)\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+3b}{2b}}\right) \\ &- \operatorname{sat}_{\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+b}{b}}}\left[\left|v_{1l}\right|^{1+b}\right] \left(\left|v_{2l}\right| - \operatorname{sat}_{\left(\frac{\varepsilon_{2l}}{k_{2l}}\right)^{\frac{1+b}{2b}}\left[\left|v_{2l}\right|\right]\right)\left(\frac{k_{1l}}{1+b}\varepsilon_{2l} - r_{1}\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+b}{2b}}k_{al}\right) \\ &- \operatorname{sat}_{\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+b}{b}}}\left[\left|v_{1l}\right|^{1+b}\right] \operatorname{sat}_{\varepsilon_{2l}}\left[k_{2l}\left|v_{2l}\right|^{\frac{2b}{1+b}}\right]\left(\frac{k_{1l}}{1+b}\varepsilon_{2l} - r_{1}\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+b}{2b}}k_{al}\right) \\ &- \operatorname{sat}_{\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+b}{b}}}\left[\left|v_{1l}\right|^{1+b}\right] \operatorname{sat}_{\varepsilon_{2l}}\left[k_{2l}\left|v_{2l}\right|^{\frac{2b}{1+b}}\right]\left(\frac{k_{1l}}{1+b}\varepsilon_{2l} - r_{1}\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+b}{2b}}k_{al}\right) \\ &- \operatorname{sat}_{\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+b}{b}}}\left[\left|v_{1l}\right|^{1+b}\right] \operatorname{sat}_{\varepsilon_{2l}}\left[k_{2l}\left|v_{2l}\right|^{\frac{2b}{1+b}}\right] \left(\frac{k_{1l}}{1+b}\varepsilon_{2l}\left|v_{2l}\right|^{1+b}}\left[\left|v_{2l}\right|\right] - r_{1}\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+b}{2b}}\right) \\ &- \operatorname{sat}_{\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+b}{b}}}\left[\left|v_{2l}\right|^{1+b}\right] \operatorname{sat}_{\varepsilon_{2l}}\left[k_{2l}\left|v_{2l}\right|^{\frac{2b}{1+b}}\right] \left(\frac{k_{1l}}{1+b}\varepsilon_{2l}\left|v_{2l}\right|^{1+b}\right] - r_{1}\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+b}{2b}}\right) \\ &- \operatorname{sat}_{\left(\frac{\varepsilon_{1l}}{k_{1l}}\right)^{\frac{1+b}{b}}}\left[\left|v_{2l}\right|^{1+b}\right] \operatorname{sat}_{\varepsilon_{2l}}\left[\frac{\varepsilon_{2l}}{1+b}\right] \left(\frac{\varepsilon_{2l}}{1+b}\right)^{\frac{1+b}{2b}}\left[\frac{\varepsilon_{2l}}{1+b}\right] \left(\frac{\varepsilon_{2l}}{1+b}\right)^{\frac{1+b}{2b}}\left[\frac{\varepsilon_{2l}}{1+b}\right] \\ &- \operatorname{sat}_{\varepsilon_{2l}}\left[\frac{\varepsilon_{2l}}{1+b}\right] \left(\frac{\varepsilon_{2l}}{1+b}\right] \left(\frac{\varepsilon_{2l}}{1+b}\right)^{\frac{1+b}{2b}}\left[\frac{\varepsilon_{2l}}{1+b}\right] \left(\frac{\varepsilon_{2l}}{1+b}\right)^{\frac{1+b}{2b}}\left[\frac{\varepsilon_{2l}}{1+b}\right] \left(\frac{\varepsilon_{2l}}{1+b}\right)^{\frac{1+b}{2b}}\left[\frac{\varepsilon_{2l}}{1+b}\right] \\ &- \operatorname{sat}_{\varepsilon_{2l}}\left[\frac{\varepsilon_{2l}}{1+b}\right] \left(\frac{\varepsilon_{2l}}{$$

Thus, $\dot{V}_1 < 0$ for:

$$\min\left\{ \left(\frac{2}{3+3b}\right) k_{1i}^{\frac{3}{2}} \left(\frac{\varepsilon_{2i}}{\varepsilon_{1i}}\right), \quad \left(\frac{1}{6+6b}\right) \left(\frac{k_{1i}^{\frac{1+3b}{2b}}}{k_{2i}^{\frac{1+b}{2b}}}\right) \left(\frac{\varepsilon_{2i}}{\varepsilon_{1i}}\right)^{\frac{1+3b}{2b}}, \\ \left(\frac{k_{1i}^{\frac{1+3b}{2b}}}{k_{ai}(1+b)}\right) \left(\frac{\varepsilon_{2i}}{\varepsilon_{1i}^{\frac{1+b}{2b}}}\right), \quad \left(\frac{1}{1+b}\right) \left(\frac{k_{1i}^{\frac{1+3b}{2b}}}{k_{2i}^{\frac{1+b}{2b}}}\right) \left(\frac{\varepsilon_{2i}}{\varepsilon_{1i}}\right)^{\frac{1+b}{2b}} \right\} > r_{1}$$

For region:
$$\Omega_2 = \left\{ \left(v_{1i}, v_{2i} \right) \in \mathbb{R}^2 : \left| v_{1i} \right| \ge \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{1}{b}}, \left| v_{2i} \right| \le \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \right\}$$

Note that in this region the term $|v_{1i}|$ is lower bounded, thus

$$\int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |r|^{b} \operatorname{sign}(r) \right] dr = \frac{k_{1i}}{1+b} \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{1+b}{b}} + \varepsilon_{1i} \left(|v_{1i}| - \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{1}{b}} \right)$$

also in this region the term $|v_{2i}|$ is upper bounded, thus we have,

$$k_{ai} |v_{2i}| - k_{ai} \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} \left[|v_{2i}| \right] = 0 \text{ and}$$

$$\begin{aligned} |v_{2i}|^{2} &= \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{b}}} \left\| v_{2i} \right\|^{2} \right] &= \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+3b}{3b}}} \left[|v_{2i}|^{\frac{2+6b}{3+3b}} \right] \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{2}{3b}}} \left[|v_{2i}|^{\frac{4}{3+3b}} \right] \\ &\leq \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+3b}{3b}}} \left[|v_{2i}|^{\frac{2+6b}{3+3b}} \right] \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{2}{3b}} \end{aligned}$$

or

$$\begin{aligned} \left| v_{2i} \right|^{2} &= \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{b}}} \left[\left| v_{2i} \right|^{2} \right] &= \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)} \left[\left| v_{2i} \right|^{\frac{2b}{1+b}} \right] \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1}{b}}} \left[\left| v_{2i} \right|^{\frac{2}{1+b}} \right] \\ &\leq \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)} \left[\left| v_{2i} \right|^{\frac{2b}{1+b}} \right] \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1}{b}} \end{aligned}$$

and using lemma 2.2:

$$|v_{1i}|^{\frac{1}{2}} = \left(|v_{1i}|^{\frac{3}{2}} - \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}} + \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}} \right)^{\frac{1}{3}} \le \left(|v_{1i}|^{\frac{3}{2}} - \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}} \right)^{\frac{1}{3}} + \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{2b}}$$

Then,

$$\begin{split} \dot{V_{1}} &\leq -\varepsilon_{1i} \Biggl(\left| |v_{1i}| - \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}} \Biggr) \text{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}} \varepsilon_{2i}} \Biggl[k_{2i} |v_{2i}|^{\frac{1+3b}{1+b}} \Biggr] \\ &\quad - \frac{r_{1}(1+b)}{2} \Biggl(\frac{\varepsilon_{1i}}{k_{1i}} \Biggr)^{\frac{3}{2}} \varepsilon_{3i} \Biggl(\left| |v_{1i}| \right|^{\frac{3}{2}} - \Biggl(\frac{\varepsilon_{1i}}{k_{1i}} \Biggr)^{\frac{3}{2b}} \Biggr) - \Biggl(\frac{k_{1i}}{1+b} \Biggr) \Biggl(\frac{\varepsilon_{1i}}{k_{1i}} \Biggr)^{\frac{1+b}{b}} k_{2i} \text{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+3b}{2b}}} \Biggl[|v_{2i}|^{\frac{1+3b}{1+b}} \Biggr] \\ &\quad + r_{1} \Biggl(\frac{3+3b}{2} \Biggr) \Biggl(\frac{\varepsilon_{1i}}{k_{1i}} \Biggr)^{\frac{3}{2}} \Biggl(\frac{\varepsilon_{2i}}{k_{2i}} \Biggr)^{\frac{2}{3b}} \Biggl(\left| |v_{1i}|^{\frac{3}{2}} - \Biggl(\frac{\varepsilon_{1i}}{k_{1i}} \Biggr)^{\frac{3}{2b}} \Biggr)^{\frac{1}{3}} \text{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+3b}{2b}}} \Biggl[|v_{2i}|^{\frac{1+3b}{1+b}} \Biggr] \\ &\quad - \frac{r_{1}}{4} \Biggl(\frac{\varepsilon_{1i}}{k_{1i}} \Biggr)^{\frac{3}{2}} \Biggl(\frac{\varepsilon_{2i}}{k_{2i}} \Biggr)^{\frac{2}{3b}} \Biggr) - \frac{1}{2} k_{2i} \text{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+3b}{3b}}} \Biggl[|v_{2i}|^{\frac{2+6b}{3+3b}} \Biggr] \\ &\quad + r_{1} \Biggl(\frac{\varepsilon_{1i}}{k_{1i}} \Biggr)^{\frac{3+3b}{2b}} \Biggr) - \frac{r_{1}}{4} \Biggl(\frac{\varepsilon_{1i}}{k_{1i}} \Biggr)^{\frac{3+3b}{2b}} \Biggr) - \frac{1}{2} k_{2i} \text{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{3+3b}{2b}}} \Biggl[|v_{2i}|^{\frac{3+3b}{1+b}} \Biggr] \\ &\quad + r_{1} \Biggl(\frac{\varepsilon_{1i}}{k_{1i}} \Biggr)^{\frac{1+3b}{2b}} \Biggl(\Biggl(\frac{3+3b}{2} \Biggr) \Biggl(\frac{\varepsilon_{2i}}{k_{2i}} \Biggr)^{\frac{1}{b}} + \Biggl(\frac{\varepsilon_{1i}}{k_{1i}} \Biggr)^{\frac{1}{b}} k_{2i} \Biggr] \text{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)} \Biggl[|v_{2i}|^{\frac{2b}{1+b}} \Biggr]$$

From lemma 2.1,

$$-\frac{r_{1}(1+b)}{2}\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}\varepsilon_{3i}\left(|v_{1i}|^{\frac{3}{2}}-\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}\right)-\left(\frac{k_{1i}}{1+b}\right)\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}k_{2i}\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+3b}{2b}}}\left[|v_{2i}|^{\frac{1+3b}{1+b}}\right]$$

$$\leq -r_{1}^{\frac{1}{3}}\left(\frac{3+3b}{2}\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}\varepsilon_{3i}\left(|v_{1i}|^{\frac{3}{2}}-\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}\right)\right)^{\frac{1}{3}}\left(\frac{3k_{1i}k_{2i}}{2+2b}\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+3b}{2b}}}\left[|v_{2i}|^{\frac{1+3b}{1+b}}\right]\right)^{\frac{3}{3}}\right)^{\frac{3}{3}}$$

and

$$-\frac{r_{1}}{4}\left(\frac{\varepsilon_{1i}^{\frac{3+5b}{2b}}}{k_{1i}^{\frac{3+3b}{2b}}}\right) - \frac{1}{2}k_{2i}\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{3+5b}{2b}}}\left[\left|v_{2i}\right|^{\frac{3+5b}{1+b}}\right]$$

$$\leq -r_{1}^{\frac{3+3b}{3+5b}}\left(\left(\frac{3+5b}{3+3b}\right)\left(\frac{1}{4}\right)\left(\frac{\varepsilon_{1i}^{\frac{3+5b}{2b}}}{k_{1i}^{\frac{3+3b}{2b}}}\right)\right)^{\frac{3+3b}{3+5b}}\left(\left(\frac{3+5b}{4b}\right)k_{2i}\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{3+5b}{2b}}}\left[\left|v_{2i}\right|^{\frac{3+5b}{1+b}}\right]\right)^{\frac{2b}{3+5b}}$$

Thus, $\dot{V}_1 < 0$ in this region for:

$$r_{1}^{\frac{1}{3}}\left(\left(\frac{3+3b}{2}\right)\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}\varepsilon_{3i}\left(\left|v_{1i}\right|^{\frac{3}{2}}-\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}\right)\right)^{\frac{1}{3}}\left(\frac{3k_{1i}k_{2i}}{2+2b}\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+3b}{2b}}}\left[\left|v_{2i}\right|^{\frac{1+3b}{1+b}}\right]\right)^{\frac{2}{3}}$$

$$>r_{1}\left(\frac{3+3b}{2}\right)\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{3}{2b}}\left(\left|v_{1i}\right|^{\frac{3}{2}}-\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}\right)^{\frac{1}{3}}\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+3b}{3b}}}\left[\left|v_{2i}\right|^{\frac{2+6b}{3+3b}}\right]$$

$$\Leftrightarrow \varepsilon_{3i}^{\frac{1}{2}} \left(\frac{2}{3+3b}\right) \left(\frac{3k_{1i}k_{2i}}{2+2b}\right) \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}} \left(\frac{k_{1i}}{\varepsilon_{1i}}\right)^{\frac{3}{2}} \left(\frac{k_{2i}}{\varepsilon_{2i}}\right)^{\frac{1}{b}} > r_1$$

Also,

$$r_{1}^{\frac{3+3b}{3+5b}}\left(\left(\frac{3+5b}{3+3b}\right)\left(\frac{1}{4}\right)\left(\frac{\varepsilon_{1i}\frac{3+5b}{2b}}{k_{1i}\frac{3+3b}{2b}}\right)\right)^{\frac{3+3b}{3+5b}}\left(\left(\frac{3+5b}{4b}\right)k_{2i}\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{3+5b}{2b}}}\left[\left|v_{2i}\right|^{\frac{3+5b}{1+b}}\right]\right)^{\frac{2b}{3+5b}}$$

$$> r_{1}\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+3b}{2b}}\left(\left(\frac{3+3b}{2}\right)\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1}{b}} + \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}k_{2i}\right)\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)}\left[\left|v_{2i}\right|^{\frac{2b}{1+b}}\right]$$

$$\Leftrightarrow \frac{\left(\frac{3+5b}{12+12b}\right)^{\frac{3+3b}{2b}} \left(\frac{3+5b}{4b}\right) \left(\frac{\varepsilon_{1i}}{k_{1i}}^{\frac{3+5b}{2b}}}{k_{1i}}\right)^{\frac{3+3b}{2b}} \left(\left(\frac{k_{1i}}{\varepsilon_{1i}}\right)^{\frac{1+3b}{2b}}\right)^{\frac{3+5b}{2b}} k_{2i}}{\left(\left(\frac{3+3b}{2}\right) \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1}{b}} + \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}} k_{2i}}\right)^{\frac{3+5b}{2b}}} > r_{1}}$$
For region: $\Omega_{3} = \left\{ \left(v_{1i}, v_{2i}\right) \in \mathbb{R}^{2} : v_{1i} \le \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}, v_{2i} \le \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}\right\}$

Note that in this region:

$$\begin{split} \dot{V_1} &= -\left(\frac{1}{2} \left| v_{2i} \right|^2 + \frac{k_{1i}}{1+b} \left| v_{1i} \right|^{1+b} \right) \times \left(k_{2i} \left| v_{2i} \right|^{\frac{1+3b}{1+b}} \right) \\ &+ r_1 \left(\frac{3+3b}{2} \right) \left| v_{1i} \right|^{\frac{1+3b}{2}} \left| v_{2i} \right|^2 + r_1 \left| v_{1i} \right|^{\frac{3+3b}{2}} k_{2i} \left| v_{2i} \right|^{\frac{2b}{1+b}} - \frac{r_1}{2} k_{1i} \left| v_{1i} \right|^{\frac{3+5b}{2}} \end{split}$$

It can be easily verified that, for any c > 0 and $(v_{1i}, v_{2i}) \in \Omega_3$:

$$\dot{V}_{1}(c^{\frac{2}{1+b}}v_{1i}, cv_{2i}) = c^{\frac{3+5b}{1+b}}\dot{V}_{1}(v_{1i}, v_{2i})$$
(B1 - 1)

Let us define the boundary of this region, bd Ω_3 , which encircles the origin that can be written as the union of the sets

bd $\Omega_3 = \Omega_{3,1} \bigcup \Omega_{3,2}$ where the sets

$$\Omega_{3,1} = \left\{ \left(v_{1i}, v_{2i} \right) : \left| v_{1i} \right| \le \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{1}{b}}, \left| v_{2i} \right| = \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \right\},\$$
$$\Omega_{3,2} = \left\{ \left(v_{1i}, v_{2i} \right) : \left| v_{1i} \right| = \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{1}{b}}, \left| v_{2i} \right| \le \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \right\}.$$

Due to this homogeneity property (B1 - 1), to show sign definiteness, it suffices to prove sign definiteness on the set bd Ω_3 , which encircles the origin, since for every $(v_{1i}, v_{2i}) \in \Omega_3 \setminus \{(0,0)\}$ there exists a c > 0 such that $(c^{2/(1+b)} v_{1i}, cv_{2i}) \in bd \Omega_3$. Note that the sign definiteness of $\dot{V_1}$ has been shown to be negative definite on bd Ω_3 already since $\Omega_{3,1} \in \Omega_1$ and $\Omega_{3,2} \in \Omega_2$. Thus, by the homogeneity property (B1 - 1), we have $\dot{V_1} < 0$ in this region as well.

As a result, $\dot{V}_1 < 0$, negative definite for all $(v_{1i}, v_{2i}) \in \mathbb{R}^2$ with $r_1 > 0$ chosen as:

$$\min \left\{ \frac{\left(\frac{2}{3+3b}\right)k_{1i}^{\frac{3}{2}}\left(\frac{\varepsilon_{2i}}{\varepsilon_{1i}}\right)}{\left(\frac{k_{1i}}{\varepsilon_{1i}}^{\frac{1+3b}{2b}}\right)} \left(\frac{1}{6+6b}\right)\left(\frac{k_{1i}}{\varepsilon_{2i}}^{\frac{1+3b}{2b}}}{k_{2i}}^{\frac{1+3b}{2b}}\right)\left(\frac{\varepsilon_{2i}}{\varepsilon_{1i}}\right)^{\frac{1+3b}{2b}}, \\ \left(\frac{k_{1i}}{\varepsilon_{2i}}^{\frac{1+3b}{2b}}}{k_{ai}(1+b)}\right)\left(\frac{\varepsilon_{2i}}{\varepsilon_{1i}}^{\frac{1+b}{2b}}\right), \quad \left(\frac{1}{1+b}\right)\left(\frac{k_{1i}}{\varepsilon_{2i}}^{\frac{1+3b}{2b}}}{k_{2i}}\right)\left(\frac{\varepsilon_{2i}}{\varepsilon_{1i}}\right)^{\frac{1+b}{2b}}, \\ \left\{\varepsilon_{3i}^{\frac{1}{2}}\left(\frac{2}{3+3b}\right)\left(\frac{3k_{1i}k_{2i}}{2+2b}\right)\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}\left(\frac{k_{1i}}{\varepsilon_{1i}}\right)^{\frac{3}{2}}\left(\frac{k_{2i}}{\varepsilon_{2i}}\right)^{\frac{1}{b}}, \\ \left(\frac{3+5b}{12+12b}\right)^{\frac{3+3b}{2b}}\left(\frac{3+5b}{4b}\right)\left(\frac{\varepsilon_{1i}}{k_{1i}}^{\frac{3+5b}{2b}}\right)^{\frac{3+3b}{2b}}\left(\left(\frac{k_{1i}}{\varepsilon_{1i}}\right)^{\frac{1+3b}{2b}}\right)^{\frac{3+5b}{2b}}k_{2i}} \\ \left(\left(\frac{3+3b}{2}\right)\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1}{b}} + \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}k_{2i}\right)^{\frac{3+5b}{2b}}\right) \\ \end{array}\right\}$$

Observe that such an $r_1 > 0$, always exists for any $k_{1i} > 0$, $k_{2i} > 0$, $\varepsilon_{1i} > 0$, $\varepsilon_{2i} > 0$, and b > 0.

Sign definiteness of $\dot{V}_2 + \dot{V}_s$:

From the previous development, we have shown that \dot{V}_1 is negative definite and since \dot{V}_2 is negative definite, the idea is to dominate the \dot{V}_s term with it. Before we proceed, from Appendix B.2-proposition 1, we have the following inequalities

$$\begin{vmatrix} \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} [v_{2i}] - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} [v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]] \end{vmatrix} \leq \operatorname{sat}_{2\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} [s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]] \end{vmatrix}$$
$$\begin{vmatrix} \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}] \right]^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]) \right] \\ -\operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} | \frac{2b}{1+b} \operatorname{sign}(v_{2i}) \right] \end{vmatrix} \leq \operatorname{sat}_{2\varepsilon_{2i}} \left[\chi(|s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]) \right] \\ \leq \operatorname{sat}_{2\varepsilon_{2i}} \left[\beta(|s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]) \right] \end{vmatrix}$$

where β is a class \mathcal{K} function. (Refer to the definition of class \mathcal{K} function in [71]). Hence,

after simple algebraic manipulation:

$$\begin{split} \dot{\mathcal{V}}_{2} + \dot{\mathcal{V}}_{s} &\leq -\frac{r}{2} \operatorname{sat}_{\left(\frac{\bar{v}_{11}}{\bar{k}_{11}}\right)^{\frac{1+3}{2}}} \left[\left| v_{1i} \right|^{\frac{3+3b}{2}} \right] \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} | v_{1i} \right|^{\frac{5}{2}} - \operatorname{sat}_{\left(\frac{\bar{v}_{11}}{\bar{k}_{11}}\right)^{\frac{3}{2}}} \left[\left| v_{1i} \right|^{\frac{3}{2}} \right] \right] \\ &- \left(\frac{1}{2} | v_{2i} |^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} | r |^{b} \operatorname{sign}(r) \right] dr \right] \times \left(| v_{2i} | \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} \right|^{\frac{2b}{1+b}} \right] \right) \\ &- 2 \left(\frac{1}{2} | v_{2i} |^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} | r |^{b} \operatorname{sign}(r) \right] dr \right) \times \left(| v_{2i} | k_{ai} | v_{2i} | - | v_{2i} | k_{ai} \operatorname{sat}_{\left(\frac{\bar{v}_{2i}}{\bar{k}_{2i}}\right)^{\frac{1+b}{2b}}} \left[| v_{2i} | \right] \right) \\ &+ \left(2 \left(\frac{1}{2} | v_{2i} |^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} | r |^{b} \operatorname{sign}(r) \right] dr \right) \times \left(| v_{2i} | k_{ai} | v_{2i} | - | v_{2i} | k_{ai} \operatorname{sat}_{\left(\frac{\bar{v}_{2i}}{\bar{k}_{2i}}\right)^{\frac{1+b}{2b}}} \left[| v_{2i} | \right] \right) \\ &+ \left(2 \left(\frac{1}{2} | v_{2i} |^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} | r |^{b} \operatorname{sign}(r) \right] dr \right) \\ &+ \left(2 \left(\frac{1}{2} | v_{2i} |^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[\tilde{v}_{2i} | + | v_{2i} | k_{ai} \operatorname{sat}_{\left(\frac{\bar{v}_{2i}}{\bar{k}_{2i}}\right)^{\frac{1+b}{2b}}} \left[\left| v_{2i} | \frac{1}{\bar{k}_{2i}} \right| \right] \right) \\ &+ 2 \left(\frac{1}{2} | v_{2i} |^{2} + \int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[\tilde{v}_{1i} | r |^{b} \operatorname{sign}(r) \right] dr \right) \varepsilon_{1i} | s_{1i} | \\ &+ r_{i} \left(\frac{3+3b}{2} \right) v_{1i} |^{\frac{1}{2}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} | r |^{b} \operatorname{sign}(r) \right] dr \right) \varepsilon_{1i} | s_{1i} | \\ &+ r_{i} \left(\frac{3+3b}{2} \right) v_{1i} |^{\frac{1}{2}} \operatorname{sat}_{\varepsilon_{1i}} \left[\tilde{v}_{1i} | \frac{3^{b}}{2} \right] | v_{2i} | | s_{1i} | \\ &+ \left(r_{1i} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{\bar{k}_{1i}}\right)^{\frac{1+b}{2}} \left[v_{1i} | \frac{3+3b}{2} \right] \\ &+ \left(\kappa_{ai} | s_{i} + \operatorname{sat}_{ii} \left[\tilde{v}_{2i} \right] \right] \right) \right) \right) \right) \\ \end{array}$$

The $z_{1i} - z_{2i}$ plane is divided into the following three regions for analysis:

$$\begin{split} \Psi_{1i}(|\zeta_{i}|) &= \{ (v_{1i}, v_{2i}) \in \mathbb{R}^{2} : |v_{1i}| \in \mathbb{R}, |v_{2i}| \geq \alpha_{2}(|\zeta_{i}|) \}, \\ \Psi_{2i}(|\zeta_{i}|) &= \{ (v_{1i}, v_{2i}) \in \mathbb{R}^{2} : |v_{1i}| \geq \alpha_{1}(|\zeta_{i}|), |v_{2i}| \leq \alpha_{2}(|\zeta_{i}|) \}, \\ \Psi_{3i}(|\zeta_{i}|) &= \{ (v_{1i}, v_{2i}) \in \mathbb{R}^{2} : |v_{1i}| \leq \alpha_{1}(|\zeta_{i}|), |v_{2i}| \leq \alpha_{2}(|\zeta_{i}|) \}, \end{split}$$

where $|\zeta_i|$ is defined as $|\zeta_i| = \max \{ |s_i|, \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}|] \}$,

while $\alpha_1(|\zeta_i|)$ and $\alpha_2(|\zeta_i|)$ are class \mathcal{K} functions defined by

$$\alpha_{1}(\boldsymbol{\zeta}_{i}|) \coloneqq \max \begin{cases} \left(\frac{\operatorname{sat}}{\left(\frac{\delta_{i_{l}}}{k_{i_{l}}}\right)^{\frac{1+2\delta}{2b}} \varepsilon_{i_{l}}} \left[\alpha_{7}(\boldsymbol{\zeta}_{i}|)\right]}{k_{1i}} \right)^{\frac{2}{1+5b}} + \left(\frac{\alpha_{7}(\boldsymbol{\zeta}_{i}|)}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \varepsilon_{1i}} \right)^{2} - \left(\frac{\operatorname{sat}}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{1+2\delta}{2b}}} \left[\alpha_{7}(\boldsymbol{\zeta}_{i}|)\right]}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \varepsilon_{1i}} \right)^{2} \\ \left(\frac{\operatorname{sat}}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{2+3\delta}{2b}}} \left[\alpha_{8}(\boldsymbol{\zeta}_{i}|)\right]}{k_{1i}} \right)^{\frac{2}{2+5b}} + \left(\frac{\alpha_{8}(\boldsymbol{\zeta}_{i}|) - \operatorname{sat}}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{2+3\delta}{2b}}} \left[\alpha_{8}(\boldsymbol{\zeta}_{i}|)\right]}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{2}{2}}} \varepsilon_{1i}} \right) \\ \left(\frac{\operatorname{sat}}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{9}(\boldsymbol{\zeta}_{i}|)\right]}{k_{1i}} \right)^{\frac{2}{3+5b}} + \left(\frac{\alpha_{9}(\boldsymbol{\zeta}_{i}|)}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \varepsilon_{1i}} \right)^{2} - \left(\frac{\operatorname{sat}}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{9}(\boldsymbol{\zeta}_{i}|)\right]}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \varepsilon_{1i}} \right)^{2} \\ \left(\frac{\operatorname{sat}}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{9}(\boldsymbol{\zeta}_{i}|)\right]}{k_{1i}} \right)^{\frac{2}{3+5b}}} + \left(\frac{\alpha_{9}(\boldsymbol{\zeta}_{i}|)}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \varepsilon_{1i}} \left(\frac{\delta_{1i}}{\delta_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{9}(\boldsymbol{\zeta}_{i}|)\right]}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3}{2}}} \varepsilon_{1i}} \right)^{2} \\ \left(\frac{\operatorname{sat}}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{1i}}\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{1i}}\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{1i}}\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \varepsilon_{1i}} \right)^{2} \\ \left(\frac{\delta_{1i}}{\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{1i}}\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{1i}}\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{1i}}\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}} \left[\alpha_{1i}}\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}} \left[\alpha_{1i}}\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{1i}}\left(\frac{\delta_{1i}}{k_{1i}}\right)^{\frac{3+3\delta}{2b}}} \left[\alpha_{1i}}\left(\frac{\delta_{$$

$$\begin{split} & \left(\frac{\left(\frac{\operatorname{sat}_{z_{2i}} \left[\delta \, \alpha_{3}(2|\zeta_{i}|) \right]}{k_{2i}} \right)^{\frac{1+b}{2b}} + \left(\frac{\delta \, \alpha_{3}(2|\zeta_{i}|) - \operatorname{sat}_{\varepsilon_{2i}} \left[\delta \, \alpha_{3}(2|\zeta_{i}|) \right]}{6k_{ai}} \right), \\ & \left(\frac{\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{k_{2i}} \left[\delta \, \varepsilon_{1i} |\zeta_{i}| \right]}{k_{2i}} \right)^{\frac{1+b}{1+3b}} + \left(\frac{\delta \, \varepsilon_{1i} |\zeta_{i}| - \operatorname{sat}_{\varepsilon_{2i}} \left[\frac{1+b}{2b} \varepsilon_{2i} \right]}{\varepsilon_{2i}} \left[\delta \, \varepsilon_{1i} |\zeta_{i}| \right]} \right), \\ & \left(\frac{\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{k_{2i}} \left[\alpha_{4}(|\zeta_{i}|) \right]}{k_{2i}} \right)^{\frac{1+b}{1+3b}} + \left(\frac{\alpha_{4}(|\zeta_{i}|) - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{\varepsilon_{2i}} \left[\alpha_{4}(|\zeta_{i}|) \right]} \right), \\ & \left(\frac{\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{k_{2i}} \left[\alpha_{5}(|\zeta_{i}|) \right]} \right)^{\frac{1+b}{2+4b}} + \left(\frac{\alpha_{5}(|\zeta_{i}|)}{\varepsilon_{2i}} \right)^{\frac{1}{2}} - \left(\frac{\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{\varepsilon_{2i}} \right), \\ & \left(\frac{\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{k_{2i}} \left[\alpha_{6}(2|\zeta_{i}|) \right]} \right)^{\frac{1+b}{2+4b}} + \left(\frac{\alpha_{6}(2|\zeta_{i}|) - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{\varepsilon_{2i}} \left[\alpha_{6}(2|\zeta_{i}|) \right]} \right)^{\frac{1}{2}}, \\ & \left(\frac{\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{k_{2i}} \left[\alpha_{6}(2|\zeta_{i}|) \right]} \right)^{\frac{1+b}{1+3b}} + \left(\frac{\alpha_{6}(2|\zeta_{i}|) - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{\varepsilon_{2i}} \left[\alpha_{6}(2|\zeta_{i}|) \right]} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}, \\ & \left(\frac{\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{k_{2i}} \left[\alpha_{6}(2|\zeta_{i}|) \right]} \right)^{\frac{1+b}{1+3b}} + \left(\frac{\alpha_{6}(2|\zeta_{i}|) - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{\varepsilon_{2i}} \left[\alpha_{6}(2|\zeta_{i}|) \right]} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}, \\ & \left(\frac{\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{\varepsilon_{2i}} \left[\alpha_{6}(2|\zeta_{i}|) \right]} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}, \\ & \left(\frac{\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{\varepsilon_{2i}} \left[\alpha_{6}(2|\zeta_{i}|) \right]} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}, \\ & \left(\frac{\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{\varepsilon_{2i}} \left[\alpha_{6}(2|\zeta_{i}|) \right]} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}, \\ & \left(\frac{\operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \varepsilon_{2i}}}{\varepsilon_{2i}} \left[\alpha_{6}(2|\zeta_{i}|) \right]} \right)^{\frac{1+b}{2b$$

with the additional class \mathcal{K} functions defined as:

$$\begin{aligned} &\alpha_{3}(2|\zeta_{i}|) \coloneqq k_{ai}(2|\zeta_{i}|) + k_{ai} \operatorname{sat}_{2\left(\frac{\varphi_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} \left[2|\zeta_{i}|\right] + \operatorname{sat}_{2\varepsilon_{2i}}\left[\beta(2|\zeta_{i}|)\right] \\ &\alpha_{4}(|\zeta_{i}|) \coloneqq r_{1}\varepsilon_{1i}\left(\frac{9+9b}{2}\right)\left(\frac{1}{k_{1i}}\right)^{\frac{3}{2}}|\zeta_{i}| \\ &\alpha_{5}(|\zeta_{i}|) \coloneqq r_{1}(18+18b)\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{2b}}|\zeta_{i}| \\ &\alpha_{6}(2|\zeta_{i}|) \coloneqq r_{1}\left(\frac{3+3b}{k_{1i}}\right)\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{2b}}\alpha_{3}(2|\zeta_{i}|) \\ &\alpha_{7}(|\zeta_{i}|) \coloneqq \left(\frac{12}{r_{1}}\right)\left(\frac{\varepsilon_{1i}}{\varepsilon_{3i}}\right)\left(\alpha_{2}(|\zeta_{i}|) \cdot \alpha_{3}(2|\zeta_{i}|) + |\zeta_{i}|\varepsilon_{1i}\right) \\ &\alpha_{8}(|\zeta_{i}|) \coloneqq \left(9+9b\left(\frac{\varepsilon_{1i}}{\varepsilon_{3i}}\right)\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}(\alpha_{2}(|\zeta_{i}|) \cdot |\zeta_{i}|) \\ &\alpha_{9}(|\zeta_{i}|) \coloneqq \left(\frac{6}{r_{1}}\right)\left(\frac{\varepsilon_{1i}}{\varepsilon_{3i}}\right)\left(\left[\alpha_{2}(|\zeta_{i}|)\right)^{2}(\alpha_{2}(|\zeta_{i}|) \cdot \alpha_{3}(2|\zeta_{i}|) + |\zeta_{i}|\varepsilon_{1i}\right) + r_{1}\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3+3b}{2b}}\alpha_{3}(2|\zeta_{i}|)\right) \end{aligned}$$

Note that, $2|\zeta_i| \ge |s_i| + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]|$.

For region:
$$\Psi_{1i}(\zeta_i) = \{ (v_{1i}, v_{2i}) \in \mathbb{R}^2 : |v_{1i}| \in \mathbb{R}, |v_{2i}| \ge \alpha_2(\zeta_i) \}$$

Consider the following properties,

$$\int_{0}^{v_{1i}} \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |r|^{b} \operatorname{sign}(r) \right] dr = \frac{k_{1i}}{1+b} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}} \left[|v_{1i}|^{1+b} \right] + \varepsilon_{1i} \left[|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} \left(|v_{1i}| \right) \right],$$

$$\left[1 - 1 + b \right] \left[1$$

$$\operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{3+3b}{2b}}}\left[\left|v_{1i}\right|^{\frac{3+3b}{2}}\right] = \operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}}\left[\left|v_{1i}\right|^{1+b}\right]\operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{1+b}{2b}}}\left[\left|v_{1i}\right|^{\frac{1+b}{2}}\right] \le \operatorname{sat}_{\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{1+b}{b}}}\left[\left|v_{1i}\right|^{1+b}\right]\left(\frac{\mathcal{E}_{1i}}{k_{1i}}\right)^{\frac{1+b}{2b}},$$

and using lemma 2.2, we have:

$$\begin{split} |v_{1i}|^{\frac{1}{2}} = & \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} (|v_{1i}|) + \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} (|v_{1i}|) \right)^{\frac{1}{2}} \leq & \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} (|v_{1i}|) \right)^{\frac{1}{2}} + \left(\operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} (|v_{1i}|) \right)^{\frac{1}{2}} \\ \leq & \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}} (|v_{1i}|) \right)^{\frac{1}{2}} + \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{1}{b}} (|v_{1i}|) \\ \end{cases}$$

Then, after rearrangement:

$$\begin{split} \dot{V}_{2} + \dot{V}_{s} &\leq -\frac{1}{3} |v_{2i}| \left(\frac{1}{2} |v_{2i}|^{2} \\ + \int_{0}^{n_{i}} \operatorname{sat}_{\varepsilon_{ii}} \left[k_{1i} |r|^{b} \operatorname{sign}(r)\right] dr \\ &+ 6k_{ai} \left(|v_{2i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{2i}}\right)^{\frac{1+b}{2b}}}\right] \\ &+ 6k_{ai} \left(|v_{2i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{2i}}\right)^{\frac{1+b}{2b}}}\right] \\ &- 6\alpha_{3} \left(s_{i} + \operatorname{sat}_{\mu_{i}} \left[\tilde{\varepsilon}_{2i}\right]\right) \\ &- \frac{1}{3} \left(\frac{1}{2} |v_{2i}|^{2} + \int_{0}^{n_{i}} \operatorname{sat}_{\varepsilon_{ii}} \left[k_{1i} |r|^{b} \operatorname{sign}(r)\right] dr \right) \left(|v_{2i}| \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} |v_{2i}|^{\frac{2b}{1+b}}\right] - 6\varepsilon_{1i} |s_{i}|\right) \\ &- \frac{1}{3} \left(\frac{1}{2} |v_{2i}|^{2} + \int_{0}^{n_{i}} \operatorname{sat}_{\varepsilon_{ii}} \left[k_{1i} |r|^{b} \operatorname{sign}(r)\right] dr \right) \left(|v_{2i}| \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} |v_{2i}|^{\frac{2b}{1+b}}\right] - 6\varepsilon_{1i} |s_{i}|\right) \\ &- \frac{1}{3} \left(\frac{1}{2} |v_{2i}| - \varepsilon_{1i}^{\frac{1}{2}} \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{0i}}\right)^{\frac{1}{b}}} \left(|v_{1i}|\right)\right)^{\frac{1}{2}} \right)^{\frac{1}{2}} \times \left(|v_{2i}| \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} |v_{2i}|^{\frac{2b}{1+b}}\right] - 6\varepsilon_{1i} |s_{i}|\right) \\ &- \left(\frac{\varepsilon_{1i}^{\frac{1}{2}}}{3} \left|v_{2i}| \left(|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{0i}}\right)^{\frac{1}{b}}} \left(|v_{1i}|\right)\right)^{\frac{1}{2}} \left(|v_{2i}| \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} |v_{2i}|^{\frac{2b}{1+b}}\right] \\ &- r_{1}\varepsilon_{1i} \left(\frac{9 + 9b}{2} \right) \left(\frac{1}{k_{1i}}\right)^{\frac{3}{2}} |s_{i}|\right) \\ &- \left(\frac{k_{1i}}{3 + 3b}\right) \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{0i}}\right)^{\frac{1+b}{b}}} \left\|v_{1i}|^{1+b}} \left(|v_{2i}| \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} |v_{2i}|^{\frac{2b}{1+b}}\right] \\ &- r_{1} \left(\frac{3 + 3b}{k_{1i}}\right) \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{0i}}\right)^{\frac{1+b}{b}}} \left|v_{1i}|^{1+b}} \left(|v_{2i}| \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} |v_{2i}|^{\frac{2b}{1+b}}\right] \\ &- r_{1} \left(\frac{3 + 3b}{k_{1i}}\right) \left(\frac{\varepsilon_{1i}}{k_{0i}}\right)^{\frac{1+b}{2b}} \times \alpha_{3} \left(s_{i} + \operatorname{sat}_{\mu_{i}} \left[\widetilde{\varepsilon_{2i}}\right]\right)\right) \\ \end{array}\right)$$

Thus, $\dot{V}_2 + \dot{V}_s \le 0$ in this region.

Next, for region: $\Psi_{2i}(\zeta_i) = \{ (v_{1i}, v_{2i}) \in \mathbb{R}^2 : |v_{1i}| \ge \alpha_1(\zeta_i), |v_{2i}| \le \alpha_2(\zeta_i) \}$

Consider the following property,

$$-\frac{r_{1}}{2}\operatorname{sat}_{\left(\frac{\mathcal{E}_{ll}}{k_{ll}}\right)^{\frac{3+3b}{2b}}}\left[\left|v_{1l}\right|^{\frac{3+3b}{2}}\right]\operatorname{sat}_{\mathcal{E}_{ll}}\left[k_{1l}\left|v_{1l}\right|^{b}\right] \leq -\left(\frac{r_{1}}{2}\right)\left(\frac{\mathcal{E}_{3l}}{\mathcal{E}_{1l}}\right)\operatorname{sat}_{\left(\frac{\mathcal{E}_{ll}}{k_{ll}}\right)^{\frac{3+3b}{2b}}}\left[\left|v_{1l}\right|^{\frac{3+3b}{2}}\right]\operatorname{sat}_{\mathcal{E}_{ll}}\left[k_{1l}\left|v_{1l}\right|^{b}\right]$$

since from the above definition, $\left(\frac{\varepsilon_{3i}}{\varepsilon_{1i}}\right) < 1$.

$$-\left(\frac{r_{1}(1+b)}{2}\right)\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}\varepsilon_{3i}\left(|v_{1i}|^{\frac{3}{2}}-\operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}}\left[|v_{1i}|^{\frac{3}{2}}\right]\right)$$
$$\leq -\left(\frac{r_{1}}{2}\right)\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}}\varepsilon_{3i}\left(|v_{1i}|^{\frac{3}{2}}-\operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}}\left[|v_{1i}|^{\frac{3}{2}}\right]\right),$$

$$\begin{pmatrix} -\left(\frac{r_{1}}{2}\right)\left(\frac{\varepsilon_{3i}}{\varepsilon_{1i}}\right) \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3+3b}{2b}}} \left[|v_{1i}|^{\frac{3+3b}{2}} \right] \operatorname{sat}_{\varepsilon_{1i}} \left[k_{1i} |v_{1i}|^{b} \right] \\ -\left(\frac{r_{1}}{2}\right)\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}} \varepsilon_{3i} \left[|v_{1i}|^{\frac{3}{2}} - \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2b}}} \left[|v_{1i}|^{\frac{3}{2}} \right] \right] \\ = -\left(\frac{r_{1}}{2}\right)\left(\frac{\varepsilon_{3i}}{\varepsilon_{1i}}\right) |v_{1i}|^{\frac{3}{2}} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}} \varepsilon_{1i}} \left[k_{1i} |v_{1i}|^{\frac{5b}{2}} \right],$$

$$\int_{0}^{v_{li}} \operatorname{sat}_{\varepsilon_{li}} \left[k_{1i} |r|^{b} \operatorname{sign}(r) \right] dr = \frac{k_{1i}}{1+b} \operatorname{sat}_{\left(\frac{\varepsilon_{li}}{k_{li}}\right)^{\frac{1+b}{b}}} \left[|v_{1i}|^{1+b} \right] + \varepsilon_{1i} \left[|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{li}}{k_{li}}\right)^{\frac{1}{b}}} \left(|v_{1i}| \right) \right]$$

$$= \frac{1}{1+b} \operatorname{sat}_{\varepsilon_{li}} \left[k_{1i} |v_{1i}|^{b} \right] \operatorname{sat}_{\left(\frac{\varepsilon_{li}}{k_{li}}\right)^{\frac{1}{b}}} \left[|v_{1i}| \right]$$

$$+ \operatorname{sat}_{\varepsilon_{li}} \left[k_{1i} |v_{1i}|^{b} \right] \left[|v_{1i}| - \operatorname{sat}_{\left(\frac{\varepsilon_{li}}{k_{li}}\right)^{\frac{1}{b}}} \left(|v_{1i}| \right) \right]$$

$$= \operatorname{sat}_{\varepsilon_{li}} \left[k_{1i} |v_{1i}|^{b} \right] v_{1i} \left| - \left(\frac{b}{1+b}\right) \operatorname{sat}_{\varepsilon_{li}} \left[k_{1i} |v_{1i}|^{b} \right] \operatorname{sat}_{\left(\frac{\varepsilon_{li}}{k_{li}}\right)^{\frac{1}{b}}} \left(|v_{1i}| \right]$$

$$\leq \varepsilon_{1i} |v_{1i}|,$$

and in this region, $|v_{2i}|$ is upper bounded by $|v_{2i}| \le \alpha_2 (|\zeta_i|)$,

Then, after rearrangement:

$$\begin{split} \dot{V}_{2} + \dot{V}_{s} &\leq -\left(\frac{r_{1}}{6}\right) \left(\frac{\varepsilon_{3i}}{\varepsilon_{1i}}\right) |v_{1i}| \begin{pmatrix} |v_{1i}|^{\frac{1}{2}} \operatorname{sat} & \left[\frac{\varepsilon_{1i}}{k_{1i}}\right]^{\frac{3}{2}} \varepsilon_{1i} \\ -\left(\frac{12}{r_{1}}\right) \left(\frac{\varepsilon_{1i}}{\varepsilon_{3i}}\right) (\alpha_{2} \left(|\zeta_{i}|\right) \cdot \alpha_{3} \left(|s_{i}| + \operatorname{sat}_{\mu_{i}}\left[\widetilde{\varepsilon}_{2i}\right]\right) + \varepsilon_{1i} |s_{i}| \right) \end{pmatrix} \\ &- \left(\frac{r_{1}}{6}\right) \left(\frac{\varepsilon_{3i}}{\varepsilon_{1i}}\right) |v_{1i}|^{\frac{1}{2}} \left(|v_{1i}| \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}} \varepsilon_{1i}} \left[k_{1i} |v_{1i}|^{\frac{5b}{2}}\right] - \left(9 + 9b \left(\frac{\varepsilon_{1i}}{\varepsilon_{3i}}\right) \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}} \alpha_{2} \left(|\zeta_{i}|\right) \cdot |s_{i}|\right) \\ &- \left(\frac{r_{1}}{6} \left(\frac{\varepsilon_{3i}}{\varepsilon_{1i}}\right) |v_{1i}|^{\frac{3}{2}} \operatorname{sat}_{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3}{2}} \varepsilon_{1i}} \left[k_{1i} |v_{1i}|^{\frac{5b}{2}}\right] \\ &+ \left(\left[\alpha_{2} \left(|\zeta_{i}|\right)\right]^{2} \left(\alpha_{2} \left(|\zeta_{i}|\right) \cdot \alpha_{3} \left(|s_{i}| + \operatorname{sat}_{\mu_{i}}\left[\widetilde{\varepsilon}_{2i}\right]\right) + \varepsilon_{1i} |s_{i}|\right) + r_{1} \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{3+3b}{2b}} \alpha_{2} \left(|s_{i}| + \operatorname{sat}_{\mu_{i}}\left[\widetilde{\varepsilon}_{2i}\right]\right) \right) \end{split}$$

Thus, $\dot{V}_2 + \dot{V}_s \le 0$ in this region.

Hence, we have shown that $\dot{\widetilde{V}} \leq \dot{V}_1 < 0$, outside the compact set:

$$\begin{split} \Psi_{3i}\left(\left|\zeta_{i}\right|\right) &= \left\{\left(v_{1i}, v_{2i}\right) \in \mathbb{R}^{2} : \left|v_{1i}\right| \leq \alpha_{1}\left(\left|\zeta_{i}\right|\right), \left|v_{2i}\right| \leq \alpha_{2}\left(\left|\zeta_{i}\right|\right)\right\} \\ \dot{\widetilde{V}} \leq \dot{V}_{1} < 0, \text{ for } \left(v_{1i}, v_{2i}\right) \notin \Psi_{3}\left(\left|\zeta_{i}\right|\right) \end{split}$$

Now using Lyapunov argument, define a Lyapunov level set,

$$\Sigma_{v}\left(\left|\zeta_{i}\right|\right) = \left\{\left(v_{1i}, v_{2i}\right) \in \mathbb{R}^{2} : V \leq \rho_{1i}\left(\left|\zeta_{i}\right|\right)\right\}$$

where the class \mathcal{K} function, ρ_{1i} is defined as,

$$\rho_{1i}(|\zeta_i|) = \max_{(v_{1i}, v_{2i}) \in bd \ \Psi_3(|\zeta_i|)} V$$

which exists since the boundary of the sets ψ_3 are compact. Note that $\psi_3(|\zeta_i|) \subset \Sigma_{\nu}(|\zeta_i|)$. Consequently,

$$\dot{\widetilde{V}} \leq \dot{V}_1 < 0$$
, for $V \geq \rho_{1i}(|\zeta_i|)$,

since $\psi_3(|\zeta_i|) \subset \Sigma_v(|\zeta_i|)$ which implies that the trajectories of (v_{1i}, v_{2i}) will enter the set $\Sigma_v(|\zeta_i|)$ in finite time and stay in it once entered.

For sufficiently small $|\zeta_i|$

Consider the compact set $\Psi_{4i} = \left\{ \left(v_{1i}, v_{2i} \right) \in \mathbb{R}^2 : \left| v_{1i} \right| \le \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{1}{b}}, \quad \left| v_{2i} \right| \le \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}} \right\}$

and the Lyapunov level set, $\Sigma_{sat} = \{ (v_{1i}, v_{2i}) \in \mathbb{R}^2 : V \le \rho_{i,sat} \},\$

where $\rho_{i,sat} = \min_{(v_{1i}, v_{2i}) \in bd} \Psi_{4i} V$

which exists since the boundary of the sets ψ_{4i} are compact. Note that $\Sigma_{\text{sat}} \subset \psi_{4i}$. Then, if $|\zeta_i|$ is sufficiently small, such that $\rho_{1i}(|\zeta_i|) \leq \rho_{i,sat}$, we have $\Sigma_{\nu}(|\zeta_i|) \subset \Sigma_{\text{sat}}$. Note that inside Σ_{sat} , the Lyapunov function becomes,

$$V = \left(\frac{1}{2}v_{2i}^{2} + \frac{k_{1i}}{1+b}|v_{1i}|^{1+b}\right)^{2} + r_{1}|v_{1i}|^{\frac{3+3b}{2}}\operatorname{sign}(|v_{1i}|)v_{2i}$$

where from section 2.2, we have $\underline{\pi}_1 \left(\left| v_{2i} \right|^4 + \left| v_{1i} \right|^{2+2b} \right) \le V$

with
$$\underline{\pi}_1 := \min\left\{\frac{1}{8}, \frac{1}{2}\left(\frac{k_{1i}}{1+b}\right)^2\right\}$$

Then, inside the set $\Sigma_{v}(|\zeta_{i}|)$, the upper bound on (v_{1i}, v_{2i}) can be found. In particular,

$$\underline{\pi}_{1}\left(\left|v_{2i}\right|^{4}+\left|v_{1i}\right|^{2+2b}\right) \leq V \leq \rho_{1i}\left(\left|\zeta_{i}\right|\right) \leq \rho_{i,sat}$$

$$\Rightarrow \underline{\pi}_{1}\left|v_{1i}\right|^{2+2b} \leq \rho_{1i}\left(\left|\zeta_{i}\right|\right), \text{ and } \underline{\pi}_{1}\left|v_{2i}\right|^{4} \leq \rho_{1i}\left(\left|\zeta_{i}\right|\right),$$

$$\Rightarrow \left|v_{1i}\right| \leq \left(\frac{\rho_{1i}\left(\left|\zeta_{i}\right|\right)}{\underline{\pi}_{1}}\right)^{\frac{1}{2+2b}}, \text{ and } \left|v_{2i}\right| \leq \left(\frac{\rho_{1i}\left(\left|\zeta_{i}\right|\right)}{\underline{\pi}_{1}}\right)^{\frac{1}{4}},$$

Note that, to ensure that the desired error dynamics is unsaturated, it is necessary that:

$$|v_{1i}| \leq \left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}$$
, and $|v_{2i} + s_i + \operatorname{sat}_{\mu_i}\left[\widetilde{e}_{2i}\right] \leq \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}$

Hence, from the above upper bounds on the states, to ensure that no saturation occurs, it is sufficient for the following class \mathcal{K} function to satisfy the inequality,

$$\rho_{1i}(|\zeta_i|) \le \rho_{i,sat}, \text{ and } \alpha_{10}(|\zeta_i|) \le \min\left\{\left(\frac{\varepsilon_{1i}}{k_{1i}}\right)^{\frac{1}{b}}, \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}\right\},$$

where
$$\alpha_{10}(|\zeta_i|) := \max\left\{\left(\frac{\rho_{1i}(|\zeta_i|)}{\underline{\pi}_1}\right)^{\frac{1}{2+2b}}, \left(\frac{\rho_{1i}(|\zeta_i|)}{\underline{\pi}_1}\right)^{\frac{1}{4}} + 2|\zeta_i|\right\},\$$

since

$$\begin{aligned} \left| v_{1i} \right| &\leq \left(\frac{\rho_{1i} \left(\left| \zeta_i \right| \right)}{\underline{\pi}_1} \right)^{\frac{1}{2+2b}} \leq \left(\frac{\varepsilon_{1i}}{k_{1i}} \right)^{\frac{1}{b}}, \text{ and} \\ \left| v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right| &\leq \left| v_{2i} \right| + \left| s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right| \leq \left(\frac{\rho_{1i} \left(\left| \zeta_i \right| \right)}{\underline{\pi}_1} \right)^{\frac{1}{4}} + 2\left| \zeta_i \right| \leq \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+b}{2b}}, \end{aligned}$$

which is possible for sufficiently small $|\zeta_i| > 0$. Thus, the control will be unsaturated in finite time and remain so thereafter for sufficiently small $|\zeta_i| > 0$. Once the states are unsaturated, the desired error dynamics become

$$\dot{v}_{1i} = v_{2i} + s_i$$

$$\dot{v}_{2i} = -k_{1i} |v_{1i}|^b \operatorname{sign}(v_{1i}) - k_{2i} |v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}])$$

which is similar to the one considered in the section where no saturation is considered the

control law (see chapter 3, 4).

Special case, when $|\zeta_i| = 0$

When $|\zeta_i| = 0$ and remains so for all future times, the system is asymptotical stable which implies that the trajectories of the system will reach the Lyapunov level set Σ_{sat} in finite time and remain in it for all future times, where we can see that for (v_{1i}, v_{2i}) inside the region Σ_{sat} , the control will be unsaturated. Essentially, it will become homogenous double integrator system and the Lyapunov function is homogeneous as well, where its stability has been studied by the author in [133]. In essence, once the system enters the unsaturated Lyapunov level set in finite time, its convergence properties is dependent on the parameter *b*, in particular for 0 < b < 1, we have finite time convergence, for b = 1 we have exponential convergence, and for b > 1 we have asymptotical stability to the origin.

For the case of b = 0:

Consider the following dynamics:

$$v_{1i} = v_{2i} + s_i$$

$$\dot{v}_{2i} = -k_{ai} \left(v_{2i} + s_i + \text{sat}_{\mu_i} [\widetilde{e}_{2i}] \right) + k_{ai} \text{sat} \left[v_{2i} + s_i + \text{sat}_{\mu_i} [\widetilde{e}_{2i}] \right]$$

$$-k_{2i} \text{sign} \left(v_{2i} + s_i + \text{sat}_{\mu_i} [\widetilde{e}_{2i}] \right) - k_{1i} \text{sign} (v_{1i})$$

where v_{1i} , v_{2i} , s_i , $\tilde{e}_{2i} \in \mathbb{R}$, are the scalar state variables, k_{1i} , k_{2i} , k_{ai} , μ_i are positive constants, $b \ge 0$ real number with

$$k_{3i} := \begin{cases} k_{1i} - k_{ai} \mu_i - k_{ai} - k_{2i}, & \text{for } |\widetilde{e}_{2i}| \neq 0, \\ k_{1i} - k_{ai} - k_{2i}, & \text{for } \widetilde{e}_{2i} = 0 & (\text{for the case of full state feedback}) \end{cases}$$

The results in this section are applicable to the desired dynamics section of both chapter 5 & 6. In particular, for Chapter 3, let $\tilde{e}_{2i} = 0$, while for Chapter 4, no changes required, and the same differential equations are obtained.

Lyapunov function

$$V = \left(\frac{1}{2}v_{2i}^{2} + k_{1i}\int_{0}^{v_{1i}}\operatorname{sign}(r) dr\right)^{2} + r_{1}|v_{1i}|^{\frac{3}{2}}\operatorname{sign}(v_{1i}) v_{2i} + r_{1}\left(\frac{2}{5}\right)k_{ai}|v_{1i}|^{\frac{5}{2}} - r_{1}k_{ai}\int_{0}^{v_{1i}}\left[\operatorname{sat}_{1}[|r|^{\frac{3}{2}}]\right]\operatorname{sign}(r) dr ,$$

where r_1 is a positive constant scalar, will be shown as a strict Lyapunov function. Note that the above function is locally Lipschitz and regular. It is differentiable everywhere for b > 0, and not differentiable on $v_{1i} = 0$ for b = 0. In particular, when none of the terms of the dynamics are saturated, the Lyapunov function is indeed identical to that proposed in Section 2.2 for the twisting-based family of algorithms.

Sign definiteness of V:

Firstly, note that the term

$$r_1 k_{ai} \left(\int_{0}^{v_{1i}} |r|^{\frac{3}{2}} \operatorname{sign}(r) - \operatorname{sat}_1 \left[|r|^{\frac{3}{2}} \operatorname{sign}(r) \right] dr \right) \geq 0,$$

the function is upper bounded by

$$V \leq \left(\frac{1}{2} v_{2i}^{2} + k_{1i} \int_{0}^{v_{1i}} \operatorname{sign}(r) dr\right)^{2} + r_{1} |v_{1i}|^{\frac{3}{2}} |v_{2i}| + r_{1} \left(\frac{2}{5}\right) k_{ai} |v_{1i}|^{\frac{5}{2}} - r_{1} k_{ai} \int_{0}^{v_{1i}} \left[\operatorname{sat}_{1}[|r|^{\frac{3}{2}}]\right] \operatorname{sign}(r) dr$$

which is positive definite and radially unbounded, next for the lower bound,

$$V \ge \left(\frac{1}{2} v_{2i}^{2} + k_{1i} |v_{1i}|\right)^{2} - r_{1} |v_{1i}|^{\frac{3}{2}} |v_{2i}|$$

$$\ge \frac{1}{8} |v_{2i}|^{4} + \frac{k_{1i}^{2}}{2} |v_{1i}|^{2} + \frac{1}{8} |v_{2i}|^{4} + \frac{k_{1i}^{2}}{2} |v_{1i}|^{2} - r_{1} |v_{1i}|^{\frac{3}{2}} |v_{2i}|$$

Using lemma 2.1,

$$\frac{1}{8} |v_{2i}|^4 + \frac{k_{1i}^2}{2} |v_{1i}|^2 \ge \left(\frac{1}{2}\right)^{\frac{1}{4}} |v_{2i}| \left(\frac{4k_{1i}^2}{6}\right)^{\frac{3}{4}} |v_{1i}|^{\frac{3}{2}}$$

Consequently:

$$V \ge \frac{1}{8} |v_{2i}|^4 + \frac{k_{1i}^2}{2} |v_{1i}|^2 \text{ with } 2^{\frac{3}{4}} \left(\frac{2^{\frac{1}{2}}}{6^{\frac{3}{4}}}\right) k_{1i}^{\frac{3}{2}} > r_1 .$$

Note that such an $r_1 > 0$ always exists for any positive k_{1i} . Thus, the Lyapunov function is positive definite and radially unbounded. In accordance to lemma 2.4, the time derivative of the Lyapunov function along the solutions of the system exists almost everywhere.

<u>Time derivative of Lyapunov function b = 0:</u>

For b = 0, *V* is not differentiable on $v_{1i} = 0$:

$$\frac{d}{dt}V((v_{1i},v_{2i}),t) \stackrel{a.e.}{\in} \dot{\widetilde{V}}((v_{1i},v_{2i}),t) = \bigcap_{\xi \in \partial V(\mathbf{v}(t),t)} \xi^T \mathbf{K}\begin{bmatrix} \dot{v}_{1i} \\ \dot{v}_{2i} \end{bmatrix} ((v_{1i},v_{2i}),t) \subset \dot{\widetilde{V}}_1 + \dot{\widetilde{V}}_2,$$

where

$$\dot{\widetilde{V}}_{1} = \bigcap_{\xi \in \partial V(\mathbf{v}(t),t)} \xi^{T} K[f](v_{1i}, v_{2i}), \quad \dot{\widetilde{V}}_{2} = \bigcap_{\xi \in \partial V(\mathbf{v}(t),t)} \xi^{T} K\begin{bmatrix}s_{i}\\0\end{bmatrix},$$

$$\begin{split} & \left\{ \begin{cases} \forall v_{1i} \neq 0, v_{2i} \in \mathbb{R} : \\ & \left[\begin{pmatrix} -k_{ai} (\left(v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}] \right) - \operatorname{sat} [v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}]] \right) \\ & -k_{2i} \operatorname{SGN} (v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}]) - k_{1i} \operatorname{sign} (v_{1i}) \end{cases} \right\}, \\ & \forall v_{1i} = 0, v_{2i} \in \mathbb{R} : \\ & \left[\begin{pmatrix} -k_{ai} (\left(v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}] \right) - \operatorname{sat} [v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}]]) \\ -k_{2i} \operatorname{SGN} (v_{2i} + s + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}]) - \operatorname{sat} [v_{2i} + s_{i} + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}]]) \\ & -k_{2i} \operatorname{SGN} (v_{2i} + s + \operatorname{sat}_{\mu_{i}} [\tilde{e}_{2i}]) - k_{1i} [-1, 1] \end{cases} \right] \end{cases} \right\}, \\ & \left\{ \partial V = \mathbf{K} [\nabla V] = \mathbf{K} \begin{bmatrix} \frac{\partial V}{\partial v_{1i}} \\ \frac{\partial V}{\partial v_{2i}} \end{bmatrix} \\ & \left\{ \left[\left(2k_{1i}^{2} v_{1i} + \frac{3}{2}r_{1} |v_{1i}|^{\frac{1}{2}} v_{2i} + k_{1i} \operatorname{sign} (v_{1i}) |v_{2i}|^{2} \\ + r_{1}k_{ai} (|v_{1i}|^{\frac{3}{2}} - \operatorname{sat} [|v_{1i}|^{\frac{3}{2}}] \right) \operatorname{sign} (v_{1i}) \\ v_{2i}^{3} + r_{1} |v_{1i}|^{\frac{3}{2}} \operatorname{sign} (v_{1i}) + 2k_{1i} |v_{1i} |v_{2i} \end{bmatrix} \right\}, \forall v_{1i} \neq 0, v_{2i} \in \mathbb{R}, \\ & \left[\left[-1, 1 \frac{1}{k_{1i}} |v_{2i}|^{2} \\ v_{2i}^{3} + r_{1} |v_{1i}|^{\frac{3}{2}} \operatorname{sign} (v_{1i}) + 2k_{1i} |v_{1i} |v_{2i} \end{bmatrix} \right], \forall v_{1i} \neq 0, v_{2i} \in \mathbb{R}, \end{aligned} \right\}$$

Computing $\dot{\tilde{V}}_2$:

Since s_i is a Filippov solution, it is absolutely continuous, $K[s_i] = s_i$:

For $v_{1i} \neq 0, \forall v_{2i} \in \mathbb{R}$:

$$\dot{\tilde{V}}_{2} = 2k_{1i}^{2}v_{1i}s_{i} + \frac{3}{2}r_{1}|v_{1i}|^{\frac{1}{2}}v_{2i}s_{i} + k_{1i}\operatorname{sign}(v_{1i})|v_{2i}|^{2}s_{i} + r_{1}k_{ai}\left(|v_{1i}|^{\frac{3}{2}} - \operatorname{sat}\left[|v_{1i}|^{\frac{3}{2}}\right]\right)\operatorname{sign}(v_{1i})s_{i}$$

For $v_{1i} = 0, \forall v_{2i} \in \mathbb{R}$:

Let $(\xi_2 k_{1i} | v_{2i} |^2, v_{2i}^3)^T$ with $\xi_2 \in [-1, 1]$ be an arbitrary element of ∂V , then

$$\dot{\widetilde{V}}_{2} = \bigcap_{\xi_{2} \in [-1, 1]} \xi_{2} k_{1i} |v_{2i}|^{2} s_{i} = \emptyset$$

where by convention max $\dot{\tilde{V}}_2 = -\infty$, if $\dot{\tilde{V}}_2 = \emptyset$.(see [74])

Thus, for $\forall (v_{1i}, v_{2i}) \in \mathbb{R}^2$:

$$\dot{\tilde{V}}_{2} \leq 2k_{1i}^{2} |v_{1i}| |s_{i}| + \frac{3}{2}r_{1} |v_{1i}|^{\frac{1}{2}} |v_{2i}| |s_{i}| + k_{1i} |v_{2i}|^{2} |s_{i}| + r_{1}k_{ai} \left(|v_{1i}|^{\frac{3}{2}} - \operatorname{sat}\left[|v_{1i}|^{\frac{3}{2}} \right] \right) \operatorname{sign}(v_{1i}) s_{i}$$

Computing $\dot{\tilde{V}_1}$ for each case, we have

For $v_{1i} \neq 0, \forall v_{2i} \in \mathbb{R}$:

$$\begin{split} \dot{\widetilde{V}}_{1} &= \frac{3}{2} r_{1} |v_{1i}|^{\frac{1}{2}} |v_{2i}|^{2} - 2k_{1i}k_{2i} |v_{1i}|v_{2i} \text{SGN}(v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) \\ &- k_{2i} v_{2i}^{-3} \text{SGN}(v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - k_{1i} r_{1} |v_{1i}|^{\frac{3}{2}} \\ &- r_{1} k_{2i} |v_{1i}|^{\frac{3}{2}} \text{sign}(v_{1i}) \text{SGN}(v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - r_{1} k_{ai} \text{sat} \left[|v_{1i}|^{\frac{3}{2}} \right] \text{sign}(v_{1i}) (v_{2i}) \\ &- r_{1} k_{ai} |v_{1i}|^{\frac{3}{2}} \text{sign}(v_{1i}) (s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) + r_{1} k_{ai} |v_{1i}|^{\frac{3}{2}} \text{sign}(v_{1i}) \text{sat} \left[v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}] \right] \\ &- v_{2i}^{-3} k_{ai} ((v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - \text{sat} \left[v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}] \right] \\ &- 2k_{1i} |v_{1i}| v_{2i} k_{ai} ((v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - \text{sat} \left[v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}] \right] \end{split}$$

For $v_{1i} = 0, \forall v_{2i} \in \mathbb{R}$:

Let $(\xi_2 k_{1i} | v_{2i} |^2, v_{2i}^3)^T$ with $\xi_2 \in [-1, 1]$ be an arbitrary element of ∂V , then

$$\dot{\widetilde{V}}_{1} = \bigcap_{\xi_{2} \in [-1, 1]} [\xi_{2} - 1, \xi_{2} + 1] k_{1i} v_{2i}^{3} - k_{2i} v_{2i}^{3} \text{SGN}(v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - v_{2i}^{3} k_{ai} ((v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - \text{sat}[v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}])) = -k_{2i} v_{2i}^{3} \text{SGN}(v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - v_{2i}^{3} k_{ai} ((v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - \text{sat}[v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]])$$

Thus, for $\forall (v_{1i}, v_{2i}) \in \mathbb{R}^2$:

$$\begin{split} \dot{\widetilde{V}}_{1} &= \frac{3}{2} r_{1} |v_{1i}|^{\frac{1}{2}} |v_{2i}|^{2} - 2k_{1i}k_{2i} |v_{1i}|v_{2i} \text{SGN}(v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) \\ &- k_{2i} v_{2i}^{-3} \text{SGN}(v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - k_{1i} r_{1} |v_{1i}|^{\frac{3}{2}} \\ &- r_{1} k_{2i} |v_{1i}|^{\frac{3}{2}} \text{sign}(v_{1i}) \text{SGN}(v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - r_{1} k_{ai} \text{sat} \left[|v_{1i}|^{\frac{3}{2}} \right] \text{sign}(v_{1i}) (v_{2i}) \\ &- r_{1} k_{ai} |v_{1i}|^{\frac{3}{2}} \text{sign}(v_{1i}) (s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) + r_{1} k_{ai} |v_{1i}|^{\frac{3}{2}} \text{sign}(v_{1i}) \text{sat} \left[v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}] \right] \\ &- v_{2i}^{-3} k_{ai} ((v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - \text{sat} \left[v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}] \right] \\ &- 2k_{1i} |v_{1i}|v_{2i}k_{ai} ((v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}]) - \text{sat} \left[v_{2i} + s_{i} + \text{sat}_{\mu_{i}}[\widetilde{e}_{2i}] \right] \end{split}$$

Hence, the generalized time derivative of the Lyapunov function, after rearrangement:

$$\begin{split} \hat{\vec{V}} &= \hat{\vec{V}}_{1} + \hat{\vec{V}}_{2} \\ &\leq 2k_{1i}^{2} |v_{1i}| |s_{i}| + \frac{3}{2}r_{1} |v_{1i}|^{\frac{1}{2}} |v_{2i}| |s_{i}| + k_{1i} |v_{2i}|^{2} |s_{i}| + \frac{3}{2}r_{1} |v_{1i}|^{\frac{1}{2}} |v_{2i}|^{2} \\ &- 2k_{1i}k_{2i} |v_{1i}| v_{2i} \text{SGN} \left(v_{2i} + s_{i} + \text{sat}_{\mu_{i}} [\widetilde{e}_{2i}] \right) \\ &- k_{2i} v_{2i}^{3} \text{SGN} \left(v_{2i} + s_{i} + \text{sat}_{\mu_{i}} [\widetilde{e}_{2i}] \right) - r_{1} |v_{1i}|^{\frac{3}{2}} \left(k_{1i} - k_{2i} - k_{ai} \text{sat}_{\mu_{i}} [|\widetilde{e}_{2i}|] - k_{ai} \right) \end{split}$$
(B1 - 2)
$$&+ r_{1} k_{ai} \text{sat} \left[|v_{1i}|^{\frac{3}{2}} \right] \left(|v_{2i}| + |s_{i}| \right) \\ &- v_{2i}^{3} k_{ai} \left(\left(v_{2i} + s_{i} + \text{sat}_{\mu_{i}} [\widetilde{e}_{2i}] \right) - \text{sat} \left[v_{2i} + s_{i} + \text{sat}_{\mu_{i}} [\widetilde{e}_{2i}] \right] \right) \\ &- 2k_{1i} |v_{1i}| v_{2i} k_{ai} \left(\left(v_{2i} + s_{i} + \text{sat}_{\mu_{i}} [\widetilde{e}_{2i}] \right) - \text{sat} \left[v_{2i} + s_{i} + \text{sat}_{\mu_{i}} [\widetilde{e}_{2i}] \right] \right) \end{split}$$

Next, the state space is divided into three regions for analysis:

$$\begin{split} \phi_{1i}(|\zeta_{i}|) &= \{ (v_{1i}, v_{2i}) \in \mathbb{R}^{2} : v_{1i} \in \mathbb{R}, |v_{2i}| \geq \alpha_{12}(|\zeta_{i}|) \}, \\ \phi_{2i}(|\zeta_{i}|) &= \{ (v_{1i}, v_{2i}) \in \mathbb{R}^{2} : |v_{1i}| \geq \alpha_{11}(|\zeta_{i}|), |v_{2i}| \leq \alpha_{12}(|\zeta_{i}|) \}, \\ \phi_{3i}(|\zeta_{i}|) &= \{ (v_{1i}, v_{2i}) \in \mathbb{R}^{2} : |v_{1i}| \leq \alpha_{11}(|\zeta_{i}|), |v_{2i}| \leq \alpha_{12}(|\zeta_{i}|) \}, \end{split}$$

where $|\zeta_i|$ is defined as: $|\zeta_i| = \max\{|s_i|, \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}|]\}$ while $\alpha_{13}(|\zeta_i|)$ and $\alpha_{14}(|\zeta_i|)$ are class \mathcal{K}

functions defined by:

$$\alpha_{11}(\zeta_{i}|) := \max \begin{cases} \left(\frac{4}{r_{1}k_{3i}}\right)^{2} \left(2k_{1i}^{2}|\zeta_{i}|+2k_{1i}k_{2i}\cdot\alpha_{12}(|\zeta_{i}|)+2k_{1i}k_{ai}\cdot\alpha_{12}(|\zeta_{i}|)\cdot(\alpha_{12}(|\zeta_{i}|)+2|\zeta_{i}|)\right)^{2}, \\ \left(\frac{4}{r_{1}k_{3i}}\right) \left(\frac{3}{2}r_{1}\alpha_{12}(|\zeta_{i}|)\cdot|\zeta_{i}|+\frac{3}{2}r_{1}\left[\alpha_{12}(|\zeta_{i}|)\right]^{2}\right), \\ \left(\frac{4}{r_{1}k_{3i}}\right)^{\frac{2}{3}} \left(k_{1i}\left[\alpha_{12}(|\zeta_{i}|)\right]^{2}|\zeta_{i}|+k_{2i}\left[\alpha_{12}(|\zeta_{i}|)\right]^{3}, \\ +r_{1}k_{ai}(\alpha_{12}(|\zeta_{i}|)+|\zeta_{i}|)+\left[\alpha_{12}(|\zeta_{i}|)\right]^{3}k_{ai}(\alpha_{12}(|\zeta_{i}|)+2|\zeta_{i}|)\right)^{\frac{2}{3}} \end{cases} \right)^{\frac{2}{3}} \\ \alpha_{12}(|\zeta_{i}|) := \max \left\{ \left(\frac{3k_{1i}}{k_{2i}}\right)\zeta_{i}|, \left(\frac{3^{\frac{3}{2}}r_{1}}{2k_{1i}^{\frac{1}{2}}k_{2i}}\right)|\zeta_{i}|, \left(\left(\frac{2}{k_{2i}}\right)^{\frac{1}{3}} \left(\frac{4}{3r_{1}k_{3i}}\right)^{\frac{2}{3}}2r_{1}k_{ai}\right)|\zeta_{i}|\right\}, \end{cases}$$

For region:
$$\phi_{1i}(\zeta_i) = \{ (v_{1i}, v_{2i}) \in \mathbb{R}^2 : v_{1i} \in \mathbb{R}, |v_{2i}| \ge \alpha_{12}(\zeta_i) \}$$

Note that in this region, the following properties apply:

$$v_{2i}\operatorname{SGN}(v_{2i} + s_i + \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}]) = |v_{2i}|\operatorname{sign}(v_{2i})\operatorname{SGN}(v_{2i} + s_i + \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}]) = |v_{2i}|$$

when $|\zeta_i| = 0 \Rightarrow (s_i, \widetilde{e}_{2i})^T = (0, 0)^T$, the above is trivially satisfied, when $|\zeta_i| \neq 0$, note that $(v_{2i} + s_i + \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}]) \neq 0$, since $|v_{2i}| \ge \frac{3k_{1i}}{k_{2i}} |\zeta_i| > |s_i + \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}]$, Thus, we have $\operatorname{SGN}(v_{2i} + s_i + \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}]) = \operatorname{sign}(v_{2i} + s_i + \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}]),$ $= \operatorname{sign}(v_{2i}) \operatorname{since} |v_{2i}| > |s_i + \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}]$ in this region

Similarly, we have

$$\begin{aligned} v_{2i} \left(v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] - \operatorname{sat} \left[v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right] \right) \\ &= \left| v_{2i} \right| \left(\left| v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right] - \operatorname{sat} \left[\left| v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right] \right) \\ &\geq 0 \end{aligned}$$

Hence, we have, from (B1 - 2),

$$\begin{split} \dot{\tilde{V}} &\leq -k_{1i} |v_{1i}| \left(k_{2i} |v_{2i}| - 2k_{1i} |s_i|\right) - |v_{2i}|^2 \left(\frac{k_{2i}}{3} |v_{2i}| - k_{1i} |s_i|\right) - k_{2i} |v_{2i}| \left(\frac{1}{3^{\frac{1}{2}}} |v_{2i}| - k_{1i}^{\frac{1}{2}} |v_{1i}|^{\frac{1}{2}}\right)^2 \\ &- |v_{1i}|^{\frac{1}{2}} |v_{2i}| \left(\frac{k_{1i}^{\frac{1}{2}} k_{2i}}{3^{\frac{1}{2}}} |v_{2i}| - \frac{3}{2} r_1 |s_i|\right) - |v_{1i}|^{\frac{1}{2}} |v_{2i}|^2 \left(\frac{k_{1i}^{\frac{1}{2}} k_{2i}}{3^{\frac{1}{2}}} - \frac{3}{2} r_1\right) \\ &- \frac{k_{2i}}{6} |v_{2i}|^3 - \frac{r_1 k_{3i}}{2} |v_{1i}|^{\frac{3}{2}} + r_1 k_{ai} \operatorname{sat} \left[|v_{1i}|^{\frac{3}{2}}\right] |v_{2i}| + r_1 k_{ai} \operatorname{sat} \left[|v_{1i}|^{\frac{3}{2}}\right] |s_i| \\ &- \frac{k_{2i}}{6} |v_{2i}|^3 - \frac{r_1 k_{3i}}{2} |v_{1i}|^{\frac{3}{2}} \\ &- |v_{2i}|^3 k_{ai} \left(v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}] - \operatorname{sat} \left[|v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]\right] \right) \\ &- 2k_{1i} |v_{1i}| |v_{2i}| k_{ai} \left(v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}] - \operatorname{sat} \left[|v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]\right] \right) \end{split}$$

From lemma 2.1,

$$-\frac{k_{2i}}{6}|v_{2i}|^{3}-\frac{r_{1}k_{3i}}{2}|v_{1i}|^{\frac{3}{2}} \le -\left(\frac{k_{2i}}{2}\right)^{\frac{1}{3}}|v_{2i}\left(\frac{3r_{1}k_{3i}}{4}\right)^{\frac{2}{3}}|v_{1i}|$$

and note that

$$\operatorname{sat}\left[|v_{1i}|^{\frac{3}{2}}\right] \le \operatorname{sat}\left[|v_{1i}|\right] \le |v_{1i}|, \text{ since } \frac{3}{2} > 1$$

Thus,

$$\begin{split} \dot{\widetilde{V}} &\leq -k_{1i} |v_{1i}| \left(k_{2i} |v_{2i}| - 2k_{1i} |s_i|\right) - |v_{2i}|^2 \left(\frac{k_{2i}}{3} |v_{2i}| - k_{1i} |s_i|\right) - k_{2i} |v_{2i}| \left(\frac{1}{3^{\frac{1}{2}}} |v_{2i}| - k_{1i}^{\frac{1}{2}} |v_{1i}|^{\frac{1}{2}}\right)^2 \\ &- |v_{1i}|^{\frac{1}{2}} |v_{2i}| \left(\frac{k_{1i}^{\frac{1}{2}} k_{2i}}{3^{\frac{1}{2}}} |v_{2i}| - \frac{3}{2} r_1 |s_i|\right) - |v_{1i}|^{\frac{1}{2}} |v_{2i}|^2 \left(\frac{k_{1i}^{\frac{1}{2}} k_{2i}}{3^{\frac{1}{2}}} - \frac{3}{2} r_1\right) \\ &- |v_{1i}| |v_{2i}| \left(\frac{1}{2} \left(\frac{k_{2i}}{2}\right)^{\frac{1}{3}} \left(\frac{3r_1 k_{3i}}{4}\right)^{\frac{2}{3}} - r_1 k_{ai}\right) - |v_{1i}| \left(\frac{1}{2} \left(\frac{k_{2i}}{2}\right)^{\frac{1}{3}} \left(\frac{3r_1 k_{3i}}{4}\right)^{\frac{2}{3}} |v_{2i}| - r_1 k_{ai} |s_i|\right) \\ &- \frac{k_{2i}}{6} |v_{2i}|^3 - \frac{r_1 k_{3i}}{2} |v_{1i}|^{\frac{3}{2}} \\ &- |v_{2i}|^3 k_{ai} \left(v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}] - \operatorname{sat} [v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]]) \\ &- 2k_{1i} |v_{1i}| |v_{2i}| k_{ai} \left(v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}] - \operatorname{sat} [v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]]) \end{split}$$

where $\dot{\vec{V}} < 0$ in this region for any positive

$$0 < r_1 < \min\left\{\frac{2k_{1i}^{\frac{1}{2}}k_{2i}}{3^{\frac{3}{2}}}, \left(\frac{1}{2k_{ai}}\right)\left(\frac{k_{2i}}{2}\right)^{\frac{1}{3}}\left(\frac{3r_1k_{3i}}{4}\right)^{\frac{2}{3}}\right\}$$

For region:
$$\phi_{2i}(\zeta_i) = \{(v_{1i}, v_{2i}) \in \mathbb{R}^2 : |v_{1i}| \ge \alpha_{11}(\zeta_i), |v_{2i}| \le \alpha_{12}(\zeta_i)\}$$

Note that in this region the term $|v_{2i}|$ is upper bounded and recall that

 $|\zeta_i| = \max\{s_i|, \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}|]\}.$

Also,

$$\left| \left(v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right) - \operatorname{sat} \left[v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right] \le \left| v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right|$$

Then we have, from (B1 - 2),

$$\begin{split} \dot{\widetilde{V}} &\leq -|v_{1i}| \left(\frac{r_{1}k_{3i}}{4} |v_{1i}|^{\frac{1}{2}} - \left(2k_{1i}^{2} |\zeta_{i}| + 2k_{1i}k_{2i} \cdot \alpha_{12} (|\zeta_{i}|) + 2k_{1i}k_{ai} \cdot \alpha_{12} (|\zeta_{i}|) \right) \cdot (\alpha_{12} (|\zeta_{i}|) + 2|\zeta_{i}|)) \right) \\ &- |v_{1i}|^{\frac{1}{2}} \left(\frac{r_{1}k_{3i}}{4} |v_{1i}| - \left(\frac{3}{2}r_{1}\alpha_{12} (|\zeta_{i}|) \right) \cdot |\zeta_{i}| + \frac{3}{2}r_{1} [\alpha_{12} (|\zeta_{i}|)]^{2} \right) \right) \\ &- \frac{r_{1}k_{3i}}{4} |v_{1i}|^{\frac{3}{2}} + \left(\frac{k_{1i} [\alpha_{12} (|\zeta_{i}|)]^{2} |\zeta_{i}| + k_{2i} [\alpha_{12} (|\zeta_{i}|)]^{3} + r_{1}k_{ai} (\alpha_{12} (|\zeta_{i}|) + |\zeta_{i}|) \right) \\ &+ [\alpha_{12} (|\zeta_{i}|)]^{3} k_{ai} (\alpha_{12} (|\zeta_{i}|) + 2|\zeta_{i}|) \end{split}$$

where $\dot{\widetilde{V}} < 0$ in this region.

Now consider the compact region $\phi_{3i}(|\zeta_i|) = \{(v_{1i}, v_{2i}) \in \mathbb{R}^2 : |v_{1i}| \le \alpha_{11}(|\zeta_i|), |v_{2i}| \le \alpha_{12}(|\zeta_i|)\}$.

From the analysis in the other regions, we have shown that $\dot{\vec{V}} < 0$ outside of the region ϕ_3 . Now, define a Lyapunov level set:

$$\Sigma_{2i}(\boldsymbol{\zeta}_i|) = \{ (\boldsymbol{v}_{1i}, \boldsymbol{v}_{2i}) \in \mathbb{R}^2 : V \le \rho_{2i}(\boldsymbol{\zeta}_i|) \}$$

where ρ_{2i} is a class \mathcal{K} function defined as:

$$\rho_{2i}(|\zeta_i|) = \max_{(v_{1i}, v_{2i}) \in \mathrm{bd} \ \phi_3(|\zeta_i|)} V$$

which exists since the boundary of the set is compact and V is continuous. Then we observe

that $\phi_3(|\zeta_i|) \subset \Sigma_{2i}(|\zeta_i|)$. As a result, we have

$$\frac{d}{dt}V(v_{1i},v_{2i}) \in \widetilde{\widetilde{V}}(v_{1i},v_{2i}) < 0 \text{ for } V \ge \rho_{2i}(|\zeta_i|)$$

which implies that the trajectories will enter the compact level set $\Sigma_{2i}(|\zeta_i|)$ in finite time and stay in it once entered.

For sufficiently small $|\zeta_i|$

Consider the compact set $\Omega_{s=0} = \{ (v_{1i}, v_{2i}) \in \mathbb{R}^2 : |v_{1i}| \le 1, |v_{2i}| \le 1 \}$

and the Lyapunov level set, $\Sigma_{s=0} = \{ (v_{1i}, v_{2i}) \in \mathbb{R}^n : V \le k_{s=0} \},\$

where
$$k_{s=0} = \min_{s \in bd \ \Omega_{s=0}} V$$

which exists since the boundary of the sets $\Omega_{s=0}$ are compact. Note that $\Sigma_{s=0} \subset \Omega_{s=0}$.

Then, if $|\zeta_i|$ is sufficiently small, such that $\rho_{2i}(|\zeta_i|) \le k_{s=0}$, we have $\Sigma_{2\nu}(|\zeta_i|) \subset \Sigma_{s=0}$, note that inside $\Sigma_{s=0}$, the Lyapunov function becomes,

$$V = \left(\frac{1}{2}v_{2i}^{2} + k_{1i}|v_{1i}|\right)^{2} + r_{1}|v_{1i}|^{\frac{3}{2}}\operatorname{sign}(|v_{1i}|)v_{2i}$$

where from section 2.2, we have $\underline{\pi}_1 \left(|v_{2i}|^4 + |v_{1i}|^2 \right) \le V$

with
$$\underline{\pi}_1 := \min\left\{\frac{1}{8}, \frac{1}{2}{k_{1i}}^2\right\}$$

Then, inside the set $\Sigma_{2\nu}(|\zeta_i|)$, the upper bound on (v_{1i}, v_{2i}) can be found. In particular,

$$\underline{\pi}_{1}\left(\left|\boldsymbol{v}_{2i}\right|^{4}+\left|\boldsymbol{v}_{1i}\right|^{2}\right) \leq V \leq \rho_{2i}\left(\left|\boldsymbol{\zeta}_{i}\right|\right) \leq k_{s=0}$$

$$\Rightarrow \underline{\pi}_{1}\left|\boldsymbol{v}_{1i}\right|^{2} \leq \rho_{2i}\left(\left|\boldsymbol{\zeta}_{i}\right|\right), \text{ and } \underline{\pi}_{1}\left|\boldsymbol{v}_{2i}\right|^{4} \leq \rho_{2i}\left(\left|\boldsymbol{\zeta}_{i}\right|\right),$$

$$\Rightarrow \left|\boldsymbol{v}_{1i}\right| \leq \left(\frac{\rho_{2i}\left(\left|\boldsymbol{\zeta}_{i}\right|\right)}{\underline{\pi}_{1}}\right)^{\frac{1}{2}}, \text{ and } \left|\boldsymbol{v}_{2i}\right| \leq \left(\frac{\rho_{2i}\left(\left|\boldsymbol{\zeta}_{i}\right|\right)}{\underline{\pi}_{1}}\right)^{\frac{1}{4}},$$

Note that, to ensure that the desired error dynamics is unsaturated, it is necessary that:

$$\left|v_{2i}+s_{i}+\operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i}\right]\right|\leq1,$$

Hence, from the above upper bounds on the states, to ensure that no saturation occurs, it is sufficient for the following class \mathcal{K} function to satisfy the inequality,

$$\rho_{2i}(|\zeta_i|) \le k_{s=0}, \text{ and } \left(\frac{\rho_{2i}(|\zeta_i|)}{\underline{\pi}_1}\right)^{\frac{1}{4}} + 2|\zeta_i| \le 1,$$

since

$$\left|v_{2i}+s_{i}+\operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i}\right]\right| \leq \left|v_{2i}\right|+\left|s_{i}+\operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i}\right]\right| \leq \left(\frac{\rho_{2i}\left(\left|\zeta_{i}\right|\right)}{\underline{\pi}_{1}}\right)^{\frac{1}{4}}+2\left|\zeta_{i}\right|\leq 1,$$

which is possible for sufficiently small $|\zeta_i| > 0$. Thus, the control will be unsaturated in finite time and remain so thereafter for sufficiently small $|\zeta_i| > 0$. Once the states are unsaturated, the desired error dynamics become

$$\dot{v}_{1i} = v_{2i} + s_i$$

 $\dot{v}_{2i} = -k_{1i} \operatorname{sign}(v_{1i}) - k_{2i} \operatorname{sign}(v_{2i} + s_i + \operatorname{sat}_{\mu_i}[\tilde{e}_{2i}])$

which is similar to the one considered in the section where no saturation is considered the

control law (see chapter 3, 4).

Special case, when $|\zeta_i| = 0$

Consider the special case of b = 0 and $|\zeta_i| = 0$, (i.e. $s_i = 0$, and $\tilde{e}_{2i} = 0$), it is possible to show convergence to the origin in finite time.

In particular, from the above development, we obtain,

$$\dot{\widetilde{V}} < 0, \quad \forall (v_{1i}, v_{2i}) \in \mathbb{R}^2 \setminus (0, 0) \text{ for } |\zeta_i| = 0.$$

since $\alpha_{13}(0) = 0$ and $\alpha_{14}(0) = 0$. Hence, the system is asymptotically stable which implies that the trajectories of the system will reach the Lyapunov level set $\Sigma_{s=0} = \{(v_{1i}, v_{2i}) \in \mathbb{R}^n : V \le k_{s=0}\}$

in finite time and stay in it in all future time,

where $k_{s=0} = \min_{s \in bd \ \Omega_{s=0}} V$, and

$$\Omega_{s=0} = \{ (v_{1i}, v_{2i}) \in \mathbb{R}^2 : |v_{1i}| \le 1, |v_{2i}| \le 1 \}$$

where $k_{s=0} > 0$ exist since the boundary of the set $\Omega_{s=0}$ is a compact set.

Note that $\Sigma_{s=0} \subset \Omega_{s=0}$. Now note that once inside this region, the system becomes

$$\dot{v}_{1i} = v_{2i},$$

 $\dot{v}_{2i} = -k_{2i} \operatorname{sign}(v_{2i}) - k_{1i} \operatorname{sign}(v_{1i})$

and the Lyapunov function becomes

$$V = \frac{1}{4} |v_{2i}|^4 + k_{1i}^2 |v_{1i}|^2 + k_{1i} |v_{1i}| |v_{2i}|^2 + r_1 |v_{1i}|^{\frac{3}{2}} \operatorname{sign}(v_{1i}) v_{2i}$$

Essentially, it will become homogenous double integrator system (also known as the twisting algorithm in the literature) and the Lyapunov function is homogeneous as well, where its finite time stability has been studied by the author in [133]. In essence, once the system enters the unsaturated Lyapunov level set in finite time, it will begin to behave like a twisting control that will converge to the origin in finite time (see [133] for the time of convergence estimation).

B.2 PROPOSITION 1

For v_{2i} , s_i , $\tilde{e}_{2i} \in \mathbb{R}$, k_{2i} , ε_{2i} , μ_i are positive constants and b > 0, the following inequalities are satisfied,

$$\begin{vmatrix} \operatorname{sat}_{\left(\frac{s_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} [v_{2i}] - \operatorname{sat}_{\left(\frac{s_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} [v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]] \end{vmatrix} \leq \operatorname{sat}_{2\left(\frac{s_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} [s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]], \\ \left| \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}] \right]^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i} + s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}]) \right] - \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} |^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i}) \right] \\ \leq \operatorname{sat}_{2\varepsilon_{2i}} \left[\beta \left(s_i + \operatorname{sat}_{\mu_i} [\widetilde{e}_{2i}] \right) \right] \end{vmatrix}$$

where β is a class \mathcal{K} function.

Proof of Proposition 1: Note that from Appendix A.2-proposition 1,

$$\begin{vmatrix} \operatorname{sat}_{\left(\frac{\tilde{\varepsilon}_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} \left[v_{2i} + s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right] - \operatorname{sat}_{\left(\frac{\tilde{\varepsilon}_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} \left[v_{2i} \right] \end{vmatrix}$$

$$\leq \max \begin{cases} \operatorname{sat}_{\left(\frac{\tilde{\varepsilon}_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} \left[v_{2i} \right] + \left| s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right] - \operatorname{sat}_{\left(\frac{\tilde{\varepsilon}_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} \left[\left| v_{2i} \right| \right] \\ \operatorname{sat}_{\left(\frac{\tilde{\varepsilon}_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} \left[\left| v_{2i} \right| \right] - \operatorname{sat}_{\left(\frac{\tilde{\varepsilon}_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}} \left[\left| v_{2i} \right| - \left| s_i + \operatorname{sat}_{\mu_i} \left[\widetilde{e}_{2i} \right] \right] \right] \end{cases}$$

Note that we only need to consider, for the right hand side of the above inequality, v_{2i} on the range

$$\begin{aligned} |v_{2i}| &\in \left[0, \left|s_i + \operatorname{sat}_{\mu_i}\left[\widetilde{e}_{2i}\right] + \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}\right] \text{ since for } |v_{2i}| \geq \left|s_i + \operatorname{sat}_{\mu_i^{\frac{1}{a}}}\left[\widetilde{e}_{2i}\right] + \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}\right] \\ \max \left\{ \begin{aligned} & \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}}\left[|v_{2i}| + \left|s_i + \operatorname{sat}_{\mu_i}\left[\widetilde{e}_{2i}\right]\right] - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}}\left[|v_{2i}|\right], \\ & \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}}\left[|v_{2i}| - \left|s_i + \operatorname{sat}_{\mu_i}\left[\widetilde{e}_{2i}\right]\right] \right] \end{aligned} \right\} = 0 \end{aligned}$$

Hence, for all $|v_{2i}| \ge 0$:

$$\max \begin{cases} \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+\delta}{2b}}} \left[|v_{2i}| + |s_{i} + \operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i}\right] \right] - \operatorname{sat}_{\gamma_{i}}\left[|v_{2i}| \right] \\ \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+\delta}{2b}}} \left[|v_{2i}| \right] - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+\delta}{2b}}} \left[|v_{2i}| - |s_{i} + \operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i}\right] \right] \\ = \max_{|v_{2i}| \in \left[0, |s_{i} + \operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i} \right] + \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+\delta}{2b}} \right]} \left\{ \begin{array}{l} \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+\delta}{2b}}} \left[|v_{2i}| + |s_{i} + \operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i} \right] \right] - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+\delta}{2b}}} \left[|v_{2i}| \right] \right\} \\ = \left\{ \begin{array}{l} \operatorname{max}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+\delta}{2b}}} \left[|v_{2i}| + \left| \frac{\varepsilon_{2i}}{k_{2i}} \right|^{\frac{1+\delta}{2b}} \left[|v_{2i}| \right] - \operatorname{sat}_{\left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+\delta}{2b}}} \left[|v_{2i}| - |s_{i} + \operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i} \right] \right] \right\} \\ = \left\{ \begin{array}{l} \left| s_{i} + \operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i} \right] \right], \text{ for } |s_{i} + \operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i} \right] \right] \leq 2 \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+\delta}{2b}} \\ 2 \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+\delta}{2b}}, \text{ for } |s_{i} + \operatorname{sat}_{\mu_{i}}\left[\widetilde{e}_{2i} \right] > 2 \left(\frac{\varepsilon_{2i}}{k_{2i}} \right)^{\frac{1+\delta}{2b}} \end{array} \right\}$$

Thus, the above proposition follows for all $|v_{2i}| \ge 0$ and $|s_i + \operatorname{sat}_{\mu_i}[\widetilde{e}_{2i}] \ge 0$.

Next, for the

$$\left| \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} + s_i |^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i} + s_i) \right] - \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} |^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i}) \right] \right|$$

Note that from Appendix A.2-proposition 1,

$$\begin{vmatrix} \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} + s_i |^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i} + s_i) \right] - \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} |^{\frac{2b}{1+b}} \operatorname{sign}(v_{2i}) \right] \\ \leq \max \begin{cases} \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} | + |s_i|^{\frac{2b}{1+b}} \right] - \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} |^{\frac{2b}{1+b}} \right], \\ \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} |^{\frac{2b}{1+b}} \right] - \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} |^{\frac{2b}{1+b}} \right], \end{cases} \end{cases}$$

Note that we only need to consider, for the right hand side of the above inequality, v_{2i} on the range

$$|v_{2i}| \in \left[0, |s_i| + \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}\right] \text{ since for } |v_{2i}| \ge |s_i| + \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2b}}$$
$$\max\left\{ \begin{aligned} & \text{sat}_{\varepsilon_{2i}} \left[k_{2i} ||v_{2i}| + |s_i||^{\frac{2b}{1+b}}\right] - \text{sat}_{\varepsilon_{2i}} \left[k_{2i} |v_{2i}|^{\frac{2b}{1+b}}\right], \\ & \text{sat}_{\varepsilon_{2i}} \left[k_{2i} |v_{2i}|^{\frac{2b}{1+b}}\right] - \text{sat}_{\varepsilon_{2i}} \left[k_{2i} ||v_{2i}| - |s_i|^{\frac{2b}{1+b}} \operatorname{sign}(|v_{2i}| - |s_i|)\right] \right\} = 0 \end{aligned}$$

Hence, for all $|v_{2i}| \ge 0$:

$$\max \begin{cases} \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} \| v_{2i} \| + \| s_i \|^{\frac{2b}{1+b}} \right] - \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} |^{\frac{2b}{1+b}} \right], \\ \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} | v_{2i} |^{\frac{2b}{1+b}} \right] - \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} \| v_{2i} \| - \| s_i \|^{\frac{2b}{1+b}} \operatorname{sign}(\| v_{2i} \| - \| s_i \|) \right] \end{cases} = \chi(|s_i|)$$

where the function $\chi(s_i)$ is defined as

$$\chi(|s_{i}|) := \max_{|v_{2i}| \in \left[0, |s_{i}| + \left(\frac{\varepsilon_{2i}}{k_{2i}}\right)^{\frac{1+b}{2}}\right] \left\{ \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} ||v_{2i}| + |s_{i}||^{\frac{2b}{1+b}}\right] - \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} ||v_{2i}|^{\frac{2b}{1+b}}\right], \\ \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} ||v_{2i}|^{\frac{2b}{1+b}}\right] - \operatorname{sat}_{\varepsilon_{2i}} \left[k_{2i} ||v_{2i}| - |s_{i}||^{\frac{2b}{1+b}}\operatorname{sign}(|v_{2i}| - |s_{i}|)\right] \right\}$$

which is a continuous nondecreasing function of $|s_i|$, zero at zero, and strictly positive. Also note that due to the saturation structure, we have

$$\chi(|s_i|) = 2\varepsilon_{2i} \text{ for } |s_i| > 2\varepsilon_{2i}$$

and due to the nondecreasing nature of the function, we have

$$\chi(|s_i|) \le 2\varepsilon_{2i} \text{ for } |s_i| \le 2\varepsilon_{2i}$$

Thus, for all $|v_{2i}| \ge 0$ and $|s_i| \ge 0$,

$$\left|\operatorname{sat}_{\varepsilon_{2i}}\left[k_{2i}|v_{2i}+s_{i}|^{\frac{2b}{1+b}}\operatorname{sign}(v_{2i}+s_{i})\right]-\operatorname{sat}_{\varepsilon_{2i}}\left[k_{2i}|v_{2i}|^{\frac{2b}{1+b}}\operatorname{sign}(v_{2i})\right]\right| \leq \operatorname{sat}_{2\varepsilon_{2i}}\left[\chi\left(|s_{i}|\right)\right]$$

Since the function $\chi(|s_i|)$ is zero at zero, continuous, strictly positive, and nondecreasing, from Lemma 1 of [145], there exists a class \mathcal{K} function $\beta(|s_i|)$ such that

$$\chi(|s_i|) \leq \beta(|s_i|), \forall |s_i| \geq 0$$

Hence,

$$\left|\operatorname{sat}_{\varepsilon_{2i}}\left[k_{2i}|v_{2i}+s_{i}|^{\frac{2b}{1+b}}\operatorname{sign}(v_{2i}+s_{i})\right]-\operatorname{sat}_{\varepsilon_{2i}}\left[k_{2i}|v_{2i}|^{\frac{2b}{1+b}}\operatorname{sign}(v_{2i})\right]\right| \leq \operatorname{sat}_{2\varepsilon_{2i}}\left[\chi\left(|s_{i}|\right)\right]$$
$$\leq \operatorname{sat}_{2\varepsilon_{2i}}\left[\beta\left(|s_{i}|\right)\right]$$
REFERENCES

- [1] A. Levant, "Principles of 2-sliding mode design," *Automatica*, vol. 43, no. 4, pp. 576-586, 2007.
- [2] A. F. Filippov, Differential equations with discontinuous righthand sides, Dordrecht: Kluwer, 1988.
- [3] G. Bartolini, A. Ferrara and E. Usai, "Chattering avoidance by second-order sliding mode control," *IEEE Transactions on automatic control*, vol. 43, no. 2, pp. 241-246, 1998.
- Y. B. Shtessel, I. A. Shkolnikov and M. D. Brown, "An asymptotic second-order smooth sliding mode control," *Asian Journal of Control*, vol. 5, no. 4, pp. 498-504, 2003.
- [5] E. Punta, "Second order sliding mode control of nonlinear multivariable systems," in 5th International Conference on Technology and Automation, Thessaloniki, Greece, 2005.
- [6] G. Bartolini, A. Ferrara, E. Usai and V. I. Utkin, "On multi-input chattering-free second-order sliding mode control," *IEEE Transactions on Automatic Control*, vol. 45, no. 9, pp. 1711-1717, 2000.

- [7] S. Yu, X. Yu, B. Shirinzadeh and Z. Man, "Continuous finite-time control for robotic manipulators with terminal sliding mode," *Automatica*, vol. 41, pp. 1957-1964, 2005.
- [8] K.-B. Park and T. Tsuji, "Terminal sliding mode control of second-order nonlinear uncertain systems," *International Journal of Robust and Nonlinear Control*, vol. 9, pp. 769-780, 1999.
- [9] S. Kamal, J. A. Moreno, A. Chalanga, B. Bandyopadhyay and L. M. Fridman, "Continuous terminal sliding-mode controller," *Automatica*, vol. 69, pp. 308-314, 2016.
- [10] C. Evangelista, P. Puleston, F. Valenciaga and L. M. Fridman, "Lyapunov-designed super twisting sliding mode control for wind energy conversion optimization," *IEEE transactions on industrial electronics*, vol. 60, no. 2, pp. 538-545, 2013.
- [11] J. Davila, L. Fridman and A. Levant, "Second-order sliding mode observer for mechanical systems," *IEEE Transactions on Automatic Control*, vol. 50, no. 11, pp. 1785-1789, 2005.
- [12] M. Van, H.-J. Kang and Y.-S. Suh, "Second order sliding mode-based output feedback tracking control for uncertain robot manipulators," *International Journal of Advanced Robotic Systems*, vol. 10, p. 9 pages, 2012.
- [13] Y. Hong, J. Huang and Y. Xu, "On an output feedback finite-time stabilization problem," *IEEE Transactions on Automatic Control*, vol. 46, no. 2, pp. 305-309, 2001.

- [14] J. Yang, Q. Wang and Y. Li, "An improved second order sliding mode twisting algorithm for finite-time trajectory tracking of intelligent vehicle," *Advances in Mechanical Engineering*, vol. 2013, p. 8 pages, 2013.
- [15] C. Vasquez, L. Fridman and J. Collado, "Second order sliding mode control of a 3dimensional overhead-crane," in 51st IEEE Conference on Decision and Control, Hawaii, 2012.
- [16] C. Vazquez, L. Fridman and J. Collado, "Generalized 'Twisting' controller of a parametrically excited crane," in *12th IEEE Workshop on Variable Structure Systems*, Mumbai, 2012.
- [17] J. A. Moreno, "Lyapunov approach for analysis and design of second order sliding mode algorithms," *Lecture notes in Control and Information Sciences*, vol. 412, pp. 113-150, 2011.
- [18] Y. Orlov, Y. Aoustin and C. Chevallereau, "Finite time stabilization of a perturbed double integrator-part i: Continuous sliding mode -based output feedback synthesis," *IEEE Transactions on Automatic Control*, vol. 56, no. 3, pp. 614-618, 2011.
- [19] A. Levant, "Homogeneity approach to high-order sliding mode design," *Automatica*, vol. 41, no. 5, pp. 823-830, 2005.
- [20] Y. Orlov, "Finite time stability and robust control synthesis of uncertain switched systems," *SIAM Journal on Control and Optimization*, vol. 43, no. 4, pp. 1253-1271,

2005.

- [21] H. G. Sage, M. F. De Mathelin and E. Ostertag, "Robust control of robot manipulators:a survey," *International Journal of Control*, vol. 72, no. 16, pp. 1498-1522, 1999.
- [22] S. P. Bhat and D. S. Bernstein, "Finite-time stability of continuous autonomous systems," *SIAM Journal of Control and Optimization*, vol. 38, no. 3, pp. 751-766, 2000.
- [23] S. Yu, X. Yu and R. Stonier, "Continuous finite-time control for robotic manipulators with terminal sliding modes," in *Proceedings of the Sixth International Conference of Information Fusion*, Cairns, Queensland, 2003.
- [24] Y. Hong, Y. Xu and J. Huang, "Finite-time control for robot manipulators," Systems & Control Letters, vol. 46, no. 4, pp. 243-253, 2002.
- [25] J. Kasac, B. Novakovic, D. Majetic and D. Brezak, "Global positioning of robot manipulators with mixed revolute and prismatic joints," *IEEE Transactions on Automatic Control*, vol. 51, no. 6, pp. 1035-1040, 2006.
- [26] J. A. Moreno and M. Osorio, "A Lyapunov approach to second-order sliding mode controllers and observers," in *Proceedings of the 47th IEEE Conference on Decision* and Control, Mexico, 2008.
- [27] J. A. Moreno and M. Osorio, "Strict lyapunov functions for the super-twisting algorithm," *IEEE Transactions on Automatic Control*, vol. 57, no. 4, pp. 1035-1040,

- [28] J. A. Moreno, "On strict Lyapunov functions for some non-homogeneous super-twisting algorithms," *Journal of the Franklin Institute*, vol. 351, no. 4, pp. 1902-1919, 2014.
- [29] A. Levant, "Sliding order and sliding accuracy in sliding mode control," *International Journal of Control*, vol. 58, no. 6, pp. 1247-1263, 1993.
- [30] H. B. Oza, Y. V. Orlov and S. K. Spurgeon, "Settling time estimate for second order sliding mode controller: A homogeneity approach," in *Proceedings of the 18th IFAC World Congress*, Milano, Italy, 2011.
- [31] H. B. Oza, Y. V. Orlov and S. K. Spurgeon, "Tuning rules for second order sliding mode based output feedback synthesis," in 12th IEEE Workshop on Variable Structure Systems, Mumbai, 2012.
- [32] V. Utkin, "About second order sliding mode control, relative degree, finite-time convergence and disturbance rejection," in 11th International Workshop on Variable Structure Systems, Mexico, 2010.
- [33] A. Polykov and A. Poznyak, "Lyapunov function design for finite-time convergence analysis:"Twisting" controller for second order sliding mode realization," *Automatica*, vol. 45, pp. 444-448, 2009.
- [34] R. Santiesteban, L. Fridman and J. Moreno, "Finite time convergence analysis for

"twisting" controller via a strict Lyapubov function," in *Proceedings of 11th Workshop* on Variable Structure Systems, Mexico City, 2010.

- [35] A. Estrada, A. Loria, R. Santiesteban and L. Fridman, "Lyapunov stability analysis of a twisting based control algorithm for systems with unmatched perturbations," in *50th IEEE Conference on Decision and Control and European Control Conference*, Florida, 2011.
- [36] V. T. Haimo, "Finite time controllers," *SIAM Journal of Control and Optimization*, vol. 24, no. 4, pp. 760-770, 1986.
- [37] R. SanTiesteban, "Time convergence estimation of a perturbed double integrator: Family of continuous sliding mode based output feedback synthesis," in *Proceedings of the European Control Conference 2013*, Zurich, 2013.
- [38] R. Santiesteban, A. Garate-Garcia and R. Bautista, "A family of continuous state feedback synthesis: Lyapunov Approach," in *Congreso Nacional de Control Automatico*, Mexico, 2013.
- [39] Y. Hong, "Finite-time stabilization and stabilizability of a class of controllable systems," *Systems and Control Letters,* vol. 46, no. 4, pp. 231-236, 2002.
- [40] E. Bernuau, W. Perruquetti, D. Efimov and E. Moulay, "Finite-time output stabilization of the double integrator," in *51st IEEE Conference on Decision and Control*, Hawaii, 2012.

- [41] S. P. Bhat and D. S. Bernstein, "Geometric homogeneity with applications to finite-time stability," *Mathematics of Control, Signals and Systems*, vol. 17, no. 2, pp. 101-127, 2005.
- [42] J.-J. E. Slotine and W. Li, Applied nonlinear control, New Jersey: Prentice-Hall, Inc., 1991, pp. 74-75.
- [43] H. B. Oza, Y. V. Orlov and S. K. Spurgeon, "Robust finite time stability and stabilisation: a survey of continuous and discontinuous paradigms," in 13th IEEE Workshop on Variable Structure Systems, Nantes, 2014.
- [44] H. B. Oza, Y. V. Orlov and S. K. Spurgeon, "On robustness of a class of homogeneous continuous finite time controllers," in *IEEE Conference on Decision and Control*, Los Angeles, 2014.
- [45] A. Polyakov, Y. Orlov, H. Oza and S. Spurgeon, "Robust finite-time stabilization and observation of a planar system revisited," in *IEEE Annual Conference on Decision and Control*, Osaka, 2015.
- [46] H. B. Oza, Y. V. Orlov and S. K. Spurgeon, "Continuous uniform finite time stabilization of planar controllable systems," *SIAM Journal on Control and Optimization*, vol. 53, no. 3, pp. 1154-1181, 2015.
- [47] E. Cruz-Zavala, J. A. Moreno and L. Fridman, "Fast second-order sliding mode control design based on Lyapunov function," in *52nd IEEE Conference on Decision and*

Control, Florence, 2013.

- [48] S. P. Bhat and D. S. Bernstein, "Lyapunov analysis of finite-time differential equations," in *Proceedings of the American Control Conference*, Washington, 1995.
- [49] T. Sanchez and J. A. Moreno, "Construction of Lyapunov functions for a class of higher order sliding modes algorithms," in *51st IEEE Conference on Desicion and Control*, Hawaii, 2012.
- [50] A. Polyakov and A. Poznyak, "Unified Lyapunov function for a finite-time stability analysis of relay second-order sliding mode control systems," IMA Journal of Mathematical Control and Information, page doi: 10.1093/imamci/dns007, 2012.
- [51] A. S. Poznyak, A. Y. Polyakov and V. V. Strygin, "Analysis of finite-time convergence by the method of Lyapunov functions in systems with second-order sliding modes," *Journal of Applied Mathematics and Mechanics*, vol. 75, no. 3, pp. 289-303, 2011.
- [52] T. Sanchez and J. A. Moreno, "Lyapunov functions for twisting and terminal controllers," in *13th IEEE Workshop on Variable Structure Systems*, Nantes, 2014.
- [53] E. Bernuau, D. Efimov, W. Perruquetti and A. Polyakov, "On homogeneity and its application in sliding mode control," *Journal of the Franklin Institute*, vol. 351, no. 4, pp. 1866-1901, 2014.
- [54] M. Harmouche, S. Laghrouche and Y. Chitour, "Stabilization of perturbed integrator

chains using Lyapunov-Based Homogeneous Controllers," ArXiv e-prints, page : http://adsabs.harvard.edu/abs/2013arXiv1303.5330H, 2013.

- [55] A. Levant, "Universal single-input-single-output (SISO) sliding-mode controllers with finite-time convergence," *IEEE Transactions on Automatic Control*, vol. 46, no. 9, pp. 1447-1451, 2001.
- [56] L. Rosier, "Homogeneous Lyapunov function for homogeneous continuous vector field," Systems and Control Letters, vol. 19, no. 6, pp. 467-473, 1992.
- [57] S. P. Bhat and D. S. Bernstein, "Continuous finite-time stabilization of the translational and rotational double integrators," *IEEE Transactions on automatic control*, vol. 43, no. 5, pp. 678-682, 1998.
- [58] H. Nakamura, "Positive definiteness of generalized homogeneous functions," in 9th IFAC symposium on Nonlinear Control Systems, France, 2013.
- [59] E. Cruz-Zavala, J. A. Moreno and L. Fridman, "Second-order uniform exact sliding mode control with uniform sliding surface," in *IEEE Conference on Decision and Control and European Control Conference*, Florida, 2011.
- [60] H.-x. Zhang, J.-s. Fan, F. Meng and J.-f. Huang, "A new double power reaching law for sliding mode control," *Control and Decision*, vol. 28, no. 2, pp. 289-293, 2013.
- [61] A. Polyakov, "Nonlinear feedback design for fixed-time stabilization of linear control

systems," *IEEE Transactions on Automatic Control*, vol. 57, no. 8, pp. 2106-2110, 2012.

- [62] M. Basin, C. B. Panathula and Y. Shtessel, "Continuous second-order sliding mode control: convergence time estimation," in *IEEE 54th Annual Conference on Decision* and Control, Osaka, 2015.
- [63] E. Cruz-Zavala, J. A. Moreno and L. M. Fridman, "Uniform robust exact differentiator," *IEEE Transactions on Automatic Control*, vol. 56, no. 11, pp. 2727-2733, 2011.
- [64] P. Li and Z. -Q. Zheng, "Robust adaptive second-order sliding mode control with fast transient performance," *IET Control Theory and Applications*, vol. 6, no. 2, pp. 305-312, 2012.
- [65] J. A. Moreno, "A linear framework for the robust stability analysis of a generalized super-twisting algorithm," in *International Conference on Electrical Engineering, Computing Science and Automatic Control*, Toluca, 2009.
- [66] V. Utkin, "Mechanical energy-based Lyapunov function for twisting and super-twisting sliding mode control," *IMA Journal of Mathematical Control and Information*, 2014.
- [67] A. Barth, M. Reichhartinger, J. Reger, M. Horn and K. Wulff, "Certainty-equivalence based super-twisting control using continuous adaptation laws," in 2016 14th International Workshop on Variable Structure Systems (VSS), Nanjing, 2016.

- [68] J. A. Moreno, T. Sanchez and E. Cruz, "Una funcion de lyapunov suave para algoritmo super-twisting," in *16th CLCA*, Mexico, 2014.
- [69] T. Sanchez and J. A. Moreno, "A constructive Lyapunov function design method for a class of homogeneous systems," in *IEEE Conference on Decision and Control*, Los Angeles, 2014.
- [70] D. Shevitz and B. Paden, "Lyapunov stability theory of nonsmooth systems," *IEEE Transactions on Automatic Control*, vol. 39, no. 9, pp. 1910-1914, 1994.
- [71] H. K. Khalil, Nonlinear Systems, Third ed., New Jersey: Prentice-Hall, 2002.
- [72] M. Forti, M. Grazzini, P. Nistri and L. Pancioni, "Generalized Lyapunov approach for convergence of neural networks with discontinuous or non-lipschitz activations," *Physica D: Nonlinear Phenomena*, vol. 214, no. 1, pp. 88-99, 2006.
- [73] B. E. Paden and S. S. Sastry, "A calculus for computing Filippov's differential inclusion with application to the variable structure control of robot manipulators," *IEEE Transactions on Circuits and Systems*, vol. 34, no. 1, pp. 73-82, 1987.
- [74] A. Bacciotti, R. Conti and P. Marcellini, "Discontinuous ordinary differential equations and stabilization," Universita di Firenze, 2000.
- [75] J. Shi, H. Liu and N. Bajcinca, "Robust control of robotic manipulators based on integral sliding mode," *International Journal of Control*, vol. 81, no. 10, pp. 1537-1548,

2008.

- [76] C.-L. Chen, T.-C. Wu and C.-C. Peng, "Robust trajectories following control of a 2-link robot manipulator via coordinate transformation for manufacturing applications," *Robotics and Computer-Integrated Manufacturing*, vol. 27, no. 3, pp. 569-580, 2011.
- [77] C.-L. Chen and R.-L. Xu, "Tracking control of robot manipulators using sliding mode controller with performance robustness," *Journal of Dynamic Systems, Measurement, and Control,* vol. 121, no. 1, pp. 64-70, 1999.
- [78] M. Agwan, S. Janardhanan and L. Dewan, "A robust higher order sliding mode control of rigid robot manipulator," in *Proceedings of 2012 International Conference on Modelling, Identification and Control*, Wuhan, 2012.
- [79] K.-B. Park, S.-W. Kim and J.-J. Lee, "Smooth variable structure controller for robot manipulator using virtual plant," *Electronics Letters*, vol. 31, no. 24, pp. 2134-2136, 1995.
- [80] K.-B. Park and J.-J. Lee, "Variable structure model following control for robot manipulator using time-varying sliding hyperplane," in *in Proceedings of Korea Automation Control Conference International Sessions (KACC '93)*, Seoul, 1993.
- [81] V. Utkin and J. Shi, "Integral sliding mode in systems operating under uncertainty conditions," in *Proceedings of the 35th Conference on Decision and Control*, Kobe, 1996.

- [82] J.-X. Xu, Y.-J. Pan and T.-H. Lee, "Analysis and design of integral sliding mode control based on Lyapunov's direct method," in *Proceedings of the 2003 American Control Conference*, Colorado, 2003.
- [83] J. Geng, Y. Sheng and X. Liu, "Time-varying nonsingular terminal sliding mode control for robot manipulators," 6 December 2013. [Online]. Available: DOI: 10.1177/0142331213512367.
- [84] K. J. Binoo and G. Ray, "Trajectory tracking of a two-link robot manipulator: a terminal attractor approach," in 6th International Conference on Electrical and Computer Engineering, Dhaka, 2010.
- [85] Z. Qin, X. He and D. Zhang, "Nonsingular and fast convergent terminal sliding mode control of robotic manipulators," in *Proceedings of the 30th Chinese Control Conference*, Yantai, 2011.
- [86] J.-J. Lee, J.-H. Lee, J.-S. Ko and M.-J. Youn, "Design of efficient sliding mode controller for robot manipulator using disturbance observer," in *Robotics and Automation, 1992. Proceedings., 1992 IEEE International Conference on*, Nice, 1992.
- [87] D. K. Han and P.-h. Chang, "Robust tracking of robot manipulator with nonlinear friction using time delay control with gradient estimator," *Journal of Mechanical Science and Technology*, vol. 24, no. 8, pp. 1743-1752, 2010.

- [88] M. Jin, S. H. Kang and P. H. Chang, "Simple effective control for robot manipulators with friction," in *Robot Manipulators*, InTech, 2008, pp. 225-242.
- [89] T. C. Hsia and L. S. Gao, "Robot manipulator control using decentralized linear timeinvariant time-delayed joint controllers," in *Robotics and Automation*, 1990. *Proceedings.*, 1990 IEEE International Conference on, Cincinnati, 1990.
- [90] J. P. Kolhe, M. Shaheed, T. S. Chandar and S. E. Talole, "Robust control of robot manipulators based on uncertainty and disturbance estimation," *International Journal of Robust and Nonlinear Control*, vol. 23, no. 1, pp. 104-122, 2013.
- [91] S. E. Talole and S. B. Phadke, "Robust input-output linearisation using uncertainty and disturbance estimation," *International Journal of Control*, vol. 82, no. 10, pp. 1794-1803, 2009.
- [92] M. Jin, J. Lee, P. H. Chang and C. Choi, "Practical nonsingular terminal sliding-mode control of robot manipualtors for high-accuracy tracking control," *IEEE Transactions* on *Industrial Electronics*, vol. 56, no. 9, pp. 3593-3601, 2009.
- [93] M. Jin, Y. Jin, P. H. Chang and C. Choi, "High-accuracy tracking control of robot manipulators using time delay estimation and terminal sliding mode," *International Journal of Advanced Robotic Systems*, vol. 8, no. 4, pp. 65-78, 2011.
- [94] P. H. Chang and S. H. Park, "On improving time-delay control under certain hard

nonlinearities," Mechatronics, vol. 13, no. 4, pp. 393-412, 2003.

- [95] Q.-C. Zhong and D. Rees, "Control of uncertain LTI systems based on an uncertainty and disturbance estimator," *Journal of Dynamic Systems, Measurements, and Control,* vol. 126, no. 4, pp. 905-910, 2004.
- [96] K. Youcef-Toumi and O. Ito, "A time delay controller for systems with unknown dynamics," in *American Control Conference*, Atlanta, 1988.
- [97] P. H. Chang and J. H. Jung, "A systematic method for gain selection of robust pid control for nonlinear plants of second-order controller canonical form," *IEEE Transactions on Control Systems Technology*, vol. 17, no. 2, pp. 473-483, 2009.
- [98] J. Alvarez-Ramirez, I. Cervantes and R. Kelly, "PID regulation of robot manipulators: stability and performance," *Systems and Control Letters*, vol. 41, no. 2, pp. 73-83, 2000.
- [99] I. Cervantes and J. Alvarez-Ramirez, "On the PID tracking control of robot manipulators," *Systems & Control Letters*, vol. 42, no. 1, pp. 37-46, 2001.
- [100] Y. Karayiannidis and Z. Doulgeri, "Model-free robot joint position regulation and tracking with prescribed performance guarantees," *Robotics and Autonomous Systems*, vol. 60, no. 2, pp. 214-226, 2012.
- [101] C. Bechlioulis, Z. Doulgeri and G. Rovithakis, "Robot force\position tracking with guaranteed prescribed performance," in *IEEE International Conference on Robotics and*

Automation, Kobe, 2009.

- [102] B. Yao and M. Tomizuka, "Smooth robust adaptive sliding mode control of manipulators with guaranteed transient performance," in *Proceedings of the American Control Conference*, Baltimore, 1994.
- [103] A. Chaillet, A. Loria and R. Kelly, "Robustness of PID-controlled manipulators vis-avis actuator dynamics and external disturbances," *European Journal of Control*, vol. 13, no. 6, pp. 563-576, 2007.
- [104] J. L. Meza, V. Santibanez and R. Campa, "An estimate of the domain of attraction for the PID regulator of manipulators," *International Journal of Robotics and Automation*, vol. 22, no. 3, pp. 187-195, 2007.
- [105] V. M. Hernandez-Guzman, V. Santibanez and R. Silva-Ortigoza, "A new tuning procedure for PID control of rigid robots," *Advanced Robotics*, vol. 22, no. 9, pp. 1007-1023, 2008.
- [106] X. Mo and Q. Lan, "Finite-time integral sliding mode control for motion control of permanent-magnet linear motors," *Mathematical Problems in Engineering*, Vols. 2013, Article ID 567610, 7 pages, 2013.
- [107] Y. Su and C. Zheng, "A simple nonlinear PID control for global finite-time regulation of robot manipulators without velocity measurements," in 2010 IEEE International Conference on Robotics and Automaition Anchorage Convention District,

Anchorage, Alaska, 2010.

- [108] M. Mendoza, A. Zavala-Rio, V. Santibanez and F. Reyes, "Output-feedback proportional-integral-derivative-type ontrol with simple tuning for the global regulation of robot manipulators with input constraints," *IET Control Theory & Applications*, vol. 9, no. 14, pp. 2097-2106, 2015.
- [109] A. Zavala-Rio, E. Aguinaga-Ruiz and V. Santibanez, "Global trajectory tracking through output feedback for robot manipulators with bounded inputs," *Asian Journal of Control*, vol. 13, no. 3, pp. 430-438, 2011.
- [110] R. Kelly, R. Ortega, A. Ailon and A. Loria, "Global regulation of flexible joint robots using approximate differentiation," *IEEE Transactions on Automatic Control*, vol. 39, no. 6, pp. 1222-1224, 1994.
- [111] V. Utkin, "On convergence time and disturbance rejection of super-twisting control," *IEEE Transactions on Automatic Control*, vol. 58, no. 8, pp. 2013-2017, 2013.
- [112] Y. Yang and S. Qin, "A new modified super-twisting algorithm with double closed-loop feedback regulation," *Transactions of the Institute of Measurement and Control*, pp. 1-10, 2016.
- [113] E. Aguinaga-Ruiz, A. Zavala-Rio, V. Santibanez and F. Reyes, "Global trajectory tracking through static feedback for robot manipulators with bounded inputs," *IEEE Transactions on Control Systems Technology*, vol. 17, no. 4, pp. 934-944, 2009.

- [114] Y. Su and J. Swevers, "Finite-time tracking control for robot manipulators with actuator saturation," *Robotics and Computer-Integrated Manufacturing*, vol. 30, no. 2, pp. 91-98, 2014.
- [115] Y. Su and J. Swevers, "A new asymptotic tracking approach for robot manipulators with actuator saturation," in *IEEE International Conference on Mechatronics*, Vicenza, 2013.
- [116] F. Morabito, A. R. Teel and L. Zaccarian, "Nonlinear antiwindup applied to Euler-Langrange systems," *IEEE Transactions on Robotics and Automation*, vol. 20, no. 3, pp. 526-537, 2004.
- [117] F. L. Lewis, D. M. Dawson and C. T. Abdallah, Robot Manipulator Control: Theory and Practice, 2004: Marcel Dekker, 2004.
- [118] V. Santibanez and R. Kelly, "Global asymptotic stability of bounded output feedback tracking control for robot manipulators," in 40th IEEE Conference on Decision and Control, Orlando, FL, 2001.
- [119] A. Loria and H. Nijmeijer, "Bounded output feedback tracking control of fully-actuated euler-lagrange systems," in *35th Conference on Decision and Control*, Kobe, 1996.
- [120] J. Alvarez-Ramirez, R. Kelly and I. Cervantes, "Semiglobal stability of saturated linear PID control for robot manipulators," *Automatica*, vol. 39, no. 6, pp. 989-995, 2003.

- [121] V. Santibanez and R. Kelly, "A class of nonlinear PID global regulators for robot manipulators," in *IEEE International Conference on Robotics and Automation*, Leuven, 1998.
- [122] Y. Su, P. C. Muller and C. Zheng, "Global asymptotic saturated PID control for robot manipulators," *IEEE Transactions on Control Systems Technology*, vol. 18, no. 6, pp. 1280-1288, 2010.
- [123] R. Kelly, "Global positioning of robot manipulators via PD control plus a class of nonlinear integral actions," *IEEE Transactions on Automatic Control*, vol. 43, no. 7, pp. 934-938, 1998.
- [124] Y. Su, P. C. Muller and C. Zheng, "A global asymptotic stable output feedback PID regulator for robot manipulators," in *IEEE International Conference on Robotics and Automation*, Roma, 2007.
- [125] V. Santibanez, K. Camarillo, J. Moreno-Valenzuela and R. Campa, "A practical PID regulator with bounded torques for robot manipulators," *International Journal of Control, Automation, and Systems,* vol. 8, no. 3, pp. 544-555, 2010.
- [126] S. Seshagiri and H. K. Khalil, "Robust output feedback regulation of minimum-phase nonlinear systems using conditional integrators," *Automatica*, vol. 41, no. 1, pp. 43-54, 2005.

- [127] S. Seshagiri and H. K. Khalil, "Universal integral controllers with anti-reset windup for minimum phase nonlinear systems," in *Proceedings of the 40th IEEE Conference on Decision and Control*, Orlando, 2001.
- [128] H. Berghuis and H. Nijmeijer, "Global regulation of robots using only position measurements," Systems & Control Letters, vol. 21, no. 4, pp. 289-293, 1993.
- [129] Y. Su and V. Parra-Vega, "Global asymptotic saturated output feedback control of robot manipulators," in 7th World Congress on Intelligent Control and Automation, Chongqing, 2008.
- [130] A. Zavala-Rio, M. Mendoza, V. Santibanez and F. Reyes, "An output-feedback PIDtype global regulator for robot manipulators with bounded inputs," in *11th IFAC Symposium on Robot Control SYROCO*, Salvador, 2015.
- [131] L. B. Freidovic and H. K. Khalil, "Universal integral controllers for robotic manipulators," in *Proceedings of 5th Symposium on Nonlinear Control Systems*, St. Petersburg, Russia, 2001.
- [132] D. J. Lopez-Araujo, A. Zavala-Rio, V. Santibanez and F. Reyes, "Output-feedback adaptive control for the global regulation of robot manipulators with bounded inputs," *International Journal of Control, Automation, and Systems*, vol. 11, no. 1, pp. 105-115, 2013.

- [133] W. Jang, H. Almurib and T. Nandha Kumar, "Lyapunov method for unperturbed double integrator systems," in *ICARCV*, Singapore, 2014.
- [134] W. Jang, H. Almurib and T. Nandha Kumar, "A lyapunov method analysis for double integrator with bounded perturbation," in *ICARCV*, Singapore, 2014.
- [135] T. Nakakuki, T. Shen and K. Tamura, "robust tracking control for robot systems with discontinuous uncertainty: an approach based on Filippov's framework," in *Electrical Engineering in Japan*, 2008.
- [136] D. S. Mitrinovic, Analytic Inequalities. In cooperation with P.M. Vasic, Berlin: Springer-Verlag, 1970.
- [137] F. H. Clarke, Optimization and nonsmooth analysis, New York: John Wiley & Sons, 1990.
- [138] N. Fischer, R. Kamalapurkar and W. E. Dixon, "LaSalle-Yoshikawa corollaries for nonsmooth systems," *IEEE Transactions on Automatic Control*, vol. 58, no. 9, pp. 2333-2338, 2013.
- [139] G. Cheng, X. Mu and Z. Ding, "Uniformly ultimate boundedness for a class of discontinuous nonautonomous systems (in Chinese)," ACTA Mathematica Applicatae Sinica, vol. 30, no. 4, pp. 675-681, 2007.
- [140] W. Heemels and S. Weiland, "Input-to-state stability and interconnections of

discontinuous dynamical systems," Automatica, vol. 44, no. 12, pp. 3079-3086, 2008.

- [141] A. Barth, M. Reichhartinger, J. Reger, M. Horn and K. Wulff, "Certainty-equivalence based super-twisting control using continuous adaptation laws," in 2016 14th International Workshop on Variable Structure Systems (VSS), Nanjing, 2016.
- [142] R. Kelly, V. Santibanez and A. Loria, Control of robot manipulators in joint space, London: Springer-Verlag, 2005.
- [143] A. Bacciotti and F. Ceragioli, "Stability and stabilization of discontinuous dystems and nonsmooth lyapunov functions," *ESAIM: Control Optimisation and Calculus of Variations*, vol. 4, pp. 361-376, 1999.
- [144] E. Aguinaga-Ruiz, A. Zavala-Rio, V. Santibanez and F. Reyes, "Global trajectory tracking through static feedback for robot manipulators with input saturations," in 47th IEEE Conference on Decision and Control, Mexico, 2008.
- [145] C. M. Kellett, "A compendium of comparison function results," *Mathmetical Control Signal Systems*, vol. 26, no. 3, pp. 339-374, 2014.
- [146] J. A. Moreno and M. Osorio, "A Lyapunov approach to second-order sliding mode controllers and observers," in *Proceedings of the 47th IEEE Conference on Decision* and Control, Mexico, 2008.
- [147] A. Barth, M. Reichhartinger, J. Reger, M. Horn and K. Wulff, "Certainty-equivalene

based super-twisting control using continuous adaptation laws," in 2016 14th International Workshop on Variable Structure Systems (VSS), Nanjing, 2016.

[148] C. Qian and W. Lin, "Non-Lipschitz continuous stabilizers for nonlinear systems with uncontrollable unstable linearization," *Systems & Control Letters*, vol. 42, no. 3, pp. 185-200, 2001.